Energy Correlators, Dispersive Sum Rules, and Modular Bootstrap in Conformal Field Theories

Thesis by Cyuan-Han Chang

In Partial Fulfillment of the Requirements for the Degree of Doctor of Philosophy

Caltech

CALIFORNIA INSTITUTE OF TECHNOLOGY Pasadena, California

> 2024 Defended April 18, 2024

© 2024

Cyuan-Han Chang ORCID: 0000-0002-7133-3553

All rights reserved

ACKNOWLEDGEMENTS

This thesis is a collection of works done during my graduate study. It could not have been possible without the support of many people. There are too many people to thank, and I apologize if I fail to mention some names. I am grateful to everyone I have interacted with throughout my Ph.D. journey.

First, I would like to thank my advisor, Prof. David Simmons-Duffin. I cannot count the number of times I was amazed by his unbounded knowledge and his ability to tackle a seemingly unsolvable problem piece by piece and find the solution. From him, I have learned how to properly do research and how important it is to fully understand the big picture of the problem and ask the important questions while at the same time being able to study them in great depth and find ways to carry out the technically challenging computation. I am also grateful to him for the freedom in research he allowed me to have and for always being supportive whenever I needed help. It has been a great honor to have David as my advisor, and I cannot imagine having a better one. Thank you to my committee members, Prof. Anton Kapustin, Prof. Olexei Motrunich, and Prof. Hirosi Ooguri, for many inspiring conversations through seminars and courses.

Besides the mentorship from my advisor, I also benefited significantly from working with postdocs. I am indebted Nathan Benjamin, whom I have probably spent the most time working with during the second half of my Ph.D. I thank him for introducing me to the world of 2d CFTs and for many conversations about physics, life, and everything. Additionally, I am grateful for his enormous help and support when I was applying for postdocs. I also want to thank Julio Parra-Martinez, whom I started talking to during COVID. Thanks to him for teaching me so many things about amplitudes and QFT. Those conversations helped broaden my horizons, and it has been a pleasure to be able to work with Julio. I had the chance to collaborate with Alexander Zhiboedov, Petr Kravchuk, and Murat Kologlu for my first project as a graduate student. During the early stage of my PhD, it significantly deepened my understanding of the field, and I greatly appreciate that.

I am very fortunate to be able to discuss physics with all the fantastic people in the Caltech conformal group, including Ying-Hsuan Lin, Yuya Kusuki, Sridip Pal, Ning Su, Tzu-Chen Huang, Junyu Liu, Yakov Landau, Aike Liu, Jaeha Lee, and Yixin Xu. I additionally thank Sridip Pal for always asking me to go to Starbucks, and Yuya Kusuki for teaching me the importance of having a chain. I also want to thank all the graduate students on the fourth floor for all the fun we had together at the grad socials and birthday events.

Many thanks to Haw-Wei Lin for always being so helpful and being an awesome roommate, and to Chun-Ju Wu, Hsin-Yuan Huang, Amy Huang, and Chi-Fang Chen for the board games and hotpot on holidays. I am also grateful to all the friends from ACT and FT Dodgers for their friendship, including Benjamin Chang, Steve Lu, Po-Hsuan Lin, Sean Hu, Wen Chao, Jacky Yang, Ying-Hsuan Lin, Yu-Li Ni, and Hsiao-Yi Chen.

Finally, I would like to thank my parents, brother, and sister for their love and support. This thesis is dedicated to them.

I am also thankful for the support from the Walter Burke Institute for Theoretical Physics. This thesis is based upon work supported by the U.S. Department of Energy, Office of Science, Office of High Energy Physics, under Award Number DE-SC0011632.

ABSTRACT

Studying conformal field theories (CFTs) is an important topic in theoretical physics. CFTs describe critical phenomena, fixed points of RG flows, and UV-complete descriptions of quantum gravity. One powerful method to study CFTs is the bootstrap approach. The idea of bootstrap is to impose symmetries and basic consistency conditions and constrain the space of possible theories. In this thesis, we study constraints and observables in CFTs that come from putting the CFT in Lorentzian signature or on a torus.

In chapter 1, we give a brief overview of CFT and the idea of bootstrap. We also give a short summary of the rest of the thesis.

In chapter 2, we consider a product of two null-integrated operators on the same null plane in a CFT. We call the matrix elements of this product (two-point) energy correlators or event shapes. Energy correlators measure the distribution of energy flux at future null infinity, and they are important in real-world collider measurements. We show that the product admits an operator product expansion (OPE) in the direction transverse to the null plane. Such OPE is called the light-ray OPE. The light-ray OPE between two null-integrated stress-energy tensors contains light-ray operators at spin $J = 3, 5, 7, 9, \ldots$ The terms with spin J = 3 + 2n have transverse spin 4 + 2n, and are constructed using special conformally-invariant differential operators.

In chapter 3, we study three-point energy correlators (EEEC). We show that Lorentz symmetry implies that EEEC can be decomposed into "celestial blocks", which are functions completely fixed by symmetry. In the collinear limit where all three null-integrated operators approach each other in the transverse direction, the celestial block reduces to a four-point conformal block, with the position of the fourth point given by the momentum of the external state. Using the celestial block decomposition and leading order results for the EEEC in the collinear limit, we make all-order predictions for the EEEC in certain kinematic limits, in both weakly-coupled N = 4 super Yang-Mills (SYM) theory and QCD. We also derive a celestial inversion formula that computes the coefficients of the celestial block expansion. We then apply the formula to the EEEC in strongly-coupled N = 4 SYM and obtain a complete celestial block decomposition.

In chapter 4, we use commutativity of null-integrated operators to generate a class of dispersive CFT sum rules for general spinning operators. Dispersive sum rules have

properties that their action on conformal blocks at the double-trace locations vanish, and they are useful for studying holographic CFTs. Our definition of the spinning dispersive sum rule is an integral of the four-point function over spacetime against a kernel. We show that the action on a conformal block with heavy dimensions can be computed by a saddle point approximation of this integral. Furthermore, the heavy action has a nice interpretation in terms of the flat space limit of the bulk scattering process. This allows us to build a dictionary between spinning dispersive CFT sum rules and flat space sum rules for the scattering of photons and gravitons.

In chapter 5, we use harmonic analysis to decompose the torus partition of 2d CFTs with $U(1)^c$ symmetry into a basis that is manifestly modular-invariant. From this decomposition, we can derive a crossing equation that acts only on the scalars. The crossing equation has interesting terms that are related to the zeros of the Riemann zeta function, and we can rephrase the Riemann hypothesis as a statement about the asymptotic density of scalar operators in $U(1)^c$ 2d CFTs. Finally, we use the scalar crossing equation to obtain bounds on the dimension of the lightest scalar operator.

PUBLISHED CONTENT AND CONTRIBUTIONS

- Cyuan-Han Chang, Murat Kologlu, Petr Kravchuk, David Simmons-Duffin, and Alexander Zhiboedov. "Transverse spin in the light-ray OPE". In: *JHEP* 05 (2022), p. 059. DOI: 10.1007/JHEP05(2022)059. arXiv: 2010.04726 [hep-th]. All authors contributed equally.
- [2] Cyuan-Han Chang and David Simmons-Duffin. "Three-point energy correlators and the celestial block expansion". In: JHEP 02 (2023), p. 126. DOI: 10.1007 / JHEP02(2023) 126. arXiv: 2202.04090 [hep-th].

All authors contributed equally.

- [3] Cyuan-Han Chang, Yakov Landau, and David Simmons-Duffin. "Spinning dispersive CFT sum rules and bulk scattering". In: (Nov. 2023). arXiv: 2311.04271 [hep-th]. All authors contributed equally.
- [4] Nathan Benjamin and Cyuan-Han Chang. "Scalar modular bootstrap and zeros of the Riemann zeta function". In: *JHEP* 11 (2022), p. 143.
 DOI: 10.1007/JHEP11(2022)143. arXiv: 2208.02259 [hep-th]. All authors contributed equally.

TABLE OF CONTENTS

Acknowledgements	. iii
Abstract	. v
Published Content and Contributions	. vii
Table of Contents	. vii
List of Illustrations	. X
List of Tables	. xiv
Chapter I: Introduction	. 1
1.1 Conformal collider physics	. 4
1.2 Dispersive sum rules	. 5
1.3 Scalar modular bootstrap	. 7
Chapter II: Transverse spin in the light-ray OPE	. 8
2.1 Introduction	. 8
2.2 Event shapes, OPEs, and transverse spin	. 11
2.3 Building transverse spin with differential operators	. 20
2.4 The complete OPE of scalar detectors	. 27
2.5 The complete OPE of general detectors	. 44
2.6 Examples	. 67
2.7 Example: event shape in $\mathcal{N} = 4$ SYM	. 75
2.8 Discussion and future directions	. 88
Chapter III: Three-point energy correlators and the celestial block expansion	. 91
3.1 Introduction	. 91
3.2 Lorentz symmetry and event shapes	. 94
3.3 Extracting light-ray OPE data from the leading order collinear EEEC	2 109
3.4 Higher-order collinear EEEC	. 123
3.5 Contact terms and Ward identities	. 134
3.6 EEEC at strong coupling	. 141
3.7 Discussion and future directions	. 154
Chapter IV: Spinning dispersive CFT sum rules and bulk scattering	. 157
4.1 Introduction	. 157
4.2 Review: superconvergence sum rules	. 160
4.3 Heavy action from a spacetime saddle-point analysis	. 167
4.4 Flat space interpretation	. 191
4.5 Discussion	. 217
Chapter V: Scalar modular bootstrap and zeros of the Riemann zeta function	. 220
5.1 Introduction	. 220
5.2 Review of Harmonic Analysis	. 222
5.3 $U(1)^c$ CFTs	. 224
5.4 General 2d CFTs	. 238
5.5 2d CFTs and the Riemann Hypothesis	. 243
• 1	

5.6 Future directions
Bibliography
Appendix A: Appendices to chapter 2
A.1 Comments on existence of light-ray operators
A.2 Conventions for two- and three-point structures
A.3 A Lorentzian formula for the light-ray kernel
A.4 An alternative derivation for the light-ray OPE formula
A.5 Kernel of the celestial map
A.6 Spinor Conventions
A.7 Distributional formulas
Appendix B: Appendices to chapter 3
B.1 Alternative derivation of the three-point celestial block
B.2 $\langle R_{\delta,i}^{(1)} \rangle$ and $\langle R_{\delta,i}^{(0)} \gamma_{\delta,i}^{(1)} \rangle$ from direct decomposition
B.3 More details on the Lorentzian inversion formula calculation 299
B.4 Lightcone bootstrap and large- <i>j</i> behavior of $\langle R_{i+6,i}^{(n+1)g/q} \rangle$
B.5 Tree-level EEEC
B.6 More details on the celestial inversion formula
Appendix C: Appendices to chapter 4
C.1 Conventions for two-point and three-point structures
C.2 Heavy action formula from the large ν limit
C.3 Details on matching partial waves
C.4 Dual structures at large $v \dots $
C.5 CFT four-point structures
Appendix D: Appendices to chapter 5
D.1 Pole structure of scalar crossing equation
D.2 Functional action on crossing equation
D.3 $c = 1$ and $c = 2$ revisited

ix

LIST OF ILLUSTRATIONS

Number	·	age
2.1	The celestial sphere S^{d-2} in a two-point event shape. The positions	
	of the detectors are parametrized by $\hat{n}_1, \hat{n}_2 \in S^{d-2}$. The light-ray	
	OPE is an expansion in the angle θ between detectors (solid red arc).	
	Transverse spin j is conjugate to the angle ϕ of one detector around	
	the other on the celestial sphere (solid blue arc)	9
2.2	Chew-Frautschi plot of light-ray operators in the $\mathcal{E} \times \mathcal{E}$ OPE in	
	a hypothetical CFT. We show Regge trajectories of even signature	
	operators with transverse spins $j = 0$ (black curves), $j = 2$ (blue	
	curves), and $j = 4$ (green curves). Light-ray operators that appear	
	in the $\mathcal{E} \times \mathcal{E}$ OPE are marked with dots. They include operators	
	with spin $J = 3$ and transverse spins $j = 0, 2, 4$. (These are the	
	"low transverse spin" terms described in [15].) In addition, there are	
	primary descendants of light-ray operators with transverse spin $j = 4$	
	and spin $J = 5, 7, 9,$	10
2.3	Kinematics of a two-point energy correlator in the process $e^+e^- \rightarrow$	
	hadrons. The picture shows the spatial geometry in the center of mass	
	frame; time is suppressed. Particles e^+ and e^- propagate in along the	
	$\pm \vec{e}_z$ directions, and we measure energy flux in the directions \hat{n}_1, \hat{n}_2 .	
	(The correlator is invariant under swapping e^+ and e^- .) ψ (solid	
	green arc) is the angle between one of the detectors \hat{n}_2 and the beam	
	direction, θ (solid red arc) is the angle between detectors, and ϕ (solid	
	blue arc) parametrizes the angle of \hat{n}_1 around \hat{n}_2 on the celestial sphere.	18
2.4	A Young diagram for an irreducible representation ρ of SO($d-1, 1$).	
	The rows have length m_1, m_2, \ldots, m_n . We often write $m_1 = J$ (spin)	
	and $m_2 = j$ (transverse spin). If we remove the first row of the	
	Young diagram for ρ , the remaining rows (m_2, \ldots, m_n) make a Young	
	diagram for an irreducible representation λ of SO($d - 2$). If we	
	remove another row, the remaining rows (m_4, \ldots, m_n) make a Young	
	diagram for an irreducible representation γ of SO $(d-4)$	24

3.1	Diagrams representing the two-point and three-point celestial blocks.	
	Each vertex should be understood as an OPE differential operator	
	C_{ijk} in the CFT _{<i>d</i>-2} . The symbol \otimes represents the factor (3.29)	101

A plot showing the values of δ , *j* appearing in the conformal block 3.2 decomposition (3.64). Black dashed line is the improved unitarity bound from (3.68). The red dot appears only in QCD, and the black dots appear in both QCD and $\mathcal{N} = 4$ SYM. The coefficients $\langle R_{\delta, j; \delta'_*}^{(1)} \rangle$ and $\langle R_{\delta,i;\delta'_{*}}^{(0)} \gamma_{\delta,i}^{(1)} \rangle$ of dots with a blue circle can be obtained using the direct decomposition method. The $\langle R_{\delta,j;\delta'_*}^{(1)} \rangle$ coefficient of the green line and the $\langle R_{\delta,j;\delta'_{*}}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ coefficient of the orange line can be obtained 3.3 Definition of (r_i, θ_i) . Due to the step function in (3.162), $\mathcal{F}_0(\zeta_1, \zeta_2)$ is nonzero only in the blue region. The three points (1, 0), (0, 1), (1, 1)4.1 Contour of the light transform on the Lorentzian cylinder, where the two dashed lines should be identified. The contour starts at point x4.2 Causality configuration of the functional. The left figure shows the configuration in the original definition (4.33). Points 1, 3 both start from 0 and are integrated along two different null directions. The right figures shows the configuration after applying $\mathcal{T}_1^{-1}, \mathcal{T}_2^{-1}, \mathcal{T}_4$, Our gauge fixing condition is $x_3 = -x_1 = y$, $x_4 = -x_2 = \frac{x}{-x^2}$, $x_5 = e$. 4.3 The two delta functions $\delta(x_{10}^2)\delta(x_{30}^2)$ force x_0 to be on an S^{d-2} (the 4.4 blue curves). The directions of the variables $\tilde{t_0}, \tilde{x_0}$ defined in (4.59) 4.5 The bulk-point limit localizes the spacetime integral to a saddle locus that we call the scattering-crystal. The entire saddle satisfies the causality constraint 4 > 5 > 3 > 0 > 1 > 6 > 2. The two delta functions restrict x_0 to be on an S^{d-2} shown by the red circle in the figure. The saddle locus further restricts x_0 to an S^{d-3} , which is represented by the two red dots (0-sphere). Points 5 and 6 are localized at $\pm e$. Points 1, 2, 3, 4 are restricted to be timelike vectors, although in practice they have purely imaginary spatial components

4.6	Our functional is a special $6j$ symbol (or a tetrahedron) where one of
	the three-point structures is a celestial structure. The cross in the left
	figure stands for the contraction of the $\langle \widetilde{O}^{\dagger} \widetilde{O} \rangle$ tensor. The flat space
	limit is the limit where the two red legs have large quantum numbers.
	(More precisely, the two legs have large scaling dimensions.) 183
5.1	(a) Pole structure of the integral in (5.27) in the complex <i>s</i> plane. The
	poles are located at $s = \frac{c}{2}, \frac{1+z_n}{2}, \frac{1+z_n}{2}$ (shown here for $c = 3$), where
	z_n are the nontrivial zeros of the Riemann zeta function with positive
	imaginary part. If the Riemann hypothesis is true, the tower of poles
	in the figure all occur at real part $\frac{3}{4}$, except for the pole at $s = \frac{c}{2}$. (b)
	Contour deformation of the integral to $\operatorname{Re}(s) > \frac{c}{2} \dots \dots$
5.2	Plot of a bound on the scalar gap for $U(1)^c$ CFTs with 4 derivatives,
	up to central charge $c = 251$. The numerical data seems to be well-
	approximated by a quadratic function with leading coefficient $\frac{1}{4\pi^2}$
	(see (5.57))
5.3	Plot of a bound on the scalar gap for $U(1)^c$ CFTs at odd $c \leq 27$.
	The colors blue, orange, green, red, brown, and purple represent
	the bound we get at $10, 20, \dots, 60$ derivatives, respectively. The
	color black represents the average Narain scalar gap, for comparison.
	(However, there is no a priori reason the average Narain scalar gap
	and the optimal $U(1)^c$ scalar gap should be similar.) See Table 5.1
	for the numerical data
5.4	Scalar part of the $SU(3)_1$ WZW model with first two leading terms
	subtracted, rescaled by $T^{1/4}$, plotted up to $T = 300$. If the Rie-
	mann hypothesis is true, then at large temperature, this function will
	remain bounded. However, if the Riemann hypothesis is false, at
	large temperatures the oscillations will grow in size and become
	unbounded (modulo the subtlety explained in footnote 14). In this
	plot, $\alpha := 2E_1(e^{2\pi i/3}) + \frac{3}{\pi}(\gamma_E + \log(4\pi) + 24\zeta'(-1) - 2) \approx 0.975$
	(see (5.77)). By fitting this plot with oscillating functions in $log(T)$,
	we can numerically recover the first few nontrivial zeros of the zeta
	function. (A similar plot can be made for any $c > 1$ CFT.) 245

xii

C.1 The steepest descent flow of the saddle integral (C.64). The red point is the saddle point. We show the deformed contour passing through the saddle in green and orange. The green part starts at infinity, goes in the opposite direction of the flow, and gradually spirals in toward the saddle. Along the green contour, the integrand keeps increasing and reaches maximum at the saddle point. It passes through the saddle point along a direction that is rotated by $e^{-\frac{3\pi i}{4}}$ relative to the real axis. After passing through the saddle point, the contour (in orange) goes in the same direction as the flow and gradually spirals out to infinity. The integrand at infinity goes as $\alpha^{-i\nu}$, so by making the contour spiral many times clockwise at infinity, we ensure that the contribution $\alpha^{-i\nu} \propto e^{2\pi \nu \arg(\alpha)}$ can be made arbitrarily small. 337

LIST OF TABLES

Number 5.1

- Page Upper bounds on the scalar gap from $U(1)^c$ CFTs with odd $c \le 27$ after taking up to 10, 20, \cdots , 60 derivatives of our crossing equation (i.e., the maximum value of *n* in (5.55)) computed to three decimal

Chapter 1

INTRODUCTION

Quantum field theory (QFT) is a crucial and foundational theoretical framework for modern physics. Since the early twentieth century, it has been used to describe a wide variety of physics phenomena. These include the interactions of elementary particles in the Standard Model, many-body systems in condensed matter physics, and also the study of gravitational theories.

One useful method for studying QFTs is perturbation theory, which can be applied to theories with weak coupling, such as the QED or QCD at high energy. However, there are also numerous QFTs that do not have a small parameter and then one has to study a strong-coupled system. It is therefore important to develop techniques that utilize the non-perturbative structures of the theory, such as symmetries, anomalies, and dualities. In this thesis, we will focus on a method, called the bootstrap, that aims to constrain the theories using symmetry and consistency conditions, such as unitarity and causality.

The bootstrap method is particularly powerful when the theory has enhanced symmetry. Typically, a physical system comes with a characteristic length scale, and the correlation functions of the theory decays exponentially as a function of the distance. However, there are systems that do not have any intrinsic length scales, and the correlations only decay as power laws. Such theories are scale-invariant, meaning they are invariant under rescaling (zooming in/out). Generically, a theory with scaling symmetry (and the usual Poincaré symmetry) has an even bigger symmetry group called the conformal group, which are angle-preserving transformations.¹ The conformal group consists of translations, rotations, rescaling, and special conformal transformations, which are compositions of inversions and translation. QFTs with conformal symmetry are called conformal field theories (CFTs).

There are many reasons why studying CFTs is important. First, they describe critical phenomena in phase transition. Crucially, critical points of different systems can be described by the same CFT. For example, the uniaxial magnet and the liquid-vapor transition of water both have the same power law behavior (with identical exponents)

¹This statement can be proven when the spacetime dimension is 2. Note, however, that there are exceptions in higher dimensions for some free theories and non-unitary theories.

at their critical points, and they are both described by the 3d Ising CFT. One can then say that these two critical points are in the same universality class.

This type of behavior that multiple theories belonging to the same universality class also shows up in the study of QFTs. When changing the energy scale of a QFT, its behavior is described by the renormalization group (RG) flow. As one keeps lowering the energy, one possible scenario is that the theory flows to a nontrivial fixed point that is invariant under changing the energy scale, which corresponds to a CFT. Thus, CFTs can be thought of as important landmarks in the space of QFTs, and QFTs live on the RG flow between CFTs.

Finally, CFTs are also very important for understanding quantum gravity. This is due to the AdS/CFT correspondence, which says that a theory of quantum gravity in a (d + 1)-dimensional Anti-de Sitter (AdS) space is equivalent to a *d*-dimensional CFT living on its boundary. Therefore, through AdS/CFT, CFTs can give non-perturbative definitions of quantum gravity. Additionally, in string theory, the worldsheet theory is given by a two-dimensional CFT.

How much does one need to know about a CFT to fully characterize it? First, each local operator carries two quantum numbers, spin and scaling dimension, describing how it transforms under conformal transformations. Furthermore, conformal symmetry allows one to show that each local operator in a CFT has a one-to-one correspondence to a state in its Hilbert space on a S^{d-1} . This is called the state-operator correspondence. As a result, local operators in a CFT are endowed with an operator product expansion (OPE) structure. That is, a product of two local operators can be written as a convergent sum of local operators. The coefficients in this sum are called the OPE coefficients. All correlation functions of local operators of the CFT are completely determined by these OPE coefficients and the quantum numbers of the operators, and they are called the "CFT data."

Let us now talk about how to study CFTs using bootstrap. The bootstrap approach is based on the philosophy that one should focus on the symmetries and consistency conditions of the CFT, and not care about its microscopic description. This idea of studying CFTs using symmetries and basic consistency conditions started in 1970s by Ferrara, Gatto, Grillo [1], and Polyakov [2]. It has later led to the solutions of a class of exactly solvable two-dimensional CFTs called the 2d minimal models [3]. In the two-dimensional case, the aforementioned conformal symmetry is enhanced even further to an infinite-dimensional symmetry group, called the Virasoro symmetry. Therefore, the bootstrap approach is even more powerful in 2d.

On the other hand, not much progress was made on solving $d \ge 3$ CFTs using bootstrap until the seminal work by Rattazzi, Rychkov, Tonni, and Vichi in 2008 [4]. Their main idea is to look at the crossing equation, which comes from imposing associativity of the OPE to a four-point correlation function. One important fact about this equation is that the coefficients in the sum are all positive for unitary CFTs. Instead of trying to completely solve the equation, they made rigorous numerical statements about the lowest scaling dimension that can appear in the OPE using the technique of linear programming. This idea of numerically solving the crossing equation was then further explored and then implemented in the semidefinite programming solver SDPB [5]. One of the most impressive results from this numerical bootstrap method is the world record determination of the critical exponents of the 3d Ising and O(N) CFT [6].

Numerical conformal bootstrap mainly studies the crossing equation in Euclidean signature. More generally, CFTs can have different spacetime signature. Euclidean CFTs live on \mathbb{R}^d , and describe thermal critical points of statistical systems. Lorentzian CFTs live on $\mathbb{R}^{d-1,1}$ and describe quantum critical points. Formally, the Osterwalder-Schrader reconstruction implies that unitary Lorentzian CFTs are related to reflection-positive Euclidean CFTs by a Wick rotation. However, it turns out that many physics are deeply hidden in Euclidean signature, and Lorentzian signature can make more manifest many interesting observables and constraints. For example, consider the possible singularities of CFT correlation functions. In Euclidean signature, singularities can only appear in the OPE limit, where two local operators approach each other. On the other hand, in Lorentzian signature, one can have more singularities in different kinematic limits. One such example is the lightcone limit where two operators are lightlike separated.

Studying the crossing equation in the lightcone limit also gives interesting constraints. In [7, 8], it was shown that in the $\phi \times \phi$ OPE, there must exist a family of operators, called the double-twist operators, whose twist approaching $2\Delta_{\phi}$ at large spin, where Δ_{ϕ} is the scaling dimension of ϕ . Another important result derived in Lorentzian signature is the Lorentzian inversion formula. An inversion formula essentially inverts the OPE decomposition of the four-point function, and expresses the OPE data as an integral of the four-point against some kernel. In [9, 10], a Lorentzian inversion formula was derived by starting with an integral over the Euclidean regime and Wicking rotating to Lorentzian signature. Importantly, the expression of the integral is analytic in spin. This implies that the CFT data, which originally only make sense for integer spin, admit an analytic continuation in spin to all complex *J*, and the local operators of the CFT should organize themselves into continuous families called the Regge trajectory. It was then later understood that the operators with continuous spin are light-ray operators [11]. To better understand continuous-spin CFT dynamics encoded by these light-ray operators, it is important to study Lorentzian signature.

Besides considering CFTs in the Lorentzian signature, one can also put the theory on some other nontrivial manifold. This provides even more constraints that can be imposed when bootstrapping CFTs. In two dimensions, one concrete realization of this idea is to put the theory on a torus, and one new consistency condition is the modular invariance of the partition function.

In chapter 2, 3, 4, and 5 of this thesis, I present several different results related to d > 2 CFTs in Lorentzian signature and d = 2 CFTs on a torus. I first study an intrinsically Lorentzian observable called the energy correlator that is closely related to collider physics. Then, I show that by considering an integral of CFT four-point against a special kernel, one can obtain nontrivial constraints that are particularly useful for holographic CFTs. Finally, I consider the torus partition function of 2d CFTs, and apply harmonic analysis to prove new rigorous statements about the scalar spectrum of a special class of 2d CFTs. I discuss them in more detail below.

1.1 Conformal collider physics

We study energy correlators in CFTs in chapters 2 and 3. Energy correlators (or more generally event-shapes) are defined as the distribution of energy flux going through energy detectors placed at future null infinity. They are important observables in real-world colliders. In CFTs, they can be written as the matrix elements of light-ray operators. If there is only one energy detector, we have the one-point energy correlator. It is completely fixed by symmetry up to several OPE coefficients. It can also be shown that one-point energy correlators are always positive (known as the average null energy condition), and it leads to interesting bounds on the CFT data [12].

Energy correlators are perhaps one of the simplest intrinsically Lorentzian observables in CFT. By developing theoretical understanding of these observables and computing them in some CFTs, one might hope to measure them at a quantum critical point and confirm with theoretical predictions. This could pave the way for exploring nontrivial Lorentzian CFT dynamics through table-top experiments. Another motivation for studying energy correlator is its connection to collider physics. They are observables that were measured in real-world colliders such as the LHC. Physicists have been developing many perturbative techniques to compute them in perturbative QCD. They often use the N = 4 SYM at weak coupling, which is a close cousin of perturbative QCD, as a platform to test those methods. If we understand how to compute energy correlator in N = 4 SYM using CFT techniques, it could be a nontrivial cross check between the two different methods. Moreover, phrasing the calculation in symmetry-based field theory language can also help understand better how to organize the complicated perturbative calculation.

In chapter 2, I study the two-point energy correlator. Although the energy detectors are nonlocal operators, they are point-like objects on the celestial sphere. We derive that when the two energy detectors get close to each other, there is still an OPE between them. We call this OPE the light-ray OPE. Interestingly, in order to produce high transverse spin² terms in the light-ray OPE, one needs a class of conformally-invariant differential operators which shift the transverse spin. The objects appearing in the OPE are light-ray operators with continuous spin J = 3, 5, 7, ... (continuous in the sense that there are no local operators at those points).

In chapter 3, I study the three-point energy correlator (EEEC). The light-ray OPE of three energy detectors is not known yet. However, there are still some nontrivial statements one can make using symmetry. We show that Lorentz symmetry implies that EEEC can be decomposed into celestial blocks, which are special functions completely fixed by symmetry. In the collinear limit where all operators approach each other, the celestial block reduces to a conformal block, with the conformally-invariant cross ratios given by the ratio between the angles between the detectors. Using the celestial block expansion and known leading order data of EEEC, we can make all-order predictions in certain kinematic limits, even in QCD. We also derive a celestial inversion formula that computes the coefficients of the celestial block expansion from an integral of the EEEC over the celestial sphere.

1.2 Dispersive sum rules

Causality is an important consistency condition for any QFT. One condition that follows from causality is that operators commute when they are spacelike separated. In fact, the crossing equation can be understood as this statement by consider a four-point function with all spacelike separations. One can then try to get an even

²Transverse spins describes how operators transform under rotations on the celestial sphere.

stronger constraint by integrating over the positions of the operators, while making sure that the operators remain spacelike. This will give a "sum rule" that says an integral of the four-point function should vanish.³ Such sum rules are powerful tools that allow us to rule out many seemingly sensible theories as being unphysical.

By carefully choosing the integration kernel, one can obtain a special class of sum rules called the dispersive sum rules. These sum rules have additional properties that their actions on conformal blocks at the double-twist locations $\Delta = 2\Delta_{\phi} + 2n$ vanish. Because of this, in a large-N CFT, dispersive sum rules can cleanly separate the contributions of single-trace and double-trace operators.

Now, let us consider applying this sum rule to a four-point function in a holographic CFT. We assume that the CFT has large N and a large gap Δ_{gap} in its single-trace higher-spin spectrum. By AdS/CFT, this means that the bulk dual should be described by a weakly-coupled gravitational EFT at low energy. On the CFT side, the action of the dispersive sum rules keep only contributions of single-trace operators below the gap. If we can find a sum rule whose contribution above the gap is always positive (which is a problem very similar to numerical bootstrap), then we can prove inequalities on the OPE coefficients of single-trace operators. On the bulk side, this means that one can prove bounds on the couplings of the bulk EFT. In particular, the bounds will be in terms of the gap Δ_{gap} , which is related to the energy cutoff of the EFT. One way to phrase the result is that dimensional analysis of gravitational EFTs in AdS follows from CFT axioms.

In chapter 4, we consider this problem for general spinning operators. Conceptually, there is no major difference between the spinning case and the scalar case (which was studied in [13]). However, the spinning case is technically much more challenging due to the large number of sum rules and tensor structures. We find that by using the position space language and writing the dispersive sum rule as an integral over a certain region in spacetime, it is straightforward to generalize the calculation to the spinning case. We then derive a list of CFT dispersive sum rules for spinning operators. The actions of the sum rules on heavy blocks have nice flat space interpretations. Thus, we can write down the dictionary between these sum rules and flat space sum rules for photons and gravitons when the operators are spin-1 currents and stress-energy tensors.

³It is called sum rule because one can turn this into a sum over the operator spectrum by apply OPE to the four-point function.

1.3 Scalar modular bootstrap

Modular bootstrap is a program that aims to derive bounds on 2d CFTs from modular invariance of the torus partition function. One can decompose the partition function into Virasoro characters, and the coefficients of this decomposition must be positive due to unitarity.⁴ Modular invariance then implies a "modular crossing equation" for this decomposition. A method similar to the conformal bootstrap can then be applied to prove bounds on the spectrum of the CFT. One interesting question is how do these bounds scale in the limit of large central charge c, and whether they can be saturated by the BTZ black hole solution in AdS₃.

In chapter 5, we study the implication of modular invariance using a different approach. Using harmonic analysis, we can decompose the partition function into a basis that is manifestly modular-invariant. Therefore we do not have to impose modular invariance. However, unitarity in this decomposition becomes less clear. Eventually, we derive a new crossing equation that only acts on the scalar operators, and it has interesting terms related to the zeros of the Riemann zeta function. The crossing equation takes a simple form if the CFT has an enhanced $U(1)^c$ symmetry. One can use this crossing equation to turn the Riemann hypothesis into a statement about the asymptotic density of scalar operators in these theories. Moreover, using the scalar crossing equation, we also obtain bounds on the scaling dimension of the lightest scalar operators in 2d CFTs with $U(1)^c$ symmetry.

⁴In fact, the coefficients should be positive integers, since they count the number of operators. However, so far it has not been clearly understood how to impose integrality constraints into bootstrap.

Chapter 2

TRANSVERSE SPIN IN THE LIGHT-RAY OPE

This chapter is based on

 Cyuan-Han Chang, Murat Kologlu, Petr Kravchuk, David Simmons-Duffin, and Alexander Zhiboedov. "Transverse spin in the light-ray OPE". In: *JHEP* 05 (2022), p. 059. DOI: 10.1007/JHEP05(2022)059. arXiv: 2010.04726 [hep-th].

2.1 Introduction

In Euclidean signature, the operator product expansion (OPE) gives a convergent expansion for correlation functions around coincident-point singularities. This expansion lets us formulate nonperturbative bootstrap conditions and perform myriad computations. Lorentzian correlators are in principle determined from Euclidean ones by analytic continuation. However, a given OPE may not commute with this continuation. Furthermore, Lorentzian signature allows for a much richer set of singularities than Euclidean signature [14]. It is important to develop nonperturbative tools for understanding these singularities and efficiently computing Lorentzian observables.

The work [15] introduced an intrinsically Lorentzian OPE for products of nullintegrated operators on the same null plane.¹ This OPE can be applied to Lorentzian observables called "event shapes," which measure the distribution of energy (and other quantities) in a collider-like experiment [18, 19, 12]:

$$\langle \Psi | \mathcal{E}(\widehat{n}_1) \cdots \mathcal{E}(\widehat{n}_k) | \Psi \rangle.$$
 (2.1)

Here $|\Psi\rangle$ is a state, for example created by sending particles from past null infinity and letting them scatter. Each operator $\mathcal{E}(\hat{n}_i)$ is a stress-tensor integrated along retarded time at future null infinity, at a fixed position $\hat{n}_i \in S^{d-2}$ on the celestial sphere. The $\mathcal{E}(\hat{n}_i)$ act like calorimeters, measuring the energy flux at angle \hat{n}_i . We refer to the $\mathcal{E}(\hat{n}_i)$ as "detectors."

¹See also [16] which derived the leading term in the expansion of a particular two-point event shape in N = 4 SYM theory, and [17] which derived leading terms in collinear limits of energy-energy correlators in the setting of perturbative gauge theories.

The OPE developed in [15] gives a nonperturbative expansion for event shapes in the separation between a pair of detectors $1 - \hat{n}_1 \cdot \hat{n}_2$. Specifically, [15] gave a precise description of the low "transverse spin" terms in this OPE. Here, "transverse spin" *j* refers to spin on the celestial sphere: it is conjugate to rotation of the points \hat{n}_1, \hat{n}_2 around each other, see figure 2.1. The spin *J*, a different quantum number, is related to selection rules for the light-ray OPE.



Figure 2.1: The celestial sphere S^{d-2} in a two-point event shape. The positions of the detectors are parametrized by $\hat{n}_1, \hat{n}_2 \in S^{d-2}$. The light-ray OPE is an expansion in the angle θ between detectors (solid red arc). Transverse spin *j* is conjugate to the angle ϕ of one detector around the other on the celestial sphere (solid blue arc).

The low transverse spin terms in the $\mathcal{E} \times \mathcal{E}$ OPE are given by spin J = 3 lightray operators [15], in accordance with an earlier analysis of the light-ray OPE by Hofman and Maldacena [12]. This is the complete OPE in 3d CFTs, where the transverse direction is 1-dimensional. Furthermore, low transverse spin terms are sufficient for studying two-point event shapes in rotationally-symmetric states in d > 3 dimensions. This covers many of the cases studied in the literature, including simple energy two-point correlators in QCD [19, 20, 21, 22, 23] and $\mathcal{N} = 4$ SYM [24, 25, 26].

In this work, we derive the remaining terms in the light-ray OPE, including arbitrary transverse spin. Higher transverse spin terms are important in d > 3 dimensional theories when the initial state $|\Psi\rangle$ is not rotationally invariant or when other detectors are present. For example, in an energy three-point correlator $\langle \Psi | \mathcal{E}(\hat{n}_1) \mathcal{E}(\hat{n}_2) \mathcal{E}(\hat{n}_3) | \Psi \rangle$, transverse spin in the $\mathcal{E}(\hat{n}_1) \times \mathcal{E}(\hat{n}_2)$ OPE encodes dependence of the event shape on the direction of the tangent vector from \hat{n}_1 to \hat{n}_2 relative to the third direction \hat{n}_3 .



Figure 2.2: Chew-Frautschi plot of light-ray operators in the $\mathcal{E} \times \mathcal{E}$ OPE in a hypothetical CFT. We show Regge trajectories of even signature operators with transverse spins j = 0 (black curves), j = 2 (blue curves), and j = 4 (green curves). Light-ray operators that appear in the $\mathcal{E} \times \mathcal{E}$ OPE are marked with dots. They include operators with spin J = 3 and transverse spins j = 0, 2, 4. (These are the "low transverse spin" terms described in [15].) In addition, there are primary descendants of light-ray operators with transverse spin j = 4 and spin $J = 5, 7, 9, \ldots$.

In seeking the higher transverse spin terms in the $\mathcal{E} \times \mathcal{E}$ OPE, we initially encounter a puzzle: there are no spin J = 3 primary light-ray operators of the type defined in [11] that can do the job. The resolution is that higher transverse spin terms are "primary descendants." They are given by the action of special conformallyinvariant differential operators \mathcal{D}_{2n} on light-ray operators with higher spins J =3 + 2n.² These conformally-invariant differential operators convert spin J into transverse spin j. They are well-defined only when acting on objects with special quantum numbers — precisely the quantum numbers that arise in the light-ray OPE. We find that the $\mathcal{E} \times \mathcal{E}$ OPE takes the schematic form

$$\mathcal{E} \times \mathcal{E} = \sum_{i} \left(\mathbb{O}_{i,J=3,j=0}^{+} + \mathbb{O}_{i,J=3,j=2}^{+} + \mathbb{O}_{i,J=3,j=4}^{+} \right) + \sum_{n,i} \mathcal{D}_{2n} \mathbb{O}_{i,J=3+2n,j=4}^{+}.$$
 (2.2)

Here, $\mathbb{O}_{i,J,j}^+$ are light-ray operators on the *i*-th Regge trajectory with spin J and transverse spin *j*. The special differential operators \mathcal{D}_{2n} act on $\mathbb{O}_{i,J=3+2n,j=4}^+$ to give higher transverse spin terms. A Chew-Frautschi plot of the light-ray operators in (2.2) is depicted in figure 2.2.

²In applications to event shapes such as (2.1), where the light-ray operators appearing in the OPE are inserted at spatial infinity (and extend along the future null infinity), the operators \mathcal{D}_{2n} are actually polynomials in the special conformal generators K_{μ} . This happens because an inversion which sends a finite point to infinity maps the translation generators P_{μ} (which act by derivatives) to K_{μ} .

The higher transverse spin terms are new ingredients in the light-ray OPE. However, it is interesting that they do not require us to go outside the space of light-ray operators defined in [11]. Instead, they are hidden in an interesting way inside the usual space of light-ray operators — at higher values of J. Along the way to understanding higher transverse spin terms in the light-ray OPE, we will also find a much simpler derivation of the original light-ray OPE from [15].

We begin in section 2.2 with an introduction to the kinematics of transverse spin, focusing on the example of energy correlators in the process $e^+e^- \rightarrow$ hadrons. In section 2.3, we provide a more detailed introduction to the underlying representation theory and the special conformally-invariant differential operators that raise transverse spin. In section 2.4, we give a new derivation of the light-ray OPE, using null-integrated scalars as an example. In section 2.5, we generalize this discussion to the OPE of null integrals of arbitrary local operators. In section 2.6, we rederive the scalar light-ray OPE using the light-ray OPE formula for general operators, and give the light-ray operators that appear in the light-ray OPE of two charge detectors and the light-ray OPE of two energy detectors. In section 2.7, we check our formulas for an event shape in $\mathcal{N} = 4$ SYM with a non-rotationally-symmetric final state. We conclude in section 2.8 with discussion and future directions.

2.2 Event shapes, OPEs, and transverse spin

Event shapes describe patterns of excitations at future null infinity. Perhaps the most important examples are energy correlators

$$\langle \Psi | \mathcal{E}(\widehat{n}_1) \cdots \mathcal{E}(\widehat{n}_n) | \Psi \rangle,$$
 (2.3)

which measure the distribution of energy at future null infinity in the state $|\Psi\rangle$. Here, $\mathcal{E}(\hat{n}_i)$ are calorimeters inserted at future null infinity in the direction $\hat{n}_i \in S^{d-2}$ on the celestial sphere. (We define them more precisely below.)

Suppose that $|\Psi\rangle$ is created by a linear combination of local operators O_i acting on the vacuum,

$$|\Psi\rangle = \int d^d x \sum_i f_i(x) O_i(x) |\Omega\rangle.$$
(2.4)

The energy correlator is

$$\langle \Psi | \mathcal{E}(\widehat{n}_1) \cdots \mathcal{E}(\widehat{n}_n) | \Psi \rangle = \int d^d x d^d x' \sum_{i,j} f_i^*(x) f_j(x') \langle \Omega | O_i^{\dagger}(x) \mathcal{E}(\widehat{n}_1) \cdots \mathcal{E}(\widehat{n}_n) O_j(x') | \Omega \rangle$$

$$= \int \frac{d^d p}{(2\pi)^d} \sum_{i,j} \widetilde{f_j}^*(p) \widetilde{f_i}(p) \langle O_i(p) | \mathcal{E}(\widehat{n}_1) \cdots \mathcal{E}(\widehat{n}_n) | O_j(p) \rangle,$$

$$(2.5)$$

where

$$|O_i(p)\rangle = \int d^d x e^{ipx} O_i(x) |\Omega\rangle, \qquad (2.6)$$

and $\tilde{f}_i(p)$ is the Fourier transform of $f_i(x)$. The matrix element on the last line of (2.5) is defined by stripping off a momentum-conserving delta function:

$$\langle O_i(q)|\mathcal{E}(\widehat{n}_1)\cdots\mathcal{E}(\widehat{n}_n)|O_j(p)\rangle = \langle O_i(p)|\mathcal{E}(\widehat{n}_1)\cdots\mathcal{E}(\widehat{n}_n)|O_j(p)\rangle(2\pi)^d\delta^d(p-q).$$
(2.7)

The delta-function $\delta^d(p-q)$ appears because the detectors $\mathcal{E}(\hat{n}_i)$ are translationinvariant: translations do not alter the direction in which the excitations exit the system at future null infinity. Thus, we are naturally led to study expectation values of energy detectors in momentum eigenstates,

$$\langle O_i(p)|\mathcal{E}(\widehat{n}_1)\cdots\mathcal{E}(\widehat{n}_n)|O_j(p)\rangle.$$
 (2.8)

2.2.1 Example: $e^+e^- \rightarrow$ hadrons

As a concrete example, consider the Standard Model process $e^+e^- \rightarrow \gamma^* \rightarrow$ hadrons, treated to leading order in the fine structure constant, but to all orders in the QCD coupling.³ The relevant energy correlator is an expectation value in a state created by the electromagnetic current J_{μ} ,

$$\epsilon^{*\nu} \langle J_{\nu}(p) | \mathcal{E}(\widehat{n}_1) \cdots \mathcal{E}(\widehat{n}_n) | J_{\mu}(p) \rangle \epsilon^{\mu}, \qquad (2.9)$$

Here, ϵ_{μ} is a polarization vector for an off-shell photon that depends on the beam direction and helicities of the incoming e^+e^- pair. (We imagine that the collision occurs at high energies, so the electrons can be treated as massless.) Suppose particle 1 moves along the \vec{e}_z direction and particle 2 moves along the $-\vec{e}_z$ direction,

³In practice, one considers the processes at high energies where hadronization corrections can be argued to be relatively small [27, 28], and computes the above event shape using perturbative QCD. Such calculations are one of the ways used to measure the strong coupling constant α_s , see [29] and references therein.

with total momentum $p_1 + p_2 = p = (E, 0, 0, 0)$. We henceforth set E = 1. If the incoming helicities are 1^+2^- , the corresponding polarization vector ϵ^{μ} is

$$\epsilon^{\mu} = \lambda_2 \sigma^{\mu} \overline{\lambda}_1 = (0, 1, i, 0). \tag{2.10}$$

More generally, for helicities $1^{\pm}2^{\mp}$, we have

$$\epsilon^{\mu} = \epsilon^{\mu}_{\pm} \equiv (0, 1, \pm i, 0). \tag{2.11}$$

We can think of the ket and bra together in (2.9) as giving an (unnormalized) density matrix. For example, for helicities 1^+2^- , we have the pure state

$$\rho_{1^+2^-} = |J_{\mu}(p)\rangle \langle J_{\nu}(p)|\epsilon_+^{\mu}\epsilon_-^{\nu}, \qquad (2.12)$$

where we used $\epsilon_{+}^{*\mu} = \epsilon_{-}^{\mu}$. More generally, we may wish to study a mixed state, for example by averaging over incoming helicities

$$\rho_{12}^{\rm av} = |J_{\mu}(p)\rangle\langle J_{\nu}(p)| \cdot \frac{1}{2}(\epsilon_{+}^{\mu}\epsilon_{-}^{\nu} + \epsilon_{-}^{\mu}\epsilon_{+}^{\nu}).$$
(2.13)

It is common to additionally average over the beam direction, replacing $\epsilon^{\mu}_{+}\epsilon^{\nu}_{-} \rightarrow \frac{2}{3}(\eta^{\mu\nu} - \frac{p^{\mu}p^{\nu}}{p^{2}})$. However, this discards valuable information, as we explain below.

Symmetries and the density matrix

Let us understand how symmetries constrain energy correlators in the process $e^+e^- \rightarrow$ hadrons. Along the way, we will introduce the notion of light-ray operators with nonzero transverse spin and understand how they appear in event shapes. Instead of QCD, we will work in a general 4-dimensional CFT. However, much of our analysis will not depend on conformal symmetry. We point out when conformal symmetry is used below.

We concentrate on a two-point energy correlator in a density matrix where we average over helicities (but not the beam direction)

$$\operatorname{Tr}(\rho_{12}^{\operatorname{av}}\mathcal{E}(\widehat{n}_1)\mathcal{E}(\widehat{n}_2)) = \langle J_{\nu}(p)|\mathcal{E}(\widehat{n}_1)\mathcal{E}(\widehat{n}_2)|J_{\mu}(p)\rangle \cdot \frac{1}{2}(\epsilon_+^{\mu}\epsilon_-^{\nu} + \epsilon_-^{\mu}\epsilon_+^{\nu}).$$
(2.14)

Let us separate the tensor $\frac{1}{2}(\epsilon_{+}^{\mu}\epsilon_{-}^{\nu} + \epsilon_{-}^{\mu}\epsilon_{+}^{\nu})$ into irreducible components under the rotation group SO(3) that fixes *p*. Focusing on spatial components *i*, *j* = 1, 2, 3, we have the traceless-symmetric and trace parts

$$\frac{1}{2}(\epsilon_{+}^{i}\epsilon_{-}^{j} + \epsilon_{-}^{i}\epsilon_{+}^{j}) = \left(\frac{1}{2}(\epsilon_{+}^{i}\epsilon_{-}^{j} + \epsilon_{-}^{i}\epsilon_{+}^{j}) - \frac{2}{3}\delta^{ij}\right) + \frac{2}{3}\delta^{ij}.$$
(2.15)

The traceless-symmetric part can be written as a sum of products of null vectors $\vec{q} \in \mathbb{C}^3$:⁴

$$\frac{1}{2}(\epsilon_{+}^{i}\epsilon_{-}^{j} + \epsilon_{-}^{i}\epsilon_{+}^{j}) - \frac{2}{3}\delta^{ij} = \sum_{\vec{q}} q^{i}q^{j}.$$
(2.18)

Plugging this in, we have

$$\operatorname{Tr}(\rho_{12}^{\operatorname{av}}\mathcal{E}(\widehat{n}_1)\mathcal{E}(\widehat{n}_2)) = \sum_q \langle J_\nu(p) | \mathcal{E}(\widehat{n}_1)\mathcal{E}(\widehat{n}_2) | J_\mu(p) \rangle q^\mu q^\nu + \text{trace part}, \quad (2.19)$$

where $q = (0, \vec{q})$ and "trace part" refers to the contribution of the second term in (2.15). Note that each q appearing in the sum is null and orthogonal to p:

$$q^2 = q \cdot p = 0. (2.20)$$

The form (2.19) makes it easy to analyze the constraints of symmetries, since now we have only a single vector q instead of a tensor.

Symmetries and detectors

Next we need a more precise definition of the detector $\mathcal{E}(\hat{n})$. It can be expressed as an integral of the stress-tensor $T^{\mu\nu}$ over future null infinity. To state this more concretely, we can make a conformal transformation that maps future null infinity to the plane $x^- = 0$. We then have

$$\mathcal{E}(\hat{n}) \to 2 \int dx^+ T_{++}(x^- = 0, x^+, \vec{x}),$$
 (2.21)

where $\vec{x} \in \mathbb{R}^{d-2}$ is a function of \hat{n} . As explained in detail in [11], such null-integrated stress tensors can be interpreted in terms of the "light-transform" of *T*, denoted by

$$\mathbf{L}[T](x,z),\tag{2.22}$$

$$\frac{1}{2}(\epsilon_{+}^{i}\epsilon_{-}^{j} + \epsilon_{-}^{i}\epsilon_{+}^{j}) - \frac{2}{3}\delta^{ij} = \sum_{\vec{q}\in Q} q^{i}q^{j}$$
(2.16)

where Q contains four null vectors,

$$Q = \left\{ \frac{1}{\sqrt{6}} (1, 0, \pm i), \frac{1}{\sqrt{6}} (0, 1, \pm i) \right\}.$$
(2.17)

⁴This is a general fact about traceless symmetric tensors. An example decomposition in this case is

where x is a space-time position marking the starting point of the null integral, and z is a future-directed null vector. The definition of L is given in (2.43) below. This description is useful because L[T](x, z) transforms like a primary operator at x. If we send x to past null infinity, L[T](x, z) becomes the null integral in (2.21). If we instead send x to spatial infinity, then L[T](x, z) becomes directly related to $\mathcal{E}(\hat{n})$,

$$\mathcal{E}(\widehat{n}) = 2\mathbf{L}[T](\infty, z) \tag{2.23}$$

where $z = (1, \hat{n})$.

In this work, we derive a nonperturbative OPE between light-ray operators that takes the schematic form

$$\mathbf{L}[T](x,z_1)\mathbf{L}[T](x,z_2) = \sum_{j=0,2,\dots} \sum_i C_{\Delta_i - 1,j}(z_1, z_2, \partial_z, \partial_w) \mathbb{D}_{\Delta_i,j}(x, z, w). \quad (2.24)$$

Here, $C_{\delta,j}(z_1, z_2, \partial_z, \partial_w)$ is a differential operator that is fixed by Lorentz symmetry, and $\mathbb{D}_{\Delta_i,j}(x, z, w)$ are light-ray operators that we characterize in more detail shortly. For now, the only information about $\mathbb{D}_{\Delta_i,j}(x, z, w)$ that we need are its Lorentz transformation properties. It is a homogeneous function of null vectors $z \in \mathbb{R}^{d-1,1}, w \in \mathbb{C}^{d-1,1}$ with homogeneities

$$\mathbb{D}_{\Delta_i,j}(x,\alpha z,\beta w) = \alpha^{1-\Delta_i} \beta^j \mathbb{D}_{\Delta_i,j}(x,z,w).$$
(2.25)

Furthermore, z, w are subject to the constraints

$$z \cdot z = w \cdot w = w \cdot z = 0, \qquad (2.26)$$

and the gauge redundancy

$$w \sim w + \lambda z. \tag{2.27}$$

We can interpret z_1 , z_2 , and z as embedding space coordinates [30] on the celestial sphere S^{d-2} , and w as an embedding space polarization vector on the celestial sphere. The quantum number j labels spin on the celestial sphere, which we call "transverse spin." Further, $\Delta_i - 1$ is a dimension on the celestial sphere, and (2.24) takes the form of an OPE in a fictitious d-2-dimensional Euclidean CFT.

The dependence of $\mathcal{E}(\hat{n})$, $\mathbb{D}_{\Delta,j}$, and $C_{\delta,j}$ on the coordinates z_1, z_2, z, w relies only on Lorentz symmetry — not full conformal symmetry.⁵ We will use conformal

⁵Note that our use of Δ to denote one of the quantum numbers of $\mathbb{D}_{\Delta,j}$ might suggest that we are relying on conformal symmetry. We are not: here $1 - \Delta$ is the Lorentz spin of $\mathbb{D}_{\Delta,j}$. The reason for this convention will become clear soon.

symmetry later to derive (2.24) — in particular to constrain which $\mathbb{D}_{\Delta_i,j}$ can appear. However, it is possible that a similar OPE exists in non-conformal theories, and our analysis of the contributions of light-ray operators $\mathbb{D}_{\Delta_i,j}$ in this section will also apply in that case.

Symmetries and matrix elements

Using the OPE in (2.19), it suffices to compute matrix elements

$$\langle J_{\nu}(p)|\mathbb{D}_{\Delta_{i,j}}(\infty, z, w)|J_{\mu}(p)\rangle q^{\mu}q^{\nu}.$$
(2.28)

In fact, let us analyze a more general matrix element where the density matrix has SO(d - 1) spin *l*:

$$\mathcal{M}(z,w;p,q) = \langle \mathcal{O}_{\mu_1\cdots\mu_k}(p) | \mathbb{D}_{\Delta_i,j}(\infty,z,w) | \mathcal{O}'_{\mu_{k+1}\cdots\mu_l}(p) \rangle q^{\mu_1}\cdots q^{\mu_l}.$$
(2.29)

The virtue of having classified the density matrix and light-ray operators into irreducible components is that \mathcal{M} is fixed by symmetry. The argument is as follows. Because of the gauge redundancy (2.27), w can only appear in the gauge-invariant combination

$$[z,w]^{\mu\nu} = z^{\mu}w^{\nu} - w^{\mu}z^{\nu}.$$
 (2.30)

Because of (2.26), the only antisymmetric tensor we can contract this with is [p, q]. Since w must appear with homogeneity j, we have

$$\mathcal{M}(z, w; p, q) = ([z, w] \cdot [p, q])^j \times \text{something.}$$
(2.31)

Finally, homogeneity in q and z fix the rest of the matrix element up to an overall coefficient:

$$\mathcal{M}(z,w;p,q) \propto ([z,w] \cdot [p,q])^j (-z \cdot q)^{l-j} (-p \cdot z)^{1-\Delta_i - l}.$$
(2.32)

A selection rule

The result (2.32) manifests a selection rule: Light-ray operators with transverse spin j only have nonzero expectation values in density matrices with SO(d - 1) spin l at least j. In other words, for \mathcal{M} to be nonvanishing, we must have

$$j \le l. \tag{2.33}$$

For example, if we average over the beam direction so that only density matrices with l = 0 appear, we discard information about light-ray operators with nonzero transverse spin.

Because of the selection rule (2.33), the low transverse spin terms in the $\mathcal{E} \times \mathcal{E}$ OPE (2.2), which have j = 0, 2, 4, are sufficient for computing two-point event shapes in density matrices with SO(d - 1) spin 4 or less. This includes the scalar density matrix studied in [21, 22, 25, 26]. Higher transverse spin terms are important for density matrices with higher spin and in multi-point event shapes.

The form of the light-ray OPE

The differential operator $C_{\delta,j}$ appearing in (2.24) has an expansion in the angle θ between detectors, or equivalently in small $-2z_1 \cdot z_2 \approx \theta^2$. To leading order in this expansion, $C_{\delta,j}$ acts as

$$C_{\delta,j}(z_1, z_2, \partial_z, \partial_w)([z, w] \cdot [p, q])^j f(z) = (-2z_1 \cdot z_2)^{\frac{\delta - j - 6}{2}} (-[z_1, z_2] \cdot [p, q])^j f(z_2) + \dots,$$
(2.34)

where f(z) is any function of z with the correct homogeneity, and "..." indicates higher-order terms in θ . Applying this to (2.32), we find

$$\langle O_{\mu_{1}\cdots\mu_{k}}(p)|\mathbf{L}[T](\infty,z_{1})\mathbf{L}[T](\infty,z_{2})|O_{\mu_{k+1}\cdots\mu_{l}}'(p)\rangle q^{\mu_{1}}\cdots q^{\mu_{l}}$$

= $\sum_{j,i}\lambda_{i,j}\left((-2z_{1}\cdot z_{2})^{\frac{\Delta_{i}-j-7}{2}}(-[z_{1},z_{2}]\cdot [p,q])^{j}(-z_{2}\cdot q)^{l-j}(-z_{2}\cdot p)^{1-\Delta_{i}-l}+\ldots\right),$
(2.35)

where $\lambda_{i,j}$ are OPE coefficients that are not fixed by Lorentz symmetry. The "..." on the right-hand side are fixed by symmetry and re-sum into a celestial block [15].⁶ We have written only the leading term of the celestial block for simplicity.

Let us specialize further to the kinematics of interest. Note that

$$[z_1, z_2] \cdot [p, q] = 2\vec{n}_{12} \cdot \vec{q}, \qquad (2.36)$$

where $z_i = (1, \hat{n}_i)$, $\vec{n}_{12} = \hat{n}_1 - \hat{n}_2$, and \vec{q} are the spatial components of q. From (2.35), we can compute the leading terms in the OPE for each possible value of l

⁶We give some example calculations of celestial blocks in section 2.7.2.



Figure 2.3: Kinematics of a two-point energy correlator in the process $e^+e^- \rightarrow$ hadrons. The picture shows the spatial geometry in the center of mass frame; time is suppressed. Particles e^+ and e^- propagate in along the $\pm \vec{e}_z$ directions, and we measure energy flux in the directions \hat{n}_1, \hat{n}_2 . (The correlator is invariant under swapping e^+ and e^- .) ψ (solid green arc) is the angle between one of the detectors \hat{n}_2 and the beam direction, θ (solid red arc) is the angle between detectors, and ϕ (solid blue arc) parametrizes the angle of \hat{n}_1 around \hat{n}_2 on the celestial sphere.

and *j*:

$$\begin{aligned} |\vec{n}_{12}|^{\Delta_{l}-9} \left((\vec{\epsilon}_{+} \cdot \vec{n}_{12})(\vec{\epsilon}_{-} \cdot \vec{n}_{12}) - \frac{2}{3}\vec{n}_{12}^{2} \right) + \dots & (j = 2, l = 2), \\ |\vec{n}_{12}|^{\Delta_{l}-7} \left((\vec{\epsilon}_{+} \cdot \hat{n}_{2})(\vec{\epsilon}_{-} \cdot \hat{n}_{2}) - \frac{2}{3} \right) + \dots & (j = 0, l = 2), \\ |\vec{n}_{12}|^{\Delta_{l}-7} + \dots & (j = 0, l = 0). \end{aligned}$$

$$(2.37)$$

where we used (2.18) to replace the sum over q. In each case, the "..." are fixed by symmetry and resum into a celestial block. The fact that light-ray operators with j = 0 appear in two different ways reflects the fact that matrix elements of a given light-ray operator can admit multiple three-point structures, each of which comes with its own OPE coefficient. Light-ray operators with j = 1 do not appear in the $\mathcal{E} \times \mathcal{E}$ OPE due to permutation symmetry under $\hat{n}_1 \leftrightarrow \hat{n}_2$ (which follows from the fact that energy detectors commute).

To be completely explicit, let us parametrize the vectors as

$$\widehat{n}_2 = R_{zx}(\psi)(0, 0, 1)$$

$$\widehat{n}_1 = R_{zx}(\psi)(\sin\theta\cos\phi, \sin\theta\sin\phi, \cos\theta).$$
(2.38)

where $R_{zx}(\psi)$ is a rotation by ψ in the *zx* plane. Here, ψ is the angle between the (nearly coincident) detectors and the beam, θ is the angle between detectors, and ϕ

represents a rotation of the two detectors around each other on the celestial sphere, see figure 2.3. The expressions (2.37) become

$$\theta^{\Delta_{i}-7} \left(\frac{1}{3} - \cos^{2} \phi \sin^{2} \psi\right) + \dots \qquad (j = 2, l = 2),$$

$$\theta^{\Delta_{i}-7} \left(\sin^{2} \psi - \frac{2}{3}\right) + \dots \qquad (j = 0, l = 2),$$

$$\theta^{\Delta_{i}-7} + \dots \qquad (j = 0, l = 0). \qquad (2.39)$$

where "..." are higher order terms in θ . As a check on the first line of (2.39), note that if we set $\psi = 0$ or π , so that the detectors are both approaching the beam direction, the dependence on ϕ goes away, reflecting the fact that our density matrix is invariant under rotations around the beam direction.

Typically both theoretical and experimental analysis of the $e^+e^- \rightarrow$ hadrons process focus on observables averaged over the beam direction. As explained above, this amounts to throwing away the contribution of light-ray operators with non-zero transverse spin, which contain extra information and can provide further nontrivial tests of QCD. Event shapes that are not averaged over the beam direction, so-called *oriented event shapes*, were studied experimentally at LEP by the DELPHI [31] and OPAL [32] collaborations. For a recent discussion of oriented event shapes in QCD, see [33].

2.2.2 A transverse spin puzzle

So far, we have introduced the kinematics of the light-ray OPE inside a two-point event shape. The next question is: what are the operators $\mathbb{D}_{\Delta_i,j}$? In particular, what are the corresponding values of Δ_i and OPE coefficients? In [15], we derived the low transverse spin terms in the light-ray OPE. For simplicity, consider a product of light-transformed scalars $\mathbf{L}[\phi_1]\mathbf{L}[\phi_2]$. The result of [15] is⁷

$$\mathbf{L}[\phi_{1}](x, z_{1})\mathbf{L}[\phi_{2}](x, z_{2}) = \pi i \sum_{i} C_{\Delta_{i}-1,0}(z_{1}, z_{2}, \partial_{z_{2}})\mathbb{O}^{+}_{\Delta_{i}, J=-1}(x, z_{2})$$

+ higher transverse spin. (2.40)

The operators $\mathbb{O}^+_{\Delta_i, J=-1}(x, z)$ are analytic continuations of null-integrated operators $\mathbf{L}[O_i]$ to spin J = -1, where O_i appears in the $\phi_1 \times \phi_2$ OPE. The quantum number Δ_i becomes the analytic continuation of scaling dimensions of O_i . The operators

⁷As explained in [34], this product is only well-defined for a theory with a sufficiently low Regge intercept. Here we assume this is the case for the sake of illustration.

 $\mathbb{O}^+_{\Delta_i,J=-1}(x,z)$ have transverse spin j = 0 — in particular they depend only on x and z and not on an additional polarization vector w. The reason is that only traceless symmetric tensors O_i appear in the OPE of scalar operators. The light-transform $\mathbf{L}[O_i]$ of a traceless symmetric tensor has vanishing transverse spin, and thus so do its analytic continuations.

Note that in d = 3 all (bosonic) operators are traceless-symmetric tensors and correspondingly there is no transverse spin in d - 2 = 1 dimension. The higher transverse spin terms in (2.40) are absent in this case. However, in d > 3 the transverse spin *j* can be non-trivial and we expect infinitely many higher transverse spin terms to appear in (2.40), since there is no reason to expect the event shapes to be independent of the angle ϕ discussed above.⁸

It is natural to expect that higher transverse-spin terms in (2.40) should also be related to the $\phi_1 \times \phi_2$ OPE. However, this presents a puzzle: All the primary light-ray operators $\mathbb{O}_{\Delta_i,J}(x, z)$ built from $\phi_1 \times \phi_2$ using the construction of [11] have vanishing transverse spin. How can we build light-ray operators with nonzero transverse spin to play the role of $\mathbb{D}_{\Delta_i,j}$ with j > 0? It turns out that the $\mathbb{D}_{\Delta_i,j}$ are *primary descendants* of $\mathbb{O}_{\Delta_i,J}$ — i.e. carefully chosen derivatives of $\mathbb{O}_{\Delta_i,J}$ that nonetheless transform like conformal primaries. In the next section, we explain how such primary descendants arise.

2.3 Building transverse spin with differential operators

2.3.1 Local operators, light transforms, and shortening conditions

Consider a local operator $O^{\mu_1\cdots\mu_J}(x)$ with dimension Δ and spin J. Throughout this work, we will use index-free notation, where we contract spin indices with an auxiliary null polarization vector $z \in \mathbb{R}^{d-1,1}$:

$$O(x,z) = O^{\mu_1 \cdots \mu_J}(x) z_{\mu_1 \cdots \mu_J}, \qquad (z^2 = 0).$$
(2.41)

By construction, O(x, z) is a homogeneous polynomial of degree J in z. Under a conformal transformation U, we have

$$UO(x, z)U^{-1} = \Omega(x')^{\Delta}O(x', R(x')z), \qquad (2.42)$$

⁸The more accurate statement is that the first line of (2.40) already contains higher-transverse spin terms: they are generated by the operators $C_{\delta,0}$. However, since the operators $C_{\delta,0}$ are fixed by Lorentz symmetry, these contributions are determined in terms of j = 0 contributions and therefore the event shapes would still be over-constrained if there were no additional contributions to (2.40). For example, one would be able to write a differential equation in z_1, z_2 that the product $\mathbf{L}[\phi_1](x, z_1)\mathbf{L}[\phi_2](x, z_2)$ would have to satisfy. It can be checked that this differential equation is incompatible with the leading term of the $\mathbf{L}[\phi_2] \times \mathbf{L}[\phi_3]$ OPE in $\langle \Psi | \mathbf{L}[\phi_1] \mathbf{L}[\phi_2] \mathbf{L}[\phi_3] | \Psi \rangle$.

where $\Omega(x')$ and R(x') are the local rescaling and rotation associated to U.

Index-free notation is more than a convenience. It allows us to describe a wider class of conformal representations than those associated to local operators. As an example, consider the light-transform

$$\mathbf{L}[O](x,z) = \int_{-\infty}^{\infty} d\alpha (-\alpha)^{-\Delta - J} O\left(x - \frac{z}{\alpha}, z\right), \qquad (2.43)$$

which is an integral of O in the direction of its polarization vector. Using index-free notation, we can interpret the light-transform as a conformally-invariant transform, changing the quantum numbers (Δ, J) by

$$\mathbf{L}: (\Delta, J) \to (1 - J, 1 - \Delta). \tag{2.44}$$

In other words, $\mathbf{L}[O](x, z)$ satisfies the conformal transformation law (2.42), with Δ replaced by 1 - J and J replaced by $1 - \Delta$. (This is clearest from the definition of the light-transform in the embedding space [11].) In general, we define the spin of an object as its homogeneity in the polarization vector z. Because $\mathbf{L}[O](x, z)$ has non-integer spin $1 - \Delta$, it cannot be written in terms of an underlying tensor with $1 - \Delta$ indices.

Though index-free notation appears to treat the nonlocal operator L[O] in the same way as the local operator O, there is still something special about the representations associated to local operators. Specifically, a local operator is a polynomial in its polarization vector. This can be phrased as a kind of shortening condition. Morally speaking, J + 1 derivatives of O(x, z) with respect to z must vanish:

$${}^{``}\partial_z^{\mu_1}\cdots\partial_z^{\mu_{J+1}}O(x,z)''=0.$$
(2.45)

We must take care to write this condition correctly because z is constrained, $z^2 = 0$. Let us parametrize z by

$$z = (1, \vec{y}^2, \vec{y}), \qquad \vec{y} \in \mathbb{R}^{d-2},$$
 (2.46)

where we use lightcone coordinates $z = (z^+, z^-, \vec{z})$ with metric $dz^2 = -dz^+dz^- + d\vec{z} \cdot d\vec{z}$. A more proper shortening condition is

$$\partial_{\vec{y}}^{i_1} \cdots \partial_{\vec{y}}^{i_{J+1}} O(x, z) - \text{traces} = 0, \qquad (2.47)$$

where we subtract traces using the metric on \mathbb{R}^{d-2} . The argument for (2.47) is as follows. O(x, z) is a sum of products of J factors of 1, \vec{y}^2 and the components of

 \vec{y} . If J + 1 \vec{y} -derivatives of some term is nonzero, at least two of those derivatives must act on the same \vec{y}^2 factor, resulting in a nonzero trace. By subtracting traces, we remove such terms.

The shortening condition (2.47) is naturally a traceless symmetric tensor with spin J + 1 in the "transverse" space \mathbb{R}^{d-2} . We can write it more economically by introducing a null vector $\vec{s} \in \mathbb{C}^{d-2}$ such that $\vec{s}^2 = 0$:

$$(\vec{s} \cdot \partial_{\vec{v}})^{J+1} O(x, z) = 0.$$
 (2.48)

We can make (2.48) Lorentz-invariant by introducing a polarization vector $w = (0, 2\vec{y} \cdot \vec{s}, \vec{s}) \in \mathbb{C}^{d-1,1}$. By construction, w is null and transverse to z, i.e. $w \cdot z = w^2 = 0$. We would like w to encode only the d-2 degrees of freedom in \vec{s} , so we must additionally impose a gauge-redundancy $w \sim w + \lambda z$. The condition (2.48) finally becomes

$$(w \cdot \partial_z)^{J+1} O(x, z) = 0.$$
(2.49)

One can check that the differential operator $(w \cdot \partial_z)^{J+1}$ preserves the ideal generated by $z^2 = w \cdot z = w^2 = 0$. More nontrivially, it is gauge invariant under $w \to w + \lambda z$ precisely when acting on functions with homogeneity J in z (we show this in (2.64) below).

The shortening condition (2.49) for O implies a related shortening condition for L[O]. For simplicity, suppose O is a scalar, i.e. J = 0. By integrating by parts inside the light-transform (2.43), we find

$$\mathbf{L}(w \cdot \partial_z) = \frac{1}{2 - \Delta} \Big((z \cdot \partial_x) (w \cdot \partial_z) - (z \cdot \partial_z) (w \cdot \partial_x) \Big) \mathbf{L} \qquad \text{(acting on scalars with dimension } \Delta\text{)}.$$
(2.50)

Consequently, L[O] satisfies its own shortening condition

$$\left((z \cdot \partial_x)(w \cdot \partial_z) - (z \cdot \partial_z)(w \cdot \partial_x)\right) \mathbf{L}[O](x, z) = 0 \qquad (O \text{ scalar}), \qquad (2.51)$$

inherited from the shortening condition (2.49) for O. Just as $w \cdot \partial_z$ is Lorentzinvariant (also conformally-invariant) only when acting on scalar representations, $(z \cdot \partial_x)(w \cdot \partial_z) - (z \cdot \partial_z)(w \cdot \partial_x)$ is conformally-invariant only when acting on representations with the quantum numbers of L[O], i.e. with dimension 1 - 0 = 1.
2.3.2 Reducibility and primary descendants

The operator $(z \cdot \partial_x)(w \cdot \partial_z) - (z \cdot \partial_z)(w \cdot \partial_x)$ can be compared to other conformallyinvariant differential operators that exist for special quantum numbers. A wellknown example is the operator that takes a current to its divergence

$$\mathcal{J}^{\mu} \to \partial_{\mu} \mathcal{J}^{\mu} = (\partial_x \cdot \partial_z) \mathcal{J}(x, z).$$
(2.52)

On the right-hand side, we have written the divergence in index-free notation.

The operator $\partial_x \cdot \partial_z$ changes quantum numbers (Δ, J) by

$$\partial_x \cdot \partial_z : (d-1,1) \to (d,0). \tag{2.53}$$

It is conformally-invariant only when acting on operators with the correct dimension and spin. This has the following representation-theoretic interpretation. Let $V_{\Delta,J}$ be a long multiplet (i.e. a generalized Verma module, see e.g. [35]) of the conformal group with dimension Δ and spin J. $V_{\Delta,1}$ is irreducible for generic Δ . However it becomes reducible when $\Delta = d - 1$:

$$V_{d-1,1} \supset V_{d,0}.$$
 (2.54)

The quotient $\widetilde{V}_{d-1,1} \equiv V_{d-1,1}/V_{d,0}$ is the short representation associated to a conserved current. If $|\mathcal{J}(z)\rangle = z_{\mu}|\mathcal{J}^{\mu}\rangle$ is the highest-weight state of $V_{d-1,1}$, then the highest-weight state of $V_{d,0} \subset V_{d-1,1}$ is

$$P \cdot \partial_z |\mathcal{J}(z)\rangle$$
 (primary descendant). (2.55)

We say that (2.55) is a "primary descendant." More formally, there exists a homomorphism

$$\Phi: V_{d,0} \to V_{d-1,1}, \tag{2.56}$$

sending the highest-weight state of $V_{d,0}$ to (2.55) inside $V_{d-1,1}$. By replacing $P \rightarrow \partial_x$ inside the expression for the primary descendant, we find the conformally-invariant differential operator (2.53).

In summary, conformally-invariant differential operators are in correspondence with *reducible* generalized Verma modules — i.e. generalized Verma modules that contain primary descendants. The differential operators arising in this way have been studied in the mathematics literature, and also in the physics literature due to their relation to poles in conformal blocks [36, 37, 35, 38]. We refer the reader to [35] for a lucid discussion and classification. In order to compare our operator (2.51) to the classification in [35], we must understand the quantum numbers of the corresponding primary descendant.⁹

2.3.3 Index-free notation for general tensor representations

A general finite-dimensional tensor representation ρ of SO(d - 1, 1) has a Young diagram with rows of length (m_1, m_2, \dots, m_n) , where $n = \lfloor \frac{d}{2} \rfloor$, see figure 2.4. We define spin $J = m_1$ as the length of the first row of the Young diagram of ρ . The remaining rows $\lambda = (m_2, \dots, m_n)$ define a representation of SO(d - 2). We define transverse spin $j = m_2$ as the length of the first row of the Young diagram of λ .



Figure 2.4: A Young diagram for an irreducible representation ρ of SO(d - 1, 1). The rows have length m_1, m_2, \ldots, m_n . We often write $m_1 = J$ (spin) and $m_2 = j$ (transverse spin). If we remove the first row of the Young diagram for ρ , the remaining rows (m_2, \ldots, m_n) make a Young diagram for an irreducible representation λ of SO(d - 2). If we remove another row, the remaining rows (m_4, \ldots, m_n) make a Young diagram for an irreducible representation γ of SO(d - 4).

An operator in the representation ρ has indices $O^{\mu_1 \cdots \mu_J; \nu_1 \cdots \nu_{m_2}; \cdots; \rho_1 \cdots \rho_{m_n}}(x)$. Each index group delimited by semicolons is symmetric. Furthermore, O is traceless in all its indices, and satisfies some additional symmetry conditions that we describe below.¹⁰ To use index-free notation, we introduce null polarization vectors

⁹Strictly speaking, we need to be more careful because the notion of generalized (aka parabolic) Verma module depends on a choice of a parabolic subalgebra of conformal algebra. This choice determines, in particular, which quantum numbers can be non-integer. In [35] the parabolic subalgebra is the maximal one for Euclidean conformal algebra, and it allows for non-integer Δ . Here, we are interested in both Δ and J being non-integer, which requires us to consider the maximal parabolic subalgebra of Lorentzian conformal algebra. Therefore, the classification of [35] is not, strictly speaking, applicable here. It would be interesting to have a full classification for Lorentzian conformal group. We take a more simplistic approach: our conformally-invariant differential operators can be identified with analytic continuations in J of the differential operators classified in [35], and this will be enough for our purposes.

¹⁰For simplicity, we only consider the bosonic representations, i.e. the representations of SO(d - 1, 1) and not of Spin(d - 1, 1). Furthermore, we ignore the possible self-duality conditions in even d.

 z, w_1, \ldots, w_{n-1} for each row:

$$O(x, z, \mathbf{w}) = O(x, z, w_1, \dots, w_{n-1})$$

= $O^{\mu_1 \cdots \mu_J; v_1 \cdots v_{m_2}; \cdots; \rho_1 \cdots \rho_{m_n}}(x) z_{\mu_1} \cdots z_{\mu_J} w_{1, v_1} \cdots w_{1, v_{m_2}} \cdots w_{n-1, \rho_1} \cdots w_{n-1, \rho_{m_n}}.$
(2.57)

Here we used the notation **w** to denote the collection of vectors w_i , $i = 1 \cdots n - 1$. The polarization vectors satisfy the relations

$$0 = z^{2}$$

$$0 = w_{1}^{2} = w_{1} \cdot z$$

$$0 = w_{2}^{2} = w_{2} \cdot w_{1} = w_{2} \cdot z$$

$$\vdots$$

$$0 = w_{n-1}^{2} = w_{n-1} \cdot w_{n-2} = \dots = w_{n-1} \cdot z.$$
(2.58)

The additional symmetry properties of O are equivalent to the statement that $O(x, z, \mathbf{w})$ is invariant under the gauge redundancies

$$w_{1} \sim w_{1} + \#z$$

$$w_{2} \sim w_{2} + \#w_{1} + \#z$$

$$\vdots$$

$$w_{n-1} \sim w_{n-1} + \#w_{n-2} + \dots + \#z.$$
(2.59)

The object $O(x, z, \mathbf{w})$ is a homogeneous polynomial of the polarization vectors z, w_1, \ldots, w_{n-1} with degrees (J, m_2, \ldots, m_n) . We often abbreviate $w_1 = w$.

The index-free formalism we have just developed is essentially the embedding space formalism for the Lorentz group.¹¹ In the embedding formalism, a traceless symmetric tensor operator in *d*-dimensions becomes a homogeneous function O(X, Z)of variables $X, Z \in \mathbb{R}^{d+1,1}$ satisfying $X^2 = X \cdot Z = Z^2 = 0$ and a gauge redundancy $Z \sim Z + \lambda X$ [30]. The Lorentz group SO(d - 1, 1) is of course the conformal group in d - 2 dimensions. Thus the embedding formalism applies, with the d + 2dimensional X and Z replaced by the *d*-dimensional z and w. In particular, z can be interpreted as an embedding-space coordinate in d - 2 dimensions.

2.3.4 Raising transverse spin

We can now recognize $w \cdot \partial_z$ (with $w_1 = w$) as a differential operator that raises transverse spin *j*, since it increases the degree in *w* by 1. Specifically, it changes the

¹¹More generally, it comes from the Borel-Weil theorem, see [15] for details.

quantum numbers (Δ, J, j) by

$$w \cdot \partial_z : (\Delta, 0, 0) \to (\Delta, -1, 1). \tag{2.60}$$

The operator $(z \cdot \partial_x)(w \cdot \partial_z) - (z \cdot \partial_z)(w \cdot \partial_x)$, obtained by commuting $w \cdot \partial_z$ through the light-transform, changes (light-transformed) quantum numbers $(1 - J, 1 - \Delta, j)$ by

$$(z \cdot \partial_x)(w \cdot \partial_z) - (z \cdot \partial_z)(w \cdot \partial_x) : (1, 1 - \Delta, 0) \to (2, 1 - \Delta, 1).$$
(2.61)

More generally, let us define the operator

$$\mathcal{D}'_n = \frac{1}{n!} (w \cdot \partial_z)^n, \qquad (2.62)$$

$$\mathcal{D}'_n: (\Delta, n+j-1, j) \to (\Delta, j-1, j+n).$$
(2.63)

The operator \mathcal{D}'_n provides the shortening condition for the Lorentz representation of local operators with spin J = n + j - 1 and transverse spin j. Note that \mathcal{D}'_n preserves the ideal generated by $z^2 = w \cdot z = w^2 = 0$. More nontrivially, it is gauge-invariant under $w \to w + \lambda z$. To see this, we act with the generator of a gauge transformation $z \cdot \partial_w$ on $\mathcal{D}'_n f(z, w)$, where f(z, w) is gauge-invariant:

$$z \cdot \partial_w (w \cdot \partial_z)^n f(z, w) = \sum_{j=0}^{n-1} (w \cdot \partial_z)^j (z \cdot \partial_z - w \cdot \partial_w) (w \cdot \partial_z)^{n-j-1} f(z, w)$$
$$= \sum_{j=0}^{n-1} (2j - n + 1) (w \cdot \partial_z)^{n-1} f(z, w)$$
$$= 0.$$
(2.64)

In the first line, we commute $z \cdot \partial_w$ past other operators until it acts on f(z, w), which it kills. In the second line, we use that f(z, w) has homogeneity n + j - 1 in z and j in w. This computation shows that \mathcal{D}'_n is Lorentz-invariant. Because local operators transform in irreducible representations of the Lorentz group, Schur's lemma implies that they must be killed by \mathcal{D}'_n .

By commuting \mathcal{D}'_n through the light transform,

$$\mathbf{L}\mathcal{D}_n' = \mathcal{D}_n \mathbf{L},\tag{2.65}$$

we find a new operator \mathcal{D}_n given by

$$\mathcal{D}_{n} = \frac{(-1)^{n} \Gamma(\Delta + j - 2)}{\Gamma(\Delta + j - 2 + n) \Gamma(n + 1)} (\partial_{x} \cdot \mathcal{D}_{z,w}^{0+})^{n},$$

$$\mathcal{D}_{z,w;\mu}^{0+} = w_{\mu} (w \cdot \partial_{w} - z \cdot \partial_{z}) + z_{\mu} w \cdot \partial_{z},$$
(2.66)

Here, $(\mathcal{D}_{z,w}^{0+})^{\mu}$ is a weight-shifting operator for the Lorentz group SO(d-1,1) in the vector representation [39]. It can be obtained from the vector weight-shifting operators for the conformal group in [39] by using the analogy of (z, w) with the embedding-space coordinates (X, Z). The operator \mathcal{D}_n is conformally-invariant precisely when acting on the quantum numbers obtained by light-transforming (2.63):

$$\mathcal{D}_n: (2-n-j, 1-\Delta, j) \to (2-j, 1-\Delta, j+n).$$
(2.67)

The operator \mathcal{D}_n falls into the classification of reducible Verma modules described in [35]. In the notation of [35], it has type I_{2,n}.^{12,9}

To summarize, by commuting \mathcal{D}'_n through the light transform, we obtain a special conformally-invariant differential operator \mathcal{D}_n that raises transverse spin. By construction, \mathcal{D}_n vanishes when acting on the light transform of a local operator with the appropriate quantum numbers, since

$$\mathcal{D}_n \mathbf{L}[\mathcal{O}] = \mathbf{L}[\mathcal{D}'_n \mathcal{O}] = 0.$$
(2.68)

Shortly, we will encounter light-ray operators $\mathbb{O}_{i,J,j}^{\pm}(x, z)$ that are conformal primaries with quantum numbers $(1 - J, 1 - \Delta_i, j)$, but that are not light-transforms of local operators. When J = n + j - 1 with $n \in \mathbb{Z}_{\geq 0}$, we can act on such operators with \mathcal{D}_n to obtain new primary descendants with higher transverse spin

$$\mathcal{D}_n \mathbb{O}_{i,j=n+j-1,j}^{\pm}(x,z) : (2-j,1-\Delta_i,j+n).$$
(2.69)

Primary descendants of this type will provide the higher transverse spin terms in the light-ray OPE.

2.4 The complete OPE of scalar detectors

In this section we will explicitly derive the OPE of the form (2.24) for two scalar detectors $\mathbf{L}[\phi_1]\mathbf{L}[\phi_2]$. As discussed in [34], products of scalar detectors are not in general well-defined non-perturbatively. Nevertheless, they make sense kinematically, and give us a nice playground to demonstrate the key concepts, which we will then generalize to arbitrary detectors in section 2.5. Furthermore, in section 2.7 we will see an example where $\mathbf{L}[\phi_1]\mathbf{L}[\phi_2]$ is well-defined and the formulas from this section can be verified.

¹²Similarly, we can think of \mathcal{D}' as an invariant differential operator with respect to SO(d - 1, 1) (the Lorentz group), thought of as the conformal group in d - 2 dimensions. In the notation of [35], it has type I_{1,n}.

2.4.1 Kinematics of the OPE

Let us understand the general form that the OPE should take. First note that the product

$$\mathbb{W}(x, z_1, z_2) \equiv \mathbf{L}[\phi_1](x, z_1)\mathbf{L}[\phi_2](x, z_2), \qquad (2.70)$$

is a conformal primary, if it is well-defined. This is because

$$[K_{\mu}, \mathbb{W}(0, z_1, z_2)] = [K_{\mu}, \mathbf{L}[\phi_1](0, z_1)\mathbf{L}[\phi_2](0, z_2)]$$

=
$$[K_{\mu}, \mathbf{L}[\phi_1](0, z_1)]\mathbf{L}[\phi_2](0, z_2) + \mathbf{L}[\phi_1](0, z_1)[K_{\mu}, \mathbf{L}[\phi_2](0, z_2)] = 0.$$

(2.71)

Similarly, the scaling dimension of $\mathbb{W}(x, z_1, z_2)$ is the sum of scaling dimensions of $\mathbf{L}[\phi_i]$,

$$\Delta_{\mathbb{W}} = (1 - J_1) + (1 - J_2) = 2. \tag{2.72}$$

However, the operator $\mathbb{W}(x, z_1, z_2)$ does not transform irreducibly under the Lorentz group — it depends on two polarization vectors, but they do not satisfy any of the relations described in section 2.3 among themselves. The OPE (2.24) decomposes $\mathbb{W}(x, z_1, z_2)$ into irreducible components, since the detectors $\mathbb{D}(x, z, w)$ are irreducible under the Lorentz group.

In fact, the problem of decomposing $\mathbb{W}(x, z_1, z_2)$ into irreducible representations of the Lorentz group is a familiar one. The Lorentz group SO(d - 1, 1) is isomorphic to the Euclidean conformal group in d - 2 dimensions. Under this isomorphism, z_i become embedding-space coordinates [40, 30] for a fictitious CFT_{d-2}. By the definition of spin we have $\mathbf{L}[\phi_i](x, \lambda z_i) = \lambda^{1-\Delta_i} \mathbf{L}[\phi_i](x, z_i)$, where Δ_i is the scaling dimension of ϕ_i .¹³ At the same time, primary operators \mathcal{P}_{δ} in the embedding formalism for the fictitious CFT_{d-2} should satisfy $\mathcal{P}_{\delta}(\lambda z_1) = \lambda^{-\delta} \mathcal{P}_{\delta}(z_1)$, where δ is the (d - 2)-dimensional scaling dimension. Thus, if we interpret the Lorentz group as a (d - 2)-dimensional conformal group, the transformation properties of $\mathbb{W}(x, z_1, z_2)$ under Lorentz transformations can be described as¹⁴

$$\mathbb{W}(0, z_1, z_2) \sim \mathcal{P}_{\delta_1}(z_1) \mathcal{P}_{\delta_2}(z_2),$$
 (2.73)

where $\delta_i = \Delta_i - 1$. In particular, if we parameterize z_i as in (2.46),

$$z_i^+ = 1, \quad z_i^- = \vec{y}^2, \quad z_i^\mu = y^\mu \ (\mu = 2, \cdots, d-1),$$
 (2.74)

¹³Recall that spin of $\mathbf{L}[\phi_i]$ is $1 - \Delta_i$.

¹⁴Provided that by Lorentz group we mean the group that fixes x. We take x = 0 and use the standard Lorentz group for concreteness.

for $\vec{y} \in \mathbb{R}^{d-2}$, then

$$\mathbb{W}(0,\vec{y}_1,\vec{y}_2) \sim \mathcal{P}_{\delta_1}(\vec{y}_1)\mathcal{P}_{\delta_2}(\vec{y}_2) \tag{2.75}$$

transforms exactly like a pair of scalar primaries with dimensions δ_1 , δ_2 at coordinates \vec{y}_1 , \vec{y}_2 in the fictitious CFT_{*d*-2}.

We are familiar with taking the usual OPE between two scalar primaries,

$$\mathcal{P}_{\delta_1}(\vec{y}_1)\mathcal{P}_{\delta_2}(\vec{y}_2) = \sum_{\delta,j} C_{\delta,j;\mu_1\cdots\mu_j}(\vec{y}_1,\vec{y}_2,\partial_{\vec{y}_2})\mathcal{P}_{\delta,j}^{\mu_1\cdots\mu_j}(\vec{y}_2), \qquad (2.76)$$

where *j* is the (d - 2)-dimensional traceless-symmetric spin.¹⁵ The differential operator $C_{\delta,j;\mu_1\cdots\mu_j}(\vec{y}_1, \vec{y}_2, \partial_{\vec{y}_2})$ is fixed by (d-2)-dimensional conformal symmetry and is the usual operator in Euclidean CFT_{*d*-2}. It implicitly depends on δ_1, δ_2 . For example, the leading term of this operator in the $\vec{y}_1 \rightarrow \vec{y}_2$ limit is given by

$$C^{\mu_1\cdots\mu_j}_{\delta,j}(\vec{y}_1,\vec{y}_2,\partial_{\vec{y}_2}) = |\vec{y}_{12}|^{\delta-\delta_1-\delta_2-j}(y^{\mu_1}_{12}\cdots y^{\mu_j}_{12} - \text{traces}) + \cdots .$$
(2.77)

It will be convenient to work with (2.76) written in embedding space coordinates. For this, note that traceless-symmetric primary operators $\mathcal{P}_{\delta,j}^{\mu_1\cdots\mu_j}(\vec{y})$ are described in the embedding formalism as functions of z, w satisfying $z^2 = w^2 = z \cdot w = 0$ subject to

$$\mathcal{P}_{\delta,j}(z,w) = \mathcal{P}_{\delta,j}(z,w+\alpha z), \qquad (2.78)$$

$$\mathcal{P}_{\delta,j}(z,\lambda w) = \lambda^{j} \mathcal{P}_{\delta,j}(z,w).$$
(2.79)

In the notation of [30] we have $P_{\text{there}} = z_{\text{here}}$, $Z_{\text{there}} = w_{\text{here}}$. So we can rewrite (2.76) as

$$\mathcal{P}_{\delta_1}(z_1)\mathcal{P}_{\delta_2}(z_2) = \sum_{\delta,j} C_{\delta,j}(z_1, z_2, \partial_{z_2}, \partial_{w_2})\mathcal{P}_{\delta,j}(z_2, w_2), \qquad (2.80)$$

where $C_{\delta,j}(z_1, z_2, \partial_{z_2}, \partial_{w_2})$ is $C_{\delta,j;\mu_1\cdots\mu_j}(\vec{y}_1, \vec{y}_2, \partial_{\vec{y}_2})$, "lifted" to the embedding space. The embedding space expression for $C_{\delta,j}$ has been studied in a number of papers (e.g. [41, 42, 1, 43] to name a few), but we will not need its explicit form. All we need is that it is SO(d - 1, 1)-invariant and upon restriction to the Poincaré section it becomes (2.76).

¹⁵Normally, we would have OPE coefficients in the right-hand side. Since here we are just discussing kinematics, we omit them. One can imagine that they have been absorbed into the definition of $\mathcal{P}_{\delta,j}$.

Translating this back to $W(x, z_1, z_2)$, we see that the Lorentz symmetry requires the detector OPE to take the form

$$\mathbb{W}(x, \vec{y}_1, \vec{y}_2) = \sum_i C_{\delta_i, j_i; \mu_1 \cdots \mu_{j_i}}(\vec{y}_1, \vec{y}_2, \partial_{\vec{y}_2}) \mathbb{W}_i^{\mu_1 \cdots \mu_{j_i}}(x, \vec{y}_2)$$
(2.81)

or

$$\mathbf{L}[\phi_1](x, z_1)\mathbf{L}[\phi_2](x, z_2) = \mathbb{W}(x, z_1, z_2) = \sum_i C_{\delta_i, j_i}(z_1, z_2, \partial_{z_2}, \partial_{w_2})\mathbb{W}_i(x, z_2, w_2),$$
(2.82)

for some primary operators $\mathbb{W}_i(x, z, w)$ with (d-2)-dimensional quantum numbers δ_i, j_i .¹⁶ Note that w satisfies the same properties as the second-row polarization vector w in section 2.3, and so the operators $\mathbb{W}_i(x, z, w)$ transform in irreps of SO(d-1, 1) with two-row Young diagrams, with the first row of length $-\delta_i$ and the second row of length j_i .

2.4.2 Harmonic analysis on the celestial sphere

So far, (2.82) is an ansatz based on kinematics. In this section we will show under which conditions (2.82) holds, and provide a formula for $\mathbb{W}_i(x, z, w)$.

We will focus on a generic matrix element of $W(x, z_1, z_2)$. Specifically, we will consider an expectation value of the form

$$W(z_1, z_2) \equiv \langle O_4(p) | \mathbb{W}(\infty, z_1, z_2) | O_3(p) \rangle, \qquad (2.83)$$

for some local operators O_3 , O_4 . We allow O_3 and O_4 to carry arbitrary spin and do not require them to be primary.

In the language of the fictitious CFT_{d-2} , $W(z_1, z_2)$ is a function of two points that at each point transforms as a scalar primary of dimension δ_i . The main idea is then to decompose $W(z_1, z_2)$ into (d-2)-dimensional conformal partial waves [44, 45]. This is equivalent to constructing the SO(d-1, 1) Casimir operators from the Lorentz generators $M_{\mu\nu}$ defined by

$$M^{\mu\nu} = M_1^{\mu\nu} + M_2^{\mu\nu}, \qquad (2.84)$$

$$M_i^{\mu\nu} = z_i^{\nu} \frac{\partial}{\partial z_{\mu,i}} - z_i^{\mu} \frac{\partial}{\partial z_{\nu,i}}, \qquad (2.85)$$

¹⁶Above, we had set x = 0 but by translation invariance this is valid for any x.

and finding a basis of their common eigenfunctions.¹⁷ In fact, note that the standard conformally-invariant three-point functions

$$\langle \mathcal{P}_{\delta_1}(z_1) \mathcal{P}_{\delta_2}(z_2) \mathcal{P}_{\delta,j}(z,w) \rangle = \frac{(-4)^j (w \cdot z_2 z_1 \cdot z - w \cdot z_1 z_2 \cdot z)^j}{(-2z_1 \cdot z_2)^{\frac{\delta_1 + \delta_2 - \delta + j}{2}} (-2z_1 \cdot z)^{\frac{\delta_1 + \delta - \delta_2 + j}{2}} (-2z_2 \cdot z)^{\frac{\delta_2 + \delta - \delta_1 + j}{2}}$$
(2.86)

are eigenfunctions of all the Casimirs for any z, w and δ, j .¹⁸ This is because by conformal invariance, the action of the Casimirs on z_1, z_2 on (2.86) is equivalent to the action on z, w, and $\mathcal{P}_{\delta,j}$ is irreducible. Modulo technical details that we omit for simplicity (see [45, 15]), these eigenfunctions form a complete basis and we have

$$W(z_1, z_2) = \sum_{j=0}^{\infty} \int_{\frac{d}{2} - i\infty}^{\frac{d}{2} + i\infty} \frac{d\delta}{2\pi i} \int D^{d-2} z \langle \mathcal{P}_{\delta_1}(z_1) \mathcal{P}_{\delta_2}(z_2) \mathcal{P}_{\delta,j}(z, D_w) \rangle W'_{\widetilde{\delta},j}(z, w)$$
(2.87)

for some $W'_{\delta,j}(z,w)$, $\delta = d - 2 - \delta$. Here the integration measure in z is the standard [46]¹⁹

$$D^{d-2}z = 2\frac{d^d z \delta(z^2)\theta(z^0)}{\text{vol SO}(1,1)},$$
(2.88)

and D_w is roughly equivalent to ∂_w [30] and serves to contract the indices encoded by w between W' and $\mathcal{P}_{\delta,j}$. This can be written in a form more similar to (2.82) by using the fact

$$\langle \mathcal{P}_{\delta_1}(z_1)\mathcal{P}_{\delta_2}(z_2)\mathcal{P}_{\delta,j}(z,w)\rangle = C_{\delta,j}(z_1,z_2,\partial_{z_2},\partial_{w_2})\langle \mathcal{P}_{\delta,j}(z_2,w_2)\mathcal{P}_{\delta,j}(z,w)\rangle,$$
(2.89)

for the normalization of $C_{\delta,j}$ as in (2.77) and for the standard two-point function given by¹⁸

$$\langle \mathcal{P}_{\delta,j}(z_1, w_1) \mathcal{P}_{\delta,j}(z_2, w_2) \rangle = \frac{(-2(w_1 \cdot w_2) + 2(w_1 \cdot z_2)(w_2 \cdot z_1)/(z_1 \cdot z_2))^j}{(-2z_1 \cdot z_2)^\delta}.$$
(2.90)

¹⁷These Casimir operators are only self-adjoint if $\delta_i \in \frac{d-2}{2} + i\mathbb{R}$. The forthcoming discussion is also only strictly rigorous under this condition and for square-integrable $W(z_1, z_2)$. In practice, however, usually one can analytically continue to general δ_i and work with non-square-integrable $W(z_1, z_2)$, provided certain care is taken in the process. See the discussions in [9, 10, 11] for details.

¹⁸Our conventions for tensor structures are summarized in appendix A.2.

¹⁹Upon restriction to the Poincaré section (2.74), this measure becomes simply $d^{d-2}\vec{y}$.

Using (2.89), we find²⁰

$$W(z_1, z_2) = \sum_{j=0}^{\infty} \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta}{2\pi i} C_{\delta,j}(z_1, z_2, \partial_{z_2}, \partial_{w_2}) W_{\delta,j}(z_2, w_2),$$
(2.91)

where

$$W_{\delta,j}(z_2, w_2) = \int D^{d-2} z \langle \mathcal{P}_{\delta,j}(z_2, w_2) \mathcal{P}_{\delta,j}(z, D_w) \rangle W'_{\widetilde{\delta},j}(z, w).$$
(2.92)

Using orthogonality of eigenfunctions, we can now find the inverse to (2.87) to compute W' and thus W. It is given by taking the inner product with the eigenfunctions,

$$W_{\delta,j}(z,w) \equiv \alpha_{\delta,j} \int D^{d-2} z_1 D^{d-2} z_2 \langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_2}(z_2) \mathcal{P}_{\delta,j}(z,w) \rangle W(z_1,z_2),$$
(2.93)

where the normalization coefficient has been computed, for example, in [15]

$$\alpha_{\delta,j} = \frac{(-1)^j \Gamma(j + \frac{d-2}{2}) \Gamma(d - 2 + j - \delta) \Gamma(\delta - 1)}{2\pi^{d-2} \Gamma(j + 1) \Gamma(\delta - \frac{d-2}{2}) \Gamma(\delta + j - 1)} \frac{\Gamma(\frac{\delta + j + \delta_1 - \delta_2}{2}) \Gamma(\frac{\delta + j + \delta_2 - \delta_1}{2})}{\Gamma(\frac{d - \delta + j + \delta_2 - \delta_1}{2})},$$
(2.94)

and $\widetilde{\mathcal{P}}_{\delta} \equiv \mathcal{P}_{\widetilde{\delta}} = \mathcal{P}_{d-2-\delta}$.²¹

What is the relationship between the OPE ansatz (2.82) and the integral expression (2.91)? If $W_{\delta,j}(z,w)$ is meromorphic in the right half-plane of δ , we can deform the integration contour in (2.91).²² In this way, we would obtain

$$\langle O_4(p) | \mathbb{W}_i(\infty, z, w) | O_3(p) \rangle = -\operatorname{res}_{\delta = \delta_i} W_{\delta, j_i}(z, w).$$
(2.95)

Since p, O_3 , and O_4 were arbitrary, this completely determines $\mathbb{W}_i(\infty, z, w)$ as an operator.

²⁰A subtlety is that (2.89) does not converge for all values of the coordinates, for similar reasons that the usual Euclidean OPE does not converge in all Euclidean configurations. The proper justification of this step is the same as in the case when one passes from conformal partial waves to conformal blocks in partial conformal wave expansion of a four-point function [47, 45].

²¹For $\delta_i \in \frac{d-2}{2} + i\mathbb{R}$ we have $\tilde{\delta}_i = \delta_i^*$ and (2.93) just contains the complex conjugate of the wavefunction, as usual — recall footnote 17.

²²In such deformations of the contour in the partial wave expansions one usually encounters "spurious" poles. Instead of representing physical contributions, they are purely kinematical and cancel among each other or against the possible discrete series contributions that we omitted in (2.87). See [9] for an example of cancellation of these poles in usual four-point functions, and [15] for an example of cancellation in event shapes (for low transverse spin). To simplify the discussion, in this paper we assume that the spurious poles always cancel.

2.4.3 Relation to light-ray operators Review of light-ray operators

We now show that $W_{\delta,j}(z, w)$ is indeed expected to be meromorphic, with residues related to light-ray operators. Let us first review the definition and some basic properties of these operators.

Light-ray operators [11] are primary operators

$$\widehat{\mathbb{O}}_{i,J,\lambda}^{\pm}(x,z,\mathbf{w}), \qquad (2.96)$$

parametrized by a spin J (the length of the first row of the SO(d - 1, 1) Young diagram), an SO(d - 2) representation λ (the remaining rows of the SO(d - 1, 1) Young diagram), a Regge trajectory label i, and a signature \pm . Their defining property is that they are defined for generic complex spin J and are related to local operators for certain integer values of J. Specifically, we have

$$\widehat{\mathbb{O}}_{i,J,\lambda}^{\pm}(x,z,\mathbf{w}) = \mathbf{L}[O](x,z,\mathbf{w}) \qquad (J \in \mathbb{Z}_{\geq 0} \text{ and } (-1)^J = \pm 1)$$
(2.97)

for some local operator O with SO(d - 1, 1) representation (J, λ) whenever J is a non-negative integer such that $(-1)^J = \pm 1$ (where \pm indicates the signature). That is, the operators $\widehat{\mathbb{O}}_{i,J,\lambda}^+$ are related to even-spin local operators, while $\widehat{\mathbb{O}}_{i,J,\lambda}^-$ are related to odd-spin local operators. The light-ray operators $\widehat{\mathbb{O}}_{i,J,\lambda}^{\pm}$ thus organize the local operators into continuous families or "Regge trajectories," and we label different trajectories by the index *i*.²³

In practice we construct light-ray operators from a product of two primaries O_1 and O_2 .²⁴ In this section, we are interested in light-ray operators that appear in the ϕ_1, ϕ_2 OPE, so we use $O_1 = \phi_1, O_2 = \phi_2$ to construct them. These operators all have $\lambda = 0$ since only traceless-symmetric operators appear in the OPE of two scalars, and analytic continuation in spin does not affect the second and higher rows of the

²³The simple picture of operators organized into isolated Regge trajectories is not rigorously proven. We give a discussion of its correctness in appendix A.1. Even if it is not correct, the results of this paper are mostly unchanged, but become more awkward to phrase. For simplicity of presentation, we take it as an assumption in the main text, and delegate more nuanced discussion to appendix A.1.

²⁴We expect that we get the same light-ray operators from any pair of local operators with the appropriate quantum numbers.

Young diagram. We define, following [11]

$$\mathbb{O}_{\Delta,J}^{\pm}(x,z) \equiv \pm \int_{\substack{x_1 \approx x^+ \\ x_2 \approx x}}^{\prime} d^d x_1 d^d x_2 K_{\Delta,J}^t(x_1, x_2; x, z) \phi_1(x_1) \phi_2(x_2) + \int_{\substack{x_1 \approx x \\ x_2 \approx x^+}}^{\prime} d^d x_1 d^d x_2 K_{\Delta,J}^u(x_1, x_2; x, z) \phi_2(x_2) \phi_1(x_1),$$
(2.98)

where [11]

$$K^{t}_{\Delta,J}(x_{1}, x_{2}; x, z) = \beta_{\Delta,J} \langle 0 | \widetilde{\phi}_{1}(x_{1}) \mathbf{L}[\mathcal{O}] \widetilde{\phi}_{2}(x_{2}) | 0 \rangle,$$

$$K^{u}_{\Delta,J}(x_{1}, x_{2}; x, z) = \beta_{\Delta,J} \langle 0 | \widetilde{\phi}_{2}(x_{2}) \mathbf{L}[\mathcal{O}] \widetilde{\phi}_{1}(x_{1}) | 0 \rangle, \qquad (2.99)$$

where $\langle 0| \cdots | 0 \rangle$ denote the standard tensor structures for the Wightman three-point functions as defined in appendix A.2.²⁵ The primes on the integrals indicate that we should restrict x_1 and x_2 to an arbitrary small neighborhood of the null cone of x. The dependence on this arbitrary choice will go away momentarily. We also used the notation $a \approx b$ to indicate that points a and b are spacelike, and x^+ denotes the image of x in the next Poincaré patch on the Lorentzian cylinder. The integrals are understood to be over the Lorentzian cylinder in order to preserve manifest conformal invariance. The coefficient $\beta_{\Delta,J}$ has an expression in terms of the Plancherel measure of SO(d + 1, 1), shadow coefficients, and Euclidean three-point pairings [11]. Here we will only need the explicit expression

$$\beta_{\Delta,J} = \frac{\Gamma(J+\frac{d}{2})\Gamma(d+J-\Delta)\Gamma(\Delta-1)}{2\pi^{d}\Gamma(J+1)\Gamma(\Delta-\frac{d}{2})\Gamma(\Delta+J-1)} \frac{\Gamma(\frac{\Delta+J+\Delta_{12}}{2})\Gamma(\frac{\Delta+J-\Delta_{12}}{2})}{\Gamma(\frac{d-\Delta+J+\Delta_{12}}{2})\Gamma(\frac{d-\Delta+J-\Delta_{12}}{2})}, \quad (2.100)$$

where $\Delta_{12} = \Delta_1 - \Delta_2$.

We then finally define

$$\mathbb{O}_{i,J}^{\pm}(x,z) = \operatorname{res}_{\Delta = \Delta_i(J)} \mathbb{O}_{\Delta,J}^{\pm}(x,z), \qquad (2.101)$$

where $\Delta_i(J)$ are the scaling dimensions of the *i*-th Regge trajectory. To be more precise, from the Lorentzian inversion formula the matrix elements of $\mathbb{O}_{\Delta,J}^{\pm}(x,z)$ have poles at $\Delta = \Delta_i(J)$ for integer $J \ge 0$, $(-1)^J = \pm 1$, and these poles analytically continue to $J \in \mathbb{C}$ (modulo subtleties discussed in appendix A.1). The poles come from the integration region in (2.98) where x_1 and x_2 approach the null-cone of x, and thus the residues are independent of the precise choice of integration region for the primed integrals. The operators thus defined satisfy

$$\mathbb{O}_{i,J}^{\pm}(x,z) = f_{12O} \mathbf{L}[O](x,z), \quad \text{when } J \in \mathbb{Z}_{\geq 0}, \ (-1)^J = \pm 1, \quad (2.102)$$

²⁵In particular, the structures appearing in K^t and K^u are not related to each other by a direct analytic continuation. Instead, they differ simply by the substitution $1 \leftrightarrow 2$.

for some local operator O, where f_{12O} is the OPE coefficient appearing in $\langle \phi_1 \phi_2 O \rangle$. For general $J \in \mathbb{C}$ we have

$$\mathbb{O}_{i,J}^{\pm}(x,z) = f_{12i}(J)\widehat{\mathbb{O}}_{i,J}^{\pm}(x,z), \qquad (2.103)$$

where $f_{12i}(J)$ is the analytic continuation of f_{12O} along the *i*-th Regge trajectory. Note that $\mathbb{O}_{i,J}^{\pm}(x, z)$ was constructed from a particular pair of local operators ϕ_1, ϕ_2 . The above relation provides the connection between $\mathbb{O}_{i,J}^{\pm}(x, z)$ constructed from different pairs of local operators.

While the normalized operators $\widehat{\mathbb{O}}_{i,J}^{\pm}(x, z)$ are more fundamental, for our purposes it will be more convenient to work with the operators $\mathbb{O}_{i,J}^{\pm}(x, z)$ constructed from ϕ_1, ϕ_2 .

The transverse spin puzzle

We now relate $\mathbb{W}_i(x, z, w)$ in (2.82) to the light-ray operators $\mathbb{O}_{i,J}^{\pm}(x, z)$. Consider the quantum numbers of $\mathbb{O}_{i,J}^{\pm}(x, z)$. By construction, they have scaling dimension 1 - J and a one-row Lorentz representation with spin $1 - \Delta_i(J)$. In particular, they have j = 0. This is the analytic continuation of the fact that only traceless-symmetric operators appear in the OPE of two scalars.

On the other hand, the operators $\mathbb{W}_i(x, z, w)$ have scaling dimension 2 and two-row Lorentz representations with spin $-\delta_i$ and transverse spin j_i . If $j_i = 0$, we can match quantum numbers with light-ray operators by setting J = -1 and $\delta_i = \Delta_k (J = -1) - 1$, leading us to expect

$$\mathbb{W}_{i}(x,z) \sim \mathbb{O}_{k,-1}^{+}(x,z) + \mathbb{O}_{k,-1}^{-}(x,z) \qquad \text{(for } \mathbb{W}_{i} \text{ with } j_{i} = 0\text{)}.$$
 (2.104)

Here "~" means that the left-hand side is a linear combination of objects on the right-hand side. Note that the expressions such as

$$\mathbb{W}_i(x,z) \sim \partial_x^2 \mathbb{O}_{k',1}^+(x,z), \qquad (\text{wrong}) \tag{2.105}$$

while allowed by dilatation and Lorentz symmetry, are forbidden by the full conformal symmetry, since $\mathbb{W}_i(x, z, w)$ must transform as primary operators.²⁶

A relation of the form (2.104) was proven in [15], where \mathbb{W}_i with $j_i = 0$ were called "low transverse-spin" terms. Having an expression for the low transverse-spin terms

²⁶Note that ∂_x^2 can be conformally-invariant, i.e. produce a primary, but only if it acts on a scalar operator of dimension $\frac{d-2}{2}$, which is not the case here.

is sufficient for computing certain even shapes, as explored in [15], but does not provide a complete OPE expansion.

Finding an expression for $\mathbb{W}_i(x, z, w)$ with $j_i > 0$ in terms of $\mathbb{O}_{i,J}^{\pm}(x, z)$ might seem hopeless due to the mismatch of quantum numbers. Luckily, our discussion in section 2.3 provides us with a family of conformally-invariant differential operators \mathcal{D}_n which have the property that

$$(\mathcal{D}_n \mathbb{O}_{i,J=n-1}^{\pm})(x, z, w) \tag{2.106}$$

transforms like a primary with scaling dimension 2, spin $1 - \Delta_i$ and transverse spin *n*. We can thus conjecture that for all j_i

$$\mathbb{W}_{i}(x, z, w) \sim (\mathcal{D}_{j_{i}} \mathbb{O}_{k, J=j_{i}-1}^{+})(x, z, w) + (\mathcal{D}_{j_{i}} \mathbb{O}_{k, J=j_{i}-1}^{-})(x, z, w), \qquad (2.107)$$

where $\delta_i = \Delta_k (J = j_i - 1) - 1$. (We use the convention $\mathcal{D}_0 = 1$.) This expression is similar in spirit to the wrong example (2.105) above, except that instead of ∂_x^2 we use a carefully-constructed differential operator \mathcal{D}_n that makes the right-hand side transform as a primary. As discussed in section 2.3, such operators are rare and it is an intriguing mathematical conspiracy that the quantum numbers of $\mathbb{W}_i(x, z, w)$ are precisely such that \mathcal{D}_n exist. In particular, one can verify that no other conformallyinvariant differential operators exist in the classification of [35]⁹ that can be used in the ansatz for $\mathbb{W}_i(x, z, w)$.

Partial waves and light-ray operators

As we discussed in section 2.4.2, the matrix elements of the operators $\mathbb{W}_i(x, z, w)$ are residues of $W_{\delta,j}(x, z, w)$ defined by (2.93), which we can write as (restoring *x*-dependence)

$$W_{\delta,j}(x,z,w) = \alpha_{\delta,j} \int D^{d-2} z_1 D^{d-2} z_2 \langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_2}(z_2) \mathcal{P}_{\delta,j}(z,w) \rangle \langle \mathcal{O}_4 | \mathbf{L}[\phi_1](x,z_1) \mathbf{L}[\phi_2](x,z_2) | \mathcal{O}_3 \rangle$$
(2.108)

By expanding the definition of L, this can be understood as a particular integral of

$$\langle O_4 | \phi_1(x_1) \phi_2(x_2) | O_3 \rangle$$
 (2.109)

over x_1, x_2 .

Similarly, the matrix elements of $(\mathcal{D}_j \mathbb{O}_{i,J=j-1}^{\pm})(x, z, w)$ are residues of

$$\langle O_4 | (\mathcal{D}_j \mathbb{O}_{\Delta, J=j-1}^{\pm})(x, z, w) | O_3 \rangle, \qquad (2.110)$$

which itself is a x_1, x_2 -integral of (2.109). Therefore, we can prove the relation between \mathbb{W}_i and $\mathcal{D}_j \mathbb{O}_{i,J=j-1}^{\pm}$ by simply showing that these integrals are the same. Concretely, we will show that

$$W_{\delta,j}(x,z,w) = A \langle O_4 | (\mathcal{D}_j \mathbb{O}_{\delta+1,J=j-1}^+)(x,z,w) - (\mathcal{D}_j \mathbb{O}_{\delta+1,J=j-1}^-)(x,z,w) | O_3 \rangle$$
(2.111)

for some constant A.

Let us start with rewriting (2.108) in terms of an integral of (2.109):

$$W_{\delta,j}(x,z,w) = \int d^d x_1 d^d x_2 \mathcal{L}_{\delta,j}(x_1,x_2;x,z,w) \langle O_4 | \phi_1(x_1) \phi_2(x_2) | O_3 \rangle.$$
(2.112)

By the definition of the light-transform, the kernel $\mathcal{L}_{\delta,j}$ is given by

$$\mathcal{L}_{\delta,j}(x_1, x_2; x, z, w) = \alpha_{\delta,j} \int d\alpha_1 d\alpha_2 D^{d-2} z_1 D^{d-2} z_2 \langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_2}(z_2) \mathcal{P}_{\delta,j}(z, w) \rangle \times \\ \times (-\alpha_1)^{-\delta_1 - 1} (-\alpha_2)^{-\delta_2 - 1} \delta^d (x - z_1/\alpha_1 - x_1) \delta^d (x - z_2/\alpha_2 - x_2)$$
(2.113)

To evaluate the α_i , z_i -integrals, let us first consider the more general integral

$$\int d\alpha D^{d-2} z(-\alpha)^{-\delta-1} \delta^d (x+z/\alpha) f(z).$$
(2.114)

For the *z*-integration to make sense, we need $f(\lambda z) = \lambda^{-d+2+\delta} f(z)$. Otherwise, we keep *f* arbitrary. We restrict to *x* belonging to the first Poincaré patch, in which case α must be negative in order to satisfy $x + z/\alpha = 0$. We find

$$\begin{split} &\int_{\alpha<0} d\alpha D^{d-2} z(-\alpha)^{-\delta-1} \delta^d (x+z/\alpha) f(z) \\ &= \frac{2}{\operatorname{vol} \operatorname{SO}(1,1)} \int_{\alpha>0} d\alpha d^d z \delta(z^2) \theta(z>0) \alpha^{-\delta-1} \delta^d (x-z/\alpha) f(z) \\ &= \frac{2}{\operatorname{vol} \operatorname{SO}(1,1)} \int_{\alpha>0} d\alpha \delta(\alpha^2 x^2) \theta(\alpha x>0) \alpha^{d-\delta-1} f(\alpha x) \\ &= \frac{2}{\operatorname{vol} \operatorname{SO}(1,1)} \int_{\alpha>0} d\alpha \delta(x^2) \theta(x>0) \alpha^{-1} f(x) \\ &= 2\delta(x^2) \theta(x>0) f(x), \end{split}$$
(2.115)

where we used that vol SO(1, 1) = $\int_0^\infty d\alpha \alpha^{-1} \cdot 2^7$ Using this result in (2.113), we find

$$\mathcal{L}_{\delta,j}(x_1, x_2; x_3, z, w) = 4\alpha_{\delta,j} \langle \widetilde{\mathcal{P}}_{\delta_1}(x_{13}) \widetilde{\mathcal{P}}_{\delta_2}(x_{23}) \mathcal{P}_{\delta,j}(z, w) \rangle \delta(x_{13}^2) \delta(x_{23}^2) \theta(x_{13} > 0, x_{23} > 0)$$
(2.116)

Let us now perform the same exercise for the right-hand side of (2.111). First, it is clear why we want the difference of \mathbb{O}^+ and \mathbb{O}^- operators with equal coefficients: by comparing to (2.98), we see that this difference only involves the K^t kernel and the ordering $\phi_1\phi_2$, which is precisely the ordering we need. We find

$$A\langle O_{4}|(\mathcal{D}_{j}\mathbb{O}_{\delta+1,J=j-1}^{+})(x,z,w) - (\mathcal{D}_{j}\mathbb{O}_{\delta+1,J=j-1}^{-})(x,z,w)|O_{3}\rangle$$

= $2A\int' d^{d}x_{1}d^{d}x_{2}(\mathcal{D}_{j}K_{\delta+1,j-1}^{t}\theta(x_{1}\approx x^{+},x_{2}\approx x))(x_{1},x_{2};x,z,w)\langle O_{4}|\phi_{1}(x_{1})\phi_{2}(x_{2})|O_{3}\rangle$
(2.117)

Here we have taken into account the action of \mathcal{D}_j on x in the integration limits in (2.98) by writing these limits as an explicit θ -function.

We would like to relate the kernel in (2.117) to the kernel $\mathcal{L}_{\delta,j}$ in (2.116). In doing so, there is an apparent problem: K^t is non-zero for generic configurations of points x_1, x_2, x , while the kernel $\mathcal{L}_{\delta,j}$ only contains delta-functions. This problem is resolved by two mechanisms. Firstly, when j > 0, the differential operator \mathcal{D}_j annihilates K^t for generic configurations of points. Hence, its action in (2.117) only has support on special loci — precisely the loci where $\mathcal{L}_{\delta,j}$ has support. Meanwhile, when j = 0, we have $\mathcal{D}_0 = 1$ and this mechanism does not work. However, in this case the coefficient $\beta_{\delta+1,j-1}$ in (2.99) contains a 0 coming from $\Gamma(J + 1)$ in the denominator of (2.100). The zero is only cancelled on special loci, giving rise to delta-functions again. In the next section, we describe in more detail how deltafunctions of (2.116) emerge from (2.117).

2.4.4 Emergence of delta-functions from the light-ray kernel

The light transform entering K^t in (2.99) is given by [11]

$$\langle 0 | \widetilde{\phi}_{1}(x_{1}) \mathbf{L}[O](x_{3}, z) \widetilde{\phi}_{2}(x_{2}) | 0 \rangle$$

= $L(\widetilde{\phi}_{1} \widetilde{\phi}_{2}[O]) \frac{(2z \cdot x_{23} x_{13}^{2} - 2z \cdot x_{13} x_{23}^{2})^{1 - \Delta}}{(x_{12}^{2})^{\frac{\widetilde{\Delta}_{1} + \widetilde{\Delta}_{2} - (1 - J) + (1 - \Delta)}{2}} (-x_{13}^{2})^{\frac{\widetilde{\Delta}_{12} + (1 - J) + (1 - \Delta)}{2}} (x_{23}^{2})^{\frac{-\widetilde{\Delta}_{12} + (1 - J) + (1 - \Delta)}{2}}$
(in the configuration $x_{1} \approx x_{2}, x_{1} > x_{3}, x_{2} \approx x_{3}$), (2.118)

 $^{^{27}}$ In a more formal derivation, one would fix the SO(1, 1) freedom by Faddeev-Popov procedure.

where

$$L(\widetilde{\phi}_{1}\widetilde{\phi}_{2}[O]) = -2\pi i \frac{\Gamma(\Delta + J - 1)}{\Gamma(\frac{\Delta + \Delta_{12} + J}{2})\Gamma(\frac{\Delta - \Delta_{12} + J}{2})}.$$
(2.119)

Acting with \mathcal{D}_j given in (2.66) we find

$$\langle 0 | \widetilde{\phi}_{1}(x_{1}) (\mathcal{D}_{j} \mathbf{L}[O]) \widetilde{\phi}_{2}(x_{2}) | 0 \rangle$$

$$= \frac{(-1)^{j} 2^{2j} (-J)_{j} L(\widetilde{\phi}_{1} \widetilde{\phi}_{2}[O]) (\Delta + j - 2)}{(\Delta - 2) \Gamma(j + 1)} (w \cdot x_{13} z \cdot x_{23} - w \cdot x_{23} z \cdot x_{13})^{j}$$

$$\times \frac{(2z \cdot x_{23} x_{13}^{2} - 2z \cdot x_{13} x_{23}^{2})^{1 - \Delta - j}}{(x_{12}^{2})^{\frac{\widetilde{\Delta}_{1} + \widetilde{\Delta}_{2} - (1 - J) + (1 - \Delta)}{2}} (-x_{13}^{2})^{\frac{\widetilde{\Delta}_{12} + (1 - J) + (1 - \Delta)}{2}} (x_{23}^{2})^{\frac{-\widetilde{\Delta}_{12} + (1 - J) + (1 - \Delta)}{2}}.$$

$$(2.120)$$

If we set J = j - 1 then for j > 0 this indeed vanishes due to the $(-J)_j$ factor, and thus \mathcal{D}_j with j > 0 annihilates K^t , as promised above. This can also be immediately concluded from the fact that for $J = j - 1 \ge 0$ we have above a light transform of an integer-spin three-point structure, and thus it must be annihilated by \mathcal{D}_j based on general properties of \mathcal{D}_j discussed in section 2.3.

However, this calculation of the action of \mathcal{D}_j is not complete. This is because we have only computed the action of \mathcal{D}_j on K^t in (2.117), but not on the θ -function. Derivatives hitting the theta-function will produce delta-function contributions that are needed to match (2.116). To derive the explicit form of these delta-functions, we will take a slightly more general approach.

Interlude: distributions and analytic continuation

To illustrate the approach, it is helpful to consider a toy example

$$\frac{\partial}{\partial x}\theta(x) = \delta(x). \tag{2.121}$$

While the right-hand side is obvious, let us re-derive it in a way that will be useful later.

We start with a new definition of the left-hand side. We can think of $\theta(x)$ as the case a = 0 of the 1-parameter family of distributions $p_a(x) = x^a \theta(x)$. The derivative $p'_a(x)$ is represented by a locally-integrable function when Re a > 0:

$$p'_{a}(x) = ax^{a-1}\theta(x) + x^{a}\delta(x) = ax^{a-1}\theta(x)$$
 (Re $a > 0$). (2.122)

Specifically, when Re a > 0, we can ignore $\partial/\partial x$ acting on $\theta(x)$. Let us define the left-hand side of (2.121) as the analytic continuation of $p'_a(x) = ax^{a-1}\theta(x)$ from Re a > 0 to a = 0.

As we explain in detail in appendix A.7.1, the distribution $x^{a-1}\theta(x)$ has a pole at a = 0 with residue $\delta(x)$. It follows that $\lim_{a\to 0} ax^{a-1}\theta(x) = \delta(x)$. To see this, consider the integral against a test function f(x) with a regular Taylor series expansion at x = 0,

$$\int_0^\infty dx \, ax^{a-1} f(x) = a \int_0^\infty dx \, x^{a-1} \left(f(0) + (f(x) - f(0)) \right)$$
$$= a \left(\frac{f(0)}{a} + O(1) \right)$$
$$= f(0) + O(a) \qquad (\text{Re } a > 0). \qquad (2.123)$$

In the second line, we used the fact that $\int dx x^{a-1}(f(x) - f(0))$ is O(1) as $a \to 0$, since f(x) - f(0) vanishes at least linearly there. Finally, taking $a \to 0$, we obtain f(0), as claimed.

Action of \mathcal{D}_i on K^t

This strategy of starting in a region where the derivative of a kernel is simple (in particular where we can ignore derivatives acting on θ -functions), and analytically continuing away works well in the case at hand. First, let us define the expression

$$K_{\Lambda,J}^{t}(x_{1}, x_{2}; x, z)\theta(x_{1} \approx x^{+}, x_{2} \approx x),$$
 (2.124)

as a distribution. This is straightforward if we tune the parameters $\Delta_1, \Delta_2, \Delta, J$ to a region $U \in \mathbb{C}^4$ in which all the powers in (2.118) are such that (2.124) is locally bounded (in x, x_i, z). This is the analog of choosing Re *a* sufficiently large and positive in the toy example of section 2.4.4. For $(\Delta_1, \Delta_2, \Delta, J) \in U$ we automatically get a distribution in x, x_i, z that is analytic in $\Delta_1, \Delta_2, \Delta, J$. We can then define it for the values that we are interested in by analytic continuation. In the interior of *U* we can ignore the theta-function for the purposes of acting with \mathcal{D}_j : the theta function only jumps when some x_{ij}^2 in (2.118) vanishes, but inside *U* we have a product of all x_{ij}^2 appearing with positive powers, and so K^t vanishes on the jump of the theta-function. This is the analog of the computation (2.122) in our toy example. Therefore, in order to compute the action of \mathcal{D}_j we can act in *U* just on K^t and then analytically continue to the desired values of $\Delta_1, \Delta_2, \Delta, J$.

Thus our goal is to analytically continue

$$(\mathcal{D}_{j}K_{\Delta,J}^{t})(x_{1}, x_{2}; x_{3}, z, w)\theta(x_{1} \approx x_{3}^{+}, x_{2} \approx x_{3})$$

= $\beta_{\Delta,J}\langle 0|\tilde{\phi}_{1}(x_{1})(\mathcal{D}_{j}\mathbf{L}[O])\tilde{\phi}_{2}(x_{2})|0\rangle\theta(x_{1} \approx x_{3}^{+})\theta(x_{2} \approx x_{3})$ (2.125)

with the Wightman function given in (2.120). We introduce the following coordinates,

$$s = 2z \cdot x_{23}x_{13}^2, \quad t = -2z \cdot x_{13}x_{23}^2.$$
 (2.126)

Recall that the prime on the integral in (2.117) means that we should only include the configurations where x_2 is near the future null cone of x_3 and so we can assume $z \cdot x_{23} < 0.^{28}$ We can then rewrite (2.125), assuming for simplicity that all points are in the same Poincaré patch

$$\beta_{\Delta,J} \frac{(-1)^{j} 2^{2j} (-J)_{j} L(\tilde{\phi}_{1} \tilde{\phi}_{2}[O]) (\Delta + j - 2)}{(\Delta - 2) \Gamma(j + 1)} (w \cdot x_{13} z \cdot x_{23} - w \cdot x_{23} z \cdot x_{13})^{j} \times \frac{(-2z \cdot x_{23})^{\frac{\tilde{\Delta}_{12} + (1 - J) + (1 - \Delta)}{2}} (-2z \cdot x_{13})^{\frac{-\tilde{\Delta}_{12} + (1 - J) + (1 - \Delta)}{2}}}{(x_{12}^{2})^{\frac{\tilde{\Delta}_{1} + \tilde{\Delta}_{2} - (1 - J) + (1 - \Delta)}{2}}} \frac{(s + t)^{1 - \Delta - j} \theta(s) \theta(t)}{s^{\frac{\tilde{\Delta}_{12} + (1 - J) + (1 - \Delta)}{2}} t^{\frac{-\tilde{\Delta}_{12} + (1 - J) + (1 - \Delta)}{2}}}.$$
(2.127)

The factors that are not written in terms of *s*, *t* will not be important in our analytic continuation: they are either analytic functions of the coordinates (the *w*-dependent factor), or can only become singular when x_1 becomes almost proportional to x_2 , or when x_i becomes proportional to *z*, while at the moment we are interested in more generic configurations.²⁹

To find the analytic continuation of the s, t-dependent factors, let us define

$$a = 1 - \Delta - j, \quad b = -\frac{\widetilde{\Delta}_{12} + (1 - J) + (1 - \Delta)}{2}, \quad c = -\frac{-\widetilde{\Delta}_{12} + (1 - J) + (1 - \Delta)}{2}.$$
(2.128)

Then the *s*, *t*-dependent part becomes

$$(s+t)^a s^b t^c \theta(s)\theta(t). \tag{2.129}$$

This is well-defined as a distribution analytic in a, b, c for a, b, c > 0. In appendix A.7.1 we show that near a + b + c + 2 = 0 its analytic continuation to general a, b, c has a pole of the form

$$(s+t)^{a}s^{b}t^{c}\theta(s)\theta(t) \sim \frac{1}{a+b+c+2}\frac{\Gamma(b+1)\Gamma(c+1)}{\Gamma(b+c+2)}\delta(s)\delta(t).$$
(2.130)

²⁸This is not true when x_{23} is close to being proportional to z. However, here we focus on generic configurations.

²⁹Following the logic similar to below, one can check that no new contributions appear in these limiting configurations.

With our definitions we have

$$a+b+c+2 = J - j + 1,$$
 (2.131)

which we want to set to 0. The resulting divergence is canceled in (2.127) by $\beta_{\Delta,J}$ for j = 0 or by the $(-J)_j$ factor for j > 0. Thus, plugging (2.130) into (2.127) and setting J = j - 1, we find

$$(\mathcal{D}_{j}K_{\delta+1,j-1}^{t}\theta(x_{1}\approx x_{3}^{+},x_{2}\approx x_{3}))(x_{1},x_{2};x_{3},z,w) = \frac{(-1)^{j}2^{1+2j}}{i\pi}\alpha_{\delta,j}\frac{(w\cdot x_{13}z\cdot x_{23} - w\cdot x_{23}z\cdot x_{13})^{j}}{(-2x_{13}\cdot x_{23})^{\frac{\tilde{\delta}_{1}+\tilde{\delta}_{2}+j-\delta}{2}}(-2z\cdot x_{23})^{\frac{-\tilde{\delta}_{12}+j+\delta}{2}}(-2z\cdot x_{13})^{\frac{\tilde{\delta}_{12}+j+\delta}{2}}\delta(x_{13}^{2})\delta(x_{23}^{2}),$$

$$(2.132)$$

Comparing this to (2.116) and to the expression (2.86) for the 3-point structure, which we reproduce here with the relevant quantum numbers,

$$\langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_2}(z_2) \mathcal{P}_{\delta,j}(z,w) \rangle = \frac{4^J (w \cdot z_1 z_2 \cdot z_3 - w \cdot z_2 z_1 \cdot z_3)^J}{(-2z_1 \cdot z_2)^{\frac{\tilde{\delta}_1 + \tilde{\delta}_2 - \delta + j}{2}} (-2z_1 \cdot z_3)^{\frac{\tilde{\delta}_1 + \delta - \tilde{\delta}_2 + j}{2}} (-2z_2 \cdot z_3)^{\frac{\tilde{\delta}_2 + \delta - \tilde{\delta}_1 + j}{2}}}$$
(2.133)

we find

$$(\mathcal{D}_{j}K_{\delta+1,j-1}^{t}\theta(x_{1}\approx x_{3}^{+},x_{2}\approx x))(x_{1},x_{2};x_{3},z,w) = \frac{(-1)^{j}}{2\pi i}\mathcal{L}_{\delta,j}(x_{1},x_{2};x_{3},z,w).$$
(2.134)

Thus (2.111) indeed holds with

$$A = (-1)^{j} i\pi. (2.135)$$

2.4.5 The final form of the scalar OPE

Combining equations (2.91), (2.111), and (2.135), we find the OPE formula

$$\mathbf{L}[\phi_{1}](x, z_{1})\mathbf{L}[\phi_{2}](x, z_{2})$$

$$= \pi i \sum_{j=0}^{\infty} \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta}{2\pi i} C_{\delta,j}(z_{1}, z_{2}, \partial_{z_{2}}, \partial_{w_{2}})(-1)^{j} \left((\mathcal{D}_{j}\mathbb{O}_{\delta+1, J=j-1}^{+})(x, z_{2}, w_{2}) - (\mathcal{D}_{j}\mathbb{O}_{\delta+1, J=j-1}^{-})(x, z_{2}, w_{2}) \right)$$

$$(2.136)$$

Deforming the contour to the right and picking up the poles from light-ray operators,²² we can write

$$\mathbf{L}[\phi_{1}](x, z_{1})\mathbf{L}[\phi_{2}](x, z_{2})$$

$$= -\pi i \sum_{j=0}^{\infty} \sum_{i} C_{\Delta_{i}-1, j}(z_{1}, z_{2}, \partial_{z_{2}}, \partial_{w_{2}})(-1)^{j} \left((\mathcal{D}_{j} \mathbb{O}^{+}_{i, J=j-1})(x, z_{2}, w_{2}) - (\mathcal{D}_{j} \mathbb{O}^{-}_{i, J=j-1})(x, z_{2}, w_{2}) \right)$$
(2.137)

Here the differential operator $C_{\delta,j}$ is defined by (2.89). For j = 0, taking into account $\mathcal{D}_0 = 1$, the result (2.136) agrees with (3.96) in [15], which has been tested in a number of examples [15].³⁰ For j > 0 the result is new. We will explore an example in which the j = 1 term is important in section 2.7.

Note that in the above expressions, the terms involving \mathbb{O}^+ for odd j and terms involving \mathbb{O}^- for even j are related to light-transforms of local operators and are thus annihilated by \mathcal{D}_j for j > 0. Similarly, for j = 0 we have $\mathbb{O}^-_{\Delta,J=-1} = 0$ due to the superconvergence sum rule [34, 15], which holds whenever the leading Regge trajectory in $\phi_1 \times \phi_2$ OPE has intercept below $J_1 + J_2 - 1 = -1$. We indeed have to assume that this is the case, in order for the left-hand side of (2.136) to be well-defined [34]. We thus conclude that in fact the above expressions simplify to

$$\mathbf{L}[\phi_1](x, z_1)\mathbf{L}[\phi_2](x, z_2) = \pi i \sum_{j=0}^{\infty} \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta}{2\pi i} C_{\delta,j}(z_1, z_2, \partial_{z_2}, \partial_{w_2}) (\mathcal{D}_j \mathbb{O}_{\delta+1, J=j-1}^{(-1)^j})(x, z_2, w_2), \quad (2.138)$$

and

$$\mathbf{L}[\phi_1](x,z_1)\mathbf{L}[\phi_2](x,z_2) = -\pi i \sum_{j=0}^{\infty} \sum_i C_{\Delta_i - 1,j}(z_1,z_2,\partial_{z_2},\partial_{w_2}) (\mathcal{D}_j \mathbb{O}_{i,J=j-1}^{(-1)^j})(x,z_2,w_2).$$
(2.139)

An interesting consequence of this simplification is vanishing of the commutator

$$[\mathbf{L}[\phi_1](x, z_1), \mathbf{L}[\phi_2](x, z_2)] = 0.$$
(2.140)

Indeed, according to the discussion in [15], this commutator has signature $(-1)^{J_1+J_2-1} = -1$. We claim that all the terms in (2.138) have signature [11, 15] +1 and thus there are no contributions to the commutator. To see why

$$\mathcal{D}_j \mathbb{O}_{i,J=j-1}^{(-1)^j} \tag{2.141}$$

has signature +1, we need to analyze two cases. Note that for j = 0 we have

$$\mathcal{D}_{j}\mathbb{O}_{i,J=j-1}^{(-1)^{j}} = \mathbb{O}_{i,J=-1}^{+}, \qquad (2.142)$$

which manifestly has signature +1. For j > 0 the operator \mathcal{D}_j is non-trivial, and we claim that it changes signature by $(-1)^j$, in which case we again see that

$$\mathcal{D}_{j}\mathbb{O}_{i,J=j-1}^{(-1)^{j}}$$
 (2.143)

³⁰The minus in front of \mathbb{O}^- has to do with a more explicit treatment of the analytic continuation of $(-1)^J$ factors in this paper.

has signature +1. To understand how D_j changes signature, it is convenient to use CRT symmetry \mathcal{J}_0 under which the light-ray operators have the property

$$\left(\mathcal{J}_0 \mathbb{O}_{i,J}^{\pm}(0, z, \mathbf{w}) \mathcal{J}_0^{-1}\right)^{\dagger} = \pm \mathbb{O}_{i,J}^{\pm}(0, z, \mathbf{w})$$
(2.144)

for all $z, w.^{31}$ Writing then

$$(\mathcal{D}_{j}\mathbb{O}_{i,J}^{\pm})(0, z, \mathbf{w}) \propto \mathcal{D}_{z,w;\mu_{1}}^{0+} \cdots \mathcal{D}_{z,w;\mu_{j}}^{0+} [P^{\mu_{1}}, \cdots [P^{\mu_{j}}, \mathbb{O}_{i,J}^{\pm}(0, z, \mathbf{w})] \cdots],$$
(2.145)

where $\mathcal{D}_{z,w;\mu}^{0+}$ is a differential operator in z, w defined in (2.66), and taking into account $\mathcal{J}_0 P^{\mu} \mathcal{J}_0^{-1} = -P^{\mu}$ and $P_{\mu}^{\dagger} = -P_{\mu}$, we find

$$\left(\mathcal{J}_0(\mathcal{D}_j\mathbb{O}_{i,J}^{\pm})(0, z, \mathbf{w})\mathcal{J}_0^{-1}\right)^{\dagger} = \pm (-1)^j (\mathcal{D}_j\mathbb{O}_{i,J}^{\pm})(0, z, \mathbf{w}),$$
(2.146)

and therefore \mathcal{D}_j indeed changes the signature by $(-1)^j$. Finally, note that the differential operator $C_{\delta,j}$ does not affect signature because it acts in the transverse space (i.e. on z, \mathbf{w}) which is not affected by \mathcal{J}_0 .

2.5 The complete OPE of general detectors

In the previous section, we derived the form of the light-ray OPE for a product of light-transformed scalar operators $L[\phi_1]L[\phi_2]$. We now derive a generalization for light-transforms of operators in arbitrary Lorentz representations, of which the scalar formula (2.136) is a special case. The generalized light-ray OPE formula is

Owing to its generality, this expression is a bit unwieldy, so let us unpack it. (We also give several concrete examples in section 2.6.) The left-hand side is a product of light-transforms of operators O_1 and O_2 , which have the quantum numbers $(\Delta_i, J_i, \lambda_i)$.³² The right-hand side contains two sets of terms. The first set (on line

³¹The CRT transformation \mathcal{J}_0 can be obtained from the CRT transformation \mathcal{J}_Ω described around (2.172) below by conjugating with a conformal transformation which brings the point ($x^+ = -\infty, x^- = 0, x^i = 0$) to x = 0 and the point ($x^+ = 0, x^- = -\infty, x^i = 0$) to spatial infinity.

³²For simplicity we focus on bosonic representations only.

2) contains what we refer to as the "low transverse spin" contributions, studied in detail in [15]. The second set (line 3) contains "higher transverse spin" contributions, which are new and require the use of the differential operators \mathcal{D}_n from section 2.3.

The operators $C_{\delta,\lambda}^{(a)}$ are similar to the operators $C_{\delta,j}$ from section 2.4. They encode the kinematic structure of the OPE in a fictitious (d - 2)-dimensional CFT on the celestial sphere. There are two important differences from section 2.4. Firstly, in the general case considered here, there can be several OPE coefficients that enter a given OPE, each with its own $C_{\delta,\lambda}^{(a)}$. The structure label (a) labels these different OPE coefficients. A sum over (a) is implicit in (2.147). As an example of the role of the structure label (a), recall that deforming the δ -contour in (2.147) to the right and picking up the poles, we obtain a discrete sum of terms

$$C_{\delta,\lambda}^{(a)}(z_1, \mathbf{w}_1, z_2, \mathbf{w}_2, \partial_{z_2}, \partial_{\mathbf{w}_2}) \mathbb{O}_{i,J_1+J_2-1,\lambda(a)}^{\pm}(x, z_2, \mathbf{w}_2).$$
(2.148)

Recall that $\mathbb{O}_{i,J_1+J_2-1,\lambda(a)}^{\pm}$ are built with reference to O_1, O_2 and are related to canonically-normalized light-ray operators as

$$\mathbb{O}_{i,J_1+J_2-1,\lambda(a)}^{\pm} = f_{12O^{\dagger},(a)} \widehat{\mathbb{O}}_{i,J_1+J_2-1,\lambda}^{\pm}, \qquad (2.149)$$

where $f_{12O^{\dagger},(a)}$ are analytically-continued OPE coefficients in the $O_1 \times O_2$ OPE. Equation (2.148) then becomes

$$f_{12O^{\dagger},(a)}C^{(a)}_{\delta,\lambda}(z_1,\mathbf{w}_1,z_2,\mathbf{w}_2,\partial_{z_2},\partial_{\mathbf{w}_2})\widehat{\mathbb{O}}^{\pm}_{i,J_1+J_2-1,\lambda}(x,z_2,\mathbf{w}_2), \qquad (2.150)$$

Thus, the index (a) of $C_{\delta,\lambda}^{(a)}$ is naturally contracted with OPE coefficients. Similar statements hold for higher transverse spin terms as well.

The second distinction has to do with the normalization of $C_{\delta,\lambda}^{(a)}$. The operators $C_{\delta,j}$ were normalized by equation (2.89), which is formulated in terms of celestial three-point structures. On the other hand, the OPE coefficients $f_{12O^{\dagger},(a)}$ are defined in terms of *d*-dimensional tensor structures, and thus the normalization condition (2.89) is only correct due to our specific choice of conventions for both *d*-dimensional as well as celestial tensor structures. The main result of this section will be the proof of (2.147) together with the simple, convention-independent, equations (2.214) and (2.215) that determine the $C_{\delta,\lambda}^{(a)}$ in terms of *d*-dimensional data.³³ We refer to these formulas as "celestial map formulas" because they map

³³Note that up to an action by an invertible matrix on the index (*a*), these operators are completely fixed by the (d - 2)-dimensional conformal symmetry (equivalently, *d*-dimensional Lorentz symmetry). These equations thus simply determine a preferred basis of these operators.

the *d*-dimensional three-point tensor structures (which naturally pair with the OPE coefficients $f_{120^{\dagger},(a)}$) to celestial sphere differential operators $C_{\delta,\lambda}^{(a)}$.³⁴

The remaining notation in (2.147) has to do with quantum numbers of exchanged operators and the respective selection rules. In the low transverse spin terms we are summing over transverse spins $\lambda \in \Lambda_{12}$ which simply means all the transverse spins that appear in the usual local OPE $O_1 \times O_2$ and the celestial OPE of operators with (d - 2)-dimensional spins $\lambda_1 \times \lambda_2$. These are the transverse spins for which the operators $\mathbb{O}_{i,J,\lambda,(a)}^{\pm}$ can be constructed from O_1 and O_2 and for which the operators $C_{\delta,\lambda}^{(a)}$ make sense.

As an example, when O_1, O_2 are scalars, Λ_{12} contains only the trivial representation of SO(d - 2) because only traceless-symmetric operators appear in OPE of ddimensional scalars. In this case, the constraint that $\lambda \in \Lambda_{12}$ should appear in the OPE of (d - 2) dimensional scalars is trivially satisfied. On the other hand, in the example of energy-energy OPE discussed in section 2.6 both constraints become non-trivial.

In the higher transverse spin terms, we sum over $\lambda_{\gamma}, \gamma \in \Gamma_{12}$. These are the transverse spins for which the action of the operators \mathcal{D}_n is well-defined. Concretely, these are transverse spins with the first row of the SO(d - 2) Young diagram (the second row of the SO(d - 1, 1) Young diagram) of length $J_1 + J_2$. We write $\lambda_{\gamma} = (J_1 + J_2, \gamma)$ with γ an SO(d - 4) irrep. Since the action of \mathcal{D}_n raises transverse spin, we use the OPE differential operator $C_{\delta,\lambda_{\gamma}(+n)}^{(a)}$, where $\lambda_{\gamma}(+n) = (J_1 + J_2 + n, \gamma)$. The set Γ_{12} consists of γ for which transverse spin λ_{γ} appears in *d*-dimensional $O_1 \times O_2$ OPE and at the same time $\lambda_{\gamma}(+n)$ appears in the (d - 2)-dimensional celestial OPE. (As explained in appendix A.5, these two conditions are in fact equivalent.)

Before proceeding with the derivation, let us comment again on the relation of this section to [15]. In [15], the celestial map formula for lower transverse spin was derived by a rather non-trivial procedure using the Lorentzian inversion formula. In this section, we will give a much simpler derivation of the celestial map formula for both lower and higher transverse spin. The drawback of this simpler derivation is that it is based on the assumption that the light-ray kernel $K_{\Delta,J,\lambda(a)}^t(x_1, x_2; x, z)$ satisfying (2.152) localizes on the null cone of x as we set J to certain values and possibly act with \mathcal{D}_n , analogously to the scalar case in section 2.4. This assumption is plausible in the sense that it gives a natural generalization of the pattern observed

³⁴As we explain below, it will sometimes happen that $C_{\delta,\lambda}^{(a)}$ vanishes for some values of *a*, i.e. not all of the OPE coefficients actually appear in (2.148).

in concrete examples, and is furthermore purely kinematical. It therefore appears to be a purely technical problem to prove it. (See sections 2.5.2 and 2.5.2 for the precise statement of our assumptions and the supporting evidence.)

For completeness, in appendix A.4, we derive (2.147) by generalizing the derivation in [15]. This derivation, although being more technical than the one in sections 2.5.2 and 2.5.2 (which is why it is relegated to an appendix), does not rely on the assumptions discussed above.

2.5.1 A formula for the light-ray operator kernel

Our starting point is the following intrinsically-Lorentzian description of light-ray operators. Recall that the light-ray operators are defined by the integral [11]

$$\mathbb{O}_{i,J,\lambda(a)}^{\pm}(x,z) = \operatorname{res}_{\Delta=\Delta_{i}} \pm \int_{\substack{x\approx 2\\x\approx 1^{-}}} d^{d}x_{1}d^{d}x_{2}K_{\Delta,J,\lambda(a)}^{t}(x_{1},x_{2};x,z)O_{1}O_{2} + \int_{\substack{x\approx 1\\x\approx 2^{-}}} d^{d}x_{1}d^{d}x_{2}K_{\Delta,J,\lambda(a)}^{u}(x_{2},x_{1};x,z)O_{2}O_{1}.$$
 (2.151)

We claim that the kernel K^t is determined by the equation

$$\int_{\substack{2 > x' > 1^{-} \\ x \approx 2, 1^{-}}} \frac{d^{d}x_{1} d^{d}x_{2}}{\operatorname{vol}(\mathrm{SO}(1, 1))^{2}} K^{t}_{\Delta, J, \lambda(a)}(x_{1}, x_{2}; x, z) \langle 0 | O_{2} \mathbf{L}[O^{\dagger}](x', z') O_{1} | 0 \rangle_{+}^{(b)}$$

$$= \frac{1}{2\pi i} \langle \mathbf{L}[O](x, z) \mathbf{L}[O^{\dagger}](x', z') \rangle \delta^{(b)}_{(a)}, \qquad (2.152)$$

together with the condition that it has the analyticity and conformal transformation properties of

$$\langle 0|\widetilde{O}_1^{\dagger}(x_1)O^L(x,z)\widetilde{O}_2^{\dagger}(x_2)|0\rangle, \qquad (2.153)$$

where O^L has the quantum numbers of L[O]. The kernel K^u is defined similarly and will be described below.³⁵ In (2.151) and in what follows we keep the transverse indices (encoded previously by polarization vectors **w**) implicit in order to avoid excessive clutter in the notation. The way these indices are contracted should in all cases be clear from the context.

³⁵An intuitive picture behind (2.152) is as follows. The Wightman three-point structure in (2.152), together with the condition $2 > x' > 1^-$, can be viewed as one of two parts of the light-transform of a time-ordered three-point structure [11]. The integration against K^t is then similar to producing L[O] from O_1, O_2 inside of this time-ordered three-point function, which should be equal to the two-point function in the right-hand side of (2.152). It would be interesting if this intuitive reasoning could be made precise: our derivation of (2.152) is based on the generalized Lorentzian inversion formula of [11], where the latter is derived from Euclidean harmonic analysis. It would be instructive to bypass the Euclidean argumentation altogether.

Above, we use the notation that O_i is at point x_i . The kernel K^t carries Lorentz indices for points x_1, x_2 that are in the dual representations to O_1, O_2 , and these indices are contracted with O_1, O_2 in (2.151) and (2.152). We prove (2.152) in appendix A.3. An advantage of (2.152) relative to the definition in [11] is that it makes reference only to Lorentzian objects. By contrast, the definition in [11] includes the Plancherel measure for the Euclidean conformal group and Euclidean shadow coefficients, and is thus more awkward to use in a purely Lorentzian setting.

In order to use (2.152) for non-integer *J*, it remains to explain the meaning of the objects

$$\langle \mathbf{L}[O](x,z)\mathbf{L}[O^{\dagger}](x',z')\rangle, \qquad (2.154)$$

$$\langle 0|O_2 \mathbf{L}[O^{\dagger}](x',z')O_1|0\rangle_+^{(b)}$$
 (2.155)

for such J.

Analytic continuation in spin

We define (2.154) and (2.155) by extending the definition of the underlying structures

$$\langle O(x,z)O^{\dagger}(x',z')\rangle, \qquad (2.156)$$

$$\langle 0|O_2O^{\dagger}(x',z')O_1|0\rangle_{+}^{(b)}$$
 (2.157)

to non-integer J and then taking the necessary light-transforms.

Note that fixing the normalization of local operators and setting the conventions for three-point structures of local operators involves specifying the expressions for

$$\langle O(x,z)O^{\dagger}(x',z')\rangle,$$
 (2.158)

(1)

$$\langle O_1(x_1, z_1) O_2(x_2, z_2) O^{\dagger}(x, z) \rangle^{(b)}.$$
 (2.159)

for integer spin J. We will now impose certain constraints on these choices which will simplify our general analysis with regard to factors of $(-1)^J$ which have ambiguous analytic continuation in J.

For the two-point function, note that for any fixed $n \in \mathbb{Z}$

$$\langle O(x_1, z_1) O^{\dagger}(x_2, z_2) \rangle (-2z_1 \cdot I(x_{12}) \cdot z_2)^{n-J}$$
 (2.160)

is conformally-invariant with quantum numbers independent of J. Moreover, for sufficiently large n, the Lorentz weights with which it transforms are dominant³⁶ and

³⁶I.e. the first row of Young diagram is at least as long as the second one.

we can take this structure to be equal to a fixed, *J*-independent, two-point function of local operators

$$\langle O(x_1, z_1) O^{\dagger}(x_2, z_2) \rangle (-2z_1 \cdot I(x_{12}) \cdot z_2)^{n-J} = f_0(x_1, z_1; x_2, z_2).$$
 (2.161)

So we find that one can always choose

$$\langle O(x_1, z_1) O^{\dagger}(x_2, z_2) \rangle = f_0(x_1, z_1; x_2, z_2) (-z_1 \cdot I(x_{12}) \cdot z_2)^{J-n}.$$
 (2.162)

We allow to modify this convention by exponential factors such as 2^{J} but not by $(-1)^{J}$.

Note that the expression (2.162) can be rewritten as

$$f_0(x_1, z_1; x_2, z_2)(-z_1 \cdot I(x_{12}) \cdot z_2)^{J-n}$$

= $f_0(x_1, z_1; x_2, z_2)(2(z_1 \cdot x_{12})(z_2 \cdot x_{12}) - (z_1 \cdot z_2)x_{12}^2)^{J-n}(x_{12}^2)^{n-J}.$ (2.163)

Using the fact that $2(z_1 \cdot x_{12})(z_2 \cdot x_{12}) - (z_1 \cdot z_2)x_{12}^2 > 0$ for generic configurations,³⁷ we see that the usual time-ordered $i\epsilon$ prescription unambiguously defines the time-ordered two-point function (2.162) for generic configurations of x_i, z_i . This is sufficient to apply the light-transforms in (2.154).

A similar argument for $\langle O_1(x_1, z_1)O_2(x_2, z_2)O^{\dagger}(x, z)\rangle^{(b)}$ shows that we can write, for sufficiently large integer *J*,

$$\langle O_1(x_1, z_1)O_2(x_2, z_2)O^{\dagger}(x_3, z_3)\rangle^{(b)} = f_0^{(b)}(x_1, z_1; x_2, z_2; x_3, z_3) \frac{(2z \cdot x_{23} x_{13}^2 - 2z \cdot x_{13} x_{23}^2)^{J-n}}{x_{12}^{J-n} x_{23}^{J-n} x_{13}^{J-n}}$$
(2.164)

where *n* is sufficiently large so that the Lorentz weights of f_0 are dominant, and the basis $f_0^{(b)}(x_1, z_1; x_2, z_2; x_3, z_3)$ can be chosen to be *J*-independent. There are two possible subtleties here. Firstly, the operators O_1 and O_2 can be identical, in which case permutation invariance will typically constrain the structures for even and odd *J* differently. Secondly, some of O_1, O_2 can be conserved currents, in which case the conservation constraints will typically require some non-trivial polynomial dependence of $f_0^{(b)}$ on *J*. In both cases, our solution is to use the generic basis of structures, ignoring these constraints. The OPE coefficients computed in this basis may satisfy some linear equations, but this will not affect any of our arguments.

³⁷To see this, note that one can always write $x_{12} = \alpha z_1 + \beta z_2 + x_{\perp}$ for $x_{\perp} \cdot z_i = 0$. We have then $2(z_1 \cdot x_{12})(z_2 \cdot x_{12}) - (z_1 \cdot z_2)x_{12}^2 = (-z_1 \cdot z_2)x_{\perp}^2 > 0$, where we used that x_{\perp} is spacelike due to being orthogonal to the timelike vector $z_1 + z_2$.

The problem of defining three-point structures for non-integer *J* is complicated by the fact that $2z \cdot x_{23} x_{13}^2 - 2z \cdot x_{13} x_{23}^2$ is in general not sign-definite [11]. We will define the analytic continuation directly for $\langle 0|O_2O^{\dagger}(x_3, z_3)O_1|0\rangle^{(b)}$ in the configuration $2 > 3, 1 \approx 3, 1 \approx 2$, where we have

$$2z \cdot x_{23} x_{13}^2 - 2z \cdot x_{13} x_{23}^2 < 0.$$
 (2.165)

For integer J we have

$$\langle 0|O_2O^{\dagger}(x_3, z_3)O_1|0\rangle^{(b)} = (-1)^J (-2z \cdot x_{23}x_{13}^2 + 2z \cdot x_{13}x_{23}^2)^J \times (\cdots), \quad (2.166)$$

where the dots represent the standard analytic continuation of all the other factors. For general $J \in \mathbb{C}$ we define

$$\langle 0|O_2O^{\dagger}(x_3, z_3)O_1|0\rangle_{\pm}^{(b)} = \pm (-2z \cdot x_{23}x_{13}^2 + 2z \cdot x_{13}x_{23}^2)^J \times (\cdots).$$
 (2.167)

In all other configurations these Wightman functions are determined by the usual analytic continuation, which is unambiguous for this ordering even for $J \in \mathbb{C}$ [11]

We will additionally use the "time-ordered" structures, which are defined to be equal to

$$\langle O_1(x_1, z_1) O_2(x_2, z_2) O^{\dagger}(x_3, z_3) \rangle_{+}^{(b)} = f_0(x_1, z_1; x_2, z_2; x_3, z_3) \frac{(2z \cdot x_{23} x_{13}^2 - 2z \cdot x_{13} x_{23}^2)^{-n}}{x_{12}^{J-n} x_{23}^{J-n} x_{13}^{J-n}} \\ \times |2z \cdot x_{23} x_{13}^2 - 2z \cdot x_{13} x_{23}^2|^J$$
(2.168)

for spacelike-separated points and are defined in other configurations by usual timeordered $i\epsilon$ -prescriptions applied to everything except $|\cdots|^J$. These structures are useful because

$$\langle O_1 O_2 \mathbf{L}[O^{\dagger}] \rangle_{+}^{(b)} = \langle 0 | O_2 \mathbf{L}[O^{\dagger}] O_1 | 0 \rangle_{+}^{(b)} \theta (2 > 3 > 1^{-}) + (\cdots) \theta (1 > 3 > 2^{-}),$$
(2.169)

similarly to the relation between the light-transform of integer-spin time-ordered and Wightman correlators [11]. Here dots represent a structure that is a bit awkward to describe but which we will not need in what follows.

Our choice of conventions for traceless-symmetric operators described in appendix A.2 satisfies the above constraints.

Definition of *K^u*

The kernel K^u is defined by requiring that it has the analyticity and conformal transformation properties of

$$\langle 0|\widetilde{O}_{2}^{\dagger}(x_{2})O^{L}(x,z)\widetilde{O}_{1}^{\dagger}(x_{1})|0\rangle, \qquad (2.170)$$

and the requirement that when we set $x = -\infty z$ and $z^0 = z^1 = 1$, $z^i = 0$, the following equality holds

$$\int_{\substack{x \approx 1 \\ x \approx 2^{-}}} d^d x_1 d^d x_2 K^u_{\Delta, J, \lambda(a)}(x_2, x_1; x, z) O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \approx 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \propto 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \propto 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \propto 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_1 = \overline{\int_{\substack{x \propto 2 \\ x \approx 1^{-}}} d^d x_1 d^d x_2 K^t_{\Delta, J, \lambda(a)}(x_1, x_2; x, z) O_1 O_2 O_2 O_1 = \overline{\int_{\substack{x \propto 2 \\ x \approx 1^{-}}} d^d x_2 d$$

Here, we have defined the linear operation

$$\overline{A} = (\mathcal{J}_{\Omega} A \mathcal{J}_{\Omega}^{-1})^{\dagger}.$$
(2.172)

Here \mathcal{J}_{Ω} is the anti-unitary operator implementing the CRT symmetry, where the reflection acts as $x^1 \rightarrow -x^1$. This relation should hold for all *J*. For integer *J* it is equivalent to

$$\int_{\substack{1 > x' > 2^{-} \\ x \approx 1, 2^{-}}} \frac{d^{d}x_{1} d^{d}x_{2}}{\operatorname{vol}(\mathrm{SO}(1, 1))^{2}} K^{u}_{\Delta, J, \lambda(a)}(x_{1}, x_{2}; x, z) \langle 0 | O_{1} \mathbf{L}[O^{\dagger}](x', z') O_{2} | 0 \rangle^{(b)}$$

= $\frac{1}{2\pi i} \langle \mathbf{L}[O](x, z) \mathbf{L}[O^{\dagger}](x', z') \rangle \delta^{(b)}_{(a)}.$ (2.173)

2.5.2 Derivation of the OPE formula

Harmonic analysis on celestial sphere

We have given a review of harmonic analysis on celestial sphere in section 2.4.2. In this section we need a generalization which we state here without proof, and refer the reader instead to [15].

Suppose we have a choice of two- and three-point structures on the celestial sphere

$$\langle \mathcal{P}_{\delta,\lambda}(z)\mathcal{P}^{\dagger}_{\delta,\lambda}(z')\rangle$$
 (2.174)

$$\langle \mathcal{P}_{\delta_1,\lambda_1}(z_1)\mathcal{P}_{\delta_2,\lambda_2}(z_2)\mathcal{P}^{\dagger}_{\delta,\lambda}(z)\rangle^{(a)}$$
(2.175)

We assume that the three-point structures are linearly-independent and span the space of conformally-invariant tensor structures for the given quantum numbers. Given these structures, we can find OPE differential operators $\widehat{C}_{\delta,\lambda}^{(a)}$ satisfying

$$\widehat{C}_{\delta,\lambda}^{(a)}(z_1, z_2; \partial_{z_2}) \langle \mathcal{P}_{\delta,\lambda}(z_2) \mathcal{P}_{\delta,\lambda}^{\dagger}(z') \rangle = \langle \mathcal{P}_{\delta_1,\lambda_1}(z_1) \mathcal{P}_{\delta_2,\lambda_2}(z_2) \mathcal{P}_{\delta,\lambda}^{\dagger}(z) \rangle^{(a)}.$$
(2.176)

Here and below, the $\mathcal{P}_{\delta,\lambda}$ carry SO(d-2) indices for λ , which we suppress for brevity. These indices are implicitly contracted between $\widehat{C}_{\delta,\lambda}^{(a)}$ and $\mathcal{P}_{\delta,\lambda}(z_2)$.

Now suppose that conformally-invariant kernels $k_{\delta,\lambda,(a)}(z_1, z_2; z)$ solve the equation

$$\int \frac{D^{d-2}z_1 D^{d-2} z_2}{\operatorname{vol}(\mathrm{SO}(1,1))} k_{\delta,\lambda,(a)}(z_1, z_2; z) \langle \mathcal{P}_{\delta_1,\lambda_1}(z_1) \mathcal{P}_{\delta_2,\lambda_2}(z_2) \mathcal{P}_{\delta,\lambda}^{\dagger}(z') \rangle^{(b)} = \langle \mathcal{P}_{\delta,\lambda}(z) \mathcal{P}_{\delta,\lambda}^{\dagger}(z') \rangle \delta_{(a)}^{(b)}$$

$$(2.177)$$

Here, $k_{\delta,\lambda,(a)}(z_1, z_2; z)$ carries SO(d - 2) indices dual to $\mathcal{P}_{\delta_1,\lambda_1}$ and $\mathcal{P}_{\delta_2,\lambda_2}$ and these indices are implicitly contracted in (2.177). If the above conditions hold for all λ 's that can appear in the (d - 2)-dimensional OPE $\mathcal{P}_{\delta_1,\lambda_1} \times \mathcal{P}_{\delta_2,\lambda_2}$, then we have

$$\mathbf{L}[O_{1}](x,z_{1})\mathbf{L}[O_{2}](x,z_{2}) = \sum_{\lambda} \int_{\frac{d-2}{2}-i\infty}^{\frac{d-2}{2}+i\infty} \frac{d\delta}{2\pi i} \widehat{C}_{\delta,\lambda}^{(a)}(z_{1},z_{2};\partial_{z_{2}}) \mathbb{W}_{\delta,\lambda,(a)}(x,z_{2}),$$
(2.178)

where

$$\mathbb{W}_{\delta,\lambda,(a)}(x,z) \equiv \int D^{d-2} z_1 D^{d-2} z_2 k_{\delta,\lambda,(a)}(z_1,z_2;z) \mathbf{L}[O_1](x,z_1) \mathbf{L}[O_2](x,z_2).$$
(2.179)

For the validity of (2.178) the same caveats as in footnote 20 apply. Note that it is not important which basis of structures one chooses in (2.175) and (2.174) — the above statement is basis-independent. We utilize this freedom below, making convenient choices when appropriate.

Lower transverse spin

For lower transverse spin, we will assume that when we set $J \rightarrow J_1 + J_2 - 1$, the light-ray kernel $K^t_{\Delta,J,\lambda(a)}$ degenerates to

$$\mathcal{N}_{(c),\delta,\lambda}^{(a)} \int_{\substack{x \approx 2\\x \approx 1^{-}}} d^{d}x_{1} d^{d}x_{2} K_{\delta+1,J,\lambda,(a)}^{t}(x_{1},x_{2};x,z) O_{1}(x_{1}) O_{2}(x_{2})$$

$$\rightarrow \int D^{d-2} z_{1} D^{d-2} z_{2} k_{\delta,\lambda,(c)}(z_{1},z_{2};z) \mathbf{L}[O_{1}](x,z_{1}) \mathbf{L}[O_{2}](x,z_{2}) \qquad (2.180)$$

where $k_{\delta,\lambda,(c)}$ is a basis of Lorentz-invariant kernels, and $\mathcal{N}_{(c),\delta,\lambda}^{(a)}$ is a rectangular matrix (implicit summation over (*a*) is understood). The matrix \mathcal{N} is needed because not every $K_{\delta+1,J,\lambda,(a)}^t$ degenerates in this way. We assume that all Lorentz-invariant kernels $k_{\delta,\lambda,(c)}$ can be generated in this way i.e. \mathcal{N} is full rank.

The evidence for this assumption comes from several lines of reasoning. First of all, it agrees with the scalar example in section 2.4. Moreover, we have additionally studied the example of $\mathbf{L}[\phi]\mathbf{L}[V]$ OPE, where *V* is an operator with spin 1, which has a non-trivial matrix \mathcal{N} and verified this statement. Furthermore, from the counting of structures in appendix A.5 one can see that the dimensions of the spaces of structures match (specifically, the rank of \mathcal{N} can be predicted from (2.186) below, and it matches the number of $k_{\delta,\lambda,(c)}$). Finally, (2.180) yields the same results as in [15] and in appendix A.4, where they have been obtained by an different, although more complicated, method. It would be interesting to find a direct derivation of (2.180).

Plugging (2.180) into (2.152), we obtain

$$\int D^{d-2} z_1 D^{d-2} z_2 k_{\delta,\lambda(c)}(z_1, z_2; z) \frac{\langle 0 | \mathbf{L}^+[\mathcal{O}_2](x, z_2) \mathbf{L}[\mathcal{O}^\dagger](x', z') \mathbf{L}^-[\mathcal{O}_1](x, z_1) | 0 \rangle_+^{(a)}}{\operatorname{vol}(\mathrm{SO}(1, 1))^2}$$

= $\frac{1}{2\pi i} \langle \mathbf{L}[\mathcal{O}](x, z) \mathbf{L}[\mathcal{O}^\dagger](x', z') \rangle \mathcal{N}_{(c),\delta,\lambda}^{(a)}.$ (2.181)

where $L^+[O_2]$ indicates that the light transform contour is restricted to 2 > x' and $L^-[O_1]$ indicates that the light transform contour is restricted to $x' > 1^-$. On the left-hand side, one factor of vol(SO(1, 1)) cancels against a zero-mode in the integral over z_1, z_2 , and the other factor cancels against a zero-mode in the triple light-transform of the three-point structure.

Let us define a matrix $\mathcal{M}_{(a),\delta,\lambda}^{(c)}$ with the property

$$\mathcal{M}_{(a),\delta,\lambda}^{(c)} \mathcal{N}_{(c'),\delta,\lambda}^{(a)} = \delta_{(c')}^{(c)}.$$
(2.182)

This is always possible to do since N is full rank. However, M defined in this way is in general ambiguous because we can replace

$$\mathcal{M}_{(a),\delta,\lambda}^{(c)} \to \mathcal{M}_{(a),\delta,\lambda}^{(c)} + m_{(a)}^{(c)}$$
(2.183)

for any $m_{(a)}^{(c)}$ such that $m_{(a)}^{(c)} \mathcal{N}_{(c'),\delta,\lambda}^{(a)} = 0$. We make a choice of \mathcal{M} and rewrite (2.181) as

$$\int D^{d-2} z_1 D^{d-2} z_2 k_{\delta,\lambda(c)}(z_1, z_2; z) \frac{\langle 0 | \mathbf{L}^+[\mathcal{O}_2](x, z_2) \mathbf{L}[\mathcal{O}^\dagger](x', z') \mathbf{L}^-[\mathcal{O}_1](x, z_1) | 0 \rangle_+^{(a)}}{\operatorname{vol}(\mathrm{SO}(1, 1))^2} \mathcal{M}_{(a),\delta,\lambda}^{(c')}$$

$$= \frac{1}{2\pi i} \langle \mathbf{L}[\mathcal{O}](x, z) \mathbf{L}[\mathcal{O}^\dagger](x', z') \rangle \delta_{(c)}^{(c')}. \qquad (2.184)$$

In the left-hand side we have a celestial "bubble integral" of $k_{\delta,\lambda(c)}(z_1, z_2; z)$ against

$$\frac{\langle 0|\mathbf{L}^{+}[O_{2}](x,z_{2})\mathbf{L}[O^{\dagger}](x',z')\mathbf{L}^{-}[O_{1}](x,z_{1})|0\rangle_{+}^{(a)}}{\operatorname{vol}(\mathrm{SO}(1,1))}\mathcal{M}_{(a),\delta,\lambda}^{(c')}.$$
(2.185)

Since $k_{\delta,\lambda(c)}(z_1, z_2; z)$ form a basis, it follows that this bubble integral is nondegenerate [45], i.e. it vanishes for all (*c*) if and only if the above expression vanishes. Due to (2.183) it then follows that

$$\frac{\langle 0|\mathbf{L}^{+}[O_{2}](x,z_{2})\mathbf{L}[O^{\dagger}](x',z')\mathbf{L}^{-}[O_{1}](x,z_{1})|0\rangle_{+}^{(a)}}{\operatorname{vol}(\mathrm{SO}(1,1))}m_{(a)}^{(c)} = 0$$
(2.186)

for any $m_{(a)}^{(c)}$ such that $m_{(a)}^{(c)} \mathcal{N}_{(c'),\delta,\lambda}^{(a)} = 0.$

Equation (2.184) has almost the same structure as (2.177) if we identify, for fixed x, x',

$$\langle \mathcal{P}_{\delta_{1},\lambda_{1}}(z_{1})\mathcal{P}_{\delta_{2},\lambda_{2}}(z_{2})\mathcal{P}_{\delta,\lambda}^{\dagger}(z)\rangle^{(c)} \leftrightarrow \frac{\langle 0|\mathbf{L}^{+}[\mathcal{O}_{2}](x,z_{2})\mathbf{L}[\mathcal{O}^{\dagger}](x',z')\mathbf{L}^{-}[\mathcal{O}_{1}](x,z_{1})|0\rangle_{+}^{(a)}}{\operatorname{vol}(\mathrm{SO}(1,1))} \mathcal{M}_{(a),\delta,\lambda}^{(c)},$$

$$(2.187)$$

$$\langle \mathcal{P}_{\delta,\lambda}(z)\mathcal{P}_{\delta,\lambda}^{\dagger}(z')\rangle \leftrightarrow \frac{1}{2\pi i} \langle \mathbf{L}[\mathcal{O}](x,z)\mathbf{L}[\mathcal{O}^{\dagger}](x',z')\rangle.$$

$$(2.188)$$

However, we cannot quite write equality signs in the relations above, because while z_1, z_2 and the transverse indices of $\mathbf{L}^+[O_2], \mathbf{L}^-[O_1]$ transform under the Lorentz group at x, the polarization z' and transverse indices of $\mathbf{L}[O^{\dagger}]$ transform under the Lorentz group at x'. Fortunately, we can perform two steps to rectify this problem. First, there is a subgroup $L_{xx'} \simeq SO(d-1, 1)$ of conformal transformations that fix both x and x', and act as Lorenz transformations locally at these points.³⁸ Second, we can "translate" the indices and polarizations of $\mathbf{L}[O^{\dagger}]$ from x' to x, so that the action of $L_{xx'}$ is the same on all indices and polarizations. Specifically, we can write

$$\langle \mathcal{P}_{\delta_{1},\lambda_{1}}(z_{1})\mathcal{P}_{\delta_{2},\lambda_{2}}(z_{2})\mathcal{P}_{\delta,\lambda}^{\dagger}(z)\rangle^{(c)} = \frac{\langle 0|\mathbf{L}^{+}[\mathcal{O}_{2}](x,z_{2})(I_{xx'}\mathbf{L}[\mathcal{O}^{\dagger}])(x',I_{xx'}z')\mathbf{L}^{-}[\mathcal{O}_{1}](x,z_{1})|0\rangle_{+}^{(a)}}{\operatorname{vol}(\mathrm{SO}(1,1))}\mathcal{M}_{(a),\delta,\lambda}^{(c)},$$
(2.189)

$$\langle \mathcal{P}_{\delta,\lambda}(z)\mathcal{P}_{\delta,\lambda}^{\dagger}(z')\rangle = \frac{1}{2\pi i} \langle \mathbf{L}[O](x,z)(I_{xx'}\mathbf{L}[O^{\dagger}])(x',I_{xx'}z')\rangle.$$
(2.190)

where

$$(I_{xx'})^{\mu}{}_{\nu} = \delta^{\mu}{}_{\nu} - 2x^{\mu}x_{\nu}/x^2, \qquad (2.191)$$

and $I_{xx'}\mathbf{L}[O^{\dagger}]$ indicates the action of *I* on the transverse spin indices. Importantly, the presence of $I_{xx'}$ means that while z' and other indices now transform in the

³⁸For example, we can set x = 0 and $x' = \infty$, in which case the group in question is the standard Lorentz group.

same way under $L_{xx'}$ as the indices and polarizations at x, the representation that they form is reflected relative to that of $L[O^{\dagger}]$. This reflection is crucial, because it is precisely the difference between the d-dimensional and (d-2)-dimensional \dagger operation, and so the representations match in the above identifications.

Equation (2.184) implies that the set of three-point structures defined by (2.189) is complete, otherwise we wouldn't be able to get the Kronecker delta in the right-hand side of (2.184). We can therefore use the statement of section 2.5.2. Specifically, let us define operators $\widehat{C}_{\delta,\lambda}^{(a)}$ by (2.176), which now takes the form

$$\frac{1}{2\pi i} \widehat{C}_{\delta,\lambda}^{(c)}(z_1, z_2, \partial_{z_2}) \langle \mathbf{L}[O](x, z_2) (I_{xx'} \mathbf{L}[O^{\dagger}])(x', I_{xx'}z') \rangle
= \frac{\langle 0 | \mathbf{L}^+[O_2](x, z_2) (I_{xx'} \mathbf{L}[O^{\dagger}])(x', I_{xx'}z') \mathbf{L}^-[O_1](x, z_1) | 0 \rangle_+^{(a)}}{\operatorname{vol}(\mathrm{SO}(1, 1))} \mathcal{M}_{(a),\delta,\lambda}^{(c)}.$$
(2.192)

In this equation, we can cancel the $I_{xx'}$ matrices on both sides

$$\frac{1}{2\pi i} \widehat{C}_{\delta,\lambda}^{(c)}(z_1, z_2, \partial_{z_2}) \langle \mathbf{L}[O](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \rangle = \frac{\langle 0 | \mathbf{L}^+[O_2](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \mathbf{L}^-[O_1](x, z_1) | 0 \rangle_+^{(a)}}{\operatorname{vol}(\mathrm{SO}(1, 1))} \mathcal{M}_{(a)}^{(c)}(z_1, z_2, \partial_{z_2}) \langle \mathbf{L}[O](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \rangle = \frac{\langle 0 | \mathbf{L}^+[O_2](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \mathbf{L}^-[O_1](x, z_1) | 0 \rangle_+^{(a)}}{\operatorname{vol}(\mathrm{SO}(1, 1))} \mathcal{M}_{(a)}^{(c)}(z_1, z_2, \partial_{z_2}) \langle \mathbf{L}[O](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \rangle = \frac{\langle 0 | \mathbf{L}^+[O_2](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \mathbf{L}^-[O_1](x, z_1) | 0 \rangle_+^{(a)}}{\operatorname{vol}(\mathrm{SO}(1, 1))} \mathcal{M}_{(a)}^{(c)}(z_1, z_2, \partial_{z_2}) \langle \mathbf{L}[O](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \rangle = \frac{\langle 0 | \mathbf{L}^+[O_2](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \mathbf{L}^-[O_1](x, z_1) | 0 \rangle_+^{(a)}}{\operatorname{vol}(\mathrm{SO}(1, 1))} \mathcal{M}_{(a)}^{(c)}(z_1, z_2, \partial_{z_2}) \langle \mathbf{L}[O](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \rangle = \frac{\langle 0 | \mathbf{L}^+[O_2](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \mathbf{L}^-[O_1](x, z_1) | 0 \rangle_+^{(a)}}{\operatorname{vol}(\mathrm{SO}(1, 1))} \mathcal{M}_{(a)}^{(a)}(z_1, z_2, \partial_{z_2}) \langle \mathbf{L}[O^{\dagger}](x', z') \mathbf{L}^-[O^{\dagger}](x', z') \mathcal{M}_{(a)}^{(c)}(z_1, z_2, \partial_{z_2}) \langle \mathbf{L}[O^{\dagger}](x', z') \mathcal{M}_$$

Then we find

$$\mathbf{L}[O_{1}](x,z_{1})\mathbf{L}[O_{2}](x,z_{2}) = \sum_{\lambda} \int_{\frac{d-2}{2}-i\infty}^{\frac{d-2}{2}+i\infty} \frac{d\delta}{2\pi i} \widehat{C}_{\delta,\lambda}^{(c)}(z_{1},z_{2};\partial_{z_{2}}) \mathbb{W}_{\delta,\lambda,(c)}(x,z_{2}),$$
(2.194)

where

$$\mathbb{W}_{\delta,\lambda,(c)}(x,z) \equiv \int D^{d-2} z_1 D^{d-2} z_2 k_{\delta,\lambda,(c)}(z_1,z_2;z) \mathbf{L}[O_1](x,z_1) \mathbf{L}[O_2](x,z_2).$$
(2.195)

In our case, the kernel $k_{\delta,\lambda,(c)}$ arose from K^t through (2.180). This means that we can alternatively rewrite $\mathbb{W}_{\delta,\lambda,(c)}(x,z)$ as

$$\mathbb{W}_{\delta,\lambda,(c)}(x,z) = \mathcal{N}_{(c),\delta,\lambda}^{(a)} \frac{1}{2} \left(\mathbb{O}_{\delta+1,J_1+J_2-1,\lambda,(a)}^+(x,z) - \mathbb{O}_{\delta+1,J_1+J_2-1,\lambda,(a)}^-(x,z) \right).$$
(2.196)

Plugging this into (2.194) we conclude

$$\mathbf{L}[O_{1}](x,z_{1})\mathbf{L}[O_{2}](x,z_{2}) = \frac{1}{2} \sum_{\lambda \in \Lambda_{12}} \int_{\frac{d-2}{2}-i\infty}^{\frac{d-2}{2}+i\infty} \frac{d\delta}{2\pi i} \mathcal{N}^{(a)}_{(c),\delta,\lambda} \widehat{C}^{(c)}_{\delta,\lambda}(z_{1},z_{2};\partial_{z_{2}}) \left(\mathbb{O}^{+}_{\delta+1,J_{1}+J_{2}-1,\lambda,(a)}(x,z) - \mathbb{O}^{-}_{\delta+1,J_{1}+J_{2}-1,\lambda,(a)}(x,z) \right) + \cdots$$
(2.197)

In the last step we fixed a subtlety that we glossed over before: the identification of three-point structures in (2.189) can only be performed for low transverse spin λ since the right-hand side simply does not exist for higher transverse spin. Therefore, in (2.197) we have only correctly identified the low transverse spin contributions, and (\cdots) denotes the higher transverse spin contributions that we deal with in the next sections.

We can simplify equation (2.197) further, by defining

$$C_{\delta,\lambda}^{(a)}(z_1, z_2; \partial_{z_2}) = \frac{1}{2\pi i} \mathcal{N}_{(c),\delta,\lambda}^{(a)} \widehat{C}_{\delta,\lambda}^{(c)}(z_1, z_2; \partial_{z_2}).$$
(2.198)

Note that while $\widehat{C}_{\delta,\lambda}^{(c)}$ are linearly-independent, $C_{\delta,\lambda}^{(a)}$ are not. In terms of $C_{\delta,\lambda}^{(a)}$ we get

$$\mathbf{L}[O_{1}](x,z_{1})\mathbf{L}[O_{2}](x,z_{2}) = \pi i \sum_{\lambda \in \Lambda_{12}} \int_{\frac{d-2}{2}-i\infty}^{\frac{d-2}{2}+i\infty} \frac{d\delta}{2\pi i} C_{\delta,\lambda}^{(a)}(z_{1},z_{2};\partial_{z_{2}}) \left(\mathbb{O}_{\delta+1,J_{1}+J_{2}-1,\lambda,(a)}^{+}(x,z) - \mathbb{O}_{\delta+1,J_{1}+J_{2}-1,\lambda,(a)}^{-}(x,z) \right) + \cdots$$
(2.199)

Using (2.193) and (2.182) we can give an equivalent characterization of $C_{\delta,\lambda}^{(a)}$,

$$C_{\delta,\lambda}^{(a)}(z_1, z_2, \partial_{z_2}) \langle \mathbf{L}[O](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \rangle = \frac{\langle 0 | \mathbf{L}^+[O_2](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \mathbf{L}^-[O_1](x, z_1) | 0 \rangle_+^{(a)}}{\operatorname{vol}(\mathrm{SO}(1, 1))}$$
(2.200)

Equations (2.200) and (2.199) reproduce equations (3.97) and (3.98) of [15], respectively.³⁹ We refer to (2.200) as the "celestial map" formula because it maps the *d*-dimensional tensor structures appearing on the right-hand side to the (d-2)dimensional OPE differential operators $C_{\delta,\lambda}^{(a)}$. Note that (2.186) implies that $m_{(a)}$ such that $\mathcal{N}_{(c),\delta,\lambda}^{(a)}m_{(a)} = 0$ satisfy

$$\frac{\langle 0|\mathbf{L}^{+}[O_{2}](x,z_{2})\mathbf{L}[O^{\dagger}](x',z')\mathbf{L}^{-}[O_{1}](x,z_{1})|0\rangle_{+}^{(a)}}{\operatorname{vol}(\mathrm{SO}(1,1))}m_{(a)} = 0, \qquad (2.201)$$

i.e. the kernel of $\mathcal{N}_{(c),\delta,\lambda}^{(a)}$ is the same as the kernel of the celestial map.

Finally, to reproduce the low transverse spin terms in (2.147), we note, analogously to the discussion around (2.138), that in order for the left-hand side of (2.199) to be well-defined, we need the Regge intercept to satisfy $J_0 < J_1 + J_2 - 1$, in

³⁹There is a (-) sign in (2.199) which is not in (3.98) of [15], which is due to the more explicit treatment of $(-1)^J$ signs in section 2.5.1 of this paper.

which case we can use a superconvergence sum rule [34, 15] which states that $\mathbb{O}_{\delta+1,J_1+J_2-1,\lambda,(a)}^{(-1)^{J_1+J_2-1}}(x,z) = 0$. This allows us to rewrite (2.199) as

$$\mathbf{L}[O_{1}](x,z_{1})\mathbf{L}[O_{2}](x,z_{2})$$

$$=\pi i(-1)^{J_{1}+J_{2}}\sum_{\lambda\in\Lambda_{12}}\int_{\frac{d-2}{2}-i\infty}^{\frac{d-2}{2}+i\infty}\frac{d\delta}{2\pi i}C_{\delta,\lambda}^{(a)}(z_{1},z_{2};\partial_{z_{2}})\mathbb{O}_{\delta+1,J_{1}+J_{2}-1,\lambda,(a)}^{(-1)^{J_{1}+J_{2}}}(x,z)+\cdots,$$
(2.202)

which reproduces the first sum in (2.147).

Higher transverse spin

Now we would like to understand the higher transverse spin terms in the OPE (2.147). The logic in this case is similar to the low transverse spin case, but with some extra complications. Similarly to (2.180), we assume that

$$\mathcal{N}_{(c),\delta,\lambda_{\gamma},n}^{(a)}\mathcal{D}_{n}\left(\int_{\substack{x\approx2\\x\approx1^{-}}}d^{d}x_{1}d^{d}x_{2}K_{\delta+1,J,\lambda_{\gamma},(a)}^{t}(x_{1},x_{2};x,z)\mathcal{O}_{1}(x_{1})\mathcal{O}_{2}(x_{2})\right)$$

$$\xrightarrow{J\to J_{1}+J_{2}-1+n}\int D^{d-2}z_{1}D^{d-2}z_{2}k_{\delta,\lambda_{\gamma}(+n),(c)}(z_{1},z_{2};z)\mathbf{L}[\mathcal{O}_{1}](x,z_{1})\mathbf{L}[\mathcal{O}_{2}](x,z_{2}).$$
(2.203)

The evidence for this assumption is the same as for (2.180) and is discussed in section 2.5.2.

Using this statement in (2.152) requires some care. Indeed, plugging in this relation (again defining \mathcal{M} to be the left inverse to \mathcal{N}), we find

$$\int D^{d-2} z_1 D^{d-2} z_2 k_{\delta,\lambda_{\gamma}(+n),(c)}(z_1, z_2; z) \frac{\langle 0|\mathbf{L}^+[\mathcal{O}_2](x, z_2)\mathbf{L}[\mathcal{O}^\dagger](x', z')\mathbf{L}^-[\mathcal{O}_1](x, z_1)|0\rangle_+^{(a)}}{\operatorname{vol}(\mathrm{SO}(1, 1))^2} \mathcal{M}_{(a),\delta,\lambda_{\gamma},n}^{(c')}$$
$$= \frac{1}{2\pi i} \langle (\mathcal{D}_n \mathbf{L}[\mathcal{O}])(x, z)\mathbf{L}[\mathcal{O}^\dagger](x', z')\rangle \delta_{(c)}^{(c')}.$$
(2.204)

The equality holds if $J = J_n \equiv J_1 + J_2 - 1 + n$. But for this value of J, the action of \mathcal{D}_n is conformally-invariant, and the right-hand side of (2.204) contains a conformally-invariant two-point function for operators $\mathcal{D}_n \mathbf{L}[O]$ and $\mathbf{L}[O^{\dagger}]$. However, the quantum numbers of these operators are not Hermitian-conjugate to each other, and thus such two-point functions do not exist! This means that the right-hand side vanishes, and so the left-hand side must also vanish. Equation (2.204) at $J = J_n$ is thus trivially satisfied.

To obtain nontrivial information, we must shift away from $J = J_n$. For this, it helps to act with \mathcal{D}_n on $\mathbf{L}[O^{\dagger}]$ on both sides of (2.204). We obtain on the right-hand side

$$\frac{1}{2\pi i} \langle (\mathcal{D}_{n} \mathbf{L}[O])(x, z) (\mathcal{D}_{n} \mathbf{L}[O^{\dagger}])(x', z') \rangle \delta_{(c)}^{(c')}
= (J - J_{n}) \lim_{J \to J_{n}} \frac{1}{2\pi i} \frac{\langle (\mathcal{D}_{n} \mathbf{L}[O])(x, z) (\mathcal{D}_{n} \mathbf{L}[O^{\dagger}])(x', z') \rangle}{J - J_{n}} \delta_{(c)}^{(c')} + O((J - J_{n})^{2}).$$
(2.205)

After acting with \mathcal{D}_n on the second operator, the limit

$$\lim_{J \to J_n} \frac{1}{2\pi i} \frac{\langle (\mathcal{D}_n \mathbf{L}[\mathcal{O}])(x, z) (\mathcal{D}_n \mathbf{L}[\mathcal{O}^{\dagger}])(x', z') \rangle}{J - J_n}$$
(2.206)

is a conformally-invariant two-point function [38].

We now need to analyze the left-hand side of (2.152) away from $J = J_n$. One might worry that we would need to know the subleading term in (2.203) in order to determine the leading non-zero piece in the left-hand side of (2.204). Fortunately, this is not required. To see this, let us first write for general J, from (2.152) and the above,

$$\int \frac{d^{d}x_{1}d^{d}x_{2}}{\operatorname{vol}(\mathrm{SO}(1,1))^{2}} \mathcal{D}_{n} \left(K_{\Delta,J,\lambda(a)}^{t,\pm}(x_{1},x_{2};x,z)\theta(x\approx2,1^{-}) \right) \\ \times \mathcal{D}_{n} \left(\langle 0|O_{2}\mathbf{L}[O^{\dagger}](x',z')O_{1}|0\rangle_{+}^{(b)}\theta(2>x'>1^{-}) \right) \\ = (J-J_{n}) \lim_{J\to J_{n}} \frac{1}{2\pi i} \frac{\langle (\mathcal{D}_{n}\mathbf{L}[O])(x,z)(\mathcal{D}_{n}\mathbf{L}[O^{\dagger}])(x',z')\rangle}{J-J_{n}} \delta_{(a)}^{(b)} + O((J-J_{n})^{2})$$
(2.207)

Since the right-hand side is $O(J - J_n)$, this should be true for the left-hand side as well. We claim that in fact

$$\int \frac{d^{d}x_{1}d^{d}x_{2}}{\operatorname{vol}(\mathrm{SO}(1,1))^{2}}F(x_{1},x_{2};x,z)\mathcal{D}_{n}\left(\langle 0|O_{2}\mathbf{L}[O^{\dagger}](x',z')O_{1}|0\rangle_{+}^{(b)}\theta(2>x'>1^{-})\right) \in O(J-J_{n})$$
(2.208)

for any conformally-invariant kernel *F* that transforms at *x* with the quantum numbers of $\mathcal{D}_n \mathbf{L}[O]$. This statement implies that we can use (2.204) directly at $J = J_n = J_1 + J_2 - 1 + n$ for the purposes of determining the $O(J - J_n)$ term.

We actually need the following refined version of (2.208),

$$\int \frac{d^{d}x_{1}d^{d}x_{2}}{\operatorname{vol}(\mathrm{SO}(1,1))^{2}}F(x_{1},x_{2};x,z)\mathcal{D}_{n}\left(\langle 0|O_{2}\mathbf{L}[O^{\dagger}](x',z')O_{1}|0\rangle_{+}^{(b)}\theta(2>x'>1^{-})\right)$$

$$= (J-J_{n})\int \frac{d^{d}x_{1}d^{d}x_{2}}{\operatorname{vol}(\mathrm{SO}(1,1))}F(x_{1},x_{2};x,z)\mathcal{D}_{n}\left(\langle 0|O_{2}\mathbf{L}[O^{\dagger}](x',z')O_{1}|0\rangle_{+}^{(b)}\theta(2>x'>1^{-})\right)\Big|_{J=J_{r}}$$

$$+ O((J-J_{n})^{2}). \qquad (2.209)$$
Note that the expressions on the left-hand and right-hand sides differ by the power of the factor vol(SO(1, 1)). The reason for this is that while at $J = J_n$ and $J \neq J_n$ both integrals have 2 reparameterization zero modes, in the integral on the right, one of the modes is similar to $x \rightarrow \alpha x$ in

$$\int dx \,\delta(x) = 1, \qquad (2.210)$$

and does not require Faddeev-Popov fixing. A simple model for the above equation is provided by the integral

$$\int_0^\infty \frac{dx}{\operatorname{vol}(\operatorname{SO}(1,1))} 2 \cdot x^{-(J-J_n)} \partial_x (x^{J-J_n} \theta(x)) = 2(J-J_n).$$
(2.211)

Clearly, the coefficient 2 in front of $(J - J_n)$ can be computed by setting $J = J_n$ and removing the vol(SO(1, 1)) factor. The result (2.209) can be shown by an explicit calculation in the case when O_1, O_2 are scalars,⁴⁰ and then noting that integrands in both sides have the same transformation properties under weight-shifting operators acting on O_1, O_2, O [39, 45].

Combining everything together, we find the following equation for the kernel $k_{\delta,\lambda_{\gamma}(+n),(a)}$,

$$\int \frac{D^{d-2} z_1 D^{d-2} z_2}{\operatorname{vol}(\mathrm{SO}(1,1))} k_{\delta,\lambda_{\gamma}(+n),(c)}(z_1, z_2; z) \mathcal{D}_n\left(\langle 0 | \mathbf{L}^+[\mathcal{O}_2](x, z_2) \mathbf{L}[\mathcal{O}^\dagger](x', z') \mathbf{L}^-[\mathcal{O}_1](x, z_1) | 0 \rangle_+^{(a)}\right) \mathcal{M}_{(a),\delta,\lambda_{\gamma}}^{(c')}$$

$$= \lim_{J \to J_n} \frac{1}{2\pi i} \frac{\langle (\mathcal{D}_n \mathbf{L}[\mathcal{O}])(x, z) (\mathcal{D}_n \mathbf{L}[\mathcal{O}^\dagger])(x', z') \rangle}{J - J_n} \delta_{(c)}^{(c')}. \qquad (2.212)$$

From this equation, following precisely the same steps as in the derivation of low transverse spin terms in the previous section, we find that if the operators $C_{\delta,\lambda_{\gamma}(+n)}^{(a)}$ are defined by

$$C_{\delta,\lambda_{\gamma}(+n)}^{(a)}(z_1, z_2, \partial_{z_2}) \lim_{J \to J_n} \frac{\langle (\mathcal{D}_n \mathbf{L}[O])(x, z_2)(\mathcal{D}_n \mathbf{L}[O^{\dagger}])(x', z') \rangle}{J - J_n}$$

= $(-1)^n \mathcal{D}_n \left(\langle 0 | \mathbf{L}^+[O_2](x, z_2) \mathbf{L}[O^{\dagger}](x', z') \mathbf{L}^-[O_1](x, z_1) | 0 \rangle_+^{(a)} \right), \quad (2.213)$

then the contribution of higher transverse spin terms is given by the second sum in (2.147).⁴¹ In appendix A.5 we consider which representations $\lambda_{\gamma}(+n)$ can be generated in this way, and show that there are enough *d*-dimensional structures that survive this celestial map to account for all the celestial OPE structures.

 $^{^{40}}$ It is easy to convince oneself that it suffices to ensure that the result of this section agrees with the result of section 2.4.

⁴¹Here we introduced a sign $(-1)^n$ by hand in order to simplify (2.147).

2.5.3 Celestial map without light transforms

In the previous section we derived the celestial map formulas (2.200) and (2.213) that determine the OPE differential operators $C_{\delta,\lambda}^{(a)}$ appearing in (2.147). These formulas involve taking several light-transforms of tensor structures analytically continued to Wightman correlators, which in practice can be a difficult calculation. Fortunately, as first observed in [15] and proved for traceless-symmetric operators, there exists a simpler version of the celestial map formulas that contains only simple algebraic manipulations.

We will show that the following is equivalent to (2.200),

$$C_{\delta,\lambda}^{(a)}(z_{1}, \mathbf{w}_{1}, z_{2}, \mathbf{w}_{2}, \partial_{z_{2}}, \partial_{\mathbf{w}_{2}}) \left((-2H_{20}) \langle O(X_{2}, Z_{2}, \mathbf{W}_{2}) O^{\dagger}(X_{0}, Z_{0}, \mathbf{W}_{0}) \rangle \right) \Big|_{\text{celestial}}$$

= $X_{12} \left| -2V_{0,21} \right| \langle O_{1}(X_{1}, Z_{1}, \mathbf{W}_{1}) O^{\dagger}(X_{0}, Z_{0}, \mathbf{W}_{0}) O_{2}(X_{2}, Z_{2}, \mathbf{W}_{2}) \rangle_{+}^{(a)} \Big|_{\text{celestial}}$
(2.214)

and the following equivalent to (2.213),

$$C_{\delta,\lambda_{\gamma}(+n)}^{(a)}(z_{1},\mathbf{w}_{1},z_{2},\mathbf{w}_{2},\partial_{z_{2}},\partial_{w_{2}})\left(\lim_{J\to J_{n}}(-2H_{20})\frac{\langle (\mathcal{D}_{n}'\mathcal{O})(X_{2},Z_{2},\mathbf{W}_{2})(\mathcal{D}_{n}'\mathcal{O}^{\dagger})(X_{0},Z_{0},\mathbf{W}_{0})\rangle}{J-J_{n}}\right)\Big|_{\text{celestial}}$$

$$=\frac{(-1)^{n}X_{12}}{\delta(-2V_{0,21})}\mathcal{D}_{n}'\left(\theta(V_{0,12})\langle \mathcal{O}_{1}(X_{1},Z_{1},\mathbf{W}_{1})\mathcal{O}^{\dagger}(X_{0},Z_{0},\mathbf{W}_{0})\mathcal{O}_{2}(X_{2},Z_{2},\mathbf{W}_{2})\rangle_{+}^{(a)}\right)\Big|_{\text{celestial}}.$$

$$(2.215)$$

(Note that we have used the $\langle \cdots \rangle_{+}^{(a)}$ structures defined in section 2.5.1.) In the above equations, we have explicitly reintroduced polarization vectors \mathbf{w}_i for the second and higher rows of Young diagrams of O_i , as described in section 2.3. Furthermore, we have used embedding space notation [30] on the right-hand side, with the standard tensor structures

$$X_{ij} \equiv -2X_i \cdot X_j \tag{2.216}$$

$$V_{i,jk} \equiv \frac{Z_i \cdot X_j X_i \cdot X_k - Z_i \cdot X_k X_i \cdot X_j}{X_j \cdot X_k}, \qquad (2.217)$$

$$H_{ij} \equiv -2(Z_i \cdot Z_j X_i \cdot X_j - Z_i \cdot X_j Z_j \cdot X_i).$$
(2.218)

Finally, the notation $(\cdots)|_{celestial}$ stands for substituting the following values for the embedding space coordinates,

$$Z_0 = -(1, 0, 0), \quad Z_1 = -(0, 1, 0), \quad Z_2 = -(0, 1, 0),$$

$$X_i = (0, 0, z_i), \quad W_{i,j} = (0, 0, w_{i,j}), \quad (2.219)$$

where we specify coordinates in the order (Y^+, Y^-, Y^μ) , $\mu = 0, \dots, d-1$. Note that in (2.215) we divide by $\delta(-2V_{0,21})$. By this we mean that the result of the action of \mathcal{D}'_n is proportional to $\delta(-2V_{0,21})$, and we simply read off the coefficient of this delta-function. Note that this coefficient is only well-defined in configurations where $V_{0,21} = 0$, which is indeed the case for the celestial locus (2.219).

Factoring out the light-transforms

We start by proving (2.215). The proof of (2.214) is only a simple modification that we comment on below.

The proof proceeds with evaluation of light-transforms in both sides of (2.213). We start with the right-hand side

$$\mathcal{D}_{n}\left(\langle 0|\mathbf{L}^{+}[O_{2}](x, z_{2}, \mathbf{w}_{2})\mathbf{L}[O^{\dagger}](x_{0}, z_{0}, \mathbf{w}_{0})\mathbf{L}^{-}[O_{1}](x, z_{1}, \mathbf{w}_{1})|0\rangle_{+}^{(a)}\right).$$
 (2.220)

Here \mathcal{D}_n acts on (x_0, z_0, \mathbf{w}_0) . We can rewrite this equivalently as

$$= \mathcal{D}_{n}\mathbf{L}_{1}\mathbf{L}_{2}\left(\langle 0|O_{2}(x,z_{2},\mathbf{w}_{2})\mathbf{L}[O^{\dagger}](x_{0},z_{0},\mathbf{w}_{0})O_{1}(x,z_{1},\mathbf{w}_{1})|0\rangle_{+}^{(a)}\theta(2>0>1^{-})\right)$$

$$= \mathcal{D}_{n}\mathbf{L}_{1}\mathbf{L}_{2}\left(\langle O_{1}(x,z_{1},\mathbf{w}_{1})O_{2}(x,z_{2},\mathbf{w}_{2})\mathbf{L}[O^{\dagger}](x_{0},z_{0},\mathbf{w}_{0})\rangle_{+}^{(a)}\theta(V_{0,12})\right), \quad (2.221)$$

where \mathbf{L}_i denotes light-transform acting on (x_i, z_i, \mathbf{w}_i) . In the last equality we used the following fact. First of all,

$$V_{0,12} = \frac{z_0 \cdot x_{10} x_{20}^2 - z_0 \cdot x_{20} x_{10}^2}{x_{12}^2}$$
(2.222)

is positive for $1 \approx 2$ and $2 > 0 > 1^-$, and is negative for $1 \approx 2$ and $1 > 0 > 2^-$. In (2.169) we have two terms, and multiplying by $\theta(V_{0,12})$ selects the first term, which is the one appearing in the first line of (2.221).

Now we can use (2.65) to rewrite this further as

$$= \mathbf{L}_{1} \mathbf{L}_{2} \mathbf{L}_{0} \mathcal{D}'_{n} \left(\langle O_{1}(x, z_{1}, \mathbf{w}_{1}) O_{2}(x, z_{2}, \mathbf{w}_{2}) O^{\dagger}(x_{0}, z_{0}, \mathbf{w}_{0}) \rangle_{+}^{(a)} \theta(V_{0, 12}) \right), \quad (2.223)$$

where we took into account the easily verified fact that $V_{0,12}$ commutes with L_0 .

Appearance of $\delta(-2V_{0,12})$

Note that without the theta-function we would have

$$\mathcal{D}'_{n} \langle O_{1}(x, z_{1}, \mathbf{w}_{1}) O_{2}(x, z_{2}, \mathbf{w}_{2}) O^{\dagger}(x_{0}, z_{0}, \mathbf{w}_{0}) \rangle^{(a)} = 0.$$
(2.224)

This is because for $J = J_n$, all possible three-point tensor structures above are polynomial in z_0 , and hence killed by \mathcal{D}'_n . To see this, note that the way non-polynomial structures in z_0 appear is through the factors of the form

$$\langle O_1(x, z_1, \mathbf{w}_1) O_2(x, z_2, \mathbf{w}_2) O^{\dagger}(x_0, z_0, \mathbf{w}_0) \rangle^{(a)} = (\cdots) V_{0, 12}^{J-k}$$
 (2.225)

when J < k, where k > 0 is the degree of z_0 in (\cdots) . Non-polynomiality cannot appear in any other way, because we require that all polarizations except z_0 enter polynomially, and $V_{0,12}$ is the only invariant that involves only z_0 . We see that for sufficiently large integer J all structures are therefore polynomial, and the appearance of non-polynomial structures is indicated by the reduction in the number of polynomial structures. The number of polynomial structures can be computed using group-theoretic counting rules [48], and a simple calculation shows that the polynomial structures start disappearing for $J = J_1 + J_2 - 1$ (see appendix A.5). This means that for $J = J_1 + J_2 - 1$ the maximal non-polynomiality is $V_{0,12}^{-1}$, and for $J = J_n = J_1 + J_2 - 1 + n$ with n > 0 there are no non-polynomial structures. More generally, the smallest possible exponent of $V_{0,12}$ is

$$V_{0,12}^{J-J_1-J_2}. (2.226)$$

Since all structures are polynomial for $J = J_n$, the properties of \mathcal{D}'_n discussed in section 2.3 ensure (2.224).

However, we are interested in

$$\mathcal{D}'_{n}\left(\langle O_{1}(x, z_{1}, \mathbf{w}_{1})O_{2}(x, z_{2}, \mathbf{w}_{2})O^{\dagger}(x_{0}, z_{0}, \mathbf{w}_{0})\rangle_{+}^{(a)}\theta(V_{0, 12})\right).$$
(2.227)

We can obtain derivatives of delta functions $\delta^{(m)}(-2V_{0,12})$ from \mathcal{D}'_n hitting the theta-function. We claim that we get a result which is proportional to $\delta(-2V_{0,12})$. To see this, we regularize by analytic continuation in *J*. Note that in (2.227) the most negative power of $V_{0,12}$ is

$$V_{0,12}^{J-J_1-J_2-n} = V_{0,12}^{J-J_n-1},$$
(2.228)

because \mathcal{D}'_n has *n* derivatives which will therefore hit $V_{0,12}^{J-J_1-J_2}$, which is the most negative power of $V_{0,12}$ before acting with \mathcal{D}'_n , at most *n* times. In other words, we have for each (*a*)

$$\mathcal{D}'_{n}\left(\langle O_{1}(x, z_{1}, \mathbf{w}_{1})O_{2}(x, z_{2}, \mathbf{w}_{2})O^{\dagger}(x_{0}, z_{0}, \mathbf{w}_{0})\rangle_{+}^{(a)}\theta(V_{0,12})\right)$$

= $(J - J_{n})(\cdots)V_{0,12}^{J-J_{n}-1}\theta(V_{0,12}) + R,$ (2.229)

where *R* involves higher powers of $V_{0,12}$ or $(J - J_n)$. As we send $J \rightarrow J_n$, we then find $R \rightarrow 0$ and

$$(J - J_n)V_{0,12}^{J - J_n - 1}\theta(V_{0,12}) \to \delta(V_{0,12}).$$
(2.230)

We therefore conclude that

$$\mathcal{D}'_{n}\left(\langle O_{1}(x, z_{1}, \mathbf{w}_{1})O_{2}(x, z_{2}, \mathbf{w}_{2})O^{\dagger}(x_{0}, z_{0}, \mathbf{w}_{0})\rangle_{+}^{(a)}\theta(V_{0, 12})\right) = \delta(-2V_{0, 21}) \times \text{finite.}$$
(2.231)

Symmetries of the integrand

Defining

$$f(X_1, Z_1, \mathbf{W}_1; X_2, Z_2, \mathbf{W}_2; X_0, Z_0, \mathbf{W}_0) = \mathcal{D}'_n \left(\langle O_1(X_1, Z_1, \mathbf{W}_1) O_2(X_2, Z_2, \mathbf{W}_2) O^{\dagger}(X_0, Z_0, \mathbf{W}_0) \rangle_+^{(a)} \theta(V_{0,12}) \right), \quad (2.232)$$

and

$$f(\alpha_1, \alpha_2, \alpha_0) = f(Z_1 - \alpha_1 X_{\infty}, -X_{\infty}, \mathbf{W}_1; Z_2 - \alpha_2 X_{\infty}, -X_{\infty}, \mathbf{W}_2; Z_0 - \alpha_0 X_0, -X_0, \mathbf{W}_0),$$
(2.233)

where

$$X_{\infty} = (0, 1, 0), \quad X_0 = (1, 0, 0),$$
 (2.234)

$$Z_i = (0, 0, z_i), \quad W_{i,j} = (0, 0, w_{i,j}), \quad (2.235)$$

we can rewrite (2.223) and thus (2.220) for $x = \infty, x' = 0$ as

$$\mathcal{D}_{n}\langle 0|\mathbf{L}^{+}[\mathcal{O}_{2}](\infty, z_{2}, \mathbf{w}_{2})\mathbf{L}[\mathcal{O}^{\dagger}](0, z_{0}, \mathbf{w}_{0})\mathbf{L}^{-}[\mathcal{O}_{1}](\infty, z_{1}, \mathbf{w}_{1})|0\rangle_{+}^{(a)} = \int_{-\infty}^{+\infty} d\alpha_{1}d\alpha_{2}d\alpha_{0}f(\alpha_{1}, \alpha_{2}, \alpha_{0}).$$
(2.236)

Our goal here will be to find an expression for $f(\alpha_1, \alpha_2, \alpha_0)$ based solely on symmetries. First, we check that the boost in the embedding space which sends

$$X_{\infty} \to \lambda X_{\infty}, \quad X_0 \to \lambda^{-1} X_0$$
 (2.237)

implies that for $\lambda > 0$

$$f(\alpha_1, \alpha_2, \alpha_0) = \lambda^{J_1 + J_2 - J'} f(\lambda \alpha_1, \lambda \alpha_2, \lambda^{-1} \alpha_0), \qquad (2.238)$$

where $J' = J_n - n = J_1 + J_2 - 1$ is the spin we get after the action of \mathcal{D}'_n . That is, we have

$$f(\alpha_1, \alpha_2, \alpha_0) = \lambda f(\lambda \alpha_1, \lambda \alpha_2, \lambda^{-1} \alpha_0).$$
(2.239)

Second, the symmetry that in Minkowski space is represented by translation along z_0 acts on embedding space coordinates as

$$X_0 \to X_0 + \lambda Z_0, \quad Z_i \to Z_i + 2(z_i \cdot z_0)\lambda X_{\infty} \quad (i = 1, 2), \tag{2.240}$$

leaving all other coordinates invariant.⁴² We can check that it implies the following equation for f,

$$f(\alpha_1, \alpha_2, \alpha_0) = (1 - \alpha_0 \lambda)^{-\Delta - J'} f(\alpha_1 - 2(z_1 \cdot z_0)\lambda, \alpha_2 - 2(z_2, z_0)\lambda, (\alpha_0^{-1} - \lambda)^{-1})$$
(2.242)

for $1 - \alpha_0 \lambda > 0$.

Now recall from the previous discussion that $f(\alpha_1, \alpha_2, \alpha_0)$ is proportional to $\delta(-2V_{0,21})$, and so we can write

$$f(\alpha_1, \alpha_2, \alpha_0) = \delta\left(\frac{\alpha_1(z_0 \cdot z_2) - \alpha_2(z_0 \cdot z_2)}{(z_1 \cdot z_2)}\right) g(\alpha_2, \alpha_0),$$
(2.243)

for some g. Here we eliminated the α_1 -dependence using the delta-function. In terms of g, the symmetries discussed above read

$$g(\alpha_2, \alpha_0) = g(\lambda \alpha_2, \lambda^{-1} \alpha_0), \qquad (2.244)$$

$$g(\alpha_2, \alpha_0) = (1 - \alpha_0 \lambda)^{-\Delta - J'} g(\alpha_2 - 2(z_2, z_0)\lambda, (\alpha_0^{-1} - \lambda)^{-1}), \qquad (2.245)$$

for $\lambda > 0$ and $1 - \alpha_0 \lambda > 0$ respectively. Let us consider g_0 defined by

$$g_0(\alpha_2, \alpha_0) = |\alpha_2 \alpha_0 - 2(z_2 \cdot z_0)|^{-\Delta - J'}.$$
 (2.246)

It is easy to check that g_0 satisfies the same symmetries as g, and thus g/g_0 is simply invariant under the above transformations of α_2, α_0 . Note that we have two continuous families of transformations, and it is easy to verify that the 2 vector fields

$$W_{i,j} \to W_{i,j} + 2(w_{i,j} \cdot z_0)\lambda X_{\infty}$$
 (*i* = 1, 2), (2.241)

⁴²To be more precise, we have

but since in (2.233) $W_{i,j}$ are inserted with first-row polarization $-X_{\infty}$, the shift by X_{∞} has no effect on the value of f.

by which they are generated are generically linearly-independent. Since we have only two coordinates α_2 , α_0 , we find that g/g_0 should be locally constant and can only change value where these vector fields degenerate. This only happens when $\alpha_2\alpha_0 - 2(z_2 \cdot z_0) = 0$ or $\alpha_0 = 0$.

Note that $g(\alpha_2, \alpha_0)$ comes from a three-point structure, and the three-point structures are analytic away from $X_{ij} = 0$, which are given by

$$X_{12} = -2(z_1 \cdot z_2), \tag{2.247}$$

$$X_{10} = \alpha_1 \alpha_0 - 2(z_1 \cdot z_0), \qquad (2.248)$$

$$X_{20} = \alpha_2 \alpha_0 - 2(z_2 \cdot z_0), \qquad (2.249)$$

so g/g_0 cannot have a discontinuity at $\alpha_0 = 0$. We thus only need to determine how g/g_0 changes when crossing $X_{20} = 0$. Since g comes from a time-ordered three-point structure, a non-analyticity near $X_{20} = 0$ must have the form

$$(X_{20} + i\epsilon)^{\#},$$
 (2.250)

where we use the usual $i\epsilon$ -prescription for time-ordered correlators. It follows that

$$g(\alpha_2, \alpha_0) = A(\alpha_2 \alpha_0 - 2(z_2 \cdot z_0) + i\epsilon)^{-\Delta - J'}$$
(2.251)

for some A, for all real values of α_2, α_0 . It is convenient to express A in terms of g(0,0),

$$g(\alpha_2, \alpha_0) = (-2z_2 \cdot z_0)^{\Delta + J'} g(0, 0) (\alpha_2 \alpha_0 - 2(z_2 \cdot z_0) + i\epsilon)^{-\Delta - J'}.$$
 (2.252)

Computing the light-transforms

We can now compute the integrals in (2.236),

$$\int d\alpha_{1}d\alpha_{2}d\alpha_{0}f(\alpha_{1},\alpha_{2},\alpha_{0})$$

$$= \int d\alpha_{1}d\alpha_{2}\delta\left(\frac{\alpha_{1}(z_{0}\cdot z_{2}) - \alpha_{2}(z_{0}\cdot z_{2})}{(z_{1}\cdot z_{2})}\right)\int d\alpha_{0}(-2z_{2}\cdot z_{0})^{\Delta+J'}g(0,0)(\alpha_{2}\alpha_{0} - 2(z_{2}\cdot z_{0}) + i\epsilon)^{-\Delta-J'}$$

$$= g(0,0)(-2z_{2}\cdot z_{0})\int d\alpha_{1}d\alpha_{2}\delta\left(\frac{\alpha_{1}(z_{0}\cdot z_{2}) - \alpha_{2}(z_{0}\cdot z_{2})}{(z_{1}\cdot z_{2})}\right)\int d\alpha_{0}(\alpha_{2}\alpha_{0} + 1 + i\epsilon)^{-\Delta-J'}$$

$$= \frac{-2\pi i}{\Delta+J'-1}g(0,0)(-2z_{2}\cdot z_{0})\int d\alpha_{1}d\alpha_{2}\delta\left(\frac{\alpha_{1}(z_{0}\cdot z_{2}) - \alpha_{2}(z_{0}\cdot z_{2})}{(z_{1}\cdot z_{2})}\right)\delta(\alpha_{2})$$

$$= \frac{-2\pi i}{\Delta+J'-1}(-2z_{1}\cdot z_{2})g(0,0), \qquad (2.253)$$

where we used the equation

$$\int_{-\infty}^{+\infty} \frac{dx}{(xy+1+i\epsilon)^a} = -\frac{2\pi i}{a-1}\delta(y)$$
(2.254)

which we prove in appendix A.7.2. Unwinding the definitions, we check that

$$(-2z_{1} \cdot z_{2})g(0,0) = \frac{X_{12}}{\delta(-2V_{0,21})} \mathcal{D}'_{n} \left(\theta(V_{0,12}) \langle \mathcal{O}_{1}(X_{1}, Z_{1}, \mathbf{W}_{1}) \mathcal{O}^{\dagger}(X_{0}, Z_{0}, \mathbf{W}_{0}) \mathcal{O}_{2}(X_{2}, Z_{2}, \mathbf{W}_{2}) \rangle_{+}^{(a)} \right) \Big|_{\text{celestial}}.$$

$$(2.255)$$

It only remains to compute the light-transforms in the left-hand side of (2.213). For this, note that

$$\lim_{J \to J_n} \frac{\langle (\mathcal{D}_n \mathbf{L}[\mathcal{O}])(x_2, z_2, \mathbf{w}_2)(\mathcal{D}_n \mathbf{L}[\mathcal{O}^{\dagger}])(x_0, z_0, \mathbf{w}_0) \rangle}{J - J_n}$$

= $\mathbf{L}_2 \mathbf{L}_0 \lim_{J \to J_n} \frac{\langle (\mathcal{D}'_n \mathcal{O})(X_2, Z_2, \mathbf{W}_2)(\mathcal{D}'_n \mathcal{O}^{\dagger})(X_0, Z_0, \mathbf{W}_0) \rangle}{J - J_n}$ (2.256)

and defining

$$g'(X_2, Z_2, \mathbf{W}_2; X_0, Z_0, \mathbf{W}_0) \equiv \lim_{J \to J_n} \frac{\langle (\mathcal{D}'_n O)(X_2, Z_2, \mathbf{W}_2) (\mathcal{D}'_n O^{\dagger})(X_0, Z_0, \mathbf{W}_0) \rangle}{J - J_n},$$
(2.257)
$$g'(g_1, g_2) \equiv g'(Z_2 - g_2 X_2 - Y_2, \mathbf{W}_2; Z_2 - g_2 X_2 - Y_2, \mathbf{W}_2)$$

$$g'(\alpha_2, \alpha_0) \equiv g'(Z_2 - \alpha_2 X_{\infty}, -X_{\infty}, \mathbf{W}_2; Z_0 - \alpha_0 X_0, -X_0, \mathbf{W}_0),$$
(2.258)

we find that g' satisfies the same properties as g above, and the same arguments lead to

$$g'(\alpha_2, \alpha_0) = (-2z_2 \cdot z_0)^{\Delta + J'} g'(0, 0) (\alpha_2 \alpha_0 - 2(z_2 \cdot z_0) + i\epsilon)^{-\Delta - J'}.$$
 (2.259)

We can then similarly compute

$$\mathbf{L}_{2}\mathbf{L}_{0}\lim_{J\to J_{n}}\frac{\langle (\mathcal{D}_{n}'\mathcal{O})(X_{2}, Z_{2}, \mathbf{W}_{2})(\mathcal{D}_{n}'\mathcal{O}^{\dagger})(X_{0}, Z_{0}, \mathbf{W}_{0})\rangle}{J-J_{n}} = \int d\alpha_{2}d\alpha_{0}g'(\alpha_{2}, \alpha_{0})$$
$$= \frac{-2\pi i}{\Delta + J' - 1}(-2z_{2} \cdot z_{0})g'(0, 0).$$
(2.260)

Since $(-2z_2 \cdot z_0) = -2H_{20}|_{\text{celestial}}$, we find

$$(-2z_2 \cdot z_0)g'(0,0) = \left(\lim_{J \to J_n} (-2H_{20}) \frac{\langle (\mathcal{D}'_n \mathcal{O})(X_2, Z_2, \mathbf{W}_2)(\mathcal{D}'_n \mathcal{O}^{\dagger})(X_0, Z_0, \mathbf{W}_0) \rangle}{J - J_n} \right) \Big|_{\text{celestial}}.$$
(2.261)

Combining with (2.213), (2.236), (2.253), (2.256), (2.260), and (2.255), we arrive at (2.215).

Low transverse spin case

The only modification to the above proof required for the case of low transverse spin — (2.200) and (2.214) — concerns the appearance of the delta-function in the analogue of (2.227).

$$(\langle O_1(x, z_1, \mathbf{w}_1) O_2(x, z_2, \mathbf{w}_2) O^{\dagger}(x_0, z_0, \mathbf{w}_0) \rangle_+^{(a)} \theta(V_{0, 12})).$$
(2.262)

In principle, there is no delta-function here. Instead, from the discussion in section 2.5.3 we know that for $J = J_1 + J_2 - 1$ there are structures which contain the most negative power of $V_{0,21}$ which is

$$(\cdots)(-2V_{0,21})^{-1}\theta(V_{0,12}).$$
 (2.263)

This inverse power of $V_{0,21}$ leads to the divergence in the triple light-transform in (2.200) which is canceled by the SO(1, 1) factor. Similarly to the discussion around (2.209) we can remove the vol(SO(1, 1)) factor and replace (in fact, this is exactly the same replacement as in (2.209))

$$(-2V_{0,21})^{-1}\theta(V_{0,12}) \to \delta(-2V_{0,21})$$
 (2.264)

and then the above proof can be applied to (2.214). The only remaining difference is that in the final formula we prefer to factor out $(-2V_{0,21})^{-1}$ before the above substitution instead of factoring out $\delta(-2V_{0,21})$ after the substitution.

2.6 Examples

2.6.1 Re-deriving the $L[\phi_1] \times L[\phi_2]$ OPE from general formulas

In this section we rederive the scalar detector OPE (2.138) from the general result (2.147).

The first step is to work out the low-transverse spin terms in (2.138). The set of transverse spins Λ_{12} is constrained to consist of traceless-symmetric representations λ of spin *j* because $\mathbf{L}[\phi_1]$ and $\mathbf{L}[\phi_2]$ both transform as scalars on the celestial sphere. In *d*-dimensional language, the transverse spin *j* is the length of the second row of the Young diagram of a SO(d - 1, 1) irrep, and is therefore constrained to be j = 0 since only traceless-symmetric light-ray operators appear in $\phi_1 \times \phi_2$ OPE. The low-transverse spin contributions are then given by

$$\mathbf{L}[\phi_1](x, z_1)\mathbf{L}[\phi_2](x, z_2) = \pi i \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta}{2\pi i} C_{\delta, j=0}(z_1, z_2, \partial_{z_2})\mathbb{O}^+_{\delta+1, J=-1}(x, z_2) + \cdots,$$
(2.265)

where we also substituted $J_1 = J_2 = 0$ and removed dependence on the transverse polarizations \mathbf{w}_i since all operators are traceless-symmetric. In order for this to agree with the j = 0 term of (2.138), we need to verify that $C_{\delta,j=0}$ normalized as in (2.89) also satisfies the celestial map (2.214). We do this by computing the structures entering in (2.214) and comparing them to the structures in (2.89).

First of all, we need to determine the expression for the three-point structure entering (2.214). Our *d*-dimensional structures for integer *J* are defined in appendix A.2, and the analytic continuation should be performed following the conventions of section 2.5.1. Comparing (A.7) and (2.164) we see that n = 0 in (2.164) and f_0 is given by some product of distances x_{ij} which is positive for space-like separated points. Looking at (2.168) we see that in our case the analytically-continued three-point structures appearing in (2.214), when all points are spacelike-separated, are equal to the absolute value of (A.7).

We can therefore use (A.9) and substitute the celestial locus values (2.219) into it. We find for J = -1,

$$X_{12}|-2V_{3,12}|\langle\phi_{1}(X_{1})\phi_{2}(X_{2})O(X_{3},Z_{3})\rangle|_{\text{celestial}} = \frac{1}{X_{12}^{\frac{\Delta_{1}+\Delta_{2}-\Delta-1}{2}}X_{13}^{\frac{\Delta_{1}+\Delta-\Delta_{2}-1}{2}}X_{23}^{\frac{\Delta_{2}+\Delta-\Delta_{1}-1}{2}}}\Big|_{\text{celestial}}$$
$$= \frac{1}{(-2z_{1}\cdot z_{2})^{\frac{\delta_{1}+\delta_{2}-\delta}{2}}(-2z_{1}\cdot z_{3})^{\frac{\delta_{1}+\delta-\delta_{2}}{2}}(-2z_{2}\cdot z_{3})^{\frac{\delta_{2}+\delta-\delta_{1}}{2}}},$$
(2.266)

and since this is positive and all X_i are space-like separated in (2.219), it follows, according to the discussion above, that this is equal to the right-hand side of (2.214) after substitution $3 \rightarrow 0$.

The left-hand side of (2.214) is easily computed from (A.8) to be equal to

$$(-2H_{23})\langle O(X_2, Z_2)O(X_3, Z_3)\rangle|_{\text{celestial}} = (-2z_2 \cdot z_3)^{-\delta}, \qquad (2.267)$$

after substitution $3 \rightarrow 0$. Using these results in (2.214) we see immediately that it gives the same normalization of $C_{\delta, j=0}$ as (2.89).

A similar logic works for the higher transverse spin terms in (2.147). We have $\lambda_{\gamma}(+n) = (n, \gamma)$. The set Γ_{12} in (2.147) then consists of just the trivial representation, because only traceless-symmetric representations appear in the $\phi_1 \times \phi_2$ OPE. We

thus find that the contribution of higher transverse spins is

$$\mathbf{L}[O_{1}](x, z_{1}, \mathbf{w}_{1})\mathbf{L}[O_{2}](x, z_{2}, \mathbf{w}_{2})$$

$$= \pi i \sum_{n=1}^{\infty} \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta}{2\pi i} C_{\delta, j=n}(z_{1}, z_{2}, \partial_{z_{2}}, \partial_{\mathbf{w}_{2}}) (\mathcal{D}_{n} \mathbb{O}_{\delta+1, n-1}^{(-1)^{n}})(x, z_{2}, \mathbf{w}_{2}) + \cdots$$
(2.268)

Therefore, in order to verify that (2.147) reproduces (2.138), we need to check that the normalizations of $C_{\delta,j}$ defined by (2.89) and (2.215) are consistent.

Reasoning analogously to the lower transverse spin case, we find that we need to compute the action of \mathcal{D}'_n defined by (2.62) on the absolute value of (A.7). Since in (2.215) we have $\theta(V_{0,12})$, we need to restrict to the region where $V_{0,12}$ is positive. In this region, we have

$$\mathcal{D}'_{n} \langle \phi_{1}(x_{1})\phi_{2}(x_{2})O(x_{3},z) \rangle_{+} = \mathcal{D}'_{n} \frac{(-2z \cdot x_{23} x_{13}^{2} + 2z \cdot x_{13} x_{23}^{2})^{J}}{x_{12}^{\Delta_{1}+\Delta_{2}-\Delta_{4}J} x_{13}^{\Delta_{1}+\Delta_{-}\Delta_{2}+J} x_{23}^{\Delta_{2}+\Delta-\Delta_{1}+J}}$$

$$= \frac{J(J-1)\cdots(J-n+1)}{n!} (-2w \cdot x_{23} x_{13}^{2} + 2w \cdot x_{13} x_{23}^{2})^{n} \frac{(-2z \cdot x_{23} x_{13}^{2} + 2z \cdot x_{13} x_{23}^{2})^{J-n}}{x_{12}^{\Delta_{1}+\Delta_{2}-\Delta_{4}J} x_{23}^{\Delta_{1}+\Delta-\Delta_{2}+J} x_{23}^{\Delta_{2}+\Delta-\Delta_{1}+J}}$$

$$(2.269)$$

Multiplying by $\theta(V_{0,12})$ and taking limit $J \rightarrow n-1$, we find

$$\mathcal{D}'_{n}\left(\langle\phi_{1}(x_{1})\phi_{2}(x_{2})O(x_{3},z)\rangle_{+}\theta(V_{0,12})\right) = \frac{1}{n} \frac{(-2w \cdot x_{23} x_{13}^{2} + 2w \cdot x_{13} x_{23}^{2})^{n}}{x_{12}^{\Delta_{1} + \Delta_{2} - \Delta + n + 1} x_{13}^{\Delta_{1} + \Delta - \Delta_{2} + n - 1} x_{23}^{\Delta_{2} + \Delta - \Delta_{1} + n - 1}}\delta(-2V_{0,12}).$$
(2.270)

Lifting this to embedding space and evaluating at the celestial locus (2.219), we find that the right-hand side of (2.215) is

$$\frac{(-1)^{n} X_{12}}{\delta(-2V_{0,21})} \mathcal{D}'_{n} \left(\theta(V_{0,12}) \langle \phi_{1}(X_{1}) \mathcal{O}(X_{0}, Z_{0}) \phi_{2}(X_{2}) \rangle_{+}^{(a)} \right) \Big|_{\text{celestial}} \\
= \frac{1}{n} \frac{(-4W \cdot X_{2} X_{1} \cdot X_{3} + 4W \cdot X_{1} X_{2} \cdot X_{3})^{n}}{X_{12}^{\frac{\Delta_{1} + \Delta_{2} - \Delta_{1} + n - 1}{2}} X_{13}^{\frac{\Delta_{2} + \Delta_{-} - \Delta_{1} + n - 1}{2}} \Big|_{\text{celestial}} \\
= \frac{1}{n} \frac{(-4w \cdot z_{2} z_{1} \cdot z_{3} + 4w \cdot z_{1} z_{2} \cdot z_{3})^{n}}{(-2z_{1} \cdot z_{2})^{\frac{\delta_{1} + \delta_{2} - \delta_{1} + n}{2}} (-2z_{1} \cdot z_{3})^{\frac{\delta_{1} + \delta_{-} - \delta_{2} + n}{2}} (-2z_{2} \cdot z_{3})^{\frac{\delta_{2} + \delta_{-} - \delta_{1} + n}{2}}$$
(2.271)

which up to a factor of 1/n agrees with the standard three-point structure (2.86) which appears in (2.89).

We now need to compute the left-hand side of (2.215). We have

$$\langle (\mathcal{D}'_{n}\mathcal{O})(x_{2}, z_{2})(\mathcal{D}'_{n}\mathcal{O})(x_{0}, z_{0}) \rangle = \mathcal{D}'_{n,2}\mathcal{D}'_{n,0}\frac{(-2z_{2} \cdot I(x_{20}) \cdot z_{0})^{J}}{x_{20}^{2\Delta}}$$

$$= \frac{(J)_{(n)}}{n!}\mathcal{D}'_{n,2}(-2z_{2} \cdot I(x_{20}) \cdot w_{0})^{n}\frac{(-2z_{2} \cdot I(x_{20}) \cdot z_{0})^{J-n}}{x_{20}^{2\Delta}}$$

$$= \sum_{k=0}^{n} \binom{n}{k}\frac{(J)_{(n)}}{n!}\frac{(n)_{(k)}(J-n)_{(n-k)}}{n!}(-2w_{2} \cdot I(x_{20}) \cdot w_{0})^{k}(-2z_{2} \cdot I(x_{20}) \cdot w_{0})^{n-k}$$

$$\times (-2w_{2} \cdot I(x_{20}) \cdot z_{0})^{n-k}\frac{(-2z_{2} \cdot I(x_{20}) \cdot z_{0})^{J-2n+k}}{x_{20}^{2\Delta}}, \qquad (2.272)$$

where we have defined $(a)_{(b)} \equiv a(a-1)\cdots(a-b+1)$. We now send $J \rightarrow n-1$ to find

$$\lim_{J \to n-1} \frac{1}{J - n + 1} \langle (\mathcal{D}'_n \mathcal{O})(x_2, z_2) (\mathcal{D}'_n \mathcal{O})(x_0, z_0) \rangle
= \sum_{k=0}^n {n \choose k} \frac{1}{n} (-1)^{n-k} (-2w_2 \cdot I(x_{20}) \cdot w_0)^k (-2z_2 \cdot I(x_{20}) \cdot w_0)^{n-k}
\times (-2w_2 \cdot I(x_{20}) \cdot z_0)^{n-k} \frac{(-2z_2 \cdot I(x_{20}) \cdot z_0)^{-1-n+k}}{x_{20}^{2\Delta}},
= \frac{1}{n} \left((-2w_2 \cdot I(x_{20}) \cdot w_0) (-2z_2 \cdot I(x_{20}) \cdot z_0) - (-2w_2 \cdot I(x_{20}) \cdot z_0) (-2z_2 \cdot I(x_{20}) \cdot w_0) \right)^n
\frac{(-2z_2 \cdot I(x_{20}) \cdot z_0)^{-1-n}}{x_{20}^{2\Delta}}
= \frac{1}{n} \left((-2H_{20}^{WW}) (-2H_{20}) - (-2H_{20}^{WZ}) (-2H_{20}^{ZW}) \right)^n \frac{(-2H_{20})^{-1-n}}{X_{20}^{\Delta+n-1}}$$
(2.273)

where H_{ij}^{AB} is defined by replacing Z_i by A_i and Z_j by B_j in H_{ij} . This can now be evaluated in the configuration (2.219) which yields, after multiplying by $-2H_{02}$, for the structure in the left-hand side of (2.215)

$$\left(\lim_{J \to n-1} (-2H_{20}) \frac{\langle (\mathcal{D}'_n \mathcal{O})(X_2, Z_2, \mathbf{W}_2)(\mathcal{D}'_n \mathcal{O}^{\dagger})(X_0, Z_0, \mathbf{W}_0) \rangle}{J - J_n} \right) \Big|_{\text{celestial}}$$

= $\frac{1}{n} \frac{\left((4z_2 \cdot z_0 w_2 \cdot w_0 - 4z_2 \cdot w_0 z_0 \cdot w_2) \right)^n}{(-2z_2 \cdot z_0)^{\delta + n}}.$ (2.274)

.

This agrees up to a factor of 1/n with the standard two-point structure (2.90) which appears in (2.89). We thus find that both sides of (2.215) differ from (2.89) by a factor of 1/n, and therefore the two equations are equivalent.

2.6.2 Selection rules in the $L[\mathcal{J}] \times L[\mathcal{J}]$ OPE

In this section we consider the light-ray operators that contribute to the light-ray OPE (2.147) of two identical charge detectors, i.e. to the two light-transforms of identical U(1) currents $L[\mathcal{J}]$. (The analysis for the non-abelian case is similar.⁴³) For concreteness we focus on d = 4 and we do not assume parity symmetry. However, the results we find will be valid in any dimension $d \ge 4$.

In d = 4 the SO(d - 1, 1) = SO(3, 1) representations are parametrized by tworow Young diagrams, which can be supplemented with self- or anti-self duality constraints. However, since we are considering a non-chiral setup, it is convenient to use real tensor representations of SO(d - 1, 1) which do not have self-duality constraints. We thus parametrize these representations by pairs (J, j). Local operators always have $J \ge j$.

First, we consider the local OPE of a U(1) current \mathcal{J} with itself. Using the counting rules of [48] it is easy to see that for sufficiently large J we have operators in the $\mathcal{J} \times \mathcal{J}$ OPE in irreps (J, 0), (J, 1) with even and odd J, and in irreps (J, 2) with even J. This generic-J behavior determines the light-ray operators that appear in the $\mathcal{J} \times \mathcal{J}$ OPE. To see this explicitly, recall that for even J + j the number of structures is given by the dimension of [48]

$$\left(S^2 \square \otimes \operatorname{Res}_{\mathrm{SO}(3)}^{\mathrm{SO}(3,1)}(J,j)\right)^{\mathrm{SO}(3)}, \qquad (2.275)$$

where \Box is the SO(3) vector irrep⁴⁴ and for odd J + j we need instead

$$\left(\wedge^2 \Box \otimes \operatorname{Res}_{\mathrm{SO}(3)}^{\mathrm{SO}(3,1)}(J,j)\right)^{\mathrm{SO}(3)}.$$
(2.276)

We have

$$S^2 \Box = \Box \oplus \bullet, \quad \wedge^2 \Box = \Box, \tag{2.277}$$

where • is the trivial representation, and the restriction of (J, j) to SO(3) is

$$\operatorname{Res}_{SO(3)}^{SO(3,1)}(J,j) = \sum_{l=j}^{J} (l)$$
(2.278)

where (l) is the spin-*l* irrep of SO(3). We get tensor structures by matching SO(3) irreps between (2.277) and (2.278). We immediately see that there are no structures

⁴³Although note the discussion of contact terms in [15].

⁴⁴Using an SO(3) irrep instead of an SO(4) irrep takes into account the conservation of \mathcal{J} .

with j > 2, and that for j = 2 the spin J must be even. For j = 0, 1 J can be of any parity.

The transverse spins that appear in the $\mathcal{J} \times \mathcal{J}$ OPE are thus 0, 1, 2. All these spins are traceless-symmetric in d - 2 dimensions and thus are allowed to appear in the celestial OPE.⁴⁵ So, the set Λ_{12} in (2.147) is given by $\Lambda_{12} = \{0, 1, 2\}$. We then have the low transverse-spin contributions, schematically

$$\mathbf{L}[\mathcal{J}] \times \mathbf{L}[\mathcal{J}] = \sum_{i} \mathbb{O}_{i,J=1,j=0}^{+} + \mathbb{O}_{i,J=1,j=1}^{+} + \mathbb{O}_{i,J=1,j=2}^{+} + \cdots$$
 (2.279)

A subtlety here is that the celestial map (2.214) typically maps multiple three-point tensor structures to zero, see appendix A.5 for details. As we discuss there, the only structures that survive are those that contain $V_{0,12}^{-1}$, which are precisely those that are polynomial for $J > J_1 + J_2 - 1$ but stop being polynomial exactly at $J = J_1 + J_2 - 1$. In our case we are interested in even-spin structures which are polynomial for J = 2 but are not polynomial for J = 1, i.e. those which disappear from the counting above as we change J = 2 to $J = 1.4^6$ We see that for j = 2 the number of structures changes from 1 to 0 and for j = 0 from 2 to 1. This happens because the SO(3) content of (J, j) changes: the spin-2 irrep disappears and cannot be paired with the spin-2 irrep in $S^2 \square$. However, for j = 1 the number of structures does not change because the only structure comes from pairing with the spin-1 irrep in $\wedge^2 \square$, and thus all j = 1 structures are annihilated by the celestial map.

The final form of the low transverse spin contributions is therefore

$$\mathbf{L}[\mathcal{J}] \times \mathbf{L}[\mathcal{J}] = \sum_{i} \mathbb{O}_{i,J=1,j=0}^{+} + \mathbb{O}_{i,J=1,j=2}^{+} + \cdots .$$
(2.280)

This is of course consistent with the fact that on the celestial sphere we have two identical scalars, and thus only even j should be allowed.⁴⁷

For higher transverse spin we find that the label γ in (2.147) is trivial because SO(d - 4) is trivial in our case. We have $\lambda(+n) = 2 + n$ and thus the higher transverse spin terms take the schematic form

$$\mathbf{L}[\mathcal{J}] \times \mathbf{L}[\mathcal{J}] = \sum_{n,i} \mathcal{D}_n \mathbb{O}_{i,J=1+n,j=2}^{(-1)^n} + \cdots .$$
(2.281)

⁴⁵Note in d > 5 three-row Young diagrams would appear in the $\mathcal{J} \times \mathcal{J}$ OPE, but the corresponding transverse spins are not allowed to appear in the celestial OPE of two scalars.

⁴⁶Here we need to detach the notion of signature (even-spin or odd-spin) from parity of J since we are analytically continuing in J. That is, for J = 1 we still use $S^2 \square$ for j = 0, 2 and $\wedge^2 \square$ for j = 1.

⁴⁷Here we assume, as usual, that the product $L[\mathcal{J}] \times L[\mathcal{J}]$ is well-defined and thus the two light-transforms commute. This requires the Regge intercept J_0 to satisfy $J_0 < J_1 + J_2 - 1 = 1$ [34].

However, since $\mathbb{O}_{i,J,j=2}^{-}$ do not appear in the $\mathcal{J} \times \mathcal{J}$ OPE, we find

$$\mathbf{L}[\mathcal{J}] \times \mathbf{L}[\mathcal{J}] = \sum_{n,i} \mathcal{D}_{2n} \mathbb{O}^+_{i,J=1+2n,j=2} + \cdots .$$
(2.282)

Note that these contributions have even transverse spin j = 2 + 2n and thus this expansion is again consistent with permutation symmetry on the celestial sphere.

Summarizing, we have the following schematic contributions to the OPE of two charge detectors in d = 4,

$$\mathbf{L}[\mathcal{J}] \times \mathbf{L}[\mathcal{J}] = \sum_{i} \left(\mathbb{O}_{i,J=1,j=0}^{+} + \mathbb{O}_{i,J=1,j=2}^{+} \right) + \sum_{n,i} \mathcal{D}_{2n} \mathbb{O}_{i,J=1+2n,j=2}^{+}.$$
 (2.283)

2.6.3 Selection rules in the $L[T] \times L[T]$ OPE

We now discuss the case of the OPE of two energy detectors, i.e. two light-transforms of *T*. We use the same setup as in the previous section, i.e. we work in d = 4 and in terms of real tensor irreps of SO(3, 1).

The transverse spins appearing in the $T \times T$ OPE are analogous to the $\mathcal{J} \times \mathcal{J}$ case. We have spins j = 0, 1, 2, 3 for both even and odd J and spin j = 4 for even J. This translates to the following analogue of (2.279)

$$\mathbf{L}[T] \times \mathbf{L}[T] = \sum_{i} \mathbb{O}_{i,J=3,j=0}^{+} + \mathbb{O}_{i,J=3,j=1}^{+} + \mathbb{O}_{i,J=3,j=2}^{+} + \mathbb{O}_{i,J=3,j=3}^{+} + \mathbb{O}_{i,J=3,j=4}^{+} + \cdots$$
(2.284)

However, we again must take care of the fact that the celestial map (2.214) annihilates some tensor structures. In this case, using the same logic as before, we find that only j = 0, 2, 4 operators have tensor structures that are non-vanishing under the celestial map. Therefore, the low-transverse spin contribution is actually

$$\mathbf{L}[T] \times \mathbf{L}[T] = \sum_{i} \mathbb{O}_{i,J=3,j=0}^{+} + \mathbb{O}_{i,J=3,j=2}^{+} + \mathbb{O}_{i,J=3,j=4}^{+} + \cdots .$$
(2.285)

Again, this is consistent with the permutation symmetry on the celestial sphere that only allows even j.

The analysis of higher transverse spin contributions is also the same as in the $\mathcal{J} \times \mathcal{J}$ case. We have from (2.147)

$$\mathbf{L}[T] \times \mathbf{L}[T] = \sum_{n,i} \mathcal{D}_n \mathbb{O}_{i,J=3+n,j=4}^{(-1)^n} + \cdots .$$
(2.286)

Taking into account that only even-spin j = 4 operators appear in $T \times T$ and combining with the low transverse spin terms, we find

$$\mathbf{L}[T] \times \mathbf{L}[T] = \sum_{i} \left(\mathbb{O}_{i,J=3,j=0}^{+} + \mathbb{O}_{i,J=3,j=2}^{+} + \mathbb{O}_{i,J=3,j=4}^{+} \right) + \sum_{n,i} \mathcal{D}_{2n} \mathbb{O}_{i,J=3+2n,j=4}^{+}.$$
(2.287)

As mentioned above, although we have derived (2.287) in d = 4, the result is valid for any $d \ge 4$.

It is interesting to ask what are the leading operators appearing at various transverse spins in the above expansion in a weakly-coupled gauge theory. At j = 0 it is well-known that the leading twist is $\tau_0 = 2$ operators which can take the schematic form,

$$\overline{\phi}\partial_{\beta_1\dot{\alpha}_1}\cdots\partial_{\beta_J\dot{\alpha}_J}\phi, \quad \overline{\psi}_{\dot{\alpha}_1}\partial_{\beta_2\dot{\alpha}_2}\cdots\partial_{\beta_J\dot{\alpha}_J}\psi_{\beta_1}, \quad \overline{F}_{\dot{\alpha}_1\dot{\alpha}_2}\partial_{\beta_3\dot{\alpha}_3}\cdots\partial_{\beta_J\dot{\alpha}_J}F_{\beta_1\beta_2}, \quad (2.288)$$

where we assume that the gauge indices are implicitly contracted, and the dotted and undotted indices are implicitly symmetrized. At j = 2 the leading twist⁴⁸ is also $\tau_2 = 2$, for the operators

$$F_{\beta_1\beta_2}\partial_{\beta_5\dot{\alpha}_5}\cdots\partial_{\beta_{J+2}\dot{\alpha}_{J+2}}F_{\beta_3\beta_4}.$$
(2.289)

To see that this is the minimal possible twist, note that $\tau \ge 2$ is the unitarity bound in d = 4 for generic-J operators. To see that there are no other operators, note that the twist of a product of symbols is bounded from below by the sum of constituent twists, and the classical twist of all fundamental fields is 1, while the twist of a derivative is 0. Therefore only products of two fundamental fields and any number of derivatives can have twist ≈ 2 , provided no indices are contracted. Fixing the value of transverse spin then leaves us with the above options.

From this argument it is clear that no j = 4 operators with twist $\tau = 2$ exist. To build the lowest-twist j = 4 operators we need to use more than 2 fundamental fields but as few as possible. Since *F* carries the most transverse spin among all fundamental fields, we find that the lowest-twist operators j = 4 have twist $\tau_4 = 4$ and are schematically given by,

$$F\partial^{J_1}F\partial^{J_2}F\partial^{J_3}F, (2.290)$$

where we keep spinor indices uncontracted and symmetrized, with $J_1+J_2+J_3+4 = J$. Note that there are multiple ways in which the gauge indices can be contracted. The

⁴⁸For all transverse spins we define twist as $\Delta - J$.

fact that j = 4 operators have higher twist than required by the unitarity bound is in general a consequence of the improved unitarity bounds of [49], which state

$$\tau \ge \max\{2, j\}. \tag{2.291}$$

In terms of the celestial quantum numbers, we get the following contributions. At j = 0 and j = 2 we have the leading contributions with dimension

$$\delta = \Delta - 1 \approx \tau_i + J - 1 = \tau_i + 2 = 4. \tag{2.292}$$

Note that this dimension corresponds to the singularity of the form $\theta^{\delta-\delta_1-\delta_2} = \theta^{-2}$. For $j = 4 + 2n \ge 4$ we get

$$\delta = \Delta - 1 \approx \tau_4 + J - 1 = \tau_4 + 2n + 2 = 6 + 2n, \tag{2.293}$$

which corresponds to the leading short-angle dependence of the form θ^{2n} . Since the leading classical twists τ_j are all at the (improved) unitarity bounds, the anomalous dimensions should be positive. Therefore, the leading short-angle asymptotics from these contributions in the interacting theory should be softer than the ones given above.

2.7 Example: event shape in N = 4 SYM

In this section we consider an example of an event shape that includes the transverse spin structures discussed above. More precisely, we consider the following event shape in N = 4 SYM:

$$\langle O_{20'}(p) | \mathbf{L}[O_{20'}](\infty, z_1) \mathbf{L}[O_{20'}](\infty, z_2) | J(p, z_3) \rangle,$$
 (2.294)

where J is the R-symmetry current, which is in the same multiplet as the half-BPS operator $O_{20'}$. We will first compute (2.294) directly by performing the light transform of the relevant four-point function. Next, we compute it using the light-ray OPE formula. In both cases, we will derive a Ward identity that relates the event shape (2.294) to the energy-energy correlator calculated in [15]. The result is given by (2.323) and (2.357). Despite the simplicity of our result, the fact that the Ward identity can be derived in two independent ways still provides a nontrivial check of our formulas.

2.7.1 Direct computation

The computation takes a few steps, summarized as follows. We start with the expression for the correlator $\langle O_{20'}O_{20'}O_{20'}J\rangle$ in terms of the scalar correlator

 $\langle O_{20'}O_{20'}O_{20'}O_{20'}\rangle$. The two are related via the superconformal Ward identities [50]. Then, we go to the Mellin space representation of the correlator, and perform the light transforms there. Finally, we Fourier transform the separation of the in and out states to obtain the desired event shape.

The relevant four-point function is given by the following expression, see (3.15) in [50],

$$\langle O_{20'}(x_4) O_{20'}(x_1) O_{20'}(x_2) J_{\alpha \dot{\alpha}, aa'}(x_3) \rangle$$

$$= \frac{1}{4} (\partial_{x_3})^{\beta}_{\dot{\alpha}} (y_{12}^2 y_{14}^2 Y_{324} - v y_{12}^2 y_{24}^2 Y_{314} - u y_{24}^2 y_{14}^2 Y_{321})_{aa'}$$

$$\times \langle X_{324}, X_{314} \rangle_{(\alpha \beta)} \frac{\Phi(u, v)}{x_{12}^2 x_{24}^2 x_{14}^2},$$

$$(2.295)$$

where our spinor conventions can be found in appendix A.6. The function $\Phi(u, v)$ is the part of the four-point function of $O_{20'}$ s that contains the nontrivial dynamical data, see e.g. section 7.2 in [15]. It satisfies $\Phi(u, v) = \Phi(v, u) = \frac{1}{v} \Phi(\frac{u}{v}, \frac{1}{v})$. The rest of the ingredients are various kinematical factors that require some unpacking. The $\alpha, \dot{\alpha} = 1, 2$ are spinor indices for the Lorentz group $SU(2)_L \times SU(2)_R$. The y_i are auxiliary variables keeping track of the SU(4) R-symmetry, see [50] for details. The structures X_{ijk} and Y_{ijk} are defined as follows:

$$(X_{ijk})_{\alpha\dot{\alpha}} = \frac{(x_{ij})_{\alpha\dot{\beta}}}{x_{ij}^2} (x_{jk})^{\dot{\beta}\gamma} \frac{(x_{ki})_{\gamma\dot{\alpha}}}{x_{ik}^2} , \quad (Y_{ijk})_{aa'} = (y_{ij})_{ab'} (y_{jk})^{b'b} (y_{ki})_{ba'}.$$
(2.296)

The commutator bracket \langle , \rangle for spinor indices⁴⁹ is defined as

$$\langle a, b \rangle_{(\alpha\beta)} \equiv a_{\alpha\dot{\alpha}} b^{\dot{\alpha}\gamma} \epsilon_{\gamma\beta} - b_{\alpha\dot{\alpha}} a^{\dot{\alpha}\gamma} \epsilon_{\gamma\beta} \,. \tag{2.297}$$

Note that $\langle a, b \rangle_{(\alpha\beta)}$ is automatically symmetric under permutation of α and β .

To connect with the event shape, we set $x_4 = 0$, and x_3 will eventually be Fourier transformed with momentum p. For now, we can replace the derivative $(\partial_{x_3})^{\beta}_{\dot{\alpha}}$ with $\frac{i}{2}p^{\beta}_{\dot{\alpha}}$. We pass to index-free notation by introducing the polarization vector $\frac{1}{2}z_3^{\dot{\alpha}\alpha}$, such that $z_3^2 = 0$, and contracting $\frac{1}{2}z_3^{\dot{\alpha}\alpha}J_{\alpha\dot{\alpha}} = z_3^{\mu}J_{\mu}$. Contracting with the polarization vector produces a term

$$p^{\beta}_{\ \dot{\alpha}} z^{\dot{\alpha}\alpha}_3 = \frac{1}{2} \langle z_3, p \rangle^{(\beta\alpha)} + (z_3 \cdot p) \epsilon^{\beta\alpha} . \qquad (2.298)$$

⁴⁹We use angular brackets to denote the commutator in spinor indices to avoid a clash with the commutator in vector indices denoted by the traditional square brackets.

In terms of the polarization vector, the correlator becomes

$$\langle O_{20'}(x_4)O_{20'}(x_1)O_{20'}(x_2)J_{aa'}(x_3, z_3)\rangle = \frac{i}{32} \langle y_{12}^2 y_{14}^2 Y_{324} - v y_{12}^2 y_{24}^2 Y_{314} - u y_{24}^2 y_{14}^2 Y_{321}\rangle_{aa} \times \langle z_3, p \rangle^{(\alpha\beta)} \langle X_{324}, X_{314} \rangle_{(\alpha\beta)} \frac{\Phi(u, v)}{x_{12}^2 x_{24}^2 x_{14}^2}.$$

$$(2.299)$$

In this formula we only Fourier transformed the external derivative while keeping the rest in coordinate space. The contraction of the brackets can be performed by the identity

$$\langle a,b\rangle^{(\alpha\beta)}\langle c,d\rangle_{(\alpha\beta)} = 8(g^{\mu\sigma}g^{\nu\rho} - g^{\mu\rho}g^{\nu\sigma} + i\epsilon^{\mu\nu\rho\sigma})a_{\mu}b_{\nu}c_{\rho}d_{\sigma}.$$
(2.300)

To compute the event shape, we place the detectors at embedding space coordinates

$$X_i = X_\infty = (0, 1, 0), \quad Z_i = (0, 0, z_i), \quad \text{for } i = 1, 2,$$
 (2.301)

and take the light-transforms

$$\mathbf{L}[O_{20'}](\infty, z_i) = \int_{-\infty}^{\infty} d\alpha_i \, O_{20'}(Z_i - \alpha_i X_\infty), \qquad (2.302)$$

while the external states are placed at $X_3 = (1, x_3^2, x_3)$ and $X_4 = (1, 0, 0)$.

We find it convenient to work in the x Poincaré patch and approach the spatial infinity insertion as follows

$$\mathbf{L}[O_{20'}](\infty, z_i) = \int_{-\infty}^{\infty} d\alpha_i \lim_{r_i \to \infty} r_i^2 O_{20'}(r_i z_i + \alpha_i \overline{z}_i), \qquad (2.303)$$

where $\overline{z}_i^2 = 0$, $(-z_i \cdot \overline{z}_i) = \frac{1}{2}$, and \overline{z}_i is arbitrary otherwise. In the embedding space, (2.303) corresponds to choosing

$$Z_{i} = \lim_{r \to \infty} Z_{i}^{r} = \left(\frac{1}{r_{i}}, 0, z_{i}\right),$$

$$X_{\infty} = \lim_{r \to \infty} X_{\infty}^{r_{i}} = \left(0, 1, \frac{\overline{z_{i}}}{r_{i}}\right).$$
(2.304)

Note that $(Z_i^r)^2 = (X_{\infty}^{r_i})^2 = (Z_i^r \cdot X_{\infty}^{r_i}) = 0$. It is clear from the definition (2.302) that the final result does not depend on the particular choice of \overline{z}_i . For convenience, we also define the null coordinates

$$x_{i-} \equiv (-x_i \cdot z_i) = \alpha_i / 2 ,$$

$$x_{3i-} \equiv (-x_3 \cdot z_i) - x_{i-} , \qquad (2.305)$$

for i = 1, 2.

Next we evaluate the integrand in the kinematics above. The cross ratios u and v take the form

$$u = \frac{x_3^2 z_1 \cdot z_2}{2x_{2-}x_{31-}}, \qquad v = \frac{x_{1-}x_{32-}}{x_{2-}x_{31-}}, \qquad (2.306)$$

and the commutator becomes

$$\langle X_{324}, X_{314} \rangle_{(\alpha\beta)} = \frac{1}{2} \frac{1}{x_3^2 x_{32-} x_{31-}} \left(-x_{32-} \langle z_1, x_3 \rangle_{(\alpha\beta)} + x_{31-} \langle z_2, x_3 \rangle_{(\alpha\beta)} + \frac{1}{2} x_3^2 \langle z_2, z_1 \rangle_{(\alpha\beta)} \right).$$
(2.307)

With these expressions, the light transforms are evaluated by the integral

$$\langle O_{20'}(0) | \mathbf{L}[O_{20'}](\infty, z_1) \mathbf{L}[O_{20'}](\infty, z_2) | J_{aa'}(x_3, z_3) \rangle$$

= $\int_{-\infty}^{\infty} dx_{1-} dx_{2-} \frac{i}{32} (y_{12}^2 y_{14}^2 Y_{324} - v y_{12}^2 y_{24}^2 Y_{314} - u y_{24}^2 y_{14}^2 Y_{321})_{aa'}$
 $\times \langle z_3, p \rangle^{(\alpha\beta)} \langle X_{324}, X_{314} \rangle_{(\alpha\beta)} \frac{\Phi(u, v)}{(-2z_1 \cdot z_2)x_{1-}x_{2-}}.$ (2.308)

To perform the light transform integrals above, it is very convenient to use the Mellin representation for $\Phi(u, v)$, see e.g. [51],

$$\Phi(u,v) = \frac{v}{u} \int_{C_0} \frac{d\gamma_{12}d\gamma_{14}}{(2\pi i)^2} \Gamma(\gamma_{12})^2 \Gamma(\gamma_{14})^2 \Gamma(2-\gamma_{12}-\gamma_{14})^2 M(\gamma_{12},\gamma_{14}) u^{-\gamma_{12}} v^{-\gamma_{14}},$$

$$C_0: \quad \operatorname{Re}[\gamma_{12}] > -1, \quad \operatorname{Re}[\gamma_{14}], \operatorname{Re}[\gamma_{13}] > 1, \qquad (2.309)$$

where $\gamma_{12} + \gamma_{13} + \gamma_{14} = 2$. The condition $\operatorname{Re}[\gamma_{13}] > 1$ thus becomes $\operatorname{Re}[\gamma_{12} + \gamma_{14}] < 1$. The weak and strong coupling results take the form

$$M^{\text{weak}}(\gamma_{12}, \gamma_{14}) = -\frac{a}{4} \frac{\gamma_{12}^2}{(\gamma_{14} - 1)^2 (\gamma_{13} - 1)^2},$$

$$M^{\text{strong}}(\gamma_{12}, \gamma_{14}) = -\frac{1}{2} \frac{\gamma_{12}^2 (1 + \gamma_{12})}{(\gamma_{14} - 1)(\gamma_{13} - 1)}.$$
 (2.310)

We can plug the Mellin representation (2.309) above into (2.308), and perform the light-transform integrals using the formula

$$\int_{-\infty}^{\infty} dx_{-} (x_{-} + i\epsilon)^{-a} ((-z \cdot x) - x_{-} + i\epsilon)^{-b} = -2\pi i (-z \cdot x)^{1-a-b} \frac{\Gamma(a+b-1)}{\Gamma(a)\Gamma(b)} .$$
(2.311)

The integral converges for Re[a + b] > 1.

At this point, we observe that the terms in the correlator proportional to y_{12}^2 produce a divergent result. This means that the event shapes for the corresponding R-symmetry structures are not well-defined. To obtain a well-defined event shape, we set $y_{12}^2 = 0$, so that only the $-uy_{24}^2y_{14}^2Y_{321}$ term survives. This term produces a finite result for the integral, which converges for Re[γ_{12}] < 0. Note that the condition $y_{12}^2 = 0$ keeps representations **84**, **105**, **175** in the OPE of scalars $O_{20'}$ [51], whereas in the $J \times O_{20'}$ OPE we have the representations

$$15 \times 20' = 15 + 20' + 45 + \overline{45} + 175 . \qquad (2.312)$$

Therefore, the only representation that appears in the event shape determined by $y_{12}^2 = 0$ is **175**. From now on, we will focus on this finite event shape in the **175** R-symmetry channel:

$$\langle O_{20'} | \mathbf{L}[O_{20'}] \mathbf{L}[O_{20'}] | J_{aa'} \rangle |_{y_{12}^2 = 0} = (y_{24}^2 y_{14}^2 Y_{321})_{aa'} \langle O_{20'} | \mathbf{L}[O_{20'}] \mathbf{L}[O_{20'}] | J \rangle_{175}.$$
(2.313)

Performing the light-transforms for the event shape in the **175** channel, we arrive at the expression

$$\langle O_{20'}(p) | \mathbf{L}[O_{20'}](\infty, z_1) \mathbf{L}[O_{20'}](\infty, z_2) | J(p, z_3) \rangle_{\mathbf{175}} = -\frac{i}{32} \frac{1}{(z_1 \cdot z_2)^2} \int_{C_0} \frac{d\gamma_{12} d\gamma_{14}}{(2\pi i)^2} \frac{M(\gamma_{12}, \gamma_{14})}{\gamma_{12}} \frac{2\pi^4}{(\sin \pi \gamma_{12})^2} \mathcal{D}_p \int d^4x \frac{e^{ipx}}{x^4} \gamma^{\gamma_{12}-1},$$
(2.314)

where
$$\gamma = 2 \frac{(-x \cdot z_1)(-x \cdot z_2)}{x^2(z_1 \cdot z_2)}$$
. The differential operator \mathcal{D}_p is given by

$$\mathcal{D}_p \equiv \langle z_3, p \rangle^{(\alpha\beta)} \left((\gamma_{14} - 1) \langle z_1, \partial_p \rangle_{(\alpha\beta)} z_2 \cdot \partial_p - (\gamma_{13} - 1) \langle z_2, \partial_p \rangle_{(\alpha\beta)} z_1 \cdot \partial_p - \frac{1}{2} \gamma_{12} \langle z_2, z_1 \rangle_{(\alpha\beta)} \partial_p^2 \right)$$
(2.315)

Finally, the Fourier transform can be performed using the following master formula:

$$\int d^d x \frac{e^{ip \cdot x}}{(x^2 - i\epsilon x^0)^a} \gamma^b = \theta(p) \times 2^{1-2a+d} \pi^{1+\frac{d}{2}} (-p^2)^{a-\frac{d}{2}} \frac{\zeta^{-b} \,_2 F_1(-b, -b, 1+a-b-\frac{d}{2}, \zeta)}{\Gamma(a+b)\Gamma(1+a-b-\frac{d}{2})}$$
(2.316)

where we introduced $\theta(p) \equiv \theta(p^0)\theta(-p^2)$ and the cross ratio

$$\zeta = \frac{(-2z_1 \cdot z_2)(-p^2)}{(-2p \cdot z_1)(-2p \cdot z_2)}.$$
(2.317)

Setting a = 2, $b = \gamma_{12} - 1$ and d = 4 we obtain

$$\int d^4x \frac{e^{ipx}}{x^4} \gamma^{\gamma_{12}-1} = \frac{2\pi^2 \sin \pi \gamma_{12}}{\gamma_{12}} \int_0^\zeta dw w^{-\gamma_{12}} (1-w)^{\gamma_{12}-1}.$$
 (2.318)

Acting with \mathcal{D}_p on the result of the Fourier transform (2.318) and using crossing symmetry of the Mellin amplitude $M(\gamma_{12}, \gamma_{14}) = M(\gamma_{12}, \gamma_{13})$, we get

$$\langle O_{20'}(p) | \mathbf{L}[O_{20'}](\infty, z_1) \mathbf{L}[O_{20'}](\infty, z_2) | J(p, z_3) \rangle_{\mathbf{175}}$$

$$= i \frac{[z_3, p] \cdot [z_1, z_2] \theta(p)}{(-2z_1 \cdot z_2)^2} \int_{C_0} \frac{d\gamma_{12} d\gamma_{14}}{(2\pi i)^2} M(\gamma_{12}, \gamma_{14}) \frac{\pi^6}{\sin \pi \gamma_{12}} \frac{4}{p^2} \left(\frac{\zeta}{1-\zeta}\right)^{1-\gamma_{12}},$$

$$(2.319)$$

where the commutator $[a, b]^{\mu\nu}$ is defined in the same way as (2.30), and contracting a pair gives

$$[a,b] \cdot [c,d] = 2 [(a \cdot c)(b \cdot d) - (a \cdot d)(b \cdot c)].$$
(2.320)

We can rewrite the result above as

$$\langle O_{20'}(p) | \mathbf{L}[O_{20'}](\infty, z_1) \mathbf{L}[O_{20'}](\infty, z_2) | J(p, z_3) \rangle_{\mathbf{175}} = 16i\pi^5 \frac{[z_3, p] \cdot [z_1, z_2](-p^2)\theta(p)}{(-2p \cdot z_1)^2 (-2p \cdot z_2)^2} \mathcal{F}_{JO}(\zeta) ,$$
(2.321)

where

$$\mathcal{F}_{JO}(\zeta) = -\frac{1}{4\zeta^2} \int_{C_0} \frac{d\gamma_{12} d\gamma_{14}}{(2\pi i)^2} M(\gamma_{12}, \gamma_{14}) \frac{\pi}{\sin \pi \gamma_{12}} \left(\frac{\zeta}{1-\zeta}\right)^{1-\gamma_{12}} .$$
(2.322)

The result (2.322), see e.g. [24], immediately implies that

$$\mathcal{F}_{JO}(\zeta) = -\frac{\zeta}{2} \mathcal{F}_{\mathcal{E}}(\zeta), \qquad (2.323)$$

where $\mathcal{F}_{\mathcal{E}}(\zeta)$ was defined in [15] and computes the energy-energy correlator.

Using (2.323) and the formulas [15], or directly computing the Mellin integral in (2.322) with the Mellin amplitudes (2.310), we get the results at weak and at strong coupling

$$\mathcal{F}_{JO}^{\text{weak}}(\zeta) = \frac{a}{8} \frac{\log(1-\zeta)}{\zeta(1-\zeta)}, \qquad \mathcal{F}_{JO}^{\text{strong}}(\zeta) = -\frac{\zeta}{4}.$$
(2.324)

2.7.2 Computation using the light-ray OPE

Now we compute the event shape (2.294) using the light-ray OPE formula. The complete light-ray OPE formula for two scalars including higher transverse spin terms is given by (2.138). Plugging the formula into (2.294), we have

$$\langle O_{20'}(p) | \mathbf{L}[O_{20'}](\infty, z_1) \mathbf{L}[O_{20'}](\infty, z_2) | J(p, z_3) \rangle$$

$$= \pi i \int_{2-i\infty}^{2+i\infty} \frac{d\Delta}{2\pi i} \left(C_{\Delta-1,0}(z_1, z_2, \partial_z) \langle O_{20'}(p) | \mathbb{O}_{\Delta,J=-1}^+(\infty, z) | J(p, z_3) \rangle \right)$$

$$+ C_{\Delta-1,1}(z_1, z_2, \partial_z, \partial_w) \langle O_{20'}(p) | (\mathcal{D}_1 \mathbb{O}_{\Delta,J=0}^-)(\infty, z, w) | J(p, z_3) \rangle \right)$$

$$= -\pi i \int_{2-i\infty}^{2+i\infty} \frac{d\Delta}{2\pi i} \left(C_a^+(\Delta, -1) C_{\Delta-1,0}(z_1, z_2, \partial_z) \langle O_{20'}(p) | \mathbf{L}[O_{\Delta,-1}](\infty, z) | J(p, z_3) \rangle_+^{(a)} \right)$$

$$+ C_a^-(\Delta, 0) C_{\Delta-1,1}(z_1, z_2, \partial_z, \partial_w) \langle O_{20'}(p) | \mathcal{D}_1 \mathbf{L}[O_{\Delta,0}](\infty, z, w) | J(p, z_3) \rangle_-^{(a)}$$

$$(2.325)$$

where $C_a^{\pm}(\Delta, J)$ is a coefficient function that encodes the OPE data. It has poles of the form

$$C_a^{\pm}(\Delta, J) \sim -\frac{p_{i,J}^{(a)}}{\Delta - \Delta_{i,J}}$$
(2.326)

where $p_{i,J}^{(a)}$ and $\Delta_{i,J}$ are the product of OPE coefficients and the scaling dimension of an exchanged operator. Note that there's just one structure label in $C_a^{\pm}(\Delta, J)$ because the three-point function from the two $O_{20'}$'s only has one tensor structure. In the second equality in (2.325), we use the relation between light-ray operators $\mathbb{O}_{\Delta,J}^{\pm}$ and $C_a^{\pm}(\Delta, J)$ [11]

$$\langle O_{20'} \mathbb{O}_{\Delta,J}^{\pm} J \rangle_{\Omega} = -C_a^{\pm}(\Delta, J) \langle 0 | O_{20'} \mathbf{L}[O_{\Delta,J}] J | 0 \rangle_{\pm}^{(a)}, \qquad (2.327)$$

where $\langle 0|O_{20'}\mathbf{L}[O_{\Delta,J}]J|0\rangle_{\pm}^{(a)}$ is the analytic continuation of the continuous-spin Wightman function $\langle 0|O_{20'}\mathbf{L}[O_{\Delta,J}]J|0\rangle^{(a)}$ from either even or odd spin. It can be defined straightforwardly following section 2.5.1. We will give their explicit expressions later in section 2.7.2.

Furthermore, as discussed in section 2.7.1, we focus on the **175** R-symmetry channel event shape defined in (2.313). Since the **175** representation is antisymmetric under the exchange of $O_{20'}$'s, the OPE should only contain operators with odd spin, and we have $C^+(\Delta, J) = 0$ for all J. The **175** R-symmetry channel event shape is then

given by

$$\langle O_{20'}(p) | \mathbf{L}[O_{20'}](\infty, z_1) \mathbf{L}[O_{20'}](\infty, z_2) | J(p, z_3) \rangle_{175} = -\pi i \int_{2-i\infty}^{2+i\infty} \frac{d\Delta}{2\pi i} C_a'^{-}(\Delta, 0) C_{\Delta-1,1}(z_1, z_2, \partial_z, \partial_w) \langle O_{20'}(p) | \mathcal{D}_1 \mathbf{L}[O_{\Delta,0}](\infty, z, w) | J(p, z_3) \rangle_{-}^{(a)},$$
(2.328)

where $C_a^{\prime-}(\Delta, J)$ is simply given by the coefficient function $C_a^-(\Delta, J)$ without the R-symmetry factor $(y_{24}^2y_{14}^2Y_{321})$.

Celestial blocks

In order to compute (2.328), we first note that by Lorentz invariance and homogeneity,

$$\langle O_{20'}(p) | \mathcal{D}_1 \mathbf{L}[O_{\Delta,0}](\infty, z, w) | J(p, z_3) \rangle_{-}^{(a)} \propto (2w \cdot z_3 z \cdot p - 2w \cdot p z \cdot z_3) (-2p \cdot z)^{-\delta - 1}$$

$$(2.329)$$

Therefore, we need to solve for the j = 1 celestial block defined by

$$C_{\delta,1}(z_1, z_2, \partial_z, \partial_w) \left((2w \cdot z_3 z \cdot p - 2w \cdot p z \cdot z_3) (-2p \cdot z)^{-\delta - 1} \right).$$
(2.330)

This is the higher transverse spin version of the celestial block computed in [15]. Lorentz invariance and homogeneity imply that

$$C_{\delta,1}(z_1, z_2, \partial_z, \partial_w) \left([w, z] \cdot [z_3, p] (-2p \cdot z)^{-\delta - 1} \right)$$

= $\frac{(-p^2)^{\frac{\delta_1 + \delta_2 - \delta + 1}{2}}}{(-2p \cdot z_1)^{\delta_1 + 1} (-2p \cdot z_2)^{\delta_2 + 1}}$
× $\left([z_3, p] \cdot [z_1, z_2] g(\zeta) + \{z_3, p\} \cdot \{z_1, z_2\} h(\zeta) - \frac{4p \cdot z_1 p \cdot z_2 p \cdot z_3}{p^2} h(\zeta) \right),$
(2.331)

where once again $[a, b]^{\mu\nu}$ is defined in the same way as (2.30), and similarly

$$\{a,b\}^{\mu\nu} = a^{\mu}b^{\nu} + b^{\mu}a^{\nu}, \qquad (2.332)$$

and the cross ratio ζ is given by (2.317). One way to obtain the functions $g(\zeta)$ and $h(\zeta)$ is using the fact that (2.331) is an eigenvector of the quadratic Casimir of the Lorentz group acting simultaneously on z_1, z_2 with eigenvalue $\delta(\delta - d + 2) + d - 3$. Proceeding this way, one gets two coupled second-order inhomogeneous differential equations of $g(\zeta)$ and $h(\zeta)$, and their boundary conditions are given by the OPE

limit of (2.331). However, finding the solutions to these differential equations is a nontrivial task. Furthermore, the system of differential equations gets more and more complicated when one has even higher transverse spin. We would like a method that allows us to compute celestial blocks with general transverse spin. Fortunately, this can be achieved by using weight-shifting operators [39] and the j = 0 celestial block calculated in [15].

First, note that $[w, z] \cdot [z_3, p](-2p \cdot z)^{-\delta-1}$ can be written in terms of a "bubble diagram"

$$[w, z] \cdot [z_3, p] (-2p \cdot z)^{-\delta - 1} = \frac{\mathcal{D}_{z, w}^{0+} \cdot \mathcal{D}_{z, w}^{0-}}{(d - 4)(\delta - 1)(\delta - d + 3)} \left([w, z] \cdot [z_3, p] (-2p \cdot z)^{-\delta - 1} \right),$$
(2.333)

where $\mathcal{D}_{z,w}^{0+\mu}$ and $\mathcal{D}_{z,w}^{0-\mu}$ are weight-shifting operators defined in [39], with the embedding space coordinates (X, Z) replaced with (z, w). The explicit expression of $\mathcal{D}_{z,w}^{0+\mu}$ is also given in (2.66). The operator $\mathcal{D}_{z,w}^{0+\mu}$ increases the transverse spin and $\mathcal{D}_{z,w}^{0-\mu}$ decreases the transverse spin, so acting $\mathcal{D}_{z,w}^{0+} \cdot \mathcal{D}_{z,w}^{0-}$ will basically give us the same expression with some overall factor. One can perform crossing on the weight-shifting operator $\mathcal{D}_{z,w}^{0-\mu}$ such that

$$\mathcal{D}_{z,w}^{0-\mu}\left([w,z]\cdot[z_3,p](-2p\cdot z)^{-\delta-1}\right) = \mathcal{D}_p^{\mu}(-2p\cdot z)^{-\delta},$$
(2.334)

where \mathcal{D}_p^{μ} is a differential operator acting on *p*. By explicitly evaluating the left-hand side of (2.334), we find that \mathcal{D}_p^{μ} is given by

$$\mathcal{D}_{p}^{\mu} = -\frac{(d-4)}{\delta} \left((\delta - d + 3)p^{\mu}z_{3} \cdot \frac{\partial}{\partial p} + p^{2}z_{3} \cdot \frac{\partial}{\partial p}\frac{\partial}{\partial p_{\mu}} + (\delta + 1)z_{3} \cdot p\frac{\partial}{\partial p_{\mu}} + \delta(\delta - d + 3)z_{3}^{\mu} \right).$$
(2.335)

Alternatively, $(-2p \cdot z)^{-\delta}$ and $[w, z] \cdot [z_3, p](-2p \cdot z)^{-\delta-1}$ can be viewed as spin 0 and spin 1 bulk-to-boundary propagators in AdS_{d-1}/CFT_{d-2} , and \mathcal{D}_p^{μ} is simply an AdS weight-shifting operator [52]. In particular, it is the AdS weight-shifting operator that increases spin by 1 multiplied by a bulk-to-boundary 6*j* symbol.

On the other hand, one can also perform crossing between $\mathcal{D}_{z,w}^{0+\mu}$ and the OPE differential operator $C_{\delta,1}$:

$$C_{\delta,1}^{(\delta_1,0,\delta_2,0)}\mathcal{D}_{z,w}^{0+\mu} = c_1 \mathcal{D}_{z_1,w_1}^{0-\mu} C_{\delta,0}^{(\delta_1,1,\delta_2,0)} + c_2 \mathcal{D}_{z_1,w_1}^{+0\mu} C_{\delta,0}^{(\delta_1-1,0,\delta_2,0)} + c_3 \mathcal{D}_{z_1,w_1}^{-0\mu} C_{\delta,0}^{(\delta_1+1,0,\delta_2,0)}$$
(2.336)

where $C_{\delta,j}^{(\delta_1,j_1,\delta_2,j_2)}$ is the OPE differential operator of $\mathcal{P}_{\delta,j} \in \mathcal{P}_{\delta_1,j_1} \times \mathcal{P}_{\delta_2,j_2}$. To obtain the coefficients c_1, c_2 and c_3 , we apply (2.336) to a scalar two-point function $\langle \mathcal{P}_{\delta} \mathcal{P}_{\delta} \rangle$ and compare the two sides of the equation. The result is given by

$$c_{1} = \frac{\delta(\delta + \delta_{1} - \delta_{2} - 1)}{(d - 4)(\delta_{1} - 1)(-\delta_{1} + d - 3)}$$

$$c_{2} = \frac{\delta}{(\delta_{1} - 1)(\delta_{1} - 2)(-\delta_{1} + d - 3)(2\delta_{1} - d + 2)}$$

$$c_{3} = \frac{\delta(-\delta_{1} - \delta_{2} + \delta + d - 3)(5 - 2d + \delta + 3\delta_{1} - \delta_{2})}{2(-\delta_{1} + d - 3)(2\delta_{1} - d + 2)}.$$
(2.337)

Combining (2.333), (2.334) and (2.336), we have

$$C_{\delta,1}(z_1, z_2, \partial_z, \partial_w) \left([w, z] \cdot [z_3, p] (-2p \cdot z)^{-\delta - 1} \right)$$

= $\frac{1}{(d - 4)(\delta - 1)(\delta - d + 3)} \left(c_1 \mathcal{D}_p \cdot \mathcal{D}_{z_1, w_1}^{0-} C_{\delta, 0}^{(\delta_1, 1, \delta_2, 0)} + c_2 \mathcal{D}_p \cdot \mathcal{D}_{z_1, w_1}^{+0} C_{\delta, 0}^{(\delta_1 - 1, 0, \delta_2, 0)} + c_3 \mathcal{D}_p \cdot \mathcal{D}_{z_1, w_1}^{-0} C_{\delta, 0}^{(\delta_1 + 1, 0, \delta_2, 0)} \right) (-2p \cdot z)^{-\delta}.$
(2.338)

Now the calculation is straightforward since $C_{\delta,0}(-2p \cdot z)^{-\delta}$ is simply the j = 0 celestial block calculated in [15].⁵⁰ Finally, we obtain that the functions $g(\zeta)$ and $h(\zeta)$ in the j = 1 celestial block (2.331) are given by

$$g(\zeta) = \zeta^{\frac{\delta-\delta_{1}-\delta_{2}-1}{2}} \left(\frac{1}{2} \left(1 + \frac{\delta_{2}-\delta_{1}}{\delta-d+3} \right)_{2} F_{1} \left(\frac{\delta+\delta_{1}-\delta_{2}-1}{2}, \frac{\delta+\delta_{2}-\delta_{1}+1}{2}, \delta+2-\frac{d}{2}, \zeta \right) \right. \\ \left. + \frac{1}{2} \left(1 + \frac{\delta_{1}-\delta_{2}}{\delta-d+3} \right)_{2} F_{1} \left(\frac{\delta+\delta_{1}-\delta_{2}+1}{2}, \frac{\delta+\delta_{2}-\delta_{1}-1}{2}, \delta+2-\frac{d}{2}, \zeta \right) \right) \right. \\ h(\zeta) = \frac{\zeta^{\frac{\delta-\delta_{1}-\delta_{2}+1}{2}}}{(1-\zeta)} \left(\frac{1}{2} \left(1 + \frac{\delta_{2}-\delta_{1}}{\delta-d+3} \right)_{2} F_{1} \left(\frac{\delta+\delta_{1}-\delta_{2}-1}{2}, \frac{\delta+\delta_{2}-\delta_{1}+1}{2}, \delta+2-\frac{d}{2}, \zeta \right) \right. \\ \left. - \frac{1}{2} \left(1 + \frac{\delta_{1}-\delta_{2}}{\delta-d+3} \right)_{2} F_{1} \left(\frac{\delta+\delta_{1}-\delta_{2}-1}{2}, \frac{\delta+\delta_{2}-\delta_{1}-1}{2}, \delta+2-\frac{d}{2}, \zeta \right) \right).$$

$$(2.339)$$

One can check that (2.339) is indeed the solution to the Casimir differential equations of $g(\zeta)$ and $h(\zeta)$ obtained by applying the quadratic Casimir to (2.331). For $\delta_1 = \delta_2 = \delta_{\phi}$, we have $h(\zeta) = 0$ and

$$g(\zeta) = \zeta \frac{\delta^{-2\Delta_{\phi}-1}}{2} {}_{2}F_{1}\left(\frac{\delta-1}{2}, \frac{\delta+1}{2}, \delta+2-\frac{d}{2}, \zeta\right)$$
$$= \zeta \frac{\delta^{-2\Delta_{\phi}}}{2} {}_{2}F_{1}\left(\frac{\Delta-2}{2}, \frac{\Delta}{2}, \Delta+1-\frac{d}{2}, \zeta\right).$$
(2.340)

⁵⁰The celestial block $C_{\delta,0}^{(\delta_1,1,\delta_2,0)}(-2p \cdot z)^{-\delta}$ was not calculated in [15], but it can be easily obtained using the Casimir equation method.

Having solved for the celestial blocks, we can now calculate $\langle O_{20'}(p) | \mathcal{D}_1 \mathbf{L}[O_{\Delta,0}] | J(p, z_3) \rangle^{(a)}$ to obtain the proportionality constant in (2.329). For an operator O with weights (Δ, J) , the three-point function $\langle 0 | O_{20'}OJ | 0 \rangle^{(a)}$ has two tensor structures. With our choice of conventions in appendix A.2, their expressions in the embedding space for integer J are given by

$$\langle 0|O_{20'}(X_4)O(X_2, Z_2)J(X_3, Z_3)|0\rangle^{(1)} = \frac{(-2V_{2,34})^J(-2V_{3,42})}{X_{24}^{\frac{\Delta+J-2}{2}}X_{23}^{\frac{\Delta+J+2}{2}}X_{34}^{\frac{6-\Delta-J}{2}}}$$
(2.341)

$$\langle 0|O_{20'}(X_4)O(X_2, Z_2)J(X_3, Z_3)|0\rangle^{(2)} = \frac{(-2V_{2,34})^{J-1}(-2H_{23})}{X_{24}^{\frac{\Delta + J-2}{2}}X_{23}^{\frac{\Delta + J+2}{2}}X_{34}^{\frac{E-\Delta - J}{2}}},$$
(2.342)

where the structures $V_{i,jk}$ and H_{ij} are defined in (2.217) and (2.218). Note that after setting J = 0, the first structure (2.341) is still a valid three-point function of local operators, and hence it should be annihilated by the shortening condition \mathcal{D}'_1 (or equivalently $\mathcal{D}_1 \mathbf{L}$). So we can just consider the second structure (2.342). As explained in section 2.5.1, its analytic continuation for complex J should be given by

$$\langle 0|O_{20'}(X_4)O(X_2, Z_2)J(X_3, Z_3)|0\rangle_{\pm}^{(2)} = \mp \frac{(-2V_{2,43})^{J-1}(-2H_{23})}{X_{24}^{\frac{\Delta + J-2}{2}}X_{23}^{\frac{\Delta + J+2}{2}}X_{34}^{\frac{6-\Delta - J}{2}}}.$$
 (2.343)

where \mp is due to the $(-2V_{2,34})^{J-1}$ factor. Using the algorithm for computing light transform and Fourier transform of three-point functions described in [34] and applying the differential operator \mathcal{D}_1 , we obtain

$$\langle O_{20'}(p) | \mathcal{D}_{1} \mathbf{L}[O_{\Delta,0}](\infty, z, w) | J(p, z_{3}) \rangle_{-}^{(2)}$$

$$= 16\pi^{4} \frac{\Gamma(\Delta - 2)}{\Gamma(\frac{\Delta}{2})^{2} \Gamma(\frac{\Delta - 2}{2}) \Gamma(\frac{4 - \Delta}{2})}$$

$$\times (-2p \cdot z)^{-\Delta} (2w \cdot z_{3}z \cdot p - 2w \cdot pz \cdot z_{3}) (-p^{2})^{\frac{\Delta - 2}{2}} \theta(p),$$

$$(2.344)$$

where $\theta(p) \equiv \theta(-p^2)\theta(p^0)$. Finally, combining (2.328), (2.331) and (2.344), we have

$$\langle O_{20'}(p) | \mathbf{L}[O_{20'}](\infty, z_1) \mathbf{L}[O_{20'}](\infty, z_2) | J(p, z_3) \rangle_{\mathbf{175}} = 16i\pi^5 \frac{[z_3, p] \cdot [z_1, z_2](-p^2)\theta(p)}{(-2p \cdot z_1)^2 (-2p \cdot z_2)^2} \mathcal{F}_{JO}(\zeta),$$
(2.345)

where the function $\mathcal{F}_{JO}(\zeta)$ is

$$\mathcal{F}_{JO}(\zeta) = -\int_{2-i\infty}^{2+i\infty} \frac{d\Delta}{2\pi i} C_2^{\prime-}(\Delta, 0) \frac{\Gamma(\Delta-2)}{\Gamma(\frac{\Delta}{2})^2 \Gamma(\frac{\Delta-2}{2}) \Gamma(\frac{4-\Delta}{2})} g_{\Delta}^{2,2}(\zeta), \qquad (2.346)$$

and $g_{\Delta}^{2,2}(\zeta)$ is the j = 1 celestial block (2.340)

$$g_{\Delta}^{2,2}(\zeta) = \zeta^{\frac{\Delta-4}{2}} {}_{2}F_{1}\left(\frac{\Delta-2}{2}, \frac{\Delta}{2}, \Delta-1, \zeta\right).$$
 (2.347)

Now the remaining task is to find the OPE data $C_2^{\prime-}(\Delta, J)$ at J = 0 and calculate the event shape. In the next section, we will show that there's a superconformal Ward identity that relates $C_2^{\prime-}(\Delta, J)$ to the OPE data of the $\langle O_{20'}O_{20'}O_{20'}O_{20'}\rangle$ 4-point function. Using the identity, we can derive a simple relation between $\mathcal{F}_{JO}(\zeta)$ and the energy-energy correlator calculated in [15].

Relation to energy-energy correlator

Deforming the contour of the Δ -integral in (2.346), we get

$$\mathcal{F}_{JO}(\zeta) = -\sum_{\Delta} p_{\Delta,J=0} \frac{\Gamma(\Delta-2)}{\Gamma(\frac{\Delta}{2})^2 \Gamma(\frac{\Delta-2}{2}) \Gamma(\frac{4-\Delta}{2})} g_{\Delta}^{2,2}(\zeta), \qquad (2.348)$$

where $p_{\Delta,J}$ is the three-point coupling of $\langle O_{20'}O_{20'}O_{20'}J\rangle$ corresponding to the structure (2.342), analytically continued to J = 0, and the sum is over Regge trajectories.

We can obtain $p_{\Delta,J}$ from the four-point function $\langle O_{20'}O_{20'}O_{20'}J\rangle$, whose expression is given in (2.295). Note that the function $\Phi(u, v)$ can be written in terms of superconformal blocks as [53]

$$\Phi(u,v) = (2\pi)^4 \frac{v}{u^3} \sum_{\Delta,J} a_{\Delta,J} g_{\Delta+4,J}(u,v), \qquad (2.349)$$

where $g_{\Delta,J}$ is the usual 4d conformal block, and $a_{\Delta,J}$ is the product of the threepoint couplings to a given superconformal primary. Plugging this into (2.295) and specializing to the configuration $x_1 = 0, x_3 = 1, x_4 = \infty$, we find⁵¹

$$\langle O_{20'}(\infty)O_{20'}(0)O_{20'}(u,v)J(1)\rangle$$

$$= (2\pi)^{4} \sum_{\Delta,J} a_{\Delta,J} \left(-\frac{\Delta+1}{8} G_{\Delta+3,J+1}(u,v) + \frac{(J-1)(\Delta+1)}{8(J+1)} G_{\Delta+3,J-1}(u,v) + \frac{(\Delta+3)(\Delta+4)(\Delta+J+4)^{2}}{128(\Delta+2)(\Delta+J+3)(\Delta+J+5)} G_{\Delta+5,J+1}(u,v) - \frac{(J-1)(\Delta+3)(\Delta+4)(\Delta-J+2)^{2}}{128(J+1)(\Delta-J+3)(\Delta-J+1)(\Delta+2)} G_{\Delta+5,J-1}(u,v) \right),$$

$$(2.350)$$

⁵¹To obtain this equation, we studied the small z, \overline{z} limit on both sides and matched each term in the series expansion.

where $G_{\Delta,J}$ is the conformal block of one conserved current and three scalars with dimension 2, which can be calculated using e.g. [54]. The above expression should agree with the usual conformal block decomposition

$$\langle O_{20'}(\infty)O_{20'}(0)O_{20'}(u,v)J(1)\rangle = \sum_{\Delta,J} p_{\Delta,J}G_{\Delta,J}(u,v).$$
 (2.351)

Therefore, there is a superconformal Ward identity that relates the three-point coupling coefficients $p_{\Delta,J}$ and $a_{\Delta,J}$:

$$p_{\Delta,J} = (2\pi)^4 \left(-\frac{\Delta-2}{8} a_{\Delta-3,J-1} + \frac{J(\Delta-2)}{8(J+2)} a_{\Delta-3,J+1} + \frac{(\Delta-2)(\Delta-1)(\Delta+J-2)^2}{128(\Delta-3)(\Delta+J-3)(\Delta+J-1)} a_{\Delta-5,J-1} - \frac{J(\Delta-2)(\Delta-1)(\Delta-J-4)^2}{128(J+2)(\Delta-J-3)(\Delta-J-5)(\Delta-3)} a_{\Delta-5,J+1} \right). \quad (2.352)$$

Setting J = 0 in the above identity and plugging it into (2.348), we have

$$\mathcal{F}_{JO}(\zeta) = -(2\pi)^4 \sum_{\Delta} \left(-\frac{\Delta-2}{8} a_{\Delta-3,-1} + \frac{(\Delta-2)^3}{128(\Delta-3)^2} a_{\Delta-5,-1} \right) \frac{\Gamma(\Delta-2)}{\Gamma(\frac{\Delta}{2})^2 \Gamma(\frac{\Delta-2}{2}) \Gamma(\frac{4-\Delta}{2})} g_{\Delta}^{2,2}(\zeta)$$

$$= -(2\pi)^4 \sum_{\Delta} a_{\Delta-4,-1} \frac{\Gamma(\Delta-2)}{8\Gamma(\frac{\Delta-1}{2})^3 \Gamma(\frac{3-\Delta}{2})} \left(g_{\Delta-1}^{2,2}(\zeta) + \frac{(\Delta-1)}{4(\Delta-2)} g_{\Delta+1}^{2,2}(\zeta) \right).$$

(2.353)

The sum of j = 1 celestial blocks in the parentheses satisfies

$$g_{\Delta-1}^{2,2}(\zeta) + \frac{(\Delta-1)}{4(\Delta-2)}g_{\Delta+1}^{2,2}(\zeta) = \zeta f_{\Delta}^{4,4}(\zeta), \qquad (2.354)$$

where $f_{\Delta}^{4,4}(\zeta)$ is the j = 0 celestial block:

$$f_{\Delta}^{4,4}(\zeta) = \zeta^{\frac{\Delta-7}{2}} {}_{2}F_{1}\left(\frac{\Delta-1}{2}, \frac{\Delta-1}{2}, \Delta-1, \zeta\right).$$
(2.355)

This gives

$$\mathcal{F}_{JO}(\zeta) = -\frac{\zeta}{2} \left(\sum_{\Delta} a_{\Delta-4,-1} \frac{4\pi^4 \Gamma(\Delta-2)}{\Gamma(\frac{\Delta-1}{2})^3 \Gamma(\frac{3-\Delta}{2})} f_{\Delta}^{4,4}(\zeta) \right).$$
(2.356)

Note that the term in the parentheses is simply the function $\mathcal{F}_{\mathcal{E}}(\zeta)$ related to the energy-energy correlator calculated in [15], see (7.11) and (7.16) there. Therefore, we have

$$\mathcal{F}_{JO}(\zeta) = -\frac{\zeta}{2} \mathcal{F}_{\mathcal{E}}(\zeta), \qquad (2.357)$$

which agrees with (2.323) from direct computation.

2.8 Discussion and future directions

We have seen that a product of light-transformed local operators $L[O_1]L[O_2]$ is encoded in a nontrivial way inside the space of light-ray operators. Low transverse spin terms in the product are special linear combinations of light-ray operators with spin $J_1 + J_2 - 1$. Higher transverse spin terms are primary descendants, obtained by acting with the special conformally-invariant differential operators \mathcal{D}_n on higher-*J* light-ray operators.

The differential operators \mathcal{D}_n appear in the general classification of reducible generalized Verma modules described in [35]. Most of the operators in this classification act on multiplets with quantum numbers below the unitarity bound. Thus, when they were first identified, it was not obvious a priori what roles they could play in physical unitary CFTs. However, light-ray operators naturally have quantum numbers that violate the unitarity bound, and indeed we have identified a role for \mathcal{D}_n acting on this space. It is interesting to ask whether there are similar roles in Lorentzian observables for other conformally-invariant differential operators. In addition, it would be interesting to further explore interrelationships between conformally-invariant differential operators, conformally-invariant pairings, conformally-invariant integral transforms, and weight-shifting operators.

One way to motivate the light-ray OPE is by thinking about null-integrated operators as primaries in a fictitious d-2-dimensional CFT. Now that we have a complete description of the terms in this OPE, can we push the analogy with CFT_{d-2} further? For example, what are the implications of the light-ray OPE for multi-point eventshapes, such as the three-point energy correlators studied in [55]? Associativity of the light-ray OPE should give rise to a nontrivial crossing equation satisfied by three-point event shapes, and it would be interesting to study this "celestial" crossing equation using bootstrap techniques. One of the first lessons of the analytic bootstrap is that the crossing equations imply the existence of "double-twist" operators with arbitrarily large spin, via the lightcone bootstrap [8, 7]. Similar arguments for the light-ray OPE could imply the existence of terms with arbitrarily-large transverse spin *j*. It would be interesting to understand the relationship between these operators and the usual large-spin operators in the lightcone bootstrap.

To fully develop a celestial bootstrap program, one would need to understand OPEs of more general light-ray operators, such as a product of a null-integrated local operator L[O] with a general light-ray operator $\mathbb{O}_{\Delta,J}$, or even a product of two general light-ray operators. This is an important problem for the future. A key conceptual question

is: do new types of operators appear beyond the ones constructed in [11]? For the OPE explored in this work, the answer turned out to be "no" for rather nontrivial reasons. For more general light-ray OPEs, the answer is less clear. It is natural to conjecture, however, that any light-ray operators allowed by symmetries will appear. For example, we expect that the leading light-ray operator in the three-fold OPE of average null energy operators is $L[X_4]$, where X_4 is the lowest-twist spin-4 local operator. This claim and its implications should be testable using the results of [55].

Currently, the best available data about multi-point event shapes comes from Hofman and Maldacena's calculation of energy correlators in $\mathcal{N} = 4$ Super Yang Mills theory at large 't Hooft coupling [12]. They computed the first few terms in the large- λ expansion of a multi-point energy correlator up to order $1/\lambda^{3/2}$. It would be interesting to understand the structure of their result from the point of view of the light-ray OPE, and also the "*t*-channel" expansion of [34].

Correlators of average null energy operators do not capture complete information about the energy distribution of a state. In particular, they are blind to the value of retarded time when excitations reach future null infinity. To probe this more refined information, it is natural to weight integrals along null infinity by non-constant functions of retarded time. In [56], it was shown that meromorphic weighting functions with carefully-chosen poles can give rise to useful "dispersive sum rules" that constrain the data of a CFT. Alternative weighting functions may have other useful applications and are worth exploring.

Finally, it would be interesting to explore the structure of the short-angle expansion of energy-energy correlators and other event shapes in non-conformal theories, and in particular how the higher-transverse spin terms arise there [17]. When the conformal symmetry is present, the operators \mathcal{D}_n acting on light-ray operators inserted at spatial infinity (in which case \mathcal{D}_n become expressed in terms of special conformal generators) define some new discrete set of translationally-invariant detectors. As the conformal symmetry is broken, this relation between these detectors and the continuous Regge trajectories gets broken as well. We thus expect that in theories such as QCD some new discrete set of anomalous dimensions should appear in observables for which higher transverse spin is important (such as higher-point event shapes).

Acknowledgements

We thank Lance Dixon, Adam Levine, Juan Maldacena, Pier Monni, Ian Moult, and Emery Sokatchev for discussions. CHC and DSD are supported by Simons Foundation grant 488657 (Simons Collaboration on the Nonperturbative Bootstrap), a Sloan Research Fellowship, and a DOE Early Career Award under grant no. DE-SC0019085. PK is supported by DOE grant no. DE-SC0009988 and the Adler Family Fund at the Institute for Advanced Study. This project has received funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (grant agreement No 787185).

Chapter 3

THREE-POINT ENERGY CORRELATORS AND THE CELESTIAL BLOCK EXPANSION

This chapter is based on

[1] Cyuan-Han Chang and David Simmons-Duffin. "Three-point energy correlators and the celestial block expansion". In: *JHEP* 02 (2023), p. 126. DOI: 10.1007/JHEP02(2023)126. arXiv: 2202.04090 [hep-th].

3.1 Introduction

Energy correlators [18, 19, 57] are natural Lorentzian observables with numerous applications in collider physics, conformal field theory, and string theory, see e.g. [58, 59, 60, 61, 62, 12, 63, 64, 24, 51, 25, 65, 66, 49, 67, 68, 21, 17, 22, 26, 55, 69, 23, 70, 71, 72, 73, 74, 75]. They are given by an expectation value of a product of energy flux operators $\mathcal{E}(\vec{n}_i)$ [76] that measure the flux of energy at locations $\vec{n}_i \in S^{d-2}$ on the celestial sphere:

$$\langle \Psi | \mathcal{E}(\vec{n}_1) \cdots \mathcal{E}(\vec{n}_k) | \Psi \rangle.$$
 (3.1)

Energy correlators are examples of more general "event shapes," which are expectation values of products of detectors at different locations on the celestial sphere.

The kinematics of energy correlators and event shapes exhibit many features of a (fictitious) Euclidean d-2-dimensional CFT on the celestial sphere. In particular, the Lorentz group SO(d – 1, 1) acts as the conformal group on the celestial sphere, so event shapes exhibit conformal symmetry.

However, other aspects of event shapes are different from d-2-dimensional CFT. Event shapes are not necessarily computed (in an obvious way) by a local path integral on the celestial sphere. Consequently, structures like radial quantization and a d-2-dimensional operator product expansion (OPE) cannot obviously be used to analyze them. Nevertheless, it was argued by Hofman and Maldacena [12] that a kind of OPE should exist between energy flux operators $\mathcal{E}(\vec{n}_1) \times \mathcal{E}(\vec{n}_2)$ in the limit $\vec{n}_1 \rightarrow \vec{n}_2$, i.e. as the corresponding points on the celestial sphere approach each other. In [15, 77], the OPE of two energy flux operators $\mathcal{E}(\vec{n}_1) \times \mathcal{E}(\vec{n}_2)$ was explicitly constructed in a general nonperturbative CFT_d , and it was shown that the objects appearing are the light-ray operators $\mathbb{O}_J(\vec{n})$ of [11]. This leads to a useful expansion for two-point energy correlators in special functions called "celestial blocks," which re-sum the contributions of light-ray operators and their descendants on the celestial sphere.

If it were possible to iterate the light-ray OPE, we would obtain a simple and beautiful procedure for evaluating higher-point energy correlators. However, the arguments of [15, 77] do not extend in a simple way to describe the OPE of an energy flux operator and a more general light-ray operator $\mathcal{E}(\vec{n}_1) \times \mathbb{O}_J(\vec{n}_2)$, or to describe an OPE of general light-ray operators $\mathbb{O}_{J_1}(\vec{n}_1) \times \mathbb{O}_{J_2}(\vec{n}_2)$. Perturbative studies of these more complicated OPEs were undertaken recently in [73]. Finding an appropriate nonperturbative generalization of the light-ray OPE is an important problem. However, we will not solve it in this work. Instead, we assume that a general light-ray OPE exists and study some of its consequences for higher-point energy correlators.

One consequence is that higher-point energy correlators should admit an expansion in a discrete sum of multi-point celestial blocks. Mathematically, harmonic analysis with respect to the Lorentz group [44] guarantees that energy correlators can be expanded in an integral of celestial "partial waves." However, going from a partial wave expansion to a celestial block expansion requires a dynamical assumption about poles in partial wave coefficients. We check this assumption by studying the celestial block expansion of three-point energy correlators (EEEC) at both weak coupling (in QCD and N = 4 SYM) and strong coupling (N = 4 SYM). In all cases, we find that a discrete celestial block expansion exists, and that the quantum numbers of objects appearing can be understood from symmetries. At weak coupling, we use the recent perturbative expressions for the EEEC in [55], and at strong coupling, we study Hofman and Maldacena's famous result for the EEEC [12].

A particularly interesting limit of the EEEC is the collinear limit [55, 73], where all three operators approach each other on the celestial sphere $|\vec{n}_{ij}| \rightarrow 0$ with $|\vec{n}_{ij}|/|\vec{n}_{kl}|$ fixed, where $\vec{n}_{ij} = \vec{n}_i - \vec{n}_j$. Physically, the collinear limit is obtained by simultaneously boosting all three detectors. By Lorentz-invariance, this is equivalent to boosting the state $|\Psi\rangle$ in the opposite direction, causing its momentum p to approach the null cone. However, a point on the null cone encodes a point on the celestial sphere $(1, \vec{n}_4) = p/p^0$, so the kinematics of the EEEC in the collinear limit are the same as for a conformal four-point function in CFT_{d-2}. In particular, celestial blocks have an expansion in the collinear limit, where the leading term is the usual four-point conformal block. This observation was made for the leading term in [73], and we will extend it to a systematic expansion around the collinear limit. Furthermore, event shapes inherit crossing symmetry from the *d*-dimensional bulk theory. This allows us to apply techniques from the analytic bootstrap for CFT fourpoint functions to the collinear EEEC, including the lightcone bootstrap [7, 8, 78, 79, 80, 81] and Lorentzian inversion formula [9, 10]. (Interestingly, the Lorentzian inversion formula requires analytically continuing to Lorentzian signature on the celestial sphere, which is (d - 2, 2) signature from the point of view of the bulk theory.)

This paper is organized as follows. In section 3.2, we study implications of Lorentz symmetry for event shapes. We explain the form that the celestial block expansion should take for 2- and 3-point event shapes, and study the expansion of 3-point celestial blocks around the collinear limit. We furthermore explore general constraints of celestial crossing symmetry for the collinear EEEC using lightcone bootstrap methods. In section 3.3, we study recent leading-order weak-coupling results for the collinear EEEC in QCD and N = 4 SYM from the point of view of the celestial block expansion, using the Lorentzian inversion formula to extract celestial block coefficients. In section 3.4, we describe some predictions for higher orders in the weak coupling expansion that follow from a discrete celestial block expansion. In section 3.5, we discuss consequences of Ward identities, in particular using them to determine the leading nontrivial contact terms in the EEEC in weakly-coupled N = 4 SYM. In section 3.6, we study the EEEC in strongly-coupled N = 4 SYM for general configurations on the celestial sphere — not just the collinear limit. We explain how the corresponding celestial OPE data can be obtained from a three-point celestial inversion formula, and then apply the inversion formula to results from [12] to obtain simple analytic formulas for the full EEEC celestial OPE data at $O(1/\lambda)$. Finally, we conclude in section 3.7.

Note Added: This paper will appear simultaneously with a paper by Hao Chen, Ian Moult, Joshua Sandor, and Hua Xing Zhu, that also studies three-point correlators of light-ray operators from the perspective of the light-ray OPE. We thank these authors for coordinating submission.

3.2 Lorentz symmetry and event shapes

Because the Lorentz group SO(d - 1, 1) is also the conformal group on the celestial sphere S^{d-2} , event shapes can be decomposed into "celestial blocks," which are natural objects from the point of view of d-2 dimensional CFT. We will be particularly interested in three-point event shapes. In the "collinear" limit where the three detectors are close to each other, the kinematics of a three-point event shape become the same as a CFT four-point function, and celestial blocks become four-point conformal blocks. We will begin by reviewing event shapes in CFT. We then discuss celestial blocks for two-point event shapes, before introducing three-point celestial blocks and their collinear limit.

3.2.1 Review: event shapes and the light transform

An event shape can be thought of as a weighted cross section, or alternatively as the expectation value of an operator at future null infinity. For example, consider the three-point energy correlator (EEEC), conventionally defined by

$$EEEC(\zeta_{12}, \zeta_{13}, \zeta_{23}) = \sum_{i,j,k} \int d\sigma \frac{E_i E_j E_k}{Q^3} \delta\left(\zeta_{12} - \frac{1 - \cos\theta_{ij}}{2}\right) \delta\left(\zeta_{13} - \frac{1 - \cos\theta_{ik}}{2}\right) \delta\left(\zeta_{23} - \frac{1 - \cos\theta_{jk}}{2}\right)$$
(3.2)

Here, $d\sigma$ is the phase space measure multiplied by the squared amplitude for some state $|O(p)\rangle$ to create outgoing particles, and the sum $\sum_{i,j,k}$ runs over triplets of outgoing particles. The definition (3.2) is convenient for perturbative calculations and deriving Ward identities (see section 3.5). However, it obscures some features like IR safety, and furthermore requires the existence of asymptotic states.

An alternative definition of the EEEC, that works in any nonperturbative QFT, is [24]

$$EEEC(\zeta_{12}, \zeta_{13}, \zeta_{23}) = \int d\Omega_{\vec{n}_1} d\Omega_{\vec{n}_2} d\Omega_{\vec{n}_3} \delta(\zeta_{12} - \frac{1 - \vec{n}_1 \cdot \vec{n}_2}{2}) \delta(\zeta_{13} - \frac{1 - \vec{n}_1 \cdot \vec{n}_3}{2}) \delta(\zeta_{23} - \frac{1 - \vec{n}_2 \cdot \vec{n}_3}{2}) \times \frac{\langle O(p) | \mathcal{E}(\vec{n}_1) \mathcal{E}(\vec{n}_2) \mathcal{E}(\vec{n}_3) | O(p) \rangle}{(-p^2)^{\frac{3}{2}} \langle O(p) | O(p) \rangle},$$
(3.3)

where $\mathcal{E}(\vec{n})$ is an energy detector defined by

$$\mathcal{E}(\vec{n}) = \lim_{r \to \infty} r^{d-2} \int_0^\infty dt \ n^i T^0{}_i(t, r\vec{n}). \tag{3.4}$$

In a CFT, $\mathcal{E}(\vec{n})$ is conformally equivalent to the average null energy operator ANEC — a null integral of the stress tensor [12]. Thus, we often refer to $\mathcal{E}(\vec{n})$ as ANEC
operators. Here, and below, we use the shorthand notation where when a bra and ket have equal momenta, we implicitly strip off an overall momentum-conserving delta function. This is equivalent to Fourier-transforming only one of the operators:

$$\langle O(p)|\cdots|O(p)\rangle \equiv \int d^d x e^{ip\cdot x} \langle 0|O(x)\cdots O(0)|0\rangle.$$
 (3.5)

The ANEC operator $\mathcal{E}(\vec{n})$ measures energy flux at a point on the celestial sphere $\vec{n} \in S^{d-2}$. In a CFT, it can be understood in terms of a conformally-invariant integral transform called the light transform [11]. To describe the light transform, we use index-free notation where we contract indices of an operator with an auxiliary null vector z: $O(x, z) = O^{\mu_1 \cdots \mu_J}(x) z_{\mu_1} \cdots z_{\mu_J}$. The light transform of an operator O with scaling dimension Δ and spin J is

$$\mathbf{L}[O](x,z) = \int_{-\infty}^{\infty} d\alpha (-\alpha)^{-\Delta - J} O\left(x - \frac{z}{\alpha}, z\right).$$
(3.6)

Under conformal transformations, L[O](x, z) transforms like a primary operator at x with quantum numbers $(1 - J, 1 - \Delta)$. The ANEC operator defined in (3.4) can be written as the light transform of the stress-energy tensor placed at spatial infinity:

$$\mathcal{E}(\vec{n}) = 2\mathbf{L}[T](\infty, z = (1, \vec{n})). \tag{3.7}$$

In general, an (un-normalized) *n*-point event shape in CFT is the matrix element of a product of *n* light-transformed operators in a state $|O(p)\rangle$:

$$\langle \mathcal{O}(p) | \mathbf{L}[\mathcal{O}_1](\infty, z_1) \cdots \mathbf{L}[\mathcal{O}_n](\infty, z_n) | \mathcal{O}(p) \rangle.$$
(3.8)

For the EEEC, we have $O_1 = O_2 = O_3 = T$.

3.2.2 Lorentz symmetry and celestial blocks

Two-point event shapes

Consider a two-point scalar event shape¹

$$\langle \phi_4(p) | \mathbf{L}[\phi_1](\infty, z_1) \mathbf{L}[\phi_2](\infty, z_2) | \phi_3(p) \rangle.$$
(3.9)

For simplicity, we study event shapes built from scalars ϕ_i in this section, leaving spinning operators for later.

¹This scalar event shape is only well-defined nonperturbatively if the theory has Regge intercept $J_0 < -1$ [34]. In this section, we are only interested in kinematics, so we assume this is the case.

Let us understand how the product $\mathbf{L}[\phi_1](\infty, z_1)\mathbf{L}[\phi_2](\infty, z_2)$ transforms under the Lorentz group SO(d - 1, 1). The Lorentz group is isomorphic to the Euclidean conformal group in d - 2 dimensions. From this point of view, the polarization vector z can be thought of as an embedding-space coordinate for the celestial sphere S^{d-2} .

As convenient notation, let $\mathcal{P}_{\delta,j}(z, w)$ denote an operator with dimension δ and spin j in a fictitious CFT_{d-2} on the celestial sphere, in the embedding formalism. The embedding space coordinates are null vectors $z, w \in \mathbb{R}^{d-1,1}$ with a gauge redundancy $w \sim w + \alpha z$. $\mathcal{P}_{\delta,j}(z, w)$ is a homogeneous function of z and w with degrees $-\delta$ and j, respectively. See [15] for more details on this notation. We usually refer to the spin j on the celestial sphere as "transverse spin" to disambiguate it from the Lorentz spin J of a local operator in d-dimensions. When j = 0, we write simply $\mathcal{P}_{\delta}(z)$.

The light-transformed operator $\mathbf{L}[\phi_i](\infty, z)$ is homogeneous of degree $1 - \Delta_i$ in z. Thus, it transforms like a scalar on the celestial sphere with dimension $\delta_i = \Delta_i - 1$:

$$\mathbf{L}[\phi_i](\infty, z) \sim \mathcal{P}_{\delta_i}(z). \tag{3.10}$$

From this point of view, we can treat correlators of $\mathbf{L}[\phi_i](\infty, z)$ as if they were correlators of $\mathcal{P}_{\delta_i}(z)$ in a fictitious CFT_{d-2} . Note that we do not assert that there exists a *local* CFT on S^{d-2} . For our purposes, (3.10) is convenient notation for keeping track of symmetries.

Using this notation, a product $\mathbf{L}[\phi_1](\infty, z_1)\mathbf{L}[\phi_2](\infty, z_2)$ transforms like a product of scalars $\mathcal{P}_{\delta_1}(z_1)\mathcal{P}_{\delta_2}(z_2)$ in d-2 dimensions. It is natural to expand such a product in a d-2 dimensional OPE, where the objects that appear are spin-*j* traceless symmetric tensors:

$$\mathcal{P}_{\delta_1}(z_1)\mathcal{P}_{\delta_2}(z_2) = \sum_{\delta,j} r_{\delta,j} \mathcal{C}_{\delta,j}(z_1, z_2, \partial_{z_2}, \partial_{w_2})\mathcal{P}_{\delta,j}(z_2, w_2).$$
(3.11)

Here, the dimensions δ and "OPE coefficients" $r_{\delta,j}$ that appear depend on the theory. However, the differential operator $C_{\delta,j}$ is determined by symmetry and is defined by

$$C_{\delta,j}(z_1, z_2, \partial_{z_2}, \partial_{w_2}) \langle \mathcal{P}_{\delta,j}(z_2, w_2) \mathcal{P}_{\delta,j}(z, w) \rangle = \langle \mathcal{P}_{\delta_1}(z_1) \mathcal{P}_{\delta_2}(z_2) \mathcal{P}_{\delta,j}(z, w) \rangle,$$
(3.12)

where $\langle \mathcal{P}_{\delta,j}(z_2, w_2) \mathcal{P}_{\delta,j}(z, w) \rangle$ and $\langle \mathcal{P}_{\delta_1}(z_1) \mathcal{P}_{\delta_2}(z_2) \mathcal{P}_{\delta,j}(z, w) \rangle$ are standard twoand three-point structures in the embedding space. The light-ray OPE gives a concrete version of the expansion (3.11) where the objects on the right-hand side are light-ray operators. Taking $\mathbf{L}[\phi]\mathbf{L}[\phi]$ as an example, the operators appearing are [15, 77]

$$\mathbf{L}[\phi] \times \mathbf{L}[\phi] \sim \sum_{i} \mathbb{O}_{i,J=-1,j=0} + \sum_{i} \sum_{n=1}^{\infty} \mathcal{D}_{2n} \mathbb{O}_{i,J=-1+2n,j=0}.$$
 (3.13)

Here, $\mathbb{O}_{i,J,j}$ denotes a light-ray operator on the *i*-th Regge trajectory with spin J and transverse spin j [11]. \mathcal{D}_{2n} is a differential operator that decreases the spin J by 2n and increases the transverse spin j by 2n. Light-ray operators are analytic continuations of light transformed operators L[O] in J, so they satisfy

$$\mathbb{O}_{i,J,j} = f_{\phi\phi O_{i,J,j}} \mathbf{L}[O_{i,J,j}], \qquad J \in \mathbb{Z}_{\geq 0}, \ J \text{ even.}$$
(3.14)

The light-ray OPE thus establishes a relation between the scalar two-point event shape, defined as the matrix element of $\mathbf{L}[\phi]\mathbf{L}[\phi]$, and the OPE data of $\phi \times \phi$ analytically continued to $J = -1, 1, 3, \ldots$. For concrete calculations of two-point event shapes using the light-ray OPE, see [15, 77].

Let us apply the light-ray OPE to the event shape (3.9). As discussed in [77], there is a selection rule for the transverse spin j: in a state created by scalar operators $\langle \phi_4(p) | \cdot | \phi_3(p) \rangle$, only light-ray operators with j = 0 can have nonzero matrix elements. Thus, the sum in (3.11) collapses to just the j = 0 terms. We find

$$\langle \phi_4(p) | \mathbf{L}[\phi_1](\infty, z_1) \mathbf{L}[\phi_2](\infty, z_2) | \phi_3(p) \rangle = \sum_{\delta} r_{12\delta} C_{\delta}(z_1, z_2, \partial_{z_2}) \langle \phi_4(p) | \mathbb{W}_{\delta}(z_2) | \phi_3(p) \rangle$$
(3.15)

where \mathbb{W}_{δ} stands for transverse-spin zero light-ray operators $\mathbb{O}_{i,J=-1,j=0}$, and transforms as a scalar primary with dimension δ under SO(d-1, 1).

The form of the matrix element $\langle \phi_4(p) | \mathbb{W}_{\delta}(z_2) | \phi_3(p) \rangle$ is fixed by Lorentz symmetry, homogeneity in z_2 , and dimensional analysis to be

$$\langle \phi_4(p) | \mathbb{W}_{\delta}(z_2) | \phi_3(p) \rangle = s_{34\delta} (-2p \cdot z_2)^{-\delta} (-p^2)^{\frac{\Delta_3 + \Delta_4 + \delta - 2 - d}{2}}.$$
 (3.16)

Thus, the event shape can be written as

$$\langle \phi_4(p) | \mathbf{L}[\phi_1](\infty, z_1) \mathbf{L}[\phi_2](\infty, z_2) | \phi_3(p) \rangle = \sum_{\delta} r_{12\delta} s_{34\delta} (-p^2)^{\frac{\Delta_3 + \Delta_4 + \delta - 2 - d}{2}} C_{\delta}(z_1, z_2, \partial_{z_2}) (-2p \cdot z_2)^{-\delta}$$
(3.17)

The object $C_{\delta}(z_1, z_2, \partial_{z_2})(-2p \cdot z_2)^{-\delta}$ is called a "celestial block" and it is completely fixed by SO(d-1, 1) symmetry. It can be computed by solving a Casimir differential

equation, similar to the method used by Dolan and Osborn to compute conventional conformal blocks [82]. The result is [15]

$$C_{\delta}(z_1, z_2, \partial_{z_2})(-2p \cdot z_2)^{-\delta} = \frac{(-p^2)^{\frac{\delta_1 + \delta_2 - \delta}{2}}}{(-2p \cdot z_1)^{\delta_1}(-2p \cdot z_2)^{\delta_2}} f_{\delta}^{\delta_1, \delta_2}(\zeta),$$
(3.18)

where ζ is a Lorentz-invariant cross ratio

$$\zeta = \frac{(-p^2)(-2z_1 \cdot z_2)}{(-2p \cdot z_1)(-2p \cdot z_2)},$$
(3.19)

and the function $f_{\delta}^{\delta_1,\delta_2}(\zeta)$ is given by

$$f_{\delta}^{\delta_1,\delta_2}(\zeta) = \zeta^{\frac{\delta-\delta_1-\delta_2}{2}} {}_2F_1\left(\frac{\delta+\delta_1-\delta_2}{2},\frac{\delta+\delta_2-\delta_1}{2},\delta+2-\frac{d}{2},\zeta\right).$$
(3.20)

Combining everything, we obtain a celestial block expansion for the two-point event shape

$$\langle \phi_4(p) | \mathbf{L}[\phi_1](\infty, z_1) \mathbf{L}[\phi_2](\infty, z_2) | \phi_3(p) \rangle$$

= $\frac{(-p^2)^{\frac{\delta_1 + \delta_2 + \delta_3 + \delta_4 - d}{2}}}{(-2p \cdot z_1)^{\delta_1} (-2p \cdot z_2)^{\delta_2}} \sum_{\delta} r_{12\delta} s_{34\delta} f_{\delta}^{\delta_1, \delta_2}(\zeta).$ (3.21)

Note that the form of the celestial block expansion (3.21) is completely dictated by symmetries. The light-ray OPE formula then predicts that the δ 's appearing in the expansion (3.21) should be related to dimensions of light-ray operators in the CFT. Furthermore, it makes a prediction for the product of coefficients $r_{12\delta}s_{34\delta}$:

$$r_{12\delta}s_{34\delta} = \frac{2^{d+2-\delta_3-\delta_4}\pi^{\frac{d}{2}+3}e^{i\pi\frac{\delta_4-\delta_3}{2}}\Gamma(\delta-1)}{\Gamma(\frac{\delta+\delta_3-\delta_4}{2})\Gamma(\frac{\delta-\delta_3+\delta_4}{2})\Gamma(\frac{\delta_3+\delta_4-\delta}{2})\Gamma(\frac{\delta+\delta_3+\delta_4+2-d}{2})} \left(p_{\delta+1,J=-1}^+ + p_{\delta+1,J=-1}^-\right),$$
(3.22)

where $p_{\Delta,J}^+(p_{\Delta,J}^-)$ is the product of OPE coefficients of the four-point function $\langle \phi_4 \phi_1 \phi_2 \phi_3 \rangle$, analytically continued from even(odd) spin.

Even if we do not know the coefficients $r_{12\delta}$ and $s_{34\delta}$, we can still make some statements about the event shape. We will be particularly interested in the limit where all the detectors are close to each other. For the two-point case, this simply corresponds to $z_1 \rightarrow z_2$, or $\zeta \rightarrow 0$, and the event shape should behave as

$$\langle \phi_4(p) | \mathbf{L}[\phi_1](\infty, z_1) \mathbf{L}[\phi_2](\infty, z_2) | \phi_3(p) \rangle$$

= $r_{12\delta_*} s_{34\delta_*} \frac{(-p^2)^{\frac{\delta_1 + \delta_2 + \delta_3 + \delta_4 - d}{2}} \zeta^{\frac{\delta_* - \delta_1 - \delta_2}{2}}}{(-2p \cdot z_1)^{\delta_1} (-2p \cdot z_2)^{\delta_2}} + \cdots$
= $r_{12\delta_*} s_{34\delta_*} (-p^2)^{\frac{\delta_* + \delta_3 + \delta_4 - d}{2}} \langle \mathcal{P}_{\delta_1}(z_1) \mathcal{P}_{\delta_2}(z_2) \mathcal{P}_{\delta_*}(p) \rangle + \cdots ,$ (3.23)

where δ_* is the smallest δ appearing in the \sum_{δ} sum.

Three-point event shapes

Next, consider a three-point event shape

$$\langle \phi_5(p) | \mathbf{L}[\phi_1](\infty, z_1) \mathbf{L}[\phi_2](\infty, z_2) \mathbf{L}[\phi_3](\infty, z_3) | \phi_4(p) \rangle.$$
(3.24)

We can use the light-ray OPE to decompose the product of a pair of detectors, say $\mathbf{L}[\phi_1]\mathbf{L}[\phi_2]$, as a sum of light-ray operators \mathbb{O}_i . However, we do not currently possess a more general light-ray OPE formula that lets us further decompose the product $\mathbb{O}_i \mathbf{L}[\phi_3]$. To make progress, let us use Lorentz symmetry to predict the form that a celestial block expansion for the three-point event shape should have.

Following the analysis in section 3.2.2, we treat $\mathbf{L}[\phi_1](\infty, z_1)\mathbf{L}[\phi_2](\infty, z_2)\mathbf{L}[\phi_3](\infty, z_3)$ as a product of three scalar primary operators $\mathcal{P}_{\delta_1}(z_1)\mathcal{P}_{\delta_2}(z_2)\mathcal{P}_{\delta_3}(z_3)$ in a fictitious CFT_{*d*-2}. Formally taking consecutive OPEs, we have²

$$\mathcal{P}_{\delta_{1}}(z_{1})\mathcal{P}_{\delta_{2}}(z_{2})\mathcal{P}_{\delta_{3}}(z_{3}) = \sum_{\delta,j} r_{12\mathcal{P}_{\delta,j}} C_{12\mathcal{P}_{\delta,j}}(z_{1}, z_{2}, \partial_{z_{2}}, \partial_{w_{2}})\mathcal{P}_{\delta,j}(z_{2}, w_{2})\mathcal{P}_{\delta_{3}}(z_{3}) = \sum_{\delta',\lambda'} \sum_{\delta,j} r_{12\mathcal{P}_{\delta,j}} r'_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta',\lambda'}} C_{12\mathcal{P}_{\delta,j}}(z_{1}, z_{2}, \partial_{z_{2}}, \partial_{w_{2}}) C_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta',\lambda'}}(z_{2}, z_{3}, \partial_{\bar{x}_{3}}, \partial_{\bar{w}_{3}})\mathcal{P}_{\delta',\lambda'}(z_{3}, \bar{w}_{3}).$$
(3.25)

Therefore, the three-point event shape should have the form

$$\begin{aligned} \langle \phi_{5}(p) | \mathbf{L}[\phi_{1}](\infty, z_{1}) \mathbf{L}[\phi_{2}](\infty, z_{2}) \mathbf{L}[\phi_{3}](\infty, z_{3}) | \phi_{4}(p) \rangle \\ &= \sum_{\delta'} \sum_{\delta,j} r_{12\mathcal{P}_{\delta,j}} r'_{\mathcal{P}_{\delta,j} 3\mathcal{P}_{\delta'}} C_{12\mathcal{P}_{\delta,j}}(z_{1}, z_{2}, \partial_{z_{2}}, \partial_{w_{2}}) C_{\mathcal{P}_{\delta,j} 3\mathcal{P}_{\delta'}}(z_{2}, z_{3}, \partial_{z_{3}}) \langle \phi_{5}(p) | \mathbb{W}'_{\delta'}(z_{3}) | \phi_{4}(p) \rangle \\ &= \sum_{\delta'} \sum_{\delta,j} r_{12\mathcal{P}_{\delta,j}} r'_{\mathcal{P}_{\delta,j} 3\mathcal{P}_{\delta'}} s'_{45\delta'} (-p^{2})^{\frac{\delta_{4}+\delta_{5}+\delta'-1-d}{2}} \\ &\times C_{12\mathcal{P}_{\delta,j}}(z_{1}, z_{2}, \partial_{z_{2}}, \partial_{w_{2}}) C_{\mathcal{P}_{\delta,j} 3\mathcal{P}_{\delta'}}(z_{2}, z_{3}, \partial_{z_{3}}) (-2z_{3} \cdot p)^{-\delta'}, \end{aligned}$$
(3.26)

where from the second to the third line, we again use the homogeneity of z_3 and p. In the expansion (3.26), we have three unknown coefficients $r_{12\mathcal{P}_{\delta,j}}$, $r'_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta'}}$, and $s'_{45\delta'}$.

If an OPE expansion for a three-point event shape exists, symmetries ensure it must take the form (3.26). However, we do not know an argument guaranteeing the

²Here, we are following the notation of [15, 77], where \vec{w}_3 denotes a collection of polarization vectors for different rows of the Young diagram of an SO(d - 2) representation. However, after taking the expectation value in a scalar density matrix $\langle \phi_5(p) | \cdot | \phi_4(p) \rangle$, only scalar representations λ' are allowed, so \vec{w}_3 immediately drops out and can be ignored.

existence of such an expansion. Mathematically, the only thing that is guaranteed is that (3.24) can be decomposed into a double-integral over complex δ and δ' of "celestial partial waves," defined below in equation (3.188). An expansion like (3.26) would arise if we can additionally close the contours to the right, picking up a set of discrete poles, as described in [47] for the conventional conformal block decomposition. In this work, we assume that such a contour maneuver is possible, at least to characterize the leading behavior of the event shape in the collinear limit. In the absence of nonperturbative arguments, it is also important to compare (3.26) to perturbative data, as we do in section 3.3.

The kinematic dependence of (3.26) is accounted for by the object

$$C_{12\mathcal{P}_{\delta,i}}(z_1, z_2, \partial_{z_2}, \partial_{w_2})C_{\mathcal{P}_{\delta,i}3\mathcal{P}_{\delta'}}(z_2, z_3, \partial_{z_3})(-2z_3 \cdot p)^{-\delta'}, \qquad (3.27)$$

which is completely fixed by Lorentz symmetry. We call (3.27) a three-point celestial block. Although we do not know a compact closed-form expression for it, we can still determine its expansion around the collinear limit, where z_1, z_2, z_3 are close to each other. The reason is that the collinear limit is equivalent (up to a Lorentz transformation) to a configuration where *p* is null. Writing *p* as z_0 in this null limit, the leading term of the three-point celestial block in the collinear limit becomes

$$C_{12\mathcal{P}_{\delta,j}}(z_1, z_2, \partial_{z_2}, \partial_{w_2})C_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta'}}(z_2, z_3, \partial_{z_3})(-2z_3 \cdot p)^{-\delta'} \rightarrow C_{12\mathcal{P}_{\delta,j}}(z_1, z_2, \partial_{z_2}, \partial_{w_2})C_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta'}}(z_2, z_3, \partial_{z_3})(-2z_3 \cdot z_0)^{-\delta'} = g_{\delta,j}^{(\delta_1, \delta_2, \delta_3, \delta')}(z_1, z_2, z_3, z_0) (3.28)$$

where the second line is simply the definition of a four-point conformal block.

To characterize subleading terms in the expansion around the collinear limit (or equivalently the null *p* limit), it is helpful to introduce $|\mathcal{P}_{\delta'}, p\rangle\rangle$, defined as the state in the conformal multiplet of $|\mathcal{P}_{\delta'}\rangle$ that is invariant under the little group SO(*d* – 1) that fixes *p*. From the point of view of conformal symmetry, *p* is a point in the center of EAdS_{*d*-1}, so the overlap of $|\mathcal{P}_{\delta'}, p\rangle\rangle$ with $|\mathcal{P}_{\delta'}(z_3)\rangle$ is a bulk-to-boundary propagator:

$$\langle \mathcal{P}_{\delta'}(z_3) | \mathcal{P}_{\delta'}, p \rangle \rangle = (-2z_3 \cdot p)^{-\delta'}. \tag{3.29}$$

The three-point celestial block can be written as

$$C_{12\mathcal{P}_{\delta,i}}(z_1, z_2, \partial_{z_2}, \partial_{w_2})C_{\mathcal{P}_{\delta,i},3\mathcal{P}_{\delta'}}(z_2, z_3, \partial_{z_3})\langle \mathcal{P}_{\delta'}(z_3)|\mathcal{P}_{\delta'}, p\rangle\rangle.$$
(3.30)



Figure 3.1: Diagrams representing the two-point and three-point celestial blocks. Each vertex should be understood as an OPE differential operator C_{ijk} in the CFT_{*d*-2}. The symbol \otimes represents the factor (3.29).

This expression can be represented by the diagram in figure 3.1, where we also include the diagram for the two-point case. Theses diagrams can be thought of as the OPE decomposition of the usual three-point and four-point function in the fictitious CFT_{d-2} , but with one of the external operators replaced by $|\mathcal{P}_{\delta'}, p\rangle\rangle$.

Let us determine a more explicit expression for $|\mathcal{P}_{\delta'}, p\rangle\rangle$. Treating SO(d - 1, 1) as the (d-2)-dimensional conformal group, its generators are $\{D, M_{ab}, P_a, K_a\}$, where $a, b = 1, \dots, d-2, M_{ab}$ are SO(d - 2) rotation generators, and P_a and K_a are the translation and special conformal generators in \mathbb{R}^{d-2} . For $p = (p^0, \vec{p}) = (1, \vec{0})$, the little group that fixes p is generated by M_{ab} and $P_a + K_a$. Therefore, the $|\mathcal{P}_{\delta'}, p\rangle\rangle$ must satisfy the conditions

$$\begin{split} M_{ab} | \mathcal{P}_{\delta'}, (1, \vec{0}) \rangle \rangle &= 0, \\ (P_a + K_a) | \mathcal{P}_{\delta'}, (1, \vec{0}) \rangle \rangle &= 0. \end{split} \tag{3.31}$$

The solution to these conditions was obtained in [83] with a somewhat different motivation (studying local probes of a dual AdS geometry). The result is

$$|\mathcal{P}_{\delta'}, (1,\vec{0})\rangle\rangle = \Gamma(\delta' + 2 - \frac{d}{2}) \left(\frac{\sqrt{P^2}}{2}\right)^{\frac{d-2}{2}-\delta'} J_{\delta'-\frac{d-2}{2}}\left(\sqrt{P^2}\right) |\mathcal{P}_{\delta'}\rangle, \qquad (3.32)$$

where $|\mathcal{P}_{\delta'}\rangle$ is the CFT primary state killed by K_a , and $J_{\nu}(x)$ is a Bessel function of the first kind.

The action of momentum generators P_a on $|\mathcal{P}_{\delta'}\rangle$ can be written as derivatives of $|\mathcal{P}_{\delta'}(z)\rangle$ with respect to z. Equation (3.32) thus expresses $|\mathcal{P}_{\delta'}(1,\vec{0})\rangle\rangle$ as an infinite-order differential operator acting on $|\mathcal{P}_{\delta'}(z)\rangle$. Plugging this into (3.30), we obtain the three-point celestial block as an infinite-order differential operator acting on a

four-point conformal block:

$$C_{12\mathcal{P}_{\delta,j}}(z_1, z_2, \partial_{z_2}, \partial_{w_2})C_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta'}}(z_2, z_3, \partial_{z_3})(-2z_3 \cdot p)^{-\delta'} = \Gamma(\delta' + 2 - \frac{d}{2}) \left(\frac{\sqrt{\partial_{\tilde{y}}^2}}{2}\right)^{\frac{d-2}{2} - \delta'} J_{\delta' - \frac{d-2}{2}}\left(\sqrt{\partial_{\tilde{y}}^2}\right) g_{\delta,j}^{(\delta_1, \delta_2, \delta_3, \delta')}(z_1, z_2, z_3, z_0), \quad (3.33)$$

where $z_0 = (1, \vec{y}^2, \vec{y})$. Note that only even powers of $\sqrt{\partial_{\vec{y}}^2}$ appear in the expansion of the Bessel function, so the above differential operator is well-defined order-by-order in this expansion. Even though intermediate terms depend on the point z_0 , the final result must be independent of z_0 . In appendix B.1, we present an alternative derivation of this identity that leads to an expression where Lorentz invariance is more manifest.

Note that the state $|\mathcal{P}_{\delta'}, p\rangle\rangle$ breaks the Lorentz group SO(d-1, 1) to the little group SO(d-1). This is the same pattern of symmetry breaking that occurs in the presence of a codimension-1 spherical boundary or defect [84]. Thus, celestial blocks are equivalent to boundary/defect conformal blocks [85]. Often, boundaries and defects are studied in Euclidean space, where the symmetry breaking pattern in CFT $_{d-2}$ is SO $(d) \rightarrow$ SO(d-1). Since the signature of the corresponding orthogonal groups is different, our celestial blocks are related by analytic continuation to those blocks.

Expanding our expression to leading and subleading in the collinear limit $p^2 \rightarrow 0$, we find

$$C_{12\mathcal{P}_{\delta,j}}(z_1, z_2, \partial_{z_2}, \partial_{w_2})C_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta'}}(z_2, z_3, \partial_{z_3})(-2z_3 \cdot p)^{-\delta'} = T_{123\delta'}(z_1, z_2, z_3, p) \left(g_{\delta,j}^{(123\delta')}(u, v) + \frac{\zeta_{13}}{d - 4 - 2\delta'} \mathcal{D}_{u,v}^{(1)} g_{\delta,j}^{(123\delta')}(u, v) + O((-p^2)^2) \right),$$
(3.34)

where *u* and *v* are defined as

$$u = \frac{(-2z_1 \cdot z_2)(-2z_3 \cdot p)}{(-2z_1 \cdot z_3)(-2z_2 \cdot p)}, \quad v = \frac{(-2z_2 \cdot z_3)(-2z_1 \cdot p)}{(-2z_1 \cdot z_3)(-2z_2 \cdot p)}, \tag{3.35}$$

and the overall factor $T_{123\delta'}(z_1, z_2, z_3, p)$ is

$$T_{123\delta'}(z_1, z_2, z_3, p) = \frac{\left(\frac{-2z_2 \cdot p}{-2z_1 \cdot p}\right)^{\frac{\delta_1 - \delta_2}{2}} \left(\frac{-2z_1 \cdot p}{-2z_1 \cdot z_3}\right)^{\frac{\delta_3 - \delta'}{2}}}{(-2z_1 \cdot z_2)^{\frac{\delta_1 + \delta_2}{2}} (-2z_3 \cdot p)^{\frac{\delta_3 + \delta'}{2}}}.$$
(3.36)

The differential operator $\mathcal{D}_{u,v}^{(1)}$ generating the first subleading term is given by

$$\mathcal{D}_{u,v}^{(1)} = \frac{1}{2} (\delta_1 - \delta_2) (\delta_1 - \delta_2 - \delta_3 + \delta') u + \frac{1}{2} (\delta_3 + \delta') ((\delta_1 - \delta_2) v - \delta_1 + \delta_2 + \delta_3 - \delta') + ((2 - 2\delta_1 + 2\delta_2 + \delta_3 - \delta') u - (\delta_3 + \delta') (v - 1)) v \partial_v + ((2 - \delta_1 + \delta_2 - \delta_3 - \delta') v - (\delta_1 - \delta_2 - \delta_3 + \delta') (u - 1)) u \partial_u + 2uv (u \partial_u^2 + v \partial_v^2 + (u + v - 1) \partial_u \partial_v).$$
(3.37)

As an example, let us specialize to a three-point energy correlator in a 4d CFT. In this case, we have $\delta_1 = \delta_2 = \delta_3 = 3$, and (3.37) becomes

$$\mathcal{D}_{u,v}^{(1)\mathcal{E}\mathcal{E}\mathcal{E}} = \frac{(3-\delta')(3+\delta')}{2} + ((5-\delta')u - (3+\delta')(v-1))v\partial_v - ((-3+\delta')(u-1) + (1+\delta')v)u\partial_u + 2uv(u\partial_u^2 + v\partial_v^2 + (u+v-1)\partial_u\partial_v).$$
(3.38)

Using (3.34), we can finally write down the expansion of the three-point event shape in the collinear limit:

$$\begin{aligned} \langle \phi_{5}(p) | \mathbf{L}[\phi_{1}](\infty, z_{1}) \mathbf{L}[\phi_{2}](\infty, z_{2}) \mathbf{L}[\phi_{3}](\infty, z_{3}) | \phi_{4}(p) \rangle \\ &= \sum_{\delta'} \sum_{\delta,j} r_{12\mathcal{P}_{\delta,j}} r'_{\mathcal{P}_{\delta,j} 3\mathcal{P}_{\delta'}} s'_{45\delta'} (-p^{2})^{\frac{\delta_{4}+\delta_{5}+\delta'-1-d}{2}} T_{123\delta'}(z_{1}, z_{2}, z_{3}, p) \\ &\times \left(g_{\delta,j}^{(123\delta')}(u, v) + \frac{\zeta_{13}}{d-4-2\delta'} \mathcal{D}_{u,v}^{(1)} g_{\delta,j}^{(123\delta')}(u, v) + O((-p^{2})^{2}) \right). \end{aligned}$$
(3.39)

In sections 3.3 and 3.4 of this paper, we will mostly focus on the leading term, which is simply a four-point conformal block. In section 3.6, when we study $\mathcal{N} = 4$ at strong coupling, we will derive results related to the full structure of the three-point celestial block.

We can also take the OPE of (3.24) in a different order. If we first take the 23 OPE, we obtain

$$\begin{aligned} \langle \phi_{5}(p) | \mathbf{L}[\phi_{1}](\infty, z_{1}) \mathbf{L}[\phi_{2}](\infty, z_{2}) \mathbf{L}[\phi_{3}](\infty, z_{3}) | \phi_{4}(p) \rangle \\ &= \sum_{\delta'} \sum_{\delta, j} r_{23} \varphi_{\delta, j} r'_{\varphi_{\delta, j} 1 \varphi_{\delta'}} s'_{45\delta'} (-p^{2})^{\frac{\delta_{4} + \delta_{5} + \delta' - 1 - d}{2}} T_{321\delta'}(z_{3}, z_{2}, z_{1}, p) \\ &\times \left(g^{(321\delta')}_{\delta, j}(v, u) + \frac{\zeta_{13}}{d - 4 - 2\delta'} \mathcal{D}^{(1)}_{v, u} g^{(321\delta')}_{\delta, j}(v, u) + O((-p^{2})^{2}) \right). \end{aligned}$$
(3.40)

Thus, we obtain a crossing equation

$$\begin{split} \sum_{\delta'} \sum_{\delta,j} r_{12\mathcal{P}_{\delta,j}} r'_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta'}} s'_{45\delta'}(-p^2) & \frac{\delta_{4}+\delta_{5}+\delta'-1-d}{2} T_{123\delta'}(z_{1},z_{2},z_{3},p) \\ & \times \left(g^{(123\delta')}_{\delta,j}(u,v) + \frac{\zeta_{13}}{d-4-2\delta'} \mathcal{D}^{(1)}_{u,v} g^{(123\delta')}_{\delta,j}(u,v) + O((-p^2)^2) \right) \\ &= \sum_{\delta'} \sum_{\delta,j} r_{23\mathcal{P}_{\delta,j}} r'_{\mathcal{P}_{\delta,j}1\mathcal{P}_{\delta'}} s'_{45\delta'}(-p^2) & \frac{\delta_{4}+\delta_{5}+\delta'-1-d}{2} T_{321\delta'}(z_{3},z_{2},z_{1},p) \\ & \times \left(g^{(321\delta')}_{\delta,j}(v,u) + \frac{\zeta_{13}}{d-4-2\delta'} \mathcal{D}^{(1)}_{v,u} g^{(321\delta')}_{\delta,j}(v,u) + O((-p^2)^2) \right) . \end{split}$$

$$(3.41)$$

The leading term of this equation looks like a usual four-point crossing equation in a (d-2)-dimensional CFT. We will study some of its implications in section 3.2.4 and appendix B.4. It would also be interesting to study (3.41) with subleading terms included.

3.2.3 Expansion of the EEEC in the collinear limit

We are now ready to study the expansion of the three-point energy correlator in the collinear limit. In what follows, we only keep the leading term (the conformal block). We also specialize to four dimensions, for simplicity. We can essentially follow section 3.2.2, replacing the scalar operators ϕ_1, ϕ_2, ϕ_3 with the stress-tensor *T*. Note that the ANEC operator \mathcal{E} is still a scalar on the celestial sphere, so the only difference from our earlier analysis is the homogeneity in the momentum *p*. For sink/source states created by a scalar operator *O*, the result is³

$$\langle O(p) | \mathcal{E}(\vec{n}_{1}) \mathcal{E}(\vec{n}_{2}) \mathcal{E}(\vec{n}_{3}) | O(p) \rangle$$

= $\sum_{\delta,j} r_{\mathcal{E}\mathcal{E}\mathcal{P}_{\delta,j}} r'_{\mathcal{P}_{\delta,j}\mathcal{E}\mathcal{P}_{\delta'_{*}}} s'_{OOP_{\delta'_{*}}} \frac{(-p^{2})^{\frac{2\Delta_{O}+\delta'_{*}-1}{2}} \left(\frac{-2z_{2} \cdot p}{-2z_{1} \cdot z_{2}}\right)^{\frac{3-\delta'_{*}}{2}}}{(-2z_{2} \cdot z_{3})^{3} (-2z_{1} \cdot p)^{\frac{3+\delta'_{*}}{2}}} g^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_{*}})}_{\delta,j}(z,\overline{z}) + \dots,$
(3.42)

where "..." denotes subleading terms in the collinear limit, and δ'_* is the smallest value of δ' that appears in the $\mathcal{P}_{\delta,j} \times \mathcal{E}$ OPE in the fictitious CFT₂. We assume for now that δ'_* is isolated.⁴ We have also changed variables from u, v to z, \overline{z} , defined

³For later convenience, we relabel the points as $1 \rightarrow 2, 2 \rightarrow 3, 3 \rightarrow 1$.

⁴We study a case where two operators have degenerate δ'_* at the lowest order in perturbation theory in section 3.4.2.

by

$$u = z\overline{z} = \frac{\zeta_{23}}{\zeta_{12}}, \quad v = (1 - z)(1 - \overline{z}) = \frac{\zeta_{13}}{\zeta_{12}},$$
 (3.43)

where z, \overline{z} are complex conjugates of each other.⁵

Recall that the EEEC is defined by (3.3). Note that $\langle O(p)|\mathcal{E}(\vec{n}_1)\mathcal{E}(\vec{n}_2)\mathcal{E}(\vec{n}_3)|O(p)\rangle$ depends only on angles between $\vec{n}_1, \vec{n}_2, \vec{n}_3$, which are localized by the delta functions in the first line of (3.3). The remaining Jacobian factor is

$$\int d\Omega_{\vec{n}_1} d\Omega_{\vec{n}_2} d\Omega_{\vec{n}_3} \delta(\zeta_{12} - \frac{1 - \vec{n}_1 \cdot \vec{n}_2}{2}) \delta(\zeta_{13} - \frac{1 - \vec{n}_1 \cdot \vec{n}_3}{2}) \delta(\zeta_{23} - \frac{1 - \vec{n}_2 \cdot \vec{n}_3}{2})$$

$$\rightarrow \frac{64\pi^2}{|(\vec{n}_1 \times \vec{n}_2) \cdot \vec{n}_3|}$$

$$= \frac{32\pi^2}{\sqrt{-\zeta_{12}^2 - \zeta_{13}^2 - \zeta_{23}^2 + 2\zeta_{12}\zeta_{13} + 2\zeta_{12}\zeta_{23} + 2\zeta_{13}\zeta_{23} - 4\zeta_{12}\zeta_{13}\zeta_{23}}}.$$
(3.44)

Also, the total cross section is given by

$$\sigma_{\text{total}}^{O} \equiv \int d^4x \; e^{ip \cdot x} \langle 0|O^{\dagger}(x)O(0)|0\rangle = \frac{2^{5-2\Delta_O}\pi^3}{\Gamma(\Delta_O - 1)\Gamma(\Delta_O)} (-p^2)^{\frac{2\Delta_O - 4}{2}} \theta(p).$$
(3.45)

Combining these results with (3.42), the EEEC in the collinear limit is given by

$$EEEC(\zeta_{12},\zeta_{13},\zeta_{23}) = \frac{\zeta_{12}}{\zeta_{23}^3 \sqrt{-\zeta_{12}^2 - \zeta_{13}^2 - \zeta_{23}^2 + 2\zeta_{12}\zeta_{13} + 2\zeta_{12}\zeta_{23} + 2\zeta_{13}\zeta_{23}}}\mathcal{G}(\zeta_{12},z,\overline{z}) + \dots,$$
(3.46)

where

$$\mathcal{G}(\zeta_{12}, z, \overline{z}) \equiv \zeta_{12}^{\frac{\delta'_* - 5}{2}} \sum_{\delta, j} R_{\delta, j; \delta'_*} g_{\delta, j}^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_*})}(z, \overline{z}),$$

$$R_{\delta, j; \delta'_*} \equiv \frac{2^{2\Delta_O - 9} \Gamma(\Delta_O - 1) \Gamma(\Delta_O)}{\pi} r_{\mathcal{E}\mathcal{E}\mathcal{P}_{\delta, j}} r'_{\mathcal{P}_{\delta, j}\mathcal{E}\mathcal{P}_{\delta'_*}} s'_{OO\mathcal{P}_{\delta'_*}}.$$
(3.47)

Here, we have set $z_i = (1, \vec{n}_i)$ and $\zeta_{ij} = \frac{1 - \vec{n}_i \cdot \vec{n}_j}{2}$.

Thus, the function $\mathcal{G}(\zeta_{12}, z, \overline{z})$ describing the leading behavior of the EEEC in the collinear limit can be written as a sum of conformal blocks, up to a power of ζ_{12} .

⁵Note that earlier we used z as a null polarization vector $z \in \mathbb{R}^{d-1,1}$, whereas here it is a complex number $z \in \mathbb{C}$. We hope that no confusion will arise from this overloaded notation.

The coefficients $R_{\delta,j;\delta'_*}$ appearing in the expansion are products of light-ray OPE coefficients $r_{\mathcal{E}\mathcal{E}\mathcal{P}_{\delta,j}}$ and $r'_{\mathcal{P}_{\delta,j}\mathcal{E}\mathcal{P}_{\delta'_*}}$ and 1-point functions $s'_{OO\mathcal{P}_{\delta'_*}}$. Since most of these quantities are unknown a-priori, we will mostly just work with the coefficients $R_{\delta,j;\delta'_*}$ in this paper. The detailed definition of $R_{\delta,j;\delta'_*}$ would become useful if one could derive a three-point light-ray OPE formula that relates $r_{\mathcal{E}\mathcal{E}\mathcal{P}_{\delta,j}\mathcal{E}\mathcal{P}_{\delta'_*}}s'_{OO\mathcal{P}_{\delta'_*}}$ to the OPE data of the CFT (similar to (3.22)).

Although most of the coefficients in the expansion (3.47) are a-priori unknown, we *do know* a lot about the quantum numbers δ'_* and δ that appear. The dimension δ'_* is associated with the lowest dimension light-ray operator in the triple- \mathcal{E} OPE. It is natural to guess that it is given by the lowest-twist spin-4 operator in the theory: $\delta'_* = \Delta_{\min}(J = 4)$, as we discuss in section 3.3.1. Furthermore, the dimensions δ that appear in the conformal block expansion (3.47) are controlled by the two- \mathcal{E} light-ray OPE, which we understand much better — they are associated to light-ray operators with spin $J = 3, 5, \ldots$ [15, 77]. Below, we will confirm these expectations in examples.

3.2.4 Lightcone bootstrap constraints

The leading term of the crossing equation (3.41) can be written as⁶

$$\mathcal{G}(\zeta_{12}, z, \overline{z}) = \zeta_{12}^{\frac{\delta'_* - 5}{2}} \sum_{\delta, j} R_{\delta, j; \delta'_*} g_{\delta, j}^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_*})}(z, \overline{z})$$
$$= \left(\frac{z\overline{z}}{(1-z)(1-\overline{z})}\right)^3 \zeta_{12}^{\frac{\delta'_* - 5}{2}} \sum_{\delta, j} R_{\delta, j; \delta'_*} g_{\delta, j}^{0, \frac{3-\delta'_*}{2}}(1-z, 1-\overline{z}). \quad (3.48)$$

This looks like a four-point crossing equation in a (d-2)-dimensional CFT. It is thus interesting to ask what we can deduce about the original *d*-dimensional CFT from it. Unfortunately, the coefficients $R_{\delta,j;\delta'_*}$ do not satisfy any simple positivity conditions, so numerical bootstrap methods do not apply in an obvious way. In this section, we will instead study (3.48) from the point of view of the lightcone bootstrap [7, 8, 81], which does not require positivity conditions.

The usual lightcone bootstrap analysis begins by analytically continuing the fourpoint function into Lorentzian signature, and then considering the double lightcone limit $z \ll 1 - \overline{z} \ll 1$. In our setting, this would require analytically continuing

⁶Here, we use two different notations for conformal blocks, and we hope the meaning hereafter will be clear from context. The first notation is $g_{\delta,j}^{(O_1O_2O_3O_4)}$, where the block is labeled by the conformal multiplets of each individual external operator O_1, \dots, O_4 . The second notation is $g_{\delta,j}^{\delta_{12},\delta_{34}}$, where we use the fact that the block depends only on the differences of scaling dimensions $\delta_{12} = \delta_1 - \delta_2$ and $\delta_{34} = \delta_3 - \delta_4$.

celestial cross-ratios away from the Euclidean regime. However, our "correlator" $\mathcal{G}(\zeta_{12}, z, \overline{z})$ does not come from a local, reflection-positive Euclidean CFT_{*d*-2}, and thus it is not guaranteed that we can analytically continue it to (d-2)-dimensional Lorentzian signature (which would be (2, d-2) signature from the point of view of the full *d*-dimensional theory).

However, we believe that the main conclusion of the analytic bootstrap, i.e. the existence of double-twist families at large spin, can still be obtained by staying in Euclidean signature. The idea is that by plugging the leading *t*-channel singularity into the Euclidean inversion formula, one can still deduce that the OPE coefficient density $C(\Delta, J)$ should behave as the lightcone bootstrap predicts at large spin *J*. Similarly, it should be possible to compute subleading corrections in *J* from subleading terms in the *t*-channel singularity. Hence, we expect that analytic continuation in z, \overline{z} can be thought of as a proxy for a more complicated analysis using Euclidean partial waves.

Thus, let us proceed to studying implications of the lightcone bootstrap for the celestial crossing equation (3.48). We will actually consider a more general three-point event shape L[O]L[O]L[O], where O is an operator with spin J. The crossing equation reads

$$\sum_{\delta,j} R_{\delta,j,\delta'_*} g_{\delta,j}^{(0,s)}(z,\overline{z}) = \left(\frac{z\overline{z}}{(1-z)(1-\overline{z})}\right)^{\delta_O} \sum_{\delta,j} R_{\delta,j,\delta'_*} g_{\delta,j}^{(0,s)}(1-z,1-\overline{z}), \quad (3.49)$$

where $s = \frac{\delta_O - \delta'_*}{2}$. As we discuss later in section 3.3.1, we expect that $\delta'_* = \Delta'_* - 1$ is the celestial scaling dimension of the lowest-twist operator with spin J' = 3J - 2 in the $O \times O \times O$ OPE. Suppose the lowest celestial sphere twist $\tau_c = \delta - j$ appearing in the sum on the left-hand side is $\tau_c^* = 2h^*$. Then in the double lightcone limit $z \ll 1 - \overline{z} \ll 1$, we have

$$R_{h^*}^* z^{h^* - 2h_O} k_{2\overline{h}^*}^{0,s}(\overline{z}) + \ldots = \sum_{h,\overline{h}} R_{h,\overline{h}} (1 - \overline{z})^{h - 2h_O} k_{2\overline{h}}^{0,s}(1 - z) + \ldots,$$
(3.50)

where we have introduced

$$h = \frac{\delta - j}{2} = \frac{\tau_c}{2}, \qquad \overline{h} = \frac{\delta + j}{2}.$$
(3.51)

Near the $\overline{z} \to 1$ limit, the SL(2, \mathbb{R}) block $k_{2\overline{h}}^{0,s}(\overline{z})$ has the expansion [81]

$$k_{2\overline{h}}^{0,s}(\overline{z}) = K_0^{0,s}(\overline{h}) + \ldots + K_0^{s,0}(\overline{h})(1-\overline{z})^{-s} + \ldots,$$

$$K_0^{r,s}(\overline{h}) \equiv \frac{\Gamma(r-s)\Gamma(2\overline{h})}{\Gamma(\overline{h}+r)\Gamma(\overline{h}-s)},$$
(3.52)

where "..." represents terms like $(1-\overline{z})^k$ or $(1-\overline{z})^{k-s}$ where k is a positive integer. Therefore, (3.50) becomes

$$R_{h^*}^* z^{h^* - 2h_O} \left(K_0^{0,s}(\overline{h}) + \dots + K_0^{s,0}(\overline{h}) (1 - \overline{z})^{-s} + \dots \right) + \dots$$

= $\sum_{h,\overline{h}} R_{h,\overline{h}} (1 - \overline{z})^{h - 2h_O} k_{2\overline{h}}^{0,s} (1 - z) + \dots$ (3.53)

Note that on the left-hand side, $z^{h^*-2h_O}$ is Casimir-singular in z (i.e. it can be made arbitrarily singular by repeatedly applying the quadratic Casimir), while on the right-hand side $k_{2\bar{h}}^{0,s}(1-z)$ is Casimir-regular in z. In order to reproduce the Casimir-singular term with the correct \bar{z} behavior on the left hand side, we must have an infinite family of operators with $h \to 2h_O$ and $h \to 2h_O - s = h_O + h'_*$ as $\bar{h} \to \infty$. They can be thought of as the "celestial double-twist operators" $[\mathcal{P}_O \mathcal{P}_O]$ and $[\mathcal{P}_O \mathcal{P}_{O'_*}]$.

Since $[\mathcal{P}_O \mathcal{P}_O]$ has $h = 2h_O$ and large-*j* (which means that they must be higher transverse spin terms), the light-ray OPE [15, 77] predicts that they should come from operators with conventional twist $\tau = 2\delta_O + 2 = 2\Delta_O$. Thus, the existence of $[\mathcal{P}_O \mathcal{P}_O]$ predicts that for the maximally allowed transverse spin $(j_{max} = 2J \text{ in this case})$ in the $O \times O$ OPE, there must be a trajectory with $\tau = 2\tau_O + 2J$ at large spin. Similarly, the existence of $[\mathcal{P}_O \mathcal{P}_{O'_*}]$ predicts that there should be a trajectory with the maximally allowed transverse spin that has twist $\tau = \Delta_O + \Delta'_*$ at large spin.

The coefficients $R_{h,\overline{h}}$ at large \overline{h} for the two types of celestial double-twist operators can also be determined. Using the formula [81]

$$\sum_{\substack{\overline{h}=j+\overline{h}_{0}\\j=0,2,\dots}} S_{a}^{r,s}(\overline{h}) k_{2h}^{r,s}(1-z) = \frac{1}{2} \left(\frac{z}{1-z}\right)^{a} + \text{Casimir-regular},$$
$$S_{a}^{r,s}(\overline{h}) = \frac{1}{\Gamma(-a-r)\Gamma(-a-s)} \frac{\Gamma(\overline{h}-r)\Gamma(\overline{h}-s)}{\Gamma(2\overline{h}-1)} \frac{\Gamma(\overline{h}-a-1)}{\Gamma(\overline{h}+a+1)},$$
(3.54)

we find

$$R_{h=2h_{O}}(\overline{h}) \sim 2R_{h^{*}}^{*}K_{0}^{0,s}(\overline{h}^{*})S_{h^{*}-2h_{O}}^{0,s}(\overline{h}),$$

$$R_{h=h_{O}+h_{*}'}(\overline{h}) \sim 2R_{h^{*}}^{*}K_{0}^{s,0}(\overline{h}^{*})S_{h^{*}-2h_{O}}^{0,s}(\overline{h}),$$
(3.55)

where \sim means that both sides have the same leading behavior at large *h*.

In (3.50), the left-hand side will have a $\log(1 - \overline{z})$ term when s = 0. In the usual four-point lightcone bootstrap, the $\log(1 - \overline{z})$ term determines the behavior of the

anomalous dimensions of double-twist operators. However, unlike in the usual lightcone bootstrap, (3.50) does not possess an identity operator. As a result, the interpretation of $\log(1-\overline{z})$ as coming from anomalous dimensions does not work in this case. To see this, consider the case where O is a spin-1 conserved current \mathcal{J} . In this case, O'_* should be the lowest twist operator with spin J' = 1 in the $\mathcal{J} \times \mathcal{J} \times \mathcal{J}$ OPE, which is just \mathcal{J} itself. So, $\delta'_* = \delta_{\mathcal{J}}$ and s = 0. For s = 0, the leading term of left hand side of (3.50) becomes

$$-R_{h^*}^* z^{h^* - 2h_O} \frac{\Gamma(2\overline{h}^*)}{\Gamma(\overline{h}^*)^2} (2\psi(\overline{h}^*) - 2\psi(1) + \log(1 - \overline{z})).$$
(3.56)

To see how the above expression can be produced by the right hand side of (3.50), note that near s = 0, the coefficients $R_{h=2h_O}(\overline{h})$ and $R_{h=h_O+h'_*}(\overline{h})$ in (3.55) are singular and take the form

$$R_{h=2h_{O}}(\overline{h}) = \frac{\widetilde{R}}{s} + \widetilde{R}^{(0)} + O(s),$$

$$R_{h=h_{O}+h'_{*}}(\overline{h}) = -\frac{\widetilde{R}}{s} + \widetilde{R}^{(0)} + O(s),$$
(3.57)

where

$$\widetilde{R} = -2R_{h^*}^* \frac{\Gamma(2\overline{h}^*)}{\Gamma(\overline{h}^*)^2} S_{h^*-2h_O}^{0,s}(\overline{h}), \qquad \widetilde{R}^{(0)} = (-\psi(1) + \psi(\overline{h}^*))\widetilde{R}.$$
(3.58)

Therefore, the contribution from the two celestial double-twist operators becomes (going back to the notation $g_{\delta,j}$ for conformal blocks)

$$\begin{split} &\lim_{s \to 0} R_{h=2h_{O}}(\overline{h}) g_{4h_{O}+j,j}^{0,s} + R_{h=h_{O}+h'_{*}}(\overline{h}) g_{2h_{O}-2s+j,j}^{0,s} \\ &= \lim_{s \to 0} \left(\frac{\widetilde{R}}{s} + \widetilde{R}^{(0)} \right) \left(g_{4h_{\mathcal{J}}+j,j}^{0,0} + s \partial_{s} g_{4h_{\mathcal{J}}+j,j}^{0,s} \right) + \left(-\frac{\widetilde{R}}{s} + \widetilde{R}^{(0)} \right) \left(g_{4h_{\mathcal{J}}+j,j}^{0,0} + s \partial_{s} g_{4h_{\mathcal{J}}-2s+j,j}^{0,s} \right) \\ &= 2\widetilde{R}^{(0)} g_{4h_{\mathcal{J}}+j,j}^{0,0} + 2\widetilde{R} \partial_{\delta} g_{\delta,j}^{0,0} |_{\delta \to 4h_{\mathcal{J}}+j}, \end{split}$$
(3.59)

which correctly reproduces (3.56) after performing the sum over \overline{h} . We see that the $\log(1-\overline{z})$ of (3.56) comes from a near-cancellation of coefficients between the two celestial double-twist families at the degenerate point s = 0.

3.3 Extracting light-ray OPE data from the leading order collinear EEEC

In the previous section, we argued that the EEEC in the collinear limit can be decomposed into conformal blocks (up to a Jacobian factor). In this section, we study the decomposition of the leading-order collinear EEEC in $\mathcal{N} = 4$ SYM and

QCD recently computed in [55]. In the case of $\mathcal{N} = 4$ SYM, the authors of [55] consider sink/source states created by the operator $\text{Tr}F^2$ (with $\Delta_O = 4$). For the QCD case, they consider both the gluon jet, created by $\text{Tr}F^2$, and the quark jet, created by the quark contribution to the electromagnetic current J^{μ} . Specifically, they contract indices between the bra and the ket, so that the quark jet event shape is an expectation value in the density matrix $|J^{\mu}(p)\rangle\langle J_{\mu}(p)|$.

Note that [55] worked at low enough loop order that the β -function does not enter, so QCD can be thought of as conformal for the purposes of studying their results. At higher orders in perturbation theory, selection rules in *J*, such as those discussed in [12, 15] will be broken. However, the celestial block expansion should still be valid, since it relies on Lorentz invariance alone. We leave an investigation of these effects to the future.

The results of [55] for the leading-order collinear EEEC can be summarized by three functions of cross ratios, $G_{N=4}(z, \overline{z})$ (equation (5.2) and (5.3) in [55]), $G_{\text{QCD}}^g(z, \overline{z})$ (square bracket in equation (5.14) in [55]), and $G_{\text{QCD}}^q(z, \overline{z})$ (square bracket in equation (5.16) in [55]). The relation between $G(z, \overline{z})$ and the EEEC is

$$EEEC(\zeta_{12},\zeta_{13},\zeta_{23}) = \frac{g^4 N_c^2}{64\pi^5 \zeta_{12}^2 \sqrt{-\zeta_{12}^2 - \zeta_{13}^2 - \zeta_{23}^2 + 2\zeta_{12}\zeta_{13} + 2\zeta_{12}\zeta_{23} + 2\zeta_{13}\zeta_{23}}} \times G_{\mathcal{N}=4}(z,\overline{z}) + \dots,$$

$$EEEC(\zeta_{12},\zeta_{13},\zeta_{23}) = \frac{g^4}{32\pi^5 \zeta_{12}^2 \sqrt{-\zeta_{12}^2 - \zeta_{13}^2 - \zeta_{23}^2 + 2\zeta_{12}\zeta_{13} + 2\zeta_{12}\zeta_{23} + 2\zeta_{13}\zeta_{23}}} \times G_{QCD}^{g/q}(z,\overline{z}) + \dots.$$

(3.60)

In the weak coupling limit, we can expand (3.47) as

$$\mathcal{G} = a_0 \left(\mathcal{G}^{(0)} + a \mathcal{G}^{(1)} + a^2 \mathcal{G}^{(2)} + \ldots \right),$$

$$\delta_i = \delta_i^{(0)} + a \gamma_i^{(1)} + a^2 \gamma_i^{(2)} + \ldots,$$

$$\delta'_* = 5 + a \gamma'^{(1)}_* + a^2 \gamma'^{(2)}_* + \ldots,$$

$$R_{\delta,j;\delta'_*} = a_0 \left(R_{\delta,j;\delta'_*}^{(0)} + a R_{\delta,j;\delta'_*}^{(1)} + \ldots \right),$$
(3.61)

where for $\mathcal{N} = 4$ SYM we have $a_0 = \frac{g^2 N_c}{16\pi^3}$, $a = \frac{g^2 N_c}{4\pi^2}$, and for QCD $a_0 = \frac{g^2}{2\pi^3}$, $a = \frac{g^2}{16\pi^2}$.⁷ As we explain later in section 3.3.1, we also set $\delta'_*^{(0)} = 5$ since it is the J' = 4 point on the twist-2 trajectory. Comparing (3.46), (3.60) and (3.61), we

⁷We choose a_0 such that there is no prefactor in (3.62).

immediately see that $\mathcal{G}^{(0)} = 0$ and⁸

$$\mathcal{G}^{(1)}(\zeta_{12}, z, \overline{z}) = (z\overline{z})^3 G(z, \overline{z}) \tag{3.62}$$

for both N = 4 SYM and QCD. The expansion of (3.47) in the weak coupling limit is given by

$$\begin{aligned} \mathcal{G}^{(0)}(\zeta_{12}, z, \overline{z}) &= \sum_{\delta^{(0)}, j} \langle R^{(0)}_{\delta, j; \delta'_{*}} \rangle g^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_{*}=5})}_{\delta^{(0)}, j}(z, \overline{z}) \\ \mathcal{G}^{(1)}(\zeta_{12}, z, \overline{z}) &= \sum_{\delta^{(0)}, j} \left(\langle R^{(1)}_{\delta, j; \delta'_{*}} \rangle g^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_{*}=5})}_{\delta^{(0)}, j}(z, \overline{z}) + \langle R^{(0)}_{\delta, j; \delta'_{*}} \gamma^{(1)}_{\delta, j} \rangle \partial_{\delta} g^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_{*}=5})}_{\delta^{(0)}, j}(z, \overline{z}) \right) \\ &+ \gamma'^{(1)}_{*} \sum_{\delta^{(0)}, j} \left(\langle R^{(0)}_{\delta, j; \delta'_{*}} \rangle \partial_{\delta'_{*}} g^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_{*}})}_{\delta^{(0)}, j}(z, \overline{z}) + \frac{1}{2} \log(\zeta_{12}) \mathcal{G}^{(0)}(z, \overline{z}) \right), \end{aligned}$$
(3.63)

where the notation $\langle ... \rangle$ represents a sum of the contributions from possibly degenerate operators, following e.g. [86].⁹ Since $\mathcal{G}^{(0)} = 0$, we must have $\langle R^{(0)}_{\delta,j;\delta_*} \rangle = 0$. The function $\mathcal{G}^{(1)}$ then becomes independent of ζ_{12} (which is consistent with (3.62)), and it can be written as

$$\mathcal{G}^{(1)}(z,\overline{z}) = \sum_{\delta^{(0)},j} \left(\langle R^{(1)}_{\delta,j;\delta'_*} \rangle g^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_*=5})}_{\delta^{(0)},j}(z,\overline{z}) + \langle R^{(0)}_{\delta,j;\delta'_*} \gamma^{(1)}_{\delta,j} \rangle \partial_{\delta} g^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_*=5})}_{\delta^{(0)},j}(z,\overline{z}) \right).$$
(3.64)

The values of $\delta^{(0)}$ and *j* appearing in the decomposition (3.64) can be related to the spectrum of the theory using the light-ray OPE formula. We describe this relation in section 3.3.1. We then obtain coefficients $\langle R_{\delta,j;\delta'_*}^{(1)} \rangle$ and $\langle R_{\delta,j;\delta'_*}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ using both a direct series expansion around the OPE limit (section 3.3.2) and the Lorentzian inversion formula (section 3.3.3). Our results are summarized in figure 3.2, where we plot the allowed values of δ , *j* in (3.64) and indicate points for which we obtained the coefficients $\langle R_{\delta,j;\delta'_*}^{(1)} \gamma_{\delta,j}^{(1)} \rangle$ using either of the two methods.

⁸Note that (3.47) assumes that the sink/source states are created by scalar operators. Even though for the QCD quark jet case, the current J^{μ} is spin-1, we can still treat it as a scalar here since we contract indices between the bra and ket. This will produce an additional factor of 2 due to the fact that for a spin-1 operator V_{μ} , $\langle 0|V^{\mu}(x)V_{\mu}(0)|0\rangle = 2\langle 0|\phi(x)\phi(0)|0\rangle$ where ϕ is a scalar with dimension Δ_V , and we use the conventional two-point structures [30] for operators with spin 0 and 1 in CFT.

⁹The degeneracy can come from either operators with the same δ , *j* or operators with the same δ'_* .



Figure 3.2: A plot showing the values of δ , *j* appearing in the conformal block decomposition (3.64). Black dashed line is the improved unitarity bound from (3.68). The red dot appears only in QCD, and the black dots appear in both QCD and $\mathcal{N} = 4$ SYM. The coefficients $\langle R_{\delta,j;\delta'_*}^{(1)} \rangle$ and $\langle R_{\delta,j;\delta'_*}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ of dots with a blue circle can be obtained using the direct decomposition method. The $\langle R_{\delta,j;\delta'_*}^{(1)} \rangle$ coefficient of the green line and the $\langle R_{\delta,j;\delta'_*}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ coefficient of the orange line can be obtained using the Lorentzian inversion formula.

3.3.1 Predictions from the light-ray OPE

Schematically, the light-ray OPE for two stress tensors is given by [15, 77]

$$\mathbf{L}[T] \times \mathbf{L}[T] \sim \sum_{i} \mathbb{O}_{i,J=3,j=0}^{+} + \mathbb{O}_{i,J=3,j=2}^{+} + \mathbb{O}_{i,J=3,j=4}^{+} + \sum_{n,i} \mathcal{D}_{2n} \mathbb{O}_{i,J=3+2n,j=4}^{+}.$$
(3.65)

This formula allows us to predict which quantum numbers $\delta^{(0)}$, *j* appear in the decomposition (3.64). First, we immediately see that only even values of *j* can appear. This is consistent with the fact that the OPE of two identical scalars on the celestial sphere only includes operators with even transverse spin.

To see the allowed values of δ , let us denote the conventional twist $\tau = \Delta - J$ of a trajectory with transverse spin j by τ_j .¹⁰ Note that the $T \times T$ OPE only contains trajectories with transverse spin j = 0, 2, 4. We also define a "celestial twist" $\tau_c \equiv \delta - j$, where δ , j are the quantum numbers appearing in (3.64).

Using (3.65), we can see that for j = 0, 2, the relation between τ_i and τ_c is given by

$$\tau_c(j=0,2) = \delta - j = \Delta(J=3) - 1 - j = \tau_j + 2 - j, \qquad (3.66)$$

¹⁰We are working in perturbation theory, so the twist of each Regge trajectory is fixed and τ_j does not depend on J. In fact, we will simply use τ_j to label each Regge trajectory.

while for $j \ge 4$ we have

$$\tau_c(j) = \delta - j = \Delta(J = -1 + j) - 1 - j = \tau_{j=4} - 2.$$
(3.67)

In 4d, the conventional twist τ_i should satisfy the improved unitarity bound [49]

$$\tau_j \ge \max\{2, j\}. \tag{3.68}$$

Consequently, one might expect that the values of τ_i and $\tau_c(j)$ that can appear are

$$\tau_{j=0} = 2, 4, 6, \dots \implies \tau_c(j=0) = 4, 6, 8, \dots,$$

$$\tau_{j=2} = 2, 4, 6, \dots \implies \tau_c(j=2) = 2, 4, 6, \dots,$$

$$\tau_{j=4} = 4, 6, 8, \dots \implies \tau_c(j \ge 4) = 2, 4, 6, \dots.$$
(3.69)

However, as discussed in [87, 73], the contribution of the j = 4, $\tau_{j=4} = 4$ operator vanishes at the leading order for both $\mathcal{N} = 4$ SYM and QCD, and the contribution of the j = 2, $\tau_{j=2} = 2$ operator vanishes in $\mathcal{N} = 4$ SYM due to supersymmetry. So the actual values of τ_j and $\tau_c(j)$ appearing at the leading order should be

$$\tau_{j=0} = 2, 4, 6, \dots \implies \tau_c(j=0) = 4, 6, 8, \dots,$$

$$\tau_{j=2} = 2, 4, 6, \dots \implies \tau_c(j=2) = 2, 4, 6, \dots,$$

$$\tau_{j=4} = 6, 8, 10, \dots \implies \tau_c(j \ge 4) = 4, 6, 8, \dots,$$
(3.70)

where the bold $\tau_{j=2} = 2$ appears only in QCD.

What operators realize these quantum numbers? We will focus on operators with leading τ_c . We will identify light-ray operators by writing local operators on their Regge trajectories, with *J* as a free parameter. The actual light-ray operators are obtained by light-transforming and analytically continuing *J* to the appropriate value according to (3.65).

For N = 4 SYM, the leading τ_c is 4, coming from operators with j = 0, 2, 4. For $j = 0, \tau_{j=0} = 2$, the operators can be schematically written as¹¹

$$\overline{\phi}\partial^{J}\phi, \quad \overline{\psi}\partial^{J-1}\psi, \quad \overline{F}\partial^{J-2}F.$$
 (3.71)

For j = 2, $\tau_{j=2} = 4$, we have

$$\psi\partial^{J_1}\psi\partial^{J_2}\psi\partial^{J-J_1-J_2-2}\psi, \quad F\partial^{J-2}\Box F, \quad \overline{F}\partial^{J_1}F\partial^{J_2}F\partial^{J-J_1-J_2-4}F, \tag{3.72}$$

¹¹We use the $\mathfrak{su}(2) \oplus \mathfrak{su}(2)$ spinor indices for the 4d Lorentz indices. The derivative is $\partial_{\dot{\alpha}\beta}$, and the field contents are $\overline{\phi}, \phi, \overline{\psi}_{\dot{\alpha}}, \psi_{\beta}, \overline{F}_{\dot{\alpha}_1\dot{\alpha}_2}, F_{\beta_1\beta_2}$. In (3.71)-(3.74), the Lorentz indices are implicitly symmetrized and the gauge indices are implicitly contracted.

and for j = 4, $\tau_{j=4} = 6$,

$$F\partial^{J_1}F\partial^{J_2}F\partial^{J-J_1-J_2-4}\Box F, \quad \overline{F}\partial^{J_1}F\partial^{J_2}F\partial^{J_3}F\partial^{J_4}F\partial^{J-J_1-J_2-J_3-J_4-6}F.$$
(3.73)

Note that there are degeneracies for all three values of j. (Supersymmetry relates some of these trajectories, but we do not study the consequences of supersymmetry here.) On the other hand, in QCD the leading τ_c is 2, and it is carried by one operator with j = 2, $\tau_{j=2} = 2$:

$$F\partial^{J-2}F. (3.74)$$

Thus, the leading τ_c operator is non-degenerate in QCD.

Finally, let us discuss the value of δ'_* in (3.64). The value of δ'_* should be determined by a generalized light-ray OPE formula

$$\mathbf{L}[T] \times \mathbb{O}_{i,J=3,j} \sim \mathbb{O}'_{i,J=4,\lambda},\tag{3.75}$$

where $\mathbb{O}'_{i,J=4,\lambda}$ is some unknown object that transforms like a primary operator with scaling dimension $\Delta_{\mathbb{O}'} = 1 - J = -3$. Though we do not have a rigorous definition of the object $\mathbb{O}'_{i,J=4,\lambda}$ for a general nonperturbative CFT, we expect that it should be related to light transforms of operators in the $T \times T \times T$ OPE with J = 4. Indeed, it has been shown in [87, 73] that for perturbative QCD, this object is just the light transformed operator $\mathbf{L}[O_{i,J=4,\lambda}]$ itself (at least at the leading order). Recently, there is also evidence from LHC data showing that the scaling behavior of the three-point energy correlator in the perturbative regime is governed by the twist-2 spin-4 anomalous dimension [74]. Therefore, in this paper we will assume that δ'_{*} is given by $\delta'_{*} = \Delta'_{*} - 1 = 5$ (at leading order in perturbation theory), since $\Delta'_{*} = 6$ is the scaling dimension of the leading twist-2, spin-4 operator in the $T \times T \times T$ OPE.¹²

3.3.2 Celestial block coefficients from direct decomposition

We now explain how to obtain the coefficients $\langle R_{\delta,j;\delta'_*}^{(1)} \rangle$ and $\langle R_{\delta,j;\delta'_*}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ in (3.64) using the known result for $\mathcal{G}^{(1)}(z,\overline{z})$ computed in [55]. Firstly, we can simply expand (3.64) in the OPE limit $z = re^{i\theta}, \overline{z} = re^{-i\theta}$ with $r \to 0$ and compare both sides order-by-order in r. Recall that the 2d block $g_{\delta,j}^{(\mathcal{EE}\mathcal{EP}_{\delta'_*}=5)}(z,\overline{z})$ in (3.64) is given by

$$g_{\delta,j}^{(\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_{*}=5})}(z,\overline{z}) = \frac{1}{1+\delta_{0,j}} (k_{\delta+j}^{0,-1}(z)k_{\delta-j}^{0,-1}(\overline{z}) + k_{\delta-j}^{0,-1}(z)k_{\delta+j}^{0,-1}(\overline{z})),$$

$$k_{\beta}^{r,s}(x) = x^{\frac{\beta}{2}} {}_{2}F_{1}\left(\frac{\beta}{2} - r, \frac{\beta}{2} + s, \beta, x\right).$$
(3.76)

¹²See also [75], where they show that in QCD, the leading correction to the scaling of collinear EEEC determined by $\delta'_* = 5$ can be used for top quark mass measurements.

The OPE limit of $\mathcal{G}^{(1)}$ corresponds to a "squeezed limit" on the celestial sphere, where two of the detectors are taken to be even closer after the collinear limit. The expansion of [55] in the squeezed limit has been studied in [87, 73] up to $O(r^{10})$.¹³ Using these results, we find that the first few coefficients $\langle R_{\delta,j}^{(1)} \rangle$ and $\langle R_{\delta,j}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ in $\mathcal{N} = 4$ SYM are given by

$$\langle R_{4,0}^{(1)} \rangle = 1, \quad \langle R_{6,0}^{(1)} \rangle = \frac{41}{12} - \frac{\pi^2}{4}, \quad \langle R_{6,2}^{(1)} \rangle = \frac{107}{60} - \frac{\pi^2}{6}, \\ \langle R_{8,0}^{(1)} \rangle = \frac{471}{600} - \frac{\pi^2}{15}, \quad \langle R_{8,2}^{(1)} \rangle = \frac{4883}{2800} - \frac{9\pi^2}{56}, \quad \langle R_{8,4}^{(1)} \rangle = \frac{2843}{5040} - \frac{\pi^2}{18}, \\ \langle R_{8,0}^{(0)} \gamma_{8,0}^{(1)} \rangle = \frac{2}{5}, \quad \langle R_{8,2}^{(0)} \gamma_{8,2}^{(1)} \rangle = -\frac{1}{20}.$$
 (3.77)

In QCD, we use $R_{\delta,j}^{(1)g/q}$ to denote coefficients for the gluon/quark jet. For the QCD gluon jet, we find

$$\langle R_{4,0}^{(1)g} \rangle = \frac{98C_A^2 + 14C_A n_f T_F + 15C_F n_f T_F}{1600}, R_{4,2}^{(1)g} = \frac{C_A (C_A - 2n_f T_F)}{2880}, \langle R_{6,0}^{(1)g} \rangle = \frac{(636386 - 63000\pi^2) C_A^2 + 2 (12600\pi^2 - 120899) C_A n_f T_F - 819C_F n_f T_F}{403200}, \langle R_{6,2}^{(1)g} \rangle = \frac{(834469 - 84000\pi^2) C_A^2 + 4 (13125\pi^2 - 129587) C_A n_f T_F + 1260C_F n_f T_F}{504000}$$
(3.78)

Note that we do not use the bracket notation $\langle \cdots \rangle$ for the coefficient $R_{4,2}^{(1)}$, because from the discussion in section 3.3.1 it is non-degenerate. For the QCD quark jet we find

$$\langle R_{4,0}^{(1)q} \rangle = \frac{C_F(91C_A + 240C_F + 13n_f T_F)}{4800},$$

$$R_{4,2}^{(1)q} = \frac{C_F(C_A - 2n_f T_F)}{2880},$$

$$\langle R_{6,0}^{(1)q} \rangle = \frac{C_F\left((109200\pi^2 - 1077733)C_A - 28\left(5100\pi^2 - 50929\right)C_F + 3\left(111199 - 11200\pi^2\right)n_f T_F\right)}{403200}$$

$$\langle R_{6,2}^{(1)q} \rangle = \frac{C_F\left((157500\pi^2 - 1548703)C_A - 210\left(1300\pi^2 - 12859\right)C_F + 326n_f T_F\right)}{1008000}.$$

$$(3.79)$$

¹³We thank Hao Chen, Ian Moult, and Hua Xing Zhu for sending us a mathematica notebook containing the expansion.

We have further expanded the results of [55] up to $O(r^{12})$, which will be helpful when comparing to the Lorentzian inversion formula result in section 3.3.3. We record the coefficients up to $\delta = 12$ in appendix B.2.

3.3.3 The Lorentzian inversion formula on the celestial sphere

Direct decomposition yields OPE data at low dimensions δ , but becomes cumbersome as δ gets larger. Alternatively, we can use the Lorentzian inversion formula (LIF) [9] to extract OPE data from a four-point correlator. Like the lightcone bootstrap, the LIF requires us to analytically continue the correlator to Lorentzian signature — which in our case means complexifying the celestial sphere. Again, it is not clear whether this analytic continuation is admissible nonperturbatively. However, nothing prevents us from using the LIF as a tool in perturbation theory, as long as perturbative correlators are well-behaved. Indeed, we do not observe any pathologies when analytically continuing the results of [55] in z, \overline{z} . It would be interesting to study the analytic structure of the collinear EEEC as a function of z, \overline{z} at higher orders in perturbation theory.

The LIF is only valid for $j > j_0$, where the "Regge intercept" j_0 controls the behavior of the correlator in the Regge limit. To reach the Regge limit, one should first take \overline{z} around the branch point at 1, and then take both z and \overline{z} to zero. From the point of view of celestial CFT, this is a strange kinematic regime, and we do not have rigorous bounds (much less physical intuition) for how the correlator should behave there. Note that the Regge limit on the celestial sphere has nothing to do with the Regge limit in *d*-dimensional Minkowski space (as far as we know), and that the celestial Regge intercept j_0 is not obviously related to the usual Regge intercept J_0 . However, we can study this limit in perturbation theory.

In $\mathcal{N} = 4$ SYM, we find that the leading term of $\mathcal{G}^{(1)}(z, \overline{z})$ in the celestial Regge limit is given by

$$\mathcal{G}_{\text{Regge}}^{(1)}(z,\overline{z}) = \frac{i\pi r^3 (w^2+1) \left(2(w^2-1) \left((w^2-1)^2 \log(w+\frac{1}{w})-2w^2\right)-2(w^6-3w^4-3w^2+1) \log w\right)}{w(w^2-1)^3},$$
(3.80)

where we set z = rw and $\overline{z} = r/w$. The scaling r^3 implies that the celestial Regge intercept is given by $j_0 = -2$ at this order in perturbation theory. The expressions for $\langle R_{\delta,j;\delta'_*}^{(1)} \rangle$ and $\langle R_{\delta,j;\delta'_*}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ for $\mathcal{N} = 4$ SYM obtained from the LIF should then be valid for all $j = 0, 2, 4, \ldots$

For the QCD gluon jet, $\mathcal{G}^{(1)g}$ can be written as

$$\mathcal{G}^{(1)g} = C_F n_f T_F \mathcal{G}_1^{(1)g} + C_A n_f T_F \mathcal{G}_2^{(1)g} + C_A^2 \mathcal{G}_3^{(1)g}.$$
(3.81)

We find that the Regge intercept for $\mathcal{G}_1^{(1)g}$ is $j_0 = 0$, while for $\mathcal{G}_2^{(1)g}$ and $\mathcal{G}_3^{(1)g}$ the intercept is $j_0 = 2$. Therefore, for the coefficient proportional to $C_F n_f T_F$, the LIF result will agree with the result from direct decomposition only for j > 0. For the other two flavor structures ($C_A n_f T_F$ and C_A^2), we expect the result to agree only for j > 2. Similarly, for QCD quark jet, $\mathcal{G}^{(1)q}$ can be written as

$$\mathcal{G}^{(1)q} = C_F n_f T_F \mathcal{G}_1^{(1)q} + (C_A - 2C_F) C_F \mathcal{G}_2^{(1)q} + \mathcal{G}_3^{(1)q}, \qquad (3.82)$$

where $\mathcal{G}_3^{(1)q}$ contains both C_F^2 and $C_F C_A$ factors. We find that the Regge intercept for $\mathcal{G}_2^{(1)q}$ is $j_0 = 0$, and the intercept for $\mathcal{G}_1^{(1)q}$ and $\mathcal{G}_3^{(1)q}$ is $j_0 = 2$.

Let us now briefly review the LIF. Consider a four-point function $g(z, \overline{z})$ in 2d with conformal block expansion

$$g(z,\overline{z}) = \sum_{\delta,j} p_{\delta,j} g_{\delta,j}^{\delta_i}(z,\overline{z}), \qquad (3.83)$$

where only even j appear in the sum. The LIF in this case can be written as

$$C^{+}(\delta, j) = \frac{\kappa_{\delta+j}}{2} \int_{0}^{1} \int_{0}^{1} \frac{dz d\overline{z}}{(z\overline{z})^{2}} g_{j+1,\delta-1}^{\delta_{i}}(z,\overline{z}) d\text{Disc}_{t}[g(z,\overline{z})], \qquad (3.84)$$

where $\tilde{\delta}_i = 2 - \delta_i$, and

$$\kappa_{\beta} = \frac{\Gamma(\frac{\beta+\delta_1-\delta_2}{2})\Gamma(\frac{\beta-\delta_1+\delta_2}{2})\Gamma(\frac{\beta+\delta_3-\delta_4}{2})\Gamma(\frac{\beta-\delta_3+\delta_4}{2})}{2\pi^2\Gamma(\beta-1)\Gamma(\beta)}.$$
(3.85)

The double discontinuity $d\text{Disc}_t[g(z, \overline{z})]$ is

$$d\text{Disc}_{t}[g(z,\overline{z})] = \cos(\pi\phi)g(z,\overline{z}) - \frac{1}{2}e^{i\pi\phi}g^{\cup}(z,\overline{z}) - \frac{1}{2}e^{-i\pi\phi}g^{\cup}(z,\overline{z}),$$
$$\phi = \frac{\delta_{2} - \delta_{1} + \delta_{3} - \delta_{4}}{2},$$
(3.86)

where g^{\cup} and g^{\cup} indicate that we should take \overline{z} around 1 in the direction shown, with *z* held fixed. Finally, the OPE coefficients $p_{\delta,j}$ are given by

$$p_{\delta^*,j} = -\operatorname{Res}_{\delta=\delta^*} C^+(\delta,j).$$
(3.87)

To compute $p_{\delta,j}$, it is convenient to define a generating functional

$$C(z,\delta,j) = \kappa_{\delta+j} z^{\frac{\delta-j}{2}-1} \int_0^1 \frac{d\overline{z}}{\overline{z}^2} g_{j+1,\delta-1}^{\widetilde{\delta}_i}(z,\overline{z}) d\text{Disc}_t[g(z,\overline{z})],$$

$$C^+(\delta,j) = \int_0^1 \frac{dz}{2z} z^{-\frac{\delta-j}{2}} C(z,\delta,j).$$
(3.88)

In the small z limit, $C(z, \delta, j)$ should have the expansion

$$C(z,\delta,j) = \sum_{\tau_c} C(\tau_c,\delta,j) z^{\frac{\tau_c}{2}}.$$
(3.89)

Using (3.84), we see that $C^+(\delta, j)$ has a pole when $\delta = j + \tau_c$, and therefore the OPE coefficient $p_{\delta,j}$ can be written as¹⁴

$$p_{j+\tau_c,j} = 2C(\tau_c, j + \tau_c, j),$$
 (3.90)

where the additional factor of 2 is due to $z \leftrightarrow \overline{z}$ symmetry.

Now, suppose we have a weak-coupling expansion

$$g(z, \overline{z}) = a \left(g^{(0)}(z, \overline{z}) + a g^{(1)}(z, \overline{z}) + \cdots \right),$$

$$\delta = \delta^{(0)} + a \gamma^{(1)} + \cdots,$$

$$p_{\delta,j} = a \left(p^{(0)}_{\delta,j} + a p^{(1)}_{\delta,j} + \cdots \right),$$
(3.91)

with coupling constant *a*, and we are interested in finding $p_{\delta,j}^{(1)}$ and $p_{\delta,j}^{(0)}\gamma_{\delta,j}^{(1)}$. Expanding (3.87) near $\delta^{(0)}$, we have

$$C^{+}(\delta,j) \sim a \left(-\frac{\langle p_{\delta,j}^{(0)} \rangle}{\delta - \delta^{(0)}} - a \frac{\langle p_{\delta,j}^{(1)} \rangle}{\delta - \delta^{(0)}} - a \frac{\langle p_{\delta,j}^{(0)} \gamma_{\delta,j}^{(1)} \rangle}{(\delta - \delta^{(0)})^2} + \cdots \right), \tag{3.92}$$

where $\langle \cdots \rangle$ indicates a sum over possible degenerate operators. For the generating functional, the expansion near $\delta^{(0)}$ is

$$C(z, \delta, j) \sim a(C^{(0)}(\tau_c, \delta, j) + aC^{(1)}(\tau_c, \delta, j))z^{\frac{\delta^{(0)} + a\gamma^{(1)} - j}{2}} + \cdots \\ \sim a\left(C^{(0)}(\tau_c, \delta, j) + aC^{(1)}(\tau_c, \delta, j) + a\frac{\gamma^{(1)}}{2}C^{(0)}(\tau_c, \delta, j)\log z\right)z^{\frac{\tau_c^{(0)}}{2}},$$
(3.93)

where $\tau_c^{(0)} = \delta^{(0)} - j$. Let us plug (3.93) into (3.88) and compare it with (3.92). We see that after integrating over *z*, the $C^{(1)}(\tau_c, \delta, j)$ term becomes a simple pole corresponding to $\langle p_{\delta,j}^{(1)} \rangle$. On the other hand, the $\gamma^{(1)}C^{(0)}(\tau_c, \delta, j)$ term has an additional log *z* and becomes a double pole, corresponding to $\langle p_{\delta,j}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$. The precise formula is

$$\langle p_{j+\tau_c,j}^{(1)} \rangle = 2C(z,\delta,j) \Big|_{z^{\frac{\tau_c}{2}}},$$

$$\langle p_{j+\tau_c,j}^{(0)} \gamma_{j+\tau_c,j}^{(1)} \rangle = 4C(z,\delta,j) \Big|_{z^{\frac{\tau_c}{2}} \log z}.$$
(3.94)

¹⁴In general, there will be an additional Jacobian factor coming from the dependence of τ_c on *j*. At the order we are working in, τ_c on each trajectory is a constant and this Jacobian factor is just 1.

$\mathcal{N} = 4$ SYM

Let us now apply the Lorentzian inversion formula to $\mathcal{G}^{(1)}(z, \overline{z})$ in $\mathcal{N} = 4$ SYM. After plugging in $g(z, \overline{z}) = \mathcal{G}_{\mathcal{N}=4}^{(1)}(z, \overline{z})$ and $\delta_1, \delta_2, \delta_3 = 3, \delta_4 = 5$ in (3.84), we find that $\langle R_{\delta,j;\delta'_*}^{(1)} \rangle$ is nonzero for even j and "celestial twists" $\tau_c = \delta - j = 4, 6, 8, \ldots$. Furthermore, $\langle R_{\delta,j;\delta'_*}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ is nonzero for even j and $\tau_c = 6, 8, 10, \ldots$ (except that $\langle R_{\delta=6,j=0;\delta'_*}^{(0)} \gamma_{\delta=6,j=0}^{(1)} \rangle = 0$). We can find analytical expressions for $\langle R_{j+\tau_c,j;\delta'_*}^{(0)} \gamma_{j+\tau_c,j}^{(1)} \rangle$ for general τ_c , and also $\langle R_{j+4,j;\delta'_*}^{(1)} \rangle$.¹⁵ The results agree with those obtained from direct decomposition in section 3.3.2 and appendix B.2.

As an example, let us describe the detailed calculation for $\langle R_{\delta,j;\delta'_*}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ at celestial twist $\tau_c = 6$. For higher twists and $\langle R_{j+4,j;\delta'_*}^{(1)} \rangle$, we simply present the final result and leave details to appendix B.3. Using (3.94), we have

$$\langle R_{j+6,j;\delta_{*}}^{(0)} \gamma_{j+6,j}^{(1)} \rangle = 4 \left(\kappa_{2j+6} z^{2} \int_{0}^{1} \frac{d\overline{z}}{\overline{z}^{2}} g_{j+1,j+5}^{0,1}(z,\overline{z}) d\text{Disc}_{t} [\mathcal{G}(z,\overline{z})] \right) \Big|_{z^{3} \log z}$$

$$= 4 \kappa_{2j+6} \int_{0}^{1} \frac{d\overline{z}}{\overline{z}^{2}} k_{2j+6}^{0,1}(\overline{z}) d\text{Disc}_{t} [\mathcal{G}(z,\overline{z})] |_{z^{3} \log z},$$

$$(3.95)$$

where we have used the fact that the log z term of $\mathcal{G}(z, \overline{z})$ starts at z^3 . We have also used the SL(2, \mathbb{R}) expansion of the conformal block

$$g_{\delta,j}^{r,s}(z,\overline{z}) = z^{\frac{\delta-j}{2}} k_{\delta+j}^{r,s}(\overline{z}) + O(z^{\frac{\delta-j}{2}+1}),$$

$$k_{\beta}^{r,s}(x) = x^{\frac{\beta}{2}} {}_{2}F_{1}(\frac{\beta}{2} - r, \frac{\beta}{2} + s, \beta, x),$$

$$r = \frac{\delta_{1} - \delta_{2}}{2}, \qquad s = \frac{\delta_{3} - \delta_{4}}{2}.$$
(3.96)

Next we should compute the double discontinuity $d\text{Disc}_t[\mathcal{G}(z, \overline{z})]$. Note that the singularities of $\mathcal{G}(z, \overline{z})|_{z^3 \log z}$ at $\overline{z} = 1$ only have integer powers or single logarithms of $1 - \overline{z}$. For example,

$$\mathcal{G}(z,\overline{z})|_{z^{3}\log z} = -\frac{1}{4(1-\overline{z})} - \frac{3}{2}\log(1-\overline{z}) + O((1-\overline{z})^{0}).$$
(3.97)

Naively, such terms have vanishing dDisc. However the correct interpretation is that their dDiscs are distributions localized at $\overline{z} = 1$. See [26] for examples of dealing with such distributions. To compute them, we insert a regulator ϵ so that the powers become non-integer, removing the regulator after taking the dDisc. For example,

¹⁵It would be interesting to calculate $\langle R_{j+\tau_c,j;\delta_*}^{(1)} \rangle$ for general τ_c as well, but we leave that for future work.

inserting a regulator $1/(1-\overline{z})^{\epsilon}$, we have

$$\langle R_{j+6,j;\delta_{*}}^{(0)} \gamma_{j+6,j}^{(1)} \rangle$$

$$= \lim_{\epsilon \to 0} 4\kappa_{\beta} \int_{0}^{1} \frac{d\overline{z}}{\overline{z}^{2}} k_{\beta}^{0,1}(\overline{z}) d\text{Disc}_{t} \left[-\frac{1}{4(1-\overline{z})^{1+\epsilon}} - \frac{3}{2(1-\overline{z})^{\epsilon}} \log(1-\overline{z}) + O((1-\overline{z})^{-\epsilon}) \right]$$

$$= \lim_{\epsilon \to 0} \left(8\sin^{2}(\pi\epsilon)\kappa_{\beta} \int_{0}^{1} \frac{d\overline{z}}{\overline{z}^{2}} k_{\beta}^{0,1}(\overline{z}) \left(\frac{1}{4(1-\overline{z})^{1+\epsilon}} + \frac{3}{2(1-\overline{z})^{\epsilon}} \log(1-\overline{z}) + O((1-\overline{z})^{-\epsilon}) \right) \right)$$

$$- 8\pi \sin(2\pi\epsilon)\kappa_{\beta} \int_{0}^{1} \frac{d\overline{z}}{\overline{z}^{2}} k_{\beta}^{0,1}(\overline{z}) \frac{3}{2(1-\overline{z})^{\epsilon}} \right), \qquad (3.98)$$

where $\beta = 2j + 6$ and we have used (B.30) for the double discontinuities. The \overline{z} integrals will localize to $\overline{z} = 1$ when taking the $\epsilon \to 0$ limit, so we can expand the SL₂ block $k_{\beta}^{0,1}(\overline{z})$ in this limit. For example, in the last line of (3.98), we have

$$-8\pi\kappa_{\beta}\lim_{\epsilon\to 0}\sin(2\pi\epsilon)\int_{0}^{1}\frac{d\overline{z}}{\overline{z}^{2}}k_{\beta}^{0,1}(\overline{z})\frac{3}{2(1-\overline{z})^{\epsilon}}$$

$$=-8\pi\kappa_{\beta}\lim_{\epsilon\to 0}\sin(2\pi\epsilon)\int_{0}^{1}\frac{d\overline{z}}{\overline{z}^{2}}\left(\frac{\Gamma(\beta)}{\Gamma(\frac{\beta}{2})\Gamma(\frac{\beta}{2}+1)}\frac{1}{1-\overline{z}}+\cdots\right)\frac{3}{2(1-\overline{z})^{\epsilon}}$$

$$=-8\pi\kappa_{\beta}\frac{3\Gamma(\beta)}{2\Gamma(\frac{\beta}{2})\Gamma(\frac{\beta}{2}+1)}\lim_{\epsilon\to 0}\sin(2\pi\epsilon)\int_{0}^{1}\frac{d\overline{z}}{\overline{z}^{2}}\left(-\frac{1}{\epsilon}\delta(1-\overline{z})+O(\epsilon^{0})\right)$$

$$=\frac{12\Gamma(\frac{\beta}{2}-1)\Gamma(\frac{\beta}{2})}{\Gamma(\beta-1)},$$
(3.99)

where we have used the expansion of $k_{\beta}^{0,1}(\overline{z})$ around $\overline{z} = 1$ and the distributional identity $\frac{1}{x^{n+\epsilon}}\theta(x) = \frac{1}{\epsilon} \frac{(-1)^n}{(n-1)!} \delta^{(n-1)}(x) + O(\epsilon^0)$. Performing similar calculations for the other terms, we obtain

$$\langle R_{j+6,j;\delta_*}^{(0)} \gamma_{j+6,j}^{(1)} \rangle = -\frac{j(j+5)\Gamma(j+2)^2}{2\Gamma(2j+4)}.$$
(3.100)

Similarly, we can use (3.94) to calculate $\langle R_{j+\tau_c,j;\delta'_*}^{(0)}\gamma_{j+\tau_c,j}^{(1)}\rangle$ with higher τ_c and $\langle R_{j+4,j}^{(1)}\rangle$. For example, for $\tau_c = 8$ we have (see (B.45) for the expression for general τ_c)

$$\langle R_{j+8,j}^{(0)} \gamma_{j+8,j}^{(1)} \rangle = \frac{\left(12(j+3)(j+4)S_1(j+3) - (19j^2 + 133j + 192)\right)\Gamma(j+3)\Gamma(j+4)}{3\Gamma(2j+7)},$$
(3.101)

where $S_1(n) = \sum_{k=1}^{n} \frac{1}{k}$ is the harmonic number. For $\langle R_{j+4,j}^{(1)} \rangle$, the result is

$$\langle R_{j+4,j}^{(1)} \rangle = \frac{\Gamma(j+2)^2}{6(j+1)\Gamma(2j+3)} \\ \times \left(-9 + 2\pi^2 + j(j+3)(-6+\pi^2) + \frac{6(j+1)}{(j+3)^2(j+4)} {}_3F_2 \left(\begin{array}{cc} 2, & 3, & j+3\\ j+4, & j+5 \end{array}; 1 \right) \right).$$

$$(3.102)$$

QCD

We can perform a similar calculation for the QCD case by replacing $g(z, \overline{z})$ with $\mathcal{G}^{(1)g/q}$ in (3.84). For the gluon jet case, we again find that $\langle R_{j+\tau_c,j}^{(0)g} \gamma_{j+\tau_c,j}^{(1)} \rangle$ is nonzero for $\tau_c = 6, 8, \cdots$ and even *j*. The expressions for $\tau_c = 6$ and $\tau_c = 8$ are given by

$$\langle R_{j+6,j}^{(0)g} \gamma_{j+6,j}^{(1)} \rangle = -\frac{\Gamma(j+3)\Gamma(j+4)}{80\Gamma(2j+5)} C_A n_F T_F - \frac{(7j^2+35j-18)\Gamma(j+3)\Gamma(j+2)}{80\Gamma(2j+5)} C_A^2,$$
(3.103)

and

$$\begin{split} \langle R_{j+8,j}^{(0)g} \gamma_{j+8,j}^{(1)} \rangle \\ &= -\frac{3\Gamma(j+4)\Gamma(j+5)}{40\Gamma(2j+7)} C_F n_F T_F - \frac{\left(28j^4 + 392j^3 + 1697j^2 + 2275j + 204\right)\Gamma(j+3)\Gamma(j+4)}{336\Gamma(2j+7)} C_A n_F T_F \\ &+ \frac{\left(420\left(j^2 + 7j + 12\right)S_1(j+3) + 35j^4 + 490j^3 + 1052j^2 - 4641j - 10056\right)\Gamma(j+3)\Gamma(j+4)}{840\Gamma(2j+7)} C_A^2. \end{split}$$

For the $\langle R_{j+\tau_c,j}^{(1)g} \rangle$ coefficient, although one can see from (3.78) that it has leading twist $\tau_c = 2$, the Lorentzian inversion formula will only give nonzero $\langle R_{j+\tau_c,j}^{(1)g} \rangle$ for $\tau_c = 4, 6, \ldots$ This is because the only nonzero $\langle R_{j+\tau_c,j}^{(1)g} \rangle$ with $\tau_c = 2$ has j = 2, and only contains flavor structures with Regge intercept $j_0 = 2$ (C_A^2 and $C_A n_f T_F$). Therefore we do not expect the Lorentzian inversion formula to give the correct result for $\tau_c = 2$. For $\tau_c = 4$, we obtain

$$\begin{split} \langle R_{j+4,j}^{(1)g} \rangle \\ &= \frac{\Gamma(j+2)\Gamma(j+3)}{80\Gamma(2j+3)} C_F n_F T_F \\ &+ \left(-\frac{(31j^2+93j+17)\Gamma(j+1)\Gamma(j+2)}{7200\Gamma(2j+3)} \\ &+ \frac{\Gamma(j+1)\Gamma(j+2)^2\Gamma(j+3)}{4\Gamma(2j+3)\Gamma(j+4)\Gamma(j+5)} \left(2_3 F_2 \left(\begin{array}{ccc} 2, & 3, & j-2 \\ j+4, & j+5 \end{array}; 1 \right) \right) \\ &- 2_3 F_2 \left(\begin{array}{ccc} 2, & 3, & j-1 \\ j+4, & j+5 \end{array}; 1 \right) + {}_3 F_2 \left(\begin{array}{ccc} 2, & 3, & j \\ j+4, & j+5 \end{array}; 1 \right) \\ &+ \left(-\frac{(315+(j+1)(j+2)(-1117+150\pi^2))\Gamma(j+1)\Gamma(j+2)}{7200\Gamma(2j+3)} \\ &+ \frac{\Gamma(j+1)\Gamma(j+2)^2}{8\Gamma(2j+3)\Gamma(j+4)} \sum_{k=0}^4 c_{k3} F_2 \left(\begin{array}{ccc} 1, & 2, & j-3+k \\ j+3, & j+4 \end{array}; 1 \right) \right) C_A^2, \end{split}$$
(3.105)

where

$$c_0 = 1, \quad c_1 = -2, \quad c_2 = 3, \quad c_3 = -2, \quad c_4 = 1.$$
 (3.106)

As expected from the value of the Regge intercept, the results (3.103), (3.104) and (3.105) agree with (3.78), (B.24) and (B.25) for j > 2, but for the flavor structure $C_F n_f T_F$ they also agree at j = 2.

For the quark jet, the calculation is almost identical. We find that $\langle R_{j+\tau_c,j}^{(0)q} \gamma_{j+\tau_c,j}^{(1)} \rangle$ is nonzero for $\tau_c = 6, 8, \cdots$, and for $\tau_c = 6$ and $\tau_c = 8$ the expressions are

$$\langle R_{j+6,j}^{(0)q} \gamma_{j+6,j}^{(1)} \rangle = \frac{C_F(C_A(19j^2 + 95j - 6) - 32C_F(2j^2 + 10j - 3))\Gamma(j+2)\Gamma(j+3)}{480\Gamma(2j+5)},$$
(3.107)

and

$$\begin{split} \langle R_{j+8,j}^{(0)q} \gamma_{j+8,j}^{(1)} \rangle \\ &= -\frac{(10j^4 + 140j^3 + 637j^2 + 1029j + 204)\Gamma(j+3)\Gamma(j+4)}{120\Gamma(2j+7)} C_F n_F T_F \\ &- \frac{3(3j^2 + 21j + 16)\Gamma(j+3)\Gamma(j+4)}{160\Gamma(2j+7)} C_F (C_A - 2C_F) - \frac{(23j^2 + 161j - 24)\Gamma(j+3)\Gamma(j+4)}{60\Gamma(2j+6)} C_F^2 \\ &+ \frac{(10j^4 + 140j^3 + 401j^2 - 623j - 3018 + 120(j+3)(j+4)S_1(j+3))\Gamma(j+3)\Gamma(j+4)}{240\Gamma(2j+7)} C_F C_A. \end{split}$$

$$(3.108)$$

For the $\langle R_{j+\tau_c,j}^{(1)q} \rangle$ coefficient, the leading twist $R_{4,2}^{(1)q}$ given in (3.79) also only contains flavor structure with Regge intercept $j_0 = 2$. So from the Lorentzian inversion formula the leading nonzero coefficient starts at $\tau_c = 4$, and it is given by

Similar to the gluon jet case, (3.107), (3.108) and (3.109) agree with (3.79), (B.26) and (B.27) for j > 2, but for the flavor structure $(C_A - 2C_F)C_F$ they also agree at j = 2.

3.4 Higher-order collinear EEEC

In this section, we use the celestial block decomposition (3.47) for the collinear EEEC and the leading order coefficients obtained in section 3.3 to make predictions for higher-order terms in the expansion of the EEEC in the coupling constant. Specifically, we study the (n + 1)-st order expansion of $\mathcal{G}(\zeta_{12}, z, \overline{z})$ in *a*, which we denote by $\mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z})$. Our key physical input is that contributions to (3.47) come from individual light-ray operators, whose contributions are fixed by symmetry in terms of their quantum numbers. In particular, this implies that anomalous dimensions should "exponentiate" to create the power laws predicted by symmetry.

Exponentiation is most powerful when the operators of interest are non-degenerate in perturbation theory. Thus, the $\tau_c = 2, j = 2$ non-degenerate operator (3.74) in QCD will play a crucial role in our arguments. Expanding (3.47) in the weak coupling limit, we find that $\mathcal{G}^{(n+1)}$ contains a term

$$\mathcal{G}^{(n+1)} \supset R^{(1)}_{4,2}(\gamma^{(1)}_{4,2})^n \partial^n_\delta g^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_*}=5)}_{\delta,2}(z,\overline{z})|_{\delta\to4}.$$
(3.110)

Since $R_{4,2}^{(1)}$ is non-degenerate, and the anomalous dimension $\gamma_{4,2}^{(1)}$ is known [73], we can predict the coefficient of this term in $\mathcal{G}^{(n+1)}$ using available perturbative data. Moreover, it turns out that $\partial_{\delta}^{n} g_{\delta,j}^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_{*}}=5)}(z,\overline{z})$ dominates in certain kinematics limits, and thus we have a prediction for the behavior of $\mathcal{G}^{(n+1)}$ in those limits. It is harder to apply the same argument for $\mathcal{G}^{(n+1)}$ in $\mathcal{N} = 4$ SYM due to the fact that all the operators contributing to $\mathcal{G}^{(1)}$ in $\mathcal{N} = 4$ SYM have tree-level degeneracies (see (3.71), (3.72) and (3.73) for the leading τ_c operators). This problem could be circumvented by using higher-order perturbative data to disentangle the degeneracies, or perhaps by organizing the EEEC in $\mathcal{N} = 4$ into an appropriate super-celestial-block expansion. Regardless, we will focus on QCD in this section.

Before proceeding, let us comment on the nonzero β -function of QCD. Note that even in the presence of a nonzero β -function, the celestial block decomposition (3.47), which follows from Lorentz symmetry, should exist. However, some features will be different. Firstly, the spin selection rule J = 3 for operators in the $\mathcal{E} \times \mathcal{E}$ OPE will be violated in the absence of conformal symmetry. We now expect light ray operators with spins $J = 3 + \delta J$ to appear, where contributions proportional to δJ come with additional factors of the β -function. In addition, without conformal symmetry, the quantum numbers of light-ray operators are no longer simply related to quantum numbers of local operators via the rule $(\Delta, J) \rightarrow (1 - J, 1 - \Delta)$. Instead, light-ray operators at null infinity carry so-called "timelike" anomalous dimensions [88, 17].

While these issues are interesting to explore, here we will sidestep them by making predictions for "conformal QCD." Specifically, we work in dimensional regularization $d = 4 - \epsilon$, and tune the coupling constant *a* to a conformal fixed-point $\beta(\epsilon, a) = 0$. The resulting predictions constrain the perturbative expansion of QCD away from the fixed point, up to terms proportional to the β -function. In fact, we expect that terms proportional to β do not affect the central predictions of this section. The reason is that the β -function corrections to the spin selection rule should only affect $R_{\delta,j}^{(n\geq 2)}$ and $\gamma_{\delta,j}^{(n\geq 2)}$, and the difference between spacelike and timelike anomalous dimensions should affect $\gamma_{\delta,j}^{(n\geq 2)}$. When deriving the results in this section, we only use known values of $R_{4,2}^{(1)}$ and $\gamma_{4,2}^{(1)}$. Therefore, they should still be true in the usual 4d QCD.

Let us now give concrete predictions for $\mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z})$ in conformal QCD. We define

$$\widetilde{\mathcal{G}}(z,\overline{z}) = \sum_{\delta,j} R_{\delta,j;\delta'_*} g_{\delta,j}^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_*}=5)}(z,\overline{z}), \qquad (3.111)$$

so the collinear EEEC is

$$\mathcal{G}(\zeta_{12}, z, \overline{z}) = \zeta_{12}^{\frac{\delta'_* - 5}{2}} \widetilde{\mathcal{G}}(z, \overline{z}).$$
(3.112)

We will first derive the leading behavior of $\widetilde{\mathcal{G}}^{(n+1)}(z,\overline{z})$ in three different kinematic limits, and then study the behavior of $\mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z})$. Our main predictions for $\mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z})$ are given by (3.143), (3.144), (3.147), and (3.148).

3.4.1 Predictions for $\widetilde{\mathcal{G}}^{(n+1)}(z,\overline{z})$

To make predictions for $\widetilde{\mathcal{G}}^{(n+1)}(z,\overline{z})$, we must take a kinematic limit where the $\tau_c = 2, j = 2$ non-degenerate operator gives the leading contribution. The first limit we consider is the OPE limit/squeezed limit, where $z, \overline{z} \ll 1$ with z/\overline{z} fixed. We parametrize z and \overline{z} as $z = re^{i\theta}, \overline{z} = re^{-i\theta}$. Using (3.111), we can fix the leading log term of the (n + 1)-th order expansion of $\widetilde{\mathcal{G}}$. This is because $\log(r)$ can only come from a derivative of $g_{\delta,j}$ with respect to δ , and each derivative introduces an anomalous dimension factor $\gamma^{(1)}$. Hence, the leading log term, which has the most powers of $\log(r)$, should take the form $\langle R^{(1)}(\gamma^{(1)})^n \rangle \log^n(r) \partial_{\delta}^n g_{\delta,j}$. The quantum numbers with the lowest value of δ and nonzero $R^{(1)}$ are $(\delta, j) = (4, 0)$ and $(\delta, j) = (4, 2)$. Plugging these in, we obtain

$$\widetilde{\mathcal{G}}^{(n+1)}(r \to 0, \theta) = \frac{r^4 \log^n(r)}{n!} \left(\langle R_{4,0}^{(1)}(\gamma_{4,0}^{(1)})^n \rangle + 2R_{4,2}^{(1)}(\gamma_{4,2}^{(1)})^n \cos(2\theta) \right) + O(r^4 \log^{n-1}(r)). \quad (3.113)$$

As discussed in 3.3.1, the light-ray OPE formula implies that $\gamma_{4,2}^{(1)}$ should be the anomalous dimension of the operator (3.74) evaluated at J = 3. This has been calculated in e.g. [89, 73], and it is given by

$$\gamma_{F\partial^{J-2}F}^{(1)}(J) = 4C_A S_1(J) - \beta_0, \qquad (3.114)$$

where $\beta_0 = \frac{11}{3}C_A - \frac{4}{3}n_FT_F$. At J = 3, we then have

$$\gamma_{4,2}^{(1)} = \gamma_{F\partial^{J-2}F}^{(1)}(3) = \frac{22}{3}C_A - \left(\frac{11}{3}C_A - \frac{4}{3}n_FT_F\right) = \frac{11}{3}C_A + \frac{4}{3}n_FT_F.$$
 (3.115)

Thus, the leading log term of $\widetilde{\mathcal{G}}^{(n+1)}$ in the OPE limit should be

$$\widetilde{\mathcal{G}}^{(n+1)g/q}(r \to 0, \theta) = \langle R_{4,0}^{(1)g/q}(\gamma_{4,0}^{(1)})^n \rangle \frac{r^4 \log^n(r)}{n!} + 2R_{4,2}^{(1)g/q} \left(\frac{11}{3}C_A + \frac{4}{3}n_F T_F\right)^n \frac{r^4 \log^n(r)}{n!} \cos(2\theta) + O(r^4 \log^{n-1}(r)), \qquad (3.116)$$

where the superscript g/q denotes gluon jet or quark jet, and $R_{4,2}^{(1)g/q}$ are given in (3.78) and (3.79).

We see that the coefficient of the leading term $r^4 \log^n(r) \cos(2\theta)$ in the OPE limit is completely fixed. In fact, since the next quantum number with nonzero $R^{(1)}$ starts at $\delta = 6$, we can also predict the term proportional to $r^5 \log^n(r) \cos(3\theta)$ since it should come from the descendant of the $g_{\delta=4, i=2}$ block. The result is

$$\begin{split} \widetilde{\mathcal{G}}^{(n+1)g/q}(r \to 0, \theta) &= \langle R_{4,0}^{(1)g/q}(\gamma_{4,0}^{(1)})^n \rangle \frac{r^4 \log^n(r)}{n!} \\ &+ 2R_{4,2}^{(1)g/q} \left(\frac{11}{3}C_A + \frac{4}{3}n_F T_F\right)^n \frac{r^4 \log^n(r)}{n!} \cos(2\theta) \\ &+ r^4(\cdots) \\ &+ \langle R_{4,0}^{(1)g/q}(\gamma_{4,0}^{(1)})^n \rangle \frac{r^5 \log^n(r)}{n!} \cos(\theta) \\ &+ 2R_{4,2}^{(1)g/q} \left(\frac{11}{3}C_A + \frac{4}{3}n_F T_F\right)^n \frac{r^5 \log^n(r)}{n!} \cos(3\theta) \\ &+ O(r^5 \log^{n-1}(r)). \end{split}$$
(3.117)

To study other interesting limits where the non-degenerate operator gives the leading contribution, we can go to Lorentzian signature on the celestial sphere (where z and \overline{z} are independent real variables) and consider the $z \ll 1$, fixed \overline{z} limit. From the collider physics point of view, this limit might not be so useful since the kinematic region one can explore using collider experiments is intrinsically Euclidean (where z and \overline{z} are complex conjugates of each other). Nevertheless, we still find a nontrivial constraint on the analytic expression of $\widetilde{\mathcal{G}}^{(n+1)}(z,\overline{z})$. In the limit $z \ll 1$, we can use the SL(2, \mathbb{R}) expansion for the conformal block (3.96), and the leading term is from the operator with the leading celestial twist τ_c , which is exactly the non-degenerate operator with $\tau_c = 2$, j = 2. Using this, we find that the leading log of $\widetilde{\mathcal{G}}^{(n+1)}$ in the limit $z \ll 1$ is given by

$$\begin{split} \widetilde{\mathcal{G}}^{(n+1)g/q}(z \ll 1, \overline{z}) \\ &= R_{4,2}^{(1)g/q} (\gamma_{4,2}^{(1)})^n \frac{z \log^n(z)}{2^n n!} k_6^{0,-1}(\overline{z}) + O(z \log^{n-1} z) \\ &= 10 R_{4,2}^{(1)g/q} \left(\frac{11}{3} C_A + \frac{4}{3} n_F T_F \right)^n \frac{z \log^n(z)}{2^n n!} \frac{(\overline{z}(12 - 12\overline{z} + \overline{z}^2) + 6(2 - 3\overline{z} + \overline{z}^2) \log(1 - \overline{z}))}{\overline{z}^2} \\ &+ O(z \log^{n-1} z). \end{split}$$
(3.118)

If we also write down the subleading $O(z \log^{n-1} z)$ order, we find

$$\begin{split} \tilde{\mathcal{G}}^{(n+1)g/q}(z \ll 1, \overline{z}) \\ &= R_{4,2}^{(1)g/q}(\gamma_{4,2}^{(1)})^n \frac{z \log^n(z)}{2^n n!} k_6^{0,-1}(\overline{z}) \\ &+ \left(\sum_{j=2,4,6,\cdots} \langle R_{j+2,j}^{(2)g/q}(\gamma_{j+2,j}^{(1)})^{n-1} \rangle k_{2j+2}^{0,-1}(\overline{z}) + R_{4,2}^{(1)g/q}(\gamma_{4,2}^{(1)})^n \partial_{\delta} k_{\delta+2}^{0,-1}(\overline{z}) |_{\delta \to 4} \\ &+ \langle R_{j+2,2}^{(1)g/q} \gamma_*' \rangle (\gamma_{j+2,2}^{(1)})^{n-1} \partial_{\delta_*'} k_6^{0,\frac{3-\delta_*'}{2}}(\overline{z}) |_{\delta_*' \to 5} \right) \frac{z \log^{n-1}(z)}{2^{n-1}(n-1)!} + O(z \log^{n-2} z). \end{split}$$

$$(3.119)$$

Note that for the subleading order coefficient $\langle R^{(2)} \rangle$, we should expect to get contribution from operators with $\tau_c = 2$ and all even j. The only term we know is $R_{4,2}^{(1)g/q}(\gamma_{4,2}^{(1)})^n \partial_{\delta} k_{\delta+2}^{0,-1}(\overline{z})|_{\delta \to 4}$. We can actually isolate this term by further taking the small \overline{z} limit, in which we find $k_{2j+2}^{0,-1}(\overline{z}) = \overline{z}^{j+1} + O(\overline{z}^{j+2})$, $\partial_{\delta} k_{\delta+2}^{0,-1}(\overline{z})|_{\delta \to 4} = \frac{1}{2}\overline{z}^3 \log \overline{z} + O(\overline{z}^4)$, and $\partial_{\delta'_*} k_6^{0,\frac{3-\delta'_*}{2}}(\overline{z})|_{\delta'_*\to 5} = O(\overline{z}^4)$. Therefore, in the $z \ll \overline{z} \ll 1$ limit, we can also predict the coefficient of the $z \log^{n-1} z \overline{z}^3 \log \overline{z}$ term. It is given by

$$\begin{split} \widetilde{\mathcal{G}}^{(n+1)g/q}(z \ll \overline{z} \ll 1) \\ &= 10R_{4,2}^{(1)g/q} \left(\frac{11}{3}C_A + \frac{4}{3}n_F T_F\right)^n \frac{z \log^n(z)}{2^n n!} \frac{(\overline{z}(12 - 12\overline{z} + \overline{z}^2) + 6(2 - 3\overline{z} + \overline{z}^2) \log(1 - \overline{z}))}{\overline{z}^2} \\ &+ \frac{1}{2}R_{4,2}^{(1)g/q} \left(\frac{11}{3}C_A + \frac{4}{3}n_F T_F\right)^n \frac{z \log^{n-1}(z)}{2^{n-1}(n-1)!} \left(\overline{z}^3 \log \overline{z} + O(\overline{z}^3)\right) \\ &+ O(z \log^{n-2} z). \end{split}$$
(3.120)

It is also interesting to study the leading behavior of $\widetilde{\mathcal{G}}^{(n+1)}(z,\overline{z})$ in the double lightcone limit $z \ll 1 - \overline{z} \ll 1$. From (3.118), we obtain

$$\widetilde{\mathcal{G}}^{(n+1)g/q}(z \ll 1 - \overline{z} \ll 1) = R_{4,2}^{(1)g/q}(\gamma_{4,2}^{(1)})^n \frac{z \log^n(z)}{2^n n!} (10 + O(1 - \overline{z})) + O(z \log^{n-1} z),$$
(3.121)

and one can try to study how this term can be created using the crossing equation (3.48) and the lightcone bootstrap [81]. One will find that they come from the $R^{(n+1)}$ coefficients with $\tau_c = 6$ and large-*j* (which corresponds to double-trace operators with conventional twist $\tau = 8$ and j = 4). From the crossing equation, we can also predict the large-*j* behavior of $\langle R_{j+6,j}^{(n+1)} \rangle$. For example, we find that at large-*j*,

$$\langle R_{j+6,j}^{(1)g/q} \rangle \sim \frac{5\sqrt{\pi}}{8} R_{4,2}^{(1)g/q} 4^{-j} j^{\frac{7}{2}}.$$
 (3.122)

This result can be generalized to $\langle R_{j+6,j}^{(n+1)g/q} \rangle$ at any order. We describe the result and the details of the calculation in appendix B.4.

3.4.2 Degeneracies of O'_*

So far in our analysis, we have assumed there is a unique isolated spin-4 operator O'_{*} that controls the collinear limit. However, it can happen that the leading-twist spin-4 operator is degenerate at tree-level, and thus we must take into account the contribution of multiple O'_{*} 's in perturbation theory.

As discussed in section 3.3.1, $\Delta'_* = \delta'_* + 1$ should be the scaling dimension of the leading twist, spin-4 operator O'_* in the $T \times T \times T$ OPE. Also, since the sink/source states we consider are rotationally-invariant, O'_* should have zero transverse spin. There are only two such Regge trajectories in QCD:¹⁶

$$O_{q}(J) = \frac{(-1)^{J}}{2^{J}} \sum_{k=1}^{J} (-1)^{k-1} \alpha_{k}^{q}(J) \overline{\psi} \gamma^{\{\mu_{1}}(i\overleftarrow{D}^{\mu_{2}}) \cdots (i\overleftarrow{D}^{\mu_{k}})(iD^{\mu_{k+1}}) \cdots (iD^{\mu_{J}})\psi - \text{traces},$$

$$O_{g}(J) = \frac{(-1)^{J}}{2^{J}} \sum_{k=1}^{J-1} (-1)^{k-1} \alpha_{k}^{g}(J) F_{av}^{\{\mu_{1}}(i\overleftarrow{D}^{\mu_{2}}) \cdots (i\overleftarrow{D}^{\mu_{k}})(iD^{\mu_{k+1}}) \cdots (iD^{\mu_{J-1}}) F_{a}^{\nu\mu_{J}} - \text{traces}.$$
(3.123)

where \overleftarrow{D}^{μ} (acting on the left) and D^{μ} (acting on the right) are covariant derivatives. The coefficients $\alpha_k^{q/g}(J)$ are chosen such that the entire expression of $O_{q/g}(J)$ is a conformal primary, and they satisfy the normalization condition $\sum_{k=1}^{J} \alpha_k^q(J) =$ $\sum_{k=1}^{J-1} \alpha_k^g(J) = 1$. For example, for J = 2 we have

$$O_q(2) = \frac{1}{8} \left(\overline{\psi} \gamma^{\{\mu_1 i D^{\mu_2}\}} \psi - \overline{\psi} \gamma^{\mu_1\}} i \overleftarrow{D}^{\{\mu_2} \psi \right) - \text{traces},$$

$$O_g(2) = \frac{1}{4} F_{a\nu}{}^{\mu_1} F^{a\nu\mu_2} - \text{traces}.$$
(3.124)

One can see that the sum $O_q(2) + O_g(2)$ is proportional to the stress-energy tensor.

For general *J*, there are many different methods to determine the coefficients $\alpha_k^{q/g}(J)$ that make $O_{q/g}(J)$ a conformal primary [72, 90, 91, 92]. Here, we are only interested in the J = 4 case. So we take a simple approach: apply the generator K_{μ} of the special conformal transformation on $O_{q/g}(4)$ and demand that its action vanishes.

¹⁶We are using mostly positive metric, and our convention for indices symmetrization is $T^{\{\mu_1 \cdots \mu_n\}} = \frac{1}{n!} \sum_{P \in S_n} T^{\mu_{P(1)} \cdots \mu_{P(n)}}$

Using the basic commutation relation $[K_{\mu}, P_{\nu}] = 2\eta_{\mu\nu}D - 2M_{\mu\nu}$, we find

$$O_{q}(4) = \frac{1}{224} \left(\overline{\psi} \gamma^{\{\mu_{1}}(iD^{\mu_{2}})(iD^{\mu_{3}})(iD^{\mu_{4}}\})\psi - 6(iD^{\{\mu_{2}})\overline{\psi}\gamma^{\mu_{1}}(iD^{\mu_{3}})(iD^{\mu_{4}})\psi \right) + 6(iD^{\{\mu_{2}\}}(iD^{\mu_{3}})\overline{\psi}\gamma^{\mu_{1}}(iD^{\mu_{4}})\psi - (iD^{\{\mu_{2}\}})(iD^{\mu_{3}})(iD^{\mu_{4}})\overline{\psi}\gamma^{\mu_{1}}\psi \right) - \text{traces},$$
(3.125)

and

$$O_g(4) = \frac{3}{112} \left(F_{a\nu}^{\{\mu_1}(iD^{\mu_2})(iD^{\mu_3})F_a^{\nu\mu_4\}} - \frac{4}{3}(iD^{\{\mu_2\}})F_{a\nu}^{\mu_1}(iD^{\mu_3})F_a^{\nu\mu_4\}} \right) - \text{traces.}$$
(3.126)

The one-loop dilatation matrix for $O_q(4)$ and $O_g(4)$ is given by [93, 94, 95]

$$\widehat{\gamma} = \frac{\alpha_s}{4\pi} \begin{pmatrix} \frac{157C_F}{30} & -\frac{11n_FT_F}{15} \\ -\frac{11C_F}{15} & \frac{21C_A}{5} + \frac{4n_FT_F}{3} \end{pmatrix}.$$
(3.127)

Its eigenvalues are

$$\frac{21}{10}C_A + \frac{157}{60}C_F + \frac{2}{3}n_F T_F \mp \frac{1}{60}\sqrt{(126C_A - 157C_F)^2 + 32(315C_A - 332C_F)n_F T_F + 1600n_F^2 T_F^2)}$$
(3.128)

and its left eigenvectors are

$$\begin{pmatrix} \alpha_{\pm} & 1 \end{pmatrix}, \tag{3.129}$$

where

$$\alpha_{\pm} = \frac{126C_A - 157C_F + 40n_F T_F \pm \sqrt{(126C_A - 157C_F)^2 + 32(315C_A - 332C_F)n_F T_F + 1600n_F^2 T_F^2}}{44n_F T_F}.$$
(3.130)

To resolve the degeneracy at 1-loop, we can define the following two operators:

$$O'_{*1} = \alpha_+ O_q(4) + O_g(4),$$

$$O'_{*2} = O_q(4) + \frac{1}{\alpha_-} O_g(4).$$
(3.131)

In the O'_{*1}, O'_{*2} basis, the anomalous dimension matrix is then diagonal. The first operator has anomalous dimension (suppressing the $\frac{\alpha_s}{4\pi}$ factor)

$$\gamma_{*1}' \equiv \frac{21}{10}C_A + \frac{157}{60}C_F + \frac{2}{3}n_F T_F - \frac{1}{60}\sqrt{(126C_A - 157C_F)^2 + 32(315C_A - 332C_F)n_F T_F + 1600n_F^2 T_F^2}$$
(3.132)

and γ'_{*2} is given by replacing $-\sqrt{\cdot} \rightarrow \sqrt{\cdot}$.

Taking into account the degeneracies of O'_* , the decomposition (3.47) for $\mathcal{G}(\zeta_{12}, z, \overline{z})$ should become

$$\mathcal{G}(\zeta_{12}, z, \overline{z}) = \frac{\pi^2}{16} \sum_{i=1,2} \frac{(-p^2)^{\frac{2\Delta_O - 4}{2}}}{\sigma_{\text{total}}^O} s_{OOP_{\delta'_*}}^{\prime(i)} \zeta_{12}^{\frac{\delta'_* - 5}{2}} \sum_{\delta,j} r_{\mathcal{E}\mathcal{E}\mathcal{P}_{\delta,j}} r_{\mathcal{P}_{\delta,j}\mathcal{E}\mathcal{P}_{\delta'_*}}^{\prime(i)} g_{\delta,j}^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_*})}(z,\overline{z})$$
$$= \sum_{i=1,2} c_i^O \zeta_{12}^{\frac{\delta'_* - 5}{2}} \sum_{\delta,j} \widetilde{R}_{\delta,j;\delta'_*;i} g_{\delta,j}^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_*})}(z,\overline{z}), \qquad (3.133)$$

where in the first line we have used (3.45) to restore the total cross section in order to emphasize the dependence on the sink/source states, and in the second line we have defined

$$c_{i}^{O} \equiv \frac{(-p^{2})^{\frac{2\Delta_{O}-4}{2}}}{\sigma_{\text{total}}^{O}} s_{OO\mathcal{P}_{\delta'_{*}}}^{\prime(i)}, \qquad \widetilde{R}_{\delta,j;\delta'_{*};i} \equiv \frac{\pi^{2}}{16} r_{\mathcal{E}\mathcal{E}\mathcal{P}_{\delta,j}} r_{\mathcal{P}_{\delta,j}\mathcal{E}\mathcal{P}_{\delta'_{*}}}^{\prime(i)}.$$
(3.134)

Note that unlike $R_{\delta,j;\delta'_*}$, the newly defined coefficient $\widetilde{R}_{\delta,j;\delta'_*;i}$ is independent of the operator O, which creates the sink/source states. The only dependence on O is in the coefficient c_i^O . Using (3.16), one can show that c_i^O can be determined using

$$\frac{\langle O(p)|\mathbf{L}[O'_{*;i}](\infty,z)|O(p)\rangle}{\langle O(p)|O(p)\rangle} = c_i^O(-2p\cdot z)^{-\delta'_*}(-p^2)^{\frac{\delta'_*+3}{2}},$$
(3.135)

where we have replaced the object $\mathbb{W}_{\delta'}$ with $\mathbf{L}[O'_{*,i}]^{17}$ based on the assumptions made in section 3.3.1. For us, the two degenerate operators can be chosen to be O'_{*1} and O'_{*2} defined in (3.131). The two different O's are $\mathrm{Tr}F^2$ corresponding to the gluon jet and J^{μ} corresponding to the quark jet (averaged over polarization), so we will simply use $c_i^{g/q}$ for two cases.

We are interested in $c_i^{(0)g/q}$ at the leading order in the coupling constant. To find $c_i^{(0)g/q}$, we first determine $\langle O(x_1)O_{g/q}(4)(x_2,z_2)O(x_3)\rangle$ in position space using Wick contractions, where O is either $\text{Tr}F^2$ or J^{μ} . After doing the light transform and Fourier transform, we then obtain that at the leading order

$$\frac{\langle \operatorname{Tr} F^{2}(p) | \mathbf{L}[O_{g}(4)](\infty, z) | \operatorname{Tr} F^{2}(p) \rangle}{\langle \operatorname{Tr} F^{2}(p) | \operatorname{Tr} F^{2}(p) \rangle} = \frac{1}{64\pi} (-2p \cdot z)^{-5} (-p^{2})^{4} + O(\alpha_{s}),$$
$$\frac{\langle J_{\mu}(p) | \mathbf{L}[O_{q}(4)](\infty, z) | J^{\mu}(p) \rangle}{\langle J_{\mu}(p) | J^{\mu}(p) \rangle} = \frac{1}{64\pi} (-2p \cdot z)^{-5} (-p^{2})^{4} + O(\alpha_{s}), \quad (3.136)$$

¹⁷The coefficient relating $\mathbb{W}_{\delta'}$ and $\mathbf{L}[O'_{*,i}]$ can be absorbed into $r'^{(i)}_{\mathcal{P}_{\delta,i}\mathcal{EP}_{\delta'}}$.
which implies that

$$c_1^{(0)g} = \frac{1}{64\pi}, \qquad c_2^{(0)g} = \frac{1}{\alpha_-} \frac{1}{64\pi}, \qquad c_1^{(0)q} = \alpha_+ \frac{1}{64\pi}, \qquad c_2^{(0)q} = \frac{1}{64\pi}.$$
 (3.137)

Therefore, the coefficients appearing in (3.64) can be rewritten as (we focus on $\langle R_{\delta,j;\delta'_*}^{(1)} \rangle$)

$$\langle R_{\delta,j;\delta'_{*}}^{(1)g} \rangle = \frac{1}{64\pi} \left(\langle \widetilde{R}_{\delta,j;\delta'_{*};1}^{(1)} \rangle + \frac{1}{\alpha_{-}} \langle \widetilde{R}_{\delta,j;\delta'_{*};2}^{(1)} \rangle \right), \langle R_{\delta,j;\delta'_{*}}^{(1)q} \rangle = \frac{1}{64\pi} \left(\alpha_{+} \langle \widetilde{R}_{\delta,j;\delta'_{*};1}^{(1)} \rangle + \langle \widetilde{R}_{\delta,j;\delta'_{*};2}^{(1)} \rangle \right).$$
(3.138)

Solving (3.138) for $\langle \widetilde{R}^{(1)}_{\delta,j;\delta'_*;1/2} \rangle$, we find

$$\langle \widetilde{R}_{\delta,j;\delta'_{*};1}^{(1)} \rangle = \frac{64\pi}{\alpha_{+} - \alpha_{-}} \left(\langle R_{\delta,j;\delta'_{*}}^{(1)q} \rangle - \alpha_{-} \langle R_{\delta,j;\delta'_{*}}^{(1)g} \rangle \right),$$

$$\langle \widetilde{R}_{\delta,j;\delta'_{*};2}^{(1)} \rangle = \frac{64\pi\alpha_{-}}{\alpha_{+} - \alpha_{-}} \left(\alpha_{+} \langle R_{\delta,j;\delta'_{*}}^{(1)g} \rangle - \langle R_{\delta,j;\delta'_{*}}^{(1)q} \rangle \right).$$

$$(3.139)$$

3.4.3 Predictions for $\mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z})$

We now explain how to use (3.133) and (3.139) to deal with the degeneracy of O'_* and make predictions for $\mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z})$. If we expand (3.112) in small coupling assuming there are no degeneracies, we will get

$$\mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z}) = \sum_{k=0}^{n} \frac{\log^{n-k} \zeta_{12}}{2^{n-k}(n-k)!} \left(\sum_{p=0}^{k} \sum_{\substack{i_1+i_2+\cdots+i_{p+1}=n-k\\i_1+2i_2+\cdots+(p+1)i_{p+1}=n-k+p}} \frac{(n-k)!}{i_1!\cdots i_{p+1}!} \left(\gamma_*^{\prime(1)} \right)^{i_1} \cdots \left(\gamma_*^{\prime(p+1)} \right)^{i_{p+1}} \widetilde{\mathcal{G}}^{(k+1-p)} \right)$$

$$(3.140)$$

More explicitly, the first few terms are given by

$$\begin{aligned} \mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z}) \\ &= \frac{\log^{n} \zeta_{12}}{2^{n} n!} \left(\gamma_{*}^{\prime(1)} \right)^{n} \widetilde{\mathcal{G}}^{(1)}(z, \overline{z}) \\ &+ \frac{\log^{n-1} \zeta_{12}}{2^{n-1} (n-1)!} \left(\left(\gamma_{*}^{\prime(1)} \right)^{n-1} \widetilde{\mathcal{G}}^{(2)}(z, \overline{z}) + (n-1) \left(\gamma_{*}^{\prime(1)} \right)^{n-2} \gamma_{*}^{\prime(2)} \widetilde{\mathcal{G}}^{(1)}(z, \overline{z}) \right) \\ &+ \frac{\log^{n-2} \zeta_{12}}{2^{n-2} (n-2)!} \left(\left(\gamma_{*}^{\prime(1)} \right)^{n-2} \widetilde{\mathcal{G}}^{(3)}(z, \overline{z}) + (n-2) \left(\gamma_{*}^{\prime(1)} \right)^{n-3} \gamma_{*}^{\prime(2)} \widetilde{\mathcal{G}}^{(2)}(z, \overline{z}) \\ &+ \frac{(n-2)(n-3)}{2} \left(\gamma_{*}^{\prime(1)} \right)^{n-4} \left(\gamma_{*}^{\prime(2)} \right)^{2} \widetilde{\mathcal{G}}^{(1)}(z, \overline{z}) + (n-2) \left(\gamma_{*}^{\prime(1)} \right)^{n-3} \gamma_{*}^{\prime(3)} \widetilde{\mathcal{G}}^{(1)}(z, \overline{z}) \end{aligned}$$

The above expression will be modified in the presence of degeneracies. In general, if we know the anomalous dimension $\gamma'^{(k)}_*$ to the *k*-th order, we will be able to rewrite the above expression up to the $\log^{n-k+1} \zeta_{12}$ term. In the previous section, we have only diagonalized the $\gamma'^{(1)}_*$ matrix, which then allows us to rewrite all the $(\gamma'^{(1)}_*)^k$ terms.

First, let us focus on the leading logarithmic divergence $\log^n(\zeta_{12})$. Using (3.133), we find that in the presence of degeneracies, the leading log term of (3.140) should become

$$\begin{aligned} \mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z}) \\ &= \frac{\log^{n} \zeta_{12}}{2^{n} n!} \left(c_{1}^{(0)O}\left(\gamma_{*1}'\right)^{n} \sum_{\delta, j} \langle \widetilde{R}_{\delta, j; \delta_{*}'; 1}^{(1)} \rangle g_{\delta, j}^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta_{*}'=5})}(z, \overline{z}) + c_{2}^{(0)O}\left(\gamma_{*2}'\right)^{n} \sum_{\delta, j} \langle \widetilde{R}_{\delta, j; \delta_{*}'; 2}^{(1)} \rangle g_{\delta, j}^{(\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta_{*}'=5})}(z, \overline{z}) \right) \\ &+ O(\log^{n-1} \zeta_{12}) \,, \end{aligned}$$
(3.142)

where γ'_{*1} and γ'_{*2} are given by (3.132). Plugging in (3.139), we obtain for the gluon and quark jets

$$\mathcal{G}^{(n+1)g}(\zeta_{12}, z, \overline{z}) = \frac{\log^{n} \zeta_{12}}{2^{n} n!} \left(\frac{\alpha_{+} (\gamma_{*2}')^{n} - \alpha_{-} (\gamma_{*1}')^{n}}{\alpha_{+} - \alpha_{-}} \mathcal{G}^{(1)g} + \frac{(\gamma_{*1}')^{n} - (\gamma_{*2}')^{n}}{\alpha_{+} - \alpha_{-}} \mathcal{G}^{(1)q} \right) + O(\log^{n-1} \zeta_{12}), \qquad (3.143)$$
$$\mathcal{G}^{(n+1)q}(\zeta_{12}, z, \overline{z}) = \frac{\log^{n} \zeta_{12}}{2^{n} n!} \left(\frac{\alpha_{+} (\gamma_{*1}')^{n} - \alpha_{-} (\gamma_{*2}')^{n}}{\alpha_{+} - \alpha_{-}} \mathcal{G}^{(1)q} + \frac{\alpha_{+} \alpha_{-} ((\gamma_{*2}')^{n} - (\gamma_{*1}')^{n})}{\alpha_{+} - \alpha_{-}} \mathcal{G}^{(1)g} \right) + O(\log^{n-1} \zeta_{12}), \qquad (3.144)$$

where $\mathcal{G}^{(1)g/q}$ are simply the leading order results for the gluon/quark jet, related to the known result of [55] by (3.62). Equations (3.143) and (3.144) show that the leading logarithmic divergence of the (n + 1)-th order EEEC $\mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z})$ is $\log^n(\zeta_{12})$, and it is completely determined by the leading order result $\mathcal{G}^{(1)g}$ and $\mathcal{G}^{(1)q}$.

In fact, since we can rewrite all the $(\gamma_*^{(1)})^k$ terms in (3.140), we can make further predictions for $\mathcal{G}^{(n+1)}(\zeta_{12}, z, \overline{z})$ if we know how to separate out the contributions from $(\gamma_*^{(1)})^k$. This can be done by taking the limits considered in section 3.4.1. For

example, in the OPE limit we find

$$\begin{aligned} \mathcal{G}^{(n+1)g/q}(\zeta_{12}, r \to 0, \theta) \\ &= \sum_{k=0}^{n} \frac{\log^{n-k} \zeta_{12}}{2^{n-k}(n-k)!} \Big(c_{1}^{(0)g/q}(\gamma_{*1}')^{n-k} \widetilde{\mathcal{G}}_{1}^{(k+1)}(r \to 0, \theta) + c_{2}^{(0)g/q}(\gamma_{*2}')^{n-k} \widetilde{\mathcal{G}}_{2}^{(k+1)}(r \to 0, \theta) \\ &+ O(r^{4} \log^{k-1} r) \Big). \end{aligned}$$
(3.145)

From (3.116) we know that $\widetilde{\mathcal{G}}^{(k+1)} \sim r^4 \log^k r$ in the OPE limit. Thus, at each $\log^{n-k} \zeta_{12}$ power, all the terms involving higher-loop anomalous dimensions $(\gamma'^{(p)}_*)$ with p > 1) in (3.140) go as at most $r^4 \log^{k-1} r$, while the term involving $(\gamma'^{(1)}_*)^k$ has the most divergent piece $r^4 \log^k r$. Therefore, we can determine the leading behavior in small r for each $\log^{n-k} \zeta_{12}$ power. The functions $\widetilde{\mathcal{G}}_1^{(k+1)}$ and $\widetilde{\mathcal{G}}_2^{(k+1)}$ only include the contribution from \mathcal{O}'_{*1} and \mathcal{O}'_{*2} respectively. They are defined as

$$\widetilde{\mathcal{G}}_{1}^{(k+1)} = \frac{64\pi}{\alpha_{+} - \alpha_{-}} \left(\widetilde{\mathcal{G}}^{(k+1)q} - \alpha_{-} \widetilde{\mathcal{G}}^{(k+1)g} \right),$$

$$\widetilde{\mathcal{G}}_{2}^{(k+1)} = \frac{64\pi\alpha_{-}}{\alpha_{+} - \alpha_{-}} \left(\alpha_{+} \widetilde{\mathcal{G}}^{(k+1)g} - \widetilde{\mathcal{G}}^{(k+1)q} \right).$$
(3.146)

Using (3.116), we then obtain, for example,

$$\begin{aligned} \mathcal{G}^{(n+1)g}(\zeta_{12}, r \to 0, \theta) \\ &= \sum_{k=0}^{n} \frac{\log^{n-k} \zeta_{12}}{2^{n-k-1}k!(n-k)!} \times \\ \left(\left(\gamma_{4,2}^{(1)} \right)^{k} \left(\frac{\alpha_{+} \left(\gamma_{*2}^{\prime} \right)^{n-k} - \alpha_{-} \left(\gamma_{*1}^{\prime} \right)^{n-k}}{\alpha_{+} - \alpha_{-}} R_{4,2}^{(1)g} + \frac{\left(\gamma_{*1}^{\prime} \right)^{n-k} - \left(\gamma_{*2}^{\prime} \right)^{n-k}}{\alpha_{+} - \alpha_{-}} R_{4,2}^{(1)q} \right) r^{4} \log^{k} r \cos(2\theta) \\ &+ (\cdots) r^{4} \log^{k} r + O(r^{4} \log^{k-1} r) \right), \end{aligned}$$

$$(3.147)$$

where $\gamma_{4,2}^{(1)}$ and $R_{4,2}^{(1)g/q}$ are given in (3.115), (3.78) and (3.79) respectively. Thus, we have a prediction for the spin-2 part of the leading term of $\mathcal{G}^{(n+1)}$ in the OPE limit, at each logarithmic order in log ζ_{12} .

Similarly, $(\gamma'^{(1)}_*)^k$ terms in (3.140) also give the dominant contribution in the $z \ll 1$ limit. Following the same calculation as that of the OPE limit, we obtain (using

$$(3.118))$$

$$\mathcal{G}^{(n+1)g}(\zeta_{12}, z \ll 1, \overline{z})$$

$$= \sum_{k=0}^{n} \frac{\log^{n-k} \zeta_{12}}{2^{n}k!(n-k)!} \times \left(\left(\gamma_{4,2}^{(1)}\right)^{k} \left(\frac{\alpha_{+} \left(\gamma_{*2}^{\prime}\right)^{n-k} - \alpha_{-} \left(\gamma_{*1}^{\prime}\right)^{n-k}}{\alpha_{+} - \alpha_{-}} R_{4,2}^{(1)g} + \frac{\left(\gamma_{*1}^{\prime}\right)^{n-k} - \left(\gamma_{*2}^{\prime}\right)^{n-k}}{\alpha_{+} - \alpha_{-}} R_{4,2}^{(1)q} \right) z \log^{k}(z) k_{6}^{0,-1}(\overline{z})$$

$$+ O(z \log^{k-1}(z)) \right), \qquad (3.148)$$

and similarly for $\mathcal{G}^{(n+1)q}$. It is also straightforward to repeat the analysis in the $z \ll \overline{z} \ll 1$ limit and predict the $z \log^{k-1}(z)\overline{z}^3 \log(\overline{z})$ term at each $\log \zeta_{12}$ order using (3.120).

3.5 Contact terms and Ward identities

In this section, we study Ward identities satisfied by the EEEC, and also compute the EEEC at $O(g^0)$ and $O(g^2)$ order. Note that conventionally $O(g^4)$ is called the "leading order" (LO) since it is the lowest order at which the EEEC is nonzero for generic detector positions. However, contributions at $O(g^0)$ and $O(g^2)$ also exist: they are proportional to delta functions, which we call "contact terms", and so only become nonzero in special configurations. It was shown in [17, 15, 16] that Ward identities can be used to determine the contact terms in the two-point EEC. Here, we perform a similar analysis for the EEEC, and use perturbation theory and Ward identities to obtain the EEEC at $O(g^0)$ and $O(g^2)$ order.

3.5.1 EEEC' and Ward identities

To study Ward identities, it is more convenient to write the EEEC as a function of the explicit positions on the celestial sphere \vec{n}_i , instead of the cross-ratios ζ_{ij} . Thus, we define

$$\begin{aligned} \text{EEEC}'(\vec{n}_1, \vec{n}_2, \vec{n}_3) &= \frac{\int d^d x \ e^{ip \cdot x} \langle 0| \mathcal{O}^{\dagger}(x) \mathcal{E}(\vec{n}_1) \mathcal{E}(\vec{n}_2) \mathcal{E}(\vec{n}_3) \mathcal{O}(0) |0\rangle}{(-p^2)^{\frac{3}{2}} \int d^d x \ e^{ip \cdot x} \langle 0| \mathcal{O}^{\dagger}(x) \mathcal{O}(0) |0\rangle} \\ &= \sum_{i,j,k} \int d\sigma \frac{E_i E_j E_k}{Q^3} \delta\left(\vec{n}_1, \frac{\vec{p}_i}{E_i}\right) \delta\left(\vec{n}_2, \frac{\vec{p}_j}{E_j}\right) \delta\left(\vec{n}_3, \frac{\vec{p}_k}{E_k}\right), \end{aligned}$$
(3.149)

where the spherical delta function $\delta(\vec{n}_1, \vec{n}_2)$ is defined by

$$\int d\Omega_{\vec{n}_2} \delta(\vec{n}_1, \vec{n}_2) f(\vec{n}_2) = f(\vec{n}_1).$$
(3.150)

The first line in (3.149) is a nonperturbative definition, while the second line is suitable for perturbation theory. As before, $d\sigma$ represents an integration over phase space weighted by the scattering cross section, and $(E_i, \vec{p_i})$ are energy and momentum of outgoing particles. The parametrization (3.149) of the EEEC is related to (3.2) by

EEEC'
$$(\vec{n}_1, \vec{n}_2, \vec{n}_3) = \frac{\sin \theta_1 \sin \theta_2 |\sin \phi|}{64\pi^2}$$
EEEC $(\zeta_{12}, \zeta_{13}, \zeta_{23}),$ (3.151)

where

$$\vec{n}_1 = (\sin \theta_1, 0, \cos \theta_1), \quad \vec{n}_2 = (\sin \theta_2 \cos \phi, \sin \theta_2 \sin \phi, \cos \theta_2), \quad \vec{n}_3 = (0, 0, 1).$$
(3.152)

To derive Ward identities, we simply integrate (3.149) over $\vec{n}_1, \vec{n}_2, \vec{n}_3$ and use energy and momentum conservation. For example,

$$\int d\Omega_{\vec{n}_1} d\Omega_{\vec{n}_2} d\Omega_{\vec{n}_3} \text{EEEC}'(\vec{n}_1, \vec{n}_2, \vec{n}_3) = \sum_{i,j,k} \int d\sigma \frac{E_i E_j E_k}{Q^3} = 1, \quad (3.153)$$

where we have used energy conservation $\sum_i E_i = Q$. Similarly, we also have

$$\int d\Omega_{\vec{n}_1} d\Omega_{\vec{n}_2} d\Omega_{\vec{n}_3} (1 - \vec{n}_1 \cdot \vec{n}_2) \text{EEEC}'(\vec{n}_1, \vec{n}_2, \vec{n}_3) = \sum_{i, j, k} \int d\sigma \frac{(E_i E_j - \vec{p}_i \cdot \vec{p}_j) E_k}{Q^3} = 1,$$
(3.154)

where we have used momentum conservation $\sum_i \vec{p}_i = 0$. In this way, we can derive the following Ward identities:

$$\int d\Omega_{\vec{n}_{1}} d\Omega_{\vec{n}_{2}} d\Omega_{\vec{n}_{3}} \text{EEEC}'(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3}) = 1,$$

$$\int d\Omega_{\vec{n}_{1}} d\Omega_{\vec{n}_{2}} d\Omega_{\vec{n}_{3}} (1 - \vec{n}_{1} \cdot \vec{n}_{3}) \text{EEEC}'(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3}) = 1,$$

$$\int d\Omega_{\vec{n}_{1}} d\Omega_{\vec{n}_{2}} d\Omega_{\vec{n}_{3}} (1 - \vec{n}_{2} \cdot \vec{n}_{3}) \text{EEEC}'(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3}) = 1,$$

$$\int d\Omega_{\vec{n}_{1}} d\Omega_{\vec{n}_{2}} d\Omega_{\vec{n}_{3}} (1 - \vec{n}_{1} \cdot \vec{n}_{2}) \text{EEEC}'(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3}) = 1,$$

$$\int d\Omega_{\vec{n}_{1}} d\Omega_{\vec{n}_{2}} d\Omega_{\vec{n}_{3}} (1 - \vec{n}_{1} \cdot \vec{n}_{3}) (1 - \vec{n}_{2} \cdot \vec{n}_{3}) \text{EEEC}'(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3}) = 1,$$

$$\int d\Omega_{\vec{n}_{1}} d\Omega_{\vec{n}_{2}} d\Omega_{\vec{n}_{3}} (1 - \vec{n}_{1} \cdot \vec{n}_{3}) (1 - \vec{n}_{1} \cdot \vec{n}_{2}) \text{EEEC}'(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3}) = 1,$$

$$\int d\Omega_{\vec{n}_{1}} d\Omega_{\vec{n}_{2}} d\Omega_{\vec{n}_{3}} (1 - \vec{n}_{1} \cdot \vec{n}_{3}) (1 - \vec{n}_{1} \cdot \vec{n}_{2}) \text{EEEC}'(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3}) = 1,$$

$$\int d\Omega_{\vec{n}_{1}} d\Omega_{\vec{n}_{2}} d\Omega_{\vec{n}_{3}} (1 - \vec{n}_{2} \cdot \vec{n}_{3}) (1 - \vec{n}_{1} \cdot \vec{n}_{2}) \text{EEEC}'(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3}) = 1.$$

$$(3.155)$$

Using the Ward identities, we immediately see that the $O(g^0)$ EEEC' must be given by

$$\begin{aligned} \text{EEEC}'(\vec{n}_1, \vec{n}_2, \vec{n}_3) &= \frac{1}{16\pi} \bigg(\delta(\vec{n}_1, \vec{n}_2) \delta(\vec{n}_1, \vec{n}_3) + \delta(\vec{n}_1, \vec{n}_2) \delta(\vec{n}_1, -\vec{n}_3) \\ &+ \delta(\vec{n}_1, \vec{n}_3) \delta(\vec{n}_3, -\vec{n}_2) + \delta(\vec{n}_2, \vec{n}_3) \delta(\vec{n}_2, -\vec{n}_1) \bigg). \end{aligned}$$
(3.156)

The structure of (3.156) is easy to understand. The first term is supported when all three detectors are coincident. The other three terms appear when two of the detectors are coincident and the other is diametrically opposite on the celestial sphere. Physically, the $O(g^0)$ EEEC gets contributions only from two particle states. By energy and momentum conservation, these particles must fly in opposite directions, and thus can only be observed by detectors that are either coincident (observing the same particle) or diametrically opposite (observing the two different particles).

3.5.2 Tree-level $(O(g^2))$ EEEC' in $\mathcal{N} = 4$ SYM

Let us now consider the EEEC at $O(g^2)$. We work in $\mathcal{N} = 4$ SYM and consider the case where the sink/source states are created by the operator $\text{Tr}F^2$. At this order, there are at most three particles in the outgoing state. Therefore, if the detectors are at different positions, the three vectors $\vec{n}_1, \vec{n}_2, \vec{n}_3$ must be coplanar in order for the total momentum to be zero. It follows that the $O(g^2)$ EEEC must take the form:

$$\begin{split} & \text{EEEC}'(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3})|_{O(g^{2})} \\ &= \mathcal{F}_{0}(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3})\delta\left((\vec{n}_{1} \times \vec{n}_{2}) \cdot \vec{n}_{3}\right) \\ &+ \mathcal{F}_{1}(\vec{n}_{1}, \vec{n}_{2})\delta\left(\vec{n}_{1}, \vec{n}_{3}\right) + \mathcal{F}_{1}(\vec{n}_{1}, \vec{n}_{3})\delta\left(\vec{n}_{2}, \vec{n}_{3}\right) + \mathcal{F}_{1}(\vec{n}_{2}, \vec{n}_{3})\delta\left(\vec{n}_{1}, \vec{n}_{2}\right) \\ &+ 2\pi c_{1}\delta(\vec{n}_{1}, \vec{n}_{2})\delta(\vec{n}_{1}, \vec{n}_{3}) \\ &+ 2\pi c_{2}[\delta(\vec{n}_{1}, \vec{n}_{2})\delta(\vec{n}_{1}, -\vec{n}_{3}) + \delta(\vec{n}_{1}, \vec{n}_{3})\delta(\vec{n}_{3}, -\vec{n}_{2}) + \delta(\vec{n}_{2}, \vec{n}_{3})\delta(\vec{n}_{2}, -\vec{n}_{1})]. \end{split}$$

$$(3.157)$$

On the right-hand side, the first line describes the configuration where the detectors are coplanar. The second line are the contact terms that appear when two of the detectors are at the same position and the third detector is at a generic position. The third and the fourth lines appear in the same configurations as (3.156), and can be thought of as the higher-order corrections.

Using perturbation theory, we can obtain the functions $\mathcal{F}_1(\vec{n}_1, \vec{n}_2)$ and $\mathcal{F}_0(\vec{n}_1, \vec{n}_2, \vec{n}_3)$.

We leave the details of the calculation in appendix B.5.1. The result for \mathcal{F}_1 is ¹⁸

$$\mathcal{F}_{1}(\zeta) = -\frac{\zeta(-60 + 102\zeta - 44\zeta^{2} + 3\zeta^{3}) + (-60 + 132\zeta - 90\zeta^{2} + 19\zeta^{3})\log(1-\zeta)}{256\pi^{4}(1-\zeta)\zeta^{6}}.$$
(3.158)

This expression is only valid for $0 < \zeta < 1$. The final expression should be a distribution and include contact terms at $\zeta = 0$ and $\zeta = 1$. To see the contact terms, we can further separate \mathcal{F}_1 into a singular part and a regular part:

$$\mathcal{F}_{1}(\zeta) = \frac{1}{512\pi^{4}\zeta} - \frac{1 + \log(1 - \zeta)}{256\pi^{4}(1 - \zeta)} + \mathcal{F}_{1}^{\text{reg}}(\zeta).$$
(3.159)

The singular terms can be interpreted as

$$\mathcal{F}_{1}^{\text{sing}}(\zeta) = \frac{1}{512\pi^{4}} \left[\frac{1}{\zeta} \right]_{0}^{-} \frac{1}{256\pi^{4}} \left[\frac{1}{1-\zeta} \right]_{1}^{-} - \frac{1}{256\pi^{4}} \left[\frac{\log(1-\zeta)}{1-\zeta} \right]_{1}^{-} + a_{1}\delta\left(\zeta\right) + b_{1}\delta\left(1-\zeta\right),$$
(3.160)

where the distribution $\left\lfloor \frac{1}{\zeta} \right\rfloor_0$ is defined as the unique distribution that agrees with $\frac{1}{\zeta}$ for $\zeta > 0$ and satisfies

$$\int_0^1 d\zeta \left[\frac{1}{\zeta}\right]_0 = 0, \qquad (3.161)$$

and $[\cdots]_1$ is defined in a similar way with $\zeta \to 1 - \zeta$. The expressions (3.159) and (3.160) now specify $\mathcal{F}_1(\zeta)$ as a distribution. However, this distribution depends on unknown coefficients a_1, b_1 . We will determine them using Ward identities.

Now consider the function \mathcal{F}_0 . We find that it can be written as¹⁹

$$\mathcal{F}_{0}(\zeta_{1},\zeta_{2}) = \sqrt{\zeta_{1}\zeta_{2}(1-\zeta_{1})(1-\zeta_{2})}\widetilde{\mathcal{F}}_{0}(\zeta_{1},\zeta_{2})\theta(\zeta_{1}+\zeta_{2}-1), \qquad (3.162)$$

where $\theta(\dots)$ is a step function (we explain its appearance in appendix B.5.1). The function $\widetilde{\mathcal{F}}_0$ is given by

$$\widetilde{\mathcal{F}}_{0}(\zeta_{1},\zeta_{2}) = \frac{1}{256\pi^{4}\zeta_{1}^{2}\zeta_{2}^{2}\sqrt{\zeta_{1}\zeta_{2}(1-\zeta_{1})(1-\zeta_{2})}(\sqrt{\zeta_{1}(1-\zeta_{2})}+\sqrt{\zeta_{2}(1-\zeta_{1})})^{4}} \\ \times \left(-14\zeta_{1}^{2}\zeta_{2}^{2}-18\zeta_{1}\zeta_{2}+19\zeta_{1}\zeta_{2}(\zeta_{1}+\zeta_{2})-6(\zeta_{1}+\zeta_{2})^{2}+12(\zeta_{1}+\zeta_{2})^{2}+12(\zeta_{1}+\zeta_$$

¹⁸Note that $\mathcal{F}_1(\vec{n}_1, \vec{n}_2)$ is invariant under an overall rotation of \vec{n}_1, \vec{n}_2 , so it can be written as a

function of the cross ratio $\zeta = \frac{1 - \vec{n}_1 \cdot \vec{n}_2}{2}$. ¹⁹Again due to rotational invariance, we can write \mathcal{F}_0 as a function of two cross ratios ζ_1, ζ_2 , where $\zeta_1 = \frac{1 - \vec{n}_1 \cdot \vec{n}_3}{2}, \zeta_2 = \frac{1 - \vec{n}_2 \cdot \vec{n}_3}{2}$.

It is convenient to study $\widetilde{\mathcal{F}}_0$ instead of \mathcal{F}_0 . To interpret $\widetilde{\mathcal{F}}_0$ as a distribution, we again separate it into a singular part and regular part as $\widetilde{\mathcal{F}}_0 = \widetilde{\mathcal{F}}_0^{\text{sing}} + \widetilde{\mathcal{F}}_0^{\text{reg}}$. The singular part is given by

$$\widetilde{\mathcal{F}}_{0}^{\text{sing}}(\zeta_{1},\zeta_{2}) = \frac{f(\theta_{1})}{r_{1}^{2}} + \frac{f(\theta_{2})}{r_{2}^{2}} + \frac{g(\theta_{3})}{r_{3}^{2}},$$
(3.164)

where

$$r_{1} = \sqrt{\zeta_{1}^{2} + (1 - \zeta_{2})^{2}}, \qquad \theta_{1} = \tan^{-1}\left(\frac{1 - \zeta_{2}}{\zeta_{1}}\right),$$

$$r_{2} = \sqrt{\zeta_{2}^{2} + (1 - \zeta_{1})^{2}}, \qquad \theta_{2} = \tan^{-1}\left(\frac{1 - \zeta_{1}}{\zeta_{2}}\right),$$

$$r_{3} = \sqrt{(1 - \zeta_{1})^{2} + (1 - \zeta_{2})^{2}}, \qquad \theta_{3} = \tan^{-1}\left(\frac{1 - \zeta_{2}}{1 - \zeta_{1}}\right). \qquad (3.165)$$

 (r_i, θ_i) are the polar coordinates with respect to the points where $\widetilde{\mathcal{F}}_0$ becomes



Figure 3.3: Definition of (r_i, θ_i) . Due to the step function in (3.162), $\mathcal{F}_0(\zeta_1, \zeta_2)$ is nonzero only in the blue region. The three points (1, 0), (0, 1), (1, 1) are where $\mathcal{F}_0(\zeta_1, \zeta_2)$ becomes singular.

singular. See figure 3.3. The two functions in (3.164) are given by

$$f(\theta) = \frac{1}{256\pi^4 \sqrt{\sin(\theta)} \cos^{\frac{3}{2}}(\theta)},$$

$$g(\theta) = \frac{1}{256\pi^4 \sqrt{\sin(\theta)} \sqrt{\cos(\theta)} \left(\sqrt{\sin(\theta)} + \sqrt{\cos(\theta)}\right)^2}.$$
(3.166)

One can then show that $\widetilde{\mathcal{F}}_0^{\mathrm{sing}}$ as a distribution should be given by

$$\widetilde{\mathcal{F}}_{0}^{\text{sing}}(\zeta_{1},\zeta_{2}) = a_{0} \left(\frac{\delta(r_{1})}{r_{1}} + \frac{\delta(r_{2})}{r_{2}} \right) + b_{0} \frac{\delta(r_{3})}{r_{3}} + \frac{f_{0}(\theta_{1})}{r_{1}} \left[\frac{1}{r_{1}} \right]_{0} + \frac{f_{0}(\theta_{2})}{r_{2}} \left[\frac{1}{r_{2}} \right]_{0} + \frac{g_{0}(\theta_{3})}{r_{3}} \left[\frac{1}{r_{3}} \right]_{0} +$$

Moreover, from (3.157) we see that \mathcal{F}_0 must be crossing symmetric. This condition implies²⁰

$$b_0 - \frac{1}{2}a_0 = \frac{\pi - \sqrt{2}\pi + 2\log 2 + \sqrt{2}\log(3 - 2\sqrt{2})}{256\pi^5} \approx 0.00003291.$$
(3.168)

Note that although we have six unknown coefficients $(c_1, c_2, a_1, b_1, a_0, b_0)$, there are actually only two types of contact terms in (3.157). By integrating against test functions, we find

$$\frac{\delta(r_1)}{r_1} \sqrt{\zeta_1 \zeta_2 (1 - \zeta_1) (1 - \zeta_2)} \delta\left((\vec{n}_1 \times \vec{n}_2) \cdot \vec{n}_3\right) = \frac{\pi^2}{2} \delta\left(\vec{n}_1, \vec{n}_3\right) \delta\left(\vec{n}_2, -\vec{n}_3\right), \\
\frac{\delta(r_2)}{r_2} \sqrt{\zeta_1 \zeta_2 (1 - \zeta_1) (1 - \zeta_2)} \delta\left((\vec{n}_1 \times \vec{n}_2) \cdot \vec{n}_3\right) = \frac{\pi^2}{2} \delta\left(\vec{n}_1, -\vec{n}_3\right) \delta\left(\vec{n}_2, \vec{n}_3\right), \\
\frac{\delta(r_3)}{r_3} \sqrt{\zeta_1 \zeta_2 (1 - \zeta_1) (1 - \zeta_2)} \delta\left((\vec{n}_1 \times \vec{n}_2) \cdot \vec{n}_3\right) = \pi^2 \delta\left(\vec{n}_1, -\vec{n}_3\right) \delta\left(\vec{n}_2, -\vec{n}_3\right), \\
(3.169)$$

and in addition

$$\delta(\frac{1-\vec{n}_{1}\cdot\vec{n}_{2}}{2})\delta(\vec{n}_{2},\vec{n}_{3}) = 4\pi\delta(\vec{n}_{1},\vec{n}_{2})\delta(\vec{n}_{2},\vec{n}_{3}),$$

$$\delta(1-\frac{1-\vec{n}_{1}\cdot\vec{n}_{2}}{2})\delta(\vec{n}_{2},\vec{n}_{3}) = 4\pi\delta(\vec{n}_{1},-\vec{n}_{2})\delta(\vec{n}_{2},\vec{n}_{3}).$$
(3.170)

Collecting all the contact terms together, we find

$$2\pi (c_{1} + 6a_{1}) \,\delta(\vec{n}_{1}, \vec{n}_{2}) \delta(\vec{n}_{2}, \vec{n}_{3}) + 2\pi \left(c_{2} + 2b_{1} + \frac{\pi}{4}a_{0}\right) \left(\delta(\vec{n}_{1}, \vec{n}_{3})\delta(\vec{n}_{2}, -\vec{n}_{3}) + \delta(\vec{n}_{1}, -\vec{n}_{3})\delta(\vec{n}_{2}, \vec{n}_{3})\right) + 2\pi \left(c_{2} + 2b_{1} + \frac{\pi}{2}b_{0}\right) \delta(\vec{n}_{1}, -\vec{n}_{3}) \,\delta(\vec{n}_{2}, -\vec{n}_{3}) \,.$$
(3.171)

Only the above linear combinations of coefficients are physically meaningful. Note that the coefficients of $\delta(\vec{n}_1, \vec{n}_3)\delta(\vec{n}_2, -\vec{n}_3)$ and $\delta(\vec{n}_1, -\vec{n}_3)\delta(\vec{n}_2, -\vec{n}_3)$ are different because of our choice of coordinates (3.165) to write the distributions $[\cdots]_0$. These coordinates are convenient for analyzing the singularities of \mathcal{F}_0 , but they do not manifest crossing symmetry. When we perform a crossing transformation, we

²⁰See appendix B.5 for a derivation.

rescale the arguments of the distributions $[\cdots]_0$, which can produce δ -functions via $[1/(\lambda x)]_0 = \lambda^{-1} [1/x]_0 - (\log \lambda/\lambda) \delta(x)$. Taking these δ -functions into account, one can show that as long as (3.168) is satisfied, the full expression is crossing symmetric. See appendix B.5.2 for details.

We can now use the Ward identities (3.155) to determine the coefficients of the contact terms. Plugging (3.157) into (3.155), we obtain

$$\int_{0}^{1} d\zeta_{1} \int_{0}^{1} d\zeta_{2} \,\widetilde{\mathcal{F}}_{0}(\zeta_{1},\zeta_{2})\theta(\zeta_{1}+\zeta_{2}-1) + 6 \int_{0}^{1} d\zeta \,\mathcal{F}_{1}(\zeta) + c_{1} + 3c_{2} = 0,$$

$$2 \int_{0}^{1} d\zeta_{1} \int_{0}^{1} d\zeta_{2} \,\zeta_{1} \widetilde{\mathcal{F}}_{0}(\zeta_{1},\zeta_{2})\theta(\zeta_{1}+\zeta_{2}-1) + 8 \int_{0}^{1} d\zeta \,\zeta\mathcal{F}_{1}(\zeta) + 4c_{2} = 0,$$

$$4 \int_{0}^{1} d\zeta_{1} \int_{0}^{1} d\zeta_{2} \,\zeta_{1} \zeta_{2} \widetilde{\mathcal{F}}_{0}(\zeta_{1},\zeta_{2})\theta(\zeta_{1}+\zeta_{2}-1) + 8 \int_{0}^{1} d\zeta \,\zeta^{2}\mathcal{F}_{1}(\zeta) + 4c_{2} = 0,$$
(3.172)

where $\tilde{\mathcal{F}}_0(\zeta_1, \zeta_2)$ is defined by (3.162). The integrals in (3.172) can be computed using (3.158), (3.160), (3.163), and (3.164). The integrals of the $\tilde{\mathcal{F}}_0$ function are given by

$$\int_{0}^{1} d\zeta_{1} \int_{0}^{1} d\zeta_{2} \,\widetilde{\mathcal{F}}_{0}(\zeta_{1},\zeta_{2})\theta(\zeta_{1}+\zeta_{2}-1) = \frac{\pi}{2}(a_{0}+b_{0}) + 0.00011205,$$
$$\int_{0}^{1} d\zeta_{1} \int_{0}^{1} d\zeta_{2} \,\zeta_{1}\widetilde{\mathcal{F}}_{0}(\zeta_{1},\zeta_{2})\theta(\zeta_{1}+\zeta_{2}-1) = \frac{\pi}{2}\left(\frac{a_{0}}{2}+b_{0}\right) + 0.0000815314,$$
$$\int_{0}^{1} d\zeta_{1} \int_{0}^{1} d\zeta_{2} \,\zeta_{1}\zeta_{2}\widetilde{\mathcal{F}}_{0}(\zeta_{1},\zeta_{2})\theta(\zeta_{1}+\zeta_{2}-1) = \frac{\pi}{2}b_{0} + 0.0000884406, \quad (3.173)$$

and for the \mathcal{F}_1 integrals we get

$$\int_{0}^{1} d\zeta \mathcal{F}_{1}(\zeta) = \frac{51 + 10\pi^{2}}{15360\pi^{4}} + a_{1} + b_{1},$$
$$\int_{0}^{1} d\zeta \zeta \mathcal{F}_{1}(\zeta) = \frac{3 + 2\pi^{2}}{3072\pi^{4}} + b_{1},$$
$$\int_{0}^{1} d\zeta \zeta^{2} \mathcal{F}_{1}(\zeta) = \frac{-4 + \pi^{2}}{1536\pi^{4}} + b_{1}.$$
(3.174)

Using these results, we find the solution to (3.172):

$$b_0 - \frac{a_0}{2} = 0.0000329062,$$

$$c_2 + 2b_1 + \frac{\pi}{4}a_0 = -0.00021859,$$

$$c_1 + 6a_1 = -0.000108274.$$
 (3.175)

We see that the first line is consistent with (3.168), which comes from crossing symmetry of $\mathcal{F}_0(\zeta_1, \zeta_2)$. The second and the third line give the coefficients of the two types of contact terms that can appear in EEEC'|_{$O(g^2)$}.

3.5.3 Contact terms from the conformal block decomposition

It is interesting to ask how the δ -functions in EEEC'|_ $O(g^2)$ are reproduced from the conformal block decomposition described in section 3.2. Since the conformal block decomposition describes the leading term in the collinear limit, we should study those δ -functions that survive when all \vec{n}_i 's are close to each other. More precisely, we have an expansion around the squeezed limit, where we first take \vec{n}_2 and \vec{n}_3 close together, and then \vec{n}_1 .

Using (3.42), the collinear limit is

$$\text{EEEC}'(\vec{n}_1, \vec{n}_2, \vec{n}_3)|_{O(g^2)} = \frac{1}{32\pi^2} \frac{\zeta_{12}}{\zeta_{23}^3} \sum_{\delta,j} R_{\delta,j}|_{O(g^2)} g_{\delta,j}^{\mathcal{E}\mathcal{E}\mathcal{E}\mathcal{P}_{\delta'_*}}(z, \overline{z}) + \dots, \quad (3.176)$$

where z, \overline{z} are defined by

$$\frac{\zeta_{23}}{\zeta_{12}} = z\overline{z}, \quad \frac{\zeta_{13}}{\zeta_{12}} = (1-z)(1-\overline{z}). \tag{3.177}$$

The small z, \overline{z} limit corresponds to the limit where \vec{n}_2 and \vec{n}_3 become close. The contact term $\mathcal{F}_1(\vec{n}_1, \vec{n}_2)\delta(\vec{n}_2, \vec{n}_3)$ can then appear from the exchanged quantum numbers $\delta = 4, j = 0$. More precisely, we have

$$EEEC'(\vec{n}_1, \vec{n}_2, \vec{n}_3)|_{O(g^2)} \sim \frac{g^2 N_c}{512\pi^5} \frac{1}{\zeta_{12}^2} \lim_{\delta \to 4} \langle R_{\delta,0}^{(0)} \rangle(z\overline{z})^{\frac{\delta}{2}-3} = \frac{g^2 N_c}{64\pi^4} \left(\lim_{\delta \to 4} \frac{\langle R_{\delta,0}^{(0)} \rangle}{\delta - 4} \right) \frac{1}{\zeta_{12}} \delta(\vec{n}_2, \vec{n}_3)$$
(3.178)

This result should agree with the $\frac{1}{\zeta}$ term of $\mathcal{F}_1(\zeta)$ in (3.160). Matching the two expressions, we obtain

$$\lim_{\delta \to 4} \frac{\langle R_{\delta,0}^{(0)} \rangle}{\delta - 4} = \frac{1}{8}.$$
(3.179)

Thus, even though $\langle R_{4,0}^{(0)} \rangle$ vanishes, the zero of $\langle R_{\delta,0}^{(0)} \rangle$ at $\delta = 4$ is related to the EEEC at $O(g^2)$ order. It would be interesting to verify (3.179) using other methods.

3.6 EEEC at strong coupling

We now consider the EEEC at strong coupling. In [12], Hofman and Maldacena computed the EEEC at strong coupling in $\mathcal{N} = 4$ SYM up to $O(\lambda^{-3/2})$ order using

AdS/CFT for a sink/source state created by a massless closed string. In this paper, we focus on the leading order and $O(1/\lambda)$ correction. In terms of (3.149), the result of [12] is

$$\begin{aligned} &\text{EEEC}_{\text{strong}}'(\vec{n}_1, \vec{n}_2, \vec{n}_3) \\ &= \left(\frac{1}{4\pi}\right)^3 \left(1 + \frac{6\pi^2}{\lambda} \left((\vec{n}_1 \cdot \vec{n}_2)^2 + (\vec{n}_1 \cdot \vec{n}_3)^2 + (\vec{n}_2 \cdot \vec{n}_3)^2 - 1\right) + O(\lambda^{-3/2})\right). \end{aligned}$$
(3.180)

Let us lift this expression into a more covariant form, using embedding space coordinates for the celestial sphere $z_i \in \mathbb{R}^{d-1,1}$ with $z_i^2 = 0$. We define

$$\mathcal{F}(z_1, z_2, z_3, p) \equiv \frac{8 \int d^d x \ e^{ip \cdot x} \langle 0|O^{\dagger}(x)\mathbf{L}[T](\infty, z_1)\mathbf{L}[T](\infty, z_2)\mathbf{L}[T](\infty, z_3)O(0)|0\rangle}{(-p^2)^{\frac{3}{2}} \int d^d x \ e^{ip \cdot x} \langle 0|O^{\dagger}(x)O(0)|0\rangle}$$
(3.181)

which is a homogeneous function of the z_i . The original EEEC can be recovered by specializing: $\mathcal{F}(z_i = (1, \vec{n}_i), p = (1, \vec{0})) = \text{EEEC}'(\vec{n}_i)$. In embedding space coordinates, the expression (3.180) becomes

$$\mathcal{F}_{\text{strong}}(z_1, z_2, z_3, p) = \frac{8}{\pi^3} \frac{1}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3 (-2z_3 \cdot p)^3} \left(1 + \frac{6\pi^2}{\lambda} \left((1 - 2\zeta_{12})^2 + (1 - 2\zeta_{23})^2 + (1 - 2\zeta_{13})^2 - 1 \right) \right) + O(\lambda^{-3/2}),$$
(3.182)

where the cross ratios ζ_{ij} are defined in (3.19). For this section, we also assume that $-p^2 = 1$, since the $-p^2$ factors can be easily restored using dimensional analysis of p.

In the weak coupling limit, the EEEC at tree-level is only nonzero when the three detectors are coplanar, and the EEEC at 1-loop it is only known in the collinear limit. On the other hand, the strong-coupling EEEC given in (3.182) is valid for any detector positions on the celestial sphere.

Just as Mean Field Theory provides a simple example of a crossing-symmetric, conformally-invariant four-point function, the strong-coupling EEEC given in (3.182) gives a simple example of a crossing-symmetric, Lorentz invariant EEEC. It is therefore a perfect target for us to study its celestial block expansion and test the results of section 3.2. Because the strong-coupling EEEC is so simple, we will be able to compute its *complete* 3-point celestial block expansion — i.e. not just its conformal block expansion in the collinear limit.

The celestial block expansion can be written as²¹

$$\mathcal{F}_{\text{strong}}(z_1, z_2, z_3, p) = \sum_{\delta'} \sum_{\delta, j} P_{\delta, j; \delta'} G^c_{\delta, j; \delta'}(z_1, z_2, z_3, p), \qquad (3.183)$$

where the celestial block is defined as

$$G_{\delta,j;\delta'}^{c}(z_{1}, z_{2}, z_{3}, p) = C_{12\mathcal{P}_{\delta,j}}(z_{1}, z_{2}, \partial_{z_{2}}, \partial_{w_{2}})C_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta'}}(z_{2}, z_{3}, \partial_{z_{3}})(-2z_{3} \cdot p)^{-\delta'}.$$
(3.184)

Let us first focus on the j = 0 sector of (3.183). The light-ray OPE formula gives a relation between the OPE data and the value of δ in the sum. In the strong coupling limit, the $T \times T$ OPE should contain double-trace operators with twists $\tau = 2\tau_T + 2n = 4 + 2n$. Therefore, the values of δ appearing in (3.183) should be

$$\delta = \Delta(J=3) - 1 = 6 + 2n. \tag{3.185}$$

Similarly, the $T \times T \times T$ OPE should contain triple-trace operators with twists $\tau = 3\tau_T + 2k = 6 + 2k$. As argued in section 3.3.1, we expect that the values of δ' appearing in (3.183) are given by

$$\delta' = \Delta(J = 4) - 1 = 9 + 2k. \tag{3.186}$$

To study the celestial block expansion of the strong-coupling EEEC, we will use harmonic analysis for the Euclidean conformal group SO(d - 1, 1) [44]. A modern review of harmonic analysis for Euclidean CFTs is given in [45], where they derive a Euclidean inversion formula that expresses OPE data as an integral of CFT fourpoint functions over Euclidean space. In this section, we use techniques from [45] to derive a "celestial inversion formula" that expresses the celestial block expansion data as an integral of the EEEC over the celestial sphere. We then consider the strong coupling limit and use this celestial inversion formula to obtain the celestial block expansion (3.183) for the strong-coupling EEEC (3.182).

3.6.1 Celestial inversion formula

We first derive the celestial inversion formula, following the derivation of the Euclidean inversion formula in section 2 of [45]. By harmonic analysis for SO(d-1, 1), the EEEC $\mathcal{F}(z_1, z_2, z_3, p)$ defined in (3.181) can be written as an integral of the form

$$\mathcal{F}(z_1, z_2, z_3, p) = \sum_j \int_{\frac{d-2}{2}}^{\frac{d-2}{2} + i\infty} \frac{d\delta}{2\pi i} \int_{\frac{d-2}{2}}^{\frac{d-2}{2} + i\infty} \frac{d\delta'}{2\pi i} I(\delta, j; \delta') \Psi^c_{\delta, j; \delta'}(z_1, z_2, z_3, p),$$
(3.187)

²¹The relation between $P_{\delta,j;\delta'}$ and the coefficient $R_{\delta,j;\delta'}$ defined in section 3.2 is $P_{\delta,j;\delta'} = \frac{16}{\pi^2} R_{\delta,j;\delta'}$.

where $\Psi^c_{\delta,j;\delta'}$ is a "celestial partial wave" defined by

$$\Psi_{\delta,j;\delta'}^{c}(z_{1},z_{2},z_{3},p) = \int D^{d-2}z D^{d-2}z' \langle \mathcal{P}_{\delta_{1}}(z_{1})\mathcal{P}_{\delta_{2}}(z_{2})\mathcal{P}_{\delta,j}(z)\rangle \langle \widetilde{\mathcal{P}}_{\delta,j}(z)\mathcal{P}_{\delta_{3}}(z_{3})\mathcal{P}_{\delta'}(z')\rangle \frac{1}{(-2z'\cdot p)^{\widetilde{\delta'}}},$$
(3.188)

where $\widetilde{\mathcal{P}}_{\delta,j}$ is the shadow representation of $\mathcal{P}_{\delta,j}$, with scaling dimension $\widetilde{\delta} = d - 2 - \delta$. The measure $D^{d-2}z$ is defined by

$$D^{d-2}z = \frac{2d^d z \delta(z^2)\theta(z^0)}{\text{vol } \mathbb{R}_+},$$
(3.189)

where \mathbb{R}_+ acts by rescaling *z*. Note that the operators $\mathcal{P}_{\delta,j}(z)$ and $\widetilde{\mathcal{P}}_{\delta,j}(z)$ each carry *j* tangent-space indices on the celestial sphere. These indices are implicitly contracted in (3.188) and below.

By construction, Ψ^c is an eigenfunction of the Casimirs of SO(d - 1, 1), acting simultaneously on z_1, z_2 and simultaneously on z_1, z_2, z_3 . So, we can study the behavior of Ψ^c in the OPE limit to determine its relation to the celestial block G^c . Following the logic in [45], the relation is

$$\Psi_{\delta,j;\delta'}^{c} = S(\mathcal{P}_{\delta_{3}}\mathcal{P}_{\delta'}[\widetilde{\mathcal{P}}_{\delta,j}])I_{\delta'}(p)G_{\delta,j;\delta'}^{c} + S(\mathcal{P}_{\delta_{3}}\mathcal{P}_{\delta'}[\widetilde{\mathcal{P}}_{\delta,j}])S(\mathcal{P}_{\delta,j}\mathcal{P}_{\delta_{3}}[\mathcal{P}_{\delta'}])G_{\delta,j;\delta'}^{c} + S(\mathcal{P}_{\delta_{1}}\mathcal{P}_{\delta_{2}}[\mathcal{P}_{\delta,j}])S(\widetilde{\mathcal{P}}_{\delta,j}\mathcal{P}_{\delta_{3}}[\mathcal{P}_{\delta'}])G_{\delta,j;\delta'}^{c},$$

$$(3.190)$$

where the coefficients $S(\cdots)$, $I_{\delta'}(p)$ are defined by

$$\int D^{d-2} z \langle \widetilde{\mathcal{P}}(z') \widetilde{\mathcal{P}}(z) \rangle \langle \mathcal{P}_1 \mathcal{P}_2 \mathcal{P}(z) \rangle = S(\mathcal{P}_1 \mathcal{P}_2[\mathcal{P}]) \langle \mathcal{P}_1 \mathcal{P}_2 \widetilde{\mathcal{P}}(z') \rangle$$
$$\int D^{d-2} z' \langle \mathcal{P}_{\delta'}(z_3) \mathcal{P}_{\delta'}(z') \rangle \frac{1}{(-2z' \cdot p)^{\widetilde{\delta'}}} = I_{\delta'}(p)(-2z_3 \cdot p)^{-\delta'}. \tag{3.191}$$

More explicitly, their expressions are

$$S(\mathcal{P}_{1}\mathcal{P}_{2}[\mathcal{P}_{\delta,j}]) = \frac{\pi^{\frac{d-2}{2}}\Gamma(\delta - \frac{d-2}{2})\Gamma(\delta + j - 1)\Gamma(\frac{\tilde{\delta}+\delta_{1}-\delta_{2}+j}{2})\Gamma(\frac{\tilde{\delta}-\delta_{1}+\delta_{2}+j}{2})}{\Gamma(\delta - 1)\Gamma(d - 2 - \delta + j)\Gamma(\frac{\delta+\delta_{1}-\delta_{2}+j}{2})\Gamma(\frac{\delta-\delta_{1}+\delta_{2}+j}{2})},$$

$$S(\mathcal{P}_{\delta,j}\mathcal{P}_{3}[\mathcal{P}_{\delta'}]) = \frac{\pi^{\frac{d-2}{2}}\Gamma(\delta' - \frac{d-2}{2})\Gamma(\frac{\tilde{\delta}'+\delta_{3}-\delta+j}{2})\Gamma(\frac{\tilde{\delta}'-\delta_{3}+\delta+j}{2})}{\Gamma(d - 2 - \delta')\Gamma(\frac{\delta'+\delta_{3}-\delta+j}{2})\Gamma(\frac{\delta'-\delta_{3}+\delta+j}{2})},$$

$$I_{\delta'}(p) = \frac{\pi^{\frac{d-2}{2}}\Gamma(\frac{d-2}{2}-\delta')}{\Gamma(d - 2 - \delta')}(-p^{2})^{\delta'-\frac{d-2}{2}}.$$
(3.192)

Just like the four-point conformal partial wave, the celestial partial wave Ψ^c satisfies an orthogonality relation that can be derived using a "bubble" formula. Consider two celestial partial waves $\Psi^c_{\delta_5,j_5;\delta'_5}$ and $\Psi^{c(\tilde{\delta}_i)}_{\tilde{\delta}_6,j_6;\tilde{\delta}'_6}$, where $\delta_{5,6} = \frac{d-2}{2} + is_{5,6}$, $\delta'_{5,6} = \frac{d-2}{2} + is'_{5,6}$ are on the principal series and the external dimensions of $\Psi^{c(\tilde{\delta}_i)}$ are the shadows of Ψ^c . The orthogonality relation is (see Appendix B.6 for a derivation)

$$\int \frac{D^{d-2} z_1 D^{d-2} z_2 D^{d-2} z_3 d_{AdS}^{d-1} p}{\operatorname{vol}(\operatorname{SO}(d-1,1))} \Psi_{\delta_5,j_5;\delta_5'}^c(z_1, z_2, z_3, p) \Psi_{\widetilde{\delta}_6,j_6;\widetilde{\delta}_6'}^{c(\widetilde{\delta}_i)}(z_1, z_2, z_3, p)$$

$$= \frac{1}{2^{d-2} \operatorname{vol}(\operatorname{SO}(d-2))} B_{12\mathcal{P}_{\delta_5,j_5}} B_{\widetilde{\mathcal{P}}_{\delta_5,j_5} 3\mathcal{P}_{\delta_5'}} \delta_{\mathcal{P}_5} \mathcal{P}_6} \delta_{\mathcal{P}_5'} \mathcal{P}_6', \qquad (3.193)$$

where the integral measure for p and the delta function $\delta_{\mathcal{P}_5\mathcal{P}_6}$ are defined by

$$d_{AdS}^{d-1}p = 2d^{d}p\delta(p^{2}+1)\theta(p^{0}),$$

$$\delta_{\mathcal{P}_{5}\mathcal{P}_{6}} = 2\pi\delta(s_{5}-s_{6})\delta_{j_{5},j_{6}}.$$
(3.194)

The B-coefficients are "bubble" coefficients defined by

$$B_{12\mathcal{P}_{\delta,j}} = \frac{1}{\mu(\delta,j)} \left(\langle \mathcal{P}_1 \mathcal{P}_2 \mathcal{P}_{\delta,j} \rangle, \langle \widetilde{\mathcal{P}}_1 \widetilde{\mathcal{P}}_2 \widetilde{\mathcal{P}}_{\delta,j} \rangle \right), \qquad (3.195)$$

where $\mu(\delta, j)$ is the Plancherel measure of SO(d-1, 1), and $\left(\langle \mathcal{P}_1 \mathcal{P}_2 \mathcal{P}_{\delta,j} \rangle, \langle \widetilde{\mathcal{P}}_1 \widetilde{\mathcal{P}}_2 \widetilde{\mathcal{P}}_{\delta,j} \rangle\right)$ is a conformally-invariant three-point pairing. Their explicit expressions are given in [45].

Integrating both sides of celestial partial wave expansion (3.187) against a celestial partial wave $\Psi^{c(\tilde{\delta}_i)}_{\tilde{\delta},i;\tilde{\delta}'}$, the orthogonality relation (3.193) gives

$$I(\delta, j; \delta') = \frac{2^{d-2} \operatorname{vol}(\mathrm{SO}(d-2))}{B_{12\mathcal{P}_{\delta,j}} B_{\tilde{\mathcal{P}}_{\delta,j}3\mathcal{P}_{\delta'}}} \int \frac{D^{d-2} z_1 D^{d-2} z_2 D^{d-2} z_3 d_{\mathrm{AdS}}^{d-1} p}{\operatorname{vol}(\mathrm{SO}(d-1,1))} \mathcal{F}(z_1, z_2, z_3, p) \Psi_{\tilde{\delta}, j; \tilde{\delta}'}^{c(\tilde{\delta}_i)}(z_1, z_2, z_3, p)$$
(3.196)

This is the celestial inversion formula that expresses the celestial partial wave expansion data $I(\delta, j; \delta')$ as an integral of the EEEC $\mathcal{F}(z_1, z_2, z_3, p)$ over the celestial sphere. Finally, we must find the relation between the celestial partial wave expansion (3.187) and the celestial block expansion (3.183). Plugging (3.190) into

(3.187), we obtain

$$\mathcal{F}(z_{1}, z_{2}, z_{3}, p) = \sum_{j} \int_{\frac{d-2}{2}}^{\frac{d-2}{2}+i\infty} \frac{d\delta}{2\pi i} \int_{\frac{d-2}{2}}^{\frac{d-2}{2}+i\infty} \frac{d\delta'}{2\pi i} I(\delta, j; \delta') \times \left(S(\mathcal{P}_{\delta_{3}}\mathcal{P}_{\delta'}[\widetilde{\mathcal{P}}_{\delta,j}]) I_{\delta'}(p) G^{c}_{\delta,j;\delta'} + S(\mathcal{P}_{\delta_{3}}\mathcal{P}_{\delta'}[\widetilde{\mathcal{P}}_{\delta,j}]) S(\mathcal{P}_{\delta,j}\mathcal{P}_{\delta_{3}}[\mathcal{P}_{\delta'}]) G^{c}_{\delta,j;\widetilde{\delta'}} + S(\mathcal{P}_{\delta_{1}}\mathcal{P}_{\delta_{2}}[\mathcal{P}_{\delta,j}]) I_{\delta'}(p) G^{c}_{\widetilde{\delta},j;\delta'} + S(\mathcal{P}_{\delta_{1}}\mathcal{P}_{\delta_{2}}[\mathcal{P}_{\delta,j}]) S(\widetilde{\mathcal{P}}_{\delta,j}\mathcal{P}_{\delta_{3}}[\mathcal{P}_{\delta'}]) G^{c}_{\widetilde{\delta},j;\widetilde{\delta'}} \right)$$

$$(3.197)$$

Using the inversion formula (3.196), one can show that above expression remains the same when we keep only the first term $S(\mathcal{P}_{\delta_3}\mathcal{P}_{\delta'}[\widetilde{\mathcal{P}}_{\delta,j}])I_{\delta'}(p)G^c_{\delta,j;\delta'}$ in the parentheses and extend the integration ranges to $\int_{\frac{d-2}{2}-i\infty}^{\frac{d-2}{2}+i\infty}$. Therefore, we have

$$\mathcal{F}(z_1, z_2, z_3, p) = \sum_j \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta}{2\pi i} \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta'}{2\pi i} C(\delta, j; \delta') G^c_{\delta, j; \delta'}(z_1, z_2, z_3, p),$$
(3.198)

where

$$C(\delta, j; \delta') \equiv I(\delta, j; \delta') S(\mathcal{P}_{\delta_3} \mathcal{P}_{\delta'}[\widetilde{\mathcal{P}}_{\delta, j}]) I_{\delta'}(p).$$
(3.199)

Finally, we can close the contour of the δ and δ' integrals in (3.198) to the right and obtain the celestial block expansion.²² In particular, for the strong-coupling EEEC (3.183), we have

$$\operatorname{Res}_{\delta=\delta_{i}}\operatorname{Res}_{\delta'=\delta_{i}'}C_{\operatorname{strong}}(\delta,j;\delta') = P_{\delta_{i},j;\delta_{i}'}.$$
(3.200)

3.6.2 Leading order

We first consider the leading order term of (3.182),

$$\mathcal{F}_{\text{strong}}^{(0)}(z_1, z_2, z_3, p) = \frac{8}{\pi^3 (-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3 (-2z_3 \cdot p)^3}.$$
 (3.201)

²²As we will see later, closing the δ contour or the δ' contour first gives the same celestial block expansion, at least for the strong-coupling EEEC we consider in this paper. Moreover, in appendix B.6 we study the celestial block $G^c_{\delta,j;\delta'}$ at large δ and δ' , and show that contributions at infinity of both contours vanish.

Plugging this into the celestial inversion formula (3.196), we find

Let us study the z_1, z_2 integral,

$$\int D^{d-2} z_1 D^{d-2} z_2 \frac{1}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3} \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta,j}(z,w) \rangle. \quad (3.203)$$

After integration, the polarization vector w must appear in the combination $[z, w]^{\mu\nu} \equiv z^{\mu}w^{\nu} - z^{\nu}w^{\mu}$ due to Lorentz invariance. However, the only remaining vector that is left unintegrated is p, and [z, w] cannot be contracted with anything. Therefore, this integral must vanish except when j = 0. To compute the integral for j = 0, we can use

$$\int D^{d-2} z_1 D^{d-2} z_2 \langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_2}(z_2) \mathcal{P}_{\delta,j=0}(z) \rangle (-2p \cdot z_1)^{-\delta_1} (-2p \cdot z_2)^{-\delta_2} = C_{\delta_1,\delta_2;\delta} (-p^2)^{\frac{\delta-\delta_1-\delta_2}{2}} (-2p \cdot z)^{-\delta},$$
(3.204)

where

$$C_{\delta_1,\delta_2;\delta} = \frac{\pi^{d-2}\Gamma(\frac{\delta_1+\delta_2+\delta-d+2}{2})\Gamma(\frac{\delta_1+\delta_2-\delta}{2})\Gamma(\frac{d-2-\delta_2-\delta+\delta_1}{2})\Gamma(\frac{d-2-\delta_1-\delta+\delta_2}{2})}{\Gamma(\delta_1)\Gamma(\delta_2)\Gamma(d-2-\delta)\Gamma(\frac{d-2}{2})}.$$
 (3.205)

We give a derivation of (3.204) in appendix B.6. After integrating over z_1 and z_2 , we obtain

$$I_{\text{strong}}^{(0)}(\delta,0;\delta') = \frac{8}{\pi^3} \frac{2^{d-2} \operatorname{vol}(\operatorname{SO}(d-2))}{B_{\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta,j}} B_{\widetilde{\mathcal{P}}_{\delta,j}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta'}}} C_{3,3;\widetilde{\delta}} \times \int \frac{D^{d-2} z D^{d-2} z' D^{d-2} z_3 d_{\text{AdS}}^{d-1} p}{\operatorname{vol}(\operatorname{SO}(d-1,1))} \frac{1}{(-2z_3 \cdot p)^3 (-2z \cdot p)^{\widetilde{\delta}}} \langle \mathcal{P}_{\delta,0}(z) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_3) \mathcal{P}_{\widetilde{\delta'}}(z') \rangle \frac{1}{(-2z' \cdot p)^{\delta'}}.$$
(3.206)

The remaining *p*-integral is just a three-point Witten diagram, and it is given by [96]

$$\int d_{AdS}^{d-1} p \ (-2p \cdot z_1)^{-\delta_1} (-2p \cdot z_2)^{-\delta_2} (-2p \cdot z_3)^{-\delta_3}$$

= $D_{\delta_1, \delta_2, \delta_3} \frac{1}{(-2z_1 \cdot z_2)^{\frac{\delta_1 + \delta_2 - \delta_3}{2}} (-2z_1 \cdot z_3)^{\frac{\delta_1 + \delta_3 - \delta_2}{2}} (-2z_2 \cdot z_3)^{\frac{\delta_2 + \delta_3 - \delta_1}{2}}},$ (3.207)

where

$$D_{\delta_1,\delta_2,\delta_3} = \frac{\pi^{\frac{d-2}{2}} \Gamma(\frac{\delta_1 + \delta_2 + \delta_3 - d+2}{2})}{2\Gamma(\delta_1)\Gamma(\delta_2)\Gamma(\delta_3)} \Gamma(\frac{\delta_1 + \delta_2 - \delta_3}{2}) \Gamma(\frac{\delta_1 + \delta_3 - \delta_2}{2}) \Gamma(\frac{\delta_2 + \delta_3 - \delta_1}{2}).$$
(3.208)

So, we have

$$I_{\text{strong}}^{(0)}(\delta,0;\delta') = \frac{8}{\pi^3} \frac{2^{d-2} \operatorname{vol}(\operatorname{SO}(d-2))}{B_{\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta,0}} B_{\widetilde{\mathcal{P}}_{\delta,0}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta'}}} C_{3,3;\widetilde{\delta}} D_{\widetilde{\delta},3,\delta'} \times \int \frac{D^{d-2} z D^{d-2} z' D^{d-2} z_3}{\operatorname{vol}(\operatorname{SO}(d-1,1))} \langle \mathcal{P}_{\delta,0}(z) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_3) \mathcal{P}_{\widetilde{\delta}'}(z') \rangle \langle \mathcal{P}_{\mathcal{E}}(z_3) \mathcal{P}_{\widetilde{\delta}}(z) \mathcal{P}_{\delta'}(z') \rangle \\ = \frac{8}{\pi^3} \frac{2^{d-2} \operatorname{vol}(\operatorname{SO}(d-2))}{B_{\mathcal{P}_{\mathcal{E}}} \mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta,0}} C_{3,3;\widetilde{\delta}} D_{\widetilde{\delta},3,\delta'} \left(\langle \mathcal{P}_{\delta} \mathcal{P}_{\widetilde{\mathcal{E}}} \mathcal{P}_{\widetilde{\delta}'} \rangle, \langle \mathcal{P}_{\widetilde{\delta}} \mathcal{P}_{\mathcal{E}} \mathcal{P}_{\delta'} \rangle \right).$$
(3.209)

Plugging in the explicit expressions using (3.195), (3.205) and (3.208), we finally obtain (after setting d = 4)

$$I_{\text{strong}}^{(0)}(\delta,0;\delta') = \frac{(1-\delta')\Gamma\left(3-\frac{\delta}{2}\right)\Gamma\left(\frac{\delta}{2}\right)^2\Gamma\left(\frac{\delta+4}{2}\right)\Gamma\left(\frac{-\delta-\delta'+5}{2}\right)\Gamma\left(\frac{-\delta+\delta'-1}{2}\right)\Gamma\left(\frac{-\delta+\delta'+3}{2}\right)\Gamma\left(\frac{\delta+\delta'+1}{2}\right)}{4\pi^5\Gamma(1-\delta)\Gamma(\delta-1)\Gamma(\delta'-1)}.$$
(3.210)

Consequently, $C_{\text{strong}}^{(0)}(\delta, 0; \delta')$ is given by

$$C_{\text{strong}}^{(0)}(\delta,0;\delta') = \frac{\Gamma\left(3-\frac{\delta}{2}\right)\Gamma\left(\frac{\delta}{2}\right)^{2}\Gamma\left(\frac{\delta+4}{2}\right)\Gamma\left(\frac{\delta-\delta'+3}{2}\right)\Gamma\left(\frac{-\delta+\delta'+3}{2}\right)\Gamma\left(\frac{\delta+\delta'-3}{2}\right)\Gamma\left(\frac{\delta+\delta'+1}{2}\right)}{4\pi^{3}\Gamma(\delta-1)\Gamma(\delta)\Gamma(\delta'-1)}.$$
 (3.211)

To find the coefficients for the celestial block expansion, we have to close the δ and δ' contours in (3.198). It turns out that the resulting celestial block expansion does not depend on the order of contour deformations, so let us close the δ' contour first for simplicity. When δ is on the principal series, the only poles of δ' that are to the right of the principal series are at $\delta' = \delta + 3 + 2k$, where k is a nonnegative integer. We then find that the residues

$$\operatorname{Res}_{\delta'=\delta+3+2k} C^{(0)}_{\mathrm{strong}}(\delta,0;\delta')$$
(3.212)

have poles at $\delta = 6 + 2n$, where *n* is a nonnegative integer. Thus, the values of δ and δ' appearing in the celestial block expansion (3.183) should be

$$\delta = 6 + 2n, \delta' = 9 + 2n + 2k. \tag{3.213}$$

This agrees with our previous predictions (3.185) and (3.186) from the light-ray OPE. The coefficients $P_{\delta,j;\delta'}$ are given by

$$P_{6+2n,0;9+2n+2k}^{(0)} = \frac{(k+1)(k+2)(-1)^{k+n}\Gamma(n+3)^2\Gamma(n+5)\Gamma(k+2n+6)\Gamma(k+2n+8)}{\pi^3\Gamma(n+1)\Gamma(2n+5)\Gamma(2n+6)\Gamma(2(k+n+4))}.$$
 (3.214)

As a consistency check, we can expand $\mathcal{F}_{\text{strong}}^{(0)}$ in the collinear limit and find the celestial block expansion order by order using the expansion of the celestial block G^c in the collinear limit given by (3.33) (or (B.18)). We have checked up to $O(\zeta_{13}^6)$ that the coefficients obtained in this way agree with (3.214).

3.6.3 $O(1/\lambda)$ correction

Let us now consider the $O(1/\lambda)$ term of (3.182),

$$\mathcal{F}_{\text{strong}}^{(1)}(z_1, z_2, z_3, p) = \frac{48}{\pi (-2p \cdot z_1)^3 (-2p \cdot z_2)^3 (-2p \cdot z_3)^3} \left(2 - 4 \left(\zeta_{12} + \zeta_{13} + \zeta_{23}\right) + 4 \left(\zeta_{12}^2 + \zeta_{13}^2 + \zeta_{23}^2\right) \right).$$
(3.215)

Plugging this into the celestial inversion formula, we get

$$\begin{split} I_{\text{strong}}^{(1)}(\delta, j; \delta') &= \frac{48}{\pi} \frac{2^{d-2} \operatorname{vol}(\mathrm{SO}(d-2))}{B_{\mathcal{P}_{\mathcal{E}}} \mathcal{P}_{\mathcal{E}} \mathcal{P}_{\delta,j}} \times \int \frac{D^{d-2} z D^{d-2} z' D^{d-2} z_1 D^{d-2} z_2 D^{d-2} z_3 d_{\text{AdS}}^{d-1} p}{\operatorname{vol}(\mathrm{SO}(d-1,1))} \\ &= \frac{\left(2 - 4 \left(\zeta_{12} + \zeta_{13} + \zeta_{23}\right) + 4 \left(\zeta_{12}^2 + \zeta_{13}^2 + \zeta_{23}^2\right)\right)}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3 (-2z_3 \cdot p)^3} \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta,j}(z) \rangle \langle \mathcal{P}_{\delta,j}(z) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_3) \mathcal{P}_{\widetilde{\delta}'}(z') \rangle \frac{1}{(-2z' \cdot p)^3} \right) \\ &= \frac{(3.216)}{(3.216)} \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_3) \mathcal{P}_{\widetilde{\mathcal{E$$

There are two types of z_1, z_2 integrals to consider,²³

$$\int D^{d-2} z_1 D^{d-2} z_2 \frac{\zeta_{12}^k}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3} \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta,j}(z,w) \rangle,$$

$$\int D^{d-2} z_1 D^{d-2} z_2 \frac{\zeta_{13}^k}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3} \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta,j}(z,w) \rangle, \quad (3.217)$$

where k = 0, 1, 2. Since the dependence of the integral on w should be $([z, w] \cdot [\cdots])^j$, the first line is only nonzero when j = 0 and the second one is nonzero when

²³Note that in (3.216) we implicitly contract the indices of $\widetilde{\mathcal{P}}_{\delta,j}(z)$ and $\mathcal{P}_{\delta,j}(z)$, and in (3.217) we contract the indices with a polarization vector w.

 $j \leq k$. However, when j = 1, the three-point structure $\langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta,j}(z,w) \rangle$ is antisymmetric in $1 \leftrightarrow 2$, and the contribution from ζ_{13}^k and ζ_{23}^k will cancel. So, $I_{\text{strong}}^{(1)}(\delta, j; \delta')$ is nonzero for j = 0 and j = 2.

$$j = 0$$

We first consider the j = 0 case. The integral containing ζ_{12}^k can be computed by applying the differential operator $\partial_p \cdot \partial_p k$ times to (3.204). The result is

$$\int D^{d-2}z_1 D^{d-2}z_2 \frac{\zeta_{12}}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3} \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta}(z) \rangle = \frac{(\delta - 6)(\widetilde{\delta} - 6)}{36} C^{(0)}_{3,3;\widetilde{\delta}}(-2p \cdot z)^{-\widetilde{\delta}},$$
(3.218)

and

$$\int D^{d-2} z_1 D^{d-2} z_2 \frac{\zeta_{12}^2}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3} \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta}(z) \rangle$$
$$= \frac{(\delta - 6)(\delta - 8)(\widetilde{\delta} - 6)(\widetilde{\delta} - 8)}{2304} C^{(0)}_{3,3;\widetilde{\delta}}(-2p \cdot z)^{-\widetilde{\delta}}.$$
(3.219)

To compute the second type of integral in (3.217), which contains ζ_{13}^k or ζ_{23}^k , we can first study

$$z_1^{\mu} \langle \mathcal{P}_{\delta_1}(z_1) \mathcal{P}_{\delta_2}(z_2) \mathcal{P}_{\delta}(z) \rangle.$$
(3.220)

The factor z_1^{μ} can be viewed as a weight-shifting operator [39] that decreases the scaling dimension of $\mathcal{P}_{\delta_1}(z_1)$ by 1. One can performing crossing on (3.220) to make the weight-shifting operators act on $\mathcal{P}_{\delta}(z)$. We find

$$\begin{aligned} &(z_{3} \cdot z_{1}) \langle \mathcal{P}_{\delta_{1}}(z_{1}) \mathcal{P}_{\delta_{2}}(z_{2}) \mathcal{P}_{\delta}(z) \rangle \\ &= \frac{(\delta - \delta_{1} + \delta_{2})(4 - d + \delta - \delta_{1} + \delta_{2})}{2(d - 3 - \delta)(d - 2 - 2\delta)} (z_{3} \cdot z) \langle \mathcal{P}_{\delta_{1} - 1}(z_{1}) \mathcal{P}_{\delta_{2}}(z_{2}) \mathcal{P}_{\delta + 1}(z) \rangle \\ &+ \frac{1}{2 - d + d\delta - 2\delta^{2}} (z_{3} \cdot D_{z}) \langle \mathcal{P}_{\delta_{1} - 1}(z_{1}) \mathcal{P}_{\delta_{2}}(z_{2}) \mathcal{P}_{\delta - 1}(z) \rangle \\ &- \frac{\delta - \delta_{1} + \delta_{2}}{2(d - 4)(d - 3 - \delta)(\delta - 1)} (z_{3} \cdot \mathcal{D}_{z,w}^{0-}) \langle \mathcal{P}_{\delta_{1} - 1}(z_{1}) \mathcal{P}_{\delta_{2}}(z_{2}) \mathcal{P}_{\delta, 1}(z, w) \rangle, \quad (3.221) \end{aligned}$$

where D_z is the Todorov operator and $\mathcal{D}_{z,w}^{0-}$ is the weight-shifting operator that decreases spin by 1 defined in [39]. As discussed below (3.217), the $\mathcal{D}_{z,w}^{0-} \langle \mathcal{P}_{\delta_1-1}(z_1) \mathcal{P}_{\delta_2}(z_2) \mathcal{P}_{\delta,1}(z,w) \rangle$ term will vanish after integrating over z_1, z_2 .

After using (3.221) and a similar relation for ζ_{23} , we find

$$\int D^{d-2} z_1 D^{d-2} z_2 \frac{\zeta_{13} + \zeta_{23}}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3} \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta}(z) \rangle
= 2 \int D^{d-2} z_1 D^{d-2} z_2 \frac{-2z_{3\mu}}{(-2z_1 \cdot p)^4 (-2z_2 \cdot p)^3 (-2z_3 \cdot p)}
\left(\frac{\widetilde{\delta}^2}{4(\widetilde{\delta} - 1)^2} z^{\mu} \langle \mathcal{P}_{\delta_{\widetilde{\mathcal{E}}} - 1}(z_1) \mathcal{P}_{\mathcal{E}}(z_2) \mathcal{P}_{\widetilde{\delta} + 1}(z) \rangle - \frac{1}{2(\widetilde{\delta} - 1)^2} D_z^{\mu} \langle \mathcal{P}_{\delta_{\widetilde{\mathcal{E}}} - 1}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \mathcal{P}_{\widetilde{\delta} - 1}(z) \rangle \right)
= \frac{2}{-2z_3 \cdot p} \left(\frac{\widetilde{\delta}^2}{4(\widetilde{\delta} - 1)^2} (-2z_3 \cdot z) C_{4,3;\widetilde{\delta} + 1}^{(0)} (-2p \cdot z)^{-\widetilde{\delta} - 1} - \frac{1}{2(\widetilde{\delta} - 1)^2} (-2z_3 \cdot D_z) C_{4,3;\widetilde{\delta} - 1}^{(0)} (-2p \cdot z)^{-\widetilde{\delta} + 1} \right)
= -\frac{1}{3} C_{3,3;\widetilde{\delta}}^{(0)} (-2p \cdot z)^{-\widetilde{\delta}} \left(\frac{\widetilde{\delta} - 6}{2} - \widetilde{\delta} \zeta_{03} \right).$$
(3.222)

Similarly, for $\zeta_{13}^2 + \zeta_{23}^2$ we have

$$\int D^{d-2} z_1 D^{d-2} z_2 \frac{\zeta_{13}^2 + \zeta_{23}^2}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3} \langle \mathcal{P}_{\tilde{\mathcal{E}}}(z_1) \mathcal{P}_{\tilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta}(z) \rangle$$

= $\frac{1}{96(\tilde{\delta}-3)} C^{(0)}_{3,3;\tilde{\delta}} (-2p \cdot z)^{-\tilde{\delta}} \left(\tilde{\delta} \left(\tilde{\delta}^2 + 6(\tilde{\delta}-4)(\tilde{\delta}+2)\zeta_{03}^2 - 2(\tilde{\delta}-6)(3\tilde{\delta}-4)\zeta_{03} - 18\tilde{\delta}+104 \right) - 192 \right).$
(3.223)

Combining (3.218), (3.219), (3.222), (3.223), we obtain

$$\int D^{d-2} z_1 D^{d-2} z_2 \mathcal{F}_{\text{strong}}^{(1)}(z_1, z_2, z_3, p) \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta}(z) \rangle$$

= $\frac{48}{\pi} \frac{(\widetilde{\delta} - 4)\widetilde{\delta}(\widetilde{\delta} + 2)}{576(\widetilde{\delta} - 3)} C_{3,3;\widetilde{\delta}}^{(0)} (-2p \cdot z)^{-\widetilde{\delta}} \left(\widetilde{\delta}^2 - 5\widetilde{\delta} + 30 + 144(\zeta_{03} - 1)\zeta_{03}\right).$
(3.224)

Hence, the inversion formula (3.216) for j = 0 is now given by

$$\begin{split} I_{\text{strong}}^{(1)}(\delta,0;\delta') \\ &= \frac{48}{\pi} \frac{2^{d-2} \operatorname{vol}(\operatorname{SO}(d-2))}{B_{\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta,j}} B_{\widetilde{\mathcal{P}}_{\delta,j}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta'}}}{\frac{576(\widetilde{\delta}-3)}{576(\widetilde{\delta}-3)}} C_{3,3;\widetilde{\delta}}^{(0)} \times \int \frac{D^{d-2} z D^{d-2} z' D^{d-2} z_3 d_{\text{AdS}}^{d-1} p}{\operatorname{vol}(\operatorname{SO}(d-1,1))} \\ &= \frac{1}{(-2p \cdot z)^{\widetilde{\delta}}(-2p \cdot z_3)^3} \left(\widetilde{\delta}^2 - 5\widetilde{\delta} + 30 + 144(\zeta_{03} - 1)\zeta_{03}\right) \langle \mathcal{P}_{\delta}(z) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_3) \mathcal{P}_{\widetilde{\delta}'}(z') \rangle \frac{1}{(-2z' \cdot p)^{\delta'}}. \end{split}$$

$$(3.225)$$

The integral over p can be evaluated using (3.207). The result is

$$\begin{split} &I_{\text{strong}}^{(1)}(\delta,0;\delta') \\ &= 6\pi^2 A \frac{2^{d-2} \operatorname{vol}(\operatorname{SO}(d-2))}{B_{\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta,j}} B_{\widetilde{\mathcal{P}}_{\delta,j}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta'}}}{576(\widetilde{\delta}-3)} C_{3,3;\widetilde{\delta}}^{(0)} D_{\widetilde{\delta},3,\delta'} \\ &\times \frac{\frac{3}{4}(\widetilde{\delta}-\delta'+3)(\widetilde{\delta}-\delta'+5)(\widetilde{\delta}+\delta'+1)(\widetilde{\delta}+\delta'+3) + (\widetilde{\delta}+1)((\widetilde{\delta}-18)\widetilde{\delta}(\widetilde{\delta}+1)+12(\delta'-3)(\delta'+1))}{\widetilde{\delta}(\widetilde{\delta}+1)} \\ &\times \left(\langle \mathcal{P}_{\delta}\mathcal{P}_{\widetilde{\mathcal{E}}}\mathcal{P}_{\widetilde{\delta}'} \rangle, \langle \mathcal{P}_{\widetilde{\delta}}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta'} \rangle \right), \end{split}$$
(3.226)

which implies that

(1)

$$\begin{split} C_{\text{strong}}^{(1)}(\delta,0;\delta') &= \left(\delta^2 \left(-6\delta'^2 + 12\delta' + 10\right) + 7\delta^4 - 28\delta^3 + 12\delta \left(\delta'^2 - 2\delta' + 3\right) + 3(\delta' - 1)^2 \left(\delta'^2 - 2\delta' - 3\right)\right) \\ \times (\delta - 4)(\delta - 1)(\delta + 2)(\delta' - 1) \frac{\Gamma \left(3 - \frac{\delta}{2}\right) \Gamma \left(\frac{\delta}{2}\right)^2 \Gamma \left(\frac{\delta + 4}{2}\right) \Gamma \left(\frac{\delta - \delta' + 3}{2}\right) \Gamma \left(\frac{-\delta + \delta' + 3}{2}\right) \Gamma \left(\frac{\delta + \delta' - 3}{2}\right) \Gamma \left(\frac{\delta + \delta' + 1}{2}\right)}{1536\pi (\delta - 3)(\delta + 1) \Gamma (\delta)^2 \Gamma (\delta')} \\ &\qquad (3.227) \end{split}$$

We see that the Gamma functions are the same as the leading order result $C_{\text{strong}}^{(0)}$, and hence the locations of the poles are the same. In particular, we only get poles at

$$\delta = 6 + 2n, \quad \delta' = 9 + 2n + 2k, \tag{3.228}$$

where *n*, *k* are nonnegative integers. The coefficients $P_{\delta,0;\delta'}^{(1)}$ are given by the residues,

$$\begin{split} P_{6+2n,0;9+2n+2k}^{(1)} &= \left(3k^2\left(4n^2+38n+83\right)+12k^3(n+4)+3k^4+6k\left(6n^2+43n+76\right)+4n^4+40n^3+169n^2+381n+378\right)\right)\\ &\times \frac{(-1)^{k+n}(n+1)(n+4)(2n+5)(k+n+4)\Gamma(k+3)\Gamma(n+3)^2\Gamma(n+5)\Gamma(k+2n+6)\Gamma(k+2n+8)}{3\pi(2n+3)(2n+7)\Gamma(k+1)\Gamma(n+1)\Gamma(2n+6)^2\Gamma(2k+2n+9)}\right). \end{split}$$

We have checked that the coefficients obtained by expanding $\mathcal{F}_{\text{strong}}^{(1)}$ in the collinear limit up to $O(\zeta_{13}^6)$ agree with the above expression.

Now let us consider the j = 2 case. From the discussion below (3.217), we must only consider the z_1, z_2 -integral

$$\int D^{d-2} z_1 D^{d-2} z_2 \frac{\zeta_{13}^2 + \zeta_{23}^2}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3} \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta,j=2}(z,w) \rangle. \quad (3.230)$$

We can again use crossing equations for weight-shifting operators. Only terms with a j = 0 three-point structure will be nonvanishing after we integrate over z_1 and z_2 . There is only one such term for $(z_3 \cdot z_1)^2 \langle \mathcal{P}_{\delta_1} \mathcal{P}_{\delta_2} \mathcal{P}_{\delta,2}(z,w) \rangle$, and its 6j symbol is given by²⁴

$$(z_{3} \cdot z_{1})^{2} \langle \mathcal{P}_{\delta_{1}} \mathcal{P}_{\delta_{2}} \mathcal{P}_{\delta,2}(z,w) \rangle = \frac{(\delta - \delta_{1} + \delta_{2} - d + 2)(\delta - \delta_{1} + \delta_{2} - d + 4)}{2(d - 2)d\delta(\delta + 1)(\delta - d + 1)(\delta - d + 2)} (z_{3} \cdot \mathcal{D}_{z,w}^{0+}) \langle \mathcal{P}_{\delta_{1}-2}(z_{1})\mathcal{P}_{\delta_{2}}(z_{2})\mathcal{P}_{\delta}(z) \rangle + \dots$$
(3.231)

Our z_1, z_2 integral is thus given by

$$\int D^{d-2} z_1 D^{d-2} z_2 \frac{\zeta_{13}^2 + \zeta_{23}^2}{(-2z_1 \cdot p)^3 (-2z_2 \cdot p)^3} \langle \mathcal{P}_{\widetilde{\mathcal{E}}}(z_1) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_2) \widetilde{\mathcal{P}}_{\delta,2}(z,w) \rangle$$

= $\frac{\widetilde{\delta}}{2(\widetilde{\delta}-3)} C^{(0)}_{5,3;\widetilde{\delta}} \frac{(2z \cdot pw \cdot z_3 - 2z \cdot z_3w \cdot p)^2}{(-2z \cdot p)^{\widetilde{\delta}+2} (-2z_3 \cdot p)^2}.$ (3.232)

Plugging this into the inversion formula (3.216) for j = 2, we find

$$I_{\text{strong}}^{(1)}(\delta, 2; \delta') = \frac{48}{\pi} \frac{2^{d-2} \operatorname{vol}(\operatorname{SO}(d-2))}{B_{\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta,2}} B_{\widetilde{\mathcal{P}}_{\delta,2}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta'}}}{\frac{\delta}{2(\delta-3)}} C_{5,3;\widetilde{\delta}}^{(0)} \times \int \frac{D^{d-2} z D^{d-2} z' D^{d-2} z_3 d_{\text{AdS}}^{d-1} p}{\operatorname{vol}(\operatorname{SO}(d-1,1))} \\ \frac{(2z \cdot p)^2 z_{3\mu} z_{3\nu} - 2(2z \cdot p)(2z \cdot z_3) z_{3\mu} p_{\nu} + (2z \cdot z_3)^2 p_{\mu} p_{\nu}}{(-2z \cdot p)^{\widetilde{\delta}+2}(-2z_3 \cdot p)^5} \langle \mathcal{P}_{\widetilde{\delta},2}^{\mu\nu}(z) \mathcal{P}_{\widetilde{\mathcal{E}}}(z_3) \mathcal{P}_{\widetilde{\delta}'}(z') \rangle \frac{1}{(-2z' \cdot p)^{\delta'}}.$$
(3.233)

To perform the p integral, we can view p_{μ} as a linear combination of the AdS weight-shifting operators [52] and then perform crossing. We find

$$p_{\mu}(-2p \cdot z)^{-\delta} = \frac{1}{(\delta - 1)(d - 2 - 2\delta)} D_{z\mu}(-2p \cdot z)^{-\delta + 1} + \frac{2\delta}{2\delta - d + 2} z_{\mu}(-2p \cdot z)^{-\delta - 1}$$
(3.234)

After using this relation, the integral over p is elementary. The result is

$$I_{\text{strong}}^{(1)}(\delta,2;\delta') = \frac{48}{\pi} \frac{2^{d-2} \operatorname{vol}(\operatorname{SO}(d-2))}{B_{\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta,2}} B_{\widetilde{\mathcal{P}}_{\delta,2}\mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta'}}} \frac{(\widetilde{\delta}+\delta'-5)(\widetilde{\delta}+\delta'-3)}{8(\widetilde{\delta}+1)(\widetilde{\delta}-3)} C_{5,3;\widetilde{\delta}}^{(0)} D_{\widetilde{\delta},5,\delta'} \left(\langle \mathcal{P}_{\mathcal{E}}\mathcal{P}_{\delta'}\widetilde{\mathcal{P}}_{\delta,2} \rangle, \langle \widetilde{\mathcal{P}}_{\mathcal{E}}\widetilde{\mathcal{P}}_{\delta'}\mathcal{P}_{\delta,2} \rangle \right)$$

$$(3.235)$$

²⁴Our convention here is $\langle \mathcal{P}_{\delta_1} \mathcal{P}_{\delta_2} \mathcal{P}_{\delta_1} j(z,w) \rangle = \frac{(2z \cdot z_1 w \cdot z_2 - 2z \cdot z_2 w \cdot z_1)^j}{(-2z_1 \cdot z_2)^{\frac{\delta_1 + \delta_2 - \delta_+ j}{2}} (-2z_1 \cdot z)^{\frac{\delta_1 + \delta_2 - \delta_+ j}{2}} (-2z_2 \cdot z)^{\frac{\delta_2 + \delta_- \delta_1 + j}{2}}}.$

Collecting together all the factors, we finally obtain

$$C_{\text{strong}}^{(1)}(\delta, 2; \delta') = \frac{\Gamma\left(4 - \frac{\delta}{2}\right)\Gamma\left(\frac{\delta}{2}\right)\Gamma\left(\frac{\delta+2}{2}\right)\Gamma\left(\frac{\delta+6}{2}\right)\Gamma\left(\frac{\delta-\delta'+5}{2}\right)\Gamma\left(\frac{-\delta+\delta'+5}{2}\right)\Gamma\left(\frac{\delta+\delta'-1}{2}\right)\Gamma\left(\frac{\delta+\delta'+3}{2}\right)}{24\pi\Gamma(\delta)\Gamma(\delta+2)\Gamma(\delta'-1)}.$$
(3.236)

After contour deformations, the only poles that will contribute to the celestial block expansion are at

$$\delta = 8 + 2n, \qquad \delta' = \delta + 5 + 2k = 13 + 2n + 2k. \tag{3.237}$$

The corresponding coefficients are given by

$$P_{k+2n,2;13+2n+2k}^{(1)} = \frac{(-1)^{k+n}\Gamma(k+5)\Gamma(n+5)\Gamma(n+7)\Gamma(n+4)\Gamma(k+2n+10)\Gamma(k+2n+12)}{6\pi\Gamma(k+1)\Gamma(n+1)\Gamma(2(n+5))\Gamma(2n+8)\Gamma(2(k+n+6))}.$$
(3.238)

We have checked that this result agrees with the expansion of $\mathcal{F}_{\text{strong}}^{(1)}$ in the collinear limit up to $O(\zeta_{13}^6)$. The validity of these expressions for the celestial block expansion of the strong-coupling EEEC is a strong check on both our expressions for the collinear expansion of celestial blocks (3.33), and the celestial inversion formula (3.196).

3.7 Discussion and future directions

In this work, we studied aspects of the celestial block decomposition of the threepoint energy correlator (EEEC). We found that, in both strong and weak coupling examples, the EEEC admits an expansion in a discrete sum of celestial blocks, corresponding to light-ray operators appearing in repeated OPEs of energy detectors. We derived useful formulas for 3-point celestial blocks in an expansion around the collinear limit. We then explored the celestial block expansion in the collinear limit using lightcone bootstrap techniques and the Lorentzian inversion formula, both in QCD and N = 4 SYM. The symmetry structure of the celestial block expansion allowed us to make certain predictions for higher orders in perturbation theory. We also determined the leading and first subleading contact terms in the EEEC in N = 4 SYM. Finally, using techniques from harmonic analysis, we studied the full celestial block decomposition of the strong-coupling EEEC in N = 4 SYM, for generic configurations of detectors. Along the way, we encountered several puzzles and novel objects that we summarize below.

This celestial block expansion must be compatible with crossing symmetry, which leads to constraints similar to those studied in the traditional conformal bootstrap. It would be interesting to fully characterize the consistency conditions that multi-point energy correlators should satisfy, in order to set up a direct bootstrap program for them (and more general event shapes). It would also be interesting to incorporate "generalized detectors" [71, 72].

In the collinear EEEC, the density matrix $|\Psi\rangle\langle\Psi|$ in which we evaluate the event shape gets highly boosted, and essentially projected onto its component with boost eigenvalue δ'_* (the dimension of the lowest-twist spin-4 operator). These highlyboosted density matrices, which we might call "light-ray density matrices," are naturally dual to light-ray operators, via taking expectation values. It is interesting to ask whether they provide a useful basis for studying other physical observables. It may also be interesting to explore information-theoretic properties of light-ray density matrices.

Our lightcone bootstrap analysis of collinear event shapes reveals the existence of two important contributions to the light-ray OPE: double-twist operators built out of a pair of detectors $[\mathcal{P}_O \mathcal{P}_O]$, and double-twist objects $[\mathcal{P}_O \mathcal{P}_{O'_*}]$ that formally look like an OPE between a light-ray operator and the light-ray density matrix obtained from boosting $|\Psi\rangle\langle\Psi|$. This suggests that it may be possible more generally to make sense of an OPE between light-ray operators and light-ray density matrices.

In our analysis of the EEEC in QCD, we used the fact that QCD admits a conformal point at a particular value of the coupling in $d = 4 - \epsilon$ dimensions. The existence of this conformal point has implications for the structure of perturbation theory at each loop order, even away from the conformal point, allowing us to apply selection rules from conformal symmetry in our work. However, the existence of a celestial block expansion should require only Lorentz invariance. It would be interesting to characterize how the structure of the space of light-ray operators and light-ray OPEs changes in the presence of a nonzero β -function.

In applying the Lorentzian inversion formula to the collinear EEEC, we were led to continue the EEEC into a "doubly-Lorentzian" regime where the cross-ratios on the celestial sphere become independent real numbers. (This same analytic continuation to (a cover of) the celestial torus is frequently used in studying celestial amplitudes

[97].) Although this continuation is straightforward at the order in perturbation theory we studied, we have little understanding of whether it is admissible at higher orders or nonperturbatively. Another important question is how event shapes behave in the "celestial Regge limit" where the celestial cross ratios undergo the analytic continuation usually studied in the context of the Regge limit in CFT [98]. The two examples we studied (weakly-coupled N = 4 SYM and QCD) exhibited very different behavior in this regime, and it would be interesting to understand what the general nonperturbative behavior can be.

Evidently, Lorentzian QFT observables still hold many mysteries. To better understand their structure, it will be important to study more examples and collect more data both in perturbation theory and beyond.

Acknowledgements

We thank Hao Chen, Lance Dixon, Murat Kologlu, Petr Kravchuk, Ian Moult, Joshua Sandor, Sasha Zhiboedov, and Hua Xing Zhu for discussions. We thank Hao Chen, Ian Moult, and Hua Xing Zhu for sharing their Mathematica notebook with results for the weak-coupling collinear EEEC with us. CHC and DSD are supported by Simons Foundation grant 488657 (Simons Collaboration on the Nonperturbative Bootstrap) and a DOE Early Career Award under grant no. DE-SC0019085.

Chapter 4

SPINNING DISPERSIVE CFT SUM RULES AND BULK SCATTERING

This chapter is based on

 Cyuan-Han Chang, Yakov Landau, and David Simmons-Duffin. "Spinning dispersive CFT sum rules and bulk scattering". In: (Nov. 2023). arXiv: 2311.04271 [hep-th].

4.1 Introduction

Causality and unitarity imply constraints on the space of low-energy effective field theories (EFTs) [99, 100]. In a 2 \rightarrow 2 scattering process, causality and unitarity manifest as analyticity, Regge boundedness, and crossing symmetry of the S-matrix. Using these properties, one can derive dispersion relations that express the Wilson coefficients of a low-energy EFT in terms of data in the ultraviolet (UV), which have positive spectral density thanks to unitarity [101, 102]. By applying functionals (such as expanding around the forward limit) to dispersion relations, positivity of the UV spectral density can be used to obtain two-sided bounds on EFT Wilson coefficients, with the correct scaling in the EFT cutoff *M* expected by dimensional analysis. A systematic exploration of this approach was initiated recently in [103, 104, 105, 106, 107].

Including gravitational interactions introduces another layer of complexity, since graviton exchange produces divergences in the forward limit. This issue was overcome in [108] by considering functionals that measure the scattering amplitude at small impact parameter. This method has paved the way for deriving bounds on higher-derivative corrections to General Relativity in flat space [109, 110, 111, 112, 113].

Similar questions can be explored in Anti-de Sitter (AdS) space as well. Furthermore, via the AdS/CFT correspondence, these questions can be phrased — and potentially answered — in CFT language. For example, HPPS conjectured in [114] that a large-*N* CFT should have a local bulk EFT dual if its single trace spectrum has a large gap $\Delta_{gap} = MR_{AdS} \gg 1$ for J > 2, where R_{AdS} is the AdS radius and *M* is the cutoff scale of the bulk EFT. Significant progress towards establishing this conjecture using CFT techniques was made in e.g. [102, 115, 116, 117, 118, 9, 119, 120, 34]. Recently, part of the HPPS conjecture was established in [13] by building a dictionary between conformal bootstrap functionals and flat space dispersion relations. The authors of [13] considered a weakly-coupled EFT in AdS with a massless graviton and a scalar with mass $m_{\phi} = O(\frac{1}{R_{\text{AdS}}})$. Schematically, the low-energy effective action is

$$S = S_{\text{gravity}} + \int d^D x \sqrt{-g} \left(\frac{1}{2} \phi (\partial^2 - m_\phi^2) \phi + \sum_n g_n D^{2n} \phi^4 + \ldots \right), \qquad (4.1)$$

where S_{gravity} contains graviton interactions and will be given below. Using bootstrap methods, they derived bounds on the higher-derivative interactions g_n in (4.1) with the expected suppression in Δ_{gap} .

The main tool in this analysis is dispersive CFT sum rules [121, 122, 123, 124, 125, 56, 126, 127, 128], which have double zeros at the locations of most double-twist operators. Such sum rules allow one to separate out light double-trace contributions in holographic CFTs, and express the light contribution ($\Delta < \Delta_{gap}$) described by (4.1) in terms of a sum of heavy conformal blocks with positive coefficients. Remarkably, in a certain "flat space limit" that we review in section 4.2, dispersive CFT sum rules reduce to the flat space sum rules previously studied in [108]. As a result, flat space functionals can be uplifted to AdS and lead to bounds on the EFT couplings in (4.1).¹

Given the bounds on graviton interactions in flat space obtained recently [109, 110, 111, 112, 113], we would like to derive similar bounds on graviton interactions in AdS, in particular establishing HPPS for purely gravitational theories. More precisely, let us consider an EFT with only a massless graviton with the effective action

$$S_{\text{gravity}} = \frac{1}{16\pi G} \int d^D x \sqrt{-g} \left(-2\Lambda + R + \alpha_2 R^2 + \alpha_3 R^3 + \ldots \right), \qquad (4.2)$$

where $\Lambda = -(D-1)(D-2)/(2R_{AdS}^2)$. We would like to obtain bounds like

$$|\alpha_2| \le G \times \frac{\#}{\Delta_{\text{gap}}^2}.$$
(4.3)

On the CFT side, this means we must construct dispersive sum rules for four-point functions of stress tensors whose flat space limit agrees with flat space sum rules for gravitons.

¹The "uplifting" procedure only guarantees positivity of the functional in the "bulk point" regime discussed below. One must additionally check other regimes to derive a rigorous CFT bound.

Conceptually, this task is similar to the one undertaken in [13]. However, stress tensor four-point functions are technically more complicated due to the profusion of tensor structures, and we need to organize the calculation carefully. Our approach is to begin with "subtracted superconvergence" sum rules, which exploit the fact that the commutator of null-integrated operators on the same null plane should vanish [34]. In particular, we focus on the action of such sum rules on a heavy conformal block (with $\Delta > \Delta_{gap}$). This action is a spacetime integral that, in the flat space limit, localizes to a certain saddle configuration that we call the "scattering crystal." Thus, the computation of the heavy action turns into evaluating conformally-invariant structures at this saddle, which is straightforward for spinning operators. Our calculation gives a "spacetime integration" of the results of [13] for scalar operators.

The result of our saddle analysis has a simple interpretation in flat space. The inserted conformal blocks become flat space partial waves, and the subtracted superconvergence sum rule becomes a sum rule for flat space "shock" amplitudes [102, 34]. Thus, by comparing the action of CFT sum rules on heavy blocks to the action of flat-space sum rules on heavy states, we can deduce a concrete dictionary between dispersive CFT functionals and flat space sum rules.² Our sum rules for stress tensors become a subset of the known flat space sum rules for gravitons [110]. The full set of flat space sum rules for gravitons includes additional sum rules that cannot be expressed in terms of "shock amplitudes" (because they have different choices of external polarizations). We leave the problem of obtaining these additional sum rules using CFT techniques to future work. Our dictionary will be needed to convert the flat space functionals with positive action on partial waves into CFT functionals with positive action on heavy blocks.

This paper is organized as follows. In section 4.2, we briefly review superconvergence sum rules and the derivation of the scalar bounds in [13]. In section 4.3, we derive a simple formula for the flat space limit of superconvergence sum rules for scalars using a spacetime saddle-point analysis. We then generalize the formula to spinning operators. In section 4.4, we explain how this formula can be matched to flat space sum rules, and obtain the dictionary between spinning CFT sum rules and flat space sum rules for photons and gravitons. We conclude in section 4.5. We summarize our conventions in appendix C.1, and present an alternative derivation

²Once we know the dictionary, it follows that the contributions of light states must match between AdS and flat space as well, though we leave exploration of light states to future work.

of the flat space limit formula in appendix C.2. Technical details on matching the CFT sum rules to flat space are given in appendices C.3, C.4, and C.5.

4.2 **Review: superconvergence sum rules**

Let us start by reviewing the idea of superconvergence in CFT, and explain how it can be related to flat space dispersive sum rules. Along the way we will introduce notation that we use throughout this work.

4.2.1 The light transform and superconvergence

Subtracted superconvergence sum rules come from studying the commutator of two null-integrated operators on the same null plane [34]. For simplicity we first consider a four-point function of scalar operators. Let us choose lightcone coordinates $x = (u, v, \vec{y})$ with $x^2 = -uv + \vec{y}^2$. A subtracted superconvergence sum rule can be written explicitly as

$$\int_{-\infty}^{\infty} dv_1 \int_{-\infty}^{\infty} dv_3 f(v_1, v_3) \langle 0 | \phi_4[\phi_3(0, v_3, \vec{y}_3), \phi_1(0, v_1, \vec{y}_1)] \phi_2 | 0 \rangle = 0, \quad (4.4)$$

where ϕ_1, ϕ_3 are placed on the same null plane u = 0. This integral vanishes for a simple reason: since $x_{13}^2 = \vec{y}_{13}^2 > 0$, the two operators ϕ_1, ϕ_3 are spacelike separated in the entire integration range, and therefore their commutator must vanish.

To obtain a sum rule, we would like to separate the two orderings $\phi_3\phi_1$ and $\phi_1\phi_3$ and perform the integral separately for each ordering. However, in order for this to be valid, we must check that each integral converges — in particular that there is no divergence from the endpoints of the integration contours. See [34] for a detailed analysis of this convergence condition. In the end, convergence can be achieved by a suitable choice of the function $f(v_1, v_3)$, which we call a "subtraction" factor. Expressions for subtraction factors will be given in section 4.3.

Our goal in this section is to rewrite (4.4) as a conformally-invariant spacetime integral, using tools from [11]. First, let us introduce index-free notation. For an ordinary integer-spin operator O, we contract the indices with a null polarization vector z:

$$O(x,z) = O^{\mu_1 \cdots \mu_J} z_{\mu_1} \cdots z_{\mu_J}, \quad z^2 = 0.$$
(4.5)

More generally, for a representation ρ represented by a Young diagram with more than one row, we introduce polarizations z, w, \tilde{w}, \ldots for each row of the Young diagram. They should be null and mutually orthogonal. Due to antisymmetry

between indices in different rows, we have gauge redundancies $w \sim w + \#z$, $\tilde{w} \sim \tilde{w} + \#w + \#z$, etc. The number of boxes in each row of the Young diagram encodes the homogeneity of the corresponding polarization vector. Generalizing to arbitrary homogeneity in the polarization vector *z* allows us to describe representations with continuous spin, which will be crucial in later calculations.³

Index-free notation can be viewed as the embedding space formalism [129, 30] for the Lorentz group SO(d-1, 1). In the embedding space of the d-dimensional conformal group SO(d, 2), one promotes the position x and polarization z to $X, Z \in \mathbb{R}^{d,2}$ satisfying $X^2 = X \cdot Z = Z^2 = 0$ and $Z \sim Z + \#X$. Then, conformal transformations act linearly on X, Z, and conformally-invariant structures are simply built from dot products of X and Z. We can recover the Minkowski space operator by the dictionary

$$O(x, z) = O(X = (1, x^2, x^{\mu}), Z = (0, 2x \cdot z, z^{\mu})),$$
(4.6)

where we again use lightcone coordinates $X = (X^+, X^-, X^\mu)$, and $X^2 = -X^+X^- + X^\mu X_\mu$. We will often go between embedding space and Minkowski space when writing expressions for conformally-invariant structures. In (C.6), we review the embedding space description of operators with more complicated representations.

The embedding space vector X lives on a projective null cone in $\mathbb{R}^{d,2}$, which is topologically $S^1 \times S^{d-1}$. Correlation functions of Lorentzian CFTs live on the universal cover $\widetilde{\mathcal{M}}_d = \mathbb{R} \times S^{d-1}$, also called the Lorentzian cylinder. More precisely, CFT correlation functions on $\mathbb{R}^{d-1,1}$ can be analytically continued to their Lorentzian cylinder counterparts [130]. The Lorentzian cylinder is tiled by Poincaré patches, where each patch represents a Minkowski space $\mathbb{R}^{d-1,1}$. The conformal group acting on $\widetilde{\mathcal{M}}_d$ is the universal covering group $\widetilde{SO}(d, 2)$. There exists a symmetry \mathcal{T} such that for each point $p \in \widetilde{\mathcal{M}}_d$, all light-rays emanating from p will converge at $\mathcal{T}p$ in the next Poincaré patch. The action of \mathcal{T} on an operator is most easily described in the embedding space, where we have

$$\mathcal{T}O(X, Z_1, Z_2, \ldots)\mathcal{T}^{-1} = O(-X, -Z_1, -Z_2, \ldots).$$
 (4.7)

We will also use the notation $p^+ \equiv \mathcal{T}p, p^- \equiv \mathcal{T}^{-1}p$.

The null-integrated operators in (4.4) are an example of a conformally-invariant integral transform called the light transform [11]. The light transform of a local

³For the Lorentzian conformal group SO(d, 2), their unitary principal representations have two continuous parameters, Δ and J [11].



Figure 4.1: Contour of the light transform on the Lorentzian cylinder, where the two dashed lines should be identified. The contour starts at point x and ends at x^+ on the next Poincaré patch.

operator O with quantum numbers (Δ, J) is defined as

$$\mathbf{L}[O](x,z) = \int_{-\infty}^{\infty} d\alpha (-\alpha)^{-\Delta - J} O\left(x - \frac{z}{\alpha}, z\right).$$
(4.8)

Under conformal transformations, $\mathbf{L}[O](x, z)$ transforms like a primary operator at x. The light transform contour is shown in figure 4.1. It starts at the point x, goes along the direction of the polarization z, and eventually ends at the point $\mathcal{T}x = x^+$. The previous description of the null-integrated operator in (4.4), where one puts the operator at u = 0 and integrates along v, can be obtained by putting x at past null infinity.

In the embedding space, the light transform (4.8) becomes

$$\mathbf{L}[O](X,Z) = \int_{-\infty}^{\infty} d\alpha \ O(Z - \alpha X, -X).$$
(4.9)

From this expression, it is not hard to see that the quantum numbers of L[O] are $(1 - J, 1 - \Delta)$. The light transform of a local operator annihilates the vacuum [11]:

$$\mathbf{L}[O](x,z)|\Omega\rangle = 0. \tag{4.10}$$

With all this in place, we can now write the subtracted superconvergence sum rule

as a conformally-invariant spacetime integral. We define a functional $\Psi_{k,v}$ by⁴

$$\Psi_{k,\nu}[\mathcal{G}] \equiv \int \frac{d^d x_2 d^d x_4 d^d x_0 D^{d-2} z_1 D^{d-2} z_3}{\operatorname{vol}(\widetilde{\operatorname{SO}}(d,2))} K_{k,\nu}(x_2, x_4, x_0, z_1, z_3) \times \langle \Omega | \phi_4(x_4) [\mathbf{L}[\phi_3](x_0, z_3), \mathbf{L}[\phi_1](x_0, z_1)] \phi_2(x_2) | \Omega \rangle,$$
(4.11)

where $K_{k,v}$ is a kernel chosen such that the integral is conformally-invariant. Its explicit expression is given below in (4.33). The integral measure for the polarization $D^{d-2}z$ is defined as [46]

$$D^{d-2}z \equiv \frac{2d^d z \delta(z^2)\theta(z^0)}{\operatorname{vol} \mathbb{R}_+},\tag{4.12}$$

where \mathbb{R}_+ acts by rescaling *z*. Then, the superconvergence condition (4.4) is equivalent to

$$\Psi_{k,\nu}[(\text{subtractions}) \times \mathcal{G}] = 0, \qquad (4.13)$$

where G is a four-point function and (subtraction) corresponds to the factor $f(v_1, v_3)$ in (4.4) (we discuss them in more detail below).

Using (4.10), we can rewrite (4.11) as

$$\Psi_{k,\nu}[\mathcal{G}] = \int \frac{d^d x_2 d^d x_4 d^d x_0 D^{d-2} z_1 D^{d-2} z_3}{\operatorname{vol}(\widetilde{SO}(d,2))} K_{k,\nu}(x_2, x_4, x_0, z_1, z_3) \\ \times \langle \Omega | [\phi_4(x_4), \mathbf{L}[\phi_3](x_0, z_3)] [\mathbf{L}[\phi_1](x_0, z_1), \phi_2(x_2)] | \Omega \rangle - (1 \leftrightarrow 3).$$

$$(4.14)$$

When we replace the four-point function \mathcal{G} with an s-channel conformal block $G_{\Delta,J}^{s}$, the double commutator $\langle \Omega | [\phi_4, \mathbf{L}[\phi_3]] [\mathbf{L}[\phi_1], \phi_2] | \Omega \rangle$ will give a $\sin^2(\pi \frac{\Delta - J - 2\Delta_{\phi}}{2})$ factor. For double-twist operators $\Delta - J = 2\Delta_{\phi} + 2n$, the \sin^2 factor then becomes a double zero. Therefore, $\Psi_{k,\nu}$ is a dispersive functional, meaning that it gives double zeros for all but finitely many double-twist operators.⁵

4.2.2 The flat space limit of superconvergence sum rules

Let us now review the argument in [13] that leads to bounds on the scalar AdS EFT (4.1). As explained in the introduction, the idea is that in a certain "flat space limit," dispersive CFT sum rules reduce to flat space sum rules.

⁴This functional is equivalent to the $\tilde{C}_{k,\nu}$ functional defined in the appendix of [13].

⁵When there is a subtraction factor, the \sin^2 factor gets modified and can become nonzero at finitely many double twist locations [56].

Review: flat space sum rules

Let us first briefly review flat space sum rules for identical real scalars. Flat space sum rules for photons and gravitons will be reviewed in section 4.4.

Let $\mathcal{M}(s, u)$ be a 2 \rightarrow 2 scattering amplitude for identical scalars, which we assume satisfies analyticity, crossing symmetry, and Regge boundedness. This allows one to write down flat space dispersion relations for $\mathcal{M}(s, u)$ (see e.g., [104, 108] for more details),

$$-\oint_{\infty} \frac{ds}{2\pi i} \frac{1}{s} \frac{\mathcal{M}(s,u)}{(s(s+u))^{\frac{k}{2}}} = 0, \qquad (4.15)$$

where s, u are the Mandelstam variables,

$$s = -(p_1 + p_2)^2$$
, $t = -(p_2 + p_3)^2$, $u = -(p_1 + p_3)^2$. (4.16)

The contour in (4.15) can be deformed and separated into two parts: the low energy part, which can be computed using an EFT, and the high energy part, which can be decomposed into partial waves.

The partial wave decomposition of a scalar amplitude $\mathcal{M}(s, u)$ is given by

$$\mathcal{M}(s,u) = s^{\frac{4-D}{2}} \sum_{J} n_J^{(D)} a_J(s) \mathcal{P}_J\left(1 + \frac{2u}{s}\right). \tag{4.17}$$

Here, $n_{\rho}^{(D)} = \frac{2^{d+1}(2\pi)^{d-1}\dim\rho}{\operatorname{vol} S^{d-1}}$, where in our case ρ is the spin-*J* traceless symmetric tensor representation of SO(*d*), and d = D - 1. Meanwhile \mathcal{P}_J is a Gegenbauer polynomial,

$$\mathcal{P}_J(x) = {}_2F_1(-J, J+d-2, \frac{d-1}{2}, \frac{1-x}{2}).$$
(4.18)

After the partial wave decomposition, the high energy part of the the dispersion relation (4.15) becomes

$$\left\langle \frac{2s+u}{s+u} \frac{\mathcal{P}_J(1+\frac{2u}{s})}{(s(s+u))^{\frac{k}{2}}} \right\rangle,\tag{4.19}$$

where $\langle \cdots \rangle$ is a heavy average, defined as a sum with positive coefficients over heavy states with mass *m* and spin *J*,

$$\langle \cdots \rangle = \frac{1}{\pi} \sum_{J} n_{J}^{(D)} \int_{M^{2}}^{\infty} \frac{dm^{2}}{m^{2}} m^{4-D} \operatorname{Im} a_{J}(s) (\cdots) .$$
 (4.20)

Unitarity implies that the spectral density $\text{Im } a_J(s)$ should be positive (or more generally a positive semidefinite matrix). Therefore, by applying a functional that has positive action on all partial wave contributions (4.19), one can derive inequalities for Wilson coefficients of the low energy EFT. Without gravity, the functional can simply include taking the forward limit $u \rightarrow 0$. For amplitudes with graviton exchange, one can integrate (4.19) over u against some kernel (see [108, 109, 110]).

The flat space limit

Consider a large-*N* CFT whose single-trace spectrum consists of a scalar operator ϕ , the stress tensor $T^{\mu\nu}$, and operators with twists $\tau = \Delta - J$ greater than $\Delta_{\text{gap}} \gg 1$. By the HPPS conjecture, this theory is expected to be dual to an EFT in AdS with the effective action (4.1). The s-channel conformal block decomposition of $\langle \phi \phi \phi \phi \rangle$ is given by

$$\langle \phi \phi \phi \phi \rangle = G_1^s + f_T^2 G_{T^{\mu\nu}}^s + \sum_{n,J} f_{[\phi\phi]_{n,J}}^2 G_{[\phi\phi]_{n,J}}^s + \sum_{\tau > \Delta_{gap}} f_{\Delta,J}^2 G_{\Delta,J}^s,$$
(4.21)

where f_O is the OPE coefficient, and $[\phi\phi]$ are double-trace operators built from ϕ . Consider now a dispersive functional ω whose action on the four-point function vanishes. Applying it to the conformal block decomposition gives us

$$-\omega|_{\text{light}} = \sum_{\tau > \Delta_{\text{gap}}} f_{\Delta,J}^2 \omega[G_{\Delta,J}^s], \qquad (4.22)$$

where

$$\omega|_{\text{light}} \equiv \omega[G_1^s] + f_T^2 \omega[G_{T^{\mu\nu}}^s] + \sum_{n,J} f_{[\phi\phi]_{n,J}}^2 \omega[G_{[\phi\phi]_{n,J}}^s].$$
(4.23)

One can argue that $\omega|_{\text{light}}$ is determined by the Wilson coefficients of the low-energy EFT in AdS. Therefore, if we can find a functional ω such that

$$\omega[G_{\Lambda,J}^s] \ge 0, \quad \tau > \Delta_{\text{gap}}, \tag{4.24}$$

then positivity of $f_{\Lambda,I}^2$ implies that

$$-\omega|_{\text{light}} \ge 0, \tag{4.25}$$

which becomes an inequality on EFT Wilson coefficients.

To study the action of ω on heavy blocks, it is useful to consider two special limits.

 The "bulk point"/"flat space" limit is the regime of large Δ with J/Δ ≪ 1. As explained in [13], this corresponds to an AdS scattering process where the energy *m* and impact parameter β are given by

$$m^2 \approx \Delta^2, \quad \beta \approx \frac{2J}{\Delta} \ll 1.$$
 (4.26)

The term "flat space" comes from the fact that the impact parameter β is much smaller than the AdS radius (which is 1 in our conventions).

The Regge limit is the regime of large Δ, J with fixed ratio Δ/J. This corresponds to an AdS scattering process with impact parameter comparable to the AdS radius.

A necessary condition for a functional ω to be positive is that it should be positive in the flat space regime. In this regime, one can relate the action of ω on conformal blocks to the action of simple flat-space dispersion relations. For example,

$$\Psi_{k,\nu;f_k}[\Delta, J] = \frac{2m^2 - \nu^2}{m^2 - \nu^2} \frac{\mathcal{P}_J(1 - \frac{2\nu^2}{m^2})}{(m^2(m^2 - \nu^2))^{\frac{k}{2}}} \left(1 + O\left(\frac{J^2}{m^2}\right)\right), \quad (4.27)$$

where \mathcal{P}_J is a Gegenbauer polynomial given by (4.18). On the left-hand side $\omega[\Delta, J]$ denotes the action of ω on a conformal block, rescaled by some positive factors:

$$\omega[\Delta, J] \equiv \frac{1}{q_{\Delta, J}} \frac{\omega[G_{\Delta, J}^{s}]}{2\sin^{2}(\pi \frac{\Delta - J - 2\Delta_{\phi}}{2})},$$
(4.28)

where

$$q_{\Delta,J} = \frac{1}{\pi p_{\Delta,J}^{\text{MFT}}} \frac{2n_J}{m^{2d-4}} \frac{\Gamma(\Delta-1)(2\Delta-d)}{\Gamma(\Delta-d+2)},$$

$$n_J = \frac{2^{d+1}\pi^{\frac{d-1}{2}}}{\Gamma(\frac{d-1}{2})} (J+1)_{d-3} (2J+d-2).$$
(4.29)

 $p_{\Delta,J}^{\text{MFT}}$ is the OPE coefficient of the Mean Field Theory, whose expression can be found in [131].

The functional $\Psi_{k,\nu;f_k}$ denotes a subtracted version of $\Psi_{k,\nu}$, where we insert an additional factor f_k into the integrand. As discussed below (4.4) this factor is needed to modify the behavior near the endpoints of the integral. For example, for k = 2, an appropriate subtraction factor is given by

$$f_{k=2}(u',v') = \frac{v'-u'}{u'v'},$$
(4.30)
where u', v' are conformally-invariant cross-ratios,

$$u' = \frac{x_{12}^2 x_{34}^2}{x_{13}^2 x_{24}^2}, \quad v' = \frac{x_{14}^2 x_{23}^2}{x_{13}^2 x_{24}^2}.$$
(4.31)

We give subtraction factors for general k in section 4.3.4.

The right hand side of (4.27) is exactly the contribution of a state with mass *m* and spin *J* to the flat space scalar sum rule (4.19) after identifying $s = m^2, u = -v^2$. This guarantees that the flat space functionals with positive action on (4.19) can be uplifted to AdS and will give us a CFT functional with positive action on all blocks in the bulk-point limit ($\Delta \gg 1$).

As explained in [13], to obtain bounds on the bulk EFT Wilson coefficients, one must also ensure that the functional ω is positive outside the bulk point limit, in particular in the Regge regime (and other regimes if necessary). We leave an exploration of the Regge regime for four-point functions of spinning operators to future work. The goal in this work is to derive the bulk-point limit result (4.27) and its generalization to spinning operators from a saddle analysis of the spacetime integral.

Before we proceed, let us comment on a subtlety regarding (4.27). In [13], (4.27) was derived (for a slightly different functional $C_{k,\nu}$ which is equivalent to $\Psi_{k,\nu}$ at large ν) for all $\nu \in [0, m)$. However, the derivation in this work will be in the limit where both ν and m are large (with fixed ratio). (Therefore, sometimes we will also refer to the bulk-point limit as the $\nu, m \gg 1$ limit.) In [13] it was shown that in the scalar case, the regimes of finite ν and large- ν are continuously connected at large m. In other words, we can safely compute at large ν , and then later consider all $\nu \in [0, m)$. We expect the same to be true in the spinning case as well.

4.3 Heavy action from a spacetime saddle-point analysis

In this section, we define our functional in position space, and study its action on a conformal block $G_{\Delta,J}$ with large scaling dimension Δ . We will show that in the bulk point limit $\nu, \Delta \gg 1$, the integral completely localizes to a saddle point, and the heavy action can be obtained by evaluating conformally-invariant structures at the saddle. This allows us to derive a simple formula for the functional in the bulk point limit. We will first use this formula to reproduce the results summarized in section 4.2, and then write down the generalized functional for spinning operators.

4.3.1 Functionals as spacetime integrals

As explained in section 4.2, our starting point for the position space functional is the fact that the commutator of light-transformed operators vanishes,

$$\langle \Omega | O_4(x_4) [\mathbf{L}[O_3](x_0, z_3), \mathbf{L}[O_1](x_0, z_1)] O_2(x_2) | \Omega \rangle = 0.$$
(4.32)

To make sure the light transform integrals $L[O_1]L[O_3]$ converge, we also need the condition $J_1 + J_3 - 1 > J_0$, where J_0 is the Regge intercept [34]. For the moment, let us assume this condition is satisfied, and we will come back to this issue later in section 4.3.4 and 4.3.5 by introducing subtraction factors.

We can integrate (4.32) against appropriate conformally-invariant structures to get different sum rules. We first consider the functional for scalar four-point functions. As we will see shortly, once we have derived this formula, the generalization to spinning operators will be obvious. For simplicity, we assume the external operators all have scaling dimension Δ_{ϕ} , but we will not impose that they are identical until section 4.3.4. Our functional is defined as

$$\Psi_{k,\nu}[\mathcal{G}] \equiv A_{k,\nu} \int_{\substack{2>4\\0\approx2,4}} \frac{d^d x_2 d^d x_4 d^d x_0 D^{d-2} z_0}{\operatorname{vol}(\widetilde{\operatorname{SO}}(d,2))} \langle 0|\phi_4(x_4^+) \mathbf{L}[\mathcal{O}](x_0,z_0)\phi_2(x_2)|0\rangle^{-1} \\ \int D^{d-2} z_1 D^{d-2} z_3 \langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_3}(z_3) \mathcal{P}_{\delta}(z_0) \rangle \langle \Omega|\phi_4(x_4^+) [\mathbf{L}[\phi_3](x_0,z_3), \mathbf{L}[\phi_1](x_0,z_1)]\phi_2(x_2)|\Omega\rangle$$

$$(4.33)$$

where the coefficient $A_{k,\nu}$ is given by

$$A_{k,\nu} = 2^{-8+4\Delta_{\phi}} \pi^{-2-\frac{d}{2}} e^{\pi\nu} \nu^{2+d-2k-4\Delta_{\phi}} \Gamma\left(\frac{d-2}{2}\right) \Gamma(\Delta_{\phi})^2 \Gamma\left(\Delta_{\phi} - \frac{d-2}{2}\right)^2.$$
(4.34)

The coefficient is chosen such that the heavy action of $\Psi_{k,v}$ agrees with the flat space sum rule. We have also applied \mathcal{T}_4 to the integrand for later convenience. The causality relations between the points are shown in figure 4.2a. In (4.33), 2 > 4 indicates that x_2 is in the future lightcone of 4, and $0 \approx 2, 4$ indicates that x_0 is spacelike separated from x_2 and x_4 .

Let us be explicit about the quantum numbers and structures in the definition (4.33). The operator O in the first line has scaling dimension $\Delta = \frac{d}{2} + i\nu$, and its spin is fixed by symmetry to be J = -1 [12, 15]. The structure $\langle 0|\phi_4(x_4^+)\mathbf{L}[O](x_0, z_0)\phi_2(x_2)|0\rangle^{-1}$ is a dual structure of a light transformed three-point function with respect to a Lorentzian three-point pairing,

$$\left(\langle O_1 O_2 O \rangle, \langle \widetilde{O}_1^{\dagger} \widetilde{O}_2^{\dagger} O^{S^{\dagger}} \rangle \right)_L$$

$$\equiv \int_{\substack{2 < 1 \\ x \approx 1, 2}} \frac{d^d x_1 d^d x_2 d^d x D^{d-2} z}{\operatorname{vol}(\widetilde{SO}(d, 2))} \langle O_1(x_1) O_2(x_2) O(x, z) \rangle \langle \widetilde{O}_1^{\dagger}(x_1) \widetilde{O}_2^{\dagger}(x_2) O^{S^{\dagger}}(x, z) \rangle.$$
(4.35)

The dual structure is defined by

$$\left(\langle 0|\phi_4(x_4^+)\mathbf{L}[O](x_0, z_0)\phi_2(x_2)|0\rangle^{-1}, \langle 0|\phi_4(x_4^+)\mathbf{L}[O](x_0, z_0)\phi_2(x_2)|0\rangle\right)_L = 1.$$
(4.36)

The explicit expression is [11]⁶

$$\langle 0|\phi_{4}(x_{4}^{+})\mathbf{L}[O_{\Delta,J}](x_{0},z_{0})\phi_{2}(x_{2})|0\rangle^{-1}$$

$$= \frac{2^{2d-2}\Gamma(\frac{\Delta+J}{2})^{2}\operatorname{vol}(\mathrm{SO}(d-2))}{2\pi i\Gamma(\Delta+J-2)} \frac{(2z_{0}\cdot x_{40}x_{20}^{2} - 2z_{0}\cdot x_{20}x_{40}^{2})^{\Delta-d+1}}{(-x_{24}^{2})^{\frac{\tilde{\lambda}_{2}+\tilde{\lambda}_{4}+\Delta-J-2d+2}{2}}(x_{02}^{2})^{\frac{\tilde{\lambda}_{2}+\Delta+J-\tilde{\lambda}_{4}}{2}}(x_{02}^{2})^{\frac{\tilde{\lambda}_{4}+\Delta+J-\tilde{\lambda}_{2}}{2}}.$$

$$(4.37)$$

In the second line of (4.33), $\langle \tilde{\mathcal{P}}_{\delta_1}(z_1) \tilde{\mathcal{P}}_{\delta_3}(z_3) \mathcal{P}_{\delta}(z_0) \rangle$ is a three-point structure of a fictitious (d-2)-dimensional Euclidean CFT on the celestial sphere, whose conformal symmetry is the Lorentz symmetry SO(d-1, 1) of the original CFT. The polarizations z_i should be viewed as embedding space coordinates of this CFT_{*d*-2}, and $\mathcal{P}_{\delta}(z)$ is a primary with scaling dimension δ on the celestial sphere. In order for the integral (4.33) to be conformally-invariant, we must have $\delta = \frac{d-2}{2} + i\nu$. We also use the notation $\tilde{\mathcal{P}}_{\delta_i} = \mathcal{P}_{d-2-\delta_i}$, and $\delta_i = \Delta_i - 1$, while $\tilde{\delta}_i = d - 2 - \delta_i$ is the shadow dimension with respect to the (d-2)-dimensional fictitious CFT. The celestial three-point structure is given by

$$\langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_3}(z_3) \mathcal{P}_{\delta}(z_0) \rangle = \frac{1}{(-2z_1 \cdot z_0)^{\frac{\delta + \widetilde{\delta}_1 - \widetilde{\delta}_3}{2}} (-2z_3 \cdot z_0)^{\frac{\delta + \widetilde{\delta}_3 - \widetilde{\delta}_1}{2}} (-2z_1 \cdot z_3)^{\frac{\widetilde{\delta}_1 + \widetilde{\delta}_3 - \delta}{2}}}.$$
(4.38)

Originally, the superconvergence sum rule (4.32) is parametrized by the positions and polarizations of the operators, subject to conformal invariance. By integrating it against the dual structure and celestial structure described above, we have

⁶Here, we choose the structures to be an analytic continuation from odd J, so the $(-1)^J$ factor in [11] becomes -1.



Figure 4.2: Causality configuration of the functional. The left figure shows the configuration in the original definition (4.33). Points 1, 3 both start from 0 and are integrated along two different null directions. The right figures shows the configuration after applying \mathcal{T}_1^{-1} , \mathcal{T}_2^{-1} , \mathcal{T}_4 , which becomes 4 > 3 > 0 > 1 > 2.

transformed the statement of superconvergence into ν -space, which parametrizes the quantum numbers of the structures. In fact, one can also show that the condition $\Psi_{k,\nu}[\mathcal{G}] = 0$ is equivalent to [15]

$$C^{-}(\frac{d}{2}+i\nu, J=-1)=0, \quad J_{0}<-1,$$
 (4.39)

where $C^{-}(\Delta, J)$ is the coefficient function computed by the Lorentzian inversion formula [9, 10], and encodes the analytic continuation of the CFT data from odd spin.

To further simplify the functional, let us focus on one term in the commutator $[\mathbf{L}[\phi_3], \mathbf{L}[\phi_1]]$ and consider

Note that since light-transformed operators annihilate the vacuum, we can rewrite the four-point function as a double commutator. Moreover, in (4.40), we can rewrite

the integral over the two light transform directions z_1, z_3 as an integral over the positions of ϕ_1, ϕ_3 with additional delta function constraints [77]. Explicitly, this gives

$$\begin{split} \Psi_{k,\nu}^{+}[\mathcal{G}] &= 4A_{k,\nu} \int_{\substack{2>4\\0\approx2,4}} \frac{d^{d}x_{1}d^{d}x_{2}d^{d}x_{3}d^{d}x_{4}d^{d}x_{0}D^{d-2}z_{0}}{\operatorname{vol}(\widetilde{\operatorname{SO}}(d,2))} \langle 0|\phi_{4}(x_{4}^{+})\mathbf{L}[O](x_{0},z_{0})\phi_{2}(x_{2})|0\rangle^{-1} \\ \langle \widetilde{\mathcal{P}}_{\delta_{1}}(x_{10})\widetilde{\mathcal{P}}_{\delta_{3}}(x_{30})\mathcal{P}_{\delta}(z_{0})\rangle\delta(x_{10}^{2})\delta(x_{30}^{2})\theta(x_{10})\theta(x_{30})\langle \Omega|[\phi_{4}(x_{4}^{+}),\phi_{3}(x_{3})][\phi_{1}(x_{1}),\phi_{2}(x_{2})]|\Omega\rangle, \\ (4.41) \end{split}$$

where the theta-functions $\theta(x_{10})\theta(x_{30})$ indicate that x_{10} and x_{30} must be timelike and future-pointing. Finally, let us make a change of variables $x_1 \rightarrow x_1^+, x_2 \rightarrow x_2^+, x_4 \rightarrow x_4^-$. This makes the causality configuration equivalent to 4 > 3 > 0 > 1 > 2 (see figure 4.2b), and we arrive at the expression

$$\Psi_{k,\nu}^{+}[\mathcal{G}] = 4A_{k,\nu} \int_{4>3>0>1>2} \frac{d^{d}x_{1}d^{d}x_{2}d^{d}x_{3}d^{d}x_{4}d^{d}x_{0}D^{d-2}z_{0}}{\operatorname{vol}(\widetilde{\mathrm{SO}}(d,2))} \langle \tilde{\mathcal{P}}_{\delta_{1}}(x_{1}+_{0})\tilde{\mathcal{P}}_{\delta_{3}}(x_{30})\mathcal{P}_{\delta}(z_{0})\rangle \delta(x_{1}^{2}+_{0})\delta(x_{30}^{2})\theta(x_{1}+_{0})\theta(x_{30})\langle \Omega|[\phi_{4}(x_{4}),\phi_{3}(x_{3})][\phi_{1}(x_{1}^{+}),\phi_{2}(x_{2}^{+})]|\Omega\rangle.$$

$$(4.42)$$

4.3.2 Action on conformal blocks

Let us now apply the functional $\Psi_{k,\nu}^+$ to an s-channel conformal block $G_{\Delta,J}^s$ and study the action in the large Δ limit. In position space, the most useful expression for the block is the Lorentzian shadow representation [2], given by integrating a fifth point x_5 over a causal diamond 4 > 5 > 3:

$$G_{\Delta,J}^{s} = \frac{1}{\beta_{\Delta,J}} \int_{4>5>3} d^{d}x_{5} |\langle \phi_{1}\phi_{2}O^{\mu_{1}\cdots\mu_{J}}(x_{5})\rangle||\langle \widetilde{O}_{\mu_{1}\cdots\mu_{J}}^{\dagger}(x_{5})\phi_{3}\phi_{4}\rangle|.$$
(4.43)

The external points should satisfy the causality constraint 1 > 2, 4 > 3 with all other pairs of points spacelike-separated. On the right-hand side, the operator O has quantum numbers (Δ, J) , and the quantum numbers of \tilde{O}^{\dagger} are related to those of O by a shadow transform and Hermitian conjugate:⁷

$$\widetilde{O} : (\Delta, \rho) \mapsto (d - \Delta, \rho^R),$$

$$O^{\dagger} : (\Delta, \rho) \mapsto (\Delta, \rho^{\dagger} = (\rho^R)^*).$$
(4.44)

In (4.43), we contract the indices of the exchanged operator O_5 . This is more natural for our purpose because the spin of the inserted block corresponds to spin of the

⁷In the case where ρ is a traceless symmetric tensor representation with spin *J*, we have $\rho^R = \rho = (\rho^R)^*$.

exchanged massive particle in flat space, and hence we are only interested in the integer-spin case.⁸ Moreover, later we will find a correspondence between blocks and flat space partial waves, which also come from contracting the indices of spin-*J* tensors. The shadow coefficient $\beta_{\Delta,J}$ in (4.43) in the scalar case is given by

$$\beta_{\Delta,J} = \frac{(-1)^J 2^{J-1} \pi^{\frac{d-2}{2}} \Gamma(\Delta + 2 - d) \Gamma(\frac{J+\Delta}{2})^2 \Gamma(\frac{\Delta + 2 - d - J}{2})^2}{\Gamma(\Delta + \frac{2 - d}{2}) \Gamma(J + \Delta) \Gamma(\Delta + 2 - d - J)}.$$
(4.45)

For the three-point structures in (4.43), the notation $|\langle ... \rangle|$ means that all the x_{ij} 's in the denominator should come with absolute values (see appendix C.1 for definitions). In our case, the correct interpretation of the absolute-valued structures is in terms of a double commutator,

$$\langle \Omega | [\phi_4(x_4), \phi_3(x_3)] [\phi_1(x_1), \phi_2(x_2)] | \Omega \rangle^s_{\Delta,J}$$

$$= \frac{-2(2\sin^2(\pi \frac{\Delta - J - 2\Delta_\phi}{2}))}{\beta_{\Delta,J}} \int_{4>5>3} d^d x_5 | \langle \phi_1 \phi_2 O^{\mu_1 \cdots \mu_J}(x_5) \rangle || \langle \widetilde{O}^{\dagger}_{\mu_1 \cdots \mu_J}(x_5) \phi_3 \phi_4 \rangle |,$$

$$(4.46)$$

where the left hand side denotes the contribution from the $G^s_{\Delta,J}$ block to the double commutator. Each commutator gives $2i\sin(\pi\frac{\Delta-J-2\Delta_{\phi}}{2})$ from the difference of two phase factors, so we have an additional $-2(2\sin^2(\pi\frac{\Delta-J-2\Delta_{\phi}}{2}))$ factor, which is accounted for on the right-hand side.

In our calculation, we will use absolute-valued structures as convenient notation in intermediate steps, and replace them with commutators at the end as appropriate. For example, for the structure $|\langle \phi_1(x_1)\phi_2(x_2)O(x_5)\rangle|$, which has causality relation $5 \approx (1 > 2)$, we have the following identity:

$$\langle 0|O(x_5)[\phi_1(x_1),\phi_2(x_2)]|0\rangle = 2i\sin(\pi\frac{\Delta - J - 2\Delta\phi}{2})|\langle\phi_1(x_1)\phi_2(x_2)O(x_5)\rangle|, \quad 5 \approx (1 > 2)$$
(4.47)

where the left-hand side is a commutator of the standard Wightman structure. For the other structure $|\langle \tilde{O}^{\dagger}(x_5)\phi_3\phi_4\rangle|$, this identity is not well-defined due to the additional phase factors from x_{45}^2, x_{35}^2 . However, in the final formula for the heavy action, we will be able to apply the above identity to both structures, enabling us to remove absolute values and write the final formula in terms of standard Wightman structures.

We can now plug the Lorentzian shadow representation into our functional. In the functional (4.42), the external points of the four-point function are 1^+ , 2^+ , 3, 4 with

⁸This is in contrast to the continuous-spin version of the Lorentzian shadow representation given in [11] that involves integrating over the polarization vector of the exchanged operator.



Figure 4.3: Our gauge fixing condition is $x_3 = -x_1 = y, x_4 = -x_2 = \frac{x}{-x^2}, x_5 = e$. The causality constraint is 4 > 5 > 3 > 0 > 1 > 2.

the condition 4 > 3 > 0 > 1 > 2, so the causality configuration agrees with the Lorentzian shadow representation (4.43). See also figure 4.2b. Therefore, using (4.43), the action of $\Psi_{k,\nu}^+$ on a conformal block can be written

$$\Psi_{k,\nu}^{+}[G_{\Delta,J}^{s}] = \frac{-8(2\sin^{2}(\pi\frac{\Delta-J-2\Delta\phi}{2}))A_{k,\nu}}{\beta_{\Delta,J}} \int_{4>5>3>0>1>2} \frac{d^{d}x_{1}d^{d}x_{2}d^{d}x_{3}d^{d}x_{4}d^{d}x_{5}d^{d}x_{0}D^{d-2}z_{0}}{\operatorname{vol}(\widetilde{\mathrm{SO}}(d,2))} \times \langle 0|\phi_{4}\mathbf{L}[\mathcal{O}](x_{0},z_{0})\phi_{2^{+}}|0\rangle^{-1} \langle \widetilde{\mathcal{P}}_{\delta_{1}}(x_{1^{+}0})\widetilde{\mathcal{P}}_{\delta_{3}}(x_{30})\mathcal{P}_{\delta}(z_{0})\rangle \times \delta(x_{1^{+}0}^{2})\delta(x_{30}^{2})\theta(x_{1^{+}0})\theta(x_{30})|\langle\phi_{1^{+}}\phi_{2^{+}}\mathcal{O}^{\mu_{1}\cdots\mu_{J}}(x_{5})\rangle||\langle \widetilde{\mathcal{O}}_{\mu_{1}\cdots\mu_{J}}^{\dagger}(x_{5})\phi_{3}\phi_{4}\rangle|.$$

$$(4.48)$$

For brevity, we have used the short-hand notation $\phi_i = \phi_i(x_i), \phi_{i^+} = \phi_i(x_i^+)$.

Now we use conformal symmetry to gauge fix the integral. We will choose the gauge fixing in [13] and fix $x_1 + x_3$ to the origin, $x_{2^+} + x_{4^-}$ to spatial infinity, and x_5 to the unit time vector *e*. More precisely, we choose (see figure 4.3)

$$x_3 = -x_1 = y, \quad x_4 = -x_2 = \frac{x}{-x^2}, \quad x_5 = e, \quad x, y > 0.$$
 (4.49)

For this gauge fixing, the stabilizer group is SO(d - 1), and the Faddeev-Popov determinant is [13]

$$2^{d}(1+2x \cdot y+x^{2}y^{2})(1-2x \cdot y+x^{2}y^{2})(1-x^{2}y^{2})^{d-2}.$$
 (4.50)

After the gauge fixing, we obtain

$$\Psi_{k,\nu}^{+}[G_{\Delta,J}^{s}] = \frac{-8(2\sin^{2}(\pi\frac{\Delta-J-2\Delta_{\phi}}{2}))A_{k,\nu}}{\beta_{\Delta,J}\operatorname{vol}(\operatorname{SO}(d-1))} \int \frac{d^{d}xd^{d}yd^{d}x_{0}D^{d-2}z_{0}}{(-x^{2})^{2d}} \times 2^{d}(1+2x\cdot y+x^{2}y^{2})(1-2x\cdot y+x^{2}y^{2})(1-x^{2}y^{2})^{d-2}} \times \langle 0|\phi_{4}\mathbf{L}[\mathcal{O}](x_{0},z_{0})\phi_{2^{+}}|0\rangle^{-1}\langle \widetilde{\mathcal{P}}_{\delta_{1}}(x_{1^{+}0})\widetilde{\mathcal{P}}_{\delta_{3}}(x_{30})\mathcal{P}_{\delta}(z_{0})\rangle \times \delta(x_{1^{+}0}^{2})\delta(x_{30}^{2})\theta(x_{1^{+}0})\theta(x_{30})|\langle\phi_{1^{+}}\phi_{2^{+}}\mathcal{O}^{\mu_{1}\cdots\mu_{J}}(x_{5})\rangle||\langle \widetilde{\mathcal{O}}_{\mu_{1}\cdots\mu_{J}}^{\dagger}(x_{5})\phi_{3}\phi_{4}\rangle|\Big|_{gauge-fixed},$$

$$(4.51)$$

where the last two lines should be evaluated in the gauge-fixed configuration (4.49).

The integral over z_0 can also be fixed using SO(d - 1) invariance. After integrating over x, y, x_0 , the remaining vectors in the integrand of (4.51) are z_0 and e. Homogeneity of z_0 then implies the z_0 -integral must be of the form

$$\int D^{d-2} z_0 \ (-2z_0 \cdot e)^{2-d} = \frac{\pi^{\frac{d-2}{2}} \Gamma(\frac{d-2}{2})}{\Gamma(d-2)}.$$
(4.52)

Thus, we can eliminate the z_0 -integral by setting z_0 to be a fixed null vector z_0^* , and then introduce a factor

$$(-2z_0^* \cdot e)^{d-2} \frac{\pi^{\frac{d-2}{2}} \Gamma(\frac{d-2}{2})}{\Gamma(d-2)}.$$
(4.53)

Let us introduce lightcone coordinate $x = (u, v, \vec{x}_{\perp})$, where $x^2 = -uv + \vec{x}_{\perp}^2$, and choose $z_0^* = (1, 0, \vec{0})$. Then (4.51) becomes

$$\Psi_{k,\nu}^{+}[G_{\Delta,J}^{s}] = \frac{-8(2\sin^{2}(\pi\frac{\Delta-J-2\Delta_{\phi}}{2}))A_{k,\nu}}{\beta_{\Delta,J}\operatorname{vol}(\operatorname{SO}(d-1))} \frac{\pi^{\frac{d-2}{2}}\Gamma(\frac{d-2}{2})}{\Gamma(d-2)} \int \frac{d^{d}xd^{d}yd^{d}x_{0}}{(-x^{2})^{2d}} \times 2^{d}(1+2x\cdot y+x^{2}y^{2})(1-2x\cdot y+x^{2}y^{2})(1-x^{2}y^{2})^{d-2}} \times \langle 0|\phi_{4}\mathbf{L}[O](x_{0},z_{0}^{*})\phi_{2^{+}})|0\rangle^{-1}\langle \widetilde{\mathcal{P}}_{\delta_{1}}(x_{1^{+}0})\widetilde{\mathcal{P}}_{\delta_{3}}(x_{30})\mathcal{P}_{\delta}(z_{0}^{*})\rangle \times \delta(x_{1^{+}0}^{2})\delta(x_{30}^{2})\theta(x_{1^{+}0})\theta(x_{30})|\langle\phi_{1^{+}}\phi_{2^{+}}O^{\mu_{1}\cdots\mu_{J}}(x_{5})\rangle||\langle \widetilde{O}_{\mu_{1}\cdots\mu_{J}}^{\dagger}(x_{5})\phi_{3}\phi_{4}\rangle|\Big|_{gauge-fixed}$$

$$(4.54)$$

4.3.3 Saddle point analysis

We now study the action $\Psi_{k,\nu}^+[G_{\Delta,J}^s]$ in the bulk-point limit, where we take both ν and Δ to be large (and $\Delta \sim m$). To study the large ν , large Δ limit of (4.54), we will consider the factors that depend exponentially on ν and Δ and look for a saddle



Figure 4.4: The two delta functions $\delta(x_{10}^2)\delta(x_{30}^2)$ force x_0 to be on an S^{d-2} (the blue curves). The directions of the variables t_0, \tilde{x}_0 defined in (4.59) are also indicated.

point. The important factors in (4.54) are given by

$$\left(\frac{(2z_0 \cdot x_{40}x_{20}^2 - 2z_0 \cdot x_{20}x_{40}^2)^2(2x_{10} \cdot x_{30})}{(-x_{24}^2)(-x_{40}^2)(-x_{20}^2)(2z_0 \cdot x_{10})(-2z_0 \cdot x_{30})}\right)^{\frac{tv}{2}} \left(\frac{(-x_{35}^2)(-x_{45}^2)(-x_{12}^2)}{(-x_{15}^2)(-x_{25}^2)(-x_{34}^2)}\right)^{\frac{m}{2}}\right|_{\text{gauge-fixed}, z_0 = z_0^*} = e^{f_{v,m}(x,y,x_0)},$$
(4.55)

where the function $f_{\nu,m}(x, y, x_0)$ is given by

$$f_{\nu,m}(x, y, x_0) = \frac{i\nu}{2} \log \left(\frac{4(z_0^* \cdot x(1+x^2x_0^2) - 2z_0^* \cdot x_0x^2x \cdot x_0)^2(-y^2)}{(-x^2)(1-2x \cdot x_0 + x^2x_0^2)(1+2x \cdot x_0 + x^2x_0^2)(z_0^* \cdot (y-x_0))(z_0^* \cdot (y+x_0))} \right) + \frac{m}{2} \log \left(\frac{(e-x)^2(e-y)^2}{(e+x)^2(e+y)^2} \right).$$

$$(4.56)$$

The integral we have to consider in the bulk-point limit then takes the form

$$\int d^{d}x d^{d}y d^{d}x_{0} \delta((y+x_{0})^{2}) \delta((y-x_{0})^{2}) \theta(y-x_{0}) \theta(y+x_{0}) e^{f_{\nu,m}(x,y,x_{0})} g(x,y,x_{0}),$$
(4.57)

where $g(x, y, x_0)$ are all the factors that do not depend exponentially on ν and m in the integrand of (4.54).

To find the location of the saddle, it is convenient to consider lightcone coordinates

$$x = (u_x, v_x, \vec{x}_\perp), \quad y = (u_y, v_y, \vec{y}_\perp), \quad x_0 = (u, v, \vec{y}_0).$$
 (4.58)

Let us also define

$$\widetilde{t}_{0} = \frac{1}{2} \left(\frac{u}{u_{y}} + \frac{v}{v_{y}} - 2\frac{\vec{y}_{0} \cdot \vec{y}_{\perp}}{u_{y}v_{y}} \right), \quad \widetilde{x}_{0} = \frac{1}{2} \left(\frac{u}{u_{y}} - \frac{v}{v_{y}} \right).$$
(4.59)

In our functional, there are two delta functions $\delta(x_{1+0}^2)\delta(x_{30}^2)$ in the integrand due to the light transforms. These delta functions restrict x_0 to lie on the intersection of

the past lightcone of x_3 and the future lightcone of x_1 . So, the x_0 -integral is forced to be on an S^{d-2} , as shown by figure 4.4. In terms of the variables defined above, the delta functions will localize \tilde{t}_0 and $|\vec{y}_0|$ to be

$$\tilde{t}_0 = 0, \quad |\vec{y}_0|^2 = \frac{u_y v_y (u_y v_y (1 - \tilde{x}_0^2) - \vec{y}_\perp^2)}{u_y v_y - (\vec{n}_0 \cdot \vec{y}_\perp)^2}, \tag{4.60}$$

where $\vec{n}_0 = \vec{y}_0/|\vec{y}_0|$ is the unit vector in the \vec{y}_0 direction. The remaining integral over S^{d-2} then becomes an integral over $\tilde{x}_0 \in \left[-(1 - \frac{\vec{y}_\perp^2}{u_y v_y})^{\frac{1}{2}}, (1 - \frac{\vec{y}_\perp^2}{u_y v_y})^{\frac{1}{2}}\right]$ and $\vec{n}_0 \in S^{d-3}$.

After going to lightcone coordinates and removing the delta functions, (4.57) becomes

$$\frac{(u_{y}v_{y})^{\frac{d-2}{2}}}{32} \int d\widetilde{x}_{0} du_{x} dv_{x} du_{y} dv_{y} d\Omega_{\vec{n}_{0}} d^{d-2} \vec{x}_{\perp} d^{d-2} \vec{y}_{\perp} (u_{y}v_{y} - (\vec{n}_{0} \cdot y_{\perp})^{2})^{\frac{2-d}{2}} (u_{y}v_{y}(1 - \widetilde{x}_{0}^{2}) - \vec{y}_{\perp}^{2})^{\frac{d-4}{2}} \times e^{f_{v,m}(x,y,x_{0})} g(x,y,x_{0}).$$
(4.61)

The large v and large m limit localizes all the variables in the integral, except for \vec{n}_0 . In fact, the $f_{v,m}$ function leads to sixteen different saddles at

$$x_{0} = (0, 0, \vec{y}_{0}), \quad x = \left(-i\frac{m \pm \sqrt{m^{2} - \nu^{2}}}{\nu}, i\frac{m \pm \sqrt{m^{2} - \nu^{2}}}{\nu}, \vec{0}\right), \quad y = \left(i\frac{m \pm \sqrt{m^{2} - \nu^{2}}}{\nu}, -i\frac{m \pm \sqrt{m^{2} - \nu^{2}}}{\nu}, \vec{0}\right)$$
(4.62)

where $|\vec{y}_0|^2 = -y^2$. Each saddle corresponds to a locus S^{d-3} parametrized by \vec{n}_0 . To find the correct saddle approximation for the integral (4.61), we need to deform the integration contour into steepest descendant flows and see which saddle locus the contour goes through (see [132] for a pedagogical introduction). The analysis is essentially identical to [13], which also studies saddle point of x, y in the bulk-point limit. The result is that the dominant saddle depends on the relative size of v and m. The case we will be particularly interested in is the region related to the flat space functional, where one considers $v \in [0, \Delta_{gap}]$ and $m > \Delta_{gap}$. In this case (v < m), the saddle locus that dominates the integral is given by

$$x_{0} = (0, 0, \vec{y}_{0}), \quad x = \left(-i\frac{m-\sqrt{m^{2}-\nu^{2}}}{\nu}, i\frac{m-\sqrt{m^{2}-\nu^{2}}}{\nu}, \vec{0}\right), \quad y = \left(i\frac{m-\sqrt{m^{2}-\nu^{2}}}{\nu}, -i\frac{m-\sqrt{m^{2}-\nu^{2}}}{\nu}, \vec{0}\right)$$
(4.63)

Expanding the function $f_{\nu,m}$ around the dominant saddle (4.63), we find

$$f_{\nu,m} = -\pi\nu + i\nu\log 2 - \frac{1}{2} \begin{pmatrix} \tilde{x}_{0} & u_{x} & v_{x} & u_{y} & v_{y} & \vec{x}_{\perp} & \vec{y}_{\perp} \end{pmatrix} M_{\nu,m,\vec{y}_{0}} \begin{pmatrix} \tilde{x}_{0} \\ u_{x} \\ v_{x} \\ u_{y} \\ v_{y} \\ \vec{x}_{\perp} \\ \vec{y}_{\perp} \end{pmatrix} + \dots,$$
(4.64)

where ... are higher-order terms in the expansion. The Hessian M_{ν,m,\vec{y}_0} has determinant

$$\operatorname{Det} M_{\nu,m,\vec{y}_0} = -\frac{i\nu^{6d+1}(m^2 - \nu^2)^2(m - \sqrt{m^2 - \nu^2})^{-4d}}{16m^4}.$$
 (4.65)

Therefore, the large v, large m limit of the integral (4.57) can be written as

$$\lim_{\substack{\nu,m\gg1\\\nu
$$= \frac{\nu^{4-d} (m - \sqrt{m^2 - \nu^2})^{d-4}}{32} \sqrt{\frac{(2\pi)^{2d+1}}{\operatorname{Det} M_{\nu,m,\vec{y}_0}}} e^{-\pi\nu} 2^{i\nu} \int d\Omega_{\vec{n}_0} g(x,y,x_0)|_{\text{saddle}}, \quad (4.66)$$$$

where saddle stands for the configuration (4.63). Furthermore, the function $g(x, y, x_0)$ evaluated at the saddle does not include any (d - 2)-dimensional vectors other than \vec{n}_0 , and hence the remaining integral over the S^{d-3} locus must be trivial. The final expression of the integral is then given by

$$\lim_{\substack{v,m\gg1\\v$$

where we have plugged in the Hessian determinant and used the fact that $e^{-\pi v} 2^{iv}$ comes from evaluating $e^{f_{v,m}(x,y,x_0)}$ at the saddle.

Now, we can compare (4.67) and the functional (4.54) and find the expression for $\Psi_{k,\nu}^+[G_{\Delta,J}^s]$ in the bulk-point limit. From the above saddle analysis, we see that the integral in the bulk-point limit can be obtained by simply evaluating the integrand at the saddle and multiplying by the additional factors coming from the Jacobian and

Hessian. This implies that in the bulk-point limit, the calculation of the heavy action of our functional can be simplified to evaluating conformally-invariant structures in the saddle configuration (4.63), which leads to

$$\lim_{\substack{\nu,m\gg1\\\nu
(4.68)$$

The saddle calculation done above involves computing the determinant of a $(2d + 1) \times (2d + 1)$ Hessian matrix. An alternative derivation of the above formula that can avoid this technically involved calculation is by first studying the large ν limit of the functional, and then use the result of the large *m* saddle analysis that was already done in [13]. We describe this calculation in appendix C.2.

The shadow coefficient

Finally, the remaining task is to study the shadow coefficient $\beta_{\Delta,J}$ in the large Δ limit. For the scalar case, its explicit expression is known and given by (4.45). For the spinning case, the shadow transform will involve mixing of different tensor structures and therefore the shadow coefficients become a matrix. In principle it can be computed using e.g., weight-shifting operators [39]. However, as we discuss below, it turns out that the shadow transform at large Δ also gets localized to a saddle. Hence, the shadow transform at large Δ becomes algebraic and the coefficients can again be computed by evaluating structures at the saddle configuration.

Let us demonstrate this idea by studying the shadow transform in the scalar case. We will derive a formula for $\beta_{\Delta,J}$ that reproduces the known answer (4.45). However, the advantage of this formula is that its generalization to the spinning case is almost trivial.

Considering the OPE limit of (4.43), we see that the shadow coefficient $\beta_{\Delta,J}$ should satisfy

$$\frac{1}{\beta_{\Delta,J}} \int_{4>5>3} d^d x_5 \langle O^{\dagger}_{\nu_1 \cdots \nu_J}(x_6) O^{\mu_1 \cdots \mu_J}(x_5) \rangle |\langle \widetilde{O}^{\dagger}_{\mu_1 \cdots \mu_J}(x_5) \phi_3 \phi_4 \rangle|$$

= $|\langle O^{\dagger}_{\nu_1 \cdots \nu_J}(x_6) \phi_3 \phi_4 \rangle|.$ (4.69)

Let us choose the external points x_3, x_4, x_6 to be

$$x_3 = y|_{\text{saddle}}, \quad x_4 = \frac{x}{-x^2}|_{\text{saddle}}, \quad x_6 = (-e)^+,$$
 (4.70)

where x, y are evaluated at the saddle configuration (4.63). With this choice of external points, we find that at large Δ , the integrand in (4.69) has a saddle point at exactly $x_5 = e$, which agrees with the saddle configuration of the original integral of the functional $\Psi_{k,v}^+$. After computing the determinant factor coming from the Gaussian integral, we obtain that in the large Δ limit, (4.69) becomes

$$\lim_{\Delta \gg 1} \frac{2^d \pi^{\frac{d}{2}} \Delta^{-\frac{d}{2}}}{\beta_{\Delta,J}} \left\langle O_{\nu_1 \cdots \nu_J}^{\dagger}(x_6) O^{\mu_1 \cdots \mu_J}(x_5) \right\rangle \left| \left\langle \widetilde{O}_{\mu_1 \cdots \mu_J}^{\dagger}(x_5) \phi_3 \phi_4 \right\rangle \right| \right|_{\substack{\text{saddle}\\ x_6 = (-e)^+}} = \left| \left\langle O_{\nu_1 \cdots \nu_J}^{\dagger}(x_6) \phi_3 \phi_4 \right\rangle \right| \left| \underset{x_6 = (-e)^+}{\text{saddle}} \right|_{\substack{x_6 = (-e)^+}} .$$
(4.71)

To recap, we see that at large Δ , the shadow transform gets localized to a saddle point. Furthermore, with a good choice of external points $x_6 = (-e)^+$, we can make the saddle point location agree with the saddle configuration for the functional (4.63). We have also checked that (4.71) is consistent with the known scalar result (4.45).

To simplify the formula of the functional in the bulk-point limit given by (4.68), we need the combination $\frac{1}{\beta_{\Delta,J}} |\langle \phi_{1^+} \phi_{2^+} O^{\mu_1 \cdots \mu_J}(x_5) \rangle|| \langle \widetilde{O}_{\mu_1 \cdots \mu_J}^{\dagger}(x_5) \phi_3 \phi_4 \rangle|$, which is different from the structure appearing in (4.71). To get the correct structure, let us first note that the two-point structure at the saddle $\langle O^{\dagger \nu_1 \cdots \nu_J}((-e)^+) O^{\mu_1 \cdots \mu_J}(e) \rangle$ can be viewed as an invertible matrix for the spin-*J* representation of the Lorentz group.⁹ Therefore, we can define an inverse "*r*-tensor" that satisfies

$$r_{\Delta,J;\rho_1\cdots\rho_J}{}^{\nu_1\cdots\nu_J}\langle O^{\dagger}_{\nu_1\cdots\nu_J}((-e)^+)O^{\mu_1\cdots\mu_J}(e)\rangle = \delta_{\{\rho_1}{}^{\{\mu_1}\cdots\delta_{\rho_J\}}{}^{\mu_J\}} - \text{traces.} \quad (4.72)$$

Here, $\{\mu_1 \cdots \mu_J\}$ means we symmetrize the indices, and "-traces" means we subtract terms proportional to $\delta^{\mu_i \mu_j}$ and $\delta_{\rho_i \rho_j}$ to make the result traceless. Concretely, in this case $r_{\Delta,J}$ is given by a reflection in the time direction,

$$r_{\Delta,J;\rho_1\cdots\rho_J}{}^{\nu_1\cdots\nu_J} = (-1)^J 2^{-J+2\Delta} \left(\delta_{\{\rho_1}{}^{\{\nu_1} + 2e_{\{\rho_1}e^{\{\nu_1\}} \right) \cdots \left(\eta_{\rho_J} \right)^{\nu_J} + 2e_{\rho_J} e^{\nu_J} \right) - \text{traces.}$$

$$(4.73)$$

⁹More generally, if *O* has representation ρ , it should be a map from ρ^* to $\rho^{\dagger} = (\rho^*)^R$.

By contracting both sides of (4.71) with $|\langle \phi_{1^+}\phi_{2^+}O^{\rho_1\cdots\rho_J}(x_5)\rangle|r_{\Delta,J;\rho_1\cdots\rho_J}{}^{\nu_1\cdots\nu_J}$, we obtain

$$\begin{split} \lim_{\Delta \gg 1} \frac{1}{\beta_{\Delta,J}} \left| \langle \phi_{1^{+}} \phi_{2^{+}} O^{\mu_{1} \cdots \mu_{J}}(x_{5}) \rangle \right| \left| \langle \widetilde{O}_{\mu_{1} \cdots \mu_{J}}^{\dagger}(x_{5}) \phi_{3} \phi_{4} \rangle \right| \Big|_{\substack{\text{saddle} \\ x_{6} = (-e)^{+}}} \\ &= 2^{-d} \pi^{-\frac{d}{2}} \Delta^{\frac{d}{2}} \left| \langle \phi_{1^{+}} \phi_{2^{+}} O^{\mu_{1} \cdots \mu_{J}}(x_{5}) \rangle |r_{\Delta,J;\mu_{1} \cdots \mu_{J}}^{\nu_{1} \cdots \nu_{J}} |\langle O_{\nu_{1} \cdots \nu_{J}}^{\dagger}(x_{6}) \phi_{3} \phi_{4} \rangle \right| \Big|_{\substack{\text{saddle} \\ x_{6} = (-e)^{+}}}. \end{split}$$

$$(4.74)$$

The left hand side of the above equation now agrees with the structure in the formula (4.68). We can then plug this equation into (4.68) and find a formula without the shadow coefficient. We write down this final formula in the next subsection.

Finally, let us discuss a natural formula for the $r_{\Delta,\rho}$ -tensor that will make it easier for us to generalize to the spinning case. It also makes it clear that the tensor depends on a choice of two-point structure convention. For a general representation ρ , the corresponding *r*-tensor is defined as

$$r_{\Delta,\rho;a'}{}^{\overline{b}}\langle O^{\dagger}_{\overline{b}}((-e)^{+})O^{a}(e)\rangle = \delta^{a}{}_{a'}, \qquad (4.75)$$

where the operator O has quantum numbers (Δ, ρ) . Here, a, a' are the indices of the representations ρ, ρ^* , and \overline{b} labels the indices of the reflected representation ρ^R and its dual ρ^{\dagger} . On the right-hand side, $\delta^a{}_{a'}$ is the identity matrix of the ρ representation (i.e., $\delta^a{}_{a'}T^{a'} = T^a$ for any tensor T with representation ρ).

Motivated by symmetry, we can write down an ansatz for the $r_{\Delta,\rho}$ -tensor using the two-point function of the shadow operator \tilde{O} ,

$$r_{\Delta,\rho;a'}{}^{\overline{b}} = C_{\Delta,\rho} \langle \widetilde{O}_{a'}^{\dagger}(e) \widetilde{O}^{\overline{b}}((-e)^{+}) \rangle.$$
(4.76)

Plugging this ansatz into the definition of $r_{\Delta,\rho}$ and taking the trace, we obtain

$$C_{\Delta,\rho}\langle \widetilde{O}_{a}^{\dagger}(e)\widetilde{O}_{b}^{\overline{b}}((-e)^{+})\rangle\langle O_{\overline{b}}^{\dagger}((-e)^{+})O^{a}(e)\rangle = \delta^{a}{}_{a} = \dim(\rho).$$
(4.77)

We see that the unknown coefficient $C_{\Delta,\rho}$ in the ansatz can be expressed in terms of the dimension of the representation and a pairing of the two-point functions.

We define a natural Euclidean two-point pairing [45],¹⁰

$$\frac{\left(\langle \widetilde{O}^{\dagger} \widetilde{O} \rangle, \langle O^{\dagger} O \rangle\right)}{\operatorname{vol}(\mathrm{SO}(1,1))} = \int \frac{d^{d}x d^{d}y}{\operatorname{vol}(\mathrm{SO}(d+1,1))} \langle \widetilde{O}_{a}^{\dagger}(x) \widetilde{O}_{b}^{\overline{b}}(y) \rangle \langle O_{\overline{b}}^{\dagger}(y) O^{a}(x) \rangle,
= \frac{1}{2^{d} \operatorname{vol}(\mathrm{SO}(d)) \operatorname{vol}(\mathrm{SO}(1,1))} \langle \widetilde{O}_{a}^{\dagger}(0) \widetilde{O}_{b}^{\overline{b}}(\infty) \rangle \langle O_{\overline{b}}^{\dagger}(\infty) O^{a}(0) \rangle.$$
(4.78)

¹⁰Compared to the definition in [45], we have absorbed an infinite vol(SO(1, 1)) factor to make the pairing finite.



Figure 4.5: The bulk-point limit localizes the spacetime integral to a saddle locus that we call the scattering-crystal. The entire saddle satisfies the causality constraint 4 > 5 > 3 > 0 > 1 > 6 > 2. The two delta functions restrict x_0 to be on an S^{d-2} shown by the red circle in the figure. The saddle locus further restricts x_0 to an S^{d-3} , which is represented by the two red dots (0-sphere). Points 5 and 6 are localized at $\pm e$. Points 1, 2, 3, 4 are restricted to be timelike vectors, although in practice they have purely imaginary spatial components (see (4.81)).

Originally, in (4.77) we want to compute the pairing between Lorentzian two-point structures. Since the two points e, $(-e)^+$ are spacelike separated, the structures can be thought of as Euclidean two-point structures with distance $\sqrt{-(2e)^2} = 2$, and we can replace the pairing with a Euclidean two-point pairing (4.78). Using the two-point pairing, one can then show that (4.77) gives

$$C_{\Delta,\rho} = \frac{2^{d} \dim(\rho)}{\operatorname{vol}(\operatorname{SO}(d))\left(\langle \widetilde{O}^{\dagger} \widetilde{O} \rangle, \langle O^{\dagger} O \rangle\right)},\tag{4.79}$$

and the *r*-tensor can be written as

$$r_{\Delta,\rho;a'}{}^{\overline{b}} = \frac{2^d \dim(\rho)}{\operatorname{vol}(\operatorname{SO}(d))\left(\langle \widetilde{O}^{\dagger} \widetilde{O} \rangle, \langle O^{\dagger} O \rangle\right)} \langle \widetilde{O}_{a'}^{\dagger}(e) \widetilde{O}^{\overline{b}}((-e)^+) \rangle.$$
(4.80)

4.3.4 Scalar sum rules

Let us briefly summarize what we have done and write down the final formula for the scalar functional in the bulk-point limit. We started with the commutator of two light-transformed operators and wrote down a functional $\Psi_{k,\nu}$ ((4.33)) whose action on any physical four-point function vanishes. The functional integrates the four-point function against a specific kernel over the external points $x_{1,2,3,4}$ and internal variables x_0, z_0 . Using the Lorentzian shadow representation of conformal blocks, we wrote down the action of the functional on blocks, with an additional internal point x_5 from the shadow representation. We then found that in the bulk-point limit ($\nu, m \gg 1$ with $\nu < m$), the integral gets completely localized to a saddle configuration (4.63). Furthermore we removed the shadow coefficient by introducing a final internal point x_6 .

In the end, we obtain a formula for the action of the functional on a conformal block in the bulk-point limit. The formula is given by a known coefficient and a product of conformally-invariant structures evaluated at a configuration that fixes all the positions and polarizations, $x_{0,1,2,3,4,5,6}$, z_0 . In lightcone coordinates, the configuration is given by (see figure 4.5)

$$\begin{aligned} x_{3} &= -x_{1} = y, \quad x_{4} = -x_{2} = \frac{x}{-x^{2}}, \quad x_{5} = e = (1, 1, \vec{0}), \quad x_{6} = -e, \\ x &= \left(-i\frac{m-\sqrt{m^{2}-\nu^{2}}}{\nu}, i\frac{m-\sqrt{m^{2}-\nu^{2}}}{\nu}, \vec{0}\right), \quad y = \left(i\frac{m-\sqrt{m^{2}-\nu^{2}}}{\nu}, -i\frac{m-\sqrt{m^{2}-\nu^{2}}}{\nu}, \vec{0}\right), \\ z_{0} &= (1, 0, \vec{0}), \quad x_{0} = (0, 0, \vec{y}_{0}), \quad |\vec{y}_{0}|^{2} = \frac{(m-\sqrt{m^{2}-\nu^{2}})^{2}}{\nu^{2}}. \end{aligned}$$
(4.81)

As we will show later, evaluating the conformal structures in this configuration reproduces the heavy action of a flat space sum rule and allows us to study the bulk scattering process. This is a generalization of the saddle point that leads to the *spacelike scattering* phenomenon in [13]. We will call the configuration (4.81) the "scattering-crystal."

By plugging (4.74) in (4.68) to remove the shadow coefficient, we find

$$\lim_{\substack{\nu,m\gg1\\\nu(4.82)$$

A nice feature of the scattering-crystal is that both structures with absolute values satisfy the identity (4.47) in this configuration. Therefore, we can replace those structures with commutators of standard Wightman structures, and remove the



Figure 4.6: Our functional is a special 6j symbol (or a tetrahedron) where one of the three-point structures is a celestial structure. The cross in the left figure stands for the contraction of the $\langle \tilde{O}^{\dagger} \tilde{O} \rangle$ tensor. The flat space limit is the limit where the two red legs have large quantum numbers. (More precisely, the two legs have large scaling dimensions.)

 $-2((2\sin^2(\pi \frac{\Delta - J - 2\Delta_{\phi}}{2})))$ factor on the left-hand side. Moreover, by applying the natural formula for the *r*-tensor given by (4.80), we arrive at the final formula for the action of the scalar functional in the bulk-point limit:¹¹

$$\lim_{\substack{\nu,m\gg1\\\nu

$$(4.83)$$$$

This formula can be viewed as a kind of 6j symbol, or equivalently a tetrahedron formed by gluing conformal three-point structures (see figure 4.6). Compared to the usual 6j symbol [133], in our case one of the three-point structures becomes a celestial structure, which originally comes from setting one of the legs to have J =-1. Furthermore, the bulk-point limit, which reproduces the flat space functional, corresponds to setting two of the legs to have large quantum numbers. In this limit, one of them has large and positive dimension $\Delta \gg 1$ and the other one has dimension

¹¹To keep track of $(-1)^J$ factors, we should clarify our convention for $\langle 0|O^{\dagger}(x_6)[\phi_4,\phi_3]|0\rangle$. Our convention is that it should be the analytic continuation of (C.2) with $1 \rightarrow 3, 2 \rightarrow 4, 3 \rightarrow 6$.

 $\frac{d}{2} + iv$ with $v \gg 1$. For usual 6*j* symbols in the classical limit, where all legs have large quantum number, they can be computed as the volume of a tetrahedron [134]. It would be interesting to see if there is a similar argument for the bulk-point limit of our functional.

Before using this formula, we still need to address two more issues. First, note that the original functional $\Psi_{k,\nu}$ in (4.33) is given by

$$\Psi_{k,\nu} = \Psi_{k,\nu}^{+} - (1 \leftrightarrow 3). \tag{4.84}$$

Therefore, when we apply $\Psi_{k,\nu}$ to a four-point function of identical operators, which is manifestly symmetric under $(1 \leftrightarrow 3)$, we will get a trivial sum rule since its action on each conformal block is simply zero. The second issue is about the convergence in the Regge limit discussed below (4.32). For external scalars, in order for the light transforms to converge, we need the Regge intercept to satisfy $J_0 < -1$, while for non-perturbative CFTs we only know that $J_0 \le 1$ [115, 9, 34].

It turns out that both issues can be resolved by introducing a "subtraction factor" to the functional, that is, we consider a more general functional

$$\Psi_{k,\nu,f}[\mathcal{G}] = \Psi_{k,\nu}^{+}[f(u',v')\mathcal{G}] - (1 \leftrightarrow 3), \tag{4.85}$$

where f(u', v') is a meromorphic function of the conformally-invariant cross-ratios u', v' defined in (4.31). Note that the kernel of $\Psi_{k,v,f}$ is again manifestly antisymmetric under $1 \leftrightarrow 3$. On the other hand, points 1 and 3 are always spacelike in the range of integration of the functional, which comes from the light transforms. Therefore, causality/crossing guarantees that any physical correlator \mathcal{G} should be symmetric under $1 \leftrightarrow 3$ in the integral, and the action of the general functional $\Psi_{k,v,f}$ should vanish on physical four-point functions. Additionally, by choosing subtraction factors that are antisymmetric under $1 \leftrightarrow 3$, which swaps u' and v', we can get nontrivial sum rules for identical external scalars, meaning its action on each conformal block is nonzero. Furthermore, by restricting to subtraction factors with nice behaviors in the Regge limit, we can make sure the integral of the functional is convergent in the Regge limit.

For the scalar case, a nice choice of subtraction factors with the above properties

is¹²

$$f_k^{\text{scalar}}(u',v') = \begin{cases} \frac{v'-u'}{(u'v')^{\frac{k+2}{4}}} & k \equiv 2 \mod 4\\ \frac{(v'-u')^3 - (v'-u')}{4(u'v')^{\frac{k+4}{4}}} & k \equiv 0 \mod 4 \end{cases}, \qquad k = 2, 4, 6, \dots$$
(4.87)

At the scattering-crystal locus, these subtraction factors become

$$f_k^{\text{scalar}}(u',v')\Big|_{\text{scattering-crystal}} = -v^{2k+2} \frac{(2m^2 - v^2)}{m^2(m^2 - v^2)} \frac{1}{(m^2(m^2 - v^2))^{\frac{k}{2}}}.$$
 (4.88)

Note that this satisfies $f_k^{\text{scalar}}(u', v') = -f_k^{\text{scalar}}(v', u')$. In the Regge limit, the subtraction factor increases the effective Regge spin of the kernel by k + 1. (One can see this from the large-*m* scaling of (4.88).) Therefore, the condition from convergence in the Regge limit becomes $J_0 < k$, which is indeed true for any $k \ge 2$.

Using the subtraction factors (4.87), we can then write down nontrivial sum rules for identical scalars,

$$\Psi_{k,\nu,f_{\nu}^{\text{scalar}}}[\mathcal{G}_{\phi\phi\phi\phi}] = 0. \tag{4.89}$$

Its action on a single s-channel block is given by

$$\Psi_{k,\nu,f_k^{\text{scalar}}}[G_{\Delta,J}^s] = 2\Psi_{k,\nu}^+[f_k^{\text{scalar}}(u',v')G_{\Delta,J}^s].$$
(4.90)

In the bulk-point limit, since the subtraction factors do not have any v and m dependence, the formula (4.83) gets modified to

$$\begin{split} &\lim_{\substack{\nu,m\gg1\\\nu$$

$$\left((v'-u')^3 + \frac{3}{d-8}(v'-u') \right) G_{d-1,\frac{2-d}{2}+i\nu} \sim 8G_{d-5,\frac{2-d}{2}+i\nu}, \quad \nu \to \infty.$$
(4.86)

¹²The $k \equiv 2 \mod 4$ case follows straightforwardly from the discussion in appendix D of [13]. The other case can be motivated by taking the large- ν limit of (D.12) in the same paper, and using the identity

We can then evaluate the structures and and the subtraction factor at the scatteringcrystal locus to find an explicit expression. Moreover, for symmetric traceless tensors, the pairing $\frac{\text{vol}(SO(d))}{2^d \dim(\rho)} \left(\langle \widetilde{O}^{\dagger} \widetilde{O} \rangle, \langle O^{\dagger} O \rangle \right)$ becomes 4^{-d} [45]. After combining all the factors and writing the result in terms of the "heavy action" defined in (4.28), we obtain

$$\lim_{\substack{\nu,m\gg1\\\nu< m}} \Psi_{k,\nu,f_k^{\text{scalar}}}[\Delta,J] = \frac{2m^2 - \nu^2}{m^2 - \nu^2} \frac{\mathcal{P}_J(1 - \frac{2\nu^2}{m^2})}{(m^2(m^2 - \nu^2))^{\frac{k}{2}}}.$$
(4.92)

As reviewed in section 4.2, this is precisely the heavy state contribution of the flat space sum rule for scalars.

4.3.5 Spinning sum rules

We are now ready to write down the sum rules for spinning operators. The above analysis for the bulk-point limit can be generalized straightforwardly. A spinning functional can be defined by

$$\Psi_{k,\nu,\lambda'}^{(a),(c)}[\mathcal{G}] \equiv \int_{\substack{2>4\\0\approx2,4}}^{2>4} \frac{d^d x_2 d^d x_4 d^d x_0 D^{d-2} z_0}{\operatorname{vol}(\widetilde{SO}(d,2))} \left(\langle 0|O_4(x_4^+) \mathbf{L}[O](x_0,z_0)O_2(x_2)|0\rangle^{(a)} \right)^{-1} \int D^{d-2} z_1 D^{d-2} z_3 \langle \widetilde{\mathcal{P}}_{\delta_1,\lambda_1}^{\dagger}(z_1) \widetilde{\mathcal{P}}_{\delta_3,\lambda_3}^{\dagger}(z_3) \mathcal{P}_{\delta,\lambda'}(z_0) \rangle^{(c)} \langle \Omega|O_4(x_4^+)[\mathbf{L}[O_3](x_0,z_3),\mathbf{L}[O_1](x_0,z_1)]O_2(x_2)|\Omega\rangle$$

$$(4.93)$$

Compared to the scalar case (4.33), the main difference is that the operators O and $\mathcal{P}_{\delta,\lambda'}$ can have more complicated representations, and there are multiple allowed dual structures and celestial structures.

Moreover, in the spinning case we do not include any coefficient in front of the integral. In the scalar case, there is a natural choice for this coefficient $A_{k,\nu}$, which is the factor needed to exactly reproduce the action of flat space sum rule (4.92). However, as we will see in the next section, for spinning operators, the CFT sum rules and flat space sum rules are related by a matrix. Since there is no natural choice for the overall factor, we do not include a factor in our definition of the spinning functional.

Let us introduce some notation to describe spinning operators. A spinning operator with SO(d - 1, 1) representation ρ can be described by a Young diagram with rows of length (m_1, m_2, \ldots, m_n) , where $n = \lfloor \frac{d}{2} \rfloor$. We will also use the notation $\rho = (J, \lambda)$, where $J = m_1$ is the spin, and $\lambda = (m_2, \ldots, m_n)$ specifies an SO(d - 2) representation (which we sometimes call the "transverse" representation). We also define $j = m_2$ to be the transverse spin of the operator.

In (4.93), the external operators have quantum numbers $(\Delta_i, J_i, \lambda_i)$, and the operator O in the first line has quantum numbers $(\frac{d}{2} + i\nu, J_1 + J_3 - 1, \lambda')$. In the second line, the celestial operators, which are primary operators that live on the celestial sphere, also carry SO(d - 2) representations. $\widetilde{P}^{\dagger}_{\delta_1,\lambda_1}$ and $\widetilde{P}^{\dagger}_{\delta_3,\lambda_3}$ have the same SO(d - 2) representations as the transverse representations of the external operators O_1, O_3 , and $\mathcal{P}_{\delta,\lambda'}$ have the same transverse representation as O. Note that the SO(d - 1, 1) indices of the ρ_2, ρ_4 representations, and the SO(d - 2) indices of $\lambda_1, \lambda_3, \lambda'$ are implicitly contracted in (4.93).

In addition to the ν variable, the spinning functional also depends on a choice of SO(d-2) representation λ' . Moreover, the functional has two tensor structure labels (a), (c). The first label (a) corresponds to the tensor structure of the spinning dual structure, defined as

$$\left(\left(\langle 0|O_4(x_4^+)\mathbf{L}[O](x_0, z_0)O_2(x_2)|0\rangle^{(a)}\right)^{-1}, \langle 0|O_4(x_4^+)\mathbf{L}[O](x_0, z_0)O_2(x_2)|0\rangle_{(b)}\right)_L = \delta_b^a$$
(4.94)

The allowed number of labels (*a*) is the number of continuous-spin structures of $\langle O_2 O_4 O \rangle$, which is given by [48, 15]

$$\left(\lambda' \otimes \left(\operatorname{Res}_{\operatorname{SO}(d-2)}^{\operatorname{SO}(d-1,1)} \rho_2 \otimes \rho_4\right)\right)^{\operatorname{SO}(d-2)},\tag{4.95}$$

where $\operatorname{Res}_{H}^{G}$ denotes the restriction of a representation of G to its subgroup H, and $(\ldots)^{H}$ denotes the number of H-singlets. The second tensor structure label (c) corresponds to the structures of a celestial three-point function with $\operatorname{SO}(d-2)$ representation $\lambda_1, \lambda_3, \lambda'$. The number of those structures is [48]

$$\left(\operatorname{Res}_{\operatorname{SO}(d-2)}^{\operatorname{SO}(d-2)}\lambda_1 \otimes \lambda_3 \otimes \lambda'\right)^{\operatorname{SO}(d-3)}.$$
(4.96)

Combining (4.95) and (4.96), we see that given the representations of the external operators, the allowed choice of the transverse representation λ' is

$$\left\{\lambda'|\lambda' \in \operatorname{Res}_{\operatorname{SO}(d-2)}^{\operatorname{SO}(d-1,1)}\rho_2 \otimes \rho_4, \left(\operatorname{Res}_{\operatorname{SO}(d-3)}^{\operatorname{SO}(d-2)}\lambda_1 \otimes \lambda_3 \otimes \lambda'\right)^{\operatorname{SO}(d-3)} \neq 0\right\}.$$
 (4.97)

In section 4.4, we will give the set of allowed λ' 's for some concrete examples.

Similar to the scalar case, commutativity of the light-transformed operators implies

$$\Psi_{k,\nu,\lambda'}^{(a),(c)}[\mathcal{G}] = 0, \quad J_1 + J_3 - 1 > J_0.$$
(4.98)

Again, this condition can be equivalently written in terms of the $C_{ab}^{\pm}(\Delta, \rho)$ coefficient function,

$$C_{ab}^{(-1)^{J_1+J_3-1}}(\frac{d}{2}+i\nu, J_1+J_3-1, \lambda') = 0, \quad J_1+J_3-1 > J_0,$$
(4.99)

where *a* is the same tensor structure label of the functional (4.93), which corresponds to the structures of the $O_2 \times O_4$ OPE. On the other hand, *b* is a subset of the structures of the $O_1 \times O_3$ OPE that have the same counting rule as the celestial structures (4.96) (see [77] for more details).

The spinning heavy action

We now derive a formula for the heavy action of the spinning functional in the bulk-point limit, similar to the scalar case (4.83). For simplicity, in what follows we assume the external operators are symmetric traceless tensors with spin J_e , and they all have scaling dimensions Δ_e . By rewriting the light transforms in (4.93), the spinning functional can be written as

$$\Psi_{k,\nu,\lambda'}^{(a),(c)} = \Psi_{k,\nu,\lambda'}^{+(a),(c)} - (1 \leftrightarrow 3), \tag{4.100}$$

and

$$\begin{split} \Psi_{k,\nu,\lambda'}^{+(a),(c)}[\mathcal{G}] \\ &= 4 \int_{4>3>0>1>2} \frac{d^d x_1 d^d x_2 d^d x_3 d^d x_4 d^d x_0 D^{d-2} z_0}{\operatorname{vol}(\widetilde{\mathrm{SO}}(d,2))} \delta(x_{1^+0}^2) \delta(x_{30}^2) \theta(x_{1^+0}) \theta(x_{30}) \\ &\times \left(\langle 0|\mathcal{O}_4(x_4) \mathbf{L}[\mathcal{O}](x_0,z_0) \mathcal{O}_2(x_2^+) |0\rangle^{(a)} \right)^{-1} \langle \widetilde{\mathcal{P}}_{\delta_1,\lambda_1}^{\dagger}(x_{1^+0}) \widetilde{\mathcal{P}}_{\delta_3,\lambda_3}^{\dagger}(x_{30}) \mathcal{P}_{\delta,\lambda'}(z_0) \rangle^{(c)} \\ &\times \langle \Omega|[\mathcal{O}_4(x_4),\mathcal{O}_3(x_3,x_{30})][\mathcal{O}_1(x_1^+,x_{1^+0}),\mathcal{O}_2(x_2^+)] |\Omega\rangle. \end{split}$$
(4.101)

Note that the polarizations of O_1 , O_3 are fixed to be x_{30} , x_{1+0} (which are null vectors due to the delta functions in the integrand). This is because of the light transforms $L[O_3]L[O_1]$ in (4.93).

To study the action of $\Psi_{k,v,\lambda'}^{+(a),(c)}$ on a conformal block, we again use the Lorentzian shadow representation of the block and consider the bulk-point limit, in which $\Delta \sim m$ and $v, m \gg 1, v < m$. To perform the saddle point analysis, we separate the integrand into a quickly-varying part and slowly-varying part. Crucially, in the bulk-point limit, the quickly-varying part of the integrand is the same as the scalar case! To see why, recall that the factor in the scalar case is given by (4.55), and it comes from powers of $x_{ij}^2, z_0 \cdot x_{10}, z_0 \cdot x_{30}, x_{10} \cdot x_{30}$, and $V_{0,24}$ in the conformally-invariant

structures in the integrand. For spinning structures, the powers of these factors will only differ by integer values that depend on the representations and choice of tensor structures, and hence they do not matter in the $v, m \gg 1$ limit. Since the quickly-varying part in the scalar and spinning cases are identical, the saddle point analysis for the spinning case is identical to what we did in section 4.3.3. As a result, the formula in the scalar case immediately generalizes to the spinning case, and by rewriting (4.83) we obtain that the formula in the spinning case is given by

$$\frac{\lim_{\substack{\nu,m\gg1\\\nu$$

(4.102)

where (a'), (b') are the three-point tensor structure labels of the block. The formula should be evaluated at the same "scattering-crystal" as the scalar case, given by (4.81), and we have implicitly contracted all the SO(d - 1, 1) indices of ρ_2, ρ_4, ρ_4 and the SO(d - 2) indices of $\lambda_1, \lambda_3, \lambda'$.

Subtraction factors

To finish the discussion of spinning sum rules, let us give the subtraction factors that should be used in the spinning case. We consider a more general functional

$$\Psi_{k,\nu,\lambda',f}^{(a),(c)}[\mathcal{G}] = \Psi_{k,\nu,\lambda'}^{+(a),(c)}[f(u',v')\mathcal{G}] - (1 \leftrightarrow 3).$$
(4.103)

As discussed in section 4.3.4, to have nontrivial sum rules for identical external operators, the kernel of the $\Psi_{k,v,\lambda'}^{+(a),(c)}$ functional times the subtraction factor f(u', v') should be antisymmetric under $(1 \leftrightarrow 3)$. For the spinning functionals, the signature under swapping 1 and 3 depends on the transverse representation λ' . We will focus on the case where the external operators are symmetric traceless tensors, so the allowed λ' satisfying the selection rule (4.97) should be a symmetric traceless tensor itself (as an SO(d - 2) representation). Therefore, the spinning functionals in this case are labeled by an integer j', which is the transverse spin. Due to the

celestial structure $\langle \widetilde{\mathcal{P}}_1 \widetilde{\mathcal{P}}_3 \mathcal{P}_{\delta,j'} \rangle$, the functional $\Psi_{k,\nu,j'}^{+(a),(c)}$ has signature $(-1)^{j'}$ under $(1 \leftrightarrow 3)$. Therefore, the nontrivial spinning sum rules are given by

$$\Psi_{k,\nu,j'}^{+(a),(c)}[f^{(-1)^{j'}}(u',\nu')\mathcal{G}] = 0, \qquad (4.104)$$

where the subtraction factors should satisfy

$$f^{\pm}(v', u') = \mp f^{\pm}(u', v'). \tag{4.105}$$

We find that for external operators with spin J_e , the subtraction factors with even j' can be chosen to be

$$f_{k;J_e}^+(u',v') = \begin{cases} \frac{v'-u'}{(u'v')^{\frac{k-2J_e+2}{4}}} & k \equiv 2J_e + 2 \mod 4\\ \frac{(v'-u')^3 - (v'-u')}{4(u'v')^{\frac{k-2J_e+4}{4}}} & k \equiv 2J_e \mod 4 \end{cases}, \qquad k = 2,4,6,\dots, \quad (4.106)$$

which reduces to the scalar case (4.87) when $J_e = 0$. On the other hand, for odd j' we have

$$f_{k;J_e}^{-}(u',v') = \begin{cases} -\frac{1}{(u'v')^{\frac{k-2J_e+1}{4}}} & k \equiv 2J_e+3 \mod 4\\ -\frac{(u'-v')^2-1}{4(u'v')^{\frac{k-2J_e+3}{4}}} & k \equiv 2J_e+1 \mod 4 \end{cases}, \qquad k = 3, 5, 7, \dots.$$
(4.107)

When evaluating these subtraction factors at the scattering crystal (4.81), we find

$$\begin{aligned} f_{k;J_e}^+(u',v')\Big|_{\text{scattering-crystal}} &= -\nu^{2(k-2J_e+1)} \frac{(2m^2-\nu^2)}{m^2(m^2-\nu^2)} \frac{1}{(m^2(m^2-\nu^2))^{\frac{k-2J_e}{2}}}, \quad k = 2, 4, 6, \dots, \\ f_{k;J_e}^-(u',v')\Big|_{\text{scattering-crystal}} &= -\nu^{2(k-2J_e)} \frac{\nu^2}{m^2(m^2-\nu^2)} \frac{1}{(m^2(m^2-\nu^2))^{\frac{k-2J_e-1}{2}}}, \quad k = 3, 5, 7, \dots. \end{aligned}$$

$$(4.108)$$

From the flat space point of view, the expressions for the subtraction factors at the saddle are precisely what we need. We expect that functionals with even j' are related to the $s \leftrightarrow t$ symmetric part of the flat space amplitude. Therefore, the corresponding flat space sum rule should have a factor

$$\left(\frac{1}{s} + \frac{1}{s+u}\right) \frac{1}{(s(s+u))^{\frac{k}{2}}} = \frac{2s+u}{s(s+u)} \frac{1}{(s(s+u))^{\frac{k}{2}}}.$$
 (4.109)

On the other hand, functionals with odd j' should be related to the $s \leftrightarrow t$ antisymmetric part, and the corresponding flat space sum rule should have the factor

$$\left(\frac{1}{s} - \frac{1}{s+u}\right) \frac{1}{(s(s+u))^{\frac{k}{2}}} = \frac{u}{s(s+u)} \frac{1}{(s(s+u))^{\frac{k}{2}}}.$$
 (4.110)

Under the identification $s = m^2$, $u = -v^2$, we see that these expressions agree with $f_{k;J_e}^{\pm}(u', v')$ at the saddle. We make this connection more precise in section 4.4.

Finally, let us explain the allowed range of k in (4.106) and (4.107). In the Regge limit, the subtraction factor f^+ changes the Regge spin by $k - 2J_e + 1$, and f^- changes the Regge spin by $k - 2J_e$. Therefore, for the integral of the functional to be convergent, we must have $k > J_0$ for even j' and $k - 1 > J_0$ for odd j'. For non-perturbative CFTs, we have $J_0 \le 1$. So, we should choose $k \ge 2$ for even j' and $k \ge 3$ for odd j'. As we will show in the next section, k also agrees with the Regge spin of the corresponding flat space sum rule.

Normally, introducing subtraction factors comes with a price that the action of the functional becomes non-vanishing on finitely many light double-traces [56]. However, if J_e is large enough, sometimes we can have subtraction factors that make the Regge behavior of the kernel worse. For these unsubtracted sum rules, the action on all double-traces will remain zero. On the flat space side, the EFT series of the corresponding flat space sum rule should automatically truncate, since there is no denominator in the integrand. As an example, for $J_e = 2$ (which corresponds to flat space gravitons), we have two such subtraction factors from k = 2, k = 3. Their expressions at the saddle are given by

$$f_{k=2;J_e=2}^+(u',v')\Big|_{\text{scattering-crystal}} = -\nu^{-2}(2m^2 - \nu^2),$$

$$f_{k=3;J_e=2}^-(u',v')\Big|_{\text{scattering-crystal}} = 1.$$
 (4.111)

These unsubtracted sum rules for gravitons will be given in section 4.4.

4.4 Flat space interpretation

We now discuss the flat space interpretation of the CFT sum rules derived in the previous section. On the flat space side, we are mainly interested in sum rules of photons and gravitons, which are dual to conserved operators on the CFT side. Our goal is to relate the CFT heavy action formula (4.102) to the heavy state contribution of the flat space sum rule. The heavy action formula consists of a block part (fourth line of (4.102)) from the block insertion, and a kernel part (third line of (4.102)) from the functional. We will be able to understand both of them in flat space. Throughout the discussion, we will study the photon case in detail and explain how to match the CFT result and the flat space result. For the graviton case, we simply give the final relation between CFT sum rules and flat space sum rules.

Hereafter, we use *D* to denote the spacetime dimension of flat space, and d = D - 1 is the spacetime dimension of the CFT.

4.4.1 Conservation at large Δ

The flat space limit of a dispersive functional corresponds to an AdS scattering process where the incoming particles are boosted to high energies. Our main interest is in massless particles like photons and gravitons, which are dual to conserved CFT operators. In this highly boosted limit, conservation will substantially simplify the dictionary that we find. We expect that for external massive particles, it is most natural to use the goldstone equivalence theorem to break them into massless excitations, and use our dictionary for each excitation.

We first note that the conservation condition for CFT three-point structures simplifies when one of the operator dimensions Δ becomes large. For a three point function $\langle O_1(x_1, z_1)O_2(x_2, z_2)O_3(x_3, z_3) \rangle$, with O_1 conserved we can write its conservation condition as

$$0 = \partial_{\mu_1} \langle O_1^{\mu_1 \cdots \mu_J}(x_1) O_2(x_2, z_2) O_3(x_3, z_3) \rangle.$$
(4.112)

We claim that at large $\Delta = \Delta_3$ this simplifies to

$$v_{1,23}^{\mu_1} \langle O_{1;\mu_1\cdots\mu_J} O_2 O \rangle = 0,$$
 (4.113)

where

$$v_{1,23}^{\mu} = \frac{1}{2} \partial_1^{\mu} \log \frac{x_{12}^2}{x_{13}^2} = \frac{x_{12}^{\mu} x_{13}^2 - x_{13}^{\mu} x_{23}^2}{x_{12}^2 x_{13}^2}.$$
 (4.114)

To see why, we can isolate the Δ dependence in the three-point structure:

$$\langle O_1^{\mu_1\cdots\mu_J}(x_1)O_2(x_2,z_2)O_3(x_3,z_3)\rangle = \left(\frac{x_{12}^2}{x_{13}^2x_{23}^2}\right)^{\Delta/2} \times (\cdots),$$
 (4.115)

where (\cdots) does not depend exponentially on Δ . Applying the conservation condition (4.112), the leading contribution in the large- Δ limit comes from the prefactor in (4.115), which gives (4.113).

Let us check this claim in the explicit example $\langle JJO \rangle$, where J is a spin-1 conserved operator, and O is a symmetric traceless tensor. Before imposing conservation, the general three-point function $\langle JJO \rangle$ in the embedding space (using the conventions in (C.9)) is given by

$$\langle J(X_{1}, Z_{1})J(X_{2}, Z_{2})O(X_{3}, Z_{3})\rangle = \frac{c_{1}(-2V_{3})^{J-2}H_{23}H_{13} + c_{2}H_{12}(-2V_{3})^{J} + c_{3}V_{2}H_{13}(-2V_{3})^{J-1} + c_{4}V_{1}H_{23}(-2V_{3})^{J-1} + c_{5}V_{1}V_{2}(-2V_{3})^{J}}{X_{12}^{\frac{\Delta_{1}+\Delta_{2}-\Delta+2-J}{2}}X_{13}^{\frac{\Delta_{1}-\Delta_{2}+\Delta+J}{2}}X_{23}^{\frac{-\Delta_{1}+\Delta_{2}+\Delta+J}{2}}}$$

$$(4.116)$$

Let us now impose the conservation condition (4.112). Starting with $\langle JJO \rangle$, we can see that $\partial_{X_1} \cdot \mathcal{D}_{Z_1} \langle JJO \rangle$ transforms like $\langle \phi JO \rangle$. It therefore has two independent structures; one proportional to V_2 and one proportional to H_{23} . These must independently vanish. Furthermore, we get two more equations from imposing that J_2 be conserved. If however we impose $\Delta_1 = d - 1 = \Delta_2$ as required for a conserved spin 1 operator, we see that only 3 of these equations are independent. Imposing them and denoting $\Delta_3 = \Delta$ we get

$$(4 - 2d - J + \Delta)c_1 + 4Jc_2 + 2c_3 + 2(d - 1 + \Delta)c_4 = 0,$$

$$(2J + 2\Delta)c_2 + (2(d - 1) + J - \Delta)c_3 + 2(d - 1 + \Delta)c_5 = 0,$$

$$(4 - 2d - J + \Delta)c_1 + 4Jc_2 + 2(d - 1 + \Delta)c_3 + 2c_4 = 0,$$

$$(2J + 2\Delta)c_2 - (2(d - 1) + J - \Delta)c_4 + 2(d - 1 + \Delta)c_5 = 0.$$

(4.117)

As mentioned above, of these four equations, only 3 are linearly independent. We can then take the large Δ limit to get

$$c_1 - 2c_4 = 0 \quad 2c_2 + c_4 + 2c_5 = 0 \quad c_3 + c_4 = 0.$$
(4.118)

One can then verify that this is the same as (4.113), which is equivalent to demanding that in the embedding space

$$V_1 \cdot D_{Z_1} \langle O_1(X_1, Z_1) O_2(X_2, Z_2) O(X, Z) \rangle = 0.$$

$$V_2 \cdot D_{Z_2} \langle O_1(X_1, Z_1) O_2(X_2, Z_2) O(X, Z) \rangle = 0,$$
(4.119)

where V_i^A is the usual embedding space structure V_i defined in (C.4) with Z_i^A stripped off, and $D_{Z_i}^A = \left(\frac{d}{2} - 1 + Z_i \cdot \partial_{Z_i}\right) \partial_{Z_i}^A - \frac{1}{2} Z_i^A \partial_{Z_i}^2$ is the Todorov/Thomas operator [30]. Importantly, the simplified conservation condition (4.113) is algebraic. It allows us to describe the conserved operator using a boundary polarization vector with one fewer degree of freedom, which will be necessary for writing a simple correspondence between boundary and bulk polarizations.

4.4.2 CFT 3-point structures and flat space 3-point amplitudes

Our goal now is to interpret the last line of (4.102) in flat space language. The conformal three-point structures will become three-point amplitudes, and their contractions will become a partial wave. Recall that the partial wave decomposition of a spinning flat space amplitude is given by [110]

$$\mathcal{M}(s,u) = s^{\frac{4-D}{2}} \sum_{\rho} n_{\rho}^{(D)} \sum_{ab} (a_{\rho}(s))_{ab} \pi_{\rho,(ab)}, \qquad (4.120)$$

where ρ is an irrep of SO(*d*), *a*, *b* represent the three-point structures (see below), and $n_{\rho}^{(D)} = \frac{2^{d+1}(2\pi)^{d-1}\dim\rho}{\operatorname{vol} S^{d-1}}$ is a normalization factor. This is a generalization of the scalar partial wave decomposition (4.17), where instead of the Gegenbauer polynomial \mathcal{P}_J we now have a more general partial wave $\pi_{\rho,(ab)}$.

The partial wave $\pi_{\rho,(ab)}$ transforms in the representation ρ of the SO(*d*) group that stabilizes $P^{\mu} = p_1^{\mu} + p_2^{\mu}$. It can be obtained by gluing two vertices,

$$\pi_{\rho,(ab)} \equiv (\overline{v}_b(n', e_3, e_4), v_a(n, e_1, e_2)), \qquad (4.121)$$

where the pairing (...,..) represents the contraction of SO(*d*) indices, and $\overline{f}(x) = f(x^*)^*$ is Schwarz reflection. The vertex $v_a(n, e_1, e_2)$ is a three-point amplitude of two massless particles and a massive particle. The massive particle has momentum P^{μ} and transforms in the representation ρ under the little group SO(*d*). We use *a* to label different independent vertices, and we define n^{μ} and e_i^{μ} as

$$n^{\mu} = \frac{p_1^{\mu} - p_2^{\mu}}{\sqrt{(p_1 - p_2)^2}}, \quad e_i^{\mu} = \epsilon_i^{\mu} - p_i^{\mu} \frac{\epsilon_i \cdot P}{p_i \cdot P}.$$
(4.122)

They satisfy $n \cdot e_i = e_i^2 = 0$, $n^2 = 1$. The other vertex can be defined in the same way with momenta p_3^{μ} , p_4^{μ} .

Let us demonstrate how to obtain vertices $v_a(n, e_1, e_2)$ from CFT structures by studying the example of photon scattering. As we saw in the previous section, for a CFT three-point function between two conserved spin one currents and an operator with large Δ , conservation places a simple constraint on the allowed three point structures. For example, when *O* is a traceless symmetric tensor, we have

$$\langle J_{1}(X_{1}, Z_{1})J_{2}(X_{2}, Z_{2})O(X_{5}, Z_{5})\rangle = \frac{c_{1}(-2V_{5})^{J-2}H_{25}H_{15} + c_{2}H_{12}(-2V_{5})^{J} + c_{3}V_{2}H_{15}(-2V_{5})^{J-1} + c_{4}V_{1}H_{25}(-2V_{5})^{J-1} + c_{5}V_{1}V_{2}(-2V_{5})^{J}}{X_{12}^{\frac{2d-\Delta-J}{2}}X_{15}^{\frac{\Delta+J}{2}}X_{25}^{\frac{\Delta+J}{2}}}$$

$$(4.123)$$

At large Δ , conservation demands that we have no terms proportional to V_1 or V_2 . Moreover, at the saddle configuration of (4.102) we find

$$-v_{1,25}|_{\text{saddle}} = v_{2,51}|_{\text{saddle}} = v_{3,46}|_{\text{saddle}} = -v_{4,63}|_{\text{saddle}} = e, \qquad (4.124)$$

ī

where *e* is the unit vector in the time direction. Therefore, if we choose the polarization vectors z_1, z_2 for the conserved currents to have no time component, the conserved structures of $\langle JJO \rangle$ at large Δ will satisfy

$$\langle J_{1}J_{2}O \rangle_{\text{saddle},z_{i}=(0,z_{i}^{x},\vec{z}_{i\perp})}^{(1)} = \frac{(-2V_{5})^{J-2}H_{15}H_{25}}{X_{12}^{\frac{2d-\Delta-J}{2}}X_{15}^{\frac{\Delta+J}{2}}X_{25}^{\frac{\Delta+J}{2}}} \bigg|_{\text{saddle},z_{i}=(0,z_{i}^{x},\vec{z}_{i\perp})},$$

$$\langle J_{1}J_{2}O \rangle_{\text{saddle},z_{i}=(0,z_{i}^{x},\vec{z}_{i\perp})}^{(2)} = \frac{(-2V_{5})^{J}H_{12}}{X_{12}^{\frac{2d-\Delta-J}{2}}X_{15}^{\frac{\Delta+J}{2}}X_{25}^{\frac{\Delta+J}{2}}} \bigg|_{\text{saddle},z_{i}=(0,z_{i}^{x},\vec{z}_{i\perp})}.$$

$$(4.125)$$

We would like to interpret these expressions in such a way that they give rise to flat-space three-point vertices. We can describe the possible vertices using the formalism of [135, 110]. The two massless spin-1 particles have momenta p_1 and p_2 , as well as polarizations e_1 and e_2 . We additionally have the momentum of the third particle $p_3 := P = p_1 + p_2$ and another Lorentz invariant quantity $n \propto p_1 - p_2$. We have the freedom to normalize n, and to shift the polarizations e_i by p_i . As the third particle is massive, we can go to its rest frame and parameterize its momentum as $P = (m, 0, \dots, 0)$. We can then use the symmetry of our saddle point to find the flat space kinematics.

Since the vertices are tensors with SO(d) indices, we can introduce index free notation and contract the indices with polarization vectors $w_1, w_2, \ldots \in \mathbb{C}^d$ for each row of the Young diagram. For example, if ρ is a symmetric traceless tensor, we have a single polarization vector w_1 , and from these quantities we can build two three-point amplitudes:

$$v_{\gamma\gamma J}^{(1)}(n, e_1, e_2) = (n \cdot w_1)^{J-2} (e_1 \cdot w_1) (e_2 \cdot w_1),$$

$$v_{\gamma\gamma J}^{(2)}(n, e_1, e_2) = (n \cdot w_1)^J (e_1 \cdot e_2).$$
(4.126)

These two amplitudes should correspond to (4.125) through the appropriate dictionary.

Our saddle point (4.81) defines an SO $(d-1) \subset$ SO(d, 2) subgroup that fixes the locations of the operators J_1, J_2, O . The vector $v_{5,12}^{\mu}$ points along the direction

preserved by this SO(d-1) at the location of O. In the bulk, there is a corresponding SO(d-1) that preserves n. Thus it is natural to impose

$$(z_5 \cdot v_{5,12})^J \leftrightarrow (w_1 \cdot n)^J. \tag{4.127}$$

The left-hand side is the projection of the polarization of O along the SO(d - 1)-invariant direction, while the right-hand side is the projection of the polarization of the massive particle along the SO(d - 1)-invariant direction.

Therefore we should impose $p_1^{\mu} - p_2^{\mu} \propto (0, v_{5,12}^{\mu})$. Evaluating this on the saddle (4.81) we find $n^{\mu} = (0, \frac{\sqrt{m^2 - v^2}}{m}, \frac{v}{m}, 0, \cdots, 0)$, while we can choose the momenta of the scattering process to be:

$$p_{1} = \frac{1}{2}(m, \sqrt{m^{2} - \nu^{2}}, \nu, 0, \dots, 0), \qquad p_{2} = \frac{1}{2}(m, -\sqrt{m^{2} - \nu^{2}}, -\nu, 0, \dots, 0),$$

$$p_{3} = \frac{1}{2}(m, \sqrt{m^{2} - \nu^{2}}, -\nu, 0, \dots, 0), \qquad p_{4} = \frac{1}{2}(m, -\sqrt{m^{2} - \nu^{2}}, \nu, 0, \dots, 0).$$
(4.128)

Here, p_1 , p_2 are incoming and p_3 , p_4 are outgoing.

It now remains to parameterize the bulk polarization vectors e_1 , e_2 , w_1 in terms of CFT polarization vectors z_1 , z_2 , z_5 to get (4.125) to agree with (4.126). Our CFT polarizations can be expressed as

$$z_1 = (0, z_1^x, \vec{z}_{1\perp}), \quad z_2 = (0, z_2^x, \vec{z}_{2\perp}), \quad z_5 = (z_5^t, z_5^x, \vec{z}_{5\perp}),$$
 (4.129)

where we have explicitly set the time component of z_1 and z_2 to be zero, in accordance with (4.124). Meanwhile, we can express the bulk polarizations as

$$e_{1}^{\mu} = e_{1}^{n} n_{\perp}^{\mu} + (0, 0, 0, \vec{e}_{1\perp}), \quad e_{2}^{\mu} = e_{2}^{n} n_{\perp}^{\mu} + (0, 0, 0, \vec{e}_{2\perp}), \quad w_{1}^{\mu} = (0, w_{1}^{t}, w_{1}^{x}, \vec{w}_{1\perp}),$$
(4.130)

where n_{\perp}^{μ} is a vector perpendicular to n^{μ} , given by

$$n_{\perp}^{\mu} = (0, -\frac{\nu}{m}, \frac{\sqrt{m^2 - \nu^2}}{m}, 0, \dots, 0).$$
 (4.131)

As we argued above, the SO(d - 1) singlets should match between CFT and flat space. In particular, this imposes

$$\frac{H_{15}}{X_{15}}\Big|_{\text{saddle}} \leftrightarrow e_1 \cdot w_1, \quad \frac{H_{25}}{X_{25}}\Big|_{\text{saddle}} \leftrightarrow e_2 \cdot w_1,$$

$$\frac{H_{12}}{X_{12}}\Big|_{\text{saddle}} \leftrightarrow e_1 \cdot e_2, \quad V_{5,12}\Big|_{\text{saddle}} \leftrightarrow -in \cdot w_1, \quad (4.132)$$

which we can achieve by setting

$$z_1^x = e_1^n, \quad \vec{z}_{1\perp} = \vec{e}_{1\perp}, \quad z_2^x = -e_2^n, \quad \vec{z}_{2\perp} = \vec{e}_{2\perp}, \quad z_5^t = -iw_1^t, \quad z_5^x = w_1^x, \quad \vec{z}_5 = \vec{w}_{1\perp}, \quad (4.133)$$

Let us emphasize that even though we use a t superscript for the w_1^t component, it is actually a spatial direction in the bulk, while z_5^t is the time direction in the CFT. This is why we need an *i* factor in their relation. Note also that the relation maps the condition $z_i^2 = 0$ to $e_i^2 = w_1^2 = 0$. If the Young diagram of the representation ρ has more than one row, then we need to introduce more polarization vectors for both the flat space vertex and the CFT three-point structure. The map between these polarizations is the same as the relation between z_5 and w_1 in (4.133).

The polarization map (4.133) turns each CFT three-point structure building block into an amplitude building block, and thus the CFT three-point structures naturally become flat space vertices.

Note that in the CFT sum rule (4.102), the three-point structures appear as commutators of Wightman functions. So we will use the commutator to define vertices, and divide by the sin(...) factor from the commutator by hand. For the cases we consider in this paper (photons and gravitons), the Young diagram of the exchanged representation ρ has at most three rows. We will often write $\rho = (J, j, \tilde{j})$ to denote the length of each row. For external operators O_1, O_2 with quantum numbers (Δ_i, J_i) and exchanged representation ρ , we define the vertices as¹³

$$= (-x_{12}^2)^{\frac{\Delta_1 + \Delta_2}{2}} \left(\frac{x_{15}^2}{x_{25}^2} \right)^{\frac{\Delta_1 - \Delta_2}{2}} \frac{\langle 0 | [O_1(x_{1^+}, z_1), O_2(x_{2^+}, z_2)] O(x_5, z_5, \vec{w}_5)] | 0 \rangle_{(a)}}{2i(\sin(\pi \frac{\tilde{\tau}_{\rho} - \Delta_1 - J_1 - \Delta_2 - J_2}{2}))} \Big|_{\text{saddle},(4.133)}$$

$$(4.134)$$

where w_i on the left-hand side are the flat space polarizations, and z_5 , \vec{w}_5 on the right-hand side are the CFT polarizations. The $2i(\sin(\pi \frac{\tilde{\tau}_{\rho} - \Delta_1 - J_1 - \Delta_2 - J_2}{2}))$ factor comes from taking the commutator, and $\tilde{\tau}_{\rho} = \Delta - J + j + \tilde{j}$ for $\rho = (J, j, \tilde{j})$.¹⁴ We introduce the factor $(-x_{12}^2)^{\frac{\Delta_1 + \Delta_2}{2}} \left(\frac{x_{15}^2}{x_{25}^2}\right)^{\frac{\Delta_1 - \Delta_2}{2}}$ to remove factors of ν .

¹³Reference [136] presents a similar relation between flat space 3-point amplitudes and CFT 3-point structures, but with a different configuration of CFT positions and polarizations X_i, Z_i . We expect that our relation is equivalent up to a choice of conformal frame. ¹⁴We find this $\sin(\pi \frac{\tilde{\tau}_{\rho} - \Delta_1 - J_1 - \Delta_2 - J_2}{2})$ factor by studying the commutator of (C.9). It would be

interesting to determine what $\tilde{\tau_{\rho}}$ should be for general ρ .

As an example, we can apply the polarization map (4.133) to the $\langle JJO \rangle$ CFT three-point structures (4.125). We get

$$(-x_{12}^{2})^{d-1} \left\{ \frac{\langle 0|[J(x_{1^{+}},z_{1}),J(x_{2^{+}},z_{2})]O(x_{5},z_{5})|0\rangle_{(1,2)}}{2i\left(\sin(\pi\frac{\Delta-J-2d}{2})\right)} \right\} \bigg|_{\text{saddle},(4.133)}$$

= $-(-i)^{J}2^{J} \left\{ 2^{-2}(n\cdot w_{1})^{J-2}(e_{1}\cdot w_{1})(e_{2}\cdot w_{1}),(n\cdot w_{1})^{J}(e_{1}\cdot e_{2}) \right\}.$ (4.135)

As expected, the result correctly reproduces the three-point amplitudes (4.126).

The other vertex $\overline{v}(n', e_3, e_4)$ can be defined using the $\langle O_3 O_4 O_6 \rangle$ CFT three-point structure in a similar way. The polarization map is given by

$$z_{3}^{x} = e_{3}^{n'}, \quad \vec{z}_{3\perp} = \vec{e}_{3\perp}, \quad z_{4}^{x} = -e_{4}^{n'}, \quad z_{4\perp} = \vec{e}_{4\perp}, \quad z_{6}^{t} = -iw_{1}^{t}, \quad z_{6}^{x} = w_{1}^{x}, \quad \vec{z}_{6} = \vec{w}_{1\perp}.$$
(4.136)

We again set the time component of z_3 , z_4 to be zero, and

$$e_{3}^{\mu} = e_{3}^{n'} n_{\perp}^{\prime \mu} + (0, 0, 0, \vec{e}_{3\perp}), \quad e_{4}^{\mu} = e_{4}^{n'} n_{\perp}^{\prime \mu} + (0, 0, 0, \vec{e}_{4\perp}), \quad w_{1}^{\mu} = (0, w_{1}^{t}, w_{1}^{x}, \vec{w}_{1\perp}),$$
(4.137)

where

$$n_{\perp}^{\prime \mu} = (0, \frac{\nu}{m}, \frac{\sqrt{m^2 - \nu^2}}{m}, 0, \dots, 0),$$
 (4.138)

which is perpendicular to $n'^{\mu} = (0, \frac{\sqrt{m^2 - \nu^2}}{m}, -\frac{\nu}{m}, 0, \dots, 0) \propto p_3^{\mu} - p_4^{\mu}$. The definition of the $\overline{\nu}(n', e_3, e_4)$ is given by

$$= (-x_{34}^2)^{\frac{\Delta_3 + \Delta_4}{2}} \left(\frac{x_{36}^2}{x_{46}^2} \right)^{\frac{\Delta_3 - \Delta_4}{2}} I_e^{\rho} \left. \frac{\langle 0|O^{\dagger}(x_{6^+}, z_6, \vec{w}_6)[O_4(x_4, z_4), O_3(x_3, z_3)]|0\rangle_{(b)}}{2i(\sin(\pi \frac{\tilde{\tau}_{\rho} - \Delta_3 - J_3 - \Delta_4 - J_4}{2}))} \right|_{\text{saddle},(4.136)}$$

$$(4.139)$$

On the right hand side, I_e^{ρ} is a tensor that reflects all the indices in the time direction. More precisely, for any tensor T with representation ρ , in the index-free notation we have

$$(I_e^{\rho}T)(z,\vec{w}) = T(I_e \cdot z, I_e \cdot \vec{w}), \qquad (4.140)$$

where $I_{e_{\nu}}^{\mu} = \delta^{\mu}{}_{\nu} + 2e^{\mu}e_{\nu}$ is the usual reflection in time direction.

•

Note that (4.139) gives a definition for $\overline{v}(-n', e_3, e_4)$ instead of $\overline{v}(n', e_3, e_4)$, which appears in the actual partial wave. This is to ensure that the two definitions (4.134) and (4.139) are consistent. In particular, one can show that the partial waves are always positive in the forward limit. Furthermore, the definitions for $v(n, e_1, e_2)$ and $\overline{v}(-n', e_3, e_4)$ are related by CRT symmetry on the CFT side. We give more details in appendix C.3, in which we also write down the relation between the partial waves coming from CFT and the Young tableaux basis of [135, 110].

The appearance of I_e^{ρ} in (4.139) is due to the two-point structure $\langle \tilde{O}^{\dagger}(e)\tilde{O}((-e)^+)\rangle$ in the block part of the CFT sum rule (4.102). The effect of this two-point structure is a reflection of all the indices of O in the time direction with some overall factor. For a representation $\rho = (J, j, \tilde{j})$, we find

$$\frac{2^{d} \operatorname{dim}(\rho)}{\operatorname{vol}(\operatorname{SO}(d))} \frac{\langle \widetilde{O}_{a}^{\dagger}(x_{5}) \widetilde{O}^{\overline{b}}(x_{6}^{\dagger}) \rangle}{\left(\langle \widetilde{O}^{\dagger} \widetilde{O} \rangle, \langle O^{\dagger} O \rangle \right)} \bigg|_{\text{saddle}} = 2^{-J+2\Delta} (-1)^{J-j+\widetilde{j}} R_{\rho} \left(I_{e}^{\rho} \right)_{a}^{\overline{b}}, \quad (4.141)$$

where the lower index *a* and upper index \overline{b} are indices in the dual ρ^* and reflected ρ^R representations respectively. As one can see from the left-hand side, the coefficient R_{ρ} depends on our convention of the two-point structure. For the two-point convention (C.5), we have

$$R_{\rho=(J,j,\tilde{j})} = \frac{(j-\tilde{j}+1)(J-j+1)(J-\tilde{j}+2)}{(j+1)(J+1)(J+2)}.$$
(4.142)

This result can be obtained by using weight-shifting operators [39] to derive a recursion relation of R_{ρ} . We give an example in appendix C.3.

Using the momentum configuration and maps of polarizations discussed above, we can now write down the precise flat space interpretation of the block part of the CFT sum rule. Let us define

$$\mathcal{B}_{\Delta,\rho,(a'b')} = \frac{2^{d} \operatorname{dim}(\rho)}{\operatorname{vol}(\operatorname{SO}(d))} \times \frac{\langle 0|[O_{1}(x_{1^{+}}, z_{1}), O_{2}(x_{2^{+}}, z_{2})]O(x_{5})|0\rangle_{(a')} \langle \widetilde{O}^{\dagger}(x_{5})\widetilde{O}(x_{6}^{+})\rangle \langle 0|O^{\dagger}(x_{6}^{+})[O_{4}(x_{4}, z_{4}), O_{3}(x_{3}, z_{3})]|0\rangle_{(b')}}{\left(\langle \widetilde{O}^{\dagger}\widetilde{O}\rangle, \langle O^{\dagger}O\rangle\right)} \bigg|_{\substack{\text{saddle}\\(4.133),(4)}}$$

199

From the definitions (4.134), (4.139), and (4.141), we find

$$\frac{\mathcal{B}_{\Delta,\rho,(a'b')}}{-2(2\sin(\pi\frac{\tilde{\tau}_{\rho}-\Delta_{1}-J_{1}-\Delta_{2}-J_{2}}{2})\sin(\pi\frac{\tilde{\tau}_{\rho}-\Delta_{3}-J_{3}-\Delta_{4}-J_{4}}{2}))} = 2^{-J+2\Delta-\Delta_{1}-\Delta_{2}-\Delta_{3}-\Delta_{4}}m^{-\Delta_{1}-\Delta_{2}-\Delta_{3}-\Delta_{4}}\nu^{2(\Delta_{1}+\Delta_{3})}\left(m-\sqrt{m^{2}-\nu^{2}}\right)^{-\Delta_{1}+\Delta_{2}-\Delta_{3}+\Delta_{4}}} R_{\rho}\pi_{\rho,(a'b')}$$
(4.144)

where $\pi_{\rho,(a'b')}$ is the flat space partial wave (4.121) in the CFT three-point structure basis, and R_{ρ} is given by (4.142). Note that when turning the vertex $\overline{\nu}(-n', e_3, e_4)$ in (4.139) into $\overline{\nu}(n', e_3, e_4)$ to get the partial wave, we get a $(-1)^{J-j+\tilde{j}}$ factor, which exactly cancels with the same factor in (4.141).

Recall that in the scalar case, where the exchanged operators are symmetric traceless tensors, we define a heavy action $\Psi[\Delta, J]$ in (4.28) by rescaling the action of the functional on the block by a positive factor. Relation (4.144) in fact suggests a way to generalize this definition to more general representation ρ . Since the other parts of the sum rule should not know about the exchanged quantum numbers of the inserted block, the heavy action should remove the *m*- and ρ -dependent factors that we do not expect to appear in the flat space answer, such as the R_{ρ} coefficient. For external operators with quantum numbers (Δ_i, J_i) and a general exchanged representation ρ , we define the large $\Delta \sim m$ heavy action $\Psi[\Delta, \rho]$ to be¹⁵

$$\Psi[\Delta,\rho]_{(a'b')} \equiv \frac{2^{J-2\Delta-\Delta_1-\Delta_2-\Delta_3-\Delta_4}m^{\Delta_1+\Delta_2+\Delta_3+\Delta_4-2-\frac{d}{2}}}{R_{\rho}} \times \frac{\Psi[G^s_{\Delta,\rho,(a'b')}]}{2\sin(\pi\frac{\tilde{\tau}_{\rho}-\Delta_1-J_1-\Delta_2-J_2}{2})\sin(\pi\frac{\tilde{\tau}_{\rho}-\Delta_3-J_3-\Delta_4-J_4}{2})}.$$
(4.145)

4.4.3 Kernels and the shock frame

We now consider the kernel part of our spinning sum rules. It is given by evaluating a celestial structure $\langle \tilde{\mathcal{P}}_1 \tilde{\mathcal{P}}_3 \mathcal{P}_{\delta,j'} \rangle$ and a dual structure $\langle 0|O_4 \mathbf{L}[O]O_2|0\rangle^{-1}$ at the scattering-crystal. In the heavy action formula, we first evaluate these structures at the scattering crystal (4.81), and then we can use the maps of CFT and flat space polarizations (4.133), (4.136) to get the corresponding flat space answer. We claim that the resulting expression can be seen to correspond to a shockwave amplitude in flat space. The shockwave amplitude is defined as the amplitude of an elastic

¹⁵In the scalar case (4.28), the factor $q_{\Delta,J}$ in the heavy action definition comes from the OPE coefficient of Mean Field Theory. So, to find the analogous factor for more general external operators, one can consider the corresponding OPE coefficients in the Mean Field Theory. However, we do not do this in this paper, and the definition (4.145) is good enough for our purpose.

scattering of a probe particle X and two shockwave gravitons g^* [34],

$$g^*(p_1)X(p_2) \to g^*(p_3)X(p_4).$$
 (4.146)

In our case, we will take the probe particle *X* to be a graviton. The two shockwave gravitons are dual to the two light transformed operators $L[O_1]$, $L[O_3]$ in the CFT functional. A convenient choice of momenta for shockwave amplitudes is given by (in all incoming convention)

$$p_{1}^{\mu} = p_{\overline{s}} \overline{s}^{\mu} + q_{1}^{\mu}, \qquad p_{2}^{\mu} = p_{s} s^{\mu} - q_{3}^{\mu}$$

$$p_{3}^{\mu} = -p_{\overline{s}} \overline{s}^{\mu} + q_{3}^{\mu}, \qquad p_{4}^{\mu} = -p_{s} s^{\mu} - q_{1}^{\mu}, \qquad (4.147)$$

where s^{μ} and \overline{s}^{μ} are two null directions, and the polarization of the shock gravitons should be in the \overline{s}^{μ} direction. The transverse momenta q_1^{μ}, q_3^{μ} are perpendicular to the null directions, and in order to make the external momenta on-shell, we must have $q_1^2 = q_3^2 = 0$. We will call the momentum configuration (4.147) the shock frame. More explicitly, let D = d + 1 be the spacetime dimension of the bulk, in the shock frame we have two null vectors $s^{\mu}, \overline{s}^{\mu}$ and D - 2 transverse unit vectors $v_{1;\perp}^{\mu}, v_{2;\perp}^{\mu}, \dots, v_{D-2;\perp}^{\mu}$ satisfying

$$s^{2} = \overline{s}^{2} = s \cdot v_{i;\perp} = \overline{s} \cdot v_{i;\perp} = 0, \quad s \cdot \overline{s} = -2, \quad v_{i;\perp} \cdot v_{j;\perp} = \delta_{ij}.$$
(4.148)

In the previous subsection, we have shown that in order to match the blocks to flat space partial waves, we should choose the flat space momentum to be (4.128). The shock frame momenta (4.147) should also agree with these center of mass frame momenta (4.128). By relating the two frames, we obtain that in terms of the bulk Minkowski coordinates used in (4.128), the vectors in the shock frame can be written as

$$\overline{s}^{\mu} = \left(\frac{\sqrt{m^{2}-\nu^{2}}}{m}, 1, 0, \frac{i\nu}{m}, 0, \dots, 0\right) \\
s^{\mu} = \left(\frac{\sqrt{m^{2}-\nu^{2}}}{m}, -1, 0, \frac{i\nu}{m}, 0, \dots, 0\right) \\
v^{\mu}_{1;\perp} = \left(\frac{i\nu}{m}, 0, 0, \frac{\sqrt{m^{2}-\nu^{2}}}{m}, 0, \dots\right) \\
v^{\mu}_{2;\perp} = (0, 0, 1, 0, 0, \dots, 0) \\
v^{\mu}_{3;\perp} = (0, 0, 0, 0, 1, 0, \dots, 0) \\
\vdots \\
v^{\mu}_{D-2;\perp} = (0, 0, 0, 0, 0, 0, \dots, 1),$$
(4.149)

$$p_{\overline{s}} = p_s = \frac{\sqrt{m^2 - v^2}}{2}, \quad q_1^{\mu} = -\frac{iv}{2}v_{1;\perp}^{\mu} + \frac{v}{2}v_{2;\perp}^{\mu}, \quad q_3^{\mu} = \frac{iv}{2}v_{1;\perp}^{\mu} + \frac{v}{2}v_{2;\perp}^{\mu}. \quad (4.150)$$

One motivation for relating the CFT kernel to the shock frame is that we set the CFT polarization vectors z_1 , z_3 to be x_{1+0} , x_{30} in the CFT sum rule (4.102). Using the map of CFT and flat space polarizations (4.133),(4.136) derived in the previous subsection, we find that in flat space the two polarizations become (up to $e_i \sim e_i + \alpha p_i$ gauge redundancy)

$$x_{1+0}|_{(4.133)} \sim i \frac{m(m - \sqrt{m^2 - v^2})}{v^2} \overline{s}^{\mu}, \qquad x_{30}|_{(4.136)} \sim i \frac{m(m - \sqrt{m^2 - v^2})}{v^2} \overline{s}^{\mu},$$
(4.151)

where \overline{s}^{μ} is given by (4.149). We see that the polarizations of particles 1 and 3 should be set to the same null direction \overline{s}^{μ} , which agrees with the fact that the two shock gravitons in the shockwave amplitude should have the same polarizations along the direction of the shock.

We can now discuss the CFT kernel in the shock frame. Explicitly, evaluating the celestial three-point structure at the saddle configuration gives

$$\left. \left\langle \widetilde{\mathcal{P}}_{\delta_{1}}^{\dagger}(x_{1+0}) \widetilde{\mathcal{P}}_{\delta_{3}}^{\dagger}(x_{30}) \mathcal{P}_{\delta,j'}(z_{0},w_{0}) \right\rangle \right|_{\text{saddle}} = (-1)^{j'} (-i)^{-\delta} 2^{j'+\delta-\widetilde{\delta}_{1}-\widetilde{\delta}_{3}} v^{\widetilde{\delta}_{1}+\widetilde{\delta}_{3}} (m - \sqrt{m^{2} - v^{2}})^{-\widetilde{\delta}_{1}-\widetilde{\delta}_{3}} (\vec{n}_{0} \cdot \vec{w}_{0\perp})^{j'}, \quad (4.152)$$

where we have introduced a polarization $w_0 = (0, 0, \vec{w}_{0\perp})$ for the transverse indices of $\mathcal{P}_{\delta,j'}$, and $\vec{n}_0 = \vec{y}_0/|\vec{y}_0|$ is the unit vector in the transverse \vec{y}_0 direction in the scattering crystal (4.81).

For each transverse spin j', the celestial three-point function has a unique structure which gives (4.152) at the saddle. On the other hand, the dual structure $\langle 0|O_4 \mathbf{L}[O]O_2|0\rangle^{-1}$ can have multiple tensor structures. Moreover, they are subject to constraints from imposing conservation on O_2 and O_4 . Interestingly, the space of allowed structures becomes easier to study when we just focus on the saddle configuration. This fact can be seen from the identity

$$-V_{2,40}V_{0,24}\big|_{\text{saddle},z_i=(0,z_i^x,\vec{z}_{i\perp})} = H_{20}\big|_{\text{saddle},z_i=(0,z_i^x,\vec{z}_{i\perp})}, \qquad (4.153)$$

and a similar identity for $V_{4,02}$ and H_{40} . Note that we also set the time component of the polarizations to zero, which is due to the large- Δ conservation analysis in section

and
4.4.2. This identity allows us to just remove all $V_{2,40}$, $V_{4,02}$ structures when writing down the basis for the spinning dual structures. Additionally, it can be shown that the basis constructed without $V_{2,40}$, $V_{4,02}$ gives the correct number of independent tensor structures of continuous-spin three-point function of two conserved operators, which should be given by [48, 15]

$$\left(\lambda' \otimes \operatorname{Res}_{\operatorname{SO}(d-2)}^{\operatorname{SO}(d-1)} \rho_2 \otimes \rho_4\right)^{\operatorname{SO}(d-2)}.$$
(4.154)

We give the argument for this statement in appendix C.4, and also explain how to compute these spinning dual structures.

As an example, let us consider j' = 0. Focusing just on the spinning part of O_2, O_4 , we see that the allowed structures should be

$$H_{20}^{J_2-n}H_{40}^{J_4-n}H_{24}^n, (4.155)$$

where $n = 0, 1, ..., \min(J_2, J_4)$. The number of structures is $\min(J_2, J_4) + 1$, which agrees with (4.154).

We now study the four-photon example in detail, where all external operators should be conserved spin-1 currents. The selection rule (4.97) implies that the allowed values of transverse spin should be j' = 0, 1, 2. Using the algorithm explained in appendix C.4, we obtain that the dual structures for j' = 0 in the large v limit are given by (for $4 > 0 > 2^{-}$)

$$\left\{ \left(\langle 0|J(X_4, Z_4) \mathbf{L}[O](X_0, Z_0) J(X_2, Z_2) | 0 \rangle^{(1)} \right)^{-1}, \left(\langle 0|J(X_4, Z_4) \mathbf{L}[O](X_0, Z_0) J(X_2, Z_2) | 0 \rangle^{(2)} \right)^{-1} \right\}$$

$$= 2^{\frac{3d-5}{2} - i\nu} e^{-\frac{i\pi}{4}} \sqrt{\frac{\nu}{\pi}} \operatorname{vol} \operatorname{SO}(d-2) \frac{1}{X_{24}^{\frac{4-\Delta_F - J_F}{2}} X_{02}^{\frac{\Delta_F + J_F}{2}} (-X_{04})^{\frac{\Delta_F + J_F}{2}}}$$

$$\times \left\{ \frac{16(d-1)}{d-2} (-2V_{0,42})^{J_F - 2} H_{20} H_{40} + \frac{4}{d-2} (-2V_{0,42})^{J_F} H_{24}, \frac{4}{d-2} (-2V_{0,42})^{J_F - 2} H_{20} H_{40} + \frac{1}{d-2} (-2V_{0,42})^{J_F} H_{24} \right\} + \dots,$$

$$(4.156)$$

where $\Delta_F = d$, $J_F = \frac{2-d}{2} + i\nu$, and ... are subleading terms at large ν . The first structure is dual to the light-transformed structure $\langle 0|J\mathbf{L}[O]J|0\rangle$ with $H_{20}H_{40}$ (see (C.67)), and the second structure is dual to the one with H_{24} .

Then, the next step is to evaluate (4.156) at the saddle (after applying a \mathcal{T}_2 to (4.156) to make the causality configuration agree with the saddle). This will give us an

expression that depends on the external polarizations z_2 , z_4 (note that z_0 is fixed in (4.81)). For example, for the H_{24} structure, we have

$$\frac{H_{24}}{X_{24}}\Big|_{\text{saddle}, z_i = (0, z_i^x, \vec{z}_{i\perp})} = -z_2^x z_4^x + \vec{z}_{2\perp} \cdot \vec{z}_{4\perp}.$$
(4.157)

Using (4.133) and (4.136) to map this to flat space, we find

$$-z_{2}^{x}z_{4}^{x} + \vec{z}_{2\perp} \cdot \vec{z}_{4\perp} \to -e_{2}^{n}e_{4}^{n'} + \vec{e}_{2\perp} \cdot \vec{e}_{4\perp} = e_{2\mu_{2}}e_{4\mu_{4}} \left(-n_{\perp}^{\mu_{2}}n_{\perp}^{\prime\mu_{4}} + \delta_{\perp}^{\mu_{2}\mu_{4}}\right). \quad (4.158)$$

Our goal is to rewrite this expression in the shock frame (4.149). Because of the gauge redundancy, we can shift everything that is contracted with e_2 by p_2 . Furthermore, thanks to the momentum configuration we choose in section 4.4.2, we have $e_2 \cdot p_1 = 0$, so we can also shift by p_1 . Similarly, we can shift things that are contracted with e_4 by p_3 , p_4 . Using this, we can make a gauge choice to remove all the $s^{\mu}, \overline{s}^{\mu}$, and we find that the H_{24} structure mapped to the shock frame can be written as

$$\frac{H_{24}}{X_{24}}\Big|_{\text{saddle},(4.133),(4.136)} = e_{2\mu_2}e_{4\mu_4}\left(\delta_{\perp;D-4}^{\mu_2\mu_4} - \frac{1}{\nu^2(m^2 - \nu^2)^2}(\mathcal{V}_2^{\mu_2}\mathcal{V}_4^{\mu_4} + \mathcal{W}_2^{\mu_2}\mathcal{W}_4^{\mu_4})\right),\tag{4.159}$$

where we define

$$\begin{aligned} \mathcal{V}_{2}^{\mu_{2}} &= m^{2}q_{1}^{\mu_{2}} + (m^{2} - \nu^{2})q_{3}^{\mu_{2}}, \\ \mathcal{V}_{4}^{\mu_{4}} &= (m^{2} - \nu^{2})q_{1}^{\mu_{4}} + m^{2}q_{3}^{\mu_{4}}, \\ \mathcal{W}_{2}^{\mu_{2}} &= m^{2}q_{1}^{\mu_{2}} - (m^{2} - \nu^{2})q_{3}^{\mu_{2}}, \\ \mathcal{W}_{4}^{\mu_{4}} &= (m^{2} - \nu^{2})q_{1}^{\mu_{4}} - m^{2}q_{3}^{\mu_{4}}. \end{aligned}$$
(4.160)

The momenta q_1^{μ}, q_3^{μ} are given by (4.149), (4.150), and $\delta_{\perp;D-4}^{\mu_2\mu_4}$ is the metric in the (D-4)-dimensional transverse space spanned by $v_{3;\perp}^{\mu}, \ldots, v_{D-2;\perp}^{\mu}$. Similarly, the structure with $H_{20}H_{40}$ becomes

$$\frac{H_{20}H_{40}}{X_{20}X_{40}}\Big|_{\text{saddle},(4.133),(4.136)} = \frac{1}{4\nu^2(m^2 - \nu^2)}e_{2\mu_2}e_{4\mu_4}\mathcal{V}_2^{\mu_2}\mathcal{V}_4^{\mu_4}.$$
 (4.161)

Using (4.156), (4.159), and (4.161), we can then obtain the flat space kernels for the two j' = 0 dual structures.

For nonzero j', we also need to contract the transverse indices carried by O in the dual structure and $\mathcal{P}_{\delta,j'}$ in the celestial structure. Let us define

$$\mathcal{K}_{\nu,j'}^{4\gamma,(a)} = \left(\langle 0|J(x_4, z_4) \mathbf{L}[\mathcal{O}_{j'}](x_0, z_0) J(x_{2^+}, z_2) |0\rangle^{(a)} \right)^{-1} \left\langle \widetilde{\mathcal{P}}_{\delta_J}^{\dagger}(x_{1^+0}) \widetilde{\mathcal{P}}_{\delta_J}^{\dagger}(x_{30}) \mathcal{P}_{\delta,j'}(z_0) \right\rangle \Big|_{\substack{\text{saddle}, (4.133), (4.136)\\\nu \to \infty}},$$

$$(4.162)$$

where $\nu \to \infty$ indicates that we take the leading term at large ν . The results for the four-photon kernels are given by

$$\mathcal{K}_{\nu,j'=0}^{4\gamma,(1)} = \frac{K_{\nu}^{4\gamma}}{2} e_{2\mu_{2}} e_{4\mu_{4}} \left((\mathcal{V}_{2}^{\mu_{2}} \mathcal{V}_{4}^{\mu_{4}} + \mathcal{W}_{2}^{\mu_{2}} \mathcal{W}_{4}^{\mu_{4}} - \nu^{2} (m^{2} - \nu^{2})^{2} \delta_{\perp;D-4}^{\mu_{2}\mu_{4}}) - (D-2) \mathcal{V}_{2}^{\mu_{2}} \mathcal{V}_{4}^{\mu_{4}} \right),
\mathcal{K}_{\nu,j'=0}^{4\gamma,(2)} = \frac{K_{\nu}^{4\gamma}}{8} e_{2\mu_{2}} e_{4\mu_{4}} \left(\mathcal{W}_{2}^{\mu_{2}} \mathcal{W}_{4}^{\mu_{4}} - \nu^{2} (m^{2} - \nu^{2})^{2} \delta_{\perp;D-4}^{\mu_{2}\mu_{4}} \right),
\mathcal{K}_{\nu,j'=1}^{4\gamma,(1)} = K_{\nu}^{4\gamma} e_{2\mu_{2}} e_{4\mu_{4}} \mathcal{W}_{2}^{\mu_{2}} \mathcal{W}_{4}^{\mu_{4}},
\mathcal{K}_{\nu,j'=2}^{4\gamma} = -\frac{4K_{\nu}^{4\gamma}}{(D-1)} e_{2\mu_{2}} e_{4\mu_{4}} \left(\mathcal{W}_{2}^{\mu_{2}} \mathcal{W}_{4}^{\mu_{4}} + \frac{\nu^{2} (m^{2} - \nu^{2})^{2}}{D-4} \delta_{\perp;D-4}^{\mu_{2}\mu_{4}} \right), \tag{4.163}$$

where

$$K_{\nu}^{4\gamma} = \frac{2^{-\frac{d}{2}+i\nu}e^{-\pi\nu}m^{-d}\nu^{-4+3d}(m^2-\nu^2)^{-2-\frac{d}{2}}(m-\sqrt{m^2-\nu^2})^{2-d}}{d-2}.$$
 (4.164)

Note that for j' = 1 there are two tensor structures (see (C.9)). However, they are just related by $2 \leftrightarrow 4$, and as we will see later, kernels that are related by $2 \leftrightarrow 4$ will give the same sum rule, so we just include one of the structures here.

Finally, let us comment on the symmetry properties of the kernel. On the CFT side, the celestial three-point structure has a stabilizer group SO(d - 3), which is also the stabilizer group of the scattering crystal (4.81). In the shock frame, this group becomes the SO(D - 4) that fixes the shockwave amplitude momenta (4.147), and the group only rotates in the $v_{3;\perp}^{\mu}, \ldots, v_{D-2;\perp}^{\mu}$ directions. On the other hand, for the continuous-spin dual structure we can choose a conformal frame to fix all three points to be on the same line, and fix the continuous-spin polarization vector in a null direction. The configuration should be fixed by an SO(d-2), and the transverse spin j' is a representation of this group. In flat space, this should correspond to an SO(D - 3). From the expressions of the kernels (4.163), it is not clear what this SO(D - 3) should be. It turns out that (at least in the Regge limit) the group can be understood as the rotation group that fixes the $s^{\mu}, \overline{s}^{\mu}, v_{2;\perp}^{\mu}$ shock frame vectors. To see this we will first need to study how to apply the kernels (4.163) to four-photon amplitudes and write down the full sum rules for photons. We will therefore explain this SO(D - 3) group and the meaning of transverse spin in the bulk in section 4.4.5.

4.4.4 Review: flat space sum rules for photons

After understanding how to map each part of the CFT bulk-point limit formula (4.102) to flat space, we can now derive the dictionary between CFT and flat space sum rules for photons and gravitons. Let us first review flat space sum rules for photons [137, 110]. The graviton sum rules will be considered in section 4.4.6.

The four-photon amplitude $\mathcal{M}_{4\gamma}(p_i, \epsilon_i)$ is a function of photon momenta p_i and polarizations ϵ_i for i = 1, 2, 3, 4. They should satisfy $p_i^2 = p_i \cdot \epsilon_i = \epsilon_i^2 = 0$ and $\epsilon_i \sim \epsilon_i + \#p_i$. In general, the amplitude can be written as a sum of Lorentzinvariant polynomials of p_i, ϵ_i times functions of Mandelstam variables defined in (4.16). Depending on the choice of the polynomials, sometimes the functions of Mandelstam variables can develop spurious poles. It turns out that there exists a "local module" such that the amplitude will not have any spurious poles [138]. Namely, an amplitude that is a polynomial can always be written as generators of the local module times polynomials of Mandelstam variables.

To describe the generators of the local module, let us introduce some basic building blocks, ¹⁶

$$H_{ij} = F^{\mu}_{i\nu} F^{\nu}_{j\mu}, \qquad H_{ijk} = F^{\mu}_{i\nu} F^{\nu}_{j\sigma} F^{\sigma}_{k\mu}, H_{ijkl} = F^{\mu}_{i\nu} F^{\nu}_{j\sigma} F^{\sigma}_{k\rho} F^{\rho}_{l\mu}, \qquad V_{i,jk} = p_{k\mu} F^{\mu}_{i\nu} p^{\nu}_{j}, \qquad (4.165)$$

where $F_{i\nu}^{\mu} = p_i^{\mu} \epsilon_{i\nu} - \epsilon_i^{\mu} p_{i\nu}$. We also define

$$X_{ijkl} = H_{ijkl} - \frac{1}{4}H_{ij}H_{kl} - \frac{1}{4}H_{ik}H_{jl} - \frac{1}{4}H_{il}H_{jk},$$

$$S = V_{1,24}H_{234} + V_{2,31}H_{341} + V_{3,42}H_{412} + V_{4,13}H_{123}.$$
(4.166)

Then, the four-photon amplitude can be written as

$$\mathcal{M}_{4\gamma}(s,u) = \left(H_{14}H_{23}\mathcal{M}_{4\gamma}^{(1)}(s,u) + X_{1243}\mathcal{M}_{4\gamma}^{(2)}(s,u) + \text{permutations}\right) + S\mathcal{M}_{4\gamma}^{(3)}(s,t),$$
(4.167)

where $\mathcal{M}_{4\gamma}^{(1)}$, $\mathcal{M}_{4\gamma}^{(2)}$ are symmetric in their two arguments, and $\mathcal{M}_{4\gamma}^{(3)}$ is symmetric under all permutations of *s*, *t*, *u*.

To write down dispersion relations for $\mathcal{M}_{4\gamma}(s, u)$, we need to study how the polarization structures behave in the fixed-*u* Regge limit, where we take $s \to \infty$. This will tell us how many subtractions we should have in the dispersion relations. Moreover, for fixed-*u* sum rules it is useful to separately consider structures that are symmetric and antisymmetric under $s \leftrightarrow t$. The polarization structures in the Regge limit scale as [110]

$$H_{12}, H_{14} \sim s, \quad H_{13} \sim s^0, \quad X_{1234}, X_{1243} - X_{1324} \sim s, \quad X_{1243} + X_{1324} \sim s^2, \quad S \sim s^2,$$

$$(4.168)$$

¹⁶We are using V and H for the building blocks of both CFT tensor structures and flat space amplitudes. We hope the meaning of the notations will be clear from the context and this will not cause too much confusion.

where other structures can be obtained by permutations.

The amplitude itself $\mathcal{M}_{4\gamma}$ satisfies boundedness and analyticity properties due to unitarity and causality [108, 13, 109, 111]. As shown by [111], for a suitably smeared amplitude,

$$\mathcal{M}_{\Psi}(s) \equiv \int_0^M dp \Psi(p) \mathcal{M}(s, -p^2), \qquad (4.169)$$

along any complex direction of s it should satisfy

$$|\mathcal{M}_{\Psi}(s)|_{s \to \infty} \le s \times \text{constant.}$$
(4.170)

The Regge behaviors (4.168), (4.170) then combine to give boundedness conditions on each coefficient function $\mathcal{M}^{(i)}$. For example, since $X_{1243} - X_{1324} \sim s$ in the Regge limit, $\mathcal{M}^{(2)}(s, u) - \mathcal{M}^{(2)}(t, u)$ cannot grow faster than s^0 in the Regge limit. Performing this analysis for all the structures, we can then determine the subtractions needed for the dispersion relation of each coefficient $\mathcal{M}^{(i)}$. The complete list of flat space fixed-*u* sum rules for photons is given by

$$\left\{ C_{4\gamma;k,u}^{+(1,2,3,4,5)} \right\} = -\oint_{\infty} \frac{ds}{4\pi i} \frac{1}{(-st)^{\frac{k}{2}}} \left\{ (s-t)\mathcal{M}_{4\gamma}^{(1,2)+}(s,u), (s-t)\left(\frac{-u}{2}\right)\mathcal{M}_{4\gamma}^{(3)}(s,t), \right. \\ \left. \mathcal{M}_{4\gamma}^{(2)-}(s,u), \frac{(s-t)}{(-st)}\mathcal{M}_{4\gamma}^{(1)}(s,t) \right\}, \quad k = 2, 4, \dots, \\ \left\{ C_{4\gamma;k,u}^{-(1,2)} \right\} = -\oint_{\infty} \frac{ds}{4\pi i} \frac{1}{(-st)^{\frac{k-1}{2}}} \left\{ \mathcal{M}_{4\gamma}^{(1)-}(s,u), \frac{(s-t)}{(-st)}\mathcal{M}_{4\gamma}^{(2)}(s,t) \right\}, \quad k = 3, 5, \dots,$$

$$\left\{ (4.171) \right\} = -\int_{\infty} \frac{ds}{4\pi i} \frac{ds}{(-st)^{\frac{k-1}{2}}} \left\{ \mathcal{M}_{4\gamma}^{(1)-}(s,u), \frac{(s-t)}{(-st)}\mathcal{M}_{4\gamma}^{(2)}(s,t) \right\}, \quad k = 3, 5, \dots,$$

where we have used the notation $\mathcal{M}^{(i)\pm}(s, u) = \mathcal{M}^{(i)}(s, u) \pm \mathcal{M}^{(i)}(t, u)$. For later convenience, we introduce an additional $\frac{-u}{2}$ factor for $\mathcal{M}^{(3)}_{4\gamma}$. This would also make all the sum rules have the same power counting since the *S* structure has one more $p_i \cdot p_j$ compared to the other structures. The parameter *k* is the Regge spin of the sum rule. A sum rule with Regge spin *k* would be convergent if the amplitude \mathcal{M} satisfies $\mathcal{M}/s^k \to 0$ in the Regge limit.

At low energy, the amplitude can be computed by an EFT with cutoff M. To derive bounds from the dispersion relations, one can deform the contour in (4.171) and separate the contributions from high energy $s \ge M^2$ and low energy $s < M^2$. Our main focus is the high energy part, where one decomposes the amplitude into SO(d) partial waves and imposes unitarity. When applying the dispersion relations (4.171) to the partial wave expansion (4.120), for each sum rule *C* we can write its heavy contribution as

$$\left\langle C[m^2,\rho]_{(ab)} \right\rangle,$$
 (4.172)

where $\langle \cdots \rangle$ is a heavy average defined as

$$\langle \cdots \rangle = \frac{1}{\pi} \sum_{\rho} n_{\rho}^{(D)} \sum_{ab} \int_{M^2}^{\infty} \frac{dm^2}{m^2} m^{4-D} \operatorname{Im}(a_{\rho}(s))_{ab} (\cdots) .$$
 (4.173)

For example, for the $C_{4\gamma;k,u}^{+(1)}$ sum rule in (4.171), we have

$$C_{4\gamma;k,u}^{+(1)}[m^2,\rho]_{(ab)} = \frac{(2m^2+u)}{[m^2(m^2+u)]^{\frac{k}{2}}} \pi_{\rho,(ab)}^{(1)}(m^2,u), \qquad (4.174)$$

where we have decomposed the partial wave into different polarization structures similar to (4.167), so $\pi_{\rho,(ab)}^{(1)}$ is the $H_{14}H_{23}$ component of the partial wave.

This dictionary we obtain below in sections 4.4.5 and 4.4.6 will be a relation between the action of CFT sum rules on conformal blocks in the bulk-point limit (4.102) and the action of flat space sum rules on partial waves (4.172). This will allow us to find CFT sum rules with positive action on conformal blocks with large Δ straightforwardly from the flat space result.

4.4.5 From CFT to flat space for photons

Now, we combine the discussion about the block/partial wave in section 4.4.2 and the discussion about the kernel/shockwave amplitude in section 4.4.3 to write down the relation between four-photon flat space sum rules and the bulk-point limit of our CFT sum rules.

When combining the two parts, we first have to understand how to contract the indices of the external operators. Recall that in the CFT functional, the polarizations z_1, z_3 are set to be $z_1 = x_{1+0}, z_3 = x_{30}$. As discussed in 4.4.3, this implies that in the bulk, we should set $e_1^{\mu} = e_3^{\mu} = i \frac{m(m - \sqrt{m^2 - v^2})}{v^2} \overline{s}^{\mu}$. For 2 and 4, we have to contract the indices of the kernel and the block. Note that we have chosen the CFT polarization vectors to have no time component. In the bulk, this means that e_2 has no time and n^{μ} components (see (4.130)), and similarly e_4^{μ} has no time and n'^{μ} components. Let us consider the following index contraction in the CFT sum rule:

$$(z_2^x K^x + \vec{z}_{2\perp} \cdot \vec{K}_{\perp}) \odot (z_2^x G^x + \vec{z}_{2\perp} \cdot \vec{G}_{\perp}), \qquad (4.175)$$

where K^x , \vec{K}_{\perp} are from the kernel and G^x , \vec{G}_{\perp} are from the block, and the symbol \odot represents the index contraction. For spin-1, it is simply stripping off the z_2 's and contracting both sides, and we see that after the index contraction, we should get $K^x G^x + \vec{K}_{\perp} \cdot \vec{G}_{\perp}$. On the other hand, mapping (4.175) to the bulk, we have

$$(-e_{2}^{n}K^{x} + \vec{e}_{2\perp} \cdot \vec{K}_{\perp}) \odot (-e_{2}^{n}G^{x} + \vec{e}_{2\perp} \cdot \vec{G}_{\perp})$$

= $e_{2\mu_{2}}(-n_{\perp}^{\mu_{2}}K^{x} + \delta_{\perp}^{\mu_{2}i}\vec{K}_{\perp i}) \odot e_{2\rho_{2}}(-n_{\perp}^{\rho_{2}}G^{x} + \delta_{\perp}^{\rho_{2}i}\vec{G}_{\perp i})$
= $e_{2\mu_{2}}K^{\mu_{2}} \odot e_{2\rho_{2}}G^{\rho_{2}},$ (4.176)

where K^{μ_2} is $-n_{\perp}^{\mu_2}K^x + \delta_{\perp}^{\mu_2 i}\vec{K}_{\perp i}$ up to a shift by $p_1^{\mu_2}, p_2^{\mu_2}$ and similarly for G^{ρ_2} . For example, in the j' = 2 kernel given by (4.163) we have $K^{\mu_2} \propto m^2 q_1^{\mu_2} - (m^2 - v^2) q_3^{\mu_2}$. To correctly reproduce $K^x G^x + \vec{K}_{\perp} \cdot \vec{G}_{\perp}$, we should perform a polarization sum in e_2 using¹⁷

$$e_{2\mu_2}e_{2\rho_2} \to \eta_{\mu_2\rho_2} - \frac{p_{1\mu_2}p_{2\rho_2} + p_{1\rho_2}p_{2\mu_2}}{p_1 \cdot p_2}.$$
(4.177)

Then, we obtain

$$K^{\mu_2}G^{\rho_2}\left(\eta_{\mu_2\rho_2} - \frac{p_{1\mu_2}p_{2\rho_2} + p_{1\rho_2}p_{2\mu_2}}{p_1 \cdot p_2}\right) = K^x G^x + \vec{K}_\perp \cdot \vec{G}_\perp$$
(4.178)

In summary, for the flat space sum rules we should set the polarizations e_1 , e_3 to be proportional to \overline{s} , and perform polarization sums for e_2 , e_4 . More precisely,

$$e_{1\mu} = e_{3\mu} = i \frac{m(m - \sqrt{m^2 - v^2})}{v^2} \overline{s}_{\mu},$$

$$e_{2\mu_2} e_{2\rho_2} \to \eta_{\mu_2\rho_2} - \frac{p_{1\mu_2} p_{2\rho_2} + p_{1\rho_2} p_{2\mu_2}}{p_1 \cdot p_2}, \quad e_{4\mu_4} e_{4\rho_4} \to \eta_{\mu_4\rho_4} - \frac{p_{3\mu_4} p_{4\rho_4} + p_{3\rho_4} p_{4\mu_4}}{p_3 \cdot p_4}$$

$$(4.179)$$

To study how the spinning CFT sum rules are mapped to flat space, we first start with the heavy action formula (4.102). Then, we further decompose the conformal block $G^s_{\Lambda,\rho}$ into independent four-point tensor structures. Namely, we want to consider

$$\Psi_{k,\nu,j'}^{+(a)}[G_{\Delta,\rho,(a'b'),I}^{s}Q^{I}], \qquad (4.180)$$

where Q^{I} is a basis of CFT four-point structures. We will choose the CFT four-point structures such that under the polarization maps (4.133),(4.136), they get mapped

¹⁷For a general polarization ϵ_2 , p_1 can be replaced with any null vector q. However, here we have chosen the polarization to satisfy the condition $e_2 \cdot p_1 = e_2 \cdot p_2 = 0$, so the correct choice here is $q = p_1$.

to the generators of the local module of the four-photon amplitude. Namely, the four-point structures satisfy

$$Q^{I}|_{\text{saddle},(4.133),(4.136)} = \frac{16}{\nu^{4}} \left\{ H_{14}H_{23}, H_{13}H_{24}, H_{12}H_{34}, X_{1243}, X_{1234}, X_{1324}, \frac{2}{\nu^{2}}S \right\},$$
(4.181)

where the structures on the right hand side is the basis of polarization structures of the four-photon amplitude introduced in section 4.4.4. We give the explicit expressions of Q^{I} 's in appendix C.5.

We can now apply the polarization map (4.133),(4.136) to the right hand side of the heavy action formula (4.102). From the previous discussions, we know that the kernel part should become $\mathcal{K}_{\nu,j'}^{(a)}$ defined in (4.162) and it acts on the four-point tensor structures Q^{I} 's through (4.179). For the block part, by (4.143), (4.144), it should turn into four-photon partial waves which are further decomposed in four-point structures. In summary, we have

$$\lim_{\substack{\nu,m\gg1\\\nu
(4.182)$$

Our remaining task is to evaluate the above expression explicitly. We perform the polarization sum of $\mathcal{K}_{v,j'}^{(a)}[Q^I]$ using (4.163), (4.179), and also insert the subtraction factors $f_{k;J_e}^{(-1)j'}$ given by (4.106), (4.107) with $J_e = 1$. Finally, by comparing the result with the heavy state contribution of the four-photon flat space sum rules (4.171), we obtain the dictionary between the four-photon CFT and flat space sum rules.

where $A_{k,\nu}^{4\gamma}$ is

$$A_{k,\nu}^{4\gamma} = 2^{5d-3-2j'} e^{\pi\nu} \pi^{1-d} \nu^{6-3d-2k} \Gamma(\frac{d}{2}).$$
(4.184)

The two matrices $M_{4\gamma}^+, M_{4\gamma}^-$ are our main results. They are given by

$$M_{4\gamma}^{+} = \begin{pmatrix} -4(D-2) & -2(D-4) & 0\\ -1 & -\frac{D-4}{2} & (D-3)\\ \frac{2}{(D-1)} & -\frac{1}{(D-1)} & 0 \end{pmatrix},$$

$$M_{4\gamma}^{-} = \begin{pmatrix} 2 \end{pmatrix}.$$
 (4.185)

The left-hand side of the dictionary is written as the heavy action defined by (4.145), and the right hand side is the heavy state contribution given by e.g., (4.174). The dictionary should be true for all tensor structures of the block/partial wave, so we suppress their labels (a'), (b') in (4.183). Moreover, since $(-1)^{j'}$ gives the signature of the integrand under $1 \leftrightarrow 3$, or equivalently $s \leftrightarrow t$, we can separately consider the CFT sum rules with even/odd j' and the flat space sum rules that are symmetric/antisymmetric under $s \leftrightarrow t$. On the CFT side, k is a parameter of the functional and the subtraction factor, which controls the behavior of the integrand in the Regge limit. On the flat space side, the same k gives the Regge spin of the sum rule.

We also notice that the all the CFT sum rules correspond to the "lowest-subtracted" flat space sum rules, meaning that they have the minimal number of subtractions (for a given Regge spin) among all the sum rules in (4.171). (In other words, their corresponding polarization structures grow the fastest in the Regge limit.) Interestingly, upon setting $e_1 = e_3$ to be along the \overline{s}^{μ} direction following (4.179), we find that the only four-photon amplitude structures that are still non-vanishing are the ones that appear in the lowest subtracted sum rules. It would be nice to better understand why this happens.

Transverse spin of sum rules

By construction, our CFT sum rules are labeled by a transverse spin j'. As discussed below the dictionary (4.183), the signature $(-1)^{j'}$ tells us if the flat space sum rule is symmetric or antisymmetric under $s \leftrightarrow t$. However, j' is a representation of SO(d - 2), so it should encode more information than just a signature. On the CFT side, the transverse spin appears in the dual structure

 $(\langle 0|O_4(x_4)\mathbf{L}[O](x_0, z_0)O_2(x_2^+)\rangle^{(a)})^{-1}$, and j' can be realized as a representation of the SO(d-2) group that stabilizes the saddle configuration of x_{2^+}, x_4, x_0, z_0 . On the flat space side, this should correspond to an SO(D-3). What is this SO(D-3) in flat space? Moreover, in the dictionary (4.183), (4.185), each CFT sum rule with a fixed j' gets mapped to a linear combination of the coefficient functions $\mathcal{M}_{4\gamma}^{(i)}$. How can we interpret this linear combination as having a transverse spin j' directly in flat space?

We find that the simplest way to understand transverse spin in flat space is by considering the fixed-*u* Regge limit in the shock frame (4.149). First, note that in the Regge limit, the \mathcal{V}, \mathcal{W} vectors defined in (4.160) become $\mathcal{V}^{\mu} \sim q_1^{\mu} + q_3^{\mu} = vv_{2;\perp}^{\mu}, \mathcal{W}^{\mu} \sim q_1^{\mu} - q_3^{\mu} = -ivv_{1;\perp}^{\mu}$. Therefore, the four kernels in (4.163) become

$$\begin{aligned} \mathcal{K}_{j'=0}^{4\gamma,(1)}\Big|_{\text{Regge}} &\propto e_{2\mu_{2}}e_{4\mu_{4}}\left((D-3)v_{2;\perp}^{\mu_{2}}v_{2;\perp}^{\mu_{4}}+v_{1;\perp}^{\mu_{2}}v_{1;\perp}^{\mu_{4}}+\delta_{\perp;D-4}^{\mu_{2}\mu_{4}}\right),\\ \mathcal{K}_{j'=0}^{4\gamma,(2)}\Big|_{\text{Regge}} &\propto e_{2\mu_{2}}e_{4\mu_{4}}\left(v_{1;\perp}^{\mu_{2}}v_{1;\perp}^{\mu_{4}}+\delta_{\perp;D-4}^{\mu_{2}\mu_{4}}\right),\\ \mathcal{K}_{j'=1}^{4\gamma,(1)}\Big|_{\text{Regge}} &\propto e_{2\mu_{2}}e_{4\mu_{4}}v_{1;\perp}^{\mu_{2}}v_{2;\perp}^{\mu_{4}},\\ \mathcal{K}_{j'=2}^{4\gamma}\Big|_{\text{Regge}} &\propto e_{2\mu_{2}}e_{4\mu_{4}}\left(v_{1;\perp}^{\mu_{2}}v_{1;\perp}^{\mu_{4}}-\frac{v_{1;\perp}^{\mu_{2}}v_{1;\perp}^{\mu_{4}}+\delta_{\perp;D-4}^{\mu_{2}\mu_{4}}}{D-3}\right). \end{aligned}$$
(4.186)

From the expression of the kernels in the Regge limit, we see that the SO(D-3) that realizes the transverse spin in flat space should be the stabilizer group of $s, \bar{s}, v_{2;\perp}$, and $v_{1;\perp}^{\mu_2} v_{1;\perp}^{\mu_4} + \delta_{\perp;D-4}^{\mu_2\mu_4}$ is the metric of the corresponding (D-3)-dimensional space. In fact, we can see this more clearly by studying the four-photon amplitude in the Regge limit. To make the action of the SO(D-3) more manifest, we can parametrize the polarizations as

$$e_{2}^{\mu} = e_{2s}s^{\mu} + e_{2\overline{s}}\overline{s}^{\mu} + e_{2+}v_{2;\perp}^{\mu} + z_{2}^{\mu},$$

$$e_{4}^{\mu} = e_{4s}s^{\mu} + e_{4\overline{s}}\overline{s}^{\mu} + e_{4+}v_{2;\perp}^{\mu} + z_{4}^{\mu},$$
(4.187)

where z_2^{μ} , z_4^{μ} are (D-3)-dimensional vectors that transform under the SO(D-3) group. Using transversality $p_2 \cdot e_2 = 0$ and the gauge redundancy $e_2 \sim e_2 + \#p_2$, we can express e_{2s} , $e_{2\overline{s}}$ in terms of e_{2+} , z_2 and the shock frame momenta (and similarly for e_4). Therefore, we can parametrize the polarizations using e_{2+} , e_{4+} and z_2 , z_4 .

We now take the four-point amplitude and set $e_1 = e_3 = i \frac{m(m - \sqrt{m^2 - v^2})}{v^2} \overline{s}$ as required by our functional. As mentioned above, after this step only the polarization structures that appear in the lowest-subtracted sum rules are non-vanishing. We then take the

Regge limit and consider the leading term. With the parametrization (4.187), we find

$$\mathcal{M}_{4\gamma}^{\text{Regge}} = \frac{m^2 v^2 (D-1)}{8} \left(\frac{2}{D-1} \mathcal{M}^{(1)+}(s,u) - \frac{1}{D-1} \mathcal{M}^{(2)+}(s,u) \right) E_{24,j'=2}^{4\gamma} \\ + \frac{im^2 v^2}{8} \left(2 \mathcal{M}^{(1)-}(s,u) \right) E_{24,j'=1}^{4\gamma} \\ - \frac{m^2 v^2}{4(D-3)} \left(-\mathcal{M}^{(1)+}(s,u) - \frac{D-4}{2} \mathcal{M}^{(2)+}(s,u) + (D-3) \frac{v^2}{2} \mathcal{M}^{(3)} \right) E_{24,j'=0}^{4\gamma,(1)} \\ - \frac{m^2 v^2}{16(D-3)} \left(-4(D-2) \mathcal{M}^{(1)+}(s,u) - 2(D-4) \mathcal{M}^{(2)+}(s,u) \right) E_{24,j'=0}^{4\gamma,(2)},$$
(4.188)

where each $E_{24,j'}^{4\gamma}$ is a tensor built from e_2, e_4 that transforms as a spin-j' representation under the SO(D-3) group. They are defined as

$$E_{24,j'=2}^{4\gamma} = z_2 \cdot v_{1;\perp} z_4 \cdot v_{1;\perp} - \frac{z_2 \cdot z_4}{D-3}, \qquad E_{24,j'=1}^{4\gamma} = e_{4+} z_2 \cdot v_{1;\perp} - e_{2+} z_4 \cdot v_{1;\perp}, \\ E_{24,j'=0}^{4\gamma,(1)} = z_2 \cdot z_4 - e_{2+} e_{4+}, \qquad E_{24,j'=0}^{4\gamma,(2)} = e_{2+} e_{4+}.$$
(4.189)

We see that by decomposing $\mathcal{M}_{4\gamma}^{\text{Regge}}$ into different representations under SO(D-3), the linear combinations of the coefficient functions exactly agree with the ones in the dictionary (4.185). In other words, we can write $\mathcal{M}_{4\gamma}^{\text{Regge}}$ as

$$\mathcal{M}_{4\gamma}^{\text{Regge}} = \left(-\frac{m^2 \nu^2}{4(D-3)} E_{24,j'=0}^{4\gamma,(1)} - \frac{m^2 \nu^2}{16(D-3)} E_{24,j'=0}^{4\gamma,(2)} - \frac{m^2 \nu^2 (D-1)}{8} E_{24,j'=2}^{4\gamma}\right) M_{4\gamma}^+ \begin{pmatrix} \mathcal{M}^{(1)+}(s,u) \\ \mathcal{M}^{(2)+}(s,u) \\ \frac{\nu^2}{2} \mathcal{M}^{(3)}(s,t) \end{pmatrix} + \left(\frac{im^2 \nu^2}{8} E_{24,j'=1}^{4\gamma}\right) M_{4\gamma}^- \left(\mathcal{M}^{(1)-}(s,u)\right),$$

$$(4.190)$$

where $M_{4\gamma}^+$, $M_{4\gamma}^-$ are the matrices in the dictionary (4.185).

Another way to see why we get the same matrices that appear in the dictionary is that under the polarization sum, the pairing between the Regge limit kernels (4.186) and the spin-j' tensors (4.189) form a diagonal matrix.

For the j' = 0 case, there are two different tensors structures. By comparing $E_{j'=0}^{(1)}, E_{j'=0}^{(2)}$ and the CFT structures that give the two kernels for j' = 0, we can even write down a dictionary between the embedding space CFT structures and spin-j' tensors:

$$H_{20}H_{40} \leftrightarrow e_{2+}e_{4+}, \quad H_{24} \leftrightarrow z_2 \cdot z_4 - e_{2+}e_{4+}.$$
 (4.191)

A similar dictionary between CFT structures and polarization structures in a conformal frame is also given in [48].

4.4.6 Sum rules for gravitons

In this section we give the dictionary between the CFT sum rules and flat space sum rules for gravitons. We first review the flat space graviton sum rules.

The local module of the graviton amplitude has 29 generators, and 28 of them can be constructed from taking products of the photon generators defined in section 4.4.4. The remaining generator G can be written as the Gram determinant of all dot products between $(p_1, p_2, p_3, e_1, e_2, e_3, e_4)$. The general four-graviton amplitude in generic spacetime dimension $(D \ge 8)$ takes the form [138]

$$\mathcal{M}_{4g}(s,u) = \mathcal{G}\mathcal{M}_{4g}^{(1)}(s,u) + S^2 \mathcal{M}_{4g}^{(10)}(s,u) + \left(H_{14}^2 H_{23}^2 \mathcal{M}_{4g}^{(2)}(s,u) + H_{12} H_{13} H_{24} H_{34} \mathcal{M}_{4g}^{(3)}(s,u) + H_{14} H_{23} (X_{1243} - X_{1234} - X_{1324}) \mathcal{M}_{4g}^{(4)}(s,u) + X_{1243}^2 \mathcal{M}_{4g}^{(6)}(s,u) + X_{1234} X_{1324} \mathcal{M}_{4g}^{(7)}(s,u) + H_{14} H_{23} S \mathcal{M}_{4g}^{(8)}(s,u) + X_{1243} S \mathcal{M}_{4g}^{(9)}(s,u) + \text{triplet permutations} \right) + H_{12} H_{34} X_{1243} \mathcal{M}_{4g}^{(5)}(s,u) + \text{sextuplet permutations.}$$
(4.192)

The functions that multiply the polarization structures \mathcal{G} and S^2 are symmetric under all permutations of *s*, *t*, *u*, and the ones with "triplet permutations" are symmetric in their two arguments. For the $\mathcal{M}^{(5)}$ function one should include all six permutations.

By repeating the analysis of the Regge limit behavior reviewed in section 4.4.4, one can obtain the dispersion relations for the graviton amplitude. There are 19 independent fixed-*u* sum rules with even Regge spin *k* and are symmetric under $s \leftrightarrow t$ [110]. Similar to the photon case, the number of subtractions for a given Regge spin of these sum rules can be different. In the dictionary for photons, we see that our CFT sum rules all correspond to the lowest-subtracted flat space sum rules. This turns out to be true for gravitons as well. For graviton sum rules with even Regge spin, 7 of them are the lowest-subtracted. They are given by

$$\left\{ C_{4g;k,u}^{+(1-7)} \right\} = -\oint_{\infty} \frac{ds}{4\pi i} \frac{(s-t)}{(-st)^{\frac{k-2}{2}}} \left\{ \mathcal{M}_{4g}^{(3)}(s,t), \left(\frac{-u}{2}\right)^2 \mathcal{M}_{4g}^{(10)}(s,t), \mathcal{M}_{4g}^{(2,5)+}(s,u), \left(\frac{-u}{2}\right) \mathcal{M}_{4g}^{(8,9)+}(s,u), \mathcal{M}_{4g}^{(6)+}(s,u) + \mathcal{M}_{4g}^{(7)}(s,t) \right\}, \quad k = 2, 4, \dots$$

$$(4.193)$$

For the remaining 12 sum rules that have higher number of subtractions, see appendix A of [110]. We again use the notation $\mathcal{M}^{(i)\pm}(s, u) = \mathcal{M}^{(i)}(s, u) \pm \mathcal{M}^{(i)}(t, u)$, and

rescale some of the sum rules by $\left(\frac{-u}{2}\right)$ to make all of them have the same power counting.

For graviton sum rules with odd Regge spin, there are 10 independent sum rules, and 3 of them are the lowest-subtracted,

$$\left\{ C_{4g;k,u}^{-(1-3)} \right\} = -\oint_{\infty} \frac{ds}{4\pi i} \frac{1}{(-st)^{\frac{k-3}{2}}} \left\{ \mathcal{M}_{4g}^{(2,5)-}(s,u), \left(\frac{-u}{2}\right) \mathcal{M}_{4g}^{(8)-}(s,u) \right\}, \quad k = 3, 5, \dots$$

$$(4.194)$$

When performing the CFT analysis for gravitons, all the discussions in section 4.4.2 and 4.4.3 can be generalized straightforwardly. In particular, we can also obtain a simple relation similar to (4.144) that relates the block part of the four-stress-tensor functional to the four-graviton partial waves, which include 20 different cases in total [135, 110]. Moreover, we can obtain the graviton kernels that are similar to (4.163). The number of independent kernels are 3 for j' = 0, 2 for j' = 1, 3 for j' = 2, 1 for j' = 3, 1 for j' = 4. Therefore, we have 7 even j' and 3 odd j' sum rules in total, and they indeed agree with the number of "lowest-subtracted" flat space graviton sum rules that are symmetric or antisymmetric under $s \leftrightarrow t$ given in (4.193) and (4.194).

Finally, the only remaining difference in the graviton case is that instead of (4.179), the rules for polarization sum of e_2 , e_4 become

$$e_{i}^{\mu}e_{i}^{\nu}e_{i}^{\alpha}e_{i}^{\beta} \to \frac{1}{2}\left(P_{e_{i}}^{\mu\alpha}P_{e_{i}}^{\nu\beta} + P_{e_{i}}^{\nu\alpha}P_{e_{i}}^{\mu\beta}\right) - \frac{1}{D-2}P_{e_{i}}^{\mu\nu}P_{e_{i}}^{\alpha\beta}, \quad i = 2, 4,$$
(4.195)

where

$$P_{e_{2}}^{\mu\nu} = \eta^{\mu\nu} - \frac{p_{1}^{\mu}p_{2}^{\nu} + p_{1}^{\nu}p_{2}^{\mu}}{p_{1} \cdot p_{2}},$$

$$P_{e_{4}}^{\mu\nu} = \eta^{\mu\nu} - \frac{p_{3}^{\mu}p_{4}^{\nu} + p_{3}^{\nu}p_{4}^{\mu}}{p_{3} \cdot p_{4}}.$$
(4.196)

Our final dictionary for gravitons is given by

$$\lim_{\substack{\Delta,\nu\gg1\\\nu$$

where the coefficient $A_{k,v}^{4g}$ is

$$A_{k,\nu}^{4g} = 2^{5d+3-2j'} e^{\pi\nu} \pi^{1-d} \nu^{2-3d-2k} \Gamma(\frac{d+2}{2}), \qquad (4.198)$$

and the two matrices are given by

$$M_{4g}^{+} = \begin{pmatrix} 16(D-4)(D-2) & 0 & 16D(D+2) & 8(D-4)D & 0 & 0 & 4(D-4)(D-2) \\ 16 & 0 & 16D & 2(D^{2}-3D-2) & -4(D-2)(D-1) & -2(D-4)(D-1) & 2(D^{2}-6D+7) \\ 2 & (D-3)(D-1) & 2 & \frac{D-3}{2} & 1-D & -\frac{1}{2}(D-4)(D-1) & \frac{1}{4}(D^{2}-6D+7) \\ -\frac{8D}{D+1} & 0 & \frac{8(D+2)}{D+1} & -\frac{4}{D+1} & 0 & 0 & \frac{2}{D+1} \\ -\frac{32}{D+1} & 0 & -\frac{32(D+2)}{D+1} & \frac{16}{D+1} & 0 & 0 & \frac{4(D-1)}{D+1} \\ -\frac{8}{D+1} & 0 & -\frac{8}{D+1} & -\frac{D-3}{D+1} & 2 & -1 & \frac{D-1}{D+1} \\ \frac{24}{(D+1)(D+3)} & 0 & \frac{24}{(D+1)(D+3)} & -\frac{12}{(D+1)(D+3)} & 0 & 0 & \frac{6}{(D+1)(D+3)} \end{pmatrix},$$

$$M_{4g}^{-} = \begin{pmatrix} -16(D+2) & 4(D-4) & 0 \\ -8 & (D-3) & 2(D-1) \\ \frac{24}{D+1} & \frac{6}{D+1} & 0 \end{pmatrix}.$$
(4.199)

Finally, the transverse spin analysis done in section 4.4.5 can also be performed in the graviton case. In this case we have transverse spin up to j' = 4. As an example, let us consider the spin-4 tensor which is defined as

$$E_{24,j'=4}^{4g} = (z_2 \cdot v_{1;\perp})^2 (z_4 \cdot v_{1;\perp})^2 - \text{traces.}$$
(4.200)

`

When we isolate the spin-4 part of the four-graviton amplitude in the Regge limit (and after setting e_1 , e_3 to be in the \overline{s} direction), we obtain

$$\mathcal{M}_{4g}^{\text{Regge}} = \frac{m^4 \nu^4}{64} \left(4\mathcal{M}^{(3)}(s,t) + 4\mathcal{M}^{(2)+}(s,u) - 2\mathcal{M}^{(5)+}(s,u) + \mathcal{M}^{(6)+}(s,u) + \mathcal{M}^{(7)}(s,t) \right) E_{24,j'=4}^{4g} + \dots,$$
(4.201)

where ... are other terms with j' < 4. Indeed, we see that the linear combination of the coefficient functions agrees with the dictionary (4.199) (the last row of M_{4g}^+).

4.5 Discussion

In this work, we studied a basis of CFT dispersive sum rules for spinning operators. The basis was constructed using the fact that the commutator of two null-integrated operators on the same null plane vanishes, also known as superconvergence. Using the Lorentzian shadow representation of conformal blocks, we expressed the action of our sum rule on a conformal block as an integral over spacetime. We showed that in the bulk point limit, where ν (parameter of the sum rule) and Δ (dimension of the block) both become large, the spacetime integral gets completely localized to a "scattering-crystal." This enabled us to derive a simple formula for the block action of our sum rule in the bulk point limit, generalizing the results in [13] to the spinning case.

The main result of this paper is a dictionary between the block action of CFT sum rules in the bulk point limit and the heavy state contribution of flat space sum rules. We fixed the dictionary by exploring flat space interpretations for the inserted block and the kernel in the simple formula of the CFT sum rule. We showed that CFT three-point structures at the saddle form a natural basis for the flat space three-point amplitudes, allowing the inserted block to be directly related to flat space partial waves in this basis. On the other hand, the kernels of the sum rule can be interpreted as actions on shockwave amplitudes in flat space. Combining the two results, we derived the dictionary between our CFT sum rules and the flat space sum rules for photons and gravitons.

By construction, the CFT sum rules are labeled by a transverse spin j'. Through the dictionary, this gives a basis of flat space sum rules each labeled by a transverse spin (an SO(D-3) representation). This SO(D-3) stabilizes the shock directions and the momentum transfer $p_1 + p_3$ of the shockwave amplitude, and the transverse spin of each sum rule can be realized by studying the amplitude in the Regge limit. However, from the flat space perspective, it remains unclear how this "transverse spin basis" is useful in practice. It would be interesting to explore this further.

Our construction of spinning CFT sum rules is an important step toward lifting the flat space gravitational bounds in [110] to AdS/CFT. It provides functionals with positive action on conformal blocks with large Δ and fixed *J*. To complete the bootstrap argument, one also needs to check positivity in the Regge limit, where Δ and *J* are both large. Moreover, one has to derive a similar dictionary that relates the low energy EFT contribution of the CFT and flat space sum rules. In the scalar case [13], the light contribution analysis also involves a saddle point. Therefore we expect we should be able to generalize it to the spinning case by arguments similar to section 4.3. Once these calculations are done, one can then derive bounds on the higher derivative corrections of gravitational AdS EFTs in terms of the large CFT gap. These bounds would give a sharp form of the HPPS conjecture [114] at the level of four-point functions of stress tensors.

It would be interesting to also test the HPPS conjecture beyond four-point functions using superconvergence. For instance, one could consider commutators of light-transformed operators in a multi-point correlation function. In flat space, we expect this should correspond to commutators of multiple shockwaves. Another interesting direction is studying superconvergence in a thermal state by using a folded light transform contour proposed in [139].

So far, we have only found a dictionary that relates CFT sum rules to some of the known flat space sum rules. In particular, we only get sum rules that have the fewest subtractions. As discussed in 4.4.5, to get other sum rules we have to leave the configuration where the polarizations of the shock gravitons are along the shock direction. In CFT this condition comes from setting the polarizations of the light-transformed operators O_1, O_3 to be the null vectors x_{10}, x_{30} . Therefore, it is conceivable that new sum rules can be obtained by studying null-integrated operators that are not integrated along their polarizations. Equivalently, we can consider taking derivatives of the light transformed operators. Schematically, we would like to find a differential operator \mathcal{D} such that

$$\mathcal{D}_{x,z_1,z_3} \mathbf{L}[O_1](x,z_1) \mathbf{L}[O_3](x,z_3)$$
(4.202)

is a conformal primary at x. These conformally-invariant differential operators have interesting connections to reducible representations of the conformal group [35]. We hope to address this problem in future.

In [13], it was shown that superconvergence sum rules also exist at J = -2, -4, ... In these sum rules, one integrates a commutator against derivatives of delta functions, which again constrains the two operators in the commutator to be spacelike separated. It would be interesting to find spinning versions of these sum rules.

In flat space, unitarity additionally imposes an upper bound on the spectral density. This condition has been used in the context of EFT amplitudes through primal bootstrap methods [140, 141]. However, it is not clear what the corresponding CFT statement is. To understand how this upper bound is realized in CFT, our dictionary will be an important ingredient.

Crossing-symmetric dispersion relations in flat space [107, 142] have also proven valuable in studying dispersive bounds on EFTs. They have the advantage of making low-energy crossing almost trivial to impose, and are more well-behaved when including loop EFT corrections [143]. Analogous CFT dispersion relations have been established in Mellin space [144], and crossing-symmetric dispersion relations in position space were recently studied in [145]. We expect that position space methods will be more straightforward to generalize to spinning case.

Including loop EFT corrections often makes the forward limit in flat space divergent [103, 146]. Consequently, one has to consider a limited set of sum rules that give weaker positivity bounds. On the other hand, loop corrections in AdS are automatically regularized by R_{AdS} . By computing the loop contributions from summing over double-trace exchanged operators [147, 148, 149], one should be able to obtain finite loop corrections to the tree-level AdS bounds.

Acknowledgements

We thank Simon Caron-Huot, Yue-Zhou Li, Dalimil Mazáč, Julio Parra-Martinez, and Leonardo Rastelli for helpful discussions. We are supported by Simons Foundation grant 488657 (Simons Collaboration on the Nonperturbative Bootstrap) and a DOE Early Career Award under grant no. DE-SC0019085. YL is additionally supported by the National Science Foundation Graduate Research Fellowship under grant no. DGE-1745301

Chapter 5

SCALAR MODULAR BOOTSTRAP AND ZEROS OF THE RIEMANN ZETA FUNCTION

This chapter is based on

 [1] Nathan Benjamin and Cyuan-Han Chang. "Scalar modular bootstrap and zeros of the Riemann zeta function". In: *JHEP* 11 (2022), p. 143.
 DOI: 10.1007/JHEP11(2022)143. arXiv: 2208.02259 [hep-th].

5.1 Introduction

The conformal bootstrap is a powerful program used to highly constrain quantum field theories starting from basic consistency conditions. In two dimensional conformal field theory (CFT), one avatar of this program is the so-called modular bootstrap which uses modular invariance of the genus one partition function to constrain possible allowed spectra of 2d CFTs. This program started with the work of [150] and has led to many interesting results (see e.g. [151, 152, 153, 154, 155, 156, 157, 158, 159, 160, 161, 162] for a non-exhaustive list). This has several applications, including constraining theories of quantum gravity in AdS₃.

In many (but not all) cases, the *spinless* bootstrap equations are studied, in which one throws away information about the spin of the original operators and only looks at their energies. This is done by grading the partition function only by the energies of the operators, and using *S*-invariance, rather than the full $SL(2, \mathbb{Z})$ -invariance of the partition function. In particular, we have

$$Z(y) := \sum_{O} e^{-2\pi y (\Delta_O - \frac{c}{12})} = Z(y^{-1}),$$
(5.1)

where the sum over O is a sum over all local operators in the theory, and Δ_O is the scaling dimension of operator O. Any bound derived from (5.1) will by definition be insensitive to the spins of the operators O. For example, the current strongest bound on the lightest nontrivial Virasoro primary operator at large central charge c is in [156], which showed at large c,

$$\Delta_{\rm gap}^{\rm Virasoro} \lesssim \frac{c}{9.1}.$$
(5.2)

However it makes no claim on what the spin of that operator is, or what the lightest spin j operator is. A similar result using the spinless bootstrap was found for a simpler class of theories, those with a $U(1)^c$ chiral algebra, in [162]

$$\Delta_{\rm gap}^{U(1)^c} \lesssim \frac{c}{9.869}.\tag{5.3}$$

In this paper we derive a novel one-dimensional crossing equation using the technology of harmonic analysis. In the case of CFTs with $U(1)^c$ symmetry, this crossing equation acts *only* on the scalar primary operators of the theory (with respect to the $U(1)^c$ chiral algebra). This allows us to place new bounds on the scalar gap of all $U(1)^c$ conformal field theories for any integer c. This is more refined information than the bound in e.g. (5.3) since it provides explicit information about the spin of the operator. Indeed the scalar gap is a natural object to consider. Scalar operators can be added to the Lagrangian while still preserving Lorentz invariance. The scalar gap is then related to questions about, for instance, if the CFT has a relevant operator or not. Another application is in the study of boundary conformal field theory. There, the bulk scalars show up in some crossing equations rather than all bulk operators, which can lead to interesting bounds that are conditional on the scalar gap [163].

Remarkably, our crossing equation has an intimate relation with the nontrivial zeros of the Riemann zeta function. In a sense which we will explain, hidden inside the scalar operators of any 2d CFT with $U(1)^c$ symmetry are the nontrivial zeros of the zeta function. As a result, we can rephrase the Riemann hypothesis as a statement about the behavior of scalar operators of any $U(1)^c$ CFT.

We also discuss a generalization to Virasoro CFTs. We derive a more complicated one-dimensional crossing equation that involves operators of all spins. The nontrivial zeros of the zeta function again play an important role. This leads to the Riemann hypothesis being equivalent to a more complicated statement about the asymptotic density of a signed count of all operators (of any spin) in any CFT. Unfortunately we run into some technical obstacles in bounding physical quantities such as the scalar gap for Virasoro CFTs.

This paper is organized as follows. In Section 5.2 we review harmonic analysis on the fundamental domain of $SL(2, \mathbb{Z})$, which will play an important role in deriving our scalar crossing equation. In Section 5.3 we apply this to the study of $U(1)^c$ CFTs and derive the scalar crossing equation. We present the numerical results for the scalar gap of $U(1)^c$ theories for various values of *c*. In Section 5.4 we discuss generalizations to theories with only Virasoro symmetry. In Section 5.5 we study more explicitly the connections between 2d CFTs and the Riemann hypothesis. We discuss various potentially interesting future directions in Section 5.6. Some detailed calculations and derivations are banished to the appendices.

5.2 **Review of Harmonic Analysis**

In this section we will review harmonic analysis on the space $\mathbb{H}/SL(2,\mathbb{Z})$, where \mathbb{H} is the upper half plane. For much of this discussion, we refer to [164]. We will use the notation of [165] in this section.

The main idea is to decompose square-integrable modular invariant functions into eigenfunctions of the Laplacian on the space $\mathbb{H}/SL(2,\mathbb{Z})$. If $\tau \in \mathbb{H}$, with real and imaginary parts *x*, *y* respectively, then there is a natural metric on \mathbb{H} given by

$$ds^2 = \frac{dx^2 + dy^2}{y^2}.$$
 (5.4)

The Laplacian on this space is given by

$$\Delta = -y^2 (\partial_x^2 + \partial_y^2). \tag{5.5}$$

Square-integrable modular-invariant functions $f(\tau)$ are those with finite L^2 norm under the measure (5.4), meaning

$$\int_{-1/2}^{1/2} dx \int_{\sqrt{1-x^2}}^{\infty} \frac{dy}{y^2} |f(\tau)|^2 < \infty.$$
(5.6)

If $f(\tau)$ is a square-integrable, modular-invariant function, it has a unique decomposition into eigenfunctions of the Laplacian (5.5). These eigenfunctions have been classified and they come in three types:

- The constant function 1, with eigenvalue 0.
- An infinite, continuous family of eigenfunctions known as real analytic Eisenstein series, $E_s(\tau)$, with $s = \frac{1}{2} + it$, t real, with eigenvalue $\frac{1}{4} + t^2$. Any real t is permissible.
- An infinite, discrete family of eigenfunctions known as Maass cusp forms, denoted v[±]_n(τ), n = 1, 2, ···. These have sporadic eigenvalues, which we denote ¹/₄ + (R[±]_n)², for R[±]_n a positive real number. Both v⁺_n and v⁻_n are ordered in increasing eigenvalue, i.e. R⁺₁ < R[±]₂ < ···, and likewise for R⁻_n. The superscript ± refers to whether the cusp form is even or odd under parity.

The decomposition of $f(\tau)$ is then given by:

$$f(\tau) = \frac{(f,1)}{(1,1)} + \frac{1}{4\pi i} \int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} ds E_s(\tau)(f,E_s) + \sum_{n=1}^{\infty} \sum_{\epsilon=\pm} v_n^{\epsilon}(\tau) \frac{(f,v_n^{\epsilon})}{(v_n^{\epsilon},v_n^{\epsilon})}, \quad (5.7)$$

where the overlap function is given by the Petersson inner product:

$$(f,g) \coloneqq \int_{-1/2}^{1/2} dx \int_{\sqrt{1-x^2}}^{\infty} \frac{dy}{y^2} f(\tau) \overline{g(\tau)}.$$
 (5.8)

The decomposition (5.7) is known as the Roelcke-Selberg decomposition.

Let us be more explicit about the eigenfunctions of the Laplacian. The real analytic Eisenstein series $E_s(\tau)$, $s \in \mathbb{C}$ are defined as a modular sum of y^s :

$$E_{s}(\tau) = \sum_{\gamma \in \Gamma_{\infty} \setminus SL(2,\mathbb{Z})} y^{s}|_{\gamma}, \qquad (5.9)$$

where Γ_{∞} is the subgroup of $SL(2,\mathbb{Z})$ generated by $\tau \rightarrow \tau + 1$. The sum (5.9) converges if $\operatorname{Re}(s) > 1$. However, it admits an analytic continuation everywhere in the *s* plane:

$$E_{s}(\tau) = y^{s} + \frac{\Lambda(1-s)}{\Lambda(s)} y^{1-s} + \sum_{j=1}^{\infty} \frac{4\sigma_{2s-1}(j)\sqrt{y}K_{s-\frac{1}{2}}(2\pi jy)}{\Lambda(s)j^{s-\frac{1}{2}}}\cos(2\pi jx), \quad (5.10)$$

where $\sigma_{2s-1}(j)$ is the divisor sigma function, *K* is the modified Bessel function of second kind, and Λ is defined as

$$\Lambda(s) \coloneqq \pi^{-s} \zeta(2s) \Gamma(s). \tag{5.11}$$

The function $\Lambda(s)$ obeys a useful identity:

$$\Lambda(s) = \Lambda(\frac{1}{2} - s). \tag{5.12}$$

From (5.10) we also see that the real analytic Eisenstein series obey a useful identity:

$$\Lambda(s)E_s(\tau) = \Lambda(1-s)E_{1-s}(\tau).$$
(5.13)

The remaining eigenfunctions, the Maass cusp forms, are more mysterious. They take the following functional form:

$$v_n^+(\tau) = \sum_{j=1}^{\infty} a_j^{(n,+)} \sqrt{y} K_{iR_n^+}(2\pi j y) \cos(2\pi j x)$$
$$v_n^-(\tau) = \sum_{j=1}^{\infty} a_j^{(n,-)} \sqrt{y} K_{iR_n^-}(2\pi j y) \sin(2\pi j x),$$
(5.14)

where R_n^{\pm} and $a_j^{(n,\pm)}$ are a set of sporadic real numbers. For example, we have the following first few values of R_n^{\pm} :

$$R_1^+ \approx 13.77975, \quad R_1^- \approx 9.53370$$

$$R_2^+ \approx 17.73856, \quad R_2^- \approx 12.17301$$

$$R_3^+ \approx 19.42348, \quad R_3^- \approx 14.35851.$$
 (5.15)

For more numerical data on the Maass cusp forms, see the online database [LMFDB]. One key feature the Maass cusp forms have is, unlike the real analytic Eisenstein series, they all lack a scalar piece:

$$\int_{-1/2}^{1/2} dx v_n^{\pm}(\tau) = 0.$$
 (5.16)

5.3 $U(1)^c$ CFTs

We begin with studying a family of particularly simple conformal field theories, with an extended current algebra of $U(1)^c$. Examples of such CFTs include Narain's family of c free bosons compactified on a c-dimensional lattice, parameterized by the moduli space $O(c, c, \mathbb{Z}) \setminus O(c, c) / O(c) \times O(c)$. It is believed that this family of CFTs fully classifies all theories with $U(1)^c$ current algebra. However, this has not been proven. Our results in this section will apply to all theories with $U(1)^c$ symmetry; we do not need to assume the theory is a Narain CFT.

5.3.1 Harmonic decomposition

In [165], the harmonic decomposition of $U(1)^c$ CFT partition functions were calculated, which we review here. The characters of the $U(1)^c$ chiral algebra are given by

$$\chi^{h}(\tau) = \frac{q^{h}}{\eta(\tau)^{c}},\tag{5.17}$$

where $\eta(\tau)$ is the Dedekind eta function. Instead of decomposing the full partition function $Z(\tau)$, we instead consider the primary-counting partition function

$$\widehat{Z}^{c}(\tau,\mu) \coloneqq y^{c/2} |\eta(\tau)|^{2c} Z(\tau)$$
$$= y^{c/2} \sum_{h,\overline{h}} q^{h} \overline{q}^{h}, \qquad (5.18)$$

where in (5.18) the sum over h, \overline{h} goes over the $U(1)^c$ primary operators. In (5.18), we write $\widehat{Z}^c(\tau, \mu)$ to emphasize that the (reduced) partition function depends not

only on the worldsheet modulus τ , but also on an abstract target space coordinate μ .¹

The function (5.18) is not yet square-integrable, but once we subtract out the Eisenstein series $E_{c/2}(\tau)$ (defined in (5.10)), this yields a square-integrable function that admits a unique spectral decomposition². In [168, 169, 165] the spectral decomposition was given as follows³:

$$\begin{split} \widehat{Z}^{c}(\tau,\mu) &= E_{c/2}(\tau) + 3\pi^{-\frac{c}{2}}\Gamma\left(\frac{c}{2} - 1\right)\mathcal{E}^{c}_{\frac{c}{2} - 1}(\mu) + \frac{1}{4\pi i}\int_{\frac{1}{2} - i\infty}^{\frac{1}{2} + i\infty} ds\pi^{s-\frac{c}{2}}\Gamma\left(\frac{c}{2} - s\right)\mathcal{E}^{c}_{\frac{c}{2} - s}(\mu)E_{s}(\tau) \\ &+ \sum_{n=1}^{\infty}\sum_{\epsilon=\pm}\frac{(\widehat{Z}^{c},\nu_{n}^{\epsilon})(\mu)}{(\nu_{n}^{\epsilon},\nu_{n}^{\epsilon})}\nu_{n}^{\epsilon}(\tau). \end{split}$$
(5.19)

The coefficients $\mathcal{E}_s^c(\mu)$ were called constrained Epstein zeta series in [169], and are defined as:

$$\mathcal{E}_{s}^{c}(\mu) \coloneqq \sum_{\Delta \in \mathcal{S}} (2\Delta)^{-s}, \qquad (5.20)$$

1

where we define the set S to be the dimensions of all non-vacuum scalar primary operators under the $U(1)^c$ chiral algebra (with multiplicity). This sum converges for $\operatorname{Re}(s) > c - 1$, but like for the $SL(2,\mathbb{Z})$ Eisenstein series (5.9), they admit an analytic continuation everywhere in the complex *s* plane. They also obey a functional equation:

$$\mathcal{E}_{\frac{c}{2}-s}^{c}(\mu) = \frac{\Gamma(s)\Gamma(s+\frac{c}{2}-1)\zeta(2s)}{\pi^{2s-\frac{1}{2}}\Gamma(\frac{c}{2}-s)\Gamma(s-\frac{1}{2})\zeta(2s-1)}\mathcal{E}_{\frac{c}{2}+s-1}^{c}(\mu).$$
 (5.21)

This equation is inherited from the functional equation that the Eisenstein series obey (5.13), combined with the definition of $\mathcal{E}_{s}^{c}(\mu)$ as an overlap of $\widehat{Z}^{c}(\tau,\mu)$ with the Eisenstein series:

$$(\widehat{Z}^{c} - E_{\frac{c}{2}}, E_{s}) = \pi^{s - \frac{c}{2}} \Gamma\left(\frac{c}{2} - s\right) \mathcal{E}_{\frac{c}{2} - s}^{c}(\mu).$$
(5.22)

¹For Narain theories, we can view μ as a parameter $\mu \in O(c, c; \mathbb{Z}) \setminus O(c, c) / O(c) \times O(c)$. The target space of Narain theories is parametrized by a symmetric metric G_{ab} and an antisymmetric *B*-field B_{ab} , where a, b indices run from $1, 2, \dots, c$. Here, however, we can just view μ as some abstract coordinate.

²For Narain CFTs, $E_{c/2}(\tau)$ has the interpretation of the averaged partition function [166, 167].

³Note that due to the pole structure of $\Lambda(s)$ and the real analytic Eisenstein series $E_s(\tau)$, the decompositions of c = 1 and c = 2 are slightly different than other c, so we will assume $c \neq 1, 2$ for the rest of this section. We revisit c = 1 and c = 2 in Appendix D.3.

For Narain CFTs, (5.20) can be rewritten as

$$\mathcal{E}_{s}^{c}(\mu) = \sum_{\vec{n}, \vec{m} \in \mathbb{Z}^{c}}^{\prime} \frac{\delta_{\vec{n} \cdot \vec{w}, 0}}{M_{\vec{n}, \vec{w}}(\mu)^{2s}},$$
(5.23)

with

$$M_{\vec{n},\vec{w}}(\mu)^2 \coloneqq G^{ab}(n_a + B_{ac}w^c)(n_b + B_{bd}w^d) + G_{cd}w^cw^d, \qquad (5.24)$$

and the prime over the summation indicating we should not sum over the vacuum state (with $\vec{n} = \vec{w} = \vec{0}$).

5.3.2 Crossing equation

Since the Maass cusp forms have no scalar piece (i.e. (5.16)), the scalar part of (5.19) is particularly simple:

$$\int_{-1/2}^{1/2} dx \widehat{Z}^{c}(\tau,\mu) = y^{\frac{c}{2}} + \frac{\Lambda\left(\frac{c-1}{2}\right)}{\Lambda\left(\frac{c}{2}\right)} y^{1-\frac{c}{2}} + 3\pi^{-\frac{c}{2}} \Gamma\left(\frac{c}{2} - 1\right) \mathcal{E}^{c}_{\frac{c}{2}-1}(\mu) + \frac{1}{4\pi i} \int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} ds \pi^{s-\frac{c}{2}} \Gamma\left(\frac{c}{2} - s\right) \mathcal{E}^{c}_{\frac{c}{2}-s}(\mu) \left(y^{s} + \frac{\Lambda(1-s)}{\Lambda(s)} y^{1-s}\right),$$
(5.25)

where as usual $\tau = x + iy$, and $\Lambda(s)$ is defined as in (5.11).

As a reminder, the set S is the set of conformal weights of all non-vacuum scalar primaries under the $U(1)^c$ chiral algebra (with multiplicity). We can rewrite the LHS of (5.25) as

$$\int_{-1/2}^{1/2} dx \widehat{Z}^c(\tau,\mu) = y^{\frac{c}{2}} \left(1 + \sum_{\Delta \in \mathcal{S}} e^{-2\pi\Delta y} \right).$$
(5.26)

This gives

$$\sum_{\Delta \in \mathcal{S}} e^{-2\pi\Delta y} = \frac{\Lambda\left(\frac{c-1}{2}\right)}{\Lambda\left(\frac{c}{2}\right)} y^{1-c} + \varepsilon_c(\mu) y^{-\frac{c}{2}} + \frac{1}{2\pi i} \int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} ds \pi^{s-\frac{c}{2}} \Gamma\left(\frac{c}{2}-s\right) \mathcal{E}_{\frac{c}{2}-s}^c(\mu) y^{s-\frac{c}{2}}$$
(5.27)

where we have defined $\varepsilon_c(\mu) \coloneqq 3\pi^{-\frac{c}{2}}\Gamma\left(\frac{c}{2}-1\right)\mathcal{E}_{\frac{c}{2}-1}^c(\mu)$, and used the symmetry between $s \leftrightarrow 1-s$ in the integral over *s*.

The remaining task is to do the integral in (5.27). We will do the integral over *s* by moving the contour to the right of $s = \frac{c}{2}$. It turns out the only poles we enclose after

moving the contour are at $s = \frac{c}{2}, \frac{1+z_n}{2}, \frac{1+z_n}{2}$, where z_n are the nontrivial zeros of the Riemann zeta function with positive imaginary part (i.e. $z_1 \approx \frac{1}{2} + 14.135i$, $z_2 \approx \frac{1}{2} + 21.022i$, etc.). See Fig. 5.1 for a picture of the pole structure (shown for c = 3). We derive the pole structure in Appendix D.1. After moving the contour, (5.27) becomes

$$1 + \sum_{\Delta \in \mathcal{S}} e^{-2\pi\Delta y} = \frac{\Lambda\left(\frac{c-1}{2}\right)}{\Lambda\left(\frac{c}{2}\right)} y^{1-c} + \varepsilon_c(\mu) y^{-\frac{c}{2}} + \sum_{k=1}^{\infty} \operatorname{Re}\left(\delta_{k,c}(\mu) y^{-\frac{c}{2}+1-\frac{z_k}{2}}\right) + \frac{1}{2\pi i} \int_{\gamma-i\infty}^{\gamma+i\infty} ds \pi^{s-\frac{c}{2}} \Gamma\left(\frac{c}{2}-s\right) \mathcal{E}_{\frac{c}{2}-s}^c(\mu) y^{s-\frac{c}{2}},$$
(5.28)

where $\gamma > \frac{c}{2}$. The terms $\varepsilon_c(\mu)$ and $\delta_{k,c}(\mu)$ are moduli-dependent constants, which have an explicit formula as

$$\varepsilon_{c}(\mu) = \frac{3}{\pi} \int_{\mathcal{F}} \frac{dxdy}{y^{2}} (\widehat{Z}^{c}(\tau,\mu) - E_{c/2}(\tau))$$

$$\delta_{k,c}(\mu) = \int_{\mathcal{F}} \frac{dxdy}{y^{2}} (\widehat{Z}^{c}(\tau,\mu) - E_{c/2}(\tau)) \operatorname{Res}_{s=z_{k}/2} E_{s}(\tau), \qquad (5.29)$$

where

$$\operatorname{Res}_{s=z_{k}/2}E_{s}(\tau) = \frac{\sqrt{\pi}\zeta(z_{k}-1)\Gamma(\frac{z_{k}-1}{2})}{2\zeta'(z_{k})\Gamma(\frac{z_{k}}{2})}y^{1-\frac{z_{k}}{2}} + \sum_{j=1}^{\infty}\frac{2\pi^{\frac{z_{k}}{2}}\cos(2\pi jx)\sigma_{z_{k}-1}(j)\sqrt{y}K_{\frac{z_{k}-1}{2}}(2\pi jy)}{j^{\frac{z_{k}-1}{2}}\zeta'(z_{k})\Gamma(\frac{z_{k}}{2})}$$
(5.30)

Now let us consider the integral in (5.28). We first rewrite the integral using the functional identity (5.21):

$$\int_{\gamma-i\infty}^{\gamma+i\infty} ds \pi^{s-\frac{c}{2}} \Gamma\left(\frac{c}{2}-s\right) \mathcal{E}_{\frac{c}{2}-s}^{c}(\mu) y^{s-\frac{c}{2}} = \int_{\gamma-i\infty}^{\gamma+i\infty} ds \frac{\Gamma(s)\Gamma(s+\frac{c}{2}-1)\zeta(2s)}{\pi^{s+\frac{c-1}{2}}\Gamma(s-\frac{1}{2})\zeta(2s-1)} \mathcal{E}_{\frac{c}{2}+s-1}^{c}(\mu) y^{s-\frac{c}{2}}$$
(5.31)

Because we take $\gamma > \frac{c}{2}$, this means that $\operatorname{Re}(\frac{c}{2} + s - 1) > c - 1$, which means we can write this as the following convergent sum:

$$\mathcal{E}^{c}_{\frac{c}{2}+s-1}(\mu) = \sum_{\Delta \in \mathcal{S}} (2\Delta)^{-\frac{c}{2}-s+1}.$$
(5.32)

Moreover we will expand the ratio of zeta functions

$$\frac{\zeta(2s)}{\zeta(2s-1)} = \sum_{n=1}^{\infty} b(n)n^{-2s},$$
(5.33)



Figure 5.1: (a) Pole structure of the integral in (5.27) in the complex *s* plane. The poles are located at $s = \frac{c}{2}, \frac{1+z_n}{2}, \frac{1+z_n^*}{2}$ (shown here for c = 3), where z_n are the nontrivial zeros of the Riemann zeta function with positive imaginary part. If the Riemann hypothesis is true, the tower of poles in the figure all occur at real part $\frac{3}{4}$, except for the pole at $s = \frac{c}{2}$. (b) Contour deformation of the integral to $\operatorname{Re}(s) > \frac{c}{2}$.

where b(n) is a number-theoretic function defined as

$$b(n) \coloneqq \sum_{k|n} k\mu(k). \tag{5.34}$$

where $\mu(n)$ is the Mobius function:

$$\mu(n) \coloneqq \begin{cases} (-1)^{\text{number of prime factors of } n} & n \text{ is square-free} \\ 0 & n \text{ is divisible by a prime squared.} \end{cases}$$
(5.35)

We can then rewrite (5.31) as

$$\int_{\gamma-i\infty}^{\gamma+i\infty} ds \pi^{s-\frac{c}{2}} \Gamma\left(\frac{c}{2}-s\right) \mathcal{E}_{\frac{c}{2}-s}^{c}(\mu) y^{s-\frac{c}{2}} = \sum_{\Delta \in \mathcal{S}} \sum_{n=1}^{\infty} b(n) \int_{\gamma-i\infty}^{\gamma+i\infty} ds \frac{\Gamma(s)\Gamma(s+\frac{c}{2}-1)}{\pi^{s+\frac{c-1}{2}}\Gamma(s-\frac{1}{2})} (2\Delta)^{-\frac{c}{2}-s+1} y^{s-\frac{c}{2}} n^{-2s}.$$
(5.36)

The integral in (5.36) is related to a confluent hypergeometric function of the second kind (see 13.4.18 of [170]), which we denote as U (and is given by HypergeometricU in Mathematica):

$$\frac{1}{2\pi i} \int_{\gamma-i\infty}^{\gamma+i\infty} ds \frac{\Gamma(s)\Gamma(s+\frac{c}{2}-1)}{\pi^{s+\frac{c-1}{2}}\Gamma(s-\frac{1}{2})} (2\Delta)^{-\frac{c}{2}-s+1} y^{s-\frac{c}{2}} n^{-2s} = \frac{y^{1-c}}{\sqrt{\pi}} n^{c-2} U\left(-\frac{1}{2}, \frac{c}{2}, \frac{2\pi n^2 \Delta}{y}\right) e^{-\frac{2\pi n^2 \Delta}{y}}$$
(5.37)

Thus we get a final crossing equation of:

$$1 + \sum_{\Delta \in \mathcal{S}} e^{-2\pi\Delta y} = \frac{\Lambda\left(\frac{c-1}{2}\right)}{\Lambda\left(\frac{c}{2}\right)} y^{1-c} + \varepsilon_c(\mu) y^{-\frac{c}{2}} + \sum_{k=1}^{\infty} \operatorname{Re}\left(\delta_{k,c}(\mu) y^{-\frac{c}{2}+1-\frac{z_k}{2}}\right) + \frac{y^{1-c}}{\sqrt{\pi}} \sum_{\Delta \in \mathcal{S}} \sum_{n=1}^{\infty} b(n) n^{c-2} U\left(-\frac{1}{2}, \frac{c}{2}, \frac{2\pi n^2 \Delta}{y}\right) e^{-\frac{2\pi n^2 \Delta}{y}}.$$
(5.38)

In addition to a rigorous derivation we have also numerically checked (5.38) for various values of c, y to a precision of 1 part in 10^{70} .

Another consistency check of (5.38) one can perform analytically is to consider the large y limit. In this limit, the LHS is dominated by 1 from the identity, but each term on the RHS is perturbatively small at large y. Similar to the lightcone bootstrap of four-point functions [7, 8], it turns out that the leading term on the LHS is reproduced by the infinite sum over Δ in the RHS. More precisely, one can show that

$$\frac{y^{1-c}}{\sqrt{\pi}} \sum_{n=0}^{\infty} b(n) n^{c-2} \int_0^\infty d\Delta \frac{2\pi^c \zeta(c-1)\Delta^{c-2}}{\zeta(c)\Gamma(\frac{c}{2})^2} U\left(-\frac{1}{2}, \frac{c}{2}, \frac{2\pi n^2 \Delta}{y}\right) e^{-\frac{2\pi n^2 \Delta}{y}} = 1$$
(5.39)

where $\frac{2\pi^c \zeta(c-1)\Delta^{c-2}}{\zeta(c)\Gamma(\frac{c}{2})^2}$ is the leading large Δ behavior of the spectral density (and which is the average spectral density for Narain theories; see [166, 167]). It might also be interesting to understand how the perturbatively small terms at large y on the RHS of (5.38) cancel among each other to give the non-perturbatively small corrections on the LHS.

5.3.3 Functionals

We would now like to apply linear functionals to (5.38) to obtain sum rules that can constrain the possible sets S. In particular we would like to put a bound on the scalar gap, meaning the lightest operator present in S. One immediate problem is that not every term in (5.38) is sign-definite. The term $\varepsilon_c(\mu)$ is not sign-definite, and the infinite terms $\delta_{k,c}(\mu)$ are also not sign-definite for any k. To remove the $\varepsilon_c(\mu)$ term is straightforward. Let us start by rewriting (5.38) as:

$$\sum_{\Delta \in \mathcal{S}} \left[y^{\frac{c}{2}} e^{-2\pi\Delta y} - \frac{y^{1-\frac{c}{2}}}{\sqrt{\pi}} \sum_{n=1}^{\infty} b(n) n^{c-2} U\left(-\frac{1}{2}, \frac{c}{2}, \frac{2\pi n^2 \Delta}{y}\right) e^{-\frac{2\pi n^2 \Delta}{y}} \right]$$
$$= -y^{\frac{c}{2}} + \frac{\Lambda\left(\frac{c-1}{2}\right)}{\Lambda\left(\frac{c}{2}\right)} y^{1-\frac{c}{2}} + \varepsilon_c(\mu) + \sum_{k=1}^{\infty} \operatorname{Re}\left(\delta_{k,c}(\mu) y^{1-\frac{z_k}{2}}\right).$$
(5.40)

Taking a derivative with respect to y removes the $\varepsilon_c(\mu)$ term. If we then redefine $t^2 := y^{-1}$ we get:

$$\sum_{\Delta \in \mathcal{S}} \left[t^{-c} (4\pi\Delta - ct^2) e^{-\frac{2\pi\Delta}{t^2}} - \frac{t^c}{\sqrt{\pi}} \sum_{n=1}^{\infty} b(n) n^{c-2} e^{-2\pi\Delta n^2 t^2} \times \left((c - 2 - 4\pi n^2 t^2 \Delta) U \left(-\frac{1}{2}, \frac{c}{2}, 2\pi n^2 t^2 \Delta \right) + 2\pi n^2 \Delta t^2 U \left(\frac{1}{2}, \frac{c}{2} + 1, 2\pi n^2 t^2 \Delta \right) \right) \right]$$
$$= ct^{2-c} + \frac{\Lambda \left(\frac{c-1}{2} \right)}{\Lambda \left(\frac{c}{2} \right)} (c - 2)t^c + \sum_{k=1}^{\infty} \operatorname{Re} \left(\delta_{k,c}(\mu) (z_k - 2)t^{z_k} \right).$$
(5.41)

Consider an even function $\varphi(t)$ that satisfies the following properties:

- $\varphi(t)$ and $\widehat{\varphi}(t)$ both decay rapidly (faster than any polynomial) at infinity
- $\varphi(t)$ and $\widehat{\varphi}(t)$ have no singularities at finite t
- $\varphi(0) = \widehat{\varphi}(0) = 0$
- $\int_0^\infty \frac{dt}{t} \varphi(t) t^s$ admits an analytic continuation to all $s \in \mathbb{C}$ (which we will call $M_\varphi(s)$),

where $\widehat{\varphi}$ is the Fourier transform of φ :

$$\widehat{\varphi}(p) \coloneqq \int_{-\infty}^{\infty} dx \ e^{-2\pi i p x} \varphi(x).$$
(5.42)

We define

$$\Phi(t) \coloneqq \sum_{n=1}^{\infty} \varphi(nt).$$
 (5.43)

The function $\Phi(t)$ can also be rewritten via the Poisson resummation formula as

$$\Phi(t) = -\frac{1}{2}\varphi(0) + \frac{1}{2t}\widehat{\varphi}(0) + \frac{1}{t}\sum_{n=1}^{\infty}\widehat{\varphi}\left(\frac{n}{t}\right)$$
$$= \frac{1}{t}\sum_{n=1}^{\infty}\widehat{\varphi}\left(\frac{n}{t}\right).$$
(5.44)

Combining (5.43) and (5.44) and the properties listed above, we see that $\Phi(t)$ decays faster than any polynomial at both small *t* and large *t*.

Now, we define a functional $\mathcal{F}^{\varphi}[h(t)]$ by

$$\mathcal{F}^{\varphi}[h(t)] \coloneqq \int_0^\infty \frac{dt}{t} h(t) \Phi(t).$$
(5.45)

Let us first consider the action of the functional on a power of *t*:

$$\mathcal{F}^{\varphi}[t^{s}] = \int_{0}^{\infty} \frac{dt}{t} t^{s} \Phi(t).$$
(5.46)

⁴We are extremely grateful to Danylo Radchenko for explaining this strategy to us. See [171] for further generalizations of this. The construction of the functionals in [171] seems to be reminiscent of the analytic functionals in [121]. It might be interesting to explore the connection further.

Because of the properties of $\Phi(t)$ discussed above, $\mathcal{F}^{\varphi}[t^s]$ is an analytic function on the entire complex *s* plane. Moreover, for Re(s) > 1, we can exchange the integration and the summation, which gives

$$\mathcal{F}^{\varphi}[t^{s}] = \int_{0}^{\infty} dt \ t^{s-1} \Phi(t)$$

$$= \int_{0}^{\infty} dt \ t^{s-1} \sum_{n=1}^{\infty} \varphi(nt)$$

$$= \sum_{n=1}^{\infty} n^{-s} \int_{0}^{\infty} dt \ t^{s-1} \varphi(t)$$

$$= \zeta(s) \int_{0}^{\infty} dt \ t^{s-1} \varphi(t), \qquad \operatorname{Re}(s) > 1. \qquad (5.47)$$

Properties of analytic continuation then imply that for all $s \in \mathbb{C}$,

$$\mathcal{F}^{\varphi}[t^s] = \zeta(s) M_{\varphi}(s). \tag{5.48}$$

From (5.48) we see that the functional \mathcal{F}^{φ} will remove the final sign-indefinite terms $\delta_{k,c}t^{z_k}$ in our crossing equation (5.41). We can then apply the functional \mathcal{F}^{φ} to (5.41) to get a positive sum rule the scalar operators must satisfy. Let us consider the situation where $\varphi(t)$ is a (finite) linear combination of Gaussians, for which $M_{\varphi}(s)$ is a sum of Gamma functions. In particular we consider the following family of $\varphi(t)$ defining the functionals:

$$\varphi(t) = \sum_{i=1}^{N} \alpha_i e^{-\pi k_i t^2},$$
(5.49)

where k_i, α_i are an arbitrary set of N real numbers. In order for $\varphi(t)$ to satisfy $\varphi(0) = \widehat{\varphi}(0) = 0$, we choose k_i, α_i subject to the constraints

$$\sum_{i=1}^{N} \alpha_i = 0,$$

$$\sum_{i=1}^{N} \alpha_i k_i^{-1/2} = 0.$$
(5.50)

With this definition of φ , we can define Φ and the action of the functional \mathcal{F} by using (5.43) and (5.45). If we then apply this functional to our crossing equation (5.41), we get a positive sum rule for the operators Δ . In Appendix D.2, we write down explicit formulas for the action of this functional on (5.41) with a single Gaussian $\varphi(t) = e^{-\pi kt^2}$ as a function of Δ and k.

Although in principle we could choose any functional via (5.49) obeying (5.50), for numerical calculations it will be more convenient to use functionals consisting of derivatives with respect to k, evaluated at k = 1 instead. To be more explicit, the sum rule we get after applying the functional from (5.49) is given by

$$\sum_{i=1}^{N} \alpha_i \operatorname{vac}(k_i) + \sum_{i=1}^{N} \sum_{\Delta} \alpha_i f(k_i, \Delta) = 0,$$
(5.51)

subject to the constraints (5.50). $f(k, \Delta)$ and -vac(k) are the actions of the functional on the LHS and RHS respectively of (5.41) (with explicit formulas given in Appendix D.2, see e.g. (D.7)). Let us consider the action of a single Gaussian of width k (i.e not yet obeying the constraints above):

$$\operatorname{vac}(k) + \sum_{\Delta} f(k, \Delta).$$
 (5.52)

The expression (5.52) is not equal to 0 because we have not obeyed the constraints (5.50). However, the only functions of k that it can be equal to are a constant term and a term proportional to $k^{-1/2}$. Any other term would allow some combination of functionals obeying (5.50) to not vanish, and thus contradict (5.51). Therefore we have

$$\operatorname{vac}(k) + \sum_{\Delta} f(k, \Delta) = c_0 + c_1 k^{-1/2},$$
 (5.53)

where c_0 , c_1 are k-independent constants (they could be theory-dependent however). From an explicit calculation of vac(k) and $f(k, \Delta)$ in Appendix D.2, we see that

$$\operatorname{vac}'(1) = \partial_k f(k, \Delta)|_{k=1} = 0,$$
 (5.54)

which implies $c_1 = 0.5$ Thus we have

$$\operatorname{vac}^{(n)}(1) + \sum_{\Delta} (\partial_k)^n f(k, \Delta)|_{k=1} = 0, \ n \ge 2$$
 (5.55)

which will be the basis for our functionals. (Only even values of n will provide independent equations, however.)

Notice that

$$\operatorname{vac}^{(n)}(1) = \lim_{\Delta \to 0} (\partial_k)^n f(k, \Delta)|_{k=1}, \ n \ge 2$$
 (5.56)

so indeed the vac term in (5.55) is precisely the contribution of the vacuum ($\Delta = 0$) to the sum rule (and the same is true in (5.51)).

⁵In fact it turns out that c_0 is related to $\varepsilon_c(\mu)$ (defined in (5.29)) via $c_0 = \frac{\pi \varepsilon_c(\mu)}{6}$. This in principle leads to a stronger crossing equation but we find that numerically it gives very similar bounds on the scalar gap, so we will not explore it further in this paper.

5.3.4 Numerical results

In this section, we present the numerical results for bounds on the scalar gap of $U(1)^c$ CFTs for various values of c obtained from using the basis of functionals (5.55). Note that the hypergeometric function in (5.38) for odd values of c reduces to an elementary function, which greatly simplifies the technical calculations. We therefore focus on odd values of c (although there is nothing in principle stopping the following from working for even c). We first consider the functional obtained from taking 2 and 4 derivatives of (5.55), and obtain a bound on $\Delta_{\text{scalar gap}}$ from these two sum rules, following the approach in [150]. Since we take at most 4 derivatives, we denote this bound as $\Delta_{\text{scalar gap}}^{(4)}$ (and more generally define a bound from at most n derivatives as $\Delta_{\text{scalar gap}}^{(n)}$). Note that $\Delta_{\text{scalar gap}}^{(n)}$ is obtained from $\frac{n}{2}$ functionals.

We have computed $\Delta_{\text{scalar gap}}^{(4)}$ for odd central charge up to 251.⁶ The results are plotted in Fig. 5.2. The bound at large *c* numerically appears to grow quadratically with *c*. Fitting it to a quadratic function gives

$$\Delta_{\text{scalar gap}}^{(4)}(c) \sim 0.0253303c^2 + 0.13506c + 0.400.$$
 (5.57)

The coefficient of the leading term is very close to $\frac{1}{4\pi^2} \approx 0.0253302959$. It may be possible to analytically prove that $\Delta_{\text{scalar gap}}^{(4)}(c) \sim \frac{c^2}{4\pi^2}$ at large c. Note that in this analysis we only considered 4 derivatives of (5.55), but it may be the case that if we take $c \to \infty$ with fixed number of derivatives, the leading asymptotics for the bound is independent of the number of derivatives. This is indeed what happens in the spinless modular bootstrap, where the large c bound at any fixed number of derivatives scales as $\frac{c}{6}$ [152].

It would be better to do the analysis with the opposite order of limits, where we take the number of derivatives to large before taking c large (as in [154]) and then extrapolate in c. We can obtain bounds from including a larger number of derivatives in (5.55) by using the semidefinite program solver SDPB [5, 172]. More precisely, we consider the sum rule

$$\sum_{n=2,4,\dots,n_{\max}} \alpha_n \operatorname{vac}^{(n)}(1) + \sum_{n=2,4,\dots,n_{\max}} \alpha_n \sum_{\Delta} (\partial_k)^n f(k,\Delta)|_{k=1} = 0.$$
(5.58)

Unfortunately, the function $(\partial_k)^n f(k, \Delta)|_{k=1}$ in (5.55) does not have a good approximation as a product of a positive function of Δ and a polynomial in Δ . Therefore,

⁶At c = 1 the crossing equation we use is slightly different due to a divergence of the zeta function at 1; see Appendix D.3.1 for discussion.



Figure 5.2: Plot of a bound on the scalar gap for $U(1)^c$ CFTs with 4 derivatives, up to central charge c = 251. The numerical data seems to be well-approximated by a quadratic function with leading coefficient $\frac{1}{4\pi^2}$ (see (5.57)).

we discretize in Δ -space and sample the function $(\partial_k)^n f(k, \Delta)|_{k=1}$ at various points $\Delta_1, \Delta_2, \ldots, \Delta_M$ above the scalar gap assumption, and use SDPB as a linear programming solver to look for a functional that satisfies

$$\sum_{\substack{n=2,4,\ldots,n_{\max}\\n=2,4,\ldots,n_{\max}}} \alpha_n \operatorname{vac}^{(n)}(1) = 1,$$
$$\sum_{\substack{n=2,4,\ldots,n_{\max}}} \alpha_n (\partial_k)^n f(k,\Delta)|_{k=1} \ge 0, \qquad \Delta = \Delta_1,\ldots,\Delta_M.$$
(5.59)

Finally, we check the positivity of the obtained functional for all $\Delta \ge \Delta_{\text{scalar gap}}^{(n_{\text{max}})}$ by hand. If there is a negative region, we sample more points there and rerun SDPB, and repeat this procedure until the functional is positive or SDPB gives a primal feasible solution⁷.

Using the method described above, we have computed $\Delta_{\text{scalar gap}}^{(n)}$ for $n = 10, 20, \dots, 60$ for central charge odd $c \le 27$. Our bounds are summarized in Table 5.1 and plotted in Fig. 5.3.⁸ We were not able to go to high enough central charge to do a reliable extrapolation to large c. There are two obstacles in going to large central charge. The

⁷We are extremely grateful to David Simmons-Duffin for explaining this approach to us.

⁸In Table 1 of [166], a bound on the gap (of any spin) was computed using the spinless modular bootstrap. Our results in Table 5.1 are specifically for scalars, and so in general are orthogonal. However, for c = 3, the bound in [166] is less than 1 and so must be a scalar, and is stronger than the bound we found at c = 3.

72	6
23	υ

С	$\Delta_{\text{scalar gap}}^{(10)}$	$\Delta_{\text{scalar gap}}^{(20)}$	$\Delta_{\text{scalar gap}}^{(30)}$	$\Delta_{\text{scalar gap}}^{(40)}$	$\Delta_{\text{scalar gap}}^{(50)}$	$\Delta_{\text{scalar gap}}^{(60)}$	$\Delta_{\mathrm{avg sgap}}$
1	0.507	$\frac{1}{2} + 7 \times 10^{-5}$	$\frac{1}{2} + 2 \times 10^{-6}$	$\approx \frac{1}{2}$	$\approx \frac{1}{2}$	$\approx \frac{1}{2}$	ill-defined
3	0.910	0.864	0.863	0.863	0.863	0.863	0.136
5	1.444	1.310	1.304	1.303	1.302	1.302	0.324
7	2.129	1.843	1.820	1.814	1.813	1.813	0.471
9	2.972	2.476	2.419	2.400	2.397	2.396	0.606
11	3.980	3.219	3.110	3.063	3.055	3.051	0.736
13	5.155	4.078	3.897	3.808	3.789	3.779	0.863
15	6.500	5.058	4.788	4.638	4.602	4.581	0.989
17	8.018	6.614	5.786	5.558	5.497	5.458	1.113
19	9.709	7.399	6.895	6.570	6.477	6.412	1.237
21	11.576	8.765	8.118	7.680	7.545	7.445	1.360
23	13.619	10.266	9.460	8.890	8.705	8.561	1.482
25	15.839	11.903	10.922	10.202	9.959	9.762	1.604
27	18.238	13.679	12.506	11.620	11.310	11.049	1.725

Table 5.1: Upper bounds on the scalar gap from $U(1)^c$ CFTs with odd $c \le 27$ after taking up to 10, 20, \cdots , 60 derivatives of our crossing equation (i.e., the maximum value of *n* in (5.55)) computed to three decimal places. We also compare it to the average Narain scalar gap, defined in (5.60) (though note that the optimal bound is different from the average). See Fig. 5.3 for a plot.

first is that the convergence of the bound as the derivative order $n \to \infty$ becomes slower for larger c. The second obstacle is that the number of terms in the sum rule (5.55) grows as c^4 (see the sum in (D.12)), which makes evaluating derivatives with respect to k very slow. It would be good if there were a more efficient way to compute the derivatives.

It is interesting to compare the bounds on the $U(1)^c$ scalar gap we get to the average Narain scalar gap. In [166] an expression for the average scalar gap of Narain theories was computed, by first calculating the average density of states for all Narain theories (under the Zamolodchikov measure), and determining when the integral of the average density of states is 1. By looking at the average density of



Figure 5.3: Plot of a bound on the scalar gap for $U(1)^c$ CFTs at odd $c \le 27$. The colors blue, orange, green, red, brown, and purple represent the bound we get at 10, 20, \cdots , 60 derivatives, respectively. The color black represents the average Narain scalar gap, for comparison. (However, there is no a priori reason the average Narain scalar gap and the optimal $U(1)^c$ scalar gap should be similar.) See Table 5.1 for the numerical data.

scalars, [166] got an average scalar gap of ⁹

$$\Delta_{\text{avg sgap}} = \left(\frac{\zeta(c)\Gamma\left(\frac{c}{2}\right)^2 (c-1)}{\zeta(c-1)2\pi^c} \right)^{\frac{1}{c-1}} = \frac{c}{2\pi e} + \frac{\log c}{2\pi e} + O(1).$$
(5.60)

Our numerical bounds at large c (including our bounds with four derivatives extrapolated to large c) appear to be very far from both the average Narain scalar gap and the bound on the gap of the lightest operator of any spin (see (5.3)). It would be interesting to explore further if our bounds on the scalar gap can be substantially improved by considering other crossing equations. Of course, it is possible that the optimal scalar gap behaves differently from both the average Narain scalar gap and the optimal gap at large c.

⁹Note that choosing the integrated average to be 1, as opposed to any other O(1) number less than 1, is somewhat of a convention. However, if we choose another cutoff, the result (5.60) changes very little.

5.4 General 2d CFTs

So far our discussion has been restricted to a very special class of CFTs, namely those with $U(1)^c$ chiral algebra. In this section we generalize to generic 2d CFTs, which only have Virasoro symmetry and no extended chiral algebra (though we pause to note that we do not have any explicit examples of such theories, even numerically [154]).

The main obstacle to repeating our analysis to general 2d CFTs is that the partition function is not square-integrable, due to the Casimir energy of the theory on a cylinder. For theories with $U(1)^c$ chiral algebra, when we factored out the characters of the theory and considered the primary counting partition function \hat{Z} , the resulting function grew only polynomially (~ $y^{c/2}$) at the cusp (see (5.18)). For theories with only Virasoro symmetry, however, the (Virasoro) primary counting partition function will grow as ~ $e^{2\pi \frac{c-1}{12}y}$ at large y. Although there are various ways we can get around this (see Sec. 4 of [165] for some discussions of other approaches), in this section we will simply take the partition function multiply by the same cusp form as we did for theories with $U(1)^c$ symmetry, and bound the resulting function we get. This will not give us a crossing equation acting only on the Virasoro scalar operators, but instead will give us an equation acting on a more complicated combination of operators of all spin.

To be more precise, let us consider any compact 2d CFT with c > 1 and only Virasoro symmetry as its maximal chiral algebra (although generalizations to other chiral algebras are simple). Suppose the partition function of this theory is $Z(\tau)$. We define the "fake scalars" of this theory as

$$Z^{\text{fake scalars}}(y) = \int_{-1/2}^{1/2} dx |\eta(\tau)|^{2c} Z(\tau).$$
 (5.61)

Note that the central charge *c* is not necessarily an integer in this analysis. We call this function "fake scalars" because if this theory were to have a $U(1)^c$ chiral algebra, then (5.61) would be a count of the scalars (under the $U(1)^c$ algebra). However, since the theory only has Virasoro symmetry, then $Z^{\text{fake scalars}}(y)$ does not in general have a positive *q*-expansion.

Even without the full $U(1)^c$ chiral algebra, the logic in deriving the crossing equation (5.38) in Sec. 5.3 will apply to $Z^{\text{fake scalars}}(y)$. We can still apply harmonic analysis to $y^{c/2}|\eta(\tau)|^{2c}Z(\tau) - E_{c/2}(\tau)$ and derive an analogous crossing equation for $Z^{\text{fake scalars}}(y)$. To be precise, the equation we derive is the following.
Let $a_c(n)$ be defined as¹⁰

$$\sum_{n=0}^{\infty} a_c(n)q^n = \prod_{n=1}^{\infty} (1-q^n)^{c-1}.$$
(5.62)

Then we have the following crossing equation in terms of the Virasoro primary operators of any c > 1 compact CFT:

$$\begin{split} &\sum_{n=0}^{\infty} e^{-4\pi yn} a_{c}(n)^{2} + \sum_{\Delta, j \in \mathcal{S} \cup \mathcal{S}^{null}} \sum_{n=0}^{\infty} e^{-2\pi y(\Delta + j + 2n)} a_{c}(n) a_{c}(n+j) = \\ &\frac{\Lambda\left(\frac{c-1}{2}\right)}{\Lambda\left(\frac{c}{2}\right)} y^{1-c} + \varepsilon y^{-\frac{c}{2}} + \sum_{n=1}^{\infty} \operatorname{Re}\left[\delta_{n} y^{-\frac{c}{2} + 1 - \frac{z_{n}}{2}}\right] \\ &+ \frac{y^{1-c}}{\sqrt{\pi}} \sum_{\Delta, j \in \mathcal{S} \cup \mathcal{S}^{null}} \sum_{n=0}^{\infty} \sum_{k=1}^{\infty} b(k) k^{c-2} U\left(-\frac{1}{2}, \frac{c}{2}, \frac{2\pi k^{2}(\Delta + j + 2n)}{y}\right) e^{-\frac{2\pi k^{2}(\Delta + j + 2n)}{y}} a_{c}(n) a_{c}(n+j) \\ &+ \frac{y^{1-c}}{\sqrt{\pi}} \sum_{n=1}^{\infty} \sum_{k=1}^{\infty} b(k) k^{c-2} U\left(-\frac{1}{2}, \frac{c}{2}, \frac{4\pi k^{2} n}{y}\right) e^{-\frac{4\pi k^{2} n}{y}} a_{c}(n)^{2}. \end{split}$$
(5.63)

In (5.63), S is the set of all non-vacuum Virasoro primary operators, labeled by their dimension $\Delta = h + \overline{h}$ and their spin $j = |h - \overline{h}|$. Moreover we define S^{null} formally as a set containing -2 operators of weight 1, spin 1 and 1 operator of weight 2, spin 0. This is simply to take into account the level 1 null state in the Virasoro vacuum block (i.e. that L_{-1} and \overline{L}_{-1} annihilate the vacuum). The LHS of (5.63) is precisely what we called $Z^{\text{fake scalars}}(y)$ above, written in terms of the Virasoro primary operators of the theory, which we denoted by the set S. For convenience we have assumed the theory has no additional conserved currents, but it is simple to generalize (5.63) to allow for them.

We have tested (5.63) numerically on the pure gravity partition function of [173, 174], which we will denote as $Z^{\text{MWK}}(\tau)$, at various values of the central charge. For simplicity we have ignored the null state at level 1 (even though this leads to a inconsistent chiral algebra due to the lack of charged twist zero states [175], the resulting function is still modular invariant with a gap to the first primary operator, so it will obey (5.63), without including the contribution from S^{null}). Strictly speaking we glossed over a subtlety in deriving (5.63). When we derived (5.38) we used the fact that $\mathcal{E}_s^c = \sum_{\Delta \in S} \Delta^{-2s}$ for Re(s) > c - 1 because the sum converges for those values of *s*. However if we define \mathcal{E}_s^c analogously for the "fake scalars," it

¹⁰At central charge 25, $a_{c=25}(n)$ is the Ramanujan tau function (up to a shift of the argument by 1).

could potentially be the case that there is no s such that the sum converges, due to the Cardy growth of the Virasoro primary operators. Nonetheless, we numerically find that (5.63) is still satisfied. It might be interesting to present a more rigorous derivation of this step.

We can then apply the same functionals on (5.63) as discussed in Sec. 5.3.3 to remove the sign-indefinite terms related to the nontrivial zeros of the zeta function. This gives sum rules the CFT must satisfy, where now all operators (instead of just scalars) participate. For example, at c = 3, and taking two derivatives in (5.55), we get

$$\sum_{\Delta, j \in \mathcal{S} \cup \mathcal{S}^{\text{null}}} \sum_{n=0}^{\infty} a_{c=3}(n) a_{c=3}(n+j) f(\Delta+j+2n) + \sum_{n=1}^{\infty} a_{c=3}(n)^2 f(2n) = \frac{\pi}{4}, \quad (5.64)$$

where

$$f(\Delta) \coloneqq \frac{\pi \left(-3 + 8\pi^2 \Delta + (3 + 4\pi^2 \Delta) \cosh(2\sqrt{2}\pi\sqrt{\Delta}) - 6\sqrt{2}\pi\sqrt{\Delta} \sinh(2\sqrt{2}\pi\sqrt{\Delta})\right)}{8 \sinh^4 \left(\sqrt{2}\pi\sqrt{\Delta}\right)}.$$
(5.65)

(This comes from evaluating $\partial_k^2|_{k=1}$ on (D.9).) We can apply the same family of functionals discussed in Sec. 5.3.3 to (5.63) more generally and try to derive bounds on the various quantities (e.g. scalar gap, gap, etc.) from this crossing equation. Unfortunately we run into two distinct issues that stop us from bounding generic theories.

First, we see that at large Δ , (5.65) falls off as $\sim e^{-2\pi\sqrt{2\Delta}}$. In fact from Appendix D.2 we see that regardless of the central charge or derivative order, the functionals used in Sec. 5.3.3 fall off with the same leading asymptotics. However, the asymptotic growth of operators in S comes from the Cardy formula [176] and is $\sim e^{2\pi\sqrt{\Delta(c-1)}}$. We thus see that if $c \geq 7$, the sum rule does not obviously converge. Note that for $U(1)^c$ CFTs this was not an issue because there, the asymptotic density of primary operators grew polynomially in Δ ($\sim \Delta^{c-2}$). As a check we have verified (5.64) for $Z^{\text{MWK}}(\tau)$ at c = 3, but the analogous computation at c = 9 diverges (even though both obey (5.63)). We have also verified (5.64) for various rational CFTs with c < 7 where we only decompose into Virasoro characters.

It is unfortunate that we only get a falloff in $e^{-\#\sqrt{\Delta}}$ in our sum rules. This only happened after we integrated against the function $\Phi(t)$ in (5.45). Before this integral

(e.g. in (5.38) and (5.63)), we had a falloff as $e^{-\#\Delta}$, which will always overwhelm the Cardy growth at any central charge. It would be interesting to see if there were another choice of functional that would both remove the sign-indefinite terms related to nontrivial zeros of the zeta function, but still preserve the faster falloff in dimension.

Second, the asymptotically large Δ behavior of

$$\sum_{n=0}^{\infty} a_c(n) a_c(n+j) f(\Delta + j + 2n)$$
 (5.66)

does not have fixed sign: for some spins the asymptotic Δ value is positive and for some spins it is negative. (This is true when one takes any number of derivatives of the crossing equation, not just two.) The root of this problem is that $a_c(n)a_c(n+j)$ does not have a definite sign. Thus there is no obvious way to construct functionals that have fixed sign for all spin and all dimensions larger than some cutoff.

We note that we have chosen to multiply the partition function by the cusp form $y^{c/2}|\eta(\tau)|^{2c}$ to render the partition function square-integrable. However any cusp form with a gap to the first excited state and that falls off at least as fast as $(q\bar{q})^{c/24}$ would be sufficient and give a similar crossing equation as (5.63). It might be useful to explore constraints one gets from other cusp forms.

Finally we end this section with an interesting observation. Our crossing equation (5.63) for Virasoro theories involves operators of all spins, since multiplying by $y^{c/2}|\eta(\tau)|^{2c}$ does not have an obvious physical interpretation for theories without a $U(1)^c$ extended current algebra. It would be better to have a crossing equation or sum rule that only involved scalar Virasoro primary operators. Surprisingly, we find strong hints that such a sum rule exists.

In order to get a sum rule acting only on scalar Virasoro primary operators, the natural thing to do is to multiply the partition function by $y^{1/2}|\eta(\tau)|^2$. This is the same object that we multiply for $U(1)^c$ theories for c = 1. Recall that there, we derived the following sum rule (see (D.15) and App. D.3.1):

$$\log k + \sum_{\Delta \in \mathcal{S}} \left[h(k, \Delta) - h(k^{-1}, \Delta) \right] = 0, \tag{5.67}$$

where

$$h(k,\Delta) \coloneqq \sqrt{2\pi}\sqrt{k\Delta}(1 - \coth(\sqrt{2\pi}\sqrt{k\Delta})) + 2\pi^2 k\Delta\operatorname{csch}^2(\sqrt{2\pi}\sqrt{k\Delta}) + \log(1 - e^{-2\sqrt{2\pi}\sqrt{k\Delta}})$$
(5.68)

(The expression (5.68) is just (D.14) at c = 1, where we multiplied through by a factor of -4 for convenience.)

Remarkably, we numerically find that (5.67) also holds for general Virasoro CFTs, where S is now the set of conformal dimensions of scalar Virasoro primary operators (minus $\frac{c-1}{12}$) subject the following constraints. First of all, due to the null state structure of the Virasoro vacuum character, we introduce an additional term in S of $\Delta - \frac{c-1}{12} = -\frac{c-25}{12}$ (assuming no spin 1 currents). Second of all, we do not include the log *k* term in the sum rule (since there is not necessarily a state with $\Delta - \frac{c-1}{12} = 0$ in the spectrum). Finally, and most importantly, the sum rule does not converge for sufficiently large *c*. At large Δ , we have

$$h(k, \Delta) - h(k^{-1}, \Delta) \sim e^{-2\sqrt{2}\pi\sqrt{\Delta \times \min(k, k^{-1})}},$$
 (5.69)

whereas

$$\rho^{\text{scalar primaries}}(\Delta) \sim e^{2\pi\sqrt{\frac{\Delta(c-1)}{3}}},$$
(5.70)

so our sum rule only converges if

$$c < 1 + 6\min(k, k^{-1}),$$
 (5.71)

which implies $c < 7.^{\text{n}}$

For various theories obeying (5.71), we very surprisingly find that the sum rule

$$\int_{-\frac{c-1}{12}}^{\infty} d\Delta \rho^{\text{scalars}}(\Delta) \left[h(k,\Delta) - h(k^{-1},\Delta) \right] = 0$$
(5.72)

is obeyed to arbitrarily high precision. For c < 7 we can use this to bound the Virasoro scalar gap. However, our bounds from this so far seem to be substantially weaker than those found in [154]. It would be extremely interesting if there were a way to analytically continue the sum in (5.72) to arbitrary central charge (and also to prove, or more honestly derive, (5.72)). If so, this could be a way to derive a Virasoro scalar gap for all central charge¹².

¹¹Note also $h(k, \Delta) - h(k^{-1}, \Delta)$ has poles at $\Delta = -\frac{n^2}{2k}$ and $\Delta = -\frac{n^2k}{2}$, $n \in \mathbb{N}$, which may be problematic for convergence. For example, if $c = 1 + 6kn^2$ or $c = 1 + 6k^{-1}n^2$, with $n \in \mathbb{N}$, then the vacuum term contributes as a pole.

¹²In [154], it was shown that no bound on the Virasoro scalar gap could be derived for $c \ge 25$ using the traditional modular bootstrap. This was due to the existence of a "spurious solution" to crossing of $\frac{J(\tau)+\overline{J}(\overline{\tau})}{\sqrt{\tau_2}|\eta(\tau)|^2}$, which lacks scalar primary operators (see discussion around Eqn (3.2) of [154]). However, if there exists a convergent sum rule like (5.72) for all *c* that only acts on scalar primary operators, then by definition it would vanish on the spurious solution found in [154], and one may be able to find a bound for $c \ge 25$.

5.5 2d CFTs and the Riemann Hypothesis

One interesting feature of our crossing equation (5.38) is that in the small y (high temperature) limit, the asymptotics are controlled by the real parts of the nontrivial zeros of the Riemann zeta function. Let us rewrite (5.38), defining the temperature $T := y^{-1}$, as

$$1 + \sum_{\Delta \in \mathcal{S}} e^{-\frac{2\pi\Delta}{T}} = \frac{\Lambda\left(\frac{c-1}{2}\right)}{\Lambda\left(\frac{c}{2}\right)} T^{c-1} + \varepsilon_c T^{\frac{c}{2}} + \sum_{k=1}^{\infty} T^{\frac{c}{2}-1+\frac{\operatorname{Re}(z_k)}{2}} \left[\operatorname{Re}(\delta_{k,c})\cos(\operatorname{Im}(z_k)\log T) - \operatorname{Im}(\delta_{k,c})\sin(\operatorname{Im}(z_k)\log T)\right] + O\left(e^{-2\pi\Delta_{\operatorname{gap}}T}\right).$$
(5.73)

At high temperature, second line of (5.73) behaves as a highly oscillatory function with an overall envelope controlled by $\text{Re}(z_k)$. The Riemann hypothesis says that for all k,

$$\operatorname{Re}(z_k) = 1/2,$$
 (5.74)

which would fix the envelope to be $T^{\frac{c}{2}-\frac{3}{4}}$. In other words, if the Riemann hypothesis is true, (5.73) can be written as

$$1 + \sum_{\Delta \in \mathcal{S}} e^{-\frac{2\pi\Delta}{T}} = \frac{\Lambda\left(\frac{c-1}{2}\right)}{\Lambda\left(\frac{c}{2}\right)} T^{c-1} + \varepsilon_c T^{\frac{c}{2}} + \sum_{k=1}^{\infty} T^{\frac{c}{2}-\frac{3}{4}} \left[\operatorname{Re}(\delta_{k,c}) \cos(\operatorname{Im}(z_k) \log T) - \operatorname{Im}(\delta_{k,c}) \sin(\operatorname{Im}(z_k) \log T) \right] + O\left(e^{-2\pi\Delta_{\operatorname{gap}}T}\right).$$
(5.75)

However, if the Riemann hypothesis is false, then there is at least one z_k with real part greater than 1/2,¹³ which changes the large temperature scaling in the second line of (5.75).¹⁴ Since the leading term of (5.73) is essentially the Cardy formula, then in some sense, the Riemann hypothesis makes a claim about the overall size of the "subsubleading" corrections to the Cardy formula.

¹³By the functional equation (5.12) and meromorphicity, the Riemann hypothesis being false implies a pair of zeros of the zeta function with identical imaginary part: one with real part greater than 1/2, one with real part less than 1/2.

¹⁴Note that there is a possibility that the residue at that zero vanishes, meaning $\delta_{k,c}$ vanishes in (5.75). However, this will only happen in a real codimension 2 subspace of the moduli space. Thus for a generic theory the scaling will change at large temperature. We thank Per Kraus for raising this question to us.

We can illustrate this with an explicit example. Let us consider the $SU(3)_1$ WZW model, and decompose the theory under the $U(1)^2$ chiral algebra (note that this is not the maximal chiral algebra). The scalar partition function is given by

$$Z_{SU(3)_{1}}^{\text{scalars}}(T) \coloneqq 1 + \sum_{\Delta \in \mathcal{S}_{SU(3)_{1}}} e^{-\frac{2\pi\Delta}{T}}$$
$$= 1 + \sum_{n=1}^{\infty} 48 \left(\sum_{k|n} (-1)^{k} \sin(\frac{k\pi}{3}) \right)^{2} e^{-\frac{4\pi n}{T}} + 24 \left(\sum_{k|3n-2} (-1)^{k} \sin(\frac{k\pi}{3}) \right)^{2} e^{-\frac{4\pi (n-\frac{2}{3})}{T}}$$
$$= 1 + 18e^{-\frac{4\pi}{3T}} + 36e^{-\frac{4\pi}{T}} + 18e^{-\frac{16\pi}{3T}} + 72e^{-\frac{28\pi}{3T}} + \cdots$$
(5.76)

For c = 2 the crossing equation (5.75) is slightly modified due to a pole at $\Lambda(1/2)$. As derived in (D.37), the crossing equation we get for a c = 2 Narain theory is

$$1 + \sum_{\Delta \in \mathcal{S}} e^{-\frac{2\pi\Delta}{T}} = \frac{3}{\pi} T \log T + \left[\widehat{E_{1}}(\rho) + \widehat{E_{1}}(\sigma) + \frac{3}{\pi} (\gamma_{E} + \log(4\pi) + 24\zeta'(-1) - 2)\right] T \\ + \sum_{k=1}^{\infty} \operatorname{Re}\left(\frac{4\pi^{\frac{z_{k}}{2}} \Lambda(\frac{1+z_{k}}{2})^{2} E_{\frac{1+z_{k}}{2}}(\rho) E_{\frac{1+z_{k}}{2}}(\sigma)}{2\Gamma(\frac{z_{k}}{2})\zeta'(z_{k})} T^{\frac{z_{k}}{2}}\right) \\ + \frac{T}{\sqrt{\pi}} \sum_{\Delta \in \mathcal{S}} \sum_{n=1}^{\infty} b(n) U\left(-\frac{1}{2}, 1, 2\pi\Delta n^{2}T\right) e^{-2\pi\Delta n^{2}T},$$
(5.77)

where $\widehat{E_1}$ is defined in (D.34). From the explicit form of the sum over k in (5.77), we see that the coefficient in front of $T^{\frac{z_k}{2}}$ falls off exponentially in k, so the sum converges rapidly.

For the case of the $SU(3)_1$ WZW model, we have $\rho = \sigma = e^{2\pi i/3}$. At large temperature, the last line of (5.77) becomes non-perturbatively small. Therefore if we subtract the first two terms on the RHS of (5.77) and go to large temperature, we should be able to probe the real part of the nontrivial zeros of the Riemann zeta function. Indeed, by evaluating (5.76) up to T = 300, we numerically are able to recover the first few nontrivial zeros of the Riemann zeta function. We plot this in Fig. 5.4. Of course for any 2d CFT we could make a similar plot using (5.63); here we picked this particular theory for concreteness.

We pause to note that we can only numerically go up to certain fixed temperature (e.g. T = 300) because we only computed a finite number of terms in (5.76). Since the residue falls off exponentially in $\text{Im}(z_k)$, this means we only numerically test the Riemann hypothesis up to a fixed imaginary part. Since the Riemann hypothesis has already been checked up to imaginary part 3×10^{12} [177], we emphasize that our numerics are *not* an independent check of the Riemann hypothesis.



Figure 5.4: Scalar part of the $SU(3)_1$ WZW model with first two leading terms subtracted, rescaled by $T^{1/4}$, plotted up to T = 300. If the Riemann hypothesis is true, then at large temperature, this function will remain bounded. However, if the Riemann hypothesis is false, at large temperatures the oscillations will grow in size and become unbounded (modulo the subtlety explained in footnote 14). In this plot, $\alpha := 2\widehat{E}_1(e^{2\pi i/3}) + \frac{3}{\pi}(\gamma_E + \log(4\pi) + 24\zeta'(-1) - 2) \approx 0.975$ (see (5.77)). By fitting this plot with oscillating functions in $\log(T)$, we can numerically recover the first few nontrivial zeros of the zeta function. (A similar plot can be made for any c > 1 CFT.)

However, it would be extremely interesting if there were a physical reason why the scalar partition function, with the first two leading terms subtracted off, had to scale as $T^{\frac{c}{2}-\frac{3}{4}}$. This would give a "physics explanation" of the Riemann hypothesis. We leave this problem as an exercise to the reader.

5.6 Future directions

In this paper we have derived a crossing equation acting only on the scalar operators of certain 2d CFTs. Rather curiously the crossing equation is intimately related to the nontrivial zeros of the Riemann zeta function. This allows us to rephrase the Riemann hypothesis purely in terms of the growth of states of scalar operators of $U(1)^c$ CFTs. By applying clever choices of linear functionals, we are able to derive positive sum rules that the scalar operators must satisfy, which lead to nontrivial bounds on the lightest non-vacuum scalar operator in $U(1)^c$ CFTs. We discuss generalizations to theories with only Virasoro symmetry. There are various future directions that may be interesting to pursue. In Sec. 5.4 we derived a crossing equation acting on all operators for theories with Virasoro symmetry. In order to make the partition function square-integrable, we multiplied by a cusp form, namely $y^{c/2}|\eta(\tau)|^{2c}$, which led to the inclusion of all spins to the crossing equation. It would be nice if there exists a crossing equation that does not rely on this, and acts only on the scalar Virasoro primary operators. In order to derive such an equation (if it exists), it might be necessary to consider some generalization of harmonic analysis to allow for exponential divergences as $y \to \infty$.

In the end of Sec. 5.4, we guessed such a sum rule for Virasoro CFTs with c < 7. It would be interesting to derive it more rigorously and somehow analytically continue the sum rule so it makes sense for arbitrary central charge.

Crossing equation for spin j?

In this paper we considered crossing equations acting on scalar operators of $U(1)^c$ CFTs (or "fake scalars" for the case of Virasoro CFTs). This was largely to avoid the Maass cusp forms in the spectral decomposition (which lack scalars – see (5.16)). It would be interesting if there were a generalization of our crossing equation to any fixed spin partition function.

In fact, the techniques we studied almost immediately generalize to any spin $j \neq 0$ crossing equation. Let us denote \mathcal{J} as the set of spin *j* primary operators of a $U(1)^c$ CFT. The spin *j* partition function is given by

$$\sum_{\Delta \in \mathcal{J}} e^{-2\pi\Delta y} = \frac{2\sigma_{c-1}(j)y^{\frac{1-c}{2}}K_{\frac{c-1}{2}}(2\pi jy)}{\Lambda\left(\frac{c}{2}\right)j^{\frac{c-1}{2}}} + \frac{1}{2\pi i}\int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} ds\pi^{s-\frac{c}{2}}\Gamma\left(\frac{c}{2}-s\right)\mathcal{E}_{\frac{c}{2}-s}^{c}\frac{\sigma_{2s-1}(j)K_{s-\frac{1}{2}}(2\pi jy)y^{\frac{1-c}{2}}}{\Lambda(s)j^{s-\frac{1}{2}}} + \sum_{n=1}^{\infty}\sum_{\epsilon=\pm}\frac{a_{j}^{(n,\pm)}(\widehat{Z},v_{n}^{\pm})}{2(v_{n}^{\pm},v_{n}^{\pm})}y^{\frac{1-c}{2}}K_{iR_{n}^{\pm}}(2\pi jy).$$
(5.78)

Unfortunately the last line in (5.78) seems very difficult to deal with analytically due to the sporadic nature of the Maass cusp form eigenvalues, but we can in fact do the integral in the second line using the same techniques as in Sec. 5.3.2. We move the contour of integration to the right, past $\text{Re}(s) = \frac{c}{2}$, so that we can expand the function $\mathcal{E}_{\frac{c}{2}+s-1}^{c}$ in terms of the scalar primary operators and then change the order of the sum and integral. From the discussion in Appendix D.1, we know the only

poles in $\mathcal{E}_{\frac{c}{2}-s}^{c}$ to the right of the contour occur at $s = \frac{c}{2}, \frac{1+z_n}{2}, \frac{1+z_n^*}{2}$ (see Fig. 5.1). The additional terms do not introduce any additional poles to the right of the contour. We thus get a crossing equation in terms of the spin *j* operators on the LHS and the scalars on the RHS (as well as the cusp forms). It may be interesting to analyze this equation further.

Better bounds on $U(1)^c$ theories?

In Table 5.1 our numerical bounds on the scalar operators of $U(1)^c$ theories are quite far from the average Narain scalar gap. For instance our numerical bounds seem to grow quadratically with c instead of linearly. This may be an indication that our crossing equation is not strong enough to pinpoint the CFT with the largest scalar gap. It would be interesting if we could modify the set of crossing equations we consider to get better bounds. For instance we could include both our crossing and the "traditional" modular invariance (or four-point function) crossing equations to see if we can get better bounds. Other avenues to explore may be to consider different functionals from the ones used in Sec. 5.3.3 (for example not just considering $\varphi(t)$ in (5.43) to be Gaussians) or somehow incorporate the residues at the nontrivial zeros of the zeta function into the crossing equation. It would also be nice to get numerical results for even c.

Four-point functions?

There is a well-known relation between crossing symmetry acting on a four-point functions of four scalar operators and modular covariance. For four identical operators, under an appropriate coordinate transformation, the four-point function should be modular invariant. (For different operators, it will transform as some vector-valued modular function.) It would be interesting if one could derive a crossing equation on certain correlation functions where only a one-dimensional slice of operators are exchanged (e.g. only scalar operators are exchanged instead of operators of all spin). It would be especially interesting if this could generalize to higher dimensions.

Applications to N = 4 SYM?

Besides in 2d CFT, another natural place that modular invariance shows up in string theory is in S-duality of $\mathcal{N} = 4$ super Yang-Mills theory. In [178] (see also [179]), harmonic analysis was used extensively to study various integrated correlators as a

function of the complexified Yang-Mills coupling. It would be interesting if there were some sort of crossing equation acting only on the zero-instanton sector (but note that the pole structure of the overlap with Eisensteins is different because there is no notion of a "scalar gap"; see [178] for discussions on this).

Acknowledgments

We thank Scott Collier, Liam Fitzpatrick, Tom Hartman, Per Kraus, Yuya Kusuki, Ying-Hsuan Lin, Aike Liu, Alex Maloney, Dalimil Mazáč, Hirosi Ooguri, Danylo Radchenko, David Simmons-Duffin, Herman Verlinde, Yifan Wang, and Xi Yin for very helpful discussions. We are especially grateful to Scott Collier and Liam Fitzpatrick for numerous discussions on related topics; to Danylo Radchenko for telling us the functionals used around (5.45); and to David Simmons-Duffin for enormous help on the numerics. We thank Liam Fitzpatrick and Tom Hartman for very help-ful comments on a draft. NB is supported by the Sherman Fairchild Foundation. CHC is supported by Simons Foundation grant 488657 (Simons Collaboration on the Nonperturbative Bootstrap) and a DOE Early Career Award under grant no. DE-SC0019085. NB and CHC are grateful for the hospitality of the Bootstrap 2022 conference at the University of Porto, as well as the hospitality of United Airlines Flights 1990 and 144, during which part of this work was completed.

BIBLIOGRAPHY

- S. Ferrara, A. F. Grillo, and R. Gatto. "Tensor representations of conformal algebra and conformally covariant operator product expansion". In: *Annals Phys.* 76 (1973), pp. 161–188. DOI: 10.1016/0003-4916(73) 90446-6.
- [2] A. M. Polyakov. "Nonhamiltonian approach to conformal quantum field theory". In: *Zh. Eksp. Teor. Fiz.* 66 (1974), pp. 23–42.
- [3] A. A. Belavin, Alexander M. Polyakov, and A. B. Zamolodchikov. "Infinite conformal symmetry in two-dimensional quantum field theory". In: *Nucl. Phys.* B241 (1984), pp. 333–380. DOI: 10.1016/0550-3213(84)90052-X.
- [4] Riccardo Rattazzi, Vyacheslav S. Rychkov, Erik Tonni, and Alessandro Vichi. "Bounding scalar operator dimensions in 4D CFT". In: *JHEP* 12 (2008), p. 031. DOI: 10.1088/1126-6708/2008/12/031. arXiv: 0807.0004 [hep-th].
- [5] David Simmons-Duffin. "A Semidefinite Program Solver for the Conformal Bootstrap". In: JHEP 06 (2015), p. 174. DOI: 10.1007/JHEP06(2015)174. arXiv: 1502.02033 [hep-th].
- [6] Filip Kos, David Poland, David Simmons-Duffin, and Alessandro Vichi. "Precision islands in the Ising and O(N) models". In: *JHEP* 08 (2016), p. 036. DOI: 10.1007 / JHEP08(2016) 036. arXiv: 1603.04436 [hep-th].
- [7] A. Liam Fitzpatrick, Jared Kaplan, David Poland, and David Simmons-Duffin. "The Analytic Bootstrap and AdS Superhorizon Locality". In: *JHEP* 12 (2013), p. 004. DOI: 10.1007/JHEP12(2013)004. arXiv: 1212.3616 [hep-th].
- [8] Zohar Komargodski and Alexander Zhiboedov. "Convexity and Liberation at Large Spin". In: JHEP 1311 (2013), p. 140. DOI: 10.1007/ JHEP11(2013)140. arXiv: 1212.4103 [hep-th].
- [9] Simon Caron-Huot. "Analyticity in Spin in Conformal Theories". In: *JHEP* 09 (2017), p. 078. DOI: 10.1007/JHEP09(2017)078. arXiv: 1703.00278 [hep-th].
- [10] David Simmons-Duffin, Douglas Stanford, and Edward Witten. "A spacetime derivation of the Lorentzian OPE inversion formula". In: *JHEP* 07 (2018), p. 085. DOI: 10.1007/JHEP07(2018)085. arXiv: 1711.03816 [hep-th].

- [11] Petr Kravchuk and David Simmons-Duffin. "Light-ray operators in conformal field theory". In: *JHEP* 11 (2018). [,236(2018)], p. 102.
 DOI: 10.1007/JHEP11(2018)102. arXiv: 1805.00098 [hep-th].
- [12] Diego M. Hofman and Juan Maldacena. "Conformal collider physics: Energy and charge correlations". In: *JHEP* 05 (2008), p. 012. DOI: 10.1088/1126-6708/2008/05/012. arXiv: 0803.1467 [hep-th].
- [13] Simon Caron-Huot, Dalimil Mazac, Leonardo Rastelli, and David Simmons-Duffin. "AdS bulk locality from sharp CFT bounds". In: *JHEP* 11 (2021), p. 164. DOI: 10.1007/JHEP11(2021)164. arXiv: 2106.10274 [hep-th].
- [14] Juan Maldacena, David Simmons-Duffin, and Alexander Zhiboedov.
 "Looking for a bulk point". In: *JHEP* 01 (2017), p. 013. DOI: 10.
 1007/JHEP01(2017)013. arXiv: 1509.03612 [hep-th].
- [15] Murat Kologlu, Petr Kravchuk, David Simmons-Duffin, and Alexander Zhiboedov. "The light-ray OPE and conformal colliders". In: *JHEP* 01 (2021), p. 128. DOI: 10.1007/JHEP01(2021)128. arXiv: 1905. 01311 [hep-th].
- [16] G.P. Korchemsky. "Energy correlations in the end-point region". In: JHEP 01 (2020), p. 008. DOI: 10.1007/JHEP01(2020)008. arXiv: 1905.01444 [hep-th].
- [17] Lance J. Dixon, Ian Moult, and Hua Xing Zhu. "Collinear limit of the energy-energy correlator". In: *Phys. Rev. D* 100.1 (2019), p. 014009.
 DOI: 10.1103/PhysRevD.100.014009. arXiv: 1905.01310 [hep-ph].
- C. L. Basham, L. S. Brown, S. D. Ellis, and S. T. Love. "Energy Correlations in electron-Positron Annihilation in Quantum Chromodynamics: Asymptotically Free Perturbation Theory". In: *Phys. Rev. D* 19 (1979), p. 2018. DOI: 10.1103/PhysRevD.19.2018.
- [19] C. Louis Basham, Lowell S. Brown, Stephen D. Ellis, and Sherwin T. Love. "Energy Correlations in electron Positron Annihilation: Testing QCD". In: *Phys. Rev. Lett.* 41 (1978), p. 1585. DOI: 10.1103/ PhysRevLett.41.1585.
- [20] Vittorio Del Duca, Claude Duhr, Adam Kardos, Gábor Somogyi, and Zoltán Trócsányi. "Three-Jet Production in Electron-Positron Collisions at Next-to-Next-to-Leading Order Accuracy". In: *Phys. Rev. Lett.* 117.15 (2016), p. 152004. DOI: 10.1103/PhysRevLett.117. 152004. arXiv: 1603.08927 [hep-ph].
- [21] Lance J. Dixon, Ming-Xing Luo, Vladyslav Shtabovenko, Tong-Zhi Yang, and Hua Xing Zhu. "Analytical Computation of Energy-Energy Correlation at Next-to-Leading Order in QCD". In: *Phys. Rev. Lett.* 120.10 (2018), p. 102001. DOI: 10.1103/PhysRevLett.120.102001. arXiv: 1801.03219 [hep-ph].

- [22] Ming-Xing Luo, Vladyslav Shtabovenko, Tong-Zhi Yang, and Hua Xing Zhu. "Analytic Next-To-Leading Order Calculation of Energy-Energy Correlation in Gluon-Initiated Higgs Decays". In: *JHEP* 06 (2019), p. 037. DOI: 10.1007/JHEP06(2019)037. arXiv: 1903.07277 [hep-ph].
- [23] Hao Chen, Ian Moult, XiaoYuan Zhang, and Hua Xing Zhu. "Rethinking jets with energy correlators: Tracks, resummation, and analytic continuation". In: *Phys. Rev. D* 102.5 (2020), p. 054012. DOI: 10.1103/PhysRevD.102.054012. arXiv: 2004.11381 [hep-ph].
- [24] A. V. Belitsky, S. Hohenegger, G. P. Korchemsky, E. Sokatchev, and A. Zhiboedov. "From correlation functions to event shapes". In: *Nucl. Phys. B* 884 (2014), pp. 305–343. DOI: 10.1016/j.nuclphysb. 2014.04.020. arXiv: 1309.0769 [hep-th].
- [25] A. V. Belitsky, S. Hohenegger, G. P. Korchemsky, E. Sokatchev, and A. Zhiboedov. "Energy-Energy Correlations in N=4 Supersymmetric Yang-Mills Theory". In: *Phys. Rev. Lett.* 112.7 (2014), p. 071601.
 DOI: 10.1103/PhysRevLett.112.071601. arXiv: 1311.6800 [hep-th].
- J. M. Henn, E. Sokatchev, K. Yan, and A. Zhiboedov. "Energy-energy correlation in N=4 super Yang-Mills theory at next-to-next-to-leading order". In: *Phys. Rev. D* 100.3 (2019), p. 036010. DOI: 10.1103/PhysRevD.100.036010. arXiv: 1903.05314 [hep-th].
- [27] B.R. Webber. "Estimation of power corrections to hadronic event shapes".
 In: *Phys. Lett. B* 339 (1994), pp. 148–150. DOI: 10.1016/0370-2693(94)91147-9. arXiv: hep-ph/9408222.
- [28] Gregory P. Korchemsky and George F. Sterman. "Power corrections to event shapes and factorization". In: *Nucl. Phys. B* 555 (1999), pp. 335–351. DOI: 10.1016/S0550-3213(99)00308-9. arXiv: hep-ph/9902341.
- [29] Andrii Verbytskyi, Andrea Banfi, Adam Kardos, Stefan Kluth, Pier Francesco Monni, Gabor Somogyi, Zoltan Szor, Zoltán Trócsányi, Zoltan Tulipant, and Giulia Zanderighi. " α_S from energy-energy correlations and jet rates in e^+e^- collisions". In: *PoS* ALPHAS2019 (2019), p. 003. DOI: 10.22323/1.365.0003.
- [30] Miguel S. Costa, Joao Penedones, David Poland, and Slava Rychkov.
 "Spinning Conformal Correlators". In: *JHEP* 11 (2011), p. 071. DOI: 10.1007/JHEP11(2011)071. arXiv: 1107.3554 [hep-th].
- [31] P. Abreu et al. "Consistent measurements of alpha(s) from precise oriented event shape distributions". In: *Eur. Phys. J. C* 14 (2000), pp. 557–584. DOI: 10.1007/s100520000354. arXiv: hep-ex/0002026.

- [32] G. Abbiendi et al. "Measurement of the longitudinal cross-section using the direction of the thrust axis in hadronic events at LEP". In: *Phys. Lett.* B 440 (1998), pp. 393–402. DOI: 10.1016/S0370-2693(98)01172-1. arXiv: hep-ex/9808035.
- [33] Vicent Mateu and Germán Rodrigo. "Oriented Event Shapes at N³LL $+O(\alpha_S^2)$ ". In: *JHEP* 11 (2013), p. 030. doi: 10.1007/JHEP11(2013) 030. arXiv: 1307.3513 [hep-ph].
- [34] Murat Kologlu, Petr Kravchuk, David Simmons-Duffin, and Alexander Zhiboedov. "Shocks, Superconvergence, and a Stringy Equivalence Principle". In: JHEP 11 (2020), p. 096. DOI: 10.1007/JHEP11(2020) 096. arXiv: 1904.05905 [hep-th].
- [35] João Penedones, Emilio Trevisani, and Masahito Yamazaki. "Recursion Relations for Conformal Blocks". In: *JHEP* 09 (2016), p. 070. DOI: 10.1007/JHEP09(2016)070. arXiv: 1509.00428 [hep-th].
- [36] Filip Kos, David Poland, and David Simmons-Duffin. "Bootstrapping the O(N) vector models". In: *JHEP* 1406 (2014), p. 091. DOI: 10. 1007/JHEP06(2014)091. arXiv: 1307.6856 [hep-th].
- [37] Filip Kos, David Poland, and David Simmons-Duffin. "Bootstrapping Mixed Correlators in the 3D Ising Model". In: *JHEP* 11 (2014), p. 109.
 DOI: 10.1007/JHEP11(2014)109. arXiv: 1406.4858 [hep-th].
- [38] Rajeev S. Erramilli, Luca V. Iliesiu, and Petr Kravchuk. "Recursion relation for general 3d blocks". In: *JHEP* 12 (2019), p. 116. DOI: 10. 1007/JHEP12(2019)116. arXiv: 1907.11247 [hep-th].
- [39] Denis Karateev, Petr Kravchuk, and David Simmons-Duffin. "Weight Shifting Operators and Conformal Blocks". In: JHEP 02 (2018). [,91(2017)], p. 081. DOI: 10.1007 / JHEP02 (2018) 081. arXiv: 1706.07813 [hep-th].
- [40] Steven Weinberg. "Six-dimensional Methods for Four-dimensional Conformal Field Theories". In: *Phys. Rev.* D82 (2010), p. 045031. DOI: 10.1103/PhysRevD.82.045031. arXiv: 1006.3480 [hep-th].
- [41] S. Ferrara, A. F. Grillo, and R. Gatto. "Manifestly conformal covariant operator-product expansion". In: *Lett. Nuovo Cim.* 2S2 (1971), pp. 1363–1369.
- [42] S. Ferrara, A.F. Grillo, and Raoul Gatto. "Manifestly conformal-covariant expansion on the light cone". In: *Phys. Rev. D* 5 (1972), pp. 3102–3108.
 DOI: 10.1103/PhysRevD.5.3102.
- [43] Jean-François Fortin and Witold Skiba. "Conformal Bootstrap in Embedding Space". In: *Phys. Rev. D* 93.10 (2016), p. 105047. DOI: 10. 1103/PhysRevD.93.105047. arXiv: 1602.05794 [hep-th].

- [44] V. K. Dobrev, G. Mack, V. B. Petkova, S. G. Petrova, and I. T. Todorov. "Harmonic Analysis on the n-Dimensional Lorentz Group and Its Application to Conformal Quantum Field Theory". In: *Lect. Notes Phys.* 63 (1977), pp. 1–280. DOI: 10.1007/BFb0009678.
- [45] Denis Karateev, Petr Kravchuk, and David Simmons-Duffin. "Harmonic Analysis and Mean Field Theory". In: *JHEP* 10 (2019), p. 217. DOI: 10.1007/JHEP10(2019)217. arXiv: 1809.05111 [hep-th].
- [46] David Simmons-Duffin. "Projectors, Shadows, and Conformal Blocks".
 In: JHEP 04 (2014), p. 146. DOI: 10.1007/JHEP04(2014)146. arXiv: 1204.3894 [hep-th].
- [47] V.K. Dobrev, V.B. Petkova, S.G. Petrova, and I.T. Todorov. "Dynamical Derivation of Vacuum Operator Product Expansion in Euclidean Conformal Quantum Field Theory". In: *Phys. Rev. D* 13 (1976), p. 887. DOI: 10.1103/PhysRevD.13.887.
- [48] Petr Kravchuk and David Simmons-Duffin. "Counting Conformal Correlators". In: JHEP 02 (2018), p. 096. DOI: 10.1007/JHEP02(2018) 096. arXiv: 1612.08987 [hep-th].
- [49] Clay Cordova and Kenan Diab. "Universal Bounds on Operator Dimensions from the Average Null Energy Condition". In: *JHEP* 02 (2018), p. 131. DOI: 10.1007/JHEP02(2018)131. arXiv: 1712.01089 [hep-th].
- [50] A. V. Belitsky, S. Hohenegger, G. P. Korchemsky, and E. Sokatchev. "N=4 superconformal Ward identities for correlation functions". In: *Nucl. Phys.* B904 (2016), pp. 176–215. DOI: 10.1016/j.nuclphysb. 2016.01.008. arXiv: 1409.2502 [hep-th].
- [51] A. V. Belitsky, S. Hohenegger, G. P. Korchemsky, E. Sokatchev, and A. Zhiboedov. "Event shapes in N = 4 super-Yang-Mills theory". In: *Nucl. Phys. B* 884 (2014), pp. 206–256. DOI: 10.1016/j.nuclphysb. 2014.04.019. arXiv: 1309.1424 [hep-th].
- [52] Miguel S. Costa and Tobias Hansen. "AdS Weight Shifting Operators". In: *JHEP* 09 (2018), p. 040. DOI: 10.1007/JHEP09(2018)040. arXiv: 1805.01492 [hep-th].
- [53] Christopher Beem, Leonardo Rastelli, and Balt C. van Rees. "More *N*=4 superconformal bootstrap". In: (2016). arXiv: 1612.02363 [hep-th].
- [54] Miguel S. Costa, Joao Penedones, David Poland, and Slava Rychkov.
 "Spinning Conformal Blocks". In: *JHEP* 11 (2011), p. 154. DOI: 10.
 1007/JHEP11(2011)154. arXiv: 1109.6321 [hep-th].

- [55] Hao Chen, Ming-Xing Luo, Ian Moult, Tong-Zhi Yang, Xiaoyuan Zhang, and Hua Xing Zhu. "Three point energy correlators in the collinear limit: symmetries, dualities and analytic results". In: *JHEP* 08.08 (2020), p. 028. DOI: 10.1007/JHEP08(2020)028. arXiv: 1912.11050 [hep-ph].
- [56] Simon Caron-Huot, Dalimil Mazac, Leonardo Rastelli, and David Simmons-Duffin. "Dispersive CFT Sum Rules". In: JHEP 05 (2021), p. 243. DOI: 10.1007/JHEP05(2021)243. arXiv: 2008.04931 [hep-th].
- [57] C. Louis Basham, Lowell S. Brown, S. D. Ellis, and S. T. Love. "Electron Positron Annihilation Energy Pattern in Quantum Chromodynamics: Asymptotically Free Perturbation Theory". In: *Phys. Rev. D* 17 (1978), p. 2298. DOI: 10.1103/PhysRevD.17.2298.
- [58] P. Abreu et al. "Energy-energy correlations in hadronic final states from Z0 decays". In: *Phys. Lett. B* 252 (1990), pp. 149–158. DOI: 10.1016/0370-2693(90)91097-U.
- [59] M. Z. Akrawy et al. "A Measurement of energy correlations and a determination of alpha-s (M2 (Z0)) in e+ e- annihilations at s**(1/2) = 91-GeV". In: *Phys. Lett. B* 252 (1990), pp. 159–169. DOI: 10.1016/0370-2693(90)91098-V.
- [60] K. Abe et al. "Measurement of alpha-s (M(Z)**2) from hadronic event observables at the Z0 resonance". In: *Phys. Rev. D* 51 (1995), pp. 962–984. DOI: 10.1103/PhysRevD.51.962. arXiv: hep-ex/9501003.
- [61] K. Abe et al. "Measurement of alpha-s from energy-energy correlations at the Z0 resonance". In: *Phys. Rev. D* 50 (1994), pp. 5580–5590. DOI: 10.1103/PhysRevD.50.5580. arXiv: hep-ex/9405006.
- [62] Andrei V. Belitsky, G. P. Korchemsky, and George F. Sterman. "Energy flow in QCD and event shape functions". In: *Phys. Lett. B* 515 (2001), pp. 297–307. DOI: 10.1016/S0370-2693(01)00899-1. arXiv: hep-ph/0106308.
- [63] Alexander Zhiboedov. "On Conformal Field Theories With Extremal a/c Values". In: JHEP 04 (2014), p. 038. DOI: 10.1007/JHEP04(2014)
 038. arXiv: 1304.6075 [hep-th].
- [64] Andrew J. Larkoski, Gavin P. Salam, and Jesse Thaler. "Energy Correlation Functions for Jet Substructure". In: *JHEP* 06 (2013), p. 108.
 DOI: 10.1007/JHEP06(2013)108. arXiv: 1305.0007 [hep-ph].
- [65] Thomas Faulkner, Robert G. Leigh, Onkar Parrikar, and Huajia Wang.
 "Modular Hamiltonians for Deformed Half-Spaces and the Averaged Null Energy Condition". In: *JHEP* 09 (2016), p. 038. DOI: 10.1007/ JHEP09(2016)038. arXiv: 1605.08072 [hep-th].

- [66] Thomas Hartman, Sandipan Kundu, and Amirhossein Tajdini. "Averaged Null Energy Condition from Causality". In: JHEP 07 (2017), p. 066. DOI: 10.1007/JHEP07(2017)066. arXiv: 1610.05308 [hep-th].
- [67] Clay Cordova, Juan Maldacena, and Gustavo J. Turiaci. "Bounds on OPE Coefficients from Interference Effects in the Conformal Collider". In: JHEP 11 (2017), p. 032. DOI: 10.1007/JHEP11(2017)032. arXiv: 1710.03199 [hep-th].
- [68] Clay Córdova and Shu-Heng Shao. "Light-ray Operators and the BMS Algebra". In: *Phys. Rev.* D98.12 (2018), p. 125015. DOI: 10.1103/ PhysRevD.98.125015. arXiv: 1810.05706 [hep-th].
- [69] Jun Gao, Vladyslav Shtabovenko, and Tong-Zhi Yang. "Energy-energy correlation in hadronic Higgs decays: analytic results and phenomenology at NLO". In: *JHEP* 02 (2021), p. 210. DOI: 10.1007/JHEP02(2021) 210. arXiv: 2012.14188 [hep-ph].
- [70] Markus A. Ebert, Bernhard Mistlberger, and Gherardo Vita. "The Energy-Energy Correlation in the back-to-back limit at N³LO and N³LL". In: *JHEP* 08 (2021), p. 022. DOI: 10.1007/JHEP08(2021) 022. arXiv: 2012.07859 [hep-ph].
- [71] Gregory P. Korchemsky, Emery Sokatchev, and Alexander Zhiboedov.
 "Generalizing event shapes: in search of lost collider time". In: *JHEP* 08 (2022), p. 188. DOI: 10.1007/JHEP08(2022)188. arXiv: 2106. 14899 [hep-th].
- [72] Gregory P. Korchemsky and Alexander Zhiboedov. "On the light-ray algebra in conformal field theories". In: *JHEP* 02 (2022), p. 140. DOI: 10.1007/JHEP02(2022)140. arXiv: 2109.13269 [hep-th].
- [73] Hao Chen, Ian Moult, and Hua Xing Zhu. "Spinning gluons from the QCD light-ray OPE". In: *JHEP* 08 (2022), p. 233. DOI: 10.1007/JHEP08(2022)233. arXiv: 2104.00009 [hep-ph].
- [74] Patrick T. Komiske, Ian Moult, Jesse Thaler, and Hua Xing Zhu. "Analyzing N-point Energy Correlators Inside Jets with CMS Open Data". In: (Jan. 2022). arXiv: 2201.07800 [hep-ph].
- [75] Jack Holguin, Ian Moult, Aditya Pathak, and Massimiliano Procura. "A New Paradigm for Precision Top Physics: Weighing the Top with Energy Correlators". In: (Jan. 2022). arXiv: 2201.08393 [hep-ph].
- [76] N. A. Sveshnikov and F. V. Tkachov. "Jets and quantum field theory". In: *Phys. Lett. B* 382 (1996). Ed. by B. B. Levchenko and V. I. Savrin, pp. 403–408. DOI: 10.1016/0370-2693(96)00558-8. arXiv: hepph/9512370.

- [77] Cyuan-Han Chang, Murat Kologlu, Petr Kravchuk, David Simmons-Duffin, and Alexander Zhiboedov. "Transverse spin in the light-ray OPE". In: JHEP 05 (2022), p. 059. DOI: 10.1007/JHEP05(2022)059. arXiv: 2010.04726 [hep-th].
- [78] Luis F. Alday, Agnese Bissi, and Tomasz Lukowski. "Large spin systematics in CFT". In: *JHEP* 11 (2015), p. 101. DOI: 10.1007/JHEP11(2015)101. arXiv: 1502.07707 [hep-th].
- [79] Luis F. Alday and Alexander Zhiboedov. "An Algebraic Approach to the Analytic Bootstrap". In: (2015). arXiv: 1510.08091 [hep-th].
- [80] Luis F. Alday. "Large Spin Perturbation Theory for Conformal Field Theories". In: *Phys. Rev. Lett.* 119.11 (2017), p. 111601. DOI: 10. 1103/PhysRevLett.119.111601. arXiv: 1611.01500 [hep-th].
- [81] David Simmons-Duffin. "The Lightcone Bootstrap and the Spectrum of the 3d Ising CFT". In: JHEP 03 (2017), p. 086. DOI: 10.1007/JHEP03(2017)086. arXiv: 1612.08471 [hep-th].
- [82] F.A. Dolan and H. Osborn. "Conformal partial waves and the operator product expansion". In: *Nucl.Phys.* B678 (2004), pp. 491–507. DOI: 10.1016/j.nuclphysb.2003.11.016. arXiv: hep-th/0309180 [hep-th].
- [83] Yu Nakayama and Hirosi Ooguri. "Bulk Locality and Boundary Creating Operators". In: *JHEP* 10 (2015), p. 114. DOI: 10.1007/JHEP10(2015) 114. arXiv: 1507.04130 [hep-th].
- [84] Abhijit Gadde. "Conformal constraints on defects". In: JHEP 01 (2020),
 p. 038. DOI: 10.1007 / JHEP01(2020) 038. arXiv: 1602.06354
 [hep-th].
- [85] Pedro Liendo, Leonardo Rastelli, and Balt C. van Rees. "The Bootstrap Program for Boundary CFT_d ". In: *JHEP* 07 (2013), p. 113. DOI: 10. 1007/JHEP07(2013)113. arXiv: 1210.4258 [hep-th].
- [86] Luis F. Alday and Simon Caron-Huot. "Gravitational S-matrix from CFT dispersion relations". In: JHEP 12 (2018), p. 017. DOI: 10.1007/ JHEP12 (2018)017. arXiv: 1711.02031 [hep-th].
- [87] Hao Chen, Ian Moult, and Hua Xing Zhu. "Quantum Interference in Jet Substructure from Spinning Gluons". In: *Phys. Rev. Lett.* 126.11 (2021), p. 112003. DOI: 10.1103/PhysRevLett.126.112003. arXiv: 2011.02492 [hep-ph].
- [88] B. Basso and G. P. Korchemsky. "Anomalous dimensions of high-spin operators beyond the leading order". In: *Nucl. Phys. B* 775 (2007), pp. 1–30. DOI: 10.1016/j.nuclphysb.2007.03.044. arXiv: hepth/0612247.

- [89] A. P. Bukhvostov, G. V. Frolov, L. N. Lipatov, and E. A. Kuraev.
 "Evolution equations for quasi-partonic operators". In: *Nucl. Phys. B* 258 (1985), pp. 601–646. DOI: 10.1016/0550-3213(85)90628-5.
- [90] V. M. Braun, G. P. Korchemsky, and Dieter Müller. "The Uses of conformal symmetry in QCD". In: *Prog. Part. Nucl. Phys.* 51 (2003), pp. 311–398. DOI: 10.1016/S0146-6410(03)90004-4. arXiv: hep-ph/0306057.
- [91] Yu. M. Makeenko. "CONFORMAL OPERATORS IN QUANTUM CHROMODYNAMICS". In: *Sov. J. Nucl. Phys.* 33 (1981), p. 440.
- [92] T. Ohrndorf. "Constraints From Conformal Covariance on the Mixing of Operators of Lowest Twist". In: *Nucl. Phys. B* 198 (1982), pp. 26–44. DOI: 10.1016/0550-3213(82)90542-9.
- [93] David J. Gross and Frank Wilczek. "Ultraviolet Behavior of Nonabelian Gauge Theories". In: *Phys. Rev. Lett.* 30 (1973). Ed. by J. C. Taylor, pp. 1343–1346. DOI: 10.1103/PhysRevLett.30.1343.
- [94] D. J. Gross and Frank Wilczek. "Asymptotically Free Gauge Theories I". In: *Phys. Rev. D* 8 (1973), pp. 3633–3652. DOI: 10.1103/PhysRevD.8.3633.
- [95] D. J. Gross and Frank Wilczek. "ASYMPTOTICALLY FREE GAUGE THEORIES. 2." In: *Phys. Rev. D* 9 (1974), pp. 980–993. DOI: 10.1103/ PhysRevD.9.980.
- [96] Daniel Z. Freedman, Samir D. Mathur, Alec Matusis, and Leonardo Rastelli. "Correlation functions in the CFT(d) / AdS(d+1) correspondence". In: Nucl. Phys. B 546 (1999), pp. 96–118. DOI: 10.1016/ S0550-3213(99)00053-X. arXiv: hep-th/9804058.
- [97] Alexander Atanasov, Adam Ball, Walker Melton, Ana-Maria Raclariu, and Andrew Strominger. "(2, 2) Scattering and the celestial torus". In: *JHEP* 07 (2021), p. 083. DOI: 10.1007/JHEP07(2021)083. arXiv: 2101.09591 [hep-th].
- [98] Lorenzo Cornalba, Miguel S. Costa, Joao Penedones, and Ricardo Schiappa. "Eikonal Approximation in AdS/CFT: Conformal Partial Waves and Finite N Four-Point Functions". In: *Nucl. Phys. B* 767 (2007), pp. 327–351. doi: 10.1016/j.nuclphysb.2007.01.007. arXiv: hep-th/0611123.
- [99] Y. Aharonov, A. Komar, and Leonard Susskind. "Superluminal behavior, causality, and instability". In: *Phys. Rev.* 182 (1969), pp. 1400– 1403. DOI: 10.1103/PhysRev.182.1400.

- [100] T. N. Pham and Tran N. Truong. "Evaluation of the Derivative Quartic Terms of the Meson Chiral Lagrangian From Forward Dispersion Relation". In: *Phys. Rev. D* 31 (1985), p. 3027. DOI: 10.1103/PhysRevD. 31.3027.
- [101] Allan Adams, Nima Arkani-Hamed, Sergei Dubovsky, Alberto Nicolis, and Riccardo Rattazzi. "Causality, analyticity and an IR obstruction to UV completion". In: *JHEP* 0610 (2006), p. 014. DOI: 10.1088/1126-6708/2006/10/014. arXiv: hep-th/0602178 [hep-th].
- [102] Xian O. Camanho, Jose D. Edelstein, Juan Maldacena, and Alexander Zhiboedov. "Causality Constraints on Corrections to the Graviton Three-Point Coupling". In: JHEP 02 (2016), p. 020. DOI: 10.1007/ JHEP02(2016)020. arXiv: 1407.5597 [hep-th].
- Brando Bellazzini, Joan Elias Miró, Riccardo Rattazzi, Marc Riembau, and Francesco Riva. "Positive moments for scattering amplitudes". In: *Phys. Rev. D* 104.3 (2021), p. 036006. DOI: 10.1103/PhysRevD.104. 036006. arXiv: 2011.00037 [hep-th].
- [104] Simon Caron-Huot and Vincent Van Duong. "Extremal Effective Field Theories". In: *JHEP* 05 (2021), p. 280. DOI: 10.1007/JHEP05(2021) 280. arXiv: 2011.02957 [hep-th].
- [105] Nima Arkani-Hamed, Tzu-Chen Huang, and Yu-tin Huang. "The EFT-Hedron". In: *JHEP* 05 (2021), p. 259. DOI: 10.1007/JHEP05(2021) 259. arXiv: 2012.15849 [hep-th].
- [106] Andrew J. Tolley, Zi-Yue Wang, and Shuang-Yong Zhou. "New positivity bounds from full crossing symmetry". In: *JHEP* 05 (2021), p. 255. DOI: 10.1007/JHEP05(2021)255. arXiv: 2011.02400 [hep-th].
- [107] Aninda Sinha and Ahmadullah Zahed. "Crossing Symmetric Dispersion Relations in Quantum Field Theories". In: *Phys. Rev. Lett.* 126.18 (2021), p. 181601. DOI: 10.1103/PhysRevLett.126.181601. arXiv: 2012.04877 [hep-th].
- [108] Simon Caron-Huot, Dalimil Mazac, Leonardo Rastelli, and David Simmons-Duffin. "Sharp boundaries for the swampland". In: JHEP 07 (2021), p. 110. DOI: 10.1007/JHEP07(2021)110. arXiv: 2102.08951 [hep-th].
- [109] Simon Caron-Huot, Yue-Zhou Li, Julio Parra-Martinez, and David Simmons-Duffin. "Causality constraints on corrections to Einstein gravity". In: JHEP 05 (2023), p. 122. DOI: 10.1007/JHEP05(2023)122. arXiv: 2201.06602 [hep-th].
- [110] Simon Caron-Huot, Yue-Zhou Li, Julio Parra-Martinez, and David Simmons-Duffin. "Graviton partial waves and causality in higher dimensions". In: *Phys. Rev. D* 108.2 (2023), p. 026007. DOI: 10.1103/ PhysRevD.108.026007. arXiv: 2205.01495 [hep-th].

- [111] Kelian Häring and Alexander Zhiboedov. "Gravitational Regge bounds". In: SciPost Phys. 16 (2024), p. 034. DOI: 10.21468/SciPostPhys. 16.1.034. arXiv: 2202.08280 [hep-th].
- [112] Johan Henriksson, Brian McPeak, Francesco Russo, and Alessandro Vichi. "Bounding violations of the weak gravity conjecture". In: JHEP 08 (2022), p. 184. DOI: 10.1007/JHEP08(2022)184. arXiv: 2203.08164 [hep-th].
- [113] Brian McPeak, Marco Venuti, and Alessandro Vichi. "Adding subtractions: comparing the impact of different Regge behaviors". In: (Oct. 2023). arXiv: 2310.06888 [hep-th].
- [114] Idse Heemskerk, Joao Penedones, Joseph Polchinski, and James Sully.
 "Holography from Conformal Field Theory". In: *JHEP* 10 (2009), p. 079. DOI: 10.1088/1126-6708/2009/10/079. arXiv: 0907.0151 [hep-th].
- Thomas Hartman, Sachin Jain, and Sandipan Kundu. "Causality Constraints in Conformal Field Theory". In: *JHEP* 05 (2016), p. 099. DOI: 10.1007/JHEP05(2016)099. arXiv: 1509.00014 [hep-th].
- [116] Nima Afkhami-Jeddi, Thomas Hartman, Sandipan Kundu, and Amirhossein Tajdini. "Einstein gravity 3-point functions from conformal field theory". In: *JHEP* 12 (2017), p. 049. DOI: 10.1007/JHEP12(2017) 049. arXiv: 1610.09378 [hep-th].
- [117] David Meltzer and Eric Perlmutter. "Beyond a = c: gravitational couplings to matter and the stress tensor OPE". In: JHEP 07 (2018), p. 157.
 DOI: 10.1007/JHEP07(2018)157. arXiv: 1712.04861 [hep-th].
- [118] Manuela Kulaxizi, Andrei Parnachev, and Alexander Zhiboedov. "Bulk Phase Shift, CFT Regge Limit and Einstein Gravity". In: *JHEP* 06 (2018), p. 121. DOI: 10.1007/JHEP06(2018) 121. arXiv: 1705. 02934 [hep-th].
- [119] Miguel S. Costa, Tobias Hansen, and João Penedones. "Bounds for OPE coefficients on the Regge trajectory". In: *JHEP* 10 (2017), p. 197.
 DOI: 10.1007/JHEP10(2017)197. arXiv: 1707.07689 [hep-th].
- [120] Alexandre Belin, Diego M. Hofman, and Grégoire Mathys. "Einstein gravity from ANEC correlators". In: *JHEP* 08 (2019), p. 032. DOI: 10.1007/JHEP08(2019)032. arXiv: 1904.05892 [hep-th].
- [121] Dalimil Mazac. "Analytic bounds and emergence of AdS₂ physics from the conformal bootstrap". In: *JHEP* 04 (2017), p. 146. DOI: 10.1007/ JHEP04(2017)146. arXiv: 1611.10060 [hep-th].
- [122] Dalimil Mazac and Miguel F. Paulos. "The analytic functional bootstrap. Part I: 1D CFTs and 2D S-matrices". In: *JHEP* 02 (2019), p. 162.
 DOI: 10.1007/JHEP02(2019)162. arXiv: 1803.10233 [hep-th].

- [123] Dean Carmi and Simon Caron-Huot. "A Conformal Dispersion Relation: Correlations from Absorption". In: *JHEP* 09 (2020), p. 009. DOI: 10.1007/JHEP09(2020)009. arXiv: 1910.12123 [hep-th].
- [124] Dalimil Mazáč, Leonardo Rastelli, and Xinan Zhou. "A basis of analytic functionals for CFTs in general dimension". In: *JHEP* 08 (2021), p. 140.
 DOI: 10.1007/JHEP08(2021)140. arXiv: 1910.12855 [hep-th].
- [125] Joao Penedones, Joao A. Silva, and Alexander Zhiboedov. "Nonperturbative Mellin Amplitudes: Existence, Properties, Applications". In: *JHEP* 08 (2020), p. 031. DOI: 10.1007/JHEP08(2020)031. arXiv: 1912.11100 [hep-th].
- [126] Dean Carmi, Joao Penedones, Joao A. Silva, and Alexander Zhiboedov.
 "Applications of dispersive sum rules: ε-expansion and holography". In: SciPost Phys. 10.6 (2021), p. 145. DOI: 10.21468/SciPostPhys. 10.6.145. arXiv: 2009.13506 [hep-th].
- [127] David Meltzer. "Dispersion Formulas in QFTs, CFTs, and Holography". In: *JHEP* 05 (2021), p. 098. DOI: 10.1007/JHEP05(2021)098. arXiv: 2103.15839 [hep-th].
- [128] Waltraut Knop and Dalimil Mazac. "Dispersive sum rules in AdS₂". In: *JHEP* 10 (2022), p. 038. DOI: 10.1007/JHEP10(2022)038. arXiv: 2203.11170 [hep-th].
- [129] P. A. M. Dirac. "Wave Equations in Conformal Space". In: Annals of Mathematics 37.2 (1936), pp. 429–442. ISSN: 0003486X. URL: http: //www.jstor.org/stable/1968455 (visited on 11/05/2023).
- [130] M. Luscher and G. Mack. "Global Conformal Invariance in Quantum Field Theory". In: *Commun. Math. Phys.* 41 (1975), pp. 203–234. DOI: 10.1007/BF01608988.
- [131] A. Liam Fitzpatrick and Jared Kaplan. "Unitarity and the Holographic S-Matrix". In: JHEP 1210 (2012), p. 032. DOI: 10.1007/JHEP10(2012) 032. arXiv: 1112.4845 [hep-th].
- [132] Edward Witten. "Analytic Continuation Of Chern-Simons Theory". In: AMS/IP Stud. Adv. Math. 50 (2011). Ed. by Joergen E. Andersen, Hans U. Boden, Atle Hahn, and Benjamin Himpel, pp. 347–446. arXiv: 1001.2933 [hep-th].
- [133] Junyu Liu, Eric Perlmutter, Vladimir Rosenhaus, and David Simmons-Duffin. "d-dimensional SYK, AdS Loops, and 6j Symbols". In: JHEP 03 (2019), p. 052. DOI: 10.1007/JHEP03(2019)052. arXiv: 1808. 00612 [hep-th].
- [134] Jun Murakami and Masakazu Yano. "On the volume of a hyperbolic and spherical tetrahedron". In: *Communications in analysis and geometry* 13.2 (2005), pp. 379–400.

- [135] Soumangsu Chakraborty, Subham Dutta Chowdhury, Tushar Gopalka, Suman Kundu, Shiraz Minwalla, and Amiya Mishra. "Classification of all 3 particle S-matrices quadratic in photons or gravitons". In: *JHEP* 04 (2020), p. 110. DOI: 10.1007/JHEP04(2020)110. arXiv: 2001.07117 [hep-th].
- [136] Yue-Zhou Li. "Notes on flat-space limit of AdS/CFT". In: JHEP 09 (2021), p. 027. DOI: 10.1007/JHEP09(2021)027. arXiv: 2106.04606 [hep-th].
- [137] S. Caron-Huot. "unpublished notes". In: ().
- [138] Subham Dutta Chowdhury, Abhijit Gadde, Tushar Gopalka, Indranil Halder, Lavneet Janagal, and Shiraz Minwalla. "Classifying and constraining local four photon and four graviton S-matrices". In: JHEP 02 (2020), p. 114. DOI: 10.1007/JHEP02(2020)114. arXiv: 1910.14392 [hep-th].
- [139] Simon Caron-Huot. "Holographic cameras: an eye for the bulk". In: JHEP 03 (2023), p. 047. DOI: 10.1007/JHEP03(2023)047. arXiv: 2211.11791 [hep-th].
- [140] Li-Yuan Chiang, Yu-tin Huang, Laurentiu Rodina, and He-Chen Weng."De-projecting the EFThedron". In: (Apr. 2022). arXiv: 2204.07140 [hep-th].
- [141] Hongbin Chen, A. Liam Fitzpatrick, and Denis Karateev. "Nonperturbative bounds on scattering of massive scalar particles in d ≥ 2". In: JHEP 12 (2022), p. 092. DOI: 10.1007/JHEP12(2022)092. arXiv: 2207.12448 [hep-th].
- [142] Subham Dutta Chowdhury, Kausik Ghosh, Parthiv Haldar, Prashanth Raman, and Aninda Sinha. "Crossing Symmetric Spinning S-matrix Bootstrap: EFT bounds". In: *SciPost Phys.* 13.3 (2022), p. 051. DOI: 10. 21468/SciPostPhys.13.3.051. arXiv: 2112.11755 [hep-th].
- [143] Yue-Zhou Li. "Effective field theory bootstrap, large-N χ PT and holographic QCD". In: (Oct. 2023). arXiv: 2310.09698 [hep-th].
- [144] Rajesh Gopakumar, Aninda Sinha, and Ahmadullah Zahed. "Crossing Symmetric Dispersion Relations for Mellin Amplitudes". In: *Phys. Rev. Lett.* 126.21 (2021), p. 211602. DOI: 10.1103/PhysRevLett.126. 211602. arXiv: 2101.09017 [hep-th].
- [145] Agnese Bissi and Aninda Sinha. "Positivity, low twist dominance and CSDR for CFTs". In: *SciPost Phys.* 14.4 (2023), p. 083. DOI: 10. 21468/SciPostPhys.14.4.083. arXiv: 2209.03978 [hep-th].
- [146] Brando Bellazzini, Marc Riembau, and Francesco Riva. "IR side of positivity bounds". In: *Phys. Rev. D* 106.10 (2022), p. 105008. DOI: 10.1103/PhysRevD.106.105008. arXiv: 2112.12561 [hep-th].

- [147] Ofer Aharony, Luis F. Alday, Agnese Bissi, and Eric Perlmutter. "Loops in AdS from Conformal Field Theory". In: *JHEP* 07 (2017), p. 036.
 DOI: 10.1007/JHEP07(2017)036. arXiv: 1612.03891 [hep-th].
- [148] David Meltzer, Eric Perlmutter, and Allic Sivaramakrishnan. "Unitarity Methods in AdS/CFT". In: JHEP 03 (2020), p. 061. DOI: 10.1007/ JHEP03(2020)061. arXiv: 1912.09521 [hep-th].
- [149] David Meltzer and Allic Sivaramakrishnan. "CFT unitarity and the AdS Cutkosky rules". In: JHEP 11 (2020), p. 073. DOI: 10.1007/ JHEP11(2020)073. arXiv: 2008.11730 [hep-th].
- [150] Simeon Hellerman. "A Universal Inequality for CFT and Quantum Gravity". In: JHEP 08 (2011), p. 130. DOI: 10.1007/JHEP08(2011) 130. arXiv: 0902.2790 [hep-th].
- [151] Christoph A. Keller and Hirosi Ooguri. "Modular Constraints on Calabi-Yau Compactifications". In: *Commun. Math. Phys.* 324 (2013), pp. 107–127. DOI: 10.1007/s00220-013-1797-8. arXiv: 1209.4649 [hep-th].
- [152] Daniel Friedan and Christoph A. Keller. "Constraints on 2d CFT partition functions". In: JHEP 10 (2013), p. 180. DOI: 10.1007/ JHEP10(2013)180. arXiv: 1307.6562 [hep-th].
- [153] Nathan Benjamin, Ethan Dyer, A. Liam Fitzpatrick, and Shamit Kachru.
 "Universal Bounds on Charged States in 2d CFT and 3d Gravity". In: *JHEP* 08 (2016), p. 041. DOI: 10.1007/JHEP08(2016)041. arXiv: 1603.09745 [hep-th].
- [154] Scott Collier, Ying-Hsuan Lin, and Xi Yin. "Modular Bootstrap Revisited". In: *JHEP* 09 (2018), p. 061. DOI: 10.1007/JHEP09(2018)061. arXiv: 1608.06241 [hep-th].
- [155] Jin-Beom Bae, Sungjay Lee, and Jaewon Song. "Modular Constraints on Conformal Field Theories with Currents". In: *JHEP* 12 (2017), p. 045. doi: 10.1007/JHEP12(2017)045. arXiv: 1708.08815 [hep-th].
- [156] Nima Afkhami-Jeddi, Thomas Hartman, and Amirhossein Tajdini.
 "Fast Conformal Bootstrap and Constraints on 3d Gravity". In: *JHEP* 05 (2019), p. 087. DOI: 10.1007/JHEP05(2019)087. arXiv: 1903.06272 [hep-th].
- [157] Baur Mukhametzhanov and Alexander Zhiboedov. "Modular invariance, tauberian theorems and microcanonical entropy". In: *JHEP* 10 (2019), p. 261. DOI: 10.1007/JHEP10(2019)261. arXiv: 1904.06359 [hep-th].

- [158] Thomas Hartman, Dalimil Mazáč, and Leonardo Rastelli. "Sphere Packing and Quantum Gravity". In: JHEP 12 (2019), p. 048. DOI: 10.1007/JHEP12(2019)048. arXiv: 1905.01319 [hep-th].
- [159] Nathan Benjamin, Hirosi Ooguri, Shu-Heng Shao, and Yifan Wang.
 "Light-cone modular bootstrap and pure gravity". In: *Phys. Rev. D* 100.6 (2019), p. 066029. DOI: 10.1103/PhysRevD.100.066029. arXiv: 1906.04184 [hep-th].
- [160] Sridip Pal and Zhengdi Sun. "Tauberian-Cardy formula with spin". In: *JHEP* 01 (2020), p. 135. DOI: 10.1007/JHEP01(2020)135. arXiv: 1910.07727 [hep-th].
- [161] Luis F. Alday and Jin-Beom Bae. "Rademacher Expansions and the Spectrum of 2d CFT". In: *JHEP* 11 (2020), p. 134. DOI: 10.1007/ JHEP11(2020)134. arXiv: 2001.00022 [hep-th].
- [162] Nima Afkhami-Jeddi, Henry Cohn, Thomas Hartman, David de Laat, and Amirhossein Tajdini. "High-dimensional sphere packing and the modular bootstrap". In: JHEP 12 (2020), p. 066. DOI: 10.1007/ JHEP12(2020)066. arXiv: 2006.02560 [hep-th].
- [163] Scott Collier, Dalimil Mazac, and Yifan Wang. "Bootstrapping Boundaries and Branes". In: (Dec. 2021). arXiv: 2112.00750 [hep-th].
- [164] Audrey Terras. Harmonic Analysis on Symmetric Spaces—Euclidean Space, the Sphere, and the Poincaré Upper Half-Plane. Springer New York, 2013. DOI: 10.1007/978-1-4614-7972-7. URL: https: //doi.org/10.1007%2F978-1-4614-7972-7.
- [165] Nathan Benjamin, Scott Collier, A. Liam Fitzpatrick, Alexander Maloney, and Eric Perlmutter. "Harmonic analysis of 2d CFT partition functions". In: *JHEP* 09 (2021), p. 174. DOI: 10.1007/JHEP09(2021)174. arXiv: 2107.10744 [hep-th].
- [LMFDB] The LMFDB Collaboration. *The L-functions and Modular Forms Database*. http://www.lmfdb.org. [Online]. 2022.
 - [166] Nima Afkhami-Jeddi, Henry Cohn, Thomas Hartman, and Amirhossein Tajdini. "Free partition functions and an averaged holographic duality". In: JHEP 01 (2021), p. 130. DOI: 10.1007/JHEP01(2021)130. arXiv: 2006.04839 [hep-th].
 - [167] Alexander Maloney and Edward Witten. "Averaging over Narain moduli space". In: *JHEP* 10 (2020), p. 187. DOI: 10.1007/JHEP10(2020)187. arXiv: 2006.04855 [hep-th].
 - [168] N. A. Obers and B. Pioline. "Eisenstein series and string thresholds". In: *Commun. Math. Phys.* 209 (2000), pp. 275–324. DOI: 10.1007/ s002200050022. arXiv: hep-th/9903113.

- [169] Carlo Angelantonj, Ioannis Florakis, and Boris Pioline. "A new look at one-loop integrals in string theory". In: *Commun. Num. Theor. Phys.* 6 (2012), pp. 159–201. DOI: 10.4310/CNTP.2012.v6.n1.a4. arXiv: 1110.5318 [hep-th].
- [170] NIST Digital Library of Mathematical Functions. http://dlmf.nist.gov/, Release 1.1.6 of 2022-06-30. F. W. J. Olver, A. B. Olde Daalhuis, D. W. Lozier, B. I. Schneider, R. F. Boisvert, C. W. Clark, B. R. Miller, B. V. Saunders, H. S. Cohl, and M. A. McClain, eds. URL: http: //dlmf.nist.gov/.
- [171] Andriy Bondarenko, Danylo Radchenko, and Kristian Seip. "Fourier interpolation with zeros of zeta and *L*-functions". In: (2020). arXiv: 2005.02996 [math.NT].
- [172] Walter Landry and David Simmons-Duffin. "Scaling the semidefinite program solver SDPB". In: (Sept. 2019). arXiv: 1909.09745 [hep-th].
- [173] Alexander Maloney and Edward Witten. "Quantum Gravity Partition Functions in Three Dimensions". In: *JHEP* 02 (2010), p. 029. DOI: 10.1007/JHEP02(2010)029. arXiv: 0712.0155 [hep-th].
- [174] Christoph A. Keller and Alexander Maloney. "Poincare Series, 3D Gravity and CFT Spectroscopy". In: JHEP 02 (2015), p. 080. DOI: 10.1007/JHEP02(2015)080. arXiv: 1407.6008 [hep-th].
- [175] Nathan Benjamin, Hirosi Ooguri, Shu-Heng Shao, and Yifan Wang.
 "Twist gap and global symmetry in two dimensions". In: *Phys. Rev. D* 101.10 (2020), p. 106026. DOI: 10.1103/PhysRevD.101.106026. arXiv: 2003.02844 [hep-th].
- [176] John L. Cardy. "Operator Content of Two-Dimensional Conformally Invariant Theories". In: *Nucl. Phys. B* 270 (1986), pp. 186–204. DOI: 10.1016/0550-3213(86)90552-3.
- [177] David Platt and Tim Trudgian. "The Riemann hypothesis is true up to 3.10¹²". In: *Bull. London Math. Soc.* 53 (2021), p. 792. DOI: 10.1112/blms.12460. arXiv: 2004.09765 [math.NT].
- [178] Scott Collier and Eric Perlmutter. "Harnessing S-Duality in $\mathcal{N} = 4$ SYM & Supergravity as $SL(2,\mathbb{Z})$ -Averaged Strings". In: (Jan. 2022). arXiv: 2201.05093 [hep-th].
- [179] Michael B. Green, Stephen D. Miller, and Pierre Vanhove. " $SL(2,\mathbb{Z})$ invariance and D-instanton contributions to the D^6R^4 interaction". In: *Commun. Num. Theor. Phys.* 09 (2015), pp. 307–344. DOI: 10.4310/ CNTP.2015.v9.n2.a3. arXiv: 1404.2192 [hep-th].

- [180] Petr Kravchuk. "Casimir recursion relations for general conformal blocks". In: *JHEP* 02 (2018). [,164(2017)], p. 011. DOI: 10.1007/JHEP02(2018)011. arXiv: 1709.05347 [hep-th].
- [181] Vasiliy Sergeyevich Vladimirov. *Methods of the theory of functions of many complex variables*. MIT Press: Cambridge, Massachusetts, 1966.
- Petr Kravchuk, Jiaxin Qiao, and Slava Rychkov. "Distributions in CFT. Part I. Cross-ratio space". In: *JHEP* 05 (2020), p. 137. DOI: 10.1007/ JHEP05(2020)137. arXiv: 2001.08778 [hep-th].
- [183] M. F. Atiyah. "Resolution of singularities and division of distributions". In: Comm. Pure Appl. Math. 23 (1970), pp. 145–150. ISSN: 0010-3640.
 DOI: 10.1002/cpa.3160230202. URL: https://doi.org/10.1002/cpa.3160230202.
- [184] I. N. Bernstein and S. I. Gelfand. "Meromorphy of the function P^{λ} ". In: *Funkcional. Anal. i Prilozen.* 3.1 (1969), pp. 84–85. ISSN: 0374-1990.
- [185] Matthijs Hogervorst and Slava Rychkov. "Radial Coordinates for Conformal Blocks". In: *Phys.Rev.* D87.10 (2013), p. 106004. DOI: 10. 1103/PhysRevD.87.106004. arXiv: 1303.1111 [hep-th].
- [186] Connor Behan. "Conformal blocks for highly disparate scaling dimensions". In: JHEP 09 (2014), p. 005. DOI: 10.1007/JHEP09(2014)005. arXiv: 1402.5698 [hep-th].
- [187] Baur Mukhametzhanov and Alexander Zhiboedov. "Analytic Euclidean Bootstrap". In: *JHEP* 10 (2019), p. 270. DOI: 10.1007/JHEP10(2019) 270. arXiv: 1808.03212 [hep-th].
- [188] R. Dijkgraaf, Erik P. Verlinde, and Herman L. Verlinde. "On Moduli Spaces of Conformal Field Theories with c ≥ 1". In: Perspectives in String Theory: Proceedings of the Niels Bohr Institute/Nordita Meeting. Copenhagen: World Scientific, 1987.

APPENDICES TO CHAPTER 2

A.1 Comments on existence of light-ray operators

In section 2.4.3 we described a picture in which light-ray operators exist in any CFT and provide analytic continuation of local operators in the sense of equation (2.97). This picture is supported, for example, by perturbative examples, where explicit expressions can be given for $\widehat{\mathbb{O}}_{i,J,\lambda}^{\pm}(x, z, \mathbf{w})$ in terms of fundamental fields and (2.97) can be verified. However, this picture is not rigorously known to be valid in non-perturbative CFTs as we now review.

In [11] a construction was given for light-ray operators in general CFTs, which we reviewed in 2.4.3. The following statements need to be established before it can be claimed that the story of section 2.4.3 is correct.

First of all, it needs to be shown that the functions

$$\langle \Psi | \mathbb{O}_{\Lambda, J}^{\pm}(x, z) | \Phi \rangle$$
 (A.1)

are meromorphic in Δ for general *J*. Furthermore, we need to prove that the positions of the poles are independent of the choice of the states Ψ , Φ , as well as of the operators ϕ_1, ϕ_2 used to define $\mathbb{O}_{\Delta,J}^{\pm}$. Finally, we have to argue that the residues of these poles depend on ϕ_1, ϕ_2 only through an OPE coefficient $f_{12O^{\dagger}}$.

All these statements are known to be true for non-negative integer J (with $(-1)^J = \pm$) [11] but, to the best of our knowledge, no proof is known for other values. In the main text we assume that these are true, since this simplifies the statement of our results. However, even if none of the above statements hold, the results of this paper, and in particular (2.147), continue to hold in the following sense. One needs to take matrix element of (2.147) between the states of interest, which for concreteness we take to be created by single insertions of primaries O_3 , O_4 . The right-hand side is then given by δ integrals of

$$\langle O_3 \mathbb{O}^{\pm}_{\delta+1,J,\lambda,(a)} O_4 \rangle,$$
 (A.2)

which can be expressed in terms of the function $C_{ab}(\Delta, J, \lambda)$ which is computed by the Lorentzian inversion integral [11] for the four-point function

$$\langle O_1 O_2 O_3 O_4 \rangle. \tag{A.3}$$

The Lorentzian inversion integral and thus $C_{ab}(\Delta, J, \lambda)$ is well-defined on the δ integration contour of (2.147), and so we get a rigorous interpretation of (2.147) without assuming any of the above facts about the light-ray operators. Moreover, the entire derivation of (2.147) can be carried out in this language, which is what is essentially done in [15] and appendix A.4.

In this language, where the matrix elements

$$\langle O_3 \mathbf{L}[O_1] \mathbf{L}[O_2] O_4 \rangle$$
 (A.4)

are expressed as an integral of $C_{ab}(\Delta, J, \lambda)$ along principal series in Δ , the analysis of the small angle limit between the detectors can be carried out in the usual way, by analytically continuing $C_{ab}(\Delta, J, \lambda)$ away from the principal series and deforming the integration contour to the right, picking up the singularities that one encounters on the way.

A.2 Conventions for two- and three-point structures

We follow the same conventions for two- and three-point structures of tracelesssymmetric operators as in [11]. In particular, we define

$$\langle O(x_1, z_1)O(x_2, z_2) \rangle = \frac{(-2z_1 \cdot I(x_{12}) \cdot z_2)^J}{x_{12}^{2\Delta}}$$
 (A.5)

where

$$I^{\mu}{}_{\nu}(x) = \delta^{\mu}{}_{\nu} - 2\frac{x^{\mu}x_{\nu}}{x^2}.$$
 (A.6)

For three-point structures we define

$$\langle \phi_1(x_1)\phi_2(x_2)O(x_3,z)\rangle = \frac{(2z \cdot x_{23} x_{13}^2 - 2z \cdot x_{13} x_{23}^2)^J}{x_{12}^{\Delta_1 + \Delta_2 - \Delta + J} x_{13}^{\Delta_1 + \Delta - \Delta_2 + J} x_{23}^{\Delta_2 + \Delta - \Delta_1 + J}}.$$
 (A.7)

In terms of embedding-space formalism these become

$$\langle O(X_1, Z_1) O(X_2, Z_2) \rangle = \frac{(-2H_{12})^J}{X_{12}^{\Delta + J}},$$
 (A.8)

$$\langle \phi_1(X_1)\phi_2(X_2)O(X_3, Z_3) \rangle = \frac{(-2V_{3,12})^J}{X_{12}^{\frac{\Delta_1+\Delta_2-\Delta-J}{2}}X_{13}^{\frac{\Delta_1+\Delta-\Delta_2+J}{2}}X_{23}^{\frac{\Delta_2+\Delta-\Delta_1+J}{2}}},$$
 (A.9)

where as usual

$$X_{ij} \equiv -2X_i \cdot X_j \tag{A.10}$$

$$V_{i,jk} \equiv \frac{Z_i \cdot X_j X_i \cdot X_k - Z_i \cdot X_k X_i \cdot X_j}{X_j \cdot X_k}, \tag{A.11}$$

$$H_{ij} \equiv -2(Z_i \cdot Z_j X_i \cdot X_j - Z_i \cdot X_j Z_j \cdot X_i).$$
(A.12)

$$(X^+, X^-, X^\mu) = (1, x^2, x^\mu), \quad (Z^+, Z^-, Z^\mu) = (0, 2x \cdot z, z^\mu),$$
 (A.13)

and the embedding space metric is

$$X^{2} = -X^{+}X^{-} + X^{\mu}X_{\mu}.$$
 (A.14)

We use these conventions both for *d*-dimensional structures as well as for (d - 2)dimensional celestial structures. For example, replacing $X \to z, Z \to w, \Delta \to \delta$ and $J \to j$ we find

$$\langle \phi_{1}(z_{1})\phi_{2}(z_{2})O(z,w)\rangle = \frac{(-2\frac{w\cdot z_{1}z\cdot z_{2}-w\cdot z_{2}z\cdot z_{1}}{z_{1}\cdot z_{2}})^{j}}{(-2z_{1}\cdot z_{2})^{\frac{\delta_{1}+\delta_{2}-\delta_{-j}}{2}}(-2z_{1}\cdot z)^{\frac{\delta_{1}+\delta-\delta_{2}+j}{2}}(-2z_{2}\cdot z)^{\frac{\delta_{2}+\delta-\delta_{1}+j}{2}}} \\ = \frac{(4w\cdot z_{1}z\cdot z_{2}-4w\cdot z_{2}z\cdot z_{1})^{j}}{(-2z_{1}\cdot z_{2})^{\frac{\delta_{1}+\delta_{2}-\delta+j}{2}}(-2z_{1}\cdot z)^{\frac{\delta_{1}+\delta-\delta_{2}+j}{2}}(-2z_{2}\cdot z)^{\frac{\delta_{2}+\delta-\delta_{1}+j}{2}}} \\ = \frac{(-4)^{j}(w\cdot z_{2}z\cdot z_{1}-w\cdot z_{1}z\cdot z_{2})^{j}}{(-2z_{1}\cdot z_{2})^{\frac{\delta_{1}+\delta_{2}-\delta+j}{2}}(-2z_{2}\cdot z)^{\frac{\delta_{2}+\delta-\delta_{1}+j}{2}}},$$
(A.15)

in agreement with (2.86).

Sometimes we need the standard tensor structures for continuous spin. We define them for the Wightman functions as

$$\langle 0|\phi_1(x_1)O(x_3,z)\phi_2(x_2)|0\rangle = \frac{(2z \cdot x_{23} x_{13}^2 - 2z \cdot x_{13} x_{23}^2)^J}{x_{12}^{\Delta_1 + \Delta_2 - \Delta + J} x_{13}^{\Delta_1 + \Delta - \Delta_2 + J} x_{23}^{\Delta_2 + \Delta - \Delta_1 + J}}.$$
 (A.16)

For non-integer *J*, this is defined to be positive for $x_{ij}^2 > 0$ and $z \cdot x_{23} x_{13}^2 - z \cdot x_{13} x_{23}^2 > 0$. The values in other configurations are obtained by analytic continuation assuming standard analyticity properties of Wightman functions.

A.3 A Lorentzian formula for the light-ray kernel

In this appendix, we prove the Lorentzian formula for the light-ray kernel (2.152). We follow the notation of [11]. Our starting point is the generalized Lorentzian inversion formula

$$C_{ab}^{t}(\Delta, J, \lambda) = \frac{-1}{2\pi i} \int_{\substack{3>4\\1>2}}^{3>4} \frac{d^{d}x_{1}\cdots d^{d}x_{4}}{\operatorname{vol}(\widetilde{SO}(d, 2))} \mathcal{T}_{2}\mathcal{T}_{4}\langle\Omega|[O_{4}, O_{1}][O_{2}, O_{3}]|\Omega\rangle G_{O}, \quad (A.17)$$
$$G_{O} = \frac{\left(\mathcal{T}_{2}\langle0|O_{2}\mathbf{L}[O^{\dagger}]O_{1}|0\rangle^{(a)}\right)^{-1} \left(\mathcal{T}_{4}\langle0|O_{4}\mathbf{L}[O]O_{3}|0\rangle^{(b)}\right)^{-1}}{\langle\mathbf{L}[O]\mathbf{L}[O^{\dagger}]\rangle^{-1}}. \quad (A.18)$$

We have written the formula in slightly different conventions relative to [11]. Firstly, we made the change of variables $x_2 \rightarrow T_2 x_2$ and $x_4 \rightarrow T_4 x_4$, so that the causal relationships become 3 > 4 and 1 > 2 with all other pairs spacelike separated. This choice makes it simpler to apply the Lorentzian two and three-point pairings defined in [11]. In addition, we only wrote the *t*-channel term in the inversion formula. The treatment of the *u*-channel term is analogous. In our notation, *O* is a representation with quantum numbers (Δ, J, λ) .

The object G_O is a conformal block with internal quantum numbers $(J + d - 1, \Delta - d + 1, \lambda)$. In (A.18), we have written it in schematic notation, where the three-point structures in the numerator should be glued using the two-point structure in the denominator. In more precise notation, G_O is defined by

$$G_{O} = \int_{1>x>2} d^{d}x D^{d-2} z A_{(a)}(x_{1}, x_{2}, x, z) \left(\mathcal{T}_{4} \langle 0|O_{4} \mathbf{L}[O]O_{3}|0\rangle^{(b)} \right)^{-1}$$
(A.19)

where the kernel $A_{(a)}$ satisfies

$$\int_{1>x>2} d^d x D^{d-2} z A_{(a)}(x_1, x_2, x, z) \langle \mathbf{L}[O](x, z) \mathbf{L}[O^{\dagger}](x', z') \rangle^{-1} = \left(\mathcal{T}_2 \langle 0|O_2 \mathbf{L}[O^{\dagger}](x', z')O_1|0\rangle^{(a)} \right)^{-1}$$
(A.20)

Recall that dual structures $\langle \cdots \rangle^{-1}$ are defined using the Lorentzian two and threepoint pairings defined in [11]. In [11], the block G_O was defined by specifying its behavior in the OPE limit $x_1 \rightarrow x_2$. The above definition in terms of the integral kernel $A_{(a)}$ is equivalent and more convenient for our purposes.

The light-ray kernel should satisfy

$$\int_{1>x>2^{-}} d^{d}x_{1} d^{d}x_{2} K^{t}_{\Delta,J,\lambda(a)}(x_{1},x_{2};x,z) \langle \Omega | O_{4} O_{1} O_{2} O_{3} | \Omega \rangle = -C^{t}_{ab}(\Delta,J,\lambda) \langle 0 | O_{4} \mathbf{L}[O] O_{3} | 0 \rangle^{(b)}$$
(A.21)

In the expression (A.21), we can replace $\langle \Omega | O_4 O_1 O_2 O_3 | \Omega \rangle$ with a double commutator. The reason is that the kernel $K_{\Delta,J,\lambda(a)}^t$ factors through null integrals of x_1 and x_2 , which annihilate the past and future vacuum, see [11] for details. Furthermore, let us make the change of variables $x_2 \rightarrow \mathcal{T}_2 x_2$ and $x_4 \rightarrow \mathcal{T}_4 x_4$, so that we have

$$\int_{1>x>2} d^d x_1 d^d x_2 (\mathcal{T}_2 K^t_{\Delta, J, \lambda(a)}) \mathcal{T}_2 \mathcal{T}_4 \langle \Omega | [O_4, O_1] [O_2, O_3] | \Omega \rangle = -C^t_{ab} (\Delta, J, \lambda) \mathcal{T}_4 \langle 0 | O_4 \mathbf{L}[O] O_3 | 0 \rangle^{(b)}$$
(A.22)

For brevity, here and in the following, we assume the arguments of $\mathcal{T}_2 K^t_{\Delta,J,\lambda(a)}$ are $(x_1, x_2; x, z)$. The arguments of $\mathbf{L}[O]$ will always be (x, z), and we use the notation $\mathbf{L}[O'^{\dagger}]$ to indicate $\mathbf{L}[O^{\dagger}](x', z')$.

Pairing both sides with the dual of the right-hand structure, we obtain

$$-C_{ab}^{t}(\Delta, J, \lambda) = \int_{\substack{1 > x > 2\\ 3 > 4}} \frac{d^{d}x_{1} d^{d}x_{2} d^{d}x_{3} d^{d}x_{4} d^{d}x D^{d-2}z}{\operatorname{vol} \widetilde{\mathrm{SO}}(d, 2)} (\mathcal{T}_{2} K_{\Delta, J, \lambda(a)}^{t}) \left(\mathcal{T}_{4} \langle 0|O_{4} \mathbf{L}[O]O_{3}|0\rangle^{(b)}\right)^{-1} \times \mathcal{T}_{2} \mathcal{T}_{4} \langle \Omega|[O_{4}, O_{1}][O_{2}, O_{3}]|\Omega\rangle.$$
(A.23)

Comparing with (A.17), (A.19), and (A.20), we conclude

$$\int_{1>x>2} d^d x D^{d-2} z(\mathcal{T}_2 K^t_{\Delta, J, \lambda(a)}) \langle \mathbf{L}[O] \mathbf{L}[O'^{\dagger}] \rangle^{-1} = \frac{1}{2\pi i} \left(\mathcal{T}_2 \langle 0|O_2 \mathbf{L}[O'^{\dagger}]O_1|0\rangle^{(a)} \right)^{-1}.$$
(A.24)

This is essentially the desired result, written in terms of dual structures. To put it in a more conventional form, we must apply the two- and three-point pairings that appear in the definition of the dual structures. Pairing both sides with $\mathcal{T}_2\langle 0|O_2\mathbf{L}[O^{\dagger}](x',z')O_1|0\rangle^{(b)}$, we find

$$\int_{\substack{1>x>2\\x'\approx1,2}} \frac{d^d x_1 d^d x_2 d^d x D^{d-2} z d^d x' D^{d-2} z'}{\operatorname{vol} \widetilde{\operatorname{SO}}(d,2)} (\mathcal{T}_2 K^t_{\Delta,J,\lambda(a)}) \left(\mathcal{T}_2 \langle 0|O_2 \mathbf{L}[O'^{\dagger}]O_1|0\rangle^{(b)}\right) \langle \mathbf{L}[O] \mathbf{L}[O'^{\dagger}]\rangle^{-1}$$
$$= \frac{1}{2\pi i} \delta^{(b)}_{(a)} \tag{A.25}$$

Comparing with the definition of the Lorentzian two-point pairing in [11], finally gives

$$\int_{1>x>2} \frac{d^d x_1 d^d x_2}{(\operatorname{vol}\operatorname{SO}(1,1))^2} (\mathcal{T}_2 K^t_{\Delta,J,\lambda(a)}) \mathcal{T}_2 \langle 0|O_2 \mathbf{L}[O'^{\dagger}]O_1|0\rangle^{(b)} = \frac{1}{2\pi i} \delta^{(b)}_{(a)} \langle \mathbf{L}[O] \mathbf{L}[O'^{\dagger}]\rangle,$$
(A.26)

where the equality is valid if $x' \approx 1, 2$. Finally, after changing variables $x_2 \rightarrow \mathcal{T}_2^{-1}x_2$, we obtain (2.152).

A.4 An alternative derivation for the light-ray OPE formula

In this appendix, we give another derivation for the light-ray OPE formula (2.147). We will first review the derivation for the low transverse spin terms given in [15]. Then we will derive the "higher transverse spin" terms by generalizing the derivation in [15].

A.4.1 Review: derivation of [15]

We first briefly review the proof given in [15]. We are interested in an expansion for

$$L[O_1](x, z_1)L[O_2](x, z_2).$$
 (A.27)

For simplicity, we assume O_1, O_2 are traceless symmetric tensors. Generalization to arbitrary representations will become straightforward after finishing the proof. We will study the matrix element

$$W(z_1, z_2) = \langle \Omega | O_4 \mathbf{L}[O_1](x, z_1) \mathbf{L}[O_2](x, z_2) O_3 | \Omega \rangle,$$
(A.28)

where O_3 , O_4 are some local primary operators. Then we can apply harmonic analysis on the celestial sphere to expand $W(z_1, z_2)$ into partial waves. The result is given by (2.91) and (2.93). The object $W_{\delta,i}(z_1, z_2)$ can be further written as

$$W_{\delta,j}(x,z) = \alpha_{\delta,j} \int D^{d-2} z_1 D^{d-2} z_2 \langle \widetilde{\mathcal{P}}_{\delta_1}^{\dagger}(z_1) \widetilde{\mathcal{P}}_{\delta_2}^{\dagger}(z_2) \mathcal{P}_{\delta,j}(z) \rangle \langle \Omega | O_4 \mathbf{L}[O_1](x,z_1) \mathbf{L}[O_2](x,z_2) O_3 | \Omega \rangle$$

= $\int d^d x_1 d^d x_2 D^{d-2} z_1 D^{d-2} z_2 \mathcal{L}_{\delta,j}(x_1,z_1,x_2,z_2;x,z) \langle \Omega | O_4 O_1(x_1,z_1) O_2(x_2,z_2) O_3 | \Omega \rangle$,
(A.29)

and the kernel $\mathcal{L}_{\delta,j}$ is

$$\mathcal{L}_{\delta,j}(x_1, z_1, x_2, z_2; x, z) = \alpha_{\delta,j} \langle \widetilde{\mathcal{P}}_{\delta_1}^{\dagger}(z_1) \widetilde{\mathcal{P}}_{\delta_2}^{\dagger}(z_2) \mathcal{P}_{\delta,j}(z) \rangle \times \int_{-\infty}^{\infty} d\alpha_1 d\alpha_2 (-\alpha_1)^{-\delta_1 - J_1 - 1} (-\alpha_2)^{-\delta_2 - J_2 - 1} \delta^{(d)} (x - z_1/\alpha_1 - x_1) \, \delta^{(d)} (x - z_2/\alpha_2 - x_2) \,.$$
(A.30)

Note that we have suppressed the Lorentz indices carried by $W_{\delta,j}$ and $\mathcal{L}_{\delta,j}$. They are contracted with the indices carried by $C_{\delta,j}$ in (2.91).

Using (A.29) and the fact that $L[O_i]$ annihilates the vacuum, we have

$$W_{\delta,j}(x,z) = \int d^d x_1 d^d x_2 D^{d-2} z_1 D^{d-2} z_2 \mathcal{L}_{\delta,j}(x_1, z_1, x_2, z_2; x, z) \theta(4 > 1) \theta(2 > 3)$$

$$\times \langle \Omega | [O_4, O_1(x_1, z_1)] [O_2(x_2, z_2), O_3] | \Omega \rangle.$$
(A.31)

By conformal invariance, we also have

$$W_{\delta,j}(x,z) = A_b(\delta,j) \langle 0|O_4 \mathbf{L}[O](x,z)O_3|0\rangle_+^{(b)}.$$
 (A.32)

where O has quantum numbers $(\Delta, J, \lambda) = (\delta + 1, J_1 + J_2 - 1, j)$, (b) is the tensor structure index, and $\langle 0|O_4 \mathbf{L}[O](x, z)O_3|0\rangle^{(b)}_+$ is the continuous-spin structure analytically continued from even spin.¹ In order to proceed, we need to introduce

¹The analytic continuation of $\langle 0|O_4 \mathbf{L}[O](x, z)O_3|0\rangle^{(b)}$ to complex spin has to be done separately for even and odd spin due to our convention of the three-point function, so we have to make a choice in the right-hand side of (A.32). However, this choice does not affect our final result. It will only change how we relate $C^{\pm}(\Delta, J)$ and $\mathbb{O}^{\pm}(\Delta, J)$ in the final step of our derivation. With our choice in (A.32), we need to identify $C^+(\Delta, J) \to \mathbb{O}^+_{\Delta,J}$ and $C^-(\Delta, J) \to -\mathbb{O}^-_{\Delta,J}$.

conformally-invariant pairings for two-point and three-point continuous spin structures in the Lorentzian signature. The pairings are described in detail in appendix E of [11]. For a two-point structure of O in representation (Δ, J, λ) , it can be paired with a two-point structure of O^S in representation $(d-\Delta, 2-d-J, \lambda)$. The two-point pairing is defined by

$$\frac{\langle \langle OO^{\dagger} \rangle, \langle O^{S}O^{S\dagger} \rangle \rangle_{L}}{\operatorname{vol}(\mathrm{SO}(1,1))^{2}} = \int_{x_{1} \approx x_{2}} \frac{d^{d}x_{1}d^{d}x_{2}D^{d-2}z_{1}D^{d-2}z_{2}}{\operatorname{vol}(\widetilde{\mathrm{SO}}(d,2))} \langle O^{a}(x_{1},z_{1})O^{b\dagger}(x_{2},z_{2}) \rangle \langle O^{S}_{b}(x_{2},z_{2})O^{S\dagger}_{a}(x_{1},z_{1}) \rangle \\
= \frac{\langle O^{a}(0,z_{1})O^{b\dagger}(\infty,z_{2}) \rangle \langle O^{S}_{b}(\infty,z_{2})O^{S\dagger}_{a}(0,z_{1}) \rangle}{2^{2d-2}\operatorname{vol}(\mathrm{SO}(d-2))} \frac{1}{(-2z_{1} \cdot z_{2})^{2-d}}, \quad (A.33)$$

where in the last line we use $\widetilde{SO}(d, 2)$ transformations to gauge-fix to $x_1 = 0, x_2 = \infty$, and *a*, *b* are the indices carried by the representation λ . The three-point pairing is defined by

$$\begin{split} &\left(\langle O_{1}O_{2}O\rangle, \langle \widetilde{O}_{1}^{\dagger}\widetilde{O}_{2}^{\dagger}O^{S^{\dagger}}\rangle\right)_{L} \\ &\equiv \int_{\substack{2<1\\x\approx1,2}} \frac{d^{d}x_{1}d^{d}x_{2}d^{d}xD^{d-2}z}{\operatorname{vol}(\widetilde{SO}(d,2))} \langle O_{1}(x_{1})O_{2}(x_{2})O(x,z)\rangle \langle \widetilde{O}_{1}^{\dagger}(x_{1})\widetilde{O}_{2}^{\dagger}(x_{2})O^{S^{\dagger}}(x,z)\rangle \\ &= \frac{1}{2^{2d-2}\operatorname{vol}(SO(d-2))} \frac{\langle O_{1}(e^{0})O_{2}(0)O(\infty,z)\rangle \langle \widetilde{O}_{1}^{\dagger}(e^{0})\widetilde{O}_{2}^{\dagger}(0)O^{S^{\dagger}}(\infty,z)\rangle}{(-2z\cdot e^{0})^{2-d}}. \end{split}$$
(A.34)

Similarly, in the last line, we gauge-fixed $x_1 = e^0, x_2 = 0, x = \infty$.

We can then obtain $A_b(\delta, j)$ by taking a Lorentzian three-point pairing of both sides with a dual structure

$$A_{b}(\delta, j) = \left(\left(\mathcal{T}_{4} \langle 0|O_{4}\mathbf{L}[O](x, z)O_{3}|0\rangle_{+}^{(b)} \right)^{-1}, \mathcal{T}_{4}W_{\delta,j}(x, z) \right)_{L}$$

$$= \int_{\substack{4>1\\2>3}} \frac{d^{d}x_{1}d^{d}x_{2}d^{d}x_{3}d^{d}x_{4}D^{d-2}z_{1}D^{d-2}z_{2}}{\mathrm{vol}\,\widetilde{\mathrm{SO}}(d, 2)} \langle \Omega| \left[O_{4}, O_{1}(x_{1}, z_{1}) \right] \left[O_{2}(x_{2}, z_{2}), O_{3} \right] |\Omega\rangle$$

$$\times \mathcal{T}_{2}^{-1}\mathcal{T}_{4}^{-1} \left[\int d^{d}x D^{d-2}z \left(\mathcal{T}_{4} \langle 0|O_{4}\mathbf{L}[O](x, z)O_{3}|0\rangle_{+}^{(b)} \right)^{-1} \right]$$

$$\times (\mathcal{T}_{2}\mathcal{L}_{\delta,j})(x_{1}, z_{1}, x_{2}, z_{2}; x, z)\theta(4^{+} > 1)\theta(2^{+} > 3) \right],$$
(A.35)

where T_2 , T_4 translate the points x_2 , x_4 to the next Poincaré patch on the Lorentzian cylinder. They are introduced so that the causality configuration in the three-point

Lorentzian pairing is satisfied. The structure $\left(\mathcal{T}_{4}\langle 0|O_{4}\mathbf{L}[O](x,z)O_{3}|0\rangle_{+}^{(b)}\right)^{-1}$ is defined by

$$\left(\left(\mathcal{T}_{4}\langle 0|O_{4}\mathbf{L}[O](x,z)O_{3}|0\rangle_{+}^{(b)}\right)^{-1},\mathcal{T}_{4}\langle 0|O_{4}\mathbf{L}[O](x,z)O_{3}|0\rangle_{+}^{(d)}\right)_{L}=\delta_{(b)}^{(d)}.$$
 (A.36)

The derivation up to this point is true for all transverse spin *j*. In the rest of this section we first finish the derivation assuming $j \leq j_{max}$, and explain what goes wrong when $j > j_{max}$, where j_{max} is the maximal allowed transverse spin in the $O_1 \times O_2$ OPE. We then give the derivation for $j > j_{max}$ in section A.4.3. Note that since we assume O_1, O_2 are traceless symmetric tensors, in what follows we will simply use $j_{\text{max}} = J_1 + J_2$.

In (A.35), the term in the bracket is a conformally-invariant four-point structure, and is an eigenfunction of the quadratic Casimir acting on 1,2 (or 3,4). Therefore it is a linear combination of conformal blocks, and we can study it by taking the OPE limit. In the OPE limit $x_3, x_4 \rightarrow x'$, the 34 three-point structure should be given by a linear operator B_{34O} acting on a two-point function:²

$$\left(\mathcal{T}_4 \langle 0 | O_4 \mathbf{L}[O](x, z) O_3 | 0 \rangle_+^{(b)} \right)^{-1}$$

= $B_{34O}(x_3, x_4, \partial_{x'}, \partial_{z'}) \langle O^F(x', z') O^{F^{\dagger}}(x, z) \rangle,$ (A.37)

where O^F has quantum numbers $(\Delta_F, J_F, j_F) = (J + d - 1, \Delta - d + 1, j)$, where $J = J_1 + J_2 - 1$ and $\Delta = \delta + 1$. Plugging this into (A.35), one can then show that the bracketed term is the conformal block that appears in the Lorentzian inversion formula, and therefore we can relate $A_b(\delta, j)$ to $C^{\pm}(\delta + 1, J_1 + J_2 - 1, j)$. Using (A.32) and the relation between light-ray operators $\mathbb{O}_{\Lambda,J}^{\pm}$ and $C^{\pm}(\Delta, J)$, we obtain that

$$\mathbf{L}[O_{1}](x, z_{1})\mathbf{L}[O_{2}](x, z_{2}) = \sum_{j \leq J_{1}+J_{2}} \int_{\frac{d-2}{2}-i\infty}^{\frac{d-2}{2}+i\infty} \frac{d\delta}{2\pi i} C_{\delta,j}^{(a)}(z_{1}, z_{2}, \partial_{z}) \left(\mathbb{O}_{\delta+1,J_{1}+J_{2}-1,j(a)}^{+}(x, z) - \mathbb{O}_{\delta+1,J_{1}+J_{2}-1,j(a)}^{-}(x, z) \right) + \text{higher transverse spin,}$$
(A.38)

+ higher transverse spin,

where the differential operator $C_{\delta,i}^{(a)}(z_1, z_2, \partial_z)$ can be obtained from a celestial map formula

$$C_{\delta,j}^{(a)}(z_1, z_2, \partial_z) \langle \mathbf{L}[O](\infty, z_2) \mathbf{L}[O^{\dagger}](0, z_1) \rangle = \frac{\langle 0 | \mathbf{L}^+[O_2](\infty, z_2) \mathbf{L}[O^{\dagger}](0, z) \mathbf{L}^-[O_1](\infty, z_1) | 0 \rangle_+^{(a)}}{\operatorname{vol} \operatorname{SO}(1, 1)}$$
(A.39)

273

²For continuous spin, the operator B_{34Q} should be an integral operator. See appendix H of [11].

This result agrees with the first sum in (2.147).

When $j > J_1 + J_2$, the invalid step in this derivation is (A.37). The reason is that the linear operator B_{34O} becomes divergent when $j > J_1 + J_2$. To see this, consider a conformal block

$$C_{34O}(x_3, x_4, \partial_{x'}) \langle O_1(x_1) O_2(x_2) O(x') \rangle.$$
 (A.40)

It has been shown that this conformal block has simple poles in Δ , the scaling dimension of O, due to null descendant states [35, 38].³ Furthermore, the poles come from the differential operator C_{34O} and have the form

$$C_{34O} \sim \frac{N}{\Delta - \Delta_*} C_{34O'} \mathcal{D}, \qquad (A.41)$$

where N is some coefficient, O' is a primary descendant of O, D is a differential operator such that O' = DO (at $\Delta = \Delta_*$), and $C_{34O'}$ is the OPE operator for $O' \in O_3 \times O_4$. The possible pole positions Δ_* are classified in [35] using representation theory of the conformal group. For us, the relevant cases are what are called type I and type II poles in [35]. If the exchanged operator has representation (l_1, l_2, \ldots, l_N) , where l_k is the number of boxes of the k-th row of the Young diagram, then the positions of the type I and type II poles are

$$\Delta_{\mathrm{I}_{k,n}}^{*} = k - l_{k} - n \qquad (n = 1, 2, \dots, l_{k-1} - l_{k})$$

$$\Delta_{\mathrm{II}_{k,n}}^{*} = d + l_{k} - k - n \qquad (n = 1, 2, \dots, l_{k} - l_{k+1}), \qquad (A.42)$$

and we call \mathcal{D} for the type I and type II case $\mathcal{D}_{I_k,n}$ and $\mathcal{D}_{II_k,n}$ respectively.⁴ In (A.37), the operator O^F has scaling dimension $\Delta_F = J + d - 1 = d + J_1 + J_2 - 2$. Therefore, if $j = J_1 + J_2 + n$ for some $n \ge 1$, then $\Delta^*_{II_{k=2},n} = d + j - 2 - n = \Delta_F$. This implies that the linear operator B_{34O} is divergent for all $j > J_1 + J_2$. To fix (A.37), we can separate the pole part and the finite part of B_{34O} near $\Delta_F = \Delta^*_{II_2,n}$:

$$B_{34O} = \frac{1}{\Delta_F - \Delta_{\text{II}_2,n}^*} C_{34O'} \mathcal{D}_{\text{II}_2,n} + B_{34O}^{\text{finite}}, \qquad (A.43)$$

where $n = j - J_1 - J_2$, and $C_{34O'}$ is a new linear operator proportional to the OPE operator of $\mathcal{D}_{\text{II}_2,n}O^F \in O_3 \times O_4$. Plugging the above expression for B_{34O} into (A.37),

³In even d, some of the poles can become double poles when more than one simple poles are at the same position. However, one can explicitly check that for the case we are discussing (type II poles with k = 2), the poles do not overlap with other poles and hence remain simple poles in even $d \ge 4$.

 $^{{}^{4}\}mathcal{D}_{I_{2},n}$ is the differential operator \mathcal{D}_{n} we use extensively in the main text.
we obtain

$$\left(\mathcal{T}_{4} \langle 0 | O_{4} \mathbf{L}[O](x, z) O_{3} | 0 \rangle_{+}^{(b)} \right)^{-1}$$

= $B_{34O}^{\text{finite}}(x_{3}, x_{4}, \partial_{x'}, \partial_{z'}) \langle O^{F}(x', z') O^{F^{\dagger}}(x, z) \rangle$
+ $C_{34O'}(x_{3}, x_{4}, \partial_{x'}, \partial_{z'}) \langle \mathcal{D}_{\text{II}_{2},n} O^{F}(x', z') O^{F^{\dagger}}(x, z) \rangle_{\text{lim}},$ (A.44)

where

$$\langle \mathcal{D}_{\mathrm{II}_{2,n}} O^{F}(x',z') O^{F^{\dagger}}(x,z) \rangle_{\mathrm{lim}} = \lim_{\Delta_{F} \to \Delta^{*}_{\mathrm{II}_{2,n}}} \frac{\langle \mathcal{D}_{\mathrm{II}_{2,n}} O^{F}(x',z') O^{F^{\dagger}}(x,z) \rangle}{\Delta_{F} - \Delta^{*}_{\mathrm{II}_{2,n}}}.$$
 (A.45)

Note that $\langle \mathcal{D}_{\Pi_2,n} O^F(x',z') O^{F^{\dagger}}(x,z) \rangle \sim O(\Delta_F - \Delta_{\Pi_2,n}^*)$ because when $\Delta_F = \Delta_{\Pi_2,n}^*$ it is a two-point function between two primaries with different scaling dimensions and therefore is zero. So the above limit should be finite. This structure is not conformally-invariant, since it does not vanish under the special conformal transformation K_{μ} . However, the result we get by integrating it against $\mathcal{T}_2 \mathcal{L}_{\delta,j}$ is still a conformally-invariant three-point structure. To see this, note that $[K_{\mu}, \mathcal{D}_{\Pi_2,n}] = O(\Delta_2 - \Delta_{\Pi_2,n}^*)$ since $\mathcal{D}_{\Pi_2,n}$ is conformally-invariant at $\Delta_{\Pi_2,n}^*$. So, we can define

$$\mathcal{D}_{\mu} \equiv \lim_{\Delta_F \to \Delta_{\mathrm{II}_{2},n}^*} \frac{[K_{\mu}, \mathcal{D}_{\mathrm{II}_{2},n}]}{\Delta_F - \Delta_{\mathrm{II}_{2},n}^*}.$$
 (A.46)

Then, we have

$$K_{\mu} \int d^{d}x D^{d-2} z \langle \mathcal{D}_{\mathrm{II}_{2},n} O^{F}(x',z') O^{F^{\dagger}}(x,z) \rangle_{\lim} \mathcal{T}_{2} \mathcal{L}_{\delta,j}(x_{1},z_{1},x_{2},z_{2},x,z)$$

=
$$\int d^{d}x D^{d-2} z \langle \mathcal{D}_{\mu} O^{F}(x',z') O^{F^{\dagger}}(x,z) \rangle \mathcal{T}_{2} \mathcal{L}_{\delta,j}(x_{1},z_{1},x_{2},z_{2},x,z).$$
(A.47)

This is simply a derivative of the Lorentzian shadow transform of $\mathcal{T}_2 \mathcal{L}_{\delta,j}$. In [15] it has been shown that the Lorentzian shadow transform of $\mathcal{T}_2 \mathcal{L}_{\delta,j}$ vanishes for $j > J_1 + J_2$, and therefore (A.47) should vanish as well.

The operator $\mathcal{D}_{\text{II}_{2,n}}O^{F}$ has quantum numbers $(J + d - 1 + n, \Delta - d + 1, j - n)$. Thus, the appearance of $C_{34O'}$ in (A.44) suggests that for $j = J_1 + J_2 + n$, $A_b(\delta, j)$ should come from exchanged operators with quantum numbers $(\delta+1, J_1+J_2-1+n, J_1+J_2)$. As we will see briefly, this is indeed the case.

A.4.2 Relation between $\mathcal{D}_{I_2,n}$ and $\mathcal{D}_{II_2,n}$

We now describe an interesting relation between the differential operators $\mathcal{D}_{I_2,n}$ and $\mathcal{D}_{II_2,n}$. This relation will be used later in the derivation for the $j > J_1 + J_2$ case.

First, note that $\mathcal{D}_{I_2,n}$ and $\mathcal{D}_{II_2,n}$ change the quantum numbers in the following way:

$$\mathcal{D}_{I_{2,n}}: (2-j, l_{1}, j-n) \to (2-j+n, l_{1}, j)$$

$$\mathcal{D}_{II_{2,n}}: (d+j-n-2, 2-d-l_{1}, j) \to (d+j-2, 2-d-l_{1}, j-n), \quad (A.48)$$

and their explicit expressions can be chosen to be 5

$$\mathcal{D}_{I_{2,n}} = \frac{(-1)^{n} \Gamma(j-n-1-l_{1})}{\Gamma(j-1-l_{1}) \Gamma(n+1)} \left(\partial_{x} \cdot \mathcal{D}_{z,w}^{0+}\right)^{n}$$
$$\mathcal{D}_{II_{2,n}} = \frac{(-1)^{n} \Gamma(j-n-1-l_{1})}{\Gamma(j-1-l_{1}) \Gamma(n+1)} \left(\partial_{x} \cdot \mathcal{D}_{z,w}^{0-}\right)^{n},$$
(A.49)

where $\mathcal{D}_{z,w}^{0+}$ and $\mathcal{D}_{z,w}^{0-}$ are weight-shifting operators that increase and decrease the transverse spin, respectively [39]. As shown in [45], in a pairing between operators one can always integrate weight-shifting operators by parts. In particular, we have

$$(\mathcal{D}_{z,w}^{0-}|_{J,j})^* = -2j(h-2+j)\mathcal{D}_{z,w}^{0+}|_{2-d-J,j-1},$$
(A.50)

where $h = \frac{d-2}{2}$, and $\mathcal{D}|_{J,j}$ indicates that \mathcal{D} acts on a multiplet with usual spin *J* and transverse spin *j*. Using this relation, one can show that

$$(\mathcal{D}_{\mathrm{II}_{2,n}}|_{d-\Delta-n,2-d-l_{1,j}})^{*} = N_{n,j}\mathcal{D}_{\mathrm{I}_{2,n}}|_{\Delta,l_{1,j-n}},$$
(A.51)

where

$$N_{n,j} = 2^n \frac{\Gamma(j+1)\Gamma(h-1+j)}{\Gamma(j-n+1)\Gamma(h-1+j-n)}.$$
 (A.52)

Note that this relation holds for general Δ , but $\mathcal{D}_{I_2,n}$ and $\mathcal{D}_{II_2,n}$ are conformallyinvariant only when $\Delta = 2 - j$. More explicitly, for an operator O_1 with quantum numbers $(\Delta, l_1, j - n)$ and O_2 with quantum numbers $(d - \Delta - n, 2 - d - l_1, j)$, we have

$$\int d^d x D^{d-2} z O_1(x, z) (\mathcal{D}_{\mathrm{II}_2, n} O_2)(x, z) = N_{n, j} \int d^d x D^{d-2} z (\mathcal{D}_{\mathrm{I}_2, n} O_1)(x, z) O_2(x, z)$$
(A.53)

In other words, $\mathcal{D}_{\mathrm{II}_{2},n}$ and $N_{n,j}\mathcal{D}_{\mathrm{I}_{2},n}$ are adjoint to each other.

We end this section by deriving two relations that will be used later in the derivation for the higher-transverse spin terms. First, consider operators O_1 with quantum

⁵After replacing $j - n \rightarrow j$ and $l_1 \rightarrow 1 - \Delta$, the definition of $\mathcal{D}_{l_2,n}$ agrees with (2.66).

numbers $(\Delta_1, l_1, j - n)$ and O_2 with quantum numbers $(\Delta_2, 2 - d - l_1, j)$, where $\Delta_1 + \Delta_2 = d$. Then by (A.51), we have for general Δ_1, Δ_2

$$\int_{x\approx x'} d^{d}x d^{d}x' D^{d-2}z D^{d-2}z' \langle \mathcal{D}_{\mathrm{II}_{2},n} \mathcal{O}_{2} \mathcal{D}_{\mathrm{II}_{2},n} \mathcal{O}_{2}^{\dagger} \rangle \langle \mathcal{O}_{1} \mathcal{O}_{1}^{\dagger} \rangle$$

= $N_{n,j}^{2} \int_{x\approx x'} d^{d}x d^{d}x' D^{d-2}z D^{d-2}z' \langle \mathcal{O}_{2} \mathcal{O}_{2}^{\dagger} \rangle \langle \mathcal{D}_{\mathrm{I}_{2},n} \mathcal{O}_{1} \mathcal{D}_{\mathrm{I}_{2},n} \mathcal{O}_{1}^{\dagger} \rangle$, (A.54)

where we use the short-hand notation that O_i is at point (x, z) and O'_i is at point (x', z'). Setting $\Delta_1 \to \Delta^*_{I_2,n}$ and $\Delta_2 \to \Delta^*_{II_2,n}$, this equation can be rewritten as

$$\begin{aligned} (\Delta_{2} - \Delta_{\mathrm{II}_{2,n}}^{*}) \int_{x \approx x'} d^{d}x d^{d}x' D^{d-2}z D^{d-2}z' \lim_{\Delta_{2} \to \Delta_{\mathrm{II}_{2,n}}^{*}} \frac{\langle \mathcal{D}_{\mathrm{II}_{2,n}} \mathcal{O}_{2} \mathcal{D}_{\mathrm{II}_{2,n}} \mathcal{O}_{2}'^{\dagger} \rangle}{\Delta_{2} - \Delta_{\mathrm{II}_{2,n}}^{*}} \langle \mathcal{O}_{1} \mathcal{O}_{1}'^{\dagger} \rangle \\ &= N_{n,j}^{2} (\Delta_{1} - \Delta_{\mathrm{I}_{2,n}}^{*}) \int_{x \approx x'} d^{d}x d^{d}x' D^{d-2}z D^{d-2}z' \langle \mathcal{O}_{2} \mathcal{O}_{2}'^{\dagger} \rangle \lim_{\Delta_{1} \to \Delta_{\mathrm{I}_{2,n}}^{*}} \frac{\langle \mathcal{D}_{\mathrm{I}_{2,n}} \mathcal{O}_{1} \mathcal{O}_{1} \mathcal{O}_{1}'^{\dagger} \rangle}{\Delta_{1} - \Delta_{\mathrm{I}_{2,n}}^{*}} \end{aligned}$$
(A.55)

Finally, using $(\Delta_2 - \Delta_{II_2,n}^*) = -(\Delta_1 - \Delta_{I_2,n}^*)$, we can conclude that

$$\left(\lim_{\Delta_{2}\to\Delta_{\mathrm{II}_{2,n}}^{*}}\frac{\langle\mathcal{D}_{\mathrm{II}_{2,n}}O_{2}\mathcal{D}_{\mathrm{II}_{2,n}}O_{2}^{\dagger}\rangle}{\Delta_{2}-\Delta_{\mathrm{II}_{2,n}}^{*}},\langle O_{1}O_{1}^{\dagger}\rangle\right)_{L}=-N_{n,j}^{2}\left(\langle O_{2}O_{2}^{\dagger}\rangle,\lim_{\Delta_{1}\to\Delta_{\mathrm{I}_{2,n}}^{*}}\frac{\langle\mathcal{D}_{\mathrm{I}_{2,n}}O_{1}\mathcal{D}_{\mathrm{I}_{2,n}}O_{1}^{\dagger}\rangle}{\Delta_{1}-\Delta_{\mathrm{I}_{2,n}}^{*}}\right)_{L}$$
(A.56)

The second relation is about the integral

$$\int d^{d}x_{1}d^{d}x_{2}d^{d}x'D^{d-2}z'\langle \mathcal{D}_{\mathrm{II}_{2},n}O^{F}(x',z')O^{F^{\dagger}}(x,z)\rangle F(x_{1},x_{2},x,z)\mathcal{T}_{2}\langle 0|O_{2}\mathbf{L}[O^{\dagger}](x',z')O_{1}|0\rangle_{+}\theta((1>2)$$
(A.57)

where O^F has quantum numbers (Δ_F, J_F, j) , O has $(\Delta, J, j - n)$, and $F(x_1, x_2, x, z)$ is a conformally-invariant kernel that transforms as $\langle \widetilde{O}_1^{\dagger} \widetilde{O}_2^{\dagger} \mathcal{D}_{I_2,n} \mathbf{L}[O] \rangle$. Note that $\Delta_F + n + (1 - J) = d$ in order for the integral to be conformally-invariant. By applying (A.51), we have for general Δ_F

$$\int d^{d}x_{1}d^{d}x_{2}d^{d}x'D^{d-2}z'\langle \mathcal{D}_{\mathrm{II}_{2,n}}O^{F}(x',z')O^{F^{\dagger}}(x,z)\rangle F(x_{1},x_{2},x,z)\mathcal{T}_{2}\langle 0|O_{2}\mathbf{L}[O^{\dagger}](x',z')O_{1}|0\rangle_{+}\theta((1>2)$$

$$= N_{n,j}\int d^{d}x_{1}d^{d}x_{2}d^{d}x'D^{d-2}z'\langle O^{F}(x',z')O^{F^{\dagger}}(x,z)\rangle F(x_{1},x_{2},x,z)$$

$$\times \mathcal{D}_{\mathrm{I}_{2,n}}\left(\mathcal{T}_{2}\langle 0|O_{2}\mathbf{L}[O^{\dagger}](x',z')O_{1}|0\rangle_{+}\theta((1>2)\approx x')\right).$$
(A.58)

Now we set $\Delta_F \to \Delta_{\mathrm{II}_{2,n}}^*$ and $J \to J_I^* = \Delta_{\mathrm{II}_{2,n}}^* - d + 1 + n$, then $\mathcal{D}_{\mathrm{I}_{2,n}}$ and $\mathcal{D}_{\mathrm{II}_{2,n}}$ are conformally-invariant, and the above equation becomes

Moreover, in the right-hand side we can apply (2.209) and finally obtain

$$\int d^{d}x_{1}d^{d}x_{2}d^{d}x'D^{d-2}z' \lim_{\Delta_{F}\to\Delta_{\mathrm{H}_{2},n}^{*}} \frac{\langle \mathcal{D}_{\mathrm{H}_{2},n}\mathcal{O}^{F}(x',z')\mathcal{O}^{F^{\dagger}}(x,z)\rangle}{\Delta_{F}-\Delta_{\mathrm{H}_{2},n}^{*}}F(x_{1},x_{2},x,z) \times \mathcal{T}_{2}\langle 0|\mathcal{O}_{2}\mathbf{L}[\mathcal{O}^{\dagger}](x',z')\mathcal{O}_{1}|0\rangle_{+}\theta((1>2)\approx x')$$

$$= \operatorname{vol}(\mathrm{SO}(1,1))N_{n,j}\int d^{d}x_{1}d^{d}x_{2}d^{d}x'D^{d-2}z'\langle \mathcal{O}^{F}(x',z')\mathcal{O}^{F^{\dagger}}(x,z)\rangle F(x_{1},x_{2},x,z) \times \mathcal{D}_{\mathrm{I}_{2},n}\left(\mathcal{T}_{2}\langle 0|\mathcal{O}_{2}\mathbf{L}[\mathcal{O}^{\dagger}](x',z')\mathcal{O}_{1}|0\rangle_{+}\theta((1>2)\approx x')\right),$$
(A.60)

where we also use the fact that $(\Delta_F - \Delta_{II_2,n}^*) = (J - J_I^*)$.

A.4.3 Derivation for the higher transverse spin case

Now we give the derivation for the higher-transverse spin terms, where $j > J_1 + J_2$. We can follow the same steps for the $j \le J_1 + J_2$ until (A.31), but we should change (A.32) to ⁶

$$W_{\delta,j}(x,z) = A_b(\delta,j) \langle 0|\mathcal{O}_4 \mathcal{D}_{\mathrm{I}_2,n} \mathbf{L}[\mathcal{O}](x,z)\mathcal{O}_3|0\rangle_+^{(b)}, \tag{A.61}$$

where *O* has quantum numbers $(\Delta, J, j) = (\delta + 1, J_1 + J_2 - 1 + n, J_1 + J_2)$ and $n = j - J_1 - J_2$ in order for $\mathcal{D}_{I_2,n}$ to be conformally-invariant. We expect that $A_b(\delta, j)$ is related to the OPE data of $O_1 \times O_2$ OPE. In particular, we want to show that $A_b(\delta, j)$ is proportional to $C^+(\delta+1, J_1+J_2-1+n, J_1+J_2) + C^-(\delta+1, J_1+J_2-1+n, J_1+J_2)$.

⁶Note that this is just a rewriting of (A.32), but in (A.32) the operator O has quantum numbers $(\delta + 1, J_1 + J_2 - 1, j)$. Here we use the relation $\mathbf{L}[O_{\delta+1, J_1+J_2-1, j}] \propto \mathcal{D}_{I_2, n} \mathbf{L}[O_{\delta+1, J_1+J_2-1+n, J_1+J_2}]$ and call the operator $O_{\delta+1, J_1+J_2-1+n, J_1+J_2}$ as O. We hope that this does not cause confusion.

We follow the old derivation and take the Lorentz pairing with a dual structure:

$$\begin{aligned} A_{b}(\delta,j) &= \left(\left(\mathcal{T}_{4} \langle 0|O_{4}\mathcal{D}_{I_{2},n} \mathbf{L}[O](x,z)O_{3}|0\rangle_{+}^{(b)} \right)^{-1}, \mathcal{T}_{4} W_{\delta,j}(x,z) \right)_{L} \\ &= \int_{\substack{4>1\\2>3}} \frac{d^{d}x_{1} d^{d}x_{2} d^{d}x_{3} d^{d}x_{4} D^{d-2}z_{1} D^{d-2}z_{2}}{\text{vol } \widetilde{SO}(d,2)} \langle \Omega| \left[O_{4}, O_{1}(x_{1},z_{1}) \right] \left[O_{2}(x_{2},z_{2}), O_{3} \right] |\Omega \rangle \\ &\times \mathcal{T}_{2}^{-1} \mathcal{T}_{4}^{-1} \left[\int d^{d}x D^{d-2}z \left(\mathcal{T}_{4} \langle 0|O_{4}\mathcal{D}_{I_{2},n} \mathbf{L}[O](x,z)O_{3}|0\rangle_{+}^{(b)} \right)^{-1} \\ &\times (\mathcal{T}_{2}\mathcal{L}_{\delta,j})(x_{1},z_{1},x_{2},z_{2};x,z)\theta(4^{+}>1)\theta(2^{+}>3) \right]. \end{aligned}$$
(A.62)

In the OPE limit where $x_3, x_4 \to x'$, the step functions $\theta(4^+ > 1)\theta(2^+ > 3)$ should become $\theta((1 > 2) \approx x')$. Also, from the discussion in section A.4.1, we know that the three-point function $\left(\mathcal{T}_4 \langle 0|O_4 \mathcal{D}_{I_2,n} \mathbf{L}[O](x,z)O_3|0\rangle^{(b)}\right)^{-1}$ in the OPE limit is given by

$$\left(\mathcal{T}_{4} \langle 0 | O_{4} \mathcal{D}_{\mathrm{I}_{2},n} \mathbf{L}[O](x,z) O_{3} | 0 \rangle_{+}^{(b)} \right)^{-1}$$

$$= B_{34O}^{\mathrm{finite}}(x_{3}, x_{4}, \partial_{x'}, \partial_{z'}) \langle O^{F}(x',z') O^{F^{\dagger}}(x,z) \rangle$$

$$+ C_{34O'}(x_{3}, x_{4}, \partial_{x'}, \partial_{z'}) \langle \mathcal{D}_{\mathrm{II}_{2},n} O^{F}(x',z') O^{F^{\dagger}}(x,z) \rangle_{\mathrm{lim}}.$$
(A.63)

As shown in [15], one gets zero after integrating the finite part $B_{34O}^{\text{finite}} \langle O^F O^{F^{\dagger}} \rangle$ against $\mathcal{T}_2 \mathcal{L}_{\delta,j}$. Therefore, we have

$$\int d^{d}x D^{d-2}z \left(\mathcal{T}_{4} \langle 0 | O_{4} \mathcal{D}_{I_{2,n}} \mathbf{L}[O] O_{3} | 0 \rangle_{+}^{(b)} \right)^{-1} \left(\mathcal{T}_{2} \mathcal{L}_{\delta,j}(x_{1}, z_{1}, x_{2}, z_{2}; x, z) \right) \theta(4^{+} > 1) \theta(2^{+} > 3)$$

$$= C_{34O'}(x_{3}, z_{3}, x_{4}, z_{4}, \partial_{x'}, \partial_{z'}) \lim_{\Delta_{F} \to \Delta_{II_{2,n}}^{*}} \frac{1}{\Delta_{F} - \Delta_{II_{2,n}}^{*}}$$

$$\times \int_{x \approx x'} d^{d}x D^{d-2}z \langle \mathcal{D}_{II_{2,n}} O^{F}(x', z') O^{F^{\dagger}}(x, z) \rangle \mathcal{T}_{2} \mathcal{L}_{\delta,j}(x_{1}, z_{1}, x_{2}, z_{2}; x, z) \theta((1 > 2) \approx x')$$

$$= C_{34O'}(x_{3}, z_{3}, x_{4}, z_{4}, \partial_{x'}, \partial_{z'}) (\mathcal{D}_{II_{2,n}} \mathbf{S}[\mathcal{T}_{2} \mathcal{L}_{\delta,j}])_{\text{lim}}(x_{1}, z_{1}, x_{2}, z_{2}; x', z') \theta((1 > 2) \approx x'),$$
(A.64)

where S represents the Lorentzian shadow transform, and we have defined

$$(\mathcal{D}_{\mathrm{II}_{2},n}\mathbf{S}[\mathcal{T}_{2}\mathcal{L}_{\delta,j}])_{\mathrm{lim}}(x_{1},z_{1},x_{2},z_{2};x',z') = \lim_{\Delta_{F}\to\Delta_{\mathrm{II}_{2},n}^{*}} \frac{1}{\Delta_{F}-\Delta_{\mathrm{II}_{2},n}^{*}} \int_{x\approx x'} d^{d}x D^{d-2}z \langle \mathcal{D}_{\mathrm{II}_{2},n}O^{F}(x',z')O^{F^{\dagger}}(x,z)\rangle \mathcal{T}_{2}\mathcal{L}_{\delta,j}(x_{1},z_{1},x_{2},z_{2};x,z).$$
(A.65)

This is a conformally-invariant three-point function that transforms like $\langle \tilde{O}_1^{\dagger} \tilde{O}_2^{\dagger} \mathcal{D}_{\text{II}_2,n} O^F \rangle$. The expression in (A.64) should then give a conformal block. To compute it, let us first act $\mathcal{D}_{\text{II}_2,n}$ on both sides of (A.63):

$$\mathcal{D}_{\mathrm{II}_{2,n}} \left(\mathcal{T}_{4} \langle 0 | \mathcal{O}_{4} \mathcal{D}_{\mathrm{I}_{2,n}} \mathbf{L}[\mathcal{O}](x, z) \mathcal{O}_{3} | 0 \rangle_{+}^{(b)} \right)^{-1} = C_{34\mathcal{O}'}(x_{3}, x_{4}, \partial_{x'}, \partial_{z'}) \lim_{\Delta_{F} \to \Delta_{\mathrm{II}_{2,n}}^{*}} \frac{\langle \mathcal{D}_{\mathrm{II}_{2,n}} \mathcal{O}^{F}(x', z') \mathcal{D}_{\mathrm{II}_{2,n}} \mathcal{O}^{F^{\dagger}}(x, z) \rangle}{\Delta_{F} - \Delta_{\mathrm{II}_{2,n}}^{*}}.$$
 (A.66)

Using (A.51) and the definition of the three-point dual structure $\left(\mathcal{T}_{4}\langle 0|\mathcal{O}_{4}\mathbf{L}[\mathcal{O}](x,z)\mathcal{O}_{3}|0\rangle^{(b)}\right)^{-1}$, one can find

$$\mathcal{D}_{\mathrm{II}_{2,n}}\left(\mathcal{T}_{4}\langle 0|O_{4}\mathcal{D}_{\mathrm{I}_{2,n}}\mathbf{L}[O](x,z)O_{3}|0\rangle_{+}^{(b)}\right)^{-1} = N_{n,j}\left(\mathcal{T}_{4}\langle 0|O_{4}\mathbf{L}[O](x,z)O_{3}|0\rangle_{+}^{(b)}\right)^{-1}.$$
(A.67)

Combining (A.64), (A.66), and (A.67), we obtain

$$\int d^{d}x D^{d-2}z \left(\mathcal{T}_{4} \langle 0|O_{4}\mathcal{D}_{\mathrm{I}_{2},n} \mathbf{L}[O]O_{3}|0\rangle_{+}^{(b)} \right)^{-1} \left(\mathcal{T}_{2}\mathcal{L}_{\delta,j}(x_{1},z_{1},x_{2},z_{2};x,z) \right)$$

$$= N_{n,j} \frac{\left((\mathcal{D}_{\mathrm{II}_{2},n} \mathbf{S}[\mathcal{T}_{2}\mathcal{L}_{\delta,j}])_{\lim} \theta((1>2)\approx x') \right) \left(\mathcal{T}_{4} \langle 0|O_{4}\mathbf{L}[O]O_{3}|0\rangle_{+}^{(b)} \right)^{-1}}{\lim_{\Delta_{F}\to\Delta_{\mathrm{II}_{2},n}^{*}} \frac{\langle \mathcal{D}_{\mathrm{II}_{2},n}\mathcal{O}^{F}\mathcal{D}_{\mathrm{II}_{2},n}\mathcal{O}^{F^{\dagger}}\rangle}{\Delta_{F}-\Delta_{\mathrm{II}_{2},n}^{*}}}.$$
(A.68)

This is a conformal block whose exchanged operator has quantum numbers $(J_1 + J_2 + d - 2 + n, \delta - d + 2, J_1 + J_2)$. Plugging this into (A.62) and comparing with the Lorentzian inversion formula, we find

$$A_{b}(\delta, j) = -2\pi i \times \frac{1}{2} (C_{ab}^{+}(\delta + 1, J_{1} + J_{2} - 1 + n, J_{1} + J_{2}) + C_{ab}^{-}(\delta + 1, J_{1} + J_{2} - 1 + n, J_{1} + J_{2}))$$

$$\times N_{n,j} \frac{\langle \mathbf{L}[O]\mathbf{L}[O^{\dagger}]\rangle^{-1}}{\lim_{\Delta_{F} \to \Delta_{\Pi_{2,n}}^{*}} \frac{\langle \mathcal{D}_{\Pi_{2,n}}O^{F}\mathcal{D}_{\Pi_{2,n}}O^{F^{\dagger}}\rangle}{\Delta_{F} - \Delta_{\Pi_{2,n}}^{*}}}$$

$$\times \left((\mathcal{D}_{\Pi_{2,n}}\mathbf{S}[\mathcal{T}_{2}\mathcal{L}_{\delta,j}])_{\lim}\theta((1 > 2) \approx x'), \mathcal{T}_{2}\langle 0|O_{2}\mathbf{L}[O^{\dagger}]O_{1}|0\rangle_{+}^{(a)} \right)_{L}.$$
(A.69)

This result can be simplified by integrating $\mathcal{D}_{\mathrm{II}_2,n}$ by parts. For the second line, using (A.56) we have

$$N_{n,j} \frac{\langle \mathbf{L}[\mathcal{O}] \mathbf{L}[\mathcal{O}^{\dagger}] \rangle^{-1}}{\lim_{\Delta_{F} \to \Delta_{\mathrm{II}_{2},n}^{*}} \frac{\langle \mathcal{D}_{\mathrm{II}_{2},n} \mathcal{O}^{F} \mathcal{D}_{\mathrm{II}_{2},n} \mathcal{O}^{F^{\dagger}} \rangle}{\Delta_{F} - \Delta_{\mathrm{II}_{2},n}^{*}}} = N_{n,j}^{-1} \frac{\left(\lim_{J \to J_{1}^{*}} \frac{\langle \mathcal{D}_{\mathrm{I}_{2},n} \mathbf{L}[\mathcal{O}] \mathcal{D}_{\mathrm{I}_{2},n} \mathbf{L}[\mathcal{O}^{\dagger}] \rangle}{J - J_{1}^{*}}\right)^{-1}}{\langle \mathcal{O}^{F} \mathcal{O}^{F^{\dagger}} \rangle},$$
(A.70)

where $J_1^* = J_1 + J_2 - 1 + n$. For the third line, using (A.60) we have

$$\left((\mathcal{D}_{\text{II}_{2},n} \mathbf{S}[\mathcal{T}_{2}\mathcal{L}_{\delta,j}])_{\text{lim}} \theta((1 > 2) \approx x'), \mathcal{T}_{2} \langle 0|O_{2}\mathbf{L}[O^{\dagger}]O_{1}|0\rangle_{+}^{(a)} \right)_{L}$$

$$= \text{vol}(\text{SO}(1,1))N_{n,j} \int_{x\approx x'} \frac{d^{d}x_{1}d^{d}x_{2}d^{d}x'd^{d}xD^{d-2}z_{1}D^{d-2}z_{2}D^{d-2}z'D^{d-2}z}{\text{vol}\,\widetilde{\text{SO}}(d,2)} \langle O^{F}(x',z')O^{F^{\dagger}}(x,z) \rangle$$

$$\times \mathcal{T}_{2}\mathcal{L}_{\delta,j}(x_{1},z_{1},x_{2},z_{2};x,z)\mathcal{D}_{\text{I}_{2},n} \left(\mathcal{T}_{2} \langle 0|O_{2}(x_{2},z_{2})\mathbf{L}[O^{\dagger}](x',z')O_{1}(x_{1},z_{1})|0\rangle_{+}^{(a)}\theta((1 > 2) \approx x') \right) .$$

$$(A.71)$$

Plugging in the definition of $\mathcal{L}_{\delta,j}$ in (A.30), we find that $A_b(\delta, j)$ is given by

$$A_{b}(\delta, j) = -\pi i (C_{ab}^{+}(\delta + 1, J_{1} + J_{2} - 1 + n, J_{1} + J_{2}) + C_{ab}^{-}(\delta + 1, J_{1} + J_{2} - 1 + n, J_{1} + J_{2})) \\ \times \frac{\left(\left(\lim_{J \to J_{1}^{*}} \frac{\langle \mathcal{D}_{I_{2},n} \mathbf{L}[\mathcal{O}] \mathcal{D}_{I_{2},n} \mathbf{L}[\mathcal{O}^{\dagger}] \rangle}{J - J_{1}^{*}} \right)^{-1}, \mathcal{Q}_{\delta,j}^{(a)} \right)_{L}}{\operatorname{vol} \operatorname{SO}(1, 1)},$$
(A.72)

where

$$Q_{\delta,j}^{(a)}(x,z,x',z') = \alpha_{\delta,j} \int D^{d-2} z_1 D^{d-2} z_2 \left\langle \widetilde{\mathcal{P}}_{\delta_1}^{\dagger}(z_1) \widetilde{\mathcal{P}}_{\delta_2}^{\dagger}(z_2) \mathcal{P}_{\delta,j}(z) \right\rangle \\ \times \left\langle 0 | \mathbf{L}^+[\mathcal{O}_2](x,z_2) \mathcal{D}_{\mathbf{I}_2,n} \mathbf{L}[\mathcal{O}^\dagger](x',z') \mathbf{L}^-[\mathcal{O}_1(x,z_1)] | 0 \right\rangle_+^{(a)},$$
(A.73)

where $L^+[O_2]$ indicates that the light transform contour is restricted to 2 > x', and $L^{-}[O_1]$ is restricted to $1 \approx x'$. Finally, comparing the expression of $A_b(\delta, j)$ with the old derivation for the lower transverse spin case, we find that the for higher transverse spin $j > J_1 + J_2$, we have

$$\begin{split} \mathbf{L}[O_{1}](x,z_{1})\mathbf{L}[O_{2}](x,z_{2}) \\ &= \sum_{n=1}^{\infty} (-1)^{n} \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta}{2\pi i} C_{\delta,j}^{(a)}(z_{1},z_{2},\partial_{z}) \\ &\qquad \times \left(\mathcal{D}_{\mathrm{I}_{2},n} \mathbb{O}_{\delta+1,J_{1}+J_{2}-1+n,J_{1}+J_{2}(a)}^{+}(x,z) - \mathcal{D}_{\mathrm{I}_{2},n} \mathbb{O}_{\delta+1,J_{1}+J_{2}-1+n,J_{1}+J_{2}(a)}^{-}(x,z) \right) \\ &\qquad + \text{lower transverse spin,} \end{split}$$
(A.74)

+ lower transverse spin,

where $n = j - J_1 - J_2$. The celestial map formula for $C_{\delta,j}^{(a)}$ is given by

$$C_{\delta,j}^{(a)}(z_{1}, \mathbf{w}_{1}, z_{2}, \mathbf{w}_{2}, \partial_{z_{2}}, \partial_{\mathbf{w}_{2}}) \left(\lim_{J \to J_{1}^{*}} \frac{\langle \mathcal{D}_{I_{2,n}} \mathbf{L}[O](\infty, z_{2}, \mathbf{w}_{2}) \mathcal{D}_{I_{2,n}} \mathbf{L}[O^{\dagger}](0, z_{0}, \mathbf{w}_{0}) \rangle}{J - J_{I^{*}}} \right)$$

= $(-1)^{n} \langle 0 | \mathbf{L}^{+}[O_{2}](\infty, z_{2}, \mathbf{w}_{2}) \mathcal{D}_{I_{2,n}} \mathbf{L}[O^{\dagger}](0, z_{0}, \mathbf{w}_{0}) \mathbf{L}^{-}[O_{1}](\infty, z_{1}, \mathbf{w}_{1}) | 0 \rangle_{+}^{(a)}.$
(A.75)

This result agrees with the second sum in (2.147).

A.5 Kernel of the celestial map

In the main text we have given the celestial map formulas (2.214) and (2.215) which map *d*-dimensional three-point tensor structures to OPE differential operators in (d-2)-dimensional space. The latter are in turn in one-to-one correspondence with three-point tensor structures in (d-2)-dimensional space. In general the space T_d of three-point structures in *d* and the space T_{d-2} of three-point structures in d-2dimensions have different dimensionality. Therefore, the celestial map $T_d \rightarrow T_{d-2}$ in general has non-trivial kernel or cokernel. In this appendix we identify a part $K_0 \subseteq K$ of the kernel $K \subseteq T_d$ of this map and conjecture that it is in fact the entire kernel ($K = K_0$) and that the cokernel is trivial, i.e. that the celestial map is surjective. We give support to this conjecture by matching the dimension of T_d/K_0 with the dimension of T_{d-2} . Finally, we consider the SO(d-2) representations λ that can be generated by the celestial map and show that they cover all the representations appearing in the (d-2)-dimensional OPE.

We begin by identifying K_0 . Let us start with the low transverse spin case (2.200). We are instructed to evaluate the structure from T_d in configuration (2.219) after multiplying by $V_{0,12}$. Since in this configuration $V_{0,12}$ vanishes, the only structures that survive are those which contain $V_{0,12}^{-1}$. Structures with more negative powers of $V_{0,12}$ will be singular and the structures with non-negative powers will vanish. We claim that no structures have more singular power of $V_{0,12}$ and that there is a simple rule for counting those with $V_{0,12}^{-1}$.

To see this, let us label the SO(d-1, 1) irreps of the operators as $\rho_1 = (J_1, \lambda_1), \rho_2 = (J_2, \lambda_2)$ and $\rho = (J, \lambda)$. For large J the number of three-point tensor structures is J-independent. This is because the number of structures is given [11] by the dimension of

$$(\rho_1 \otimes \rho_2 \otimes \rho)^{\mathrm{SO}(d-1)}. \tag{A.76}$$

The dependence on *J* can be exhibited by first computing the SO(d - 1) content of $\rho_1 \otimes \rho_2$ and matching SO(d - 1) irreps there to dual irreps in SO(d - 1) content of ρ . As *J* is increased, all that happens is that new irreps appear in SO(d - 1) decomposition of ρ , but any given irrep appears at most once.⁷ Since $\rho_1 \otimes \rho_2$ contains finitely many SO(d - 1) irreps, at some point the number of matching dual pairs stabilizes.

⁷In what follows we use facts about dimensional reduction of SO(N) irreps. See, e.g., [180] for a review.

As *J* is decreased to sufficiently low values, some structures disappear from this counting. In terms of explicit expressions this happens because the structures at large *J* depend on $V_{0,12}^{J-n}$ for various *n*, and as *J* becomes less than *n* such structures cease to be polynomial in *z* and have to disappear from the above counting. We are interested in the maximal *n* among all structures, which is therefore the same as the value of *J* at which the number of structures stabilizes. As *J* is increased by 1, the SO(*d* – 1) content of ρ is appended by representations with the first row of length *J*. Since the maximal length of the first row of SO(*d* – 1) representations in $\rho_1 \otimes \rho_2$ is $J_1 + J_2$,⁸ it follows that the stable number of representations is achieved starting from at most $J = J_1 + J_2$. This implies that $n \leq J_1 + J_2$. This means that the power of $V_{0,12}$ is no smaller than

$$V_{0,12}^{J-J_1-J_2}.$$
 (A.77)

This finishes the proof of the claim that structures analytically continued to $J = J_1 + J_2 - 1$ have at most $V_{0,12}^{-1}$ singularity.

We thus find that the map (2.214) is well-defined and the structures with $V_{0,12}$ to non-negative powers get mapped to 0. These structures constitute the set K_0 . We conjecture that (2.214) is non-degenerate on the remaining structures in T_d , i.e. those which contain $V_{0,12}^{-1}$. To support this conjecture, let us count these structures and match their number to the number of structures in T_{d-2} .

To do that, note that the above discussion implies that the number of structures with $V_{0,12}^{-1}$ is precisely the difference between the number of polynomial structures for $J = J_1 + J_2$ and the number of polynomial structures for $J = J_1 + J_2 - 1$, i.e. the dimension of⁹

$$(\rho_1 \otimes \rho_2 \otimes (J_1 + J_2, \lambda))^{\text{SO}(d-1)} \ominus (\rho_1 \otimes \rho_2 \otimes (J_1 + J_2 - 1, \lambda))^{\text{SO}(d-1)}$$

= $(\rho_1 \otimes \rho_2 \otimes \text{Res}_{\text{SO}(d-1)}^{\text{SO}(d-1,1)} ((J_1 + J_2, \lambda) \ominus (J_1 + J_2 - 1, \lambda)))^{\text{SO}(d-1)}.$ (A.78)

We can simplify this by noting that

$$\operatorname{Res}_{\operatorname{SO}(d-1)}^{\operatorname{SO}(d-1,1)}((J_1+J_2,\lambda) \ominus (J_1+J_2-1,\lambda)) = \bigoplus_{\tau \in \operatorname{Res}_{\operatorname{SO}(d-2)}^{\operatorname{SO}(d-2)}\lambda} (J_1+J_2,\tau). \quad (A.79)$$

⁸To see this, note that the length of the first row of the Young diagram gives the maximal eigenvalue under any given boost. Since it is J_i for ρ_i , it must be $J_1 + J_2$ for $\rho_1 \otimes \rho_2$. Choosing a boost in (a complexification of) SO(d-1) yields the desired result.

⁹Here we use the formal difference \ominus . To make this precise one can interpret all identities involving it as character identities.

In particular, this only involves SO(d-1) irreps with the first row of length $J_1 + J_2$. As mentioned above, this is the maximal length of the first row in SO(d-1) irreps in $\rho_1 \otimes \rho_2$, and this part of the tensor product $\rho_1 \otimes \rho_2$ simplifies¹⁰

$$\operatorname{Res}_{\operatorname{SO}(d-1,1)}^{\operatorname{SO}(d-1,1)} \rho_1 \otimes \rho_2 = \bigoplus_{\tau \in \operatorname{Res}_{\operatorname{SO}(d-2)}^{\operatorname{SO}(d-2)} \lambda_1 \otimes \lambda_2} (J_1 + J_2, \tau) \oplus \cdots, \qquad (A.80)$$

where the dots represent irreps with shorter first row. By comparing the last two equations we see that

$$(\rho_1 \otimes \rho_2 \otimes (J_1 + J_2, \lambda))^{\mathrm{SO}(d-1)} \ominus (\rho_1 \otimes \rho_2 \otimes (J_1 + J_2 - 1, \lambda))^{\mathrm{SO}(d-1)}$$

= $(\lambda_1 \otimes \lambda_2 \otimes \lambda)^{\mathrm{SO}(d-3)}$ (A.81)

which is the same rule as for counting the structures in T_{d-2} .

We now turn to the higher transverse spin case. In this case the celestial map is applied to three-point structures with representations (J_i, λ_i) and (J, λ) where $J = J_1 + J_2 - 1 + n$ and $\lambda = (J_1 + J_2, \gamma)$. As discussed in section 2.5.3, the structures that contain $V_{0,12}$ to powers higher than the minimal possible $J - J_1 - J_2$ are mapped to zero by the celestial map (2.215). Similarly to the above, we can determine the number of structures which contain $V_{0,12}^{J-J_1-J_2}$ by taking the difference between the number of polynomial structures at $J = J_1 + J_2$ and $J = J_1 + J_2 - 1$. However, since $\lambda = (J_1 + J_2, \gamma)$, there are no polynomial structures for $J = J_1 + J_2 - 1$. Therefore, all tensor structures with such λ contain $V_{0,12}^{J-J_1-J_2}$. In this case K_0 is trivial and we simply would like to match the dimensions of T_d and T_{d-2} .

Since $\lambda = (J_1 + J_2, \gamma)$, the minimal length of the first row in SO(d - 1) irreps contained in (J, λ) with sufficiently large J^{11} is $J_1 + J_2$, in particular

$$\operatorname{Res}_{\operatorname{SO}(d-1)}^{\operatorname{SO}(d-1,1)}(J,\lambda) = \bigoplus_{\tau \in \operatorname{Res}_{\operatorname{SO}(d-2)}^{\operatorname{SO}(d-2)}\lambda} (J_1 + J_2, \tau) + \cdots,$$
(A.82)

where the dots represent irreps with longer first row. Taking into account (A.80), we find

$$(\rho_1 \otimes \rho_2 \otimes (J,\lambda))^{\mathrm{SO}(d-1)} = (\lambda_1 \otimes \lambda_2 \otimes \lambda)^{\mathrm{SO}(d-3)}.$$
 (A.83)

¹⁰One can see this by treating SO(d - 1, 1) irreps as shortened parabolic Verma modules of SO(d - 1, 1), in which case it is analogous to the statement that all primaries of dimension $\Delta_1 + \Delta_2$ one can build out of primaries O_1^a, O_2^b of dimensions Δ_1, Δ_2 are those given by decomposing $O_1^a O_2^b$ into irreducible Lorentz irreps.

¹¹We need the number of analytically-continued tensor structures, which is the same as for very large J.

Since λ_i has first row that is no larger than J_i , it follows that $\lambda_1 \otimes \lambda_2$ only contains SO(d-3) irreps with first row at most of length $J_1 + J_2$. Since the first row of λ is already $J_1 + J_2$, it makes no difference to increase it by n,

$$(\rho_1 \otimes \rho_2 \otimes (J,\lambda))^{\mathrm{SO}(d-1)} = (\lambda_1 \otimes \lambda_2 \otimes \lambda(+n))^{\mathrm{SO}(d-3)}.$$
 (A.84)

This establishes the equality of dimensions of T_d and T_{d-2} .

Finally, let us discuss which representations λ can be generated through celestial map. The low transverse spin terms contain λ 's which are parts of the $\rho = (J, \lambda)$ representations which appear in $O_1 \times O_2$ OPE. We claim that they cover all λ 's that can appear in (d - 2)-dimensional OPE and have first row length at most $J_1 + J_2$. Indeed, using any such λ in (A.81) we find that

$$\dim(\rho_1 \otimes \rho_2 \otimes (J,\lambda))^{\mathrm{SO}(d-1)} \ge \dim(\lambda_1 \otimes \lambda_2 \otimes \lambda)^{\mathrm{SO}(d-3)}$$
(A.85)

for $J = J_1 + J_2$ and thus for any larger J. Since the right-hand side is non-zero whenever λ appears in (d - 2)-dimensional OPE, and the left-hand side being non-zero implies that (J, λ) appears in d-dimensional OPE, we obtain the desired result.

It remains to establish that the higher-transverse spin terms cover all λ 's in (d-2)dimensional OPE with first row of length more than $J_1 + J_2$. Any such λ' can be represented as $\lambda(+n)$ where λ has first row $J_1 + J_2$. This can be then used in (A.84) to conclude that (J, λ) appears in $O_1 \times O_2$ OPE for generic J.

A.6 Spinor Conventions

We use conventions from [50] but switch to mostly plus signature, see appendix A in that paper. A four-dimensional vector $z^{\mu} = (z^0, \vec{z})$ is represented by 2×2 matrix

$$z_{\alpha\dot{\alpha}} = z_{\mu}(\overline{\sigma}^{\mu})_{\alpha\dot{\alpha}} = z^{0}\sigma^{0} + \vec{z}\cdot\vec{\sigma},$$

$$z^{\dot{\alpha}\alpha} = z_{\mu}(\sigma^{\mu})^{\dot{\alpha}\alpha} = -z^{0}\sigma^{0} + \vec{z}\cdot\vec{\sigma}$$
(A.86)

where $\sigma^{\mu} = (1, \vec{\sigma}), \ \overline{\sigma}^{\mu} = (-1, \vec{\sigma}) \text{ and } \vec{\sigma} = (\sigma^1, \sigma^2, \sigma^3).$

Convention for lowering and raising of indices is

$$x^{\beta\beta} = \epsilon^{\beta\alpha} x_{\alpha\dot{\alpha}} \epsilon^{\dot{\alpha}\beta},$$

$$a^{\alpha\beta} = \epsilon^{\alpha\gamma} a_{\gamma\delta} \epsilon^{\delta\beta}.$$
 (A.87)

The Levi-Civita tensors are normalized as follows

$$\epsilon^{12} = \epsilon_{12} = \epsilon_{\dot{1}\dot{2}} = \epsilon^{\dot{1}\dot{2}} = 1 . \tag{A.88}$$

The X matrices are defined as follows

$$(X_{ijk})_{\alpha\dot{\alpha}} = \frac{(x_{ij})_{\alpha\dot{\beta}}}{x_{ij}^2} (x_{jk})^{\dot{\beta}\gamma} \frac{(x_{ki})_{\gamma\dot{\alpha}}}{x_{ik}^2} .$$
(A.89)

Using sigma matrix identities we can simplify

$$X_{ijk} = -\frac{x_{ij} \cdot x_{jk} x_{ki\mu} + x_{jk} \cdot x_{ki} x_{ij\mu} - x_{ij} \cdot x_{ki} x_{jk\mu}}{x_{ij}^2 x_{ki}^2} \overline{\sigma}^{\mu} .$$
(A.90)

A.7 Distributional formulas

In this appendix we formally derive some of the expressions involving distributions that were used in the main text.

A.7.1 Analytic continuation of distributions

In this section we prove (2.130), i.e. we study the analytic continuation of the distribution

$$(s+t)^a s^b t^c \theta(s)\theta(t) \tag{A.91}$$

from the region a, b, c > 0, where it is represented by a locally-integrable function, to general complex a, b, c.

First, let us clarify the idea of analytic continuation of a distribution. Let us restrict to one-variable case with one parameter, i.e. we consider a distribution $g_a(x)$ defined for values of parameter $a \in U \subseteq \mathbb{C}$. Assume that this distribution depends on *a* holomorphically. That is, for any test function f(x) the pairing

$$\langle g_a, f \rangle \equiv \int dx g_a(x) f(x)$$
 (A.92)

is a holomorphic function of *a*. We say that a distribution $h_a(x)$ defined and holomorphic for $a \in V \subseteq \mathbb{C}$ is an analytic continuation of $g_a(x)$ if $U \subseteq V$ and for any test function f(x) we have

$$\langle h_a, f \rangle = \langle g_a, f \rangle \tag{A.93}$$

for all $a \in U$. Similarly, we say that $g_a(x)$ is meromorphic for $a \in U$ if $\langle g_a, f \rangle$ is meromorphic for any test function f and the set of poles is independent of f (of course, some poles may disappear for a specially chosen f). We say that h(x) is the residue of $g_a(x)$ at a_* if $\langle h, f \rangle = \operatorname{res}_{a=a_*}\langle g_a, f \rangle$ for any f, etc. All these notions generalize straightforwardly to the case of several variables and several parameters. Before studying (A.91), let us consider a simpler example,

$$g_a(x) = x^a \theta(x). \tag{A.94}$$

For Re a > -1 this is an integrable function of x, and is holomorphic in a as a distribution. We claim that it admits analytic continuation to $\mathbb{C} \setminus \mathbb{Z}_{<0}$ that is meromorphic in \mathbb{C} with simple poles at negative integer a. As a simple example, consider a test function f(x) that is equal to e^{-x} for $x \ge 0$ and for x < 0 is completed in some smooth way so that it decays quickly at $x \to -\infty$.¹² We have then

$$\langle g_a, f \rangle = \int_0^\infty e^{-x} x^a dx = \Gamma(a+1),$$
 (A.95)

which is indeed meromorphic and has simple poles at negative integer a. To see that this statement holds for more general test functions, recall that

$$(x \pm i\epsilon)^a \tag{A.96}$$

is a distribution that is an entire function of a.¹³ For Re a > -1 we can write the equality of distributions

$$x^{a}\theta(x) = \frac{(x-i\epsilon)^{a}e^{i\pi a} - (x+i\epsilon)^{a}e^{-i\pi a}}{2i\sin\pi a},$$
(A.97)

where the right-hand side is in fact analytic for all $a \in \mathbb{C} \setminus \mathbb{Z}$. We can compute the residue at a = -n as

$$\operatorname{res}_{a=-n} \frac{(x-i\epsilon)^{a} e^{i\pi a} - (x+i\epsilon)^{a} e^{-i\pi a}}{2i\sin \pi a} = \frac{(x-i\epsilon)^{-n} - (x+i\epsilon)^{-n}}{2i\pi}$$
$$= \frac{(-1)^{n-1}}{(n-1)!} \delta^{(n-1)}(x).$$
(A.98)

In particular, this vanishes for $n \le 0$, consistently with $g_a(x)$ being analytic for Re a > -1.

We conclude that $g_a(x) = x^a \theta(x)$ can be analytically continued so that it has simple poles at a = -n with residues

$$\operatorname{res}_{a=-n} x^{a} \theta(x) = \frac{(-1)^{n-1}}{(n-1)!} \delta^{(n-1)}(x).$$
(A.99)

¹²Everywhere in this section we can work with tempered distributions, so that e^{-x} is an appropriate test function for x > 0 (i.e. it is Schwartz).

¹³The notation $(x \pm i\epsilon)^a$ means the boundary value of x^a on real line, approached either from above or below. Since x^a has at most power-law singularity for any a, Vladimirov's theorem [181] ensures that its boundary values are well-defined tempered distributions in x, analytic in the parameter a. See, e.g. [182] for a review of these facts.

Let us perform a simple check with the f(x) defined above. We have

$$\operatorname{res}_{a=-n}\langle g_a, f \rangle = \operatorname{res}_{a=-n}\Gamma(a+1) = \frac{(-1)^{n-1}}{(n-1)!}$$
 (A.100)

and this is indeed equal to

$$\langle \operatorname{res}_{a=-n}g_a, f \rangle = \frac{(-1)^{n-1}}{(n-1)!} \langle \delta^{(n-1)}, f \rangle = \frac{(-1)^{n-1}}{(n-1)!}.$$
 (A.101)

Note that the non-trivial part of this analytic continuation is taking care of the singularity at x = 0, since for $x > x_0 > 0$ the function $x^a \theta(x)$ is locally-integrable for any $a \in \mathbb{C}$.

This simple result for $x^a \theta(x)$ can be extended a more general setup: consider a finite set of k smooth functions $q_i(x)$, $x \in \mathbb{R}^n$ and consider the function

$$g_a(x) = \prod_{i=1}^k \theta(q_i(x)) q_i(x)^{a_i}.$$
 (A.102)

For Re $\alpha_i > 0$ this defines a locally-integrable function. Provided that the functions q_i are in general position (clarified below), we claim that the distribution $g_a(x)$ can be analytically continued to a distribution meromorphic for $a \in \mathbb{C}^k$. To see this, suppose we want to define the analytic continuation in a neighborhood of some point x_0 where r functions $q_{i_1}(x), \dots, q_{i_r}(x)$ vanish. Provided that the matrix of derivatives

$$M_{jl} = \partial_l q_{ij}(x_0) \tag{A.103}$$

has rank r (in particular, $r \le n$), we can use $y_j \equiv q_{i_j}(x)$ as the first r coordinates in a neighborhood U of x_0 . This condition is what we mean by "general position" above. With this choice of coordinates, we simply have

$$g_a(y) = \tilde{g}_a(y) \prod_{j=1}^r y_j^{a_{i_j}} \theta(y_j), \qquad (A.104)$$

where $\tilde{g}_a(y)$ is a smooth function in *U*. Each of $y_j^{a_{i_j}}\theta(y_j)$ can be analytically continued as above, and we can take their product because they are distributions in different variables y_j . We can then finally multiply the resulting distribution by the smooth function $\tilde{g}_a(y)$.

This more general result still does not apply to (A.91) because in (A.91) we have functions

$$q_1(s,t) = s, \quad q_2(s,t) = t, \quad q_3(s,t) = s+t$$
 (A.105)

which are not in general position near s, t = 0. The conclusion about analytic continuation, however, still holds. To establish it, one needs to use a general result about resolution of singularities. We do not reproduce this argument, and instead refer to [183, 184]. Here we simply work out the required resolution in the concrete example of (A.91). Our goal is to define the integral

$$\int ds dt (s+t)^a s^b t^c \theta(s) \theta(t) f(s,t)$$
(A.106)

for test functions f. To do so, we define new coordinates u, v by

$$s = uv, \quad t = u(1 - v).$$
 (A.107)

The region s, t > 0 is mapped one-to-one onto the region u > 0, 0 < v < 1, and the integral (A.106) can be written as

$$\int du dv u^{a+b+c+1} v^b (1-v)^c \theta(u) \theta(v) \theta(1-v) \widetilde{f}(u,v)$$
(A.108)

where

$$\widetilde{f}(u,v) \equiv f(uv,u(1-v)). \tag{A.109}$$

Importantly, $\tilde{f}(u, v)$ is a test function in u, v. Therefore, if we manage to define

$$u^{a+b+c+1}v^b(1-v)^c\theta(u)\theta(v)\theta(1-v)$$
(A.110)

as a distribution meromorphic in a, b, c, we are done. In this case, setting

$$q_1(u, v) = u, \quad q_2(u, v) = v, \quad q_3(u, v) = 1 - v,$$
 (A.111)

we find that q_i are in general position at all points x_0 where at least one function vanishes. (The coordinates u, v "resolve the singularity" that we had at s, t = 0.) In particular, we find poles at

$$a + b + c + 1 = -n \tag{A.112}$$

with residues proportional to $\delta^{(n)}(u)$, poles at

$$b = -n \tag{A.113}$$

with residues proportional to $\delta^{(n)}(v)$, and poles at

$$c = -n \tag{A.114}$$

with residues proportional to $\delta^{(n)}(1-v)$. As explained in the main text, we are interested in the pole near

$$a + b + c + 1 = -1, \tag{A.115}$$

in which case from the analysis above we get

$$u^{a+b+c+1}v^{b}(1-v)^{c}\theta(u)\theta(v)\theta(1-v) \sim \frac{1}{a+b+c+2}\delta(u)v^{b}(1-v)^{c}\theta(v)\theta(1-v).$$
(A.116)

Now we only need to pull this back to s, t coordinates, i.e. evaluate (A.108) with \tilde{f} given by (A.109). We find

$$\int ds dt (s+t)^{a} s^{b} t^{c} \theta(s) \theta(t) f(s,t)$$

$$= \int du dv u^{a+b+c+1} v^{b} (1-v)^{c} \theta(u) \theta(v) \theta(1-v) \widetilde{f}(u,v)$$

$$\sim \frac{1}{a+b+c+2} \int du dv \delta(u) v^{b} (1-v)^{c} \theta(v) \theta(1-v) \widetilde{f}(u,v)$$

$$= \frac{1}{a+b+c+2} \int dv v^{b} (1-v)^{c} \theta(v) \theta(1-v) \widetilde{f}(0,v)$$

$$= \frac{f(0,0)}{a+b+c+2} \int dv v^{b} (1-v)^{c} \theta(v) \theta(1-v)$$

$$= \frac{f(0,0)}{a+b+c+2} \frac{\Gamma(b+1)\Gamma(c+1)}{\Gamma(b+c+2)}.$$
(A.117)

This implies that

$$(s+t)^{a}s^{b}t^{c}\theta(s)\theta(t) \sim \frac{1}{a+b+c+2} \frac{\Gamma(b+1)\Gamma(c+1)}{\Gamma(b+c+2)}\delta(s)\delta(t), \qquad (A.118)$$

as stated in (2.130).

A.7.2 An identity

In this section, we show

$$\int_{-\infty}^{+\infty} \frac{dx}{(xy+1+i\epsilon)^a} = -\frac{2\pi i}{a-1}\delta(y)$$
(A.119)

We define the integral by analytic continuation from the region Re a > 1. Thus, let us evaluate it assuming Re a > 1. Suppose first that y is nonzero. If y > 0, the integrand is holomorphic in the upper-half plane for x. Furthermore, because Re a > 1, it decays sufficiently quickly at infinity that the integration contour can be deformed into the upper half-plane, giving zero. If y < 0, a similar argument shows that the integral can be deformed into the lower half-plane, giving zero. It follows that the distribution (A.119) is supported at y = 0.

Now consider the integral against a test function f(y). Because (A.119) is supported at y = 0, we can restrict the y integral to the range [-1, 1] (or any finite-size interval containing the origin). We furthermore substitute the Taylor expansion of f(y) and integrate term by term:

$$\int_{-1}^{1} dy \int_{-\infty}^{+\infty} \frac{dx}{(xy+1+i\epsilon)^a} \sum_{n=0}^{\infty} \frac{f^{(n)}(0)}{n!} y^n$$
(A.120)

Let us evaluate the term proportional to f(0). Swapping the order of integration, we have

$$f(0) \int_{-\infty}^{+\infty} dx \int_{-1}^{1} dy \frac{1}{(xy+1+i\epsilon)^{a}} = f(0) \int_{-\infty}^{+\infty} dx \frac{(1+i\epsilon+x)^{1-a} - (1+i\epsilon-x)^{1-a}}{(1-a)x}$$
(A.121)

Because the integrand on the right-hand side is holomorphic at x = 0, we can deform the contour so that it moves slightly above the origin (staying below the singularity at $x = 1 + i\epsilon$). We denote this by $\int_{-\infty}^{\infty} \rightarrow \int_{\infty}$. After this deformation, we split the integrand into two terms

$$\frac{f(0)}{1-a} \int_{\infty} dx \left(\frac{(1+i\epsilon+x)^{1-a}}{x} - \frac{(1+i\epsilon-x)^{1-a}}{x} \right)$$
(A.122)

The first term is holomorphic in the positive imaginary direction for x, so we can deform the contour that direction and obtain zero. The second term is holomorphic in the negative imaginary direction for x, except for the pole at x = 0. Thus, we can deform the contour that direction and pick up only the residue at x = 0. We obtain

$$\frac{2\pi i}{1-a}f(0).$$
 (A.123)

Finally, consider the terms in (A.120) proportional to $f^{(n)}(0)$. For these terms, note that

$$\frac{y^n}{(xy+1+i\epsilon)^a} \propto \partial_x^n \frac{1}{(xy+1+i\epsilon)^{a-n}}$$
(A.124)

This is a total derivative in x, and hence integrates to zero. Together with (A.123), this establishes (A.119).

APPENDICES TO CHAPTER 3

B.1 Alternative derivation of the three-point celestial block

In this section we give an alternative derivation of an expression for the three-point celestial block

$$C_{12\mathcal{P}_{\delta,j}}(z_1, z_2, \partial_{z_2}, \partial_{w_2})C_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta'}}(z_2, w_2, z_3, \partial_{z_3})(-2p \cdot z_3)^{-\delta'}.$$
 (B.1)

Lorentz invariance and homogeneity imply that¹

$$C_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta'}}(z_2, z_3, \partial_{z_3})(-2p \cdot z_3)^{-\delta'} = \frac{(-[z_2, w_2] \cdot [z_3, p])^j (-p^2)^{\frac{\delta + \delta_3 + j - \delta'}{2}}}{(-2p \cdot z_2)^{\delta + j} (-2p \cdot z_3)^{\delta_3 + j}} f_{\delta'}'(\zeta_{23}).$$
(B.2)

From the definition of $C_{\mathcal{P}_{\delta,j}3\mathcal{P}_{\delta'}}$, the leading term of $f'_{\delta'}(\zeta_{23})$ should be given by

$$f_{\delta'}'(\zeta_{23}) = \zeta_{23}^{\frac{\delta' - \delta_{-j} - \delta_3}{2}} \left(1 + O(\zeta_{23})\right).$$
(B.3)

Solving the Casimir equation with this boundary condition, we find

$$f_{\delta'}'(\zeta_{23}) = \zeta_{23}^{\frac{\delta'-\delta-j-\delta_3}{2}} {}_2F_1\left(\frac{\delta'-\delta_3+\delta+j}{2}, \frac{\delta'+\delta_3-\delta+j}{2}, \delta'+2-\frac{d}{2}, \zeta_{23}\right).$$
(B.4)

This is the two-point celestial block where one of the external operators has nonzero j, generalizing the result derived in [15].

Now, we must compute

$$C_{12\mathcal{P}_{\delta,j}}(z_1, z_2, \partial_{z_2}, \partial_{w_2}) \left(\frac{(-[z_2, w_2] \cdot [z_3, p])^j (-p^2)^{\frac{\delta + \delta_3 + j - \delta'}{2}}}{(-2p \cdot z_2)^{\delta + j} (-2p \cdot z_3)^{\delta_3 + j}} f_{\delta'}'(\zeta_{23}) \right).$$
(B.5)

Expanding the two-point celestial block $f'_{\delta'}(\zeta_{23})$ given in (B.4), we can rewrite (B.5) as

$$\sum_{n=0}^{\infty} \frac{\left(\frac{\delta'-\delta_{3}+\delta+j}{2}\right)_{n} \left(\frac{\delta'+\delta_{3}-\delta+j}{2}\right)_{n}}{\left(\delta'+2-\frac{d}{2}\right)_{n} n!} (-p^{2})^{n} C_{12} \mathcal{P}_{\delta,j}(z_{1},z_{2},\partial_{z_{2}},\partial_{w_{2}}) \left(\langle \mathcal{P}_{\delta,j}(z_{2},w_{2})\mathcal{P}_{\delta_{3}}(z_{3})\mathcal{P}_{\delta'+2n}(p)\rangle\right),$$
(B.6)

¹We define $[a, b] \cdot [c, d] = 2[(a \cdot c)(b \cdot d) - (a \cdot d)(b \cdot c)]$

where

$$\langle \mathcal{P}_{\delta,j}(z_2, w_2) \mathcal{P}_{\delta_3}(z_3) \mathcal{P}_{\delta'+2n}(p) \rangle = \frac{(-[z_2, w_2] \cdot [z_3, p])^j}{(-2p \cdot z_2)^{\frac{\delta'+\delta+j-\delta_3}{2}+n}(-2p \cdot z_3)^{\frac{\delta'+\delta_3+j-\delta}{2}+n}(-2z_2 \cdot z_3)^{\frac{\delta+j+\delta_3-\delta'}{2}-n}}.$$
(B.7)

Note that the function $\langle \mathcal{P}_{\delta,j}(z_2, w_2) \mathcal{P}_{\delta_3}(z_3) \mathcal{P}_{\delta'+2n}(p) \rangle$ is not a conformally invariant three-point function since *p* is not null, and therefore $C_{12\mathcal{P}_{\delta,j}}$ acting on it does not give a conformal block. To compute the action of $C_{12\mathcal{P}_{\delta,j}}$ on $\langle \mathcal{P}_{\delta,j}(z_2, w_2) \mathcal{P}_{\delta_3}(z_3) \mathcal{P}_{\delta'+2n}(p) \rangle$, we would like to express it as an expansion in conformal three-point functions in the limit that *p* becomes null. To do so, we express *p* as a linear combination of two null vectors $p = z_0 + v$. In the limit $v \to 0$, *p* approaches the point z_0 . We define the null vectors by

$$z_0^{\mu} = p^{\mu} - \frac{p^2}{2p \cdot z_1} z_1^{\mu}, \quad v^{\mu} = \frac{p^2}{2p \cdot z_1} z_1^{\mu},$$
 (B.8)

essentially using z_1 as a reference direction. Consequently, subsequent expressions will not be manifestly symmetric with respect to $1 \leftrightarrow 2$, but the symmetry will be restored in the final answer. Note that z_0 and v satisfy

$$z_0^2 = v^2 = 0, \ p = z_0 + v, \ -p^2 = -2z_0 \cdot v.$$
 (B.9)

The cross ratios can then be written

$$\zeta_{12} = \frac{-2z_2 \cdot v}{-2z_2 \cdot (z_0 + v)}, \quad \zeta_{13} = \frac{-2z_3 \cdot v}{-2z_3 \cdot (z_0 + v)}, \quad \zeta_{23} = \frac{(-2z_2 \cdot z_3)(-2z_0 \cdot v)}{(-2z_2 \cdot (z_0 + v))(-2z_3 \cdot (z_0 + v))}$$
(B.10)

In these variables, the expansion around the collinear limit is an expansion in small *v*. We can write the three-point celestial block as

$$\sum_{n=0}^{\infty} \frac{\left(\frac{\delta'-\delta_{3}+\delta+j}{2}\right)_{n} \left(\frac{\delta'+\delta_{3}-\delta+j}{2}\right)_{n}}{\left(\delta'+2-\frac{d}{2}\right)_{n} n!} (-2z_{0}\cdot v)^{n} C_{12} \mathcal{P}_{\delta,j}(z_{1},z_{2},\partial_{z_{2}},\partial_{w_{2}}) \left(\langle \mathcal{P}_{\delta,j}(z_{2},w_{2})\mathcal{P}_{\delta_{3}}(z_{3})\mathcal{P}_{\delta'+2n}(z_{0}+v)\rangle\right)$$
(B.11)

We find that the function $\langle \mathcal{P}_{\delta,j}(z_2, w_2) \mathcal{P}_{\delta_3}(z_3) \mathcal{P}_{\delta'+2n}(z_0 + v) \rangle$ has the following expansion:

$$\langle \mathcal{P}_{\delta,j}(z_2, w_2) \mathcal{P}_{\delta_3}(z_3) \mathcal{P}_{\delta'+2n}(z_0 + v) \rangle$$

$$= \sum_{m=0}^{\infty} \sum_{k=0}^{m} c_{m,k} (-2v \cdot z_0)^k (v \cdot D_{z_0})^{m-k} \langle \mathcal{P}_{\delta,j}(z_2, w_2) \mathcal{P}_{\delta_3}(z_3) \mathcal{P}_{\delta'+2n+2k}(z_0) \rangle,$$
(B.12)

where D_{z_0} is the Todorov operator acting on z_0 . The coefficients $c_{m,k}$ can be determined by expanding both sides of the above equation order by order in small v:

$$c_{m,k} = \frac{\left(\frac{j+2n+\delta-\delta_{3}+\delta'}{2}\right)_{k} \left(\frac{j+2n-\delta+\delta_{3}+\delta'}{2}\right)_{k}}{(m-k)!k!} \frac{\left(\frac{d}{2} - (2k+1+2n+\delta')\right)\Gamma\left(\frac{d}{2} - (m+k+1+2n+\delta')\right)}{\Gamma\left(\frac{d}{2} - (k+2n+\delta')\right)}.$$
(B.13)

Since the Todorov operator D_{z_0} commutes with $C_{12\mathcal{P}_{\delta,j}}$, the three-point celestial block is therefore given by

$$C_{12\mathcal{P}_{\delta,j}}(z_{1}, z_{2}, \partial_{z_{2}}, \partial_{w_{2}}) \left(\frac{(-[z_{2}, w_{2}] \cdot [z_{3}, p])^{j} (-p^{2})^{\frac{\delta+\delta_{3}+j-\delta'}{2}}}{(-2p \cdot z_{2})^{\delta+j} (-2p \cdot z_{3})^{\delta_{3}+j}} f_{\delta'}'(\zeta_{23}) \right)$$

$$= \sum_{n=0}^{\infty} \sum_{m=0}^{\infty} \sum_{k=0}^{m} \frac{\left(\frac{\delta'-\delta_{3}+\delta+j}{2}\right)_{n} \left(\frac{\delta'+\delta_{3}-\delta+j}{2}\right)_{n}}{\left(\delta'+2-\frac{d}{2}\right)_{n} n!} c_{m,k} (-2v \cdot z_{0})^{n+k} (v \cdot D_{z_{0}})^{m-k} g_{\delta,j}^{(\delta_{1},\delta_{2},\delta_{3},\delta'+2n+2k)}(z_{1}, z_{2}, z_{3}, z_{0})$$

$$= \sum_{n=0}^{\infty} \sum_{m=0}^{\infty} \sum_{k=0}^{m} B_{n,m,k} (-2v \cdot z_{0})^{n+k} (v \cdot D_{z_{0}})^{m-k} g_{\delta,j}^{(\delta_{1},\delta_{2},\delta_{3},\delta'+2n+2k)}(z_{1}, z_{2}, z_{3}, z_{0}),$$
(B.14)

where $g_{\delta,j}^{(\delta_1,\delta_2,\delta_3,\delta'+2n+2k)}(z_1, z_2, z_3, z_0)$ is a usual four-point conformal block, and

$$B_{n,m,k} = \frac{\left(\frac{j+\delta-\delta_{3}+\delta'}{2}\right)_{n+k} \left(\frac{j-\delta+\delta_{3}+\delta'}{2}\right)_{n+k}}{\left(\delta'+2-\frac{d}{2}\right)_{n} n!(m-k)!k!} \frac{\left(\frac{d}{2}-(2k+1+2n+\delta')\right)\Gamma\left(\frac{d}{2}-(m+k+1+2n+\delta')\right)}{\Gamma\left(\frac{d}{2}-(k+2n+\delta')\right)}$$
(B.15)

The expression (B.14) can be further simplified. To do so, we redefine $n' \equiv n+k$, $N \equiv n+m$, which gives

$$\sum_{n=0}^{\infty} \sum_{m=0}^{\infty} \sum_{k=0}^{m} B_{n,m,k} (-2v \cdot z_0)^{n+k} (v \cdot D_{z_0})^{m-k} g_{\delta,j}^{(\delta_1,\delta_2,\delta_3,\delta'+2n+2k)}(z_1, z_2, z_3, z_0)$$

=
$$\sum_{N=0}^{\infty} \sum_{n'=0}^{N} \sum_{k=0}^{n'} B_{n'-k,N-n'+k,k} (-2v \cdot z_0)^{n'} (v \cdot D_{z_0})^{N-n'} g_{\delta,j}^{(\delta_1,\delta_2,\delta_3,\delta'+2n')}(z_1, z_2, z_3, z_0).$$

(B.16)

It turns out that the coefficients $B_{n,m,k}$ given in (B.15) satisfy

$$\sum_{k=0}^{n'} B_{n'-k,N-n'+k,k} = B_{0,N,0}\delta_{n',0}.$$
(B.17)

Therefore, two of the sums in (B.14) collapse, and the three-point celestial block becomes

$$C_{12\mathcal{P}_{\delta,j}}(z_1, z_2, \partial_{z_2}, \partial_{w_2}) \left(\frac{(-[z_2, w_2] \cdot [z_3, p])^j (-p^2)^{\frac{\delta + \delta_3 + j - \delta'}{2}}}{(-2p \cdot z_2)^{\delta + j} (-2p \cdot z_3)^{\delta_3 + j}} f'_{\delta'}(\zeta_{23}) \right)$$

$$= \sum_{N=0}^{\infty} \frac{\Gamma(\frac{d}{2} - 1 - \delta' - N)}{\Gamma(\frac{d}{2} - 1 - \delta')N!} (v \cdot D_{z_0})^N g^{(\delta_1, \delta_2, \delta_3, \delta')}_{\delta, j}(z_1, z_2, z_3, z_0)$$

$$= \Gamma(\delta' + 2 - \frac{d}{2}) (\sqrt{v \cdot D_{z_0}})^{\frac{d}{2} - \delta' - 1} J_{\delta' + 1 - \frac{d}{2}} \left(2\sqrt{v \cdot D_{z_0}} \right) g^{(\delta_1, \delta_2, \delta_3, \delta')}_{\delta, j}(z_1, z_2, z_3, z_0).$$
(B.18)

This expression is more manifestly Lorentz-invariant than (3.33), at the cost of singling out z_1 as special.

It is convenient to write the result in terms of conformally-invariant cross ratios. Before doing so, we relabel the vector $v \rightarrow v'$ to avoid confusion with the cross-ratio v. Factoring out the homogeneity factor in the conformal block, we find (up to linear order in v', for simplicity)

$$C_{12\mathcal{P}_{\delta,j}}(z_1, z_2, \partial_{z_2}, \partial_{w_2}) \left(\frac{(-[z_2, w_2] \cdot [z_3, p])^j (-p^2)^{\frac{\delta + \delta_3 + j - \delta'}{2}}}{(-2p \cdot z_2)^{\delta + j} (-2p \cdot z_3)^{\delta_3 + j}} f_{\delta'}'(\zeta_{23}) \right)$$
$$= \left(1 + \frac{1}{\frac{d}{2} - 2 - \delta'} (v' \cdot D_{z_0}) + \cdots \right) T_{123\delta'}(z_1, z_2, z_3, z_0) g_{\delta,j}^{(123\delta')}(u, v), \quad (B.19)$$

where u, v and $T_{123\delta'}$ are defined in (3.35) and (3.36) (with p replaced by z_0). Finally, we can replace $z_0 \rightarrow p - \frac{p^2}{2p \cdot z_1} z_1, v' \rightarrow \frac{p^2}{2p \cdot z_1} z_1$. After expanding in $(-p^2)$, we obtain (3.34) in the main text.

We can also derive the same result from (3.33). To compute the celestial block using (3.32), we start with the conformal block

$$T_{123\delta'}(z_1, z_2, z_3, z_0)g_{\delta,j}^{(123\delta')}\left(\frac{(-2z_1 \cdot z_2)(-2z_3 \cdot z_0)}{(-2z_1 \cdot z_3)(-2z_2 \cdot z_0)}, \frac{(-2z_2 \cdot z_3)(-2z_1 \cdot z_0)}{(-2z_1 \cdot z_3)(-2z_2 \cdot z_0)}\right),$$
(B.20)

and go to the frame where z_0 is near the origin and z_1, z_2, z_3 are near infinity. Explicitly, let us set

$$z_0 = (1, \vec{y}^2, \vec{y}), \quad z_i = (\lambda^2 \vec{y}_i^2, 1, \lambda \vec{y}_i),$$
 (B.21)

where i = 1, 2, 3, and λ is an expansion parameter. The first two terms, for example, from expanding out the Bessel function in (3.33) then become

$$\left(1 - \frac{\partial_{\vec{y}}^2}{4(\delta' + 2 - \frac{d}{2})}\right) \left(T_{123\delta'}(z_1, z_2, z_3, z_0)g_{\delta,j}^{(123\delta')}\left(\frac{(-2z_1 \cdot z_2)(-2z_3 \cdot z_0)}{(-2z_1 \cdot z_3)(-2z_2 \cdot z_0)}, \frac{(-2z_2 \cdot z_3)(-2z_1 \cdot z_0)}{(-2z_1 \cdot z_3)(-2z_2 \cdot z_0)}\right)\right) \Big|_{\vec{y}_0 \to 0}$$

$$= T_{123\delta'}(z_1, z_2, z_3, p) \left(g_{\delta,j}^{(123\delta')}(u, v) + \frac{\zeta_{13}}{d - 4 - 2\delta'}\mathcal{D}_{u,v}^{(1)}g_{\delta,j}^{(123\delta')}(u, v)\right) \Big|_{\substack{p \to (1,1,0) \\ z_i = (\lambda^2 \vec{y}_i^2, 1, \lambda \vec{y}_i)}} + O(\lambda^4),$$
(B.22)

which agrees with (3.34). More generally, when acting on the appropriate class of functions, $v' \cdot D_{z_0} = \partial_{\vec{v}}^2/4$, so (3.33) and (B.18) are equivalent.

B.2 $\langle R_{\delta,j}^{(1)} \rangle$ and $\langle R_{\delta,j}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ from direct decomposition In this appendix, we give the coefficients $\langle R_{\delta,j}^{(1)} \rangle$ and $\langle R_{\delta,j}^{(0)} \gamma_{\delta,j}^{(1)} \rangle$ up to $\delta = 12$ for N = 4 SYM, the QCD gluon jet, and the QCD quark jet obtained by expanding the collinear EEEC order-by-order in small r, as in section 3.3.2.

B.2.1 $\mathcal{N} = 4$ SYM

$$\langle R_{10,0}^{(1)} \rangle = \frac{141301}{352800} - \frac{2\pi^2}{49}, \qquad \langle R_{10,2}^{(1)} \rangle = \frac{107129}{529200} - \frac{19\pi^2}{1260}, \\ \langle R_{10,4}^{(1)} \rangle = \frac{33394601}{85377600} - \frac{5\pi^2}{132}, \qquad \langle R_{10,6}^{(1)} \rangle = \frac{189283}{1801800} - \frac{3\pi^2}{286}, \\ \langle R_{12,0}^{(1)} \rangle = \frac{84401}{3175200} - \frac{\pi^2}{378}, \qquad \langle R_{12,2}^{(1)} \rangle = \frac{547098707}{5378788800} - \frac{9\pi^2}{1232}, \\ \langle R_{12,4}^{(1)} \rangle = \frac{2674437767}{6412281000} - \frac{16\pi^2}{6435}, \qquad \langle R_{12,6}^{(1)} \rangle = \frac{1220098669}{19675656000} - \frac{7\pi^2}{1144}, \\ \langle R_{12,8}^{(1)} \rangle = \frac{1030567}{68612544} - \frac{\pi^2}{663}, \\ \langle R_{10,0}^{(0)} \gamma_{10,0}^{(1)} \rangle = \frac{89}{210}, \qquad \langle R_{10,2}^{(0)} \gamma_{10,2}^{(1)} \rangle = \frac{8}{105}, \qquad \langle R_{10,4}^{(0)} \gamma_{10,4}^{(1)} \rangle = -\frac{1}{154}, \\ \langle R_{12,0}^{(0)} \gamma_{12,0}^{(1)} \rangle = \frac{253}{630}, \qquad \langle R_{12,2}^{(0)} \gamma_{12,2}^{(1)} \rangle = \frac{28247}{194040}, \qquad \langle R_{12,4}^{(0)} \gamma_{12,4}^{(1)} \rangle = \frac{893}{90090}, \\ \langle R_{12,6}^{(0)} \gamma_{12,6}^{(1)} \rangle = -\frac{1}{1560}. \qquad (B.23)$$

B.2.2 QCD

For the gluon jet, we find

$$\begin{split} & \langle R_{8,0}^{(1)g} \rangle = \frac{-2 \left(5688900 \pi^2 - 563610307 \right) C_A^2 + 2 \left(3969000 \pi^2 - 399089339 \right) C_{Anf} T_F - 3108375 C_{Fnf} T_F }{317520000} \right) \\ & \langle R_{8,2}^{(1)g} \rangle = \frac{C_A \left(\left(241456351 - 24418800 \pi^2 \right) C_A + 2 \left(11020710 \pi^2 - 110698537 \right) n_f T_F \right)}{50803200} \right) \\ & \langle R_{8,4}^{(1)g} \rangle = \frac{\left(1713863 - 173600 \pi^2 \right) C_A^2 + 2 \left(107800 \pi^2 - 1063963 \right) C_{Anf} T_F + 120 C_{Fnf} T_F \right)}{1303948800} \\ & \langle R_{10,0}^{(1)g} \rangle = \frac{\left(104601961181 - 10560904200 \pi^2 \right) C_A^2 + 2 \left(251143200 \pi^2 - 2512589293 \right) C_{Anf} T_F + 7121499 C_{Fnf} T_F \right)}{1303948800} \\ & \langle R_{10,2}^{(1)g} \rangle = \frac{\left(104601961181 - 10560904200 \pi^2 \right) C_A^2 + 2 \left(251143200 \pi^2 - 2512589293 \right) C_{Anf} T_F + 7121499 C_{Fnf} T_F \right)}{23051952000} \\ & - \frac{389 C_{Fnf} T_F}{23051952000} + \frac{\left(13728260700 \pi^2 - 136102341667 \right) C_{Anf} T_F \right)}{23051952000} \\ & - \frac{389 C_{Fnf} T_F}{1411200} , \\ & \langle R_{10,4}^{(1)g} \rangle = \frac{\left(4(51371371326 - 363812248800 \pi^2 \right) C_A^2 + 14 \left(333659400 \pi^2 - 3323123011 \right) n_f T_F \right)}{4098124800} \\ & \langle R_{10,6}^{(1)g} \rangle = \frac{\left(4613713731326 - 363812248800 \pi^2 \right) C_A^2 }{1558311955200} + \frac{\left(254307853800 \pi^2 - 2498045171789 \right) C_{Anf} T_F }{779155977600} \\ & + \frac{788981 C_{Fnf} T_F}{13836800} , \\ & \langle R_{12,2}^{(1)g} \rangle = -\frac{\left(885896411400 \pi^2 - 9950914029409 \right) C_A^2}{2727045921600} + \frac{\left(139361777100 \pi^2 - 1376830794827 \right) C_{Anf} T_F }{283329446400} \\ & + \frac{9197 C_{Fnf} T_F}{1288983292000} + \frac{\left(655611356400 \pi^2 - 6475204191751 \right) C_{Anf} T_F}{865728864000} \\ & - \frac{86819 C_{Fnf} T_F}{4122518400} , \\ & \langle R_{12,4}^{(1)g} \rangle = -\frac{\left(56981204280 \pi^2 - 577018051339 \right) C_A^2}{129859329000} + \frac{\left(655611356400 \pi^2 - 6475204191751 \right) C_{Anf} T_F}{865728864000} \\ & - \frac{86819 C_{Fnf} T_F}{4122518400} , \\ & \langle R_{12,6}^{(1)g} \rangle = \frac{\left(3609756605 - 365743840 \pi^2 - 3679625658343 \right) n_f T_F - 7 \left(91148557500 \pi^2 - 899626801693 \right) C_A }{865728864000} \\ & - \frac{86819 C_{Fnf} T_F}{914833920} , \\ & \langle R_{12,8}^{(1)g} \rangle = \frac{\left(3609756605 - 365743840 \pi^2 \right) C_A^2 + 16 \left(41815620 \pi^2 - 412703677 \right) C_{Anf} T_F + 2352 C_{Fnf} T_F$$

and

$$\langle R_{8,0}^{(0)g} \gamma_{8,0}^{(1)} \rangle = \frac{-62C_A^2 - 110C_A n_f T_F - 135C_F n_f T_F}{5040}, \qquad \langle R_{8,2}^{(0)g} \gamma_{8,2}^{(1)} \rangle = -\frac{C_A (73C_A + 16n_f T_F)}{20160}, \\ \langle R_{10,0}^{(0)g} \gamma_{10,0}^{(1)} \rangle = \frac{1986C_A^2 - 6740C_A n_f T_F - 1407C_F n_f T_F}{47040}, \qquad \langle R_{10,2}^{(0)g} \gamma_{10,2}^{(1)} \rangle = \frac{5743C_A^2 - 11981C_A n_f T_F - 594C_F n_f T_F}{332640}, \qquad \langle R_{10,4}^{(0)g} \gamma_{10,4}^{(1)} \rangle = -\frac{C_A (39C_A + 7n_f T_F)}{73920}, \\ \langle R_{12,0}^{(0)g} \gamma_{12,0}^{(1)} \rangle = \frac{491846C_A^2 - 1164457C_A n_f T_F - 102141C_F n_f T_F}{6486480}, \\ \langle R_{12,2}^{(0)g} \gamma_{12,2}^{(1)} \rangle = \frac{999298C_A^2 - 1763153C_A n_f T_F - 34398C_F n_f T_F}{24216192}, \\ \langle R_{12,4}^{(0)g} \gamma_{12,4}^{(1)} \rangle = \frac{3197C_A^2 - 6135C_A n_f T_F - 126C_F n_f T_F}{720720}, \qquad \langle R_{12,6}^{(0)g} \gamma_{12,6}^{(1)} \rangle = -\frac{C_A (37C_A + 6n_f T_F)}{686400}. \\ \end{cases}$$

For the quark jet, we find

$$\begin{split} \langle \mathcal{R}_{8,0}^{(1)q} \rangle &= \frac{(27930000\pi^2 - 275120246)C_A C_F}{35280000} - \frac{(200976\pi^2 - 2041751)C_F^2}{282240} + \frac{(201264317 - 20580000\pi^2)C_F n_f T_F}{35280000}, \\ \langle \mathcal{R}_{8,2}^{(1)q} \rangle &= \frac{(10224900\pi^2 - 100838071)C_A C_F}{16934400} - \frac{(1774500\pi^2 - 17530127)C_F^2}{2822400} + \frac{(39243247 - 3981600\pi^2)C_F n_f T_F}{8467200}, \\ \langle \mathcal{R}_{8,4}^{(1)q} \rangle &= \frac{C_F (7 (15000\pi^2 - 148003)C_A + (1685981 - 170800\pi^2)C_F)}{1792929600} - \frac{(21218400\pi^2 - 209515081)C_F^2}{39513600} \\ &+ \frac{(168438023821 - 17142602400\pi^2)C_F n_f T_F}{14343436800}, \\ \langle \mathcal{R}_{10,2}^{(1)q} \rangle &= \frac{(1387767150\pi^2 - 16812815347)C_A C_F}{15367968000} - \frac{(64499400\pi^2 - 636790073)C_F^2}{127008000} \\ &+ \frac{(2801569019 - 284592000\pi^2)C_F n_f T_F}{13367661000}, \\ \langle \mathcal{R}_{10,4}^{(1)q} \rangle &= \frac{(6747741000\pi^2 - 66583998913)C_A C_F}{12294374400} + \frac{(46333633219 - 4693827600\pi^2)C_F^2}{12294374400} \\ &- \frac{(999600\pi^2 - 9863251)C_F n_f T_F}{12394374400}, \\ \langle \mathcal{R}_{10,6}^{(1)q} \rangle &= \frac{(6747741000\pi^2 - 6583998913)C_A C_F}{12394374400} - \frac{(70232400\pi^2 - 7516807777)C_F^2}{12294374400} \\ &+ \frac{(2604233505507 - 26177918400\pi^2)C_F n_f T_F}{1330560}, \\ \langle \mathcal{R}_{12,0}^{(1)q} \rangle &= \frac{(112390544366100\pi^2 - 118604851318477)C_A C_F}{11307996800} - \frac{(13487161000\pi^2 - 193326296189)C_F^2}{86060620800} \\ &+ \frac{(2604233505507 - 26177918400\pi^2)C_F n_f T_F}{1904605506800}, \\ \langle \mathcal{R}_{12,4}^{(1)q} \rangle &= \frac{(31216440026800\pi^2 - 31809587301651)C_A C_F}{19437318400} - \frac{(518049932400\pi^2 - 5194727659819)C_F^2}{1943731840} \\ &+ \frac{(48851703227 - 49855085280\pi^2)C_F n_f T_F}{19437318400}, \\ \langle \mathcal{R}_{12,6}^{(1)q} \rangle &= -\frac{(31815016577491 - 322378056000\pi^2)C_F n_f T_F}{10387746368000} - \frac{(177924146400\pi^2 - 1756123265947)C_F^2}{1484106624000} \\ &+ \frac{(488517802327 - 49855085280\pi^2)C_F n_f T_F}{10387746368000} - \frac{(177924146400\pi^2 - 1756123265947)C_F^2}{1484106624000} \\ &- \frac{(9737280\pi^2 - 909763)C_F n_f T_F}{17297280}, \\ \langle \mathcal{R}_{12,8}^{(1)q} \rangle &= -\frac{(3(11826)68577491 - 322378056000\pi^2)C_F n_f T_F}{17297280}, \\ \langle \mathcal{R}_{12,8}^{(1)q} \rangle = \frac{C_F((469964880\pi^2 - 4638418021)C_A + 14(4303513889 - 436035$$

and

$$\langle R_{8,0}^{(0)q} \gamma_{8,0}^{(1)} \rangle = -\frac{C_F(24C_A + 4C_F + 67n_f T_F)}{1680}, \qquad \langle R_{8,0}^{(0)q} \gamma_{8,2}^{(1)} \rangle = \frac{C_F(13C_A - 40C_F)}{6720}, \\ \langle R_{10,0}^{(0)q} \gamma_{10,0}^{(1)} \rangle = -\frac{C_F(-23799C_A + 3652C_F + 86219n_f T_F)}{517440}, \qquad \langle R_{10,2}^{(0)q} \gamma_{10,2}^{(1)} \rangle = -\frac{C_F(-4825C_A + 682C_F + 9012n_f T_F)}{221760}, \qquad \langle R_{10,4}^{(0)q} \gamma_{10,4}^{(1)} \rangle = \frac{C_F(13C_A - 368C_F)}{443520}, \\ \langle R_{12,0}^{(0)q} \gamma_{12,0}^{(1)} \rangle = -\frac{C_F(-343735C_A + 5382C_F + 865310n_f T_F)}{4324320}, \\ \langle R_{12,2}^{(0)q} \gamma_{12,2}^{(1)} \rangle = -\frac{C_F(-38990713C_A + 463606C_F + 69682564n_f T_F)}{887927040}, \\ \langle R_{12,4}^{(0)q} \gamma_{12,4}^{(1)} \rangle = -\frac{C_F(-7296C_A + 655C_F + 13016n_f T_F)}{1441440}, \qquad \langle R_{12,6}^{(0)q} \gamma_{12,6}^{(1)} \rangle = \frac{C_F(13C_A - 43C_F)}{514800}. \\ (B.27)$$

B.3 More details on the Lorentzian inversion formula calculation

In this section, we compute in full detail the coefficients $\langle R_{j+4,j}^{(1)} \rangle$ and $\langle R_{j+\tau_c,j;\delta'_*}^{(0)} \gamma_{j+\tau_c,j}^{(1)} \rangle$ for $\mathcal{N} = 4$ SYM using the Lorentzian inversion formula. We compute $\langle R_{j+4,j}^{(0)} \rangle$ in the first subsection, and then we compute $\langle R_{j+\tau_c,j;\delta'_*}^{(0)} \gamma_{j+\tau_c,j}^{(1)} \rangle$ in the next subsection.

B.3.1 $\langle R_{j+4,j}^{(1)} \rangle$ From (3.94), we have

$$\langle R_{j+4,j}^{(1)} \rangle = 2\kappa_{2j+6} \int_0^1 \frac{d\overline{z}}{\overline{z}^2} k_{2j+6}^{0,1}(\overline{z}) d\text{Disc}_t [\mathcal{G}^{(1)}(z,\overline{z})]|_{z^2}.$$
 (B.28)

Therefore, we first need $\mathcal{G}^{(1)}(z,\overline{z})|_{z^2}$ in order to compute $\langle R_{j+4,j}^{(1)} \rangle$. We find

$$\mathcal{G}^{(1)}(z,\overline{z})|_{z^2} = \frac{1}{4(1-\overline{z})^2} \left(2\overline{z}^4(\overline{z}-2)\zeta_2 - \overline{z}(1-\overline{z})\log(1-\overline{z})(4+2\overline{z}(\overline{z}-2)+\overline{z}(1-\overline{z})^2\log(1-\overline{z})) + 2\overline{z}^3(3-3\overline{z}+\overline{z}^2)\operatorname{Li}_2(\overline{z}) \right).$$
(B.29)

Using the double discontinuities

$$d\text{Disc}_{t}[(1-\overline{z})^{\alpha}] = -2\sin^{2}(\pi\alpha)(1-\overline{z})^{\alpha},$$

$$d\text{Disc}_{t}[(1-\overline{z})^{\alpha}\log(1-\overline{z})] = -2\sin^{2}(\pi\alpha)(1-\overline{z})^{\alpha}\log(1-\overline{z}) - 2\pi\sin(2\pi\alpha)(1-\overline{z})^{\alpha},$$

$$d\text{Disc}_{t}[(1-\overline{z})^{\alpha}\text{Li}_{2}(\overline{z})] = -2\sin^{2}(\pi\alpha)(1-\overline{z})^{\alpha}\text{Li}_{2}(\overline{z}) + 2\pi\sin(2\pi\alpha)(1-\overline{z})^{\alpha}\log(\overline{z}),$$

$$d\text{Disc}_{t}[(1-\overline{z})^{\alpha}\log^{2}(1-\overline{z})] = -2\sin^{2}(\pi\alpha)(1-\overline{z})^{\alpha}\log^{2}(1-\overline{z}) - 4\pi\sin(2\pi\alpha)(1-\overline{z})^{\alpha}\log(1-\overline{z}) - 4\pi^{2}\cos(2\pi\alpha)(1-\overline{z})^{\alpha},$$

$$(B.30)$$

we obtain (after introducing a small regulator ϵ)

$$d\text{Disc}_{t}[\mathcal{G}^{(1)}(z,\overline{z})]|_{z^{2}}$$

$$= -2\sin^{2}(\pi\epsilon)\mathcal{G}^{(1)}(z,\overline{z})|_{z^{2}} + 2\pi\sin(2\pi\epsilon) \times \left(-\frac{\overline{z}^{1+\epsilon}}{4(1-\overline{z})^{1+\epsilon}}\right)(4+2\overline{z}^{2}-4\overline{z})$$

$$-2\pi\sin(2\pi\epsilon) \times \frac{\overline{z}^{3+\epsilon}}{2(1-\overline{z})^{2+\epsilon}}(3-3\overline{z}+\overline{z}^{2})\log(\overline{z})$$

$$+4\pi\sin(2\pi\epsilon) \times \left(-\frac{\overline{z}^{2+\epsilon}}{4(1-\overline{z})^{-1+\epsilon}}\right)\log(1-\overline{z})$$

$$-4\pi^{2}\cos(2\pi\epsilon) \times \left(-\frac{\overline{z}^{2+\epsilon}}{(1-\overline{z})^{-1+\epsilon}}\right).$$
(B.31)

We can then plug (B.31) into (B.28) and integrate each term over \overline{z} . After taking the $\epsilon \to 0$ limit, we obtain (3.102).

B.3.2 $\langle R_{j+\tau_c,j;\delta'_*}^{(0)} \gamma_{j+\tau_c,j}^{(1)} \rangle$

To do the calculation for general twist operators, we will need to compute dDisc of more complicated functions than in the leading twist case. It is convenient to first define the linear functional

$$I_{\beta}[f] \equiv \lim_{\epsilon \to 0} \kappa_{\beta} \int_{0}^{1} \frac{d\overline{z}}{\overline{z}^{2}} k_{\beta}^{0,1}(\overline{z}) d\text{Disc}_{t} \left[\frac{\overline{z}^{\epsilon}}{(1-\overline{z})^{\epsilon}} f(\overline{z}) \right], \quad (B.32)$$

where κ_{β} is defined in (3.85), and

$$k_{\beta}^{0,1}(\bar{z}) = \bar{z}^{\frac{\beta}{2}} {}_{2}F_{1}\left(\frac{\beta}{2}, \frac{\beta}{2} + 1, \beta; \bar{z}\right).$$
(B.33)

To compute anomalous dimensions, we will need to apply this functional to functions with power-law divergences or logarithmic divergences near $\overline{z} = 1$. Using (B.30), we can obtain

$$I_{\beta}\left[\frac{\overline{z}^{b}}{(1-\overline{z})^{a}}\right] = \frac{\Gamma(\frac{\beta}{2})^{2}\Gamma(\frac{\beta}{2}+1)\Gamma(-1+b+\frac{\beta}{2})}{\Gamma(a)\Gamma(a+1)\Gamma(\beta-1)\Gamma(1-a+\frac{\beta}{2})\Gamma(b-a+\frac{\beta}{2})} \times {}_{3}F_{2}\left(\begin{array}{cc} 1-a, & b-a, & 1+\frac{\beta}{2} \\ 1-a+\frac{\beta}{2}, & b-a+\frac{\beta}{2} \end{array};1\right),$$
(B.34)

where *a* should be a positive integer (for zero or negative *a* the right hand side is zero). Also, for logarithmic divergences we can take the derivative of (B.34) with respect to *a*. For a = 0, we have

$$I_{\beta}\left[\overline{z}^{b}\log(1-\overline{z})\right] = -\frac{\Gamma(\frac{\beta}{2})\Gamma(\frac{\beta}{2}-1)}{\Gamma(\beta-1)},$$
(B.35)

and for a = 1

where S_1 is the harmonic number, and

$${}_{3}F_{2}^{(0,0,1),(0,0)}\left(\begin{array}{cc}a_{1}, a_{2}, a_{3}\\b_{1}, b_{2}\end{array}; 1\right) = \partial_{a_{3}}\left({}_{3}F_{2}\left(\begin{array}{cc}a_{1}, a_{2}, a_{3}\\b_{1}, b_{2}\end{array}; 1\right)\right).$$
(B.37)

(B.34), (B.35), and (B.36) are the main results we need for the functional I_{β} in order to do the Lorentzian inversion formula calculation for general twists. In particular, for (B.36), it suffices to consider the b = 1 case, which is given by

$$I_{\beta}\left[\frac{\overline{z}}{(1-\overline{z})}\log(1-\overline{z})\right] = \frac{\Gamma(\frac{\beta}{2})\Gamma(\frac{\beta}{2}+1)\left(1-2S_{1}(\frac{\beta}{2}-1)\right)}{\Gamma(\beta-1)}.$$
 (B.38)

By (3.94), the anomalous dimension coefficient $\langle R_{j+\tau_c,j}^{(0)} \gamma_{j+\tau_c,j}^{(1)} \rangle$ with celestial twist τ_c should be given by

$$\langle R_{j+\tau_c,j}^{(0)} \gamma_{j+\tau_c,j}^{(1)} \rangle = 4\kappa_{2j+\tau_c} \left(z^{\frac{\tau_c}{2}-1} \int_0^1 \frac{d\overline{z}}{\overline{z}^2} g_{j+1,j+\tau_c-1}^{\widetilde{\delta}_i}(z,\overline{z}) \mathrm{dDisc}_t [\mathcal{G}(z,\overline{z})] \right) \Big|_{z^{\frac{\tau_c}{2}} \log z}.$$

$$(B.39)$$

The 2d conformal block $g_{j+1,j+\tau_c-1}^{\tilde{\delta}_i}(z,\overline{z})$ is

$$g_{j+1,j+\tau_c-1}^{\tilde{\delta}_i}(z,\bar{z}) = k_{2-\tau_c}^{0,1}(z)k_{2j+\tau_c}^{0,1}(\bar{z}) + (z\leftrightarrow\bar{z}).$$
(B.40)

Expanding the above expression in small z, we find (the $(z \leftrightarrow \overline{z})$ term does not contribute)

$$\langle R_{j+\tau_c,j}^{(0)} \gamma_{j+\tau_c,j}^{(1)} \rangle = 4 \sum_{m=0}^{\frac{\tau_c}{2} - 3} \frac{(1 - \frac{\tau_c}{2})_m (2 - \frac{\tau_c}{2})_m}{(2 - \tau_c)_m m!} I_{2j+\tau_c} \left[\mathcal{G}(z,\overline{z}) \right] \Big|_{z^{\frac{\tau_c}{2} - m} \log z}, \quad (B.41)$$

where we have used the fact that $\mathcal{G}(z, \overline{z})$ starts at z^3 to restrict the range of the sum over *m*. Therefore, our task now is to find $I_{\beta} [\mathcal{G}(z, \overline{z})]|_{z^n \log z}$ for general *n*. Focusing

on the singular terms near $\overline{z} = 1$, we find

$$\begin{aligned} \mathcal{G}(z,\overline{z})|_{\log z} \\ &= z^3 \times \left(-\frac{1+z+2z^2}{2z(1-z)^2} \frac{1}{1-\overline{z}} - \frac{1}{2(1-z)^2} \frac{\overline{z}^3}{(1-\overline{z})^2} \left(1 + \frac{z\overline{z}}{2} - 2z\overline{z} \sum_{k=1}^{\infty} \frac{z^k \overline{z}^k}{k(k+2)} \right) \right. \\ &- \frac{(1+z)\log(1-z)}{2z(1-z)} \frac{\overline{z}^2}{(1-\overline{z})^2} - \frac{(1-z+2z^3)\log(1-z)}{2z^2(1-z)^2} \frac{\overline{z}^2}{(1-\overline{z})} \\ &- \frac{\overline{z}^2(z\overline{z} - (1-z)(1-\overline{z}))(z\overline{z}(1-z)(1-\overline{z}))}{2z(1-z)(1-\overline{z})} \sum_{k=1}^{\infty} \frac{z^k}{k} \left(1 - \frac{1}{(1-\overline{z})^k} \right) \\ &- \frac{z}{2(1-z)} \frac{\overline{z}\log(1-\overline{z})}{1-\overline{z}} - \frac{3(1+2z^2-z^3)}{2(1-z)^3} \overline{z}\log(1-\overline{z}) \right) + O((1-\overline{z})^0). \end{aligned} \tag{B.42}$$

We can then apply the functional I_{β} to the above expression and expand in small z. After several lines of algebra, we obtain that for $n \ge 1$

$$\begin{split} I_{\beta}\left[\mathcal{G}(z,\overline{z})\right]_{z^{3+n}\log z} \\ &= \left(-n-2+\frac{3-7n-4n^{2}+2n(n+2)S_{1}(n)+(n+2)S_{1}(n+1)}{2(n+2)}\right)I_{\beta}\left[\frac{1}{1-\overline{z}}\right] \\ &+\frac{1}{2}\left(S_{1}(n+1)+S_{1}(n)\right)I_{\beta}\left[\frac{\overline{z}^{2}}{(1-\overline{z})^{2}}\right] \\ &-\frac{1}{2}\left((n+1)I_{\beta}\left[\frac{\overline{z}^{3}}{(1-\overline{z})^{2}}\right]+\frac{n}{2}I_{\beta}\left[\frac{\overline{z}^{4}}{(1-\overline{z})^{2}}\right]-2\sum_{k=1}^{n-1}\frac{n-k}{k(k+2)}I_{\beta}\left[\frac{\overline{z}^{k+4}}{(1-\overline{z})^{2}}\right]\right) \\ &-\frac{1}{2}\left(\frac{I_{\beta}\left[\frac{\overline{z}^{2}}{(1-\overline{z})^{n}}\right]}{n+1}-\frac{I_{\beta}\left[\frac{\overline{z}^{2}}{(1-\overline{z})^{n-1}}\right]}{n}+S_{1}(n-1)I_{\beta}\left[\frac{\overline{z}^{4}}{(1-\overline{z})}\right]-\sum_{k=1}^{n-1}\frac{1}{k}I_{\beta}\left[\frac{\overline{z}^{4}}{(1-\overline{z})^{1+k}}\right]\right) \\ &-\frac{1}{2}I_{\beta}\left[\frac{\overline{z}}{(1-\overline{z})}\log(1-\overline{z})\right]-\frac{3}{2}n(n+2)I_{\beta}\left[\log(1-\overline{z})\right]. \end{split}$$
(B.43)

For the leading order term n = 0, we simply have

$$I_{\beta} \left[\mathcal{G}(z, \overline{z}) \right] \Big|_{z^{3} \log z} = -\frac{1}{4} I_{\beta} \left[\frac{1}{1 - \overline{z}} \right] - \frac{3}{2} I_{\beta} \left[\log(1 - \overline{z}) \right]$$
$$= -\frac{(\beta - 6)(\beta + 4)\Gamma(\frac{\beta}{2})\Gamma(\frac{\beta}{2} - 1)}{16\Gamma(\beta - 1)}.$$
 (B.44)

Combining (B.34), (B.35), (B.36), (B.41), (B.43), and (B.44), we can then determine

$$\langle R_{j+\tau_{c},j}^{(0)} \gamma_{j+\tau_{c},j}^{(1)} \rangle$$
 for any twist $\tau_{c} = 6, 8, 10, \dots$ For general τ_{c} , we obtain

$$\langle R_{j+\tau_{c},j}^{(0)} \gamma_{j+\tau_{c},j}^{(1)} \rangle$$

$$= \frac{(-1)^{\frac{\tau_{c}}{2}} (\tau_{c}^{2} - 2\tau_{c} + 8(-1)^{\frac{\tau_{c}}{2}} - 16)\Gamma(\frac{\tau_{c}}{2})\Gamma(\frac{\tau_{c}}{2} + 1)\Gamma(j + \frac{\tau_{c}}{2})\Gamma(j + \frac{\tau_{c}}{2} + 1)S_{1}(j + \frac{\tau_{c}}{2} - 1)}{2\Gamma(\tau_{c} - 1)\Gamma(2j + \tau_{c} - 1)}$$

$$+ \frac{\Gamma(j + \frac{\tau_{c}}{2})\Gamma(j + \frac{\tau_{c}}{2} - 1)}{\Gamma(2j + \tau_{c} - 1)}P_{\tau_{c}}(j), \qquad (B.45)$$

where $P_{\tau_c}(j)$ is a polynomial in *j*. It is given by

$$P_{\tau_c}(j) = \frac{(-1)^{\frac{\tau_c}{2}}(\frac{\tau_c}{2} - 2)\Gamma(\frac{\tau_c}{2})\Gamma(\frac{\tau_c}{2} + 2)(j + \frac{\tau_c}{2} - 3)(j + \frac{\tau_c}{2} + 2)}{2\Gamma(\tau_c - 1)} + \sum_{m=0}^{\frac{\tau_c}{2} - 4} \frac{(1 - \frac{\tau_c}{2})_m (2 - \frac{\tau_c}{2})_m}{m!(2 - \tau_c)_m} Q_{m,\tau_c}(j),$$
(B.46)

and

$$\begin{split} & \mathcal{Q}_{m,\tau_{c}}(j) \\ = & -2\tau_{c}\left(j^{2}-2j(m+2)+4m+6\right)+j^{2}(4m+6)+\tau_{c}^{2}(-2j+m+4)+\frac{4(j+m+2)(j+m+3)}{-2m+\tau_{c}-6} \\ & +\frac{4(j+m+1)(j+m+2)}{-2m+\tau_{c}-4}+\frac{4(j+m)(j+m+1)}{-2m+\tau_{c}-2}+j(6-4m)+6(m+1)(m+4)-\frac{\tau_{c}^{3}}{2} \\ & +2\Gamma(j+\frac{\tau_{c}}{2}+1)^{2} \left(\frac{3F_{2}\left(\begin{array}{c}5+m-\frac{\tau_{c}}{2}, 5+m-\frac{\tau_{c}}{2}, j+\frac{\tau_{c}}{2}\\ j+4+m, j+6+m\end{array};1\right)}{\Gamma(j+4+m)\Gamma(j+6+m)\Gamma(\frac{\tau_{c}}{2}-m-2)\Gamma(\frac{\tau_{c}}{2}-m-4)} \\ & -\frac{3F_{2}\left(\begin{array}{c}4+m-\frac{\tau_{c}}{2}, 4+m-\frac{\tau_{c}}{2}, j+\frac{\tau_{c}}{2}\\ j+3+m, j+5+m\end{array};1\right)}{\Gamma(j+3+m)\Gamma(j+5+m)\Gamma(\frac{\tau_{c}}{2}-m-1)\Gamma(\frac{\tau_{c}}{2}-m-3)} \\ & +2\sum_{k=1}^{\frac{\tau_{c}}{2}-4-m}}\frac{\Gamma(j+\frac{\tau_{c}}{2})\Gamma(j+\frac{\tau_{c}}{2}+1)\Gamma(j+\frac{\tau_{c}}{2}+3)_{3}F_{2}\left(\begin{array}{c}3-k, -k, j+\frac{\tau_{c}}{2}+1\\ j+\frac{\tau_{c}}{2}-k, j+\frac{\tau_{c}}{2}+3-k\end{array};1\right)}{k\Gamma(k+1)\Gamma(k+2)\Gamma(j+\frac{\tau_{c}}{2}-k)\Gamma(j+\frac{\tau_{c}}{2}+3-k)\Gamma(j+\frac{\tau_{c}}{2}-1)} \\ & (B.47) \end{split}$$

B.4 Lightcone bootstrap and large-*j* behavior of $\langle R_{j+6,j}^{(n+1)g/q} \rangle$

In this appendix, we describe how to obtain the large-*j* behavior of the coefficients $\langle R_{j+6,j}^{(n+1)g/q} \rangle$ using the double lightcone limit result (3.121) and the crossing equation (3.48).

B.4.1 Infinite sums of $SL(2, \mathbb{R})$ blocks

We will use lightcone bootstrap techniques from [81] to study the large-*j* behavior of the coefficients. An important ingredient for the lightcone bootstrap are identities for infinite sums of $SL(2, \mathbb{R})$ blocks. One identity derived in [81] is given in (3.54). For the calculation in this appendix, we can write it as (assuming *a* is negative)

$$\sum_{\substack{h=j+h_0\\j=0,2,\dots}} S_a^{r,s}(h) k_{2h}^{r,s}(1-z) = \frac{1}{2} z^a + O(z^{a+1}),$$

$$S_a^{r,s}(h) = \frac{1}{\Gamma(-a-r)\Gamma(-a-s)} \frac{\Gamma(\overline{h}-r)\Gamma(\overline{h}-s)}{\Gamma(2\overline{h}-1)} \frac{\Gamma(\overline{h}-a-1)}{\Gamma(\overline{h}+a+1)}.$$
(B.48)

For our calculation, we will also need to compute an infinite sum of derivatives of the SL(2, \mathbb{R}) blocks, such as $\partial_h k_{2h}^{r,s}(1-z)$ or $\partial_s k_{2h}^{r,s}(1-z)$. In particular, we want to compute

$$\sum_{j=0,2,\cdots} S_a^{0,-1} (j+3) \partial_{\delta}^n k_{\delta+j}^{0,-1} (1-z)|_{\delta \to j+6}, \tag{B.49}$$

$$\sum_{j=0,2,\cdots} S_a^{0,-1}(j+3)\partial_{\delta'_*}^n k_{2j+6}^{0,\frac{3-\delta'_*}{2}}(1-z)|_{\delta'_*\to 5}.$$
(B.50)

Let us first consider (B.49) with n = 1. One way of computing it is to follow the original argument for lightcone bootstrap given in [7, 8]. In the limit $h \to \infty$, $\partial_h k_{2h}(z) \sim \frac{(4\rho)^h}{\sqrt{1-\rho^2}} \log(4\rho)$, and therefore the dominant contribution of the sum in h is at $h \sim \frac{1}{1-\rho} \sim \frac{1}{\sqrt{1-z}}$. Thus, we should study the large h limit of $k_{2h}^{0,-1}(z)$ with $h\sqrt{1-z}$ fixed. Using the Euler integral representation of $k_{2h}(z)$, we find that

$$k_{2h}^{0,-1}(z) \sim 2^{2h} \sqrt{\frac{y}{\pi h}} K_1(2\sqrt{y}), \quad y = h^2(1-z),$$
 (B.51)

in the fixed y, $h \to \infty$ limit. Now, we can replace $1 - z \to z$ in (B.51) and plug it into (B.49) with n = 1. Focusing on the leading term in the small z limit, we find

$$\sum_{h=0,2,\dots} S_a^{0,-1}(h+3) \frac{1}{2} \partial_h k_{2h+6}^{0,-1}(1-z)$$

$$\approx \frac{1}{4} \int_0^\infty dh \frac{2^{-4-2h} h^{-2a-\frac{1}{2}} \sqrt{\pi}}{\Gamma(1-a)\Gamma(-a)} \partial_h \left(2^{2h+6} \sqrt{\frac{h^2(1-\overline{z})}{\pi h}} K_1(2h\sqrt{z}) \right)$$

$$= \frac{1}{2} z^a \left(\log 2 + \sqrt{z} \frac{(1+4a)\Gamma(-\frac{1}{2}-a)\Gamma(\frac{1}{2}-a)}{4\Gamma(1-a)\Gamma(-a)} \right) + O(z^{a+1}), \quad (B.52)$$

where \approx means that the Casimir-singular terms are the same. We have checked numerically that this formula is correct in the $z \rightarrow 0$ limit.

Interestingly, we find that if we simply move the derivative ∂_h of $\sum_{h=0,2,...} S_a^{0,-1}(h+3) \frac{1}{2} \partial_h k_{2h+6}^{0,-1}(1-z)$ to the function $S_a^{0,-1}(h+3)$, we will also get the same result in the small z limit. One way of understanding this is by following the discussion in [81]. We can write (B.49) as a contour integral

$$\int_{-\epsilon-i\infty}^{\epsilon+i\infty} dh \frac{\pi(1+e^{i\pi h})}{2\tan(\pi h)} S_a^{0,-1}(h+3) \frac{1}{2^n} \partial_h^n k_{2h+6}^{0,-1}(1-z),$$
(B.53)

and then integrate by parts to move all the derivatives ∂_h . The leading term at small z should come from $\partial_h^n S_a^{0,-1}(h+3)$. In conclusion, the infinite sum (B.49) in the small z limit is given by

$$\sum_{j=0,2,\cdots} S_a^{0,-1}(j+3)\partial_{\delta}^n k_{\delta+j}^{0,-1}(1-z)|_{\delta \to j+6} = \left(\frac{\log 2}{2}\right)^n z^a + O(z^{a+\frac{1}{2}}).$$
(B.54)

We now consider (B.50). We claim that it is given by

$$\sum_{j=0,2,\cdots} S_a^{0,-1} (j+3) \partial_{\delta_*}^n k_{2j+6}^{0,\frac{3-\delta_*}{2}} (1-z)|_{\delta_*\to 5} = c_n z^a \log^n z + O(z^a \log^{n-1} z).$$
(B.55)

The coefficient c_n can be determined by applying $\partial_{\delta'_*}$ to (B.48) *n* times. Since (B.48) is independent of δ'_* , we should get zero. Hence, we have

$$\sum_{j=0,2,\cdots} S_a^{0,-1}(j+3)\partial_{\delta'_*}^n k_{2j+6}^{0,\frac{3-\delta'_*}{2}}(1-z)|_{\delta'_*\to 5}$$

$$= -\sum_{m=1}^n \sum_{j=0,2,\cdots} \frac{n!}{m!(n-m)!} \left(\partial_{\delta'_*}^m S_a^{0,\frac{3-\delta'_*}{2}}(j+3)|_{\delta'_*\to 5} \right) \left(\partial_{\delta'_*}^{n-m} k_{2j+6}^{0,\frac{3-\delta'_*}{2}}(1-z)|_{\delta'_*\to 5} \right)$$

$$= -\sum_{m=1}^n \sum_{j=0,2,\cdots} \frac{n!}{m!(n-m)!} \frac{(-1)^m}{4^m} \partial_a^m S_a^{0,-1}(j+3) \left(\partial_{\delta'_*}^{n-m} k_{2j+6}^{0,\frac{3-\delta'_*}{2}}(1-z)|_{\delta'_*\to 5} \right) + O(z^a \log^{n-1} z)$$

$$= -\sum_{m=1}^n \frac{n!}{m!(n-m)!} \frac{(-1)^m}{4^m} c_{n-m} z^a \log^n z + O(z^a \log^{n-1} z). \tag{B.56}$$

This leads to the recursion relation

$$c_n = -\sum_{m=1}^n \frac{n!}{m!(n-m)!} \frac{(-1)^m}{4^m} c_{n-m}.$$
(B.57)

With the initial condition $c_0 = \frac{1}{2}$, one can then determine that $c_n = 2^{-1-2n}$. Thus,

$$\sum_{j=0,2,\cdots} S_a^{0,-1} (j+3) \partial_{\delta'_*}^n k_{2j+6}^{0,\frac{5-\delta'_*}{2}} (1-z)|_{\delta'_* \to 5} = \frac{1}{2^{1+2n}} z^a \log^n z + O(z^a \log^{n-1} z).$$
(B.58)

B.4.2 $\langle R_{i+6,i}^{(n+1)} \rangle$ in the large-*j* limit

We are now ready to compute the large-*j* behavior of $\langle R_{j+6,j}^{(n+1)g/q} \rangle$. As a warmup, we first consider the leading order (n = 0) case of the double lightcone limit (3.121). Since we know that $R_{j+2,j}^{(1)}$ is nonzero only when j = 2, the $z(1 - \overline{z})^0$ term of $\widetilde{\mathcal{G}}^{(1)}(z \ll 1 - \overline{z} \ll 1)$ should be produced by the first subleading term in the SL(2, \mathbb{R}) expansion of the $\tau_c = 4$ operator or the leading SL(2, \mathbb{R}) expansion of the $\tau_c = 6$ operator. More explicitly, the crossing equation (3.48) at leading order in the double lightcone limit is given by (suppressing the g/q superscript for brevity)

$$10R_{4,2}^{(1)}z(1-\overline{z})^{0} + \cdots$$

$$= \left(\frac{z\overline{z}}{(1-z)(1-\overline{z})}\right)^{3} \left[\sum_{j=0,2,\cdots} \langle R_{j+4,j}^{(1)} \rangle \left((1-\overline{z})^{2}k_{2j+4}^{0,-1}(1-z) + (1-\overline{z})^{3}\sum_{n=-1}^{1} C_{n}k_{2j+4+2n}^{0,-1}(1-z) + \cdots\right) + \sum_{j=0,2,\cdots} \langle R_{j+6,j}^{(1)} \rangle (1-\overline{z})^{3}k_{2j+6}^{0,-1}(1-z) + \cdots\right], \quad (B.59)$$

where \cdots are all subleading terms in the double lightcone limit. The coefficients C_n are from the SL(2, \mathbb{R}) block expansion of the conformal block, but their expressions will not be important for our discussion. The reason is that the $\langle R_{j+4,j}^{(1)} \rangle$ coefficient actually has no contribution to the leading term in the left hand side. There are two ways of seeing this. First, we can simply look at the expression for $\langle R_{j+4,j}^{(1)} \rangle$ in (3.105) and (3.109) we obtained from the Lorentzian inversion formula and take the large-*j* limit.² For both the gluon jet and quark jet, we find

$$\langle R_{j+4,j}^{(1)} \rangle \sim 4^{-j} j^{\frac{3}{2}} \sim S_{a=-1}^{0,-1}(j),$$
 (B.61)

and therefore

$$z^{3} \sum_{j=0,2,4,\cdots} \langle R_{j+4,j}^{(1)} \rangle k_{2j+4+2n}^{0,-1} (1-z) \sim z^{2},$$
(B.62)

$${}_{3}F_{2}\left(\begin{array}{ccc}2, & 3, & j+1\\ j+4, & j+5\end{array}; 1\right) = \frac{\Gamma(j+5)}{\Gamma(j+1)\Gamma(4)} \int_{0}^{1} dz \, z^{j} (1-z)^{3} {}_{2}F_{1}\left(2, 3, j+4, z\right) \sim 1 + \frac{6}{j} - \frac{18}{j^{2}} \tag{B.60}$$

Sometimes the $_2F_1$ is even explicitly known, and one can study the large-*j* limit after evaluating the integral.

²When taking the large-*j* limit of the ${}_{3}F_{2}$ functions in $\langle R_{j+4,j}^{(1)} \rangle$, it is more convenient to first use the Euler representation of the ${}_{3}F_{2}$ function to write it as an integral of the ${}_{2}F_{1}$ function, and then study the large-*j* limit of the integral numerically. For example,

which is indeed subleading in (B.59). The other argument is to consider the term

$$\left(\frac{z\overline{z}}{(1-z)(1-\overline{z})}\right)^3 (1-\overline{z})^2 \sum_{j} \langle R_{j+4,j}^{(1)} \rangle k_{2j+4}^{0,-1}(1-z).$$
(B.63)

If $\langle R_{j+4,j}^{(1)} \rangle$ grows like $S_{a=-2}^{0,-1}(j)$ or faster at large-*j*, we should expect to see $z(1-\overline{z})^{-1}$ on the left hand side of (B.59). However, such terms do not exist, so $z^3 \sum_j \langle R_{j+4,j}^{(1)} \rangle k_{2j+4+2n}^{0,-1}(1-z)$ must give subleading contribution to *z*.

Based on the above discussion, the crossing equation (B.59) becomes

$$10R_{4,2}^{(1)}z(1-\overline{z})^0 + \dots = \left(\frac{z\overline{z}}{(1-z)(1-\overline{z})}\right)^3 \sum_{j=0,2,\dots} \langle R_{j+6,j}^{(1)} \rangle (1-\overline{z})^3 k_{2j+6}^{0,-1}(1-z) + \dots$$
(B.64)

Using (B.48), we obtain

$$\langle R_{j+6,j}^{(1)} \rangle \sim 20 R_{4,2}^{(1)} S_{a=-2}^{0,-1} (j+3) \sim \frac{5\sqrt{\pi}}{8} R_{4,2}^{(1)} 4^{-j} j^{\frac{7}{2}}.$$
 (B.65)

which is (3.122) in the main text.

Let us now consider n = 1. In this case, we should allow nonzero $\langle R_{j+\tau_c,j}^{(2)} \rangle$ for general τ_c . At this order, the crossing equation in the double lightcone limit becomes

$$\begin{split} 10R_{4,2}^{(1)}\gamma_{4,2}^{(1)}z\log z(1-\overline{z})^{0} + \cdots \\ &= \left(\frac{z\overline{z}}{(1-z)(1-\overline{z})}\right)^{3} \left[\sum_{\tau_{c}<6}\sum_{j}\langle R_{j+\tau_{c},j}^{(2)} \rangle \sum_{n=0}^{\infty} (1-\overline{z})^{\frac{\tau_{c}}{2}+n} \sum_{m=-n}^{n} C_{n,m}^{\tau_{c}} k_{2j+\tau_{c}+2m}^{0,-1}(1-z) \right. \\ &+ \sum_{\tau_{c}<6}\sum_{j}\langle R_{j+\tau_{c},j}^{(1)} \gamma_{j+\tau_{c},j}^{(1)} \rangle \left(\sum_{n=0}^{\infty}\frac{1}{2}(1-\overline{z})^{\frac{\tau_{c}}{2}+n} \log(1-\overline{z}) \sum_{m=-n}^{n} C_{n,m}^{\tau_{c}} k_{2j+\tau_{c}+2m}^{0,-1}(1-z) \right. \\ &+ \sum_{\tau_{c}<6}\sum_{j}\langle R_{j+\tau_{c},j}^{(1)} \gamma_{j}^{(1)} \rangle \sum_{n=0}^{\infty} (1-\overline{z})^{\frac{\tau_{c}}{2}+n} \sum_{m=-n}^{n} C_{n,m}^{\tau_{c}} \partial_{\delta_{k}} k_{\delta+j+2m}^{0,-1}(1-z) |_{\delta \to j+\tau_{c}} \right) \\ &+ \sum_{\tau_{c}<6}\sum_{j}\langle R_{j+\tau_{c},j}^{(1)} \gamma_{*}^{(1)} \rangle \sum_{n=0}^{\infty} (1-\overline{z})^{\frac{\tau_{c}}{2}+n} \sum_{m=-n}^{n} C_{n,m}^{\tau_{c}} \partial_{\delta_{k}} k_{2j+\tau_{c}+2m}^{0,-1}(1-z) |_{\delta \to j+\tau_{c}} \right) \\ &+ \sum_{j}\langle R_{j+6,j}^{(2)} \rangle (1-\overline{z})^{3} k_{2j+6}^{0,-1}(1-z) \\ &+ \sum_{j}\langle R_{j+6,j}^{(1)} \gamma_{j+6,j}^{(1)} \rangle \left(\frac{1}{2}(1-\overline{z})^{3}\log(1-\overline{z})k_{2j+6}^{0,-1}(1-z) + (1-\overline{z})^{3} \partial_{\delta_{k}} k_{\delta+j}^{0,-1}(1-z) |_{\delta \to j+\tau_{c}} \right] \\ &+ \sum_{j}\langle R_{j+6,j}^{(1)} \gamma_{j}^{(1)} \rangle (1-\overline{z})^{3} \partial_{\delta_{k}} k_{2j+6}^{0,\frac{3-\delta_{k}'}{2}}(1-z) |_{\delta_{k}^{*}\to 5} + \cdots \right]. \\ &+ \sum_{j}\langle R_{j+6,j}^{(1)} \gamma_{j}^{\prime(1)} \rangle (1-\overline{z})^{3} \partial_{\delta_{k}} k_{2j+6}^{0,\frac{3-\delta_{k}'}{2}}(1-z) |_{\delta_{k}^{*}\to 5} + \cdots \right]. \end{aligned}$$

We can use (B.48) and (B.54) to argue that for $\tau_c < 6$, $\langle R_{j+\tau_c,j}^{(2)} \rangle$ and $\langle R_{j+\tau_c,j}^{(1)} \gamma_{j+\tau_c,j}^{(1)} \rangle$ cannot grow as fast as $\partial_a S_a^{0,-1}|_{a\to-2}$ at large *j*, otherwise the *z* log *z* term in the left hand side will have the wrong leading behavior for $1 - \overline{z} \ll 1$. Similarly, $\langle R_{j+6,j}^{(1)} \gamma_{j+6,j}^{(1)} \rangle$ should not give any leading contribution, or we will get *z* log *z* log $(1 - \overline{z})$. Thus, we have

$$10R_{4,2}^{(1)}\gamma_{4,2}^{(1)}z\log z + \cdots$$

= $z^{3}\left(\sum_{j} \langle R_{j+6,j}^{(2)} \rangle k_{2j+6}^{0,-1}(1-z) + \sum_{j} \langle R_{j+6,j}^{(1)}\gamma_{*}^{\prime(1)} \rangle \partial_{\delta_{*}^{\prime}}k_{2j+6}^{0,\frac{3-\delta_{*}^{\prime}}{2}}(1-z)|_{\delta_{*}^{\prime} \to 5} + \cdots\right).$
(B.67)

As discussed in section 3.4, we also need to deal with the degeneracies coming from γ'_* . To do so, let us make an ansatz for the large-*j* behavior of $R_{j+6,j}^{(n+1)}$. We will assume

$$\begin{split} R_{j+6,j}^{(n+1)g} &= \left(c_1^{(0)g} A_1^{(n+1)} + c_2^{(0)g} A_2^{(n+1)} \right) \partial_a^n S_a^{0,-1} (j+3)|_{a \to -2} + O(4^{-j} j^{\frac{7}{2}} \log^{n-1} j), \\ R_{j+6,j}^{(n+1)q} &= \left(c_1^{(0)q} A_1^{(n+1)} + c_2^{(0)q} A_2^{(n+1)} \right) \partial_a^n S_a^{0,-1} (j+3)|_{a \to -2} + O(4^{-j} j^{\frac{7}{2}} \log^{n-1} j), \\ (B.68) \end{split}$$

where the coefficients $c_i^{(0)g/q}$ are given in (3.137), and our goal is to determine $A_1^{(n+1)}$ and $A_2^{(n+1)}$. In general, the coefficient c^O at subleading order can also contribute, but we expect them to appear as $c_i^{(k)O}A_i^{(n+1-k)}\partial_a^{n-k}S_a|_{a\to-2}$. Since $\partial_a^{n-k}S_a|_{a\to-2}$ is subleading at large-*j*, we can just consider $c_i^{(0)O}$ for the leading large-*j* behavior. Note that the ansatz (B.68) implies that for any nonnegative integers *n* and *k*,

$$\langle R_{j+6,j}^{(n+1)g/q}(\gamma'_{*})^{k} \rangle = \left(c_{1}^{(0)g/q} A_{1}^{(n+1)}(\gamma'_{*1})^{k} + c_{2}^{(0)g/q} A_{2}^{(n+1)}(\gamma'_{*2})^{k} \right) \partial_{a}^{n} S_{a}^{0,-1}(j+3)|_{a \to -2}$$
$$+ O(4^{-j} j^{\frac{7}{2}} \log^{n-1} j),$$
(B.69)

where γ'_{*1} and γ'_{*2} are given by (3.132). Comparing (B.68) to the n = 0 result (B.65), we find

.....

$$A_{1}^{(1)} = \frac{1280\pi}{\alpha_{+} - \alpha_{-}} (R_{4,2}^{(1)q} - \alpha_{-} R_{4,2}^{(1)g}),$$

$$A_{2}^{(1)} = \frac{1280\pi}{\alpha_{+} - \alpha_{-}} \alpha_{-} (\alpha_{+} R_{4,2}^{(1)g} - R_{4,2}^{(1)q}).$$
(B.70)

For the n = 1 crossing equation (B.67), we can also plug in the ansatz (B.68), and use (B.48) and (B.58) to compute the infinite sums. After taking care of the

degeneracies, we obtain

$$A_{1}^{(2)} = \frac{320\pi}{\alpha_{+} - \alpha_{-}} (R_{4,2}^{(1)q} - \alpha_{-} R_{4,2}^{(1)g}) (4\gamma_{4,2}^{(1)} - \gamma_{*1}'),$$

$$A_{2}^{(2)} = \frac{320\pi}{\alpha_{+} - \alpha_{-}} \alpha_{-} (\alpha_{+} R_{4,2}^{(1)g} - R_{4,2}^{(1)q}) (4\gamma_{4,2}^{(1)} - \gamma_{*2}').$$
(B.71)

For n > 1, one can repeat the above argument and determine $A_1^{(n+1)}$, $A_2^{(n+1)}$. In particular, terms involving $R_{j+\tau_c,j}^{(n+1-p)}$ with $\tau_c < 6$ should not contribute to the $z \log^n z (1-\overline{z})^0$ term. Moreover, only $\langle R_{j+6,j}^{(n+1)} \rangle$ and $\langle R_{j+6,j}^{(n+1-m)} (\gamma_*^{\prime(1)})^m \rangle$ can produce $z \log^n z$. So, we have

$$10R_{4,2}^{(1)} \left(\gamma_{4,2}^{(1)}\right)^{n} z \log^{n} z + \cdots$$

$$= z^{3} \left(\sum_{j} \langle R_{j+6,j}^{(n+1)} \rangle k_{2j+6}^{0,-1} (1-z) + \sum_{m=1}^{n} \sum_{j} \langle R_{j+6,j}^{(n+1-m)} (\gamma_{*}^{\prime(1)})^{m} \rangle \partial_{\delta_{*}^{*}}^{m} k_{2j+6}^{0,\frac{3-\delta_{*}^{\prime}}{2}} (1-z)|_{\delta_{*}^{\prime} \to 5} + \cdots \right)$$
(B.72)

After using the infinite sum formula (B.48), (B.58) and the relation (B.69), we find that the solution takes a suprisingly simple form for $n \ge 1$:

$$A_{1}^{(n+1)} = \frac{320\pi (R_{4,2}^{(1)q} - \alpha_{-}R_{4,2}^{(1)g})(\gamma_{4,2}^{(1)})^{n-1}(4\gamma_{4,2}^{(1)} - \gamma_{*1}')}{\alpha_{+} - \alpha_{-}},$$

$$A_{2}^{(n+1)} = \frac{320\pi\alpha_{-}(\alpha_{+}R_{4,2}^{(1)g} - R_{4,2}^{(1)q})(\gamma_{4,2}^{(1)})^{n-1}(4\gamma_{4,2}^{(1)} - \gamma_{*2}')}{\alpha_{+} - \alpha_{-}}.$$
 (B.73)

It would be interesting to use some other methods to verify our results for the large-*j* behavior of $\langle R_{j+6,j}^{(n+1)} \rangle$. For example, $\langle R_{j+6,j}^{(1)} \rangle$ can be obtained by performing a higher-twist version of the Lorentzian inversion formula calculation described in section 3.3.3. If the collinear EEEC at subleading order is known, one should also be able to find $\langle R_{j+6,j}^{(n+1)} \rangle$ at higher values of *n*.

B.5 Tree-level EEEC

In this appendix, we give the details of the calculation of tree-level EEEC in section 3.5. In the first section, we compute the functions \mathcal{F}_0 and \mathcal{F}_1 in (3.157). In the second section, we derive the relation (3.168) using crossing symmetry.

B.5.1 Computing \mathcal{F}_0 and \mathcal{F}_1

We first calculate the squared amplitude for the initial state created by $\text{Tr}F^2$. We focus on processes with three out-going particles since \mathcal{F}_0 and \mathcal{F}_1 only get contributions from those processes. At tree-level, there are three possible processes with three outgoing particles. The first one includes three gluons, and its amplitude squared is given by (we use mostly positive metric)

$$\begin{split} |\mathcal{M}_{g+g+g}|^2 &= -64g^2 N_c (N_c^2 - 1) \times \\ & \left(6(p_1 \cdot p_2 + p_1 \cdot p_3 + p_2 \cdot p_3) \right. \\ & + 2 \left(\frac{(p_1 \cdot p_2)^2 + (p_2 \cdot p_3)^2}{p_1 \cdot p_3} + \frac{(p_1 \cdot p_3)^2 + (p_2 \cdot p_3)^2}{p_1 \cdot p_2} + \frac{(p_1 \cdot p_2)^2 + (p_1 \cdot p_3)^2}{p_2 \cdot p_3} \right. \\ & + \frac{(p_1 \cdot p_2)^3}{p_1 \cdot p_3 p_2 \cdot p_3} + \frac{(p_1 \cdot p_3)^3}{p_1 \cdot p_2 p_2 \cdot p_3} + \frac{(p_2 \cdot p_3)^3}{p_1 \cdot p_2 p_1 \cdot p_3} \\ & + 3 \left(\frac{p_1 \cdot p_3 p_2 \cdot p_3}{p_1 \cdot p_2} + \frac{p_1 \cdot p_2 p_2 \cdot p_3}{p_1 \cdot p_3} + \frac{p_1 \cdot p_2 p_1 \cdot p_3}{p_2 \cdot p_3} \right) \right). \end{split}$$
(B.74)

The second process has one gluon and two Weyl spinors

$$|\mathcal{M}_{g(p_1)+\lambda(p_2)+\overline{\lambda}(p_3)}|^2 = -16g^2 N_c (N_c^2 - 1) \frac{(p_1 \cdot p_2)^2 + (p_1 \cdot p_3)^2}{p_2 \cdot p_3}.$$
 (B.75)

Finally, the third process has one gluon and two scalars

$$|\mathcal{M}_{g(p_1)+\phi(p_2)+\phi(p_3)}|^2 = -32g^2 N_c (N_c^2 - 1) \frac{p_1 \cdot p_2 p_1 \cdot p_3}{p_2 \cdot p_3}.$$
 (B.76)

We can then define the total amplitude squared $|\mathcal{M}|^2$ as

$$|\mathcal{M}|^{2} = \frac{1}{3!} |\mathcal{M}_{g+g+g}|^{2} + 4|\mathcal{M}_{g(p_{1})+\lambda(p_{2})+\overline{\lambda}(p_{3})}|^{2} + \frac{6}{2!} |\mathcal{M}_{g(p_{1})+\phi(p_{2})+\phi(p_{3})}|^{2}.$$
 (B.77)

Note that due to the three identical gluons and two identical scalars in the final states, we should include symmetry factors $\frac{1}{3!}$ and $\frac{1}{2!}$ in the phase space measure for the corresponding final state. But here we choose to include those factors in $|\mathcal{M}|^2$, so that we can just use the same phase space measure for all the final states.

Comparing (3.149) and (3.157), one can show that the function \mathcal{F}_1 is defined as

$$\begin{aligned} \mathcal{F}_{1}(\vec{n}_{1},\vec{n}_{2}) \\ &= \sum_{i,j} \int d\sigma \frac{E_{i}^{2} E_{j}}{Q^{3}} \delta\left(\vec{n}_{1},\frac{\vec{p}_{i}}{E_{i}}\right) \delta\left(\vec{n}_{2},\frac{\vec{p}_{j}}{E_{j}}\right) \\ &= \frac{1}{\sigma_{\text{tot}}} \sum_{\{i,j\} \subset \{a,b,c\}} \int \frac{d^{3} \vec{p}_{a}}{(2\pi)^{3}} \frac{d^{3} \vec{p}_{b}}{(2\pi)^{3}} \frac{d^{3} \vec{p}_{c}}{(2\pi)^{3}} \frac{1}{2E_{a}} \frac{1}{2E_{b}} \frac{1}{2E_{c}} |\mathcal{M}|^{2} (2\pi)^{4} \delta(Q - E_{a} - E_{b} - E_{c}) \delta^{(3)} \left(\vec{p}_{a} + \vec{p}_{b} + \vec{p}_{c}\right) \\ &\times \frac{E_{i}^{2} E_{j}}{Q^{3}} \delta\left(\vec{n}_{1},\frac{\vec{p}_{i}}{E_{i}}\right) \delta\left(\vec{n}_{2},\frac{\vec{p}_{j}}{E_{j}}\right), \end{aligned}$$
(B.78)
and \mathcal{F}_0 is defined as

$$\mathcal{F}_{0}(\vec{n}_{1},\vec{n}_{2},\vec{n}_{3})\delta((\vec{n}_{1}\times\vec{n}_{2})\cdot\vec{n}_{3}) = \frac{1}{\sigma_{\text{tot}}}\sum_{\{i,j,k\}=\{a,b,c\}} \int \frac{d^{3}\vec{p}_{a}}{(2\pi)^{3}} \frac{d^{3}\vec{p}_{b}}{(2\pi)^{3}} \frac{d^{3}\vec{p}_{c}}{(2\pi)^{3}} \frac{1}{2E_{a}} \frac{1}{2E_{b}} \frac{1}{2E_{c}} |\mathcal{M}|^{2} (2\pi)^{4} \delta(Q - E_{a} - E_{b} - E_{c}) \delta^{(3)}(\vec{p}_{a} + \vec{p}_{b} + \vec{p}_{a}) + \vec{p}_{a} + \frac{E_{i}E_{j}E_{k}}{Q^{3}} \delta\left(\vec{n}_{1}, \frac{\vec{p}_{i}}{E_{i}}\right) \delta\left(\vec{n}_{2}, \frac{\vec{p}_{j}}{E_{j}}\right) \delta\left(\vec{n}_{3}, \frac{\vec{p}_{k}}{E_{k}}\right).$$
(B.79)

We can then plug in (B.77) for the squared amplitude $|\mathcal{M}|^2$. Also, the total cross section is

$$\sigma_{\rm tot} = \frac{N_c^2 - 1}{2\pi} Q^4.$$
(B.80)

Performing the integral in (B.78), we then obtain that \mathcal{F}_1 is given by (3.158). For (B.79), the delta function $\delta((\vec{n}_1 \times \vec{n}_2) \cdot \vec{n}_3)$ on the left-hand side will be canceled by one of the delta functions in $\delta^{(3)}$ ($\vec{p}_a + \vec{p}_b + \vec{p}_c$) on the right-hand side. After performing the calculation, we then find that \mathcal{F}_0 is given by (3.162) and (3.163). The step function $\theta(\zeta_1 + \zeta_2 - 1)$ in (3.162) comes from the condition $\vec{p}_a + \vec{p}_b + \vec{p}_c = 0$. If $\zeta_1 + \zeta_2 < 1$, one can easily draw a line such that all three momenta $\vec{p}_{a,b,c}$ lie on the same side of the line, and therefore there are no solutions to $\vec{p}_a + \vec{p}_b + \vec{p}_c = 0$.

B.5.2 Crossing symmetry of \mathcal{F}_0

We now consider crossing symmetry of $\mathcal{F}_0(\vec{n}_1, \vec{n}_2, \vec{n}_3)$. If $\vec{n}_1, \vec{n}_2, \vec{n}_3$ are all different from each other, it is not too hard to check that the function $\mathcal{F}_0(\vec{n}_1, \vec{n}_2, \vec{n}_3)$ given by (3.162) and (3.163) is crossing symmetric. So we want to focus on the delta functions and show that they are also crossing symmetric. In particular, we will consider 2 \leftrightarrow 3. Note that (3.162) can also be written as

$$\mathcal{F}_{0}(\vec{n}_{1},\vec{n}_{2},\vec{n}_{3}) = \frac{1}{16}\sqrt{(\vec{n}_{1}-\vec{n}_{3})^{2}(\vec{n}_{1}+\vec{n}_{3})^{2}(\vec{n}_{2}-\vec{n}_{3})^{2}(\vec{n}_{2}+\vec{n}_{3})^{2}}\widetilde{\mathcal{F}}_{0}(\vec{n}_{1},\vec{n}_{2},\vec{n}_{3}),$$
(B.81)

and the singular part of $\widetilde{\mathcal{F}}_0$ is given by (3.167).

We now preform the crossing $2 \leftrightarrow 3$ and first look at the contact term at $r_2 = 0$. Naively, we expect that the contact term looks like

$$\frac{\delta(r_2')}{r_2'} \frac{1}{16} \sqrt{(\vec{n}_1 - \vec{n}_2)^2 (\vec{n}_1 + \vec{n}_2)^2 (\vec{n}_2 - \vec{n}_3)^2 (\vec{n}_2 + \vec{n}_3)^2} \delta\left((\vec{n}_1 \times \vec{n}_2) \cdot \vec{n}_3\right), \quad (B.82)$$

where the new variable r'_2 is given by

$$r_2' = \frac{1}{4}\sqrt{(\vec{n}_1 + \vec{n}_2)^4 + (\vec{n}_2 - \vec{n}_3)^4} = h_2(\theta_2)r_2 + O(r_2^2), \tag{B.83}$$

312

and the function $h_2(\theta_2)$ is

$$h_2(\theta_2) = \sqrt{\cos^2 \theta_2 + (\sqrt{\cos \theta_2} - \sqrt{\sin \theta_2})^4}.$$
 (B.84)

If we integrate (B.82) against a test function $F(\vec{n}_1, \vec{n}_2, \vec{n}_3)$, we find

$$\int d\Omega_{\vec{n}_1} d\Omega_{\vec{n}_2} d\Omega_{\vec{n}_3} F(\vec{n}_1, \vec{n}_2, \vec{n}_3) \frac{\delta(r_2')}{r_2'} \frac{1}{16} \sqrt{(\vec{n}_1 - \vec{n}_2)^2 (\vec{n}_1 + \vec{n}_2)^2 (\vec{n}_2 - \vec{n}_3)^2 (\vec{n}_2 + \vec{n}_3)^2} \delta\left((\vec{n}_1 \times \vec{n}_2) \cdot \vec{n}_3\right)$$
$$= 8\pi^2 \int_0^{\frac{\pi}{4}} d\theta_2 \frac{1}{h_2(\theta_2)^2} \frac{\sqrt{\cos \theta_2} - \sqrt{\sin \theta_2}}{\sqrt{\sin \theta_2}} \int dr_2 \delta(r_2) F(\vec{n}_1, \vec{n}_2, \vec{n}_3)$$
$$= 2\pi^3 F(-\vec{n}_3, \vec{n}_3, \vec{n}_3), \qquad (B.85)$$

which agrees with the original contact term before crossing. However, there is actually another delta function coming from the $[\cdots]_0$ distribution. This is due to the relation

$$\left[\frac{1}{ax}\right]_0 = \frac{1}{a} \left[\frac{1}{x}\right]_0 + \frac{\log a}{a} \delta(x).$$
(B.86)

Therefore, from the $[\cdots]_0$ distribution we have

$$\frac{f_0(\theta_2')}{r_2'} \left[\frac{1}{r_2'}\right]_0 \to \frac{f_0(\theta_2')}{h_2(\theta_2)r_2} \frac{\log h_2(\theta_2)}{h_2(\theta_2)} \delta(r_2), \tag{B.87}$$

where

$$\theta_2' = \tan^{-1} \left(\frac{(\vec{n}_1 + \vec{n}_2)^2}{(\vec{n}_2 - \vec{n}_3)^2} \right) = \tan^{-1} \left((1 - \sqrt{\tan \theta_2})^2 \right) + O(r_2).$$
(B.88)

Therefore, we should also consider

$$8\pi^2 \int_0^{\frac{\pi}{4}} d\theta_2 \frac{f_0(\theta_2') \log h_2(\theta_2)}{h_2(\theta_2)^2} \frac{\sqrt{\cos \theta_2} - \sqrt{\sin \theta_2}}{\sqrt{\sin \theta_2}} \int dr_2 \delta(r_2) F(\vec{n}_1, \vec{n}_2, \vec{n}_3).$$
(B.89)

It turns out that the θ_2 integral actually vanishes. So, for the $r_2 = 0$ contact term, the $[\cdots]_0$ distribution does not produce new delta function after crossing $2 \leftrightarrow 3$.

Now we consider the other two contact terms. After $2 \leftrightarrow 3$, r_1 and r_3 become

$$r_{1}' = \frac{1}{4}\sqrt{(\vec{n}_{1} - \vec{n}_{2})^{4} + (\vec{n}_{2} + \vec{n}_{3})^{4}} = h_{1 \to 3}(\theta_{3})r_{3} + O(r_{3}^{2}),$$

$$r_{3}' = \frac{1}{4}\sqrt{(\vec{n}_{1} + \vec{n}_{2})^{4} + (\vec{n}_{2} + \vec{n}_{3})^{4}} = h_{3 \to 1}(\theta_{1})r_{1} + O(r_{1}^{2}),$$
 (B.90)

where

$$h_{1\to3}(\theta_3) = \sqrt{\sin^2 \theta_3 + \left(\sqrt{\cos \theta_3} + \sqrt{\sin \theta_3}\right)^4},$$

$$h_{3\to1}(\theta_1) = \sqrt{\sin^2 \theta_1 + \left(\sqrt{\cos \theta_1} - \sqrt{\sin \theta_1}\right)^4}.$$
 (B.91)

Also, θ_1 and θ_3 become

$$\theta_{1}' = \tan^{-1} \left(\frac{(\vec{n}_{2} + \vec{n}_{3})^{2}}{(\vec{n}_{1} - \vec{n}_{2})^{2}} \right) = \tan^{-1} \left(\frac{\sin \theta_{3}}{(\sqrt{\cos \theta_{3}} + \sqrt{\sin \theta_{3}})^{2}} \right) + O(r_{3}),$$

$$\theta_{3}' = \tan^{-1} \left(\frac{(\vec{n}_{2} + \vec{n}_{3})^{2}}{(\vec{n}_{1} + \vec{n}_{2})^{2}} \right) = \tan^{-1} \left(\frac{\sin \theta_{1}}{(\sqrt{\cos \theta_{1}} - \sqrt{\sin \theta_{1}})^{2}} \right) + O(r_{1}).$$
(B.92)

Integrating the $\delta(r'_1)$ term, we get

$$\int d\Omega_{\vec{n}_1} d\Omega_{\vec{n}_2} d\Omega_{\vec{n}_3} F(\vec{n}_1, \vec{n}_2, \vec{n}_3) \frac{\delta(r_1')}{r_1'} \frac{1}{16} \sqrt{(\vec{n}_1 - \vec{n}_2)^2 (\vec{n}_1 + \vec{n}_2)^2 (\vec{n}_2 - \vec{n}_3)^2 (\vec{n}_2 + \vec{n}_3)^2} \delta((\vec{n}_1 \times \vec{n}_2) \cdot \vec{n}_3)$$

$$= 8\pi^2 \int_0^{\frac{\pi}{2}} d\theta_3 \frac{1}{h_{1\to3}(\theta_3)^2} \frac{\sqrt{\cos\theta_3} + \sqrt{\sin\theta_3}}{\sqrt{\cos\theta_3}} \int dr_3 \delta(r_3) F(\vec{n}_1, \vec{n}_2, \vec{n}_3)$$

$$= 2\pi^3 F(-\vec{n}_3, -\vec{n}_3, \vec{n}_3). \tag{B.93}$$

We should also include the contribution from the $\left[\frac{1}{r'_1}\right]_0$ distribution. This term will give

$$8\pi^{2} \int_{0}^{\frac{\pi}{2}} d\theta_{3} \frac{f_{0}(\theta_{1}') \log h_{1 \to 3}(\theta_{3})}{h_{1 \to 3}(\theta_{3})^{2}} \frac{\sqrt{\cos \theta_{3}} + \sqrt{\sin \theta_{3}}}{\sqrt{\cos \theta_{3}}} \int dr_{3} \delta(r_{3}) F(\vec{n}_{1}, \vec{n}_{2}, \vec{n}_{3})$$
$$= 8\pi^{2} \times \frac{\pi - \sqrt{2}\pi + 2\log 2 + \sqrt{2}\log(3 - 2\sqrt{2})}{512\pi^{4}} F(-\vec{n}_{3}, -\vec{n}_{3}, \vec{n}_{3}).$$
(B.94)

Comparing this result with the contact terms before crossing, we find that for the delta functions to be crossing-symmetric, we must have (3.168). Also, if we consider the contact term at $r_1 = 0$, we will get the same condition.

B.6 More details on the celestial inversion formula

In this appendix, we give the derivation for the orthogonality relation of celestial partial waves (3.193) and the integral identity (3.204). We also show that the contributions at infinity of the contour deformations of (3.198) vanish.

B.6.1 Orthogonality of celestial partial waves

To derive the orthogonality relation, let us consider a natural pairing

$$\int D^{d-2} z_1 D^{d-2} z_2 D^{d-2} z_3 d^{d-1}_{\text{AdS}} p \Psi^c_{\delta_5, j_5; \delta'_5}(z_1, z_2, z_3, p) \Psi^{c(\tilde{\delta}_i)}_{\tilde{\delta}_6, j_6; \tilde{\delta}'_6}(z_1, z_2, z_3, p),$$
(B.95)

where $d_{AdS}^{d-1}p = 2d^d p \delta(p^2 + 1)\theta(p^0)$ is an integral over the AdS space defined by $p^2 = -1$. Plugging in the definition of the celestial partial wave, we get

$$\int D^{d-2} z_1 D^{d-2} z_2 D^{d-2} z_3 d_{AdS}^{d-1} p D^{d-2} z D^{d-2} z' D^{d-2} z'' D^{d-2} z'''$$

$$\langle \mathcal{P}_{\delta_1}(z_1) \mathcal{P}_{\delta_2}(z_2) \mathcal{P}_{\delta_5, j_5}(z) \rangle \langle \widetilde{\mathcal{P}}_{\delta_5, j_5}(z) \mathcal{P}_{\delta_3}(z_3) \mathcal{P}_{\delta_5'}(z') \rangle \frac{1}{(-2z' \cdot p)^{\delta_5'}}$$

$$\times \langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_2}(z_2) \widetilde{\mathcal{P}}_{\delta_6, j_6}(z'') \rangle \langle \mathcal{P}_{\delta_6, j_6}(z'') \widetilde{\mathcal{P}}_{\delta_3}(z_3) \widetilde{\mathcal{P}}_{\delta_6'}(z''') \rangle \frac{1}{(-2z''' \cdot p)^{\delta_6'}}$$

$$= B_{12\mathcal{P}_{\delta_5, j_5}} \delta_{\mathcal{P}_5 \mathcal{P}_6} \int D^{d-2} z_3 d_{AdS}^{d-1} p D^{d-2} z D^{d-2} z' D^{d-2} z''''$$

$$\langle \widetilde{\mathcal{P}}_{\delta_5, j_5}(z) \mathcal{P}_{\delta_3}(z_3) \mathcal{P}_{\delta_5'}(z') \rangle \frac{1}{(-2z' \cdot p)^{\delta_5'}} \langle \mathcal{P}_{\delta_5, j_5}(z) \widetilde{\mathcal{P}}_{\delta_3}(z_3) \widetilde{\mathcal{P}}_{\delta_6'}(z''') \rangle \frac{1}{(-2z''' \cdot p)^{\delta_6'}}$$

$$= B_{12\mathcal{P}_{\delta_5, j_5}} B_{\widetilde{\mathcal{P}}_{\delta_5, j_5} 3\mathcal{P}_{\delta_5'}} \delta_{\mathcal{P}_5 \mathcal{P}_6} \delta_{\mathcal{P}_5' \mathcal{P}_6'}$$

$$\times \int d_{AdS}^{d-1} p D^{d-2} z' \frac{1}{(-2z' \cdot p)^{\delta_5'}} \frac{1}{(-2z' \cdot p)^{\delta_5'}}.$$
(B.96)

In the first and the second equality above, we have used the bubble formula (eq. (2.32) in [45]), and $B_{12\mathcal{P}_{\delta_5,j_5}}$, $B_{\tilde{\mathcal{P}}_{\delta_5,j_5}3\mathcal{P}_{\delta'_5}}$ are bubble coefficients given by (3.195). Moreover, the integral in the last line of (B.96) is given by

$$\int \frac{D^{d-2} z d_{\text{AdS}}^{d-1} p}{\text{vol}(\text{SO}(d-1,1))} (-2p \cdot z)^{-\delta} (-2p \cdot z)^{-\widetilde{\delta}} = \frac{1}{2^{d-2} \text{vol}(\text{SO}(d-2))}, \quad (B.97)$$

where we use the conformal group to gauge fix p = (1, 0, 0, ..., 0) and z = (1, 1, 0, ..., 0). The stabilizer group after the gauge-fixing is SO(d - 2), and the Fadeev-Popov determinant for the gauge-fixing is 1.

Therefore, the orthogonality relation for the celestial partial wave is

$$\int \frac{D^{d-2}z_1 D^{d-2} z_2 D^{d-2} z_3 d_{AdS}^{d-1} p}{\operatorname{vol}(\operatorname{SO}(d-1,1))} \Psi_{\delta_5,j_5;\delta_5'}^c(z_1, z_2, z_3, p) \Psi_{\widetilde{\delta}_6,j_6;\widetilde{\delta}'_6}^{c(\widetilde{\delta}_i)}(z_1, z_2, z_3, p)$$

$$= \frac{1}{2^{d-2} \operatorname{vol}(\operatorname{SO}(d-2))} B_{12\mathcal{P}_{\delta_5,j_5}} B_{\widetilde{\mathcal{P}}_{\delta_5,j_5} 3\mathcal{P}_{\delta_5'}} \delta_{\mathcal{P}_5} \mathcal{P}_6} \delta_{\mathcal{P}_5'} \mathcal{P}_6', \qquad (B.98)$$

where

$$\delta \varphi_5 \varphi_6 = 2\pi \delta (s_5 - s_6) \delta_{j_5, j_6}, \tag{B.99}$$

for $\delta_5 = \frac{d-2}{2} + is_5$, $\delta_6 = \frac{d-2}{2} + is_6$ with s_5 , $s_6 > 0$.

B.6.2 Derivation of (3.204)

We now consider the identity

$$\int D^{d-2} z_1 D^{d-2} z_2 \langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_2}(z_2) \mathcal{P}_{\delta,j=0}(z) \rangle (-2p \cdot z_1)^{-\delta_1} (-2p \cdot z_2)^{-\delta_2} = C_{\delta_1,\delta_2;\delta} (-p^2)^{\frac{\delta-\delta_1-\delta_2}{2}} (-2p \cdot z)^{-\delta}.$$
(B.100)

By Lorentz symmetry and homogeneity of p and z, the right-hand side must be proportional to $(-p^2)^{\frac{\delta-\delta_1-\delta_2}{2}}(-2p \cdot z)^{-\delta}$. Thus, our goal here is showing that the coefficient $C_{\delta_1,\delta_2;\delta}$ is given by (3.205). Our strategy is to first fix $p^2 = -1$, and integrate both sides of (3.204) against $(-2p \cdot z)^{-\tilde{\delta}}$ over z and p. More precisely, for the right-hand side, we have

$$\int \frac{D^{d-2}z d_{\text{AdS}}^{d-1} p}{\text{vol}(\text{SO}(d-1,1))} C_{\delta_{1},\delta_{2};\delta} (-2p \cdot z)^{-\delta} (-2p \cdot z)^{-\widetilde{\delta}} = C_{\delta_{1},\delta_{2};\delta} \frac{1}{2^{d-2} \text{vol}(\text{SO}(d-2))}$$
(B.101)

which follows from (B.97). For the left-hand side, we want to compute

$$\int \frac{D^{d-2} z_1 D^{d-2} z_2 D^{d-2} z d_{AdS}^{d-1} p}{\operatorname{vol}(\mathrm{SO}(d-1,1))} \langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_2}(z_2) \mathcal{P}_{\delta,0}(z) \rangle (-2p \cdot z_1)^{-\delta_1} (-2p \cdot z_2)^{-\delta_2} (-2p \cdot z)^{-\widetilde{\delta}}.$$
(B.102)

One can immediately recognize that the *p* integral is a three-point Witten diagram, and can be evaluated using (3.207). Furthermore, the remaining integral over z_1, z_2, z is a conformally-invariant three-point pairing. Therefore, the left-hand side is given by

$$\int \frac{D^{d-2}z_1 D^{d-2}z_2 D^{d-2}z d_{AdS}^{d-1} p}{\operatorname{vol}(\operatorname{SO}(d-1,1))} \langle \widetilde{\mathcal{P}}_{\delta_1}(z_1) \widetilde{\mathcal{P}}_{\delta_2}(z_2) \mathcal{P}_{\delta,0}(z) \rangle (-2p \cdot z_1)^{-\delta_1} (-2p \cdot z_2)^{-\delta_2} (-2p \cdot z)^{-\widetilde{\delta}} \\ = D_{\delta_1,\delta_2,\widetilde{\delta}} \left(\langle \widetilde{\mathcal{P}}_{\delta_1} \widetilde{\mathcal{P}}_{\delta_2} \mathcal{P}_{\delta} \rangle, \langle \mathcal{P}_{\delta_1} \mathcal{P}_{\delta_2} \widetilde{\mathcal{P}}_{\delta} \rangle \right) \\ = \frac{\pi^{\frac{d-2}{2}} \Gamma(\frac{\delta_1 + \delta_2 + \widetilde{\delta} - d+2}{2}) \Gamma(\frac{\delta_1 + \delta_2 - \widetilde{\delta}}{2}) \Gamma(\frac{\delta_1 + \widetilde{\delta} - \delta_2}{2}) \Gamma(\frac{\delta_2 + \widetilde{\delta} - \delta_1}{2})}{2\Gamma(\delta_1) \Gamma(\delta_2) \Gamma(\widetilde{\delta})} \frac{1}{2^{d-2} \operatorname{vol}(\operatorname{SO}(d-3))}.$$
(B.103)

Finally, comparing (B.101) and (B.103), we obtain

$$C_{\delta_1,\delta_2;\delta} = \frac{\pi^{\frac{d-2}{2}} \Gamma(\frac{\delta_1 + \delta_2 + \widetilde{\delta} - d+2}{2}) \Gamma(\frac{\delta_1 + \delta_2 - \widetilde{\delta}}{2}) \Gamma(\frac{\delta_1 + \widetilde{\delta} - \delta_2}{2}) \Gamma(\frac{\delta_2 + \widetilde{\delta} - \delta_1}{2})}{2\Gamma(\delta_1) \Gamma(\delta_2) \Gamma(\widetilde{\delta})} \operatorname{vol}(S^{d-3}), \text{ (B.104)}$$

which agrees with (3.205).

B.6.3 Celestial block at large δ and δ'

In this section, we study (3.198),

$$\mathcal{F}(z_1, z_2, z_3, p) = \sum_j \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta}{2\pi i} \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta'}{2\pi i} C(\delta, j; \delta') G^c_{\delta, j; \delta'}(z_1, z_2, z_3, p),$$
(B.105)

and make sure that the contributions at infinity vanish when doing the contour deformations. For concreteness, we consider the leading order strong-coupling EEEC, so $C(\delta, j; \delta')$ is given by (3.211). When we first close the δ' contour to the right, the locations of the poles are at $\delta' = \delta + 3 + 2k$. Furthermore, (3.211) in the large δ' limit behaves like

$$C_{\text{strong}}^{(0)}(\delta, j; \delta' \to \infty) \sim 2^{-\delta'} (\cdots), \qquad (B.106)$$

where (\cdots) grows sub-exponentially at large δ' . Therefore, a sufficient condition for the contribution at infinity of the δ' contour to vanish is

$$\lim_{\operatorname{Re}(\delta')\to\infty} 2^{-\delta'} \text{ (sub-exponential) } G^c_{\delta,j;\delta'}(z_1, z_2, z_3, p) = 0.$$
 (B.107)

After closing the δ' contour, (3.198) becomes

$$\mathcal{F}(z_1, z_2, z_3, p) = \sum_j \sum_{k=0}^{\infty} \int_{\frac{d-2}{2} - i\infty}^{\frac{d-2}{2} + i\infty} \frac{d\delta}{2\pi i} \left(\operatorname{Res}_{\delta' = \delta + 3 + 2k} C_{\operatorname{strong}}^{(0)}(\delta, j; \delta') \right) G_{\delta, j; \delta' = \delta + 3 + 2k}^c(z_1, z_2, z_3, p)$$
(B.108)

On the δ -plane, $\operatorname{Res}_{\delta'=\delta+3+2k}C_{\operatorname{strong}}^{(0)}$ has poles at $\delta = 6 + 2n$. The contributions from these poles will reproduce the celestial block coefficients (3.214). Note that at large δ ,

$$\operatorname{Res}_{\delta'=\delta+3+2k} C_{\operatorname{strong}}^{(0)} \sim 2^{-\delta} (\cdots), \qquad (B.109)$$

where (\cdots) grows sub-exponentially at large δ . Thus, a sufficient condition for the contribution at infinity of the δ contour to vanish is

$$\lim_{\operatorname{Re}(\delta)\to\infty} 2^{-\delta} \text{ (sub-exponential) } G^c_{\delta,j;\delta+3+2k}(z_1, z_2, z_3, p) = 0.$$
(B.110)

If the two conditions (B.107) and (B.110) are true, the celestial block expansion (3.183) can be obtained from (3.198) by contour deformation.

To show that (B.107) and (B.110) are true, we will need to understand the behavior of the celestial block G^c at large δ or δ' . For four-point conformal blocks, one

can determine their large Δ behavior by studying the Casimir equation in the limit $\Delta \rightarrow \infty$ [36, 37, 35, 180, 38]. For the celestial block, the analogous Casimir equations are

$$\left(-\frac{1}{2} L^{(12)}_{\mu\nu} L^{(12)\mu\nu} - \delta(\delta - d + 2) - j(j + d - 4) \right) G^c_{\delta,j;\delta'} = 0,$$

$$\left(-\frac{1}{2} L^{(123)}_{\mu\nu} L^{(123)\mu\nu} - \delta'(\delta' - d + 2) \right) G^c_{\delta,j;\delta'} = 0,$$
(B.111)

where

$$L_{\mu\nu}^{(12)} = \sum_{i=1}^{2} z_{i\mu} \frac{\partial}{\partial z_{i}^{\nu}} - z_{i\nu} \frac{\partial}{\partial z_{i}^{\mu}},$$

$$L_{\mu\nu}^{(123)} = \sum_{i=1}^{3} z_{i\mu} \frac{\partial}{\partial z_{i}^{\nu}} - z_{i\nu} \frac{\partial}{\partial z_{i}^{\mu}}.$$
(B.112)

If we take the limit given by (B.107), the second line of (B.111) will give a differential equation that the leading behavior of G^c in this limit must satisfy. Similarly, for the limit given by (B.110), (B.111) will give two differential equations. However, since G^c can depend nontrivially on three cross ratios, ζ_{12} , ζ_{13} , ζ_{23} , these differential equations are not as simple as as the conformal block case, and it is difficult to solve them directly.

Thus, we are led to consider an alternative method for studying G^c at large δ or δ' . We find that this can be achieved by writing down an integral representation with finite integration range for the celestial block. When we consider the limits given by (B.107) and (B.110), the integral will be dominated by a saddle point and the behavior of G^c can be determined.

Warmup: conformal blocks at large Δ revisited

The method also applies to the conformal block, so let us first consider this simpler case. They key idea is to use the *Lorentzian* shadow representation of the block [2]. Here, we follow the notation of [11], where the block can be written

$$G_{\Delta,J}(x_i) \sim \int_{1 > x_0 > 2} d^d x_0 D^{d-2} z |T_{d-\Delta,2-d-J}(x_1, x_2, x_0, z)| T_{\Delta,J}(x_3, x_4, x_0, z),$$
(B.113)

where the causality configuration is 1 > 2, 3 > 4, and all other points are spacelike. Here, ~ means that the two sides can differ by a factor independent of the positions x_i , and

$$T_{\Delta,J}(x_1, x_2, x_0, z) = \frac{(2z \cdot x_{20} x_{10}^2 - 2z \cdot x_{10} x_{20}^2)^J}{(-x_{12}^2)^{\frac{\Delta_1 + \Delta_2 - \Delta + J}{2}} (x_{10}^2)^{\frac{\Delta_1 + \Delta - \Delta_2 + J}{2}} (x_{20}^2)^{\frac{\Delta_2 + \Delta - \Delta_1 + J}{2}}$$
(B.114)

is a conformal three-point structure. Since we are interested in the large Δ limit, we focus on the Δ -dependence of the integrand,

$$\left(\frac{(-x_{10}^2)(-x_{20}^2)(-x_{34}^2)}{(-x_{12}^2)(-x_{30}^2)(-x_{40}^2)}\right)^{\frac{1}{2}}.$$
(B.115)

Let us choose lightcone coordinates $x = (u, v, x_{\perp})$, where $x^2 = uv + x_{\perp}^2$, and set $x_1 = (u_1, v_1, 0), x_2 = (0, 0, 0), x_3 = (1, 1, 0), x_4 = \infty$. The conditions 1 > 2 and $1 \approx 3$ become $v_1 < 0, 0 < u_1 < 1$. Our integral becomes

$$(-u_1v_1)^{-\frac{\Lambda}{2}} \int du dv d^{d-2}x_{\perp} \left((-uv - r^2)(-(u - u_1)(v - v_1) - r^2) \right)^{\frac{\Lambda}{2}} \left((u - 1)(v - 1) + r^2 \right)^{-\frac{\Lambda}{2}} (\cdots)$$
(B.116)

where (\cdots) are independent of Δ . In the large Δ limit, a saddle point appears at $u = 1 - \sqrt{1 - u_1}$, $v = 1 - \sqrt{1 - v_1}$, r = 0. Therefore, the leading large Δ behavior is given by

$$\mathcal{G}_{\Delta,J} \sim (-u_1 v_1)^{-\frac{\Delta}{2}} \left((1 - \sqrt{1 - u_1})(\sqrt{1 - v_1} - 1) \right)^{\Delta} (\cdots) .$$
 (B.117)

Going back to the more familiar cross ratios $\rho, \overline{\rho}$ [185], we find

$$\frac{-u_1v_1}{(1-\sqrt{1-u_1})^2(\sqrt{1-v_1}-1)^2} = \frac{1}{\rho\overline{\rho}}.$$
 (B.118)

Therefore, in the large Δ limit, $G_{\Delta,J}$ is proportional to $(\rho \overline{\rho})^{\frac{\Delta}{2}} (\cdots)$. One can fix the position-independent factor by considering the OPE limit of $G_{\Delta,J}$. Eventually, we obtain

$$G_{\Delta,J} \sim 4^{\Delta} (\rho \overline{\rho})^{\frac{\Delta}{2}} (\cdots),$$
 (B.119)

which agrees with the known result. To match the full result in [36, 37, 35, 180, 38], we could additionally include the 1-loop determinant around the saddle point. However, that such subleading terms will not be important in our analysis.

Lorentzian integrals for the celestial block

For the three-point celestial block G^c , we can write down a similar integral representation with finite integration range by continuing to "double Lorentzian" signature, as in section 3.2.4. That is, we must analytically continue the celestial sphere to a Lorentzian signature space, so that the full spacetime has signature (2, d - 2). We then have

$$G_{\delta,j;\delta'}^{c}(z_{i},p) \sim \int_{\substack{1>0>2\\0>0'>3}} D^{d-2}z_{0}D^{d-2}z_{0'}\frac{1}{(-2p\cdot z_{0'})^{\delta'}} \langle \widetilde{\mathcal{P}}_{\delta'}(z_{0}')\mathcal{P}_{3}\mathcal{P}_{\delta,j}(z_{0})\rangle \langle \widetilde{\mathcal{P}}_{\delta,j}(z_{0})\mathcal{P}_{1}\mathcal{P}_{2}\rangle$$
(B.120)

where z_i , p are in (2, d - 2) signature. We can view z_i as the embedding space coordinates of $\mathbb{R}^{1,d-3}$, and the causality constraints 1 > 0 > 2 and 0 > 0' > 3 should be understood in this space. Since the right-hand side is a solution of the Casimir equations (B.111) by construction, one can show that it is proportional to G^c (up to a factor independent of z_i and p) by considering its various OPE limits. To study the integral more explicitly, we will pick the frame $z_i = (1, \vec{y}_i^2, \vec{y}_i), p = (p_+, 1, \vec{y}_p)$, where

$$\vec{y}_1 = (1, -1, 0), \quad \vec{y}_2 = (0, 0, 0), \quad \vec{y}_3 = (-1, 1, 0), \quad \vec{y}_p = (y_p^+, y_p^-, 0), \quad (B.121)$$

where the coordinate for \vec{y} is $(y^+, y^-, \vec{y}_\perp)$, and $\vec{y}^2 = y^+y^- + \vec{y}_\perp^2$. This satisfies the causality constraint 1 > 2 > 3. Also, the relation between p_+ , y_p^+ , y_p^- and ζ_{12} , ζ_{13} , ζ_{23} can be obtained straightforwardly using the definition of the cross ratios.

Let us now consider the limit $\delta' \to \infty$ corresponding to (B.107). For this limit, it is more convenient to study

$$\int_{1>2>0'>3} D^{d-2} z'_0 \frac{1}{(-2p \cdot z'_0)^{\delta'}} g^{(\delta_1, \delta_2, \delta_3, \widetilde{\delta'})}_{\delta, j}(z_1, z_2, z_3, z'_0).$$
(B.122)

In the limit $\delta' \to \infty$, the conformal block $g_{\delta,j}^{(\delta_1,\delta_2,\delta_3,\widetilde{\delta'})}(z_1, z_2, z_3, z'_0)$ behaves like [186]

$$g_{\delta,j}^{(\delta_1,\delta_2,\delta_3,\widetilde{\delta'})}(z_1,z_2,z_3,z_0') \sim \left(\frac{(-2z_2\cdot z_3)}{(-2z_2\cdot z_0')(-2z_3\cdot z_0')}\right)^{-\frac{\delta'}{2}}(\cdots), \qquad (B.123)$$

where (\cdots) grows sub-exponentially. Therefore, we must consider the integral

$$\int_{1>2>0'>3} D^{d-2} z_0' \frac{1}{(-2p \cdot z_0')^{\delta'}} \left(\frac{(-2z_2 \cdot z_3)}{(-2z_2 \cdot z_0')(-2z_3 \cdot z_0')} \right)^{-\frac{\delta'}{2}}.$$
 (B.124)

After considering the integral in the frame $z_i = (1, \vec{y}_i^2, \vec{y}_i), p = (p_+, 1, \vec{y}_p)$ given by (B.121) and solving for its saddle point in the large δ' limit, we can find the behavior of the integral at large δ' . We can further determine the position-independent factor in (B.120) by matching the integral with the collinear limit of G^c given by (3.34).^{3,4} We find that at large δ' ,

$$G_{\delta,j;\delta'\to\infty}^{c} \sim T_{123\delta'}(z_1, z_2, z_3, p) \left(\frac{2(1 - \sqrt{1 - \zeta_{13}})}{\zeta_{13}}\right)^{\delta'} \text{ (sub-exponential)} + T_{123\delta'}(z_1, z_2, z_3, p) \left(\frac{2(1 - \sqrt{1 - \zeta_{23}})}{\sqrt{\zeta_{13}\zeta_{23}}}\right)^{\delta'} \text{ (sub-exponential)},$$
(B.125)

where $T_{123\delta'}$ is the homogeneity factor defined in (3.36). We have checked that this result indeed solves the Casimir equations (B.111) in the δ' limit. Furthermore, if z_i 's are on the celestial sphere, all the cross ratios should satisfy $\zeta_{ij} \in (0, 1)$. Using (B.125), we find that for $\zeta_{ij} \in (0, 1), 2^{-\delta'}G^c_{\delta,j;\delta'}$ is always decaying exponentially at large δ' , and thus the condition (B.107) is true.

Finally, we consider the limit corresponding to (B.110), where we set $\delta' = \delta + 3 + 2k$ and take $\delta \rightarrow \infty$. For this limit, we must consider the integral

$$\int_{\substack{1>0>2\\0>0'>3}} D^{d-2} z_0 D^{d-2} z'_0 \frac{1}{(-2p \cdot z'_0)^{\delta}} \frac{1}{(-2z'_0 \cdot z_3)^{-\delta} (-2z_0 \cdot z_3)^{\delta}} \times \frac{1}{(-2z_0 \cdot z_1)^{-\frac{\delta}{2}} (-2z_0 \cdot z_2)^{-\frac{\delta}{2}} (-2z_1 \cdot z_2)^{\frac{\delta}{2}}}.$$
(B.126)

In the large δ limit, it turns out that the dominant contribution of the z'_0 integral comes from the top of the diamond 0 > 0' > 3. Hence, we should set $z'_0 = z_0$ and solve for the saddle point of the z_0 integral. After comparing the saddle point result

³When comparing the saddle point result to the collinear limit of the celestial block, one should note that in the main text we always set $-p^2 = 1$. However, in the frame (B.121), $-p^2$ depends on the cross ratios and is not equal to 1. Therefore, we should first factor out the homogeneity factors on both sides of (B.120) and just compare the remaining functions that depend on the cross ratios, which are independent of the choice of conformal frame.

⁴From (3.34), one can show that if we first take the collinear limit and then the $\delta' \to \infty$ limit, the leading behavior of G^c at large δ' is $(\max(\zeta_{13}, \zeta_{23}))^{\frac{\delta'}{2}} \times (\text{sub-exponential})$.

to the collinear limit of the celestial block,⁵ we obtain

$$G^{c}_{\delta \to \infty, j; \delta' = \delta + 3 + 2k \to \infty} \sim T_{123\delta'}(z_1, z_2, z_3, p) \left(\frac{2(1 - \sqrt{1 - \zeta_{12}})}{\sqrt{\zeta_{13}\zeta_{12}}}\right)^{\delta} \text{ (sub-exponential)}$$
(B.127)

The result is indeed a solution to the Casimir equations (B.111) in the corresponding limit. Moreover, (B.127) implies that the condition (B.110) holds for $\zeta_{ij} \in (0, 1)$.

⁵If we first take the collinear limit and then the limit corresponding to (B.110), the leading behavior of G^c is $\zeta_{12}^{\frac{\delta}{2}} \times ($ sub-exponential)

APPENDICES TO CHAPTER 4

C.1 Conventions for two-point and three-point structures

In this appendix, we summarize the conventions for the conformally-invariant structures we use in the main text.

For standard two-point and three-point structures, we use

$$\langle O(x_1, z_1) O(x_2, z_2) \rangle = \frac{(2z_1 \cdot I(x_{12}) \cdot z_2)^J}{x_{12}^{2\Delta}},$$
$$I^{\mu}{}_{\nu}(x) = \delta^{\mu}{}_{\nu} - \frac{2x^{\mu}x_{\nu}}{x^2},$$
(C.1)

and

$$\langle \phi_1(x_1)\phi_2(x_2)O(x_3, z_3)\rangle = \frac{(2z_3 \cdot x_{23}x_{13}^2 - 2z_3 \cdot x_{13}x_{23}^2)^J}{(x_{12}^2)^{\frac{\Delta_1 + \Delta_2 - \Delta + J}{2}}(x_{13}^2)^{\frac{\Delta_1 + \Delta + J - \Delta_2}{2}}(x_{23}^2)^{\frac{\Delta_2 + \Delta + J - \Delta_1}{2}}.$$
 (C.2)

In the embedding space, the standard structures can be written as

$$\langle O(X_1, Z_1) O(X_2, Z_2) \rangle = \frac{(2H_{12})^J}{X_{12}^{\Delta + J}},$$

$$\langle \phi_1(X_1) \phi_2(X_2) O(X_3, X_3) \rangle = \frac{(-2V_{3,12})^J}{X_{12}^{\frac{\Delta_1 + \Delta_2 - \Delta - J}{2}} X_{13}^{\frac{\Delta_1 + \Delta + J - \Delta_2}{2}} X_{23}^{\frac{\Delta_2 + \Delta + J - \Delta_1}{2}}, \quad (C.3)$$

where

$$X_{ij} = -2X_i \cdot X_j,$$

$$V_{i,jk} = \frac{Z_i \cdot X_j X_i \cdot X_k - Z_i \cdot X_k X_i \cdot X_j}{X_j \cdot X_k},$$

$$H_{ij} = -2(Z_i \cdot Z_j X_i \cdot X_j - Z_i \cdot X_j Z_j \cdot X_i).$$
 (C.4)

~

We also need the structures with more complicated representations. In the main text, the most complicated case will be when O_1, O_2 are symmetric traceless tensors with spins J_1, J_2 , and O has representation ρ whose Young diagram has three rows of length (J, j, \tilde{j}) . Our convention for the two-point structure is

$$\langle O(X_1, Z_1, W_1, \widetilde{W}_1) O(X_2, Z_2, W_2, \widetilde{W}_2) \rangle = \frac{(2H_{12})^{J-j} (2Y_{12})^{j-j} (2\widetilde{Y}_{12})^j}{X_{12}^{\Delta+J}}.$$
 (C.5)

For the additional two rows of the Young diagram, we introduce two null polarization vectors W_i , \widetilde{W}_i with the conditions

$$X_i \cdot W_i = Z_i \cdot W_i = X_i \cdot W_i = Z_i \cdot W_i = W_i \cdot W_i = 0,$$

$$W_i \sim W_i + \#X_i + \#Z_i,$$

$$\widetilde{W}_i \sim \widetilde{W}_i + \#X_i + \#Z_i + \#W_i.$$
(C.6)

The structures Y_{12} , \tilde{Y}_{12} are index contractions of the antisymmetrization of the embedding space vectors X_i , Z_i , W_i , \tilde{W}_i . Let us introduce the notation

$$[V_1, V_2, \dots, V_n] \cdot [W_1, W_2, \dots, W_n] \equiv \sum_{\sigma \in S_n} \operatorname{sgn}(\sigma) V_{1\alpha_1} V_{2\alpha_2} \dots V_{n\alpha_n} W_1^{\alpha_{\sigma(1)}} W_2^{\alpha_{\sigma(2)}} \dots W_n^{\alpha_{\sigma(n)}}$$
(C.7)

Then, the structures Y_{12} , \tilde{Y}_{12} are defined as

$$Y_{ij} = -2[X_i, Z_i, W_i] \cdot [X_j, Z_j, W_j],$$

$$\widetilde{Y}_{ij} = -2[X_i, Z_i, W_i, \widetilde{W}_i] \cdot [X_j, Z_j, W_j, \widetilde{W}_j].$$
(C.8)

One can check that subject to the conditions (C.6), (C.5) is the only conformallyinvariant structure with the correct homogeneity of the embedding space vectors. We have chosen the two-point convention such that (C.5) itself is Rindler positive.

For three-point structures, we have

$$\langle O_{1}(X_{1}, Z_{1})O_{2}(X_{2}, Z_{2})O(X_{3}, Z_{3}, W_{3}, \widetilde{W}_{3})\rangle^{(a)} = \frac{(-2V_{3,12})^{m_{3}}V_{1,23}^{m_{1}}V_{2,31}^{m_{2}}H_{12}^{n_{12}}H_{13}^{n_{13}}H_{23}^{n_{23}}(-2U_{3,12})^{k_{31}}(-2U_{3,21})^{k_{32}}(-2\widetilde{U}_{3,12})^{\widetilde{j}}}{X_{12}^{\frac{\Delta_{1}+J_{1}+\Delta_{2}+J_{2}-\Delta-J-j-\widetilde{j}}{2}}X_{13}^{\frac{\Delta_{1}+J_{1}+\Delta+J+j+\widetilde{j}-\Delta_{2}-J_{2}}{2}}X_{23}^{\frac{\Delta_{2}+J_{2}+\Delta+J+j+\widetilde{j}-\Delta_{1}-J_{1}}{2}}},$$
(C.9)

where the tensor structure is labeled by the nonnegative integers m_i , n_{ij} , k_{ij} , subject to the constraints

$$m_{1} + n_{12} + n_{13} + k_{31} = J_{1} - \tilde{j},$$

$$m_{2} + n_{12} + n_{23} + k_{32} = J_{2} - \tilde{j},$$

$$m_{3} + n_{13} + n_{23} = J - j,$$

$$k_{31} + k_{32} = j - \tilde{j}.$$
(C.10)

The structures $U_{i,jk}$, $\tilde{U}_{i,jk}$ are defined as

$$U_{i,jk} = \frac{X_{ij}}{X_{jk}} [X_i, Z_i, W_i] \cdot [X_j, Z_j, X_k],$$

$$\widetilde{U}_{i,jk} = \frac{X_{ij} X_{ik}}{X_{jk}} [X_i, Z_i, W_i, \widetilde{W}_i] \cdot [X_j, Z_j, X_k, Z_k].$$
 (C.11)

When writing the Lorentzian shadow representation of the block (e.g., (4.43)), we need to use the three-point structures with absolute values. Also, note that the two structures in (4.43) have different causality configurations. Their explicit expressions are given by

$$\begin{split} |\langle \phi_1(x_1)\phi_2(x_2)O(x_5,z_5)\rangle| &= \frac{(2z_5 \cdot x_{25}x_{15}^2 - 2z_5 \cdot x_{15}x_{25}^2)^J}{(-x_{12}^2)^{\frac{\Delta_1 + \Delta_2 - \Delta_+ J}{2}}(x_{15}^2)^{\frac{\Delta_+ J + \Delta_1 - \Delta_2}{2}}(x_{25}^2)^{\frac{\Delta_+ J + \Delta_2 - \Delta_1}{2}}, \quad 1 > 2, 5 \approx 1, 2 \\ |\langle \phi_3(x_3)\phi_4(x_4)O(x_5,z_5)\rangle| &= \frac{(2z_5 \cdot x_{45}x_{35}^2 - 2z_5 \cdot x_{35}x_{45}^2)^J}{(-x_{43}^2)^{\frac{\Delta_4 + \Delta_3 - \Delta_+ J}{2}}(-x_{45}^2)^{\frac{\Delta_+ J + \Delta_4 - \Delta_3}{2}}(-x_{35}^2)^{\frac{\Delta_+ J + \Delta_3 - \Delta_4}{2}}, \quad 4 > 5 > 3. \end{split}$$
(C.12)

From the expression of $|\langle \phi_1(x_1)\phi_2(x_2)O(x_5, z_5)\rangle|$, one can explicitly verify the identity (4.47).

In the definition of the functional (4.93), we introduce a celestial three-point structure. This structure is a standard three-point structure in a Euclidean (d - 2)dimensional CFT, where z_i are viewed as the embedding space coordinates. Therefore, we can get this structure by taking the *d*-dimensional three-point structure in (C.3) and make the replacement $X_i \rightarrow z_i, Z_i \rightarrow w_i$. This gives

$$\langle \mathcal{P}_{\delta_1}(z_1) \mathcal{P}_{\delta_2}(z_2) \mathcal{P}_{\delta,j}(z,w) \rangle = \frac{(4w \cdot z_1 z \cdot z_2 - 4w \cdot z_2 z \cdot z_1)^j}{(-2z_1 \cdot z)^{\frac{\delta_1 + \delta_2 - \delta_1 j}{2}} (-2z_2 \cdot z)^{\frac{\delta_2 + \delta_1 - \delta_1 j}{2}} (-2z_1 \cdot z_2)^{\frac{\delta_1 + \delta_2 - \delta_1 j}{2}}}$$
(C.13)

Finally, in the functional we also use a dual structure $\langle 0|O_4 \mathbf{L}[O]O_{2^+}|0\rangle^{-1}$, which should be a continuous-spin structure. In particular, we are interested in the structure with the causality constraint 4 > 0 > 2. We define the standard continuous-spin structure with this configuration as

$$\langle 0|\phi_4(x_4)O(x_0, z_0)\phi_2(x_2^+)|0\rangle = \frac{(2z_0 \cdot x_{40}x_{20}^2 - 2z_0 \cdot x_{20}x_{40}^2)^J}{(-x_{24}^2)^{\frac{\Delta_2 + \Delta_4 - \Delta + J}{2}}(-x_{02}^2)^{\frac{\Delta + J + \Delta_2 - \Delta_4}{2}}(-x_{04}^2)^{\frac{\Delta + J + \Delta_4 - \Delta_2}{2}}, \quad 4 > 0 > 2.$$
(C.14)

In the embedding space, it is given by

$$\langle 0|\phi_4(X_4)O(X_0, Z_0)\phi_2(X_2^+)|0\rangle = \frac{(-2V_{0,42}(-X_{24}))^J}{(-X_{24})^{\frac{\Lambda_2+\Lambda_4-\Lambda+J}{2}}(-X_{02})^{\frac{\Lambda+J+\Lambda_2-\Lambda_4}{2}}(-X_{04})^{\frac{\Lambda+J+\Lambda_4-\Lambda_2}{2}}, \quad 4 > 0 > 2.$$
(C.15)

Note that the combination $-2V_{0,42}(-X_{24})$ is always positive due to the causality constraint 4 > 0 > 2. The actual dual structure will be the above structure with the replacement $\Delta \rightarrow J + d - 1, J \rightarrow \Delta - d + 1, \Delta_i \rightarrow \widetilde{\Delta}_i$, multiplied by a computable prefactor. In the scalar case, it is given by (4.37). We discuss the spinning case in more detail in appendix C.4.

C.2 Heavy action formula from the large *v* limit

In the main text of this paper we have taken the large ν limit of the kernel and large Δ limit of the block at the same time. However, in this appendix we compute the large ν limit independently. Taking the large ν limit of the kernel alone will slightly differ from taking Δ and ν limits simultaneously. Recall that when we perform the gauge fixing, we also fix a fifth point x_5 coming from the conformal block we want to act the functional on. If we just study the action of the functional on a general four-point function, we can again use conformal symmetry to fix $x_1 + x_3$ to the origin and $x_{2^+} + x_{4^-}$ to spatial infinity, and since we do not have a fifth point, the stabilizer group is generated by Lorentz transformations and dilatations. We can then use Lorentz transformation to fix all four points to be on the same plane. After performing the large ν analysis (which will be done below), one can then obtain a kernel that only depends on the cross ratios (see (C.34)). Therefore, we can undo the gauge fixing and go back the the bulk-point gauge fixing again, and then the rest of the calculation is the same as [13].

We now want to take the large ν limit of

$$\begin{split} \Psi_{k,\nu}^{+}[\mathcal{G}] &= 4A_{k,\nu} \int_{4>3>0>1>2} \frac{d^{d}x_{1}d^{d}x_{2}d^{d}x_{3}d^{d}x_{4}d^{d}x_{0}D^{d-2}z_{0}}{\operatorname{vol}(\widetilde{\mathrm{SO}}(d,2))} \langle 0|\phi_{4}(x_{4})\mathbf{L}[\mathcal{O}](x_{0},z_{0})\phi_{2}(x_{2}^{+})|0\rangle^{-1} \\ \langle \widetilde{\mathcal{P}}_{\delta_{1}}(x_{1^{+}0})\widetilde{\mathcal{P}}_{\delta_{3}}(x_{30})\mathcal{P}_{\delta}(z_{0})\rangle\delta(x_{1^{+}0}^{2})\delta(x_{30}^{2})\theta(x_{1^{+}0})\theta(x_{30})\langle \Omega|[\phi_{4}(x_{4}),\phi_{3}(x_{3})][\phi_{1}(x_{1}^{+}),\phi_{2}(x_{2}^{+})]|\Omega\rangle. \\ (C.16) \end{split}$$

The only ν dependence is in dual and celestial structures, and so we need only focus on

$$\int d^{d}x_{0}D^{d-2}z\langle 0|\phi_{4}(x_{4})\mathbf{L}[O](x_{0},z_{0})\phi_{2}(x_{2}^{+})|0\rangle^{-1}\delta(x_{1+0}^{2})\delta(x_{30}^{2}) \\ \times \langle \mathcal{P}_{\delta}(z)\mathcal{P}_{\widetilde{\delta}_{1}}(x_{1+0})\mathcal{P}_{\widetilde{\delta}_{3}}(x_{30})\rangle(-x_{13}^{2})^{\widetilde{\Delta}_{\phi}}(-x_{24}^{2})^{\widetilde{\Delta}_{\phi}},$$
(C.17)

which we can rewrite using the explicit expression for the dual structure (4.37) which

gives us

$$\int d^{d}x_{0}D^{d-2}z \times \langle \mathcal{P}_{\delta}(z)\mathcal{P}_{\tilde{\delta}_{1}}(x_{1+0})\mathcal{P}_{\tilde{\delta}_{3}}(x_{30})\rangle(-x_{13}^{2})^{\tilde{\Delta}_{\phi}}(-x_{24}^{2})^{\tilde{\Delta}_{\phi}}\delta(x_{1+0}^{2})\delta(x_{30}^{2}) \\ \frac{(2z_{0}\cdot x_{40}x_{20}^{2} - 2z_{0}\cdot x_{20}x_{40}^{2})^{\Delta-d+1}}{(-x_{24}^{2})^{\frac{\tilde{\Delta}_{2}+\tilde{\Delta}_{4}+\Delta-J-2d+2}{2}}(x_{02}^{2})^{\frac{\tilde{\Delta}_{2}+\Delta+J-\tilde{\Delta}_{4}}{2}}(x_{02}^{2})^{\frac{\tilde{\Delta}_{4}+\Delta+J-\tilde{\Delta}_{2}}{2}},$$
(C.18)

where we have dropped the overall coefficient in (4.37). It will be useful to notice that the dual structure has the form $\langle \tilde{\phi}(x_4) O^F(x_0, z_0) \tilde{\phi}(x_2) \rangle$, which is a three-point function of two scalars with shadow dimensions $\tilde{\Delta}_{\phi} = d - \Delta_{\phi}$, and a third operator with $(\Delta_F, J_F) = (J + d - 1, \Delta - d + 1)$, where J and Δ are the dimension and spin of the light transformed operator. In particular we will focus on the limit $J_F = 2 - d + \delta = \frac{2-d}{2} + i\nu, \nu \to \infty$.

From the analysis in the cross-ratio space [13], we know that in the bulk-point limit x and y will get fixed to saddle points of the form

$$x = (u_x, v_x, 0), \qquad y = (u_y, v_y, 0),$$
 (C.19)

and therefore we can just study the large v limit for this particular x and y using the gauge fixing procedure described above.

Let us choose the coordinates of x_0 and z to be $x_0 = (u, v, \vec{y}_0), z = (1, \vec{y}_z^2, \vec{y}_z)$. In the integrand, the J_F dependent factors are (we replace δ with $J_F + d - 2$ as well as Δ with $J_F + d - 1$)

$$\langle 0 | \widetilde{\phi}_{4}(x_{4}) O^{F}(x_{0}, z) \widetilde{\phi}_{2}(x_{2}^{+}) | 0 \rangle \langle \mathcal{P}_{\delta}(z) \mathcal{P}_{\widetilde{\delta}_{1}}(x_{1+0}) \mathcal{P}_{\widetilde{\delta}_{3}}(x_{30}) \rangle$$

$$= \frac{(-2V_{0,24})^{J_{F}}}{(x_{2+4}^{2})^{\frac{2\widetilde{\Delta}_{\phi} - d + 2 - J_{F}}{2}} (-x_{40}^{2})^{\frac{d - 2 + J_{F}}{2}} (x_{2+0}^{2})^{\frac{d - 2 + J_{F}}{2}}} \times \frac{1}{(-2z \cdot x_{1+0})^{\frac{d - 2 + J_{F}}{2}} (-2z \cdot x_{30})^{\frac{d - 2 + J_{F}}{2}} (-2x_{1+0} \cdot x_{30})^{\frac{2\widetilde{\delta}_{\phi} - d + 2 - J_{F}}{2}}}.$$
(C.20)

Note that we have the delta functions $\delta(x_{1+0}^2)\delta(x_{30}^2)$, and therefore $-2x_{1+0} \cdot x_{30} = (x_{1+0} - x_{30})^2 = x_{1+3}^2$.

The two delta functions can be written as

$$\delta(x_{1+0}^2)\delta(x_{30}^2) = \delta((u+u_y)(v+v_y) - \vec{y}_0^2)\delta(-(u-u_y)(v-v_y) + \vec{y}_0^2).$$
(C.21)

In terms of \tilde{t}_0, \tilde{x}_0 defined in (4.59) (in this case we have $\vec{y}_{\perp} = 0$), the delta functions fix $\tilde{t}_0 = 0$ and $r_0 \equiv |\vec{y}_0| = (u_y v_y (1 - \tilde{x}_0^2))^{\frac{1}{2}}$. The Jacobian relating \tilde{t}_0, \tilde{x}_0 and u, v is

given by $dudv = 2u_y v_y d\tilde{t_0} d\tilde{x_0}$. This gives (for any function $f(u, v, \vec{y_0})$)

$$\frac{1}{2} \int du dv d^{d-2} \vec{y}_0 \,\delta(-(u-u_y)(v-v_y) + \vec{y}_0^2) \delta((u+u_y)(v+v_y) - \vec{y}_0^2) f(u,v,\vec{y}_0)$$

= $\int d\tilde{x}_0 d\Omega_{\vec{n}_0} \frac{(u_y v_y (1-\tilde{x}_0^2))^{\frac{d-4}{2}}}{8} f(\tilde{t}_0 = 0, \tilde{x}_0, r_0 = (u_y v_y (1-\tilde{x}_0^2))^{\frac{1}{2}}, \Omega_{\vec{n}_0}), \quad (C.22)$

where \vec{n}_0 is the unit vector in the \vec{y}_0 direction.

After removing the delta functions, (C.17) becomes

$$\int d\tilde{x}_{0} d\Omega_{\vec{n}_{0}} d^{d-2} \vec{y}_{z} \frac{(u_{y}v_{y}(1-\tilde{x}_{0}^{2}))^{\frac{d-4}{2}}}{8} \frac{2^{-J_{F}}}{(-x_{2^{2}+4^{-}}^{2})^{\frac{2\tilde{\Delta}_{\phi}-d+2-J_{F}}{2}} (x_{4^{-}0}^{2})^{\frac{d-2}{2}} (x_{2^{+}0}^{2})^{\frac{d-2}{2}}} \times \frac{1}{(-2z \cdot x_{1^{+}0})^{\frac{d-2}{2}} (-2z \cdot x_{30})^{\frac{d-2}{2}} (-2x_{1^{+}0} \cdot x_{30})^{\frac{2\tilde{\Delta}_{\phi}-d+2-J_{F}}{2}}} e^{J_{F} \times h(\tilde{x}_{0}, \vec{y}_{z}, \Omega_{\vec{n}_{0}})} (4u_{y}v_{y})^{(d-\Delta_{\phi})} (\frac{4}{u_{x}v_{x}})^{(d-\Delta_{\phi})}},$$
(C.23)

where the function $h(\tilde{x}_0, \vec{y}_z, \Omega_{\vec{n}_0})$ is given by

$$h(\tilde{x}_{0}, \vec{y}_{z}, \Omega_{\vec{n}_{0}}) = \log\left(\frac{1}{u_{x}} + \frac{\vec{y}_{z}^{2}}{v_{x}} - u_{y}v_{y}(1 - \tilde{x}_{0}^{2})(v_{x} + u_{x}\vec{y}_{z}^{2}) - u_{x}v_{y}^{2}\tilde{x}_{0}^{2} - u_{y}^{2}v_{x}\tilde{x}_{0}^{2}\vec{y}_{z}^{2} + 2(u_{y}v_{x} - v_{y}u_{x})\tilde{x}_{0}\vec{y}_{0} \cdot \vec{y}_{z} - \frac{1}{2}\log\left((v_{y}(1 - \tilde{x}_{0}) - 2\vec{y}_{0} \cdot \vec{y}_{z} + u_{y}(1 + \tilde{x}_{0})\vec{y}_{z}^{2})(v_{y}(1 + \tilde{x}_{0}) + 2\vec{y}_{0} \cdot \vec{y}_{z}u_{y}(1 - \tilde{x}_{0})\vec{y}_{z}^{2})\right) - \frac{1}{2}\log\left(\frac{(1 - u_{y}v_{y}u_{x}v_{x} - (u_{y}v_{x} - v_{y}u_{x})\tilde{x}_{0})(1 - u_{y}v_{y}u_{x}v_{x} + (u_{y}v_{x} - u_{x}v_{y})(1 - \tilde{x}_{0}))}{u_{x}^{2}v_{x}^{2}}\right).$$
(C.24)

We find that at large J_F , the integral has two saddle loci: one at $\tilde{x}_0 = 0$, $\vec{y}_z = 0$ and one at $\tilde{x}_0 = 0$, $\vec{y}_z = \infty$. Namely,

$$\frac{\partial}{\partial \tilde{x}_{0}} h(\tilde{x}_{0}, \vec{y}_{z}, \Omega_{\vec{n}_{0}}) \Big|_{\tilde{x}_{0} \to 0, \vec{y}_{z} \to 0} = \frac{\partial}{\partial \vec{y}_{z}} h(\tilde{x}_{0}, \vec{y}_{z}, \Omega_{\vec{n}_{0}}) \Big|_{\tilde{x}_{0} \to 0, \vec{y}_{z} \to 0} = 0$$
$$\frac{\partial}{\partial \tilde{x}_{0}} h(\tilde{x}_{0}, \vec{y}_{z}, \Omega_{\vec{n}_{0}}) \Big|_{\tilde{x}_{0} \to 0, \vec{y}_{z} \to \infty} = \frac{\partial}{\partial \vec{y}_{z}} h(\tilde{x}_{0}, \vec{y}_{z}, \Omega_{\vec{n}_{0}}) \Big|_{\tilde{x}_{0} \to 0, \vec{y}_{z} \to \infty} = 0. \quad (C.25)$$

Let us first consider the locus at $\tilde{x}_0 = \vec{y}_z = 0$. We will see what the other saddle should give shortly. Expanding $h(\tilde{x}_0, \vec{y}_z, \Omega_{\vec{n}_0})$ around $\tilde{x}_0 = \vec{y}_z = 0$, we get

$$h(\tilde{x}_{0}, \vec{y}_{z}, \Omega_{\vec{n}_{0}}) \approx -\log\left(\frac{v_{y}}{v_{x}}\right) + \frac{(1 + u_{y}^{2}v_{x}^{2} - 2u_{x}u_{y}v_{x}v_{y})(1 - u_{x}^{2}v_{y}^{2})}{2(1 - u_{y}v_{y}u_{x}v_{x})^{2}}\tilde{x}_{0}^{2}$$
$$+ \frac{2(1 - u_{x}^{2}v_{y}^{2})}{v_{y}(1 - u_{y}v_{y}u_{x}v_{x})}\tilde{x}_{0}\vec{y}_{0} \cdot \vec{y}_{z} + \frac{2v_{x}(\vec{y}_{0} \cdot \vec{y}_{z})^{2} - v_{y}(u_{y}v_{x} - v_{y}u_{x})\vec{y}_{z}^{2}}{v_{x}v_{y}^{2}} + \dots$$
$$= -\log\left(\frac{v_{y}}{v_{x}}\right) + \frac{1}{2}\left(\tilde{x}_{0} \quad \vec{y}_{z}\right)M_{\vec{y}_{0}}\left(\frac{\tilde{x}_{0}}{\vec{y}_{z}}\right). \tag{C.26}$$

Alternatively, one can also choose to not integrate \tilde{t}_0 , $|\vec{y}_0|$ and keep the delta functions. Then the function *h* will also depend on \tilde{t}_0 and $|\vec{y}_0|$, and one can show that there are saddle loci at $\tilde{t}_0 = \tilde{x}_0 = 0$, $\vec{y}_z = 0$, ∞ . We can then expand *h* around the saddle locus and use the delta functions to fix \tilde{t}_0 and $|\vec{y}_0|$. This calculation should give the same result.

Now let us compute the saddle integral in (C.23). By rotational invariance, the determinant of the Hessian $M_{\vec{y}_0}$ in (C.26) does not depend on the direction of \vec{y}_0 , so we can pick a direction that makes computing the determinant simple. For example, $\vec{y}_0 = |\vec{y}_0|(1, 0, ..., 0)$. We then find

$$\operatorname{Det} M_{\vec{y}_0} = \frac{2^{d-2} (1 - u_y^2 v_x^2) (1 - v_y^2 u_x^2) (v_y u_x - u_y v_x)^{d-2}}{(v_x v_y)^{d-2} (1 - u_y v_y u_x v_x)^2}.$$
 (C.27)

One also has to check the sign of each eigenvalue of $\text{Det}M_{\vec{y}_0}$ in order to get the correct phase in the saddle integral. We find that for $v_y u_x - u_y v_x > 0$, all eigenvalues are positive, and for $v_y u_x - u_y v_x < 0$, one eigenvalue is positive and the other d - 2 eigenvalues are negative. Both cases turn out to give the same result, so let us assume $v_y u_x - u_y v_x > 0$ here. The saddle integral in (C.23) is then given by

$$\int d\widetilde{x}_0 d^{d-2} \vec{y}_z (\cdots) e^{J_F \times h(\widetilde{x}_0, \vec{y}_z, \Omega_{\vec{n}_0})}$$

$$\approx (\cdots) |_{\widetilde{x}_0 \to 0, \vec{y}_z \to 0} \times \left(\frac{v_x}{v_y}\right)^{J_F} e^{\frac{i\pi}{2}(d-1)} J_F^{-\frac{d-1}{2}} \sqrt{\frac{(2\pi)^{d-1}}{\operatorname{Det} M_{\vec{y}_0}}} + \dots, \qquad (C.28)$$

where (\cdots) are the other factors in the integrand of (C.23), and ... are subleading terms at large J_F . Evaluating the remaining factors on the saddle locus $\tilde{x}_0 = 0$, $\vec{y}_z = 0$, we obtain

$$(4u_{y}v_{y})^{(d-\Delta_{\phi})} (\frac{4}{u_{x}v_{x}})^{(d-\Delta_{\phi})} \int d\Omega_{\vec{n}_{0}} \frac{(u_{y}v_{y})^{\frac{d-4}{2}}}{8} \left(\frac{v_{x}}{v_{y}}\right)^{J_{F}} e^{\frac{i\pi}{2}(d-1)} J_{F}^{-\frac{d-1}{2}} \sqrt{\frac{(2\pi)^{d-1}}{\text{Det}M_{\vec{y}_{0}}}} \times \frac{2^{-J_{F}}}{(-x_{2^{+}4^{-}})^{\frac{2\tilde{\Delta}_{\phi}-d+2-J_{F}}{2}} (x_{4^{-}0}^{2})^{\frac{d-2}{2}} (x_{2^{+}0}^{2})^{\frac{d-2}{2}}} (-2z \cdot x_{1^{+}0})^{\frac{d-2}{2}} (-2z \cdot x_{30})^{\frac{d-2}{2}} (-2x_{1^{+}0} \cdot x_{30})^{\frac{2\tilde{\delta}_{\phi}-d+2-J_{F}}{2}}} |_{\vec{x}_{0}\to 0, \vec{y}_{z}\to 0}} = 2^{-\frac{9}{2}+2d+J_{F}} \pi^{\frac{d-1}{2}} \operatorname{vol}(S^{d-3}) e^{\frac{i\pi}{2}(d-1)} J_{F}^{-\frac{d-1}{2}}} \times \frac{(v_{y}u_{x})^{\frac{d-2}{2}} (u_{y}v_{x})^{\frac{J_{F}}{2}+d-2} (v_{y}u_{x}-u_{y}v_{x})^{\frac{2-d}{2}} (1-u_{y}v_{y}u_{x}v_{x})^{3-d}}}{\sqrt{(1-u_{y}^{2}v_{x}^{2})(1-v_{y}^{2}u_{x}^{2})}}.$$

$$(C.29)$$

We can write the result in terms of the conformally-invariant cross-ratios r and η defined as

$$u' = \frac{(1 - 2r\eta + r^2)^2}{16r^2}, \quad v' = \frac{(1 + 2r\eta + r^2)^2}{16r^2}, \quad (C.30)$$

where u', v' are defined in (4.31). In terms of r, η , the *u*-channel Regge limit corresponds to $r \to 0$ with fixed η . In our gauge fixing, they are given by

$$r = |x||y| = (u_y v_y u_x v_x)^{\frac{1}{2}}, \qquad \eta = -\frac{x \cdot y}{|x||y|} = \frac{v_y u_x + u_y v_x}{2(u_y v_y u_x v_x)^{\frac{1}{2}}}.$$
 (C.31)

Then, in terms of r, η , (C.29) becomes

$$2^{-\frac{9}{2}+2d+J_F}\pi^{\frac{d-1}{2}}\operatorname{vol}(S^{d-3})e^{\frac{i\pi}{4}(d-1)}v^{-\frac{d-1}{2}}\frac{(1-r^2)^{3-d}r^{d-2}w^{-i\nu}(w-\frac{1}{w})^{\frac{2-d}{2}}}{((1+r^2)^2-4r^2\eta^2)^{\frac{1}{2}}},\quad(C.32)$$

where we have used $J_F = \frac{2-d}{2} + i\nu$ and $\eta = \frac{w+1/w}{2}$. One can then notice that the leading large ν behavior is similar to

$$\mathcal{P}_{\frac{2-d}{2}+i\nu}\left(\frac{w+1/w}{2}\right) \sim \frac{\Gamma(d-2)}{\Gamma(\frac{d-2}{2})} \frac{\nu^{\frac{2-d}{2}}}{(w-1/w)^{\frac{d-2}{2}}} \left(e^{-\frac{i\pi}{4}(d-2)}w^{i\nu} + e^{\frac{i\pi}{4}(d-2)}w^{-i\nu}\right), \quad \nu \to \infty, |w| > 1$$
(C.33)

In particular, the second term agrees with (C.32), and the first term comes from the other saddle locus. Therefore, after including both saddle loci, the integral (C.23) becomes

$$2^{-\frac{9}{2}+2d+J_F}\pi^{\frac{d-1}{2}}e^{\frac{i\pi}{4}}\operatorname{vol}(S^{d-3})\nu^{-\frac{1}{2}}\frac{\Gamma(\frac{d-2}{2})}{\Gamma(d-2)}\frac{(1-r^2)^{3-d}r^{d-2}}{((1+r^2)^2-4r^2\eta^2)^{\frac{1}{2}}}\mathcal{P}_{\frac{2-d}{2}+i\nu}(\eta).$$
(C.34)

The expression we start with, (C.17), is essentially the shadow representation of the conformal block $G_{\frac{d}{2}+i\nu,J=-1}^{\tilde{\Delta}_{\phi}}$, where one of the three-point structures becomes a celestial structure and delta functions due to setting J = -1 [77]. What we have shown here is that its large ν limit is given by (C.34). Although the same result can be obtained in the cross-ratio space using the Casimir equation [11], we believe that the above calculation from saddle point can be more straightforwardly generalized to spinning operators. Since $G_{\frac{d}{2}+i\nu,J=-1}^{\tilde{\Delta}_{\phi}}$ is exactly the kernel appearing in the Lorentzian inversion formula for $C(\Delta = \frac{d}{2} + i\nu, J = -1)$, the calculation here could also be helpful for understanding the OPE data at large Δ [187].

Finally, to reproduce the heavy action formula, we can take the conformally-invariant expression (C.34) and plug it back in (C.16). Similar to the main text, we can then

study its action on conformal block and choose the gauge fixing (4.49), which introduces the Faddeev-Popov factor (4.50). This gives (4.51) with the kernel part replaced with (C.34). Combining with the subtraction factors (4.87), we find that we get an integral of the form

$$\int d^{d}x d^{d}y \frac{r^{k+d-1}(1-r^{4})\eta}{((1+r^{2})^{2}-4r^{2}\eta^{2})^{\frac{k+1}{2}}} \times \frac{1}{(-x^{2})^{2d}} \frac{1}{(-x^{2}_{12})^{\widetilde{\Delta}_{\phi}}(-x^{2}_{34})^{\widetilde{\Delta}_{\phi}}} \left| \langle \phi_{1}, \phi_{2}, O^{\mu_{1}\cdots\mu_{J}}(x_{5}) \rangle \right| \left| \langle \widetilde{O}^{\dagger}_{\mu_{1}\cdots\mu_{J}}(x_{5}) \phi_{3} \phi_{4} \rangle \right|_{\text{gauge-fixed}}$$
(C.35)

The only difference between this integral and the one considered in [13] (see (3.25) and (3.37)) is a factor of η , which becomes 1 in the bulk-point limit. One can further check that the overall factors also agree. Hence, the calculation in [13] implies that we recover the same heavy action formula (4.92).

C.3 Details on matching partial waves

In this appendix, we give some more details on the partial wave discussion in section 4.4.2.

C.3.1 Partial waves in monomial basis

In the main text, we define the vertices $v(n, e_1, e_2)$ directly from evaluating the CFT three-point structures at the saddle and applying the polarization map. From the flat space perspective, it is more convenient to define the vertices as monomials, which can be easily expressed using Young tableaux, as given in [110]. Furthermore, the bootstrap calculation done in [110] also uses the monomial basis (up to a Gram-Schmidt procedure to convert it into a orthonormal basis). The dictionary found in the main text tells us that we can take the flat space functional found in [110] and construct a positive CFT functional with positive action on heavy blocks. If we want to further study the OPE coefficients in the bootstrap calculation, we will need to know the relation between the two different bases.

One can find the relation between the two bases by explicitly evaluating the CFT three-point structures at the saddle for all the exchanged representations ρ in the photon and graviton case (see [110] for a complete list). It turns out that there is a clear map between the building blocks of CFT three-point tensor structures and the building blocks of the monomial basis, given by columns of the Young tableaux. We find that the map between the structures of $\langle O_1 O_2 O_5 \rangle$ and $v(n, e_1, e_2)$ is given

where we use the notation (C.7) for the CFT structures. The Young tableaux columns represent the antisymmetrization of the vectors in the boxes (see [110] for the precise definition). For example,

$$\frac{e_1}{n} = e_1 \cdot w_1 n \cdot w_2 - n \cdot w_1 e_1 \cdot w_2,$$
(C.37)

where w_1, w_2 are the polarizations of the vertex. Both sides of the map agree up to an overall factor after one evaluates the CFT structure at the scattering-crystal configuration (4.81) and applies the polarization map (4.133).

The map given in (C.36) allows us to unambiguously relate the structure labels of the two bases, and we find that the partial waves computed in the two bases are related by

$$\pi_{\rho,(a'b')}^{\text{CFT}} = 2^{2J-2j-n_{15}^{a'}-n_{25}^{a'}-n_{36}^{b'}-n_{46}^{b'}} (-1)^{n_{12}^{a'}+n_{25}^{a'}+n_{34}^{b'}+n_{46}^{b'}} \pi_{\rho,(a'b')}^{\text{tableaux}}, \tag{C.38}$$

where $n_{15}^{a'}$ counts the number of H_{15} of the (a') tensor structure, and other $n_{ij}^{a'}$, $n_{ij}^{b'}$ are defined similarly (see (C.9)). Note that the above relation is true for each a', b'. Alternatively, one can think of the factor relating the two partial waves as two diagonal matrices that rescale the vertices.

We also see that this change of basis preserves the positivity of the partial wave actions, so we can reuse the positive functional in the monomial basis. In the forward limit, where n = n', $e_3 = e_1^*$, $e_4 = e_2^*$, the partial wave in the monomial basis is positive (or positive semi-definite for multiple tensor structures) by construction. In the CFT basis, (C.38) implies that the partial wave is also positive. (Note that if we instead define $\overline{v}(n', e_3, e_4)$ in (4.139), then the partial wave wouldn't be positive in the forward limit.)

C.3.2 Polarization map and CRT symmetry

We now explain why the two definitions of vertices (4.134), (4.139) are consistent with each other. In particular, we will show that by using the definition of $v(n, e_1, e_2)$ and the CRT symmetry, we can recover the definition of $\overline{v}(-n', e_3, e_4)$.

by

For any CFT, there is an anti-unitary CRT symmetry J satisfying $J^2 = 1$, and it acts on local operators as (assuming the operator is bosonic)

$$JO(x,z)J^{-1} = O^{\dagger}(\overline{x},\overline{z}^*), \qquad (C.39)$$

where $\overline{x} = (-x^0, -x^1, x^2, ...)$ (and similarly for \overline{z}^*) is a Rindler reflection. Since J is anti-unitary, any CFT correlation function should satisfy

$$\langle O_1(x_1, z_1) \cdots O_n(x_n, z_n) \rangle = \langle O_1^{\dagger}(\overline{x}_1, \overline{z}_1^*) \cdots O_n^{\dagger}(\overline{x}_n, \overline{z}_n^*) \rangle^*, \qquad (C.40)$$

where we have applied a CRT to each operator.

More generally, as explained in [34], for any spacelike points *A* and *B*, one can define two Rindler wedges $B > x > A^-$ and $A > x > B^-$. Then, there exists a Rindler conjugation J_{AB} that exchanges the two wedges. Explicitly,

$$J_{AB}(X) = X - 2\frac{X \cdot X_A}{X_A \cdot X_B} X_B - 2\frac{X \cdot X_B}{X_A \cdot X_B} X_A.$$
 (C.41)

It turns out that if we choose A to be the future infinity and B^- to be the origin, under the corresponding Rindler conjugation J_{AB} we have $3 \rightarrow 1^+, 4 \rightarrow 2^+, 6^+ \rightarrow 5$ in the scattering crystal configuration (4.81). Therefore, using this J_{AB} , one might expect

$$\left. \langle 0|O^{\dagger}(x_{6^{+}}, z_{6})O_{4}(x_{4}, z_{4})O_{3}(x_{3}, z_{3})|0\rangle \right|_{\text{saddle}} \stackrel{?}{=} \left. \langle 0|O(x_{5}, \overline{z}_{6}^{*})O_{4}^{\dagger}(x_{2^{+}}, \overline{z}_{4}^{*})O_{3}^{\dagger}(x_{1^{+}}, \overline{z}_{3}^{*})|0\rangle^{*} \right|_{\text{saddle}}$$
(C.42)

This is however too fast. A funny feature of our saddle configuration is that although the points $x_{1,2,3,4}$ are timelike, they all have purely imaginary spatial components. The correct way to think about this should be we first consider e.g. $x_3 = (u_y, v_y, 0)$ for real u_y, v_y , and at the end analytically continue to $u_y \rightarrow -i\frac{m-\sqrt{m^2-v^2}}{v}, v_y \rightarrow i\frac{m-\sqrt{m^2-v^2}}{v}$. To take into account the fact that there are imaginary spatial components, we should apply an additional reflection to them, which can then be shifted to act on the polarizations. In summary, the correct CRT relation becomes

$$\langle 0|O^{\dagger}(x_{6^{+}}, z_{6})O_{4}(x_{4}, z_{4})O_{3}(x_{3}, z_{3})|0\rangle \Big|_{\text{saddle}}$$

= $\langle 0|O(x_{5}, I_{e} \cdot z_{6}^{*})O_{4}^{\dagger}(x_{2^{+}}, I_{e} \cdot z_{4}^{*})O_{3}^{\dagger}(x_{1^{+}}, I_{e} \cdot z_{3}^{*})|0\rangle^{*} \Big|_{\text{saddle}},$ (C.43)

where the polarizations are reflected only in the time direction. Since (C.43) is simply a statement about the CFT structures at the saddle configuration, we can explicitly verify that it is true for different representations ρ . Using the CRT relation (C.43), we can write

$$(-x_{34}^2)^{\frac{\Delta_3 + \Delta_4}{2}} \left(\frac{x_{36}^2}{x_{46}^2} \right)^{\frac{\Delta_3 - \Delta_4}{2}} I_e^{\rho} \left. \frac{\langle 0|O^{\dagger}(x_{6^+}, z_6)[O_4(x_4, z_4), O_3(x_3, z_3)]|0\rangle_{(b)}}{2i(\sin(\pi^{\frac{\tilde{\tau}_{\rho} - \Delta_3 - J_3 - \Delta_4 - J_4}}))} \right|_{\text{saddle}}$$

$$= (-x_{12}^2)^{\frac{\Delta_3 + \Delta_4}{2}} \left(\frac{x_{15}^2}{x_{25}^2} \right)^{\frac{\Delta_3 - \Delta_4}{2}} \frac{\langle 0|[O_3(x_{1^+}, z_3^*), O_4(x_{2^+}, z_4^*)]O(x_5, z_6^*)|0\rangle_{(b)}^*}{-2i(\sin(\pi^{\frac{\tilde{\tau}_{\rho} - \Delta_3 - J_3 - \Delta_4 - J_4}{2}}))} \right|_{\text{saddle}} .$$

$$(C.44)$$

We assume there is only one polarization z_6 for simplicity, but the argument for more polarizations \vec{w}_6 is the same. We have also imposed conservation to set the time components of z_3 , z_4 to zero. So, $I_e \cdot z_3 = z_3$, $I_e \cdot z_4 = z_4$, and the reflection tensor I_e^{ρ} cancels with the I_e acting on z_6 . The additional minus sign in the denominator comes from changing the operator order in the commutator.

Since the positions are now at x_{1^+}, x_{2^+}, x_5 , we can use the definition (4.134) for $v(n, e_1, e_2)$ (although now it is a vertex for external particles with spin J_3, J_4). We can then write (C.44) as

$$(-x_{34}^2)^{\frac{\lambda_3+\lambda_4}{2}} \left(\frac{x_{36}^2}{x_{46}^2}\right)^{\frac{\lambda_3-\lambda_4}{2}} I_e^{\rho} \left. \frac{\langle 0|O^{\dagger}(x_{6^+},z_6)[O_4(x_4,z_4),O_3(x_3,z_3)]|0\rangle_{(b)}}{2i(\sin(\pi\frac{\tilde{\tau}_{\rho}-\lambda_3-J_3-\lambda_4-J_4}{2}))} \right|_{\text{saddle}}$$

= $v(n, z_3^{x*}n_{\perp} + \vec{z}_{3\perp}^*, -z_4^{x*}n_{\perp} + \vec{z}_{4\perp}^*, w_1 = (0, iz_6^{t*}, z_6^{x*}, \vec{z}_{6\perp}^*))^*.$ (C.45)

When contracting the indices of the vertex \overline{v} , we should contract the polarizations after taking the complex conjugate. So, using $(v \cdot z)^* = v^* \cdot z^*$, we can write the above equation as a Schwarz reflection,

$$(-x_{34}^{2})^{\frac{\Delta_{3}+\Delta_{4}}{2}} \left(\frac{x_{36}^{2}}{x_{46}^{2}}\right)^{\frac{\Delta_{3}-\Delta_{4}}{2}} I_{e}^{\rho} \left.\frac{\langle 0|O^{\dagger}(x_{6^{+}},z_{6})[O_{4}(x_{4},z_{4}),O_{3}(x_{3},z_{3})]|0\rangle_{(b)}}{2i(\sin(\pi\frac{\tilde{\tau}_{\rho}-\Delta_{3}-J_{3}-\Delta_{4}-J_{4}}{2}))}\right|_{\text{saddle}} = \overline{\nu}(n,z_{3}^{x}n_{\perp}+\vec{z}_{3\perp},-z_{4}^{x}n_{\perp}+\vec{z}_{4\perp},w_{1}=(0,-iz_{6}^{t},z_{6}^{x},\vec{z}_{6\perp})).$$
(C.46)

Rotational invariance demands that the vertex should just contain dot products of the polarizations and n^{μ} . Thus we are free to apply a reflection to all the vectors since it leaves the dot products invariant. In the bulk Minkowski coordinates $(t^{\text{bulk}}, x_1^{\text{bulk}}, x_2^{\text{bulk}}, \ldots)$, we will apply a reflection in the x_1^{bulk} direction. This will send $n^{\mu} \rightarrow -n'^{\mu}, n_{\perp}^{\mu} \rightarrow n'^{\mu}_{\perp}$. (See section 4.4.2 for their expressions in Minkowski

coordinates.) After this reflection, we finally arrive at

$$(-x_{34}^2)^{\frac{\Delta_3 + \Delta_4}{2}} \left(\frac{x_{36}^2}{x_{46}^2} \right)^{\frac{\Delta_3 - \Delta_4}{2}} I_e^{\rho} \left. \frac{\langle 0|O^{\dagger}(x_{6^+}, z_6)[O_4(x_4, z_4), O_3(x_3, z_3)]|0\rangle_{(b)}}{2i(\sin(\pi\frac{\tilde{\tau}_{\rho} - \Delta_3 - J_3 - \Delta_4 - J_4}{2}))} \right|_{\text{saddle}} = \bar{\nu}(-n', z_3^x n'_{\perp} + \vec{z}_{3\perp}, -z_4^x n'_{\perp} + \vec{z}_{4\perp}, w_1 = (0, iz_6^t, z_6^x, \vec{z}_{6\perp})),$$
(C.47)

which agrees with the $\overline{v}(-n', e_3, e_4)$ definition (4.139) and the polarization map (4.136).

C.3.3 Computing R_{ρ}

In the main text, we claim that

$$\frac{2^{d} \dim(\rho)}{\operatorname{vol}(\mathrm{SO}(d))} \frac{\langle \widetilde{O}_{a}^{\dagger}(x_{5}) \widetilde{O}^{\overline{b}}(x_{6}^{\dagger}) \rangle}{\left(\langle \widetilde{O}^{\dagger} \widetilde{O} \rangle, \langle O^{\dagger} O \rangle \right)} \bigg|_{\text{saddle}} = 2^{-J+2\Delta} (-1)^{J-j+\widetilde{j}} R_{\rho} \left(I_{e}^{\rho} \right)_{a}^{\overline{b}}, \qquad (C.48)$$

where the R_{ρ} coefficient is given by (4.142). Here, we give a derivation for (4.142) in the case where $\rho = (J, j)$.

By the two-point pairing definition (4.78), we can rewrite the left-hand side of the above equation as

$$\dim(\rho) \frac{\langle \widetilde{O}_{a}^{\dagger}(e) \widetilde{O}^{\overline{b}}((-e)^{+}) \rangle}{\langle \widetilde{O}_{a'}^{\dagger}(e) \widetilde{O}^{\overline{b}'}((-e)^{+}) \rangle \langle O^{a'}(e) O_{\overline{b}'}^{\dagger}((-e)^{+}) \rangle}.$$
 (C.49)

It is not hard to see that both the shadow two-point $\langle \tilde{O}^{\dagger} \tilde{O} \rangle$ and the two-point structure $\langle O^{\dagger} O \rangle$ are proportional to the reflection tensor \mathcal{I}_{e}^{ρ} . However, the coefficient R_{ρ} should only depend on the convention of the two-point structure $\langle O^{\dagger} O \rangle$. From (C.5), we get

$$\langle O^{\dagger}(e, z_5, w_5, \widetilde{w}_5) O((-e)^+, z_6, w_6, \widetilde{w}_6) \rangle = 2^{J-2\Delta} (-1)^{J-j+\widetilde{j}} \widehat{I}_e^p(z_5, w_5, \widetilde{w}_5; z_6, w_6, \widetilde{w}_6),$$
 (C.50)

where

$$\begin{split} \widetilde{I}_{e}^{p}(z_{5}, w_{5}, \widetilde{w}_{5}; z_{6}, w_{6}, \widetilde{w}_{6}) \\ &= (z_{6} \cdot I_{e} \cdot z_{5})^{J-j} \left(\sum_{\sigma \in S_{2}} z_{6\alpha_{1}} w_{6\alpha_{2}} I_{e}^{\alpha_{1}} {}_{\beta_{1}} I_{e}^{\alpha_{2}} {}_{\beta_{2}} z_{5}^{\beta_{\sigma(1)}} w_{5}^{\beta_{\sigma(2)}} \right)^{j-\widetilde{j}} \\ &\times \left(\sum_{\sigma \in S_{3}} z_{6\alpha_{1}} w_{6\alpha_{2}} \widetilde{w}_{6\alpha_{3}} I_{e}^{\alpha_{1}} {}_{\beta_{1}} I_{e}^{\alpha_{2}} {}_{\beta_{2}} I_{e}^{\alpha_{3}} {}_{\beta_{3}} z_{5}^{\beta_{\sigma(1)}} w_{5}^{\beta_{\sigma(2)}} \widetilde{w}_{5}^{\beta_{\sigma(3)}} \right)^{\widetilde{j}}, \end{split}$$
(C.51)

where $I_{e\nu}^{\mu} = \delta^{\mu}{}_{\nu} + 2e^{\mu}e_{\nu}$, and we have introduced polarization vectors z, w, \tilde{w} for the three rows of the Young diagram of ρ . By comparing (C.48), (C.49), (C.50), and using the fact that $(I_e^{\rho})_a^{\overline{b}} (I_e^{\rho})^a_{\overline{b}} = \dim \rho$, we obtain

$$I_e^{\rho} = R_{\rho} \tilde{I}_e^{\rho}. \tag{C.52}$$

This implies

$$(\widehat{I}_e^{\rho})_a{}^{\overline{b}}(\widehat{I}_e^{\rho})^a{}^{\overline{b}}_{\overline{b}} = R_{\rho}^{-2}\dim\rho.$$
(C.53)

Now, let us specialize to $\rho = (J, j)$ and compute $(\hat{l}_e^{\rho})_a^{\overline{b}} (\hat{l}_e^{\rho})^a_{\overline{b}}$. The main idea is that we can use weight-shifting operators to derive a recursion relation [39, 45]. In particular, we will use

$$\mathcal{D}_{z,w}^{0+\mu}\Big|_{J,j} = (j-J)w^{\mu} + z^{\mu}z \cdot \frac{\partial}{\partial w},$$

$$\mathcal{D}_{z,w}^{0-\mu}\Big|_{J,j} = \left((-J-d+4-j)\delta^{\mu}{}_{\nu} + z^{\mu}\frac{\partial}{\partial z^{\nu}}\right) \left((d-6+2j)\frac{\partial}{\partial w_{\nu}} - w^{\nu}\partial_{w}^{2}\right), \quad (C.54)$$

where $\mathcal{D}_{z,w}^{0+}$ increases the transverse spin *j* by 1 and $\mathcal{D}_{z,w}^{0-}$ decreases *j* by 1. These operators are the weight-shifting operators of the SO(d - 1, 1) group in the vector representation [39]. Using them, we can build a "bubble diagram,"

$$\mathcal{D}_{z_5,w_5}^{0+} \cdot \mathcal{D}_{z_5,w_5}^{0-} \widehat{I}_e^{\rho}(z_5,w_5;z_6,w_6) = j(d-6+2j)(J+j+d-4)(J-j+2)\widehat{I}_e^{\rho}(z_5,w_5;z_6,w_6)$$
(C.55)

Furthermore, one can perform crossing on a weight-shift operator and move it to the other leg of \hat{I}_e ,

$$\mathcal{D}_{z_5,w_5}^{0-\mu}\widehat{I}_e^p(z_5,w_5;z_6,w_6) = \frac{j(d-6+2j)(J+j+d-4)}{J-j+1}\mathcal{D}_{z_6,w_6}^{0+\mu}\widehat{I}_e^{p'}(z_5,w_5;z_6,w_6),$$
(C.56)

where $\rho' = (J, j - 1)$. Combining the above two equations, we get

$$\widehat{I}_{e}^{p}(z_{5}, w_{5}; z_{6}, w_{6}) = \frac{1}{(J-j+1)(J-j+2)} \mathcal{D}_{z_{5}, w_{5}}^{0+} \cdot \mathcal{D}_{z_{6}, w_{6}}^{0+} \widehat{I}_{e}^{p'}(z_{5}, w_{5}; z_{6}, w_{6}).$$
(C.57)

Our goal is to compute the index contraction of two \hat{I}_e^p tensors. Using (C.57) we can rewrite it as the index contraction of $\mathcal{D}_{z_5,w_5}^{0+} \cdot \mathcal{D}_{z_6,w_6}^{0+} \hat{I}_e^{p'}$ and \hat{I}_e^p . Then, we can "integrate by parts" to move the weight-shifting operators to act on \hat{I}_e^p . More

precisely, let us denote the index contraction by a pairing (\cdots, \cdots) . Then the integration by parts relation is

$$\left(\mathcal{D}_{z,w}^{0+\mu}\widehat{I}_{e}^{\rho'},\widehat{I}_{e}^{\rho}\right) = \frac{J-j+2}{j(d-6+2j)(J+j+d-4)} \left(I_{e}^{\rho'},\mathcal{D}_{z,w}^{0-\mu}I_{e}^{\rho}\right).$$
(C.58)

Note that this relation is different from the one given in [45]. This is simply because they are adjoint relations with respect to different pairings. To obtain the adjoint with respect to index contraction, a simple way is to use the identity of the $\mathcal{D}^{(h)}$ operator given in [110]. (Note that the \mathcal{D}^{0-} operator defined here is just a special case of $\mathcal{D}^{(h)}$ with h = 2.)

The adjoint relation enables us to turn the computation into the index contraction of $\widehat{I}_{e}^{p'}$ and $\mathcal{D}_{z_5,w_5}^{0-} \cdot \mathcal{D}_{z_6,w_6}^{0-} \widehat{I}_{e}^{p}$, which then becomes the contraction of two $\widehat{I}_{e}^{p'}$'s. Together with (C.53), this gives the recursion relation

$$R_{\rho}^{-2}\dim\rho = \frac{(j+d-5)(2j+d-4)(J-j+2)(J+j+d-3)}{j(2j+d-6)(J-j+1)(J+j+d-4)}R_{\rho'}^{-2}\dim\rho'.$$
(C.59)

The relation between dim ρ and dim ρ' can be obtained from standard formula [180, 45], or from the recursion relation for the Plancherel measure of the SO(d - 1, 1) Lorentz group [45]. Eventually, we get

$$\frac{R_{\rho'=(J,j-1)}^2}{R_{\rho=(J,j)}^2} = \frac{(J-j+2)^2}{(J-j+1)^2}.$$
(C.60)

With the initial condition $R_{\rho=(J,0)} = 1$, this leads to (4.142) in the $\tilde{j} = 0$ case.

C.4 Dual structures at large v

In this appendix, we discuss how to compute the dual structure $\left(\langle 0|O_4 \mathbf{L}[O]O_2|0\rangle^{(a)}\right)^{-1}$ that appears in the kernel of the functional. In particular, we are interested in the large ν limit, where ν parametrizes the scaling dimension of O as $\Delta = \frac{d}{2} + i\nu$.

C.4.1 Light transform at large dimension

To understand the dual structure, we should first study the light transform of the three-point function $\langle 0|O_4OO_2|0\rangle$, and then consider its Lorentzian three-point pairing. Hence, let us first consider the light transform L[O] in the limit where O has large scaling dimension. Recall that the light transformed three-point function is given by

$$\langle 0|O_4 \mathbf{L}[O_{\Delta,J,\lambda'}](X_0, Z_0)O_2|0\rangle = \int d\alpha \ \langle 0|O_4 O_{\Delta,J,\lambda'}(Z_0 - \alpha X_0, -X_0)O_2|0\rangle$$
(C.61)



Figure C.1: The steepest descent flow of the saddle integral (C.64). The red point is the saddle point. We show the deformed contour passing through the saddle in green and orange. The green part starts at infinity, goes in the opposite direction of the flow, and gradually spirals in toward the saddle. Along the green contour, the integrand keeps increasing and reaches maximum at the saddle point. It passes through the saddle point along a direction that is rotated by $e^{-\frac{3\pi i}{4}}$ relative to the real axis. After passing through the saddle point, the contour (in orange) goes in the same direction as the flow and gradually spirals out to infinity. The integrand at infinity goes as $\alpha^{-i\nu}$, so by making the contour spiral many times clockwise at infinity, we ensure that the contribution $\alpha^{-i\nu} \propto e^{2\pi \nu \arg(\alpha)}$ can be made arbitrarily small.

We will focus on the case $4 > 0 > 2^-$ and consider the limit $\Delta = \frac{d}{2} + i\nu, \nu \to \infty$. In this limit, we see that the quickly varying part of the integrand is

$$e^{-\frac{i\nu}{2}(\log(-2(Z_0 - \alpha X_0) \cdot X_2) + \log(-2(Z_0 - \alpha X_0) \cdot X_4))}.$$
(C.62)

which implies that the integral has a saddle point at

$$\alpha_* = \frac{2X_0 \cdot X_2 Z_0 \cdot X_4 + 2X_0 \cdot X_4 Z_0 \cdot X_2}{X_{02} X_{04}}.$$
(C.63)

As one can check, this point is spacelike from both 2 and 4. Therefore, the light transform gets localized to the point in the middle of the region that is spacelike from 2 and 4.

Expanding the quickly varying part around α_* , we find that the saddle integral we have to do is

$$\int d\alpha \ e^{\frac{i\nu}{2} \left(\frac{X_{02}X_{04}}{V_{0,24}X_{24}}\right)^2 (\alpha - \alpha_*)^2} = e^{-\frac{3i\pi}{4}} \sqrt{\frac{2\pi}{\nu}} \frac{(-V_{0,42})X_{24}}{X_{02}(-X_{04})}.$$
 (C.64)

The phase factor $e^{-\frac{3i\pi}{4}}$ comes from deforming the contour to a steepest descent contour. We show the steepest descent flow and the deformed contour in figure C.1. The contour passes through the saddle point with angle $-\frac{3\pi}{4}$, and hence we have the $e^{-\frac{3i\pi}{4}}$ factor in (C.64). Also, note that when taking $\left(\frac{X_{02}X_{04}}{V_{0,42}X_{24}}\right)^2$ out of the square root, we have to make sure all factors are positive. In summary, the above analysis implies that the large Δ limit of $\langle 0|O_4 \mathbf{L}[O_{\Delta,J}](X_0, Z_0)O_2|0\rangle$ should be given by

$$\lim_{\substack{\Delta = \frac{d}{2} + i\nu \\ \nu \gg 1}} \langle 0|O_4 \mathbf{L}[O_{\Delta,J,\lambda'}](X_0, Z_0)O_2|0\rangle = e^{-\frac{3i\pi}{4}} \sqrt{\frac{2\pi}{\nu} \frac{(-V_{0,42})X_{24}}{X_{02}(-X_{04})}} \langle 0|O_4 O_{\Delta,J,\lambda'}(Z_0 - \alpha_*X_0, -X_0)O_2|0\rangle.$$
(C.65)

Furthermore, we can study how each building block of the three-point structure transforms under $X_0 \rightarrow Z_0 - \alpha_* X_0, Z_0 \rightarrow -X_0$. We will focus on the structures that appear when λ' has a single row of length j'. We find

$$X_{02} \rightarrow \frac{(-V_{0,42})X_{24}}{-X_{04}}, \quad V_{2,04} \rightarrow -\left(V_{2,04} + \frac{H_{20}}{V_{0,42}}\right), \quad U_{0,24} \rightarrow \frac{(-V_{0,42})X_{24}}{(-X_{04})X_{02}}U_{0,24},$$
$$X_{04} \rightarrow \frac{(-V_{0,42})X_{24}}{X_{02}}, \quad V_{4,20} \rightarrow -\left(V_{4,20} + \frac{H_{40}}{V_{0,42}}\right), \quad U_{0,42} \rightarrow \frac{(-V_{0,42})X_{24}}{X_{04}X_{02}}U_{0,42},$$
$$(C.66)$$

and all other structures are invariant (assuming O_2 and O_4 have no transverse spin). Therefore, starting with the three-point structure (C.9), its light transform at large Δ is given by

$$\lim_{\substack{\Delta = \frac{d}{2} + i\nu \\ \nu \gg 1}} \langle 0|O_{4}\mathbf{L}[O_{\Delta,J,j'}](X_{0}, Z_{0}, W_{0})O_{2}|0\rangle = 2^{\Delta + J - 1}e^{-\frac{3i\pi}{4}}\sqrt{\frac{2\pi}{\nu}}$$

$$\times \frac{(-2V_{0,42})^{1 - \Delta - J + m_{0}} \left(-V_{2,04} - \frac{H_{20}}{V_{0,42}}\right)^{m_{2}} \left(-V_{4,20} - \frac{H_{40}}{V_{0,42}}\right)^{m_{4}} H_{24}^{n_{24}} H_{02}^{n_{02}} H_{04}^{n_{04}} (-2U_{0,24})^{k_{02}} (2U_{0,42})^{k_{04}}}{\frac{X_{24}^{\frac{\Delta_{2} + J_{2} + \Delta_{4} + J_{4} - 2 + \Delta + J - j'}{2}}{X_{24}^{\frac{\Delta_{2} + J_{2} + \Delta_{4} + J_{4} - 2 + \Delta + J - j'}{2}} (-X_{04})^{\frac{\Delta_{4} + J_{4} + 2 - \Delta - J + j' - \Delta_{2} - J_{2}}{2}} X_{02}^{\frac{\Delta_{2} + J_{2} + 2 - \Delta - J + j' + \Delta_{2} + J_{2}}{2}}} (C.67)$$

The result can also be written as

$$2^{\Delta+J-1}e^{-\frac{3i\pi}{4}}\sqrt{\frac{2\pi}{\nu}}\langle 0|O_4O^L(X_0, Z_0, W_0)O_2|0\rangle \bigg|_{V_{2,04}\to -\left(V_{2,04}+\frac{H_{20}}{V_{0,42}}\right), V_{4,20}\to -\left(V_{4,20}+\frac{H_{40}}{V_{0,42}}\right), U_{0,42}\to -U_{0,42}}$$
(C.68)

,

where the operator O^L has quantum numbers $(1 - J, 1 - \Delta, j')$.

Computation of light transform of general spinning three-point functions has been discussed in [34] for general Δ (see eq. (5.69)). One can check that the above result agrees with the large Δ_1 limit of the result in [34], which contains an Appell F_2 . The Appell F_2 function in this limit will simplify and lead to the replacement rule $V_{2,04} \rightarrow -\left(V_{2,04} + \frac{H_{20}}{V_{0,42}}\right), V_{4,20} \rightarrow -\left(V_{4,20} + \frac{H_{40}}{V_{0,42}}\right)$ given above.

We also need to understand how to impose conservation condition. In section 4.4.1, we study the conservation condition when the exchanged operator O has large dimension. Here, after the light transform, the dimension and spin are swapped, and therefore we have to instead consider conservation at large spin. However, the strategy in both cases are the same. We should identify the quickly-varying part of the three-point structure and take its derivative. By taking a x_4 -derivative of the Δ -dependent factor of (C.67), we find that the conservation condition $\partial_{x_4} \langle O_4 O O_2 \rangle = 0$ after the light transform becomes equivalent to

$$\left(V_{4,20} + \frac{H_{40}}{V_{0,42}}\right)_A D^A_{Z_4} \langle 0|O_4 \mathbf{L}[O]O_2|0\rangle = 0,$$
(C.69)

where A is an embedding space index, and its meaning as a subscript is that we should take the structures $V_{4,20}$, H_{40} and strip off the Z_4 . $D_{Z_4}^A$ is the Todorov/Thomas operator [30]. This condition is perfectly consistent with the original large Δ conservation (4.113) and the replacement rule from the light transform given by (C.66).

C.4.2 Dual structures

To get the dual structures, we can start with a standard basis of continuous-spin three-point structures $\langle 0|\tilde{O}_4^{\dagger}O^F\tilde{O}_2^{\dagger}|0\rangle^{(a)}$ by adding spinning structures to the scalar convention (C.15) following (C.9). The quantum numbers of \tilde{O}_2^{\dagger} , \tilde{O}_4^{\dagger} are $(d - \Delta_2, J_2)$ and $(d - \Delta_4, J_4)$. For O^F , we should have $(\Delta_F, J_F) = (J + d - 1, \Delta - d + 1)$, where $\Delta = \frac{d}{2} + i\nu$ and $J = J_1 + J_3 - 1$ depends on the spin of the external operators. The actual dual structures should be given by

$$\left(\langle 0|O_4\mathbf{L}[O]O_2|0\rangle^{(a)}\right)^{-1} = \alpha^{(a)}{}_{(b)}\langle 0|\widetilde{O}_4^{\dagger}O^F\widetilde{O}_2^{\dagger}|0\rangle^{(b)}, \qquad (C.70)$$

and our goal is to find the coefficients $\alpha^{(a)}{}_{(b)}$.

Following the definition of dual structures (4.94), we have

$$\alpha^{(a)}{}_{(b)}\left(\langle 0|\widetilde{O}_4^{\dagger}O^F\widetilde{O}_2^{\dagger}|0\rangle^{(b)},\langle 0|O_4\mathbf{L}[O]O_2|0\rangle_{(c)}\right)_L = \delta_c^a,\tag{C.71}$$

which gives

$$\alpha^{(a)}{}_{(b)} = \left(\langle 0 | \widetilde{O}_4^{\dagger} O^F \widetilde{O}_2^{\dagger} | 0 \rangle^{(b)}, \langle 0 | O_4 \mathbf{L}[O] O_2 | 0 \rangle_{(a)} \right)_L^{-1}$$

= 2^{2d-2} vol(SO(d - 2)) $\left(\langle 0 | \widetilde{O}_4^{\dagger}(0^+) O^F(\infty, z_0^*) \widetilde{O}_2^{\dagger}(e) | 0 \rangle^{(b)} \langle 0 | O_4(0^+) \mathbf{L}[O](\infty, z_0^*) O_2(e) | 0 \rangle_{(a)} \right)^{-1}$
(C.72)

where the in the second line we have used the conformal group to fixed all the points to the configuration $x_4 = 0^+, x_2 = e, x_0 = \infty$ and $z_0 = (1, 1, \vec{0})$ (in lightcone coordinates). The prefactor $2^{2d-2} \operatorname{vol}(\operatorname{SO}(d-2))$ comes from the Faddeev-Popov determinant and volume of the stabilizer group of the gauge-fixed configuration [11].

Then, the calculation of dual structures is now reduced to evaluating the continuousspin basis and the light transformed structures in this standard configuration. Since the continuous-spin basis can be obtained from (C.15), (C.9), and the light transformed structures at large ν are given by (C.67), the dual structures at large ν can be computed straightforwardly using (C.72).

Lastly, when we impose conservation on the external operators O_2 , O_4 , the dual structures get a gauge redundancy due to the conservation equations. At large ν , the statement becomes

$$\langle 0|O_{4}\mathbf{L}[O]O_{2}|0\rangle^{-1} \sim \langle 0|O_{4}\mathbf{L}[O]O_{2}|0\rangle^{-1} + \left(V_{4,20} + \frac{H_{40}}{V_{0,42}}\right)(\ldots) + \left(V_{2,04} + \frac{H_{20}}{V_{0,42}}\right)(\ldots) + \left(V_{2,04} + \frac{H_{20$$

The idea is that whenever we have a $\left(V_{4,20} + \frac{H_{40}}{V_{0,42}}\right)$ factor, we can always integrate it by parts in the pairing [45] and get the large- ν conservation equation (C.69). The argument for the other factor is similar. Fortunately, thanks to the identity (4.153), when we evaluate the structures at the saddle and set the time component of the external polarizations to zero, the result is independent of this gauge redundancy.

In summary, to compute the dual structures for conserved external operators, we can use the gauge redundancy to remove all the V_i structures for the external operators and simply consider a basis with just the H_{i0} , H_{ij} structures, as stated in section 4.4.3. We can then use (C.72) to compute the dual structure coefficients, where the label (*b*) only includes continuous-spin structures without the V_i 's, and (*a*) are the conserved structures. These two structures both have the same counting given by (4.154).

C.5 CFT four-point structures

In this appendix, we give the expressions of the CFT four-point structures that we use in the main text. In particular, they should satisfy (4.181). From the group-theoretic counting argument [48], we already know that the number of CFT four-point structures should agree with the number of flat space polarization structures. To write down the CFT structures, let us first define

$$\begin{split} H_{ijk}^{CFT} &= [X_i, Z_i]^{A_1 A_2} [X_j, Z_j]^{A_2 A_3} [X_k, Z_k]^{A_3 A_1}, \\ H_{ijkl}^{CFT} &= [X_i, Z_i]^{A_1 A_2} [X_j, Z_j]^{A_2 A_3} [X_k, Z_k]^{A_3 A_4} [X_l, Z_l]^{A_4 A_1}, \\ X_{ijkl}^{CFT} &= H_{ijkl}^{CFT} - \frac{1}{4} \left(H_{14}^{CFT} H_{23}^{CFT} + H_{13}^{CFT} H_{24}^{CFT} + H_{12}^{CFT} H_{34}^{CFT} \right), \\ S^{CFT} &= V_{1,24}^{CFT} H_{234}^{CFT} + V_{2,31}^{CFT} H_{341}^{CFT} + V_{3,42}^{CFT} H_{412}^{CFT} + V_{4,13}^{CFT} H_{123}^{CFT}, \end{split}$$
(C.74)

where A_i are embedding space indices, and $[X, Z]^{AB} = X^A Z^B - X^B Z^A$. The superscript *CFT* is to distinguish the CFT structure and the flat space structure, and V^{CFT} , H_{ij}^{CFT} are given in (C.4). Then, we find

$$\left\{ H_{14}^{CFT} H_{23}^{CFT}, H_{13}^{CFT} H_{24}^{CFT}, H_{12}^{CFT} H_{34}^{CFT}, X_{1243}^{CFT}, X_{1234}^{CFT}, X_{1324}^{CFT}, S^{CFT} \right\} \Big|_{\text{saddle},(4.133),(4.136)}$$

= $\frac{16}{\nu^4} \left\{ H_{14}H_{23}, H_{13}H_{24}, H_{12}H_{34}, X_{1243}, X_{1234}, X_{1324}, \frac{2}{\nu^2} S \right\}.$ (C.75)

Thus, we choose the left hand side of the above equation to be our four-point structure basis Q^{I} in the main text.

The CFT four-point structures we choose all have homogeneity 1 for all $X_{1,2,3,4}$, $Z_{1,2,3,4}$. Under the map of polarizations, they reproduce the expected amplitude structures, with the correct Regge behavior. The additional factors of v should not be a big issue since we can absorb them into the prefactors. The only different case is that we have to introduce a $\frac{2}{v^2}$ factor for the *S* structure. This is reasonable from dimensional analysis since the *S* structure has an additional $p_i \cdot p_j$. It also seems to suggest that from the CFT point of view, $\frac{2}{v^2}S$ is a more natural choice for the structure.

For the four-graviton case, the construction given above can be easily generalized. The only four-graviton flat space structure that cannot be written in terms of H_{ij}, X_{ijkl}, S is the Gram determinant of all dot products between $(p_1, p_2, p_3, e_1, e_2, e_3, e_4)$, which we write as $\mathcal{G}_{p_1,p_2,p_3,e_1,e_2,e_3,e_4}$. Even though this structure does not appear in the graviton sum rule dictionary given in section 4.4.6 as it grows more slowly in the Regge limit, let us still give the corresponding CFT structure for completeness. The corresponding CFT structure can be written as

$$X_{24} \left(\frac{X_{14}X_{34}}{X_{12}X_{23}}\right)^{\frac{1}{2}} \mathcal{G}_{X_1,X_2,X_3,Z_1,Z_2,Z_3,Z_4} - (\ldots) \mathcal{G}_{X_1,X_2,X_3,Z_1,Z_2,Z_3,X_4}.$$
 (C.76)

This structure is manifestly invariant under $Z_i \rightarrow Z_i + \#X_i$ thanks to properties of determinant. The only exception is the $Z_4 \rightarrow Z_4 + \#X_4$ gauge redundancy. This is why we introduce the second term, and (...) is a factor that fixes homogeneity. The structure is constructed such that its homogeneity is the same as the other structures (has homogeneity 1 for all X_i, Z_i). When evaluating the above expression at the saddle and applying the polarization map, we find that the second term vanishes, and we have

$$X_{24} \left(\frac{X_{14}X_{34}}{X_{12}X_{23}}\right)^{\frac{1}{2}} \mathcal{G}_{X_1, X_2, X_3, Z_1, Z_2, Z_3, Z_4} \bigg|_{\text{saddle}, (4.133), (4.136)} = \frac{256}{\nu^6} \mathcal{G}_{p_1, p_2, p_3, e_1, e_2, e_3, e_4}.$$
(C.77)

So, (C.76) should be the correct CFT four-point structure that gives $\mathcal{G}_{p_1,p_2,p_3,e_1,e_2,e_3,e_4}$.

APPENDICES TO CHAPTER 5

D.1 Pole structure of scalar crossing equation

In this appendix we will carefully derive the pole structure of the constrained Epstein zeta series $\mathcal{E}_{s}^{c}(\mu)$. Much of this analysis is in Sec. 3.2 of [165]. Let us look at the scalar sector of (5.19):

$$y^{\frac{c}{2}} \left(1 + \sum_{\Delta \in \mathcal{S}} e^{-2\pi\Delta y} \right) \coloneqq \int_{-1/2}^{1/2} dx \widehat{Z}^{c}(\tau, \mu)$$

$$= y^{\frac{c}{2}} + \frac{\Lambda\left(\frac{c-1}{2}\right)}{\Lambda\left(\frac{c}{2}\right)} y^{1-\frac{c}{2}} + 3\pi^{-\frac{c}{2}} \Gamma\left(\frac{c}{2} - 1\right) \mathcal{E}^{c}_{\frac{c}{2}-1}(\mu)$$

$$+ \frac{1}{4\pi i} \int_{\frac{1}{2} - i\infty}^{\frac{1}{2} + i\infty} ds \pi^{s-\frac{c}{2}} \Gamma\left(\frac{c}{2} - s\right) \mathcal{E}^{c}_{\frac{c}{2}-s}(\mu) \left(y^{s} + \frac{\Lambda(s - \frac{1}{2})}{\Lambda(s)} y^{1-s}\right)$$

$$= y^{\frac{c}{2}} + \frac{\Lambda\left(\frac{c-1}{2}\right)}{\Lambda\left(\frac{c}{2}\right)} y^{1-\frac{c}{2}} + 3\pi^{-\frac{c}{2}} \Gamma\left(\frac{c}{2} - 1\right) \mathcal{E}^{c}_{\frac{c}{2}-1}(\mu)$$

$$+ \frac{1}{2\pi i} \int_{\frac{1}{2} - i\infty}^{\frac{1}{2} + i\infty} ds \pi^{s-\frac{c}{2}} \Gamma\left(\frac{c}{2} - s\right) \mathcal{E}^{c}_{\frac{c}{2}-s}(\mu) y^{s}. \tag{D.1}$$

In the last line of (D.1), we used the functional equation that $\mathcal{E}_{\frac{c}{2}-s}^{c}(\mu)$ obeys, (5.21). We would like to move the contour in (D.1) to the right, so we again need to classify all simple poles of the integrand with $\operatorname{Re}(s) > \frac{1}{2}$. As was argued in [165], there can only be poles we cross at $s = \frac{c}{2}$ and $s = \frac{1+z_n}{2}, \frac{1+z_n^*}{2}$. Let us review the argument.

The idea is to take the inverse Laplace transform of (D.1) to get the scalar density of states. We then integrate from 0 to some number Δ (not including the vacuum), and demand that this vanishes for sufficiently small Δ . This is due to the fact that the spectrum for a compact CFT is discrete, so in general there is a gap between the vacuum and first excited scalar state. A simple calculation shows the number of scalar operators (excluding the vacuum) below Δ is

$$N_{0}(\Delta) = \frac{2\pi^{c}\zeta(c-1)\Delta^{c-1}}{(c-1)\Gamma(\frac{c}{2})^{2}\zeta(c)} + 12\frac{2^{\frac{c}{2}}\Delta^{\frac{c}{2}}\mathcal{E}_{\frac{c}{2}-1}^{c}(\mu)}{c(c-2)} - \frac{1}{2\pi i}\int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} ds\frac{2^{\frac{c}{2}-s}\Delta^{\frac{c}{2}-s}\mathcal{E}_{\frac{c}{2}-s}^{c}(\mu)}{s-\frac{c}{2}}$$
$$= \frac{2\pi^{c}\zeta(c-1)\Delta^{c-1}}{(c-1)\Gamma(\frac{c}{2})^{2}\zeta(c)} + 12\frac{2^{\frac{c}{2}}\Delta^{\frac{c}{2}}\mathcal{E}_{\frac{c}{2}-1}^{c}(\mu)}{c(c-2)}$$
$$+ \frac{1}{2\pi i}\int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} ds\frac{2^{\frac{c}{2}-s}\Delta^{\frac{c}{2}-s}\Gamma(s)\Gamma(s+\frac{c}{2}-1)\zeta(2s)\mathcal{E}_{\frac{c}{2}+s-1}^{c}(\mu)}{\pi^{2s-\frac{1}{2}}\Gamma(\frac{c}{2}+1-s)\Gamma(s-\frac{1}{2})\zeta(2s-1)}$$
(D.2)

Let us look at the last line of (D.2). In the limit of small Δ , we must get 0 for the integrated density of states, which means the integral must cancel the two power laws in Δ coming from the first two terms. In the integral in the last line of (D.2), we must close the contour to the left in the *s*-plane since Δ is small. This will tell us about the pole structure of $\mathcal{E}_{\frac{c}{2}+s-1}^{c}(\mu)$ for $\operatorname{Re}(s) < \frac{1}{2}$ (if we wanted to know the pole structure for $\operatorname{Re}(s) > \frac{1}{2}$ we would look at the first line of (D.2) and again close the contour to the left). In order to cancel the term that goes as Δ^{c-1} , we need a pole at $s = 1 - \frac{c}{2}$. This comes from the term $\Gamma(s + \frac{c}{2} - 1)$ in the numerator, with the others being finite. (Although the other gamma and zeta functions naively contribute poles and zeros for integer *c*, their combination is always finite.) Moreover in order for the residue to match, this fixes

$$\mathcal{E}_0^c(\mu) = -1.$$
 (D.3)

We also need to cancel the second polynomial in (D.2). This comes from a pole at s = 0, coming from the $\Gamma(s)$ term. We see the residue already matches the coefficient in (D.2) so we cannot constrain the value of $\mathcal{E}_{\frac{c}{2}-1}^{c}(\mu)$. Finally there can be no other poles with $\operatorname{Re}(s) < \frac{1}{2}$. Naively this tells us that $\mathcal{E}_{\frac{c}{2}+s-1}^{c}(\mu)$ cannot have any poles for $\operatorname{Re}(s) < \frac{1}{2}$, but this is too fast – if the prefactor vanishes then $\mathcal{E}_{\frac{c}{2}+s-1}^{c}(\mu)$ can have a pole. The only zeros with $\operatorname{Re}(s) < \frac{1}{2}$ in the prefactor of the integrand are when $s = \frac{z_n}{2}, \frac{z_n^*}{2}$, coming from the $\zeta(2s)$ term. Thus $\mathcal{E}_{\frac{c}{2}+s-1}^{c}(\mu)$ can have a pole at $s = \frac{z_n}{2}, \frac{z_n^*}{2}$. We also know that $\mathcal{E}_{\frac{c}{2}+s-1}^{c}(\mu)$ must have zeros at $s = -\frac{c}{2}, -\frac{c}{2} - 1, \cdots$ to cancel the poles from $\Gamma(s + \frac{c}{2} - 1)$.

Thus, looking at the integrand in (D.1), we see the only poles to the right of the contour of integration are at $s = \frac{c}{2}$ and $s = 1 - \frac{z_n}{2}$, $1 - \frac{z_n^*}{2}$. (Using the functional equation for the zeta function, we can rewrite the last term as $s = \frac{1+z_n}{2}$, $\frac{1+z_n^*}{2}$.) The residue of the pole at $s = \frac{c}{2}$ is given in (D.3) and the residue at $s = \frac{1+z_n}{2}$, $\frac{1+z_n^*}{2}$ is

just given by reading off the pole from integrating the partition function against an Eisenstein series at $s = \frac{z_n}{2}$ (see (5.29)).

This fully reproduces the pole structure which we used to derive (5.38).

D.2 Functional action on crossing equation

Let us consider the functional

$$\mathcal{F}_k[h(t)] \coloneqq \int_0^\infty \frac{dt}{t} h(t) \sum_{m=1}^\infty e^{-\pi k t^2 m^2}.$$
 (D.4)

We would like to apply this functional to each of the terms in (5.41). To do so let us first compute:

$$\mathcal{F}_{k}[t^{s}e^{-\frac{A}{t^{2}}}] \coloneqq \sum_{m=1}^{\infty} f_{m}^{k}(s,A)$$
$$\mathcal{F}_{k}[t^{s}e^{-Bt^{2}}U(\alpha,\beta,Bt^{2})] \coloneqq \sum_{m=1}^{\infty} g_{m}^{\alpha,\beta,k}(s,B),$$
(D.5)

with $f_m^k(s, A)$ and $g_m^{\alpha, \beta, k}(s, B)$ defined as

Applying this term by term to (5.41) we then get:

$$\begin{split} \sum_{\Delta \in \mathcal{S}} \sum_{k} \alpha_{k} \sum_{m=1}^{\infty} \left[4\pi \Delta f_{m}^{k}(-c, 2\pi\Delta) - c f_{m}^{k}(2-c, 2\pi\Delta) \\ &- \sum_{n=1}^{\infty} \frac{b(n)n^{c-2}}{\sqrt{\pi}} \Big((c-2)g_{m}^{-\frac{1}{2},\frac{c}{2},k}(c, 2\pi n^{2}\Delta) - 4\pi n^{2}\Delta g_{m}^{-\frac{1}{2},\frac{c}{2},k}(c+2, 2\pi n^{2}\Delta) \\ &+ 2\pi n^{2}\Delta g_{m}^{\frac{1}{2},\frac{c+2}{2},k}(c+2, 2\pi n^{2}\Delta) \Big) \right] \\ &= \zeta(c-1)\Gamma\left(\frac{c-1}{2}\right)\pi^{\frac{1-c}{2}} \sum_{k} \alpha_{k}\left(\frac{c}{2}k^{\frac{c}{2}-1} + \left(\frac{c}{2}-1\right)k^{-\frac{c}{2}}\right). \quad (D.7) \end{split}$$

The above equation is summed over an arbitrary choice of k's and α_k 's, subject to the constraints in (5.50).

Remarkably, for odd $c \ge 3$, we can get closed form expressions for the sums over *m* in (D.7). For c = 3, (D.7) reduces to

$$\sum_{\Delta \in \mathcal{S}} \sum_{k} \alpha_{k} \left[\frac{\sqrt{2} - e^{2\sqrt{2}\pi\sqrt{k\Delta}}(\sqrt{2} - 2\pi\sqrt{k\Delta})}{2(-1 + e^{2\sqrt{2}\pi\sqrt{k\Delta}})^{2}\sqrt{\Delta}} + \sum_{n=1}^{\infty} \frac{b(n)n\pi\cosh(\sqrt{2}n\pi\sqrt{\frac{\Delta}{k}})}{4k^{\frac{3}{2}}\sinh^{3}(\sqrt{2}n\pi\sqrt{\frac{\Delta}{k}})} \right]$$
$$= \frac{\pi}{6} \sum_{k} \alpha_{k} \left(\frac{3}{2}k^{\frac{1}{2}} + \frac{1}{2}k^{-\frac{3}{2}} \right).$$
(D.8)

The sum over n can be simplified to give

$$\sum_{\Delta \in S} \sum_{k} \alpha_{k} \left[\frac{\sqrt{2} - e^{2\sqrt{2}\pi\sqrt{k\Delta}}(\sqrt{2} - 2\pi\sqrt{k\Delta})}{2(-1 + e^{2\sqrt{2}\pi\sqrt{k\Delta}})^{2}\sqrt{\Delta}} + \frac{\pi e^{2\sqrt{2}\pi\sqrt{\frac{\Delta}{k}}}}{(-1 + e^{2\sqrt{2}\pi\sqrt{\frac{\Delta}{k}}})^{2}k^{3/2}} \right] = \frac{\pi}{6} \sum_{k} \alpha_{k} \left(\frac{3}{2}k^{\frac{1}{2}} + \frac{1}{2}k^{-\frac{3}{2}} \right).$$
(D.9)

To simplify (D.7) for c odd, $c \ge 5$, we first define the auxiliary functions:

$$\begin{split} \nu_{1}(c,0,m) &\coloneqq \frac{(-1)^{\frac{c+1}{2}+m}\Gamma(c)}{(c-2)\Gamma(m+1)\Gamma(\frac{c+1}{2}-m)}, \\ \nu_{1}(c,n,m) &\coloneqq (-1)^{n+\frac{c-1}{2}+m}2^{2n}(n+1)(c-\frac{(n+1)(n+2)}{2})\Gamma(c-n-2) \\ &\qquad \times \sum_{i=0}^{m-1}\sum_{j=0}^{i+1}\frac{(-1)^{i+j}(i-j+1)^{n}}{\Gamma(j+1)\Gamma(n-j+2)\Gamma(m-i)\Gamma(\frac{c+3}{2}-n-m+i)}, \qquad n \neq 0 \\ \nu_{2}(c,n,m) &\coloneqq 2^{-2m}(c-3-2m)(c-1-2m)(c+1-2m)(c+3-2m)\Gamma(\frac{c-3}{2}+m) \\ &\qquad \times \sum_{i=0}^{\frac{c-1}{2}-m}\sum_{j=0}^{i+1}\frac{(-1)^{m+n+i+j}(i-j+1)\frac{c+1}{2}-m}{\Gamma(j+1)\Gamma(\frac{c+5}{2}-m-j)\Gamma(n-i+1)\Gamma(m-n+i+1)} \\ \nu_{3}(c,n,m) &\coloneqq (-1)^{n+\frac{c-1}{2}+m}2^{2n}(n+1)(n+2)(n+3)\Gamma(c-n-4) \\ &\qquad \times \sum_{i=0}^{m}\sum_{j=0}^{i+1}\frac{(-1)^{i+j}(i-j+1)^{n+2}}{\Gamma(j+1)\Gamma(n-j+4)\Gamma(m-i+1)\Gamma(\frac{c-3}{2}-n-m+i)}. \end{split}$$
(D.10)
Then (D.7) becomes:

$$\begin{split} \sum_{\Delta \in \mathcal{S}} \sum_{k} \alpha_{k} \Biggl[\frac{1}{2^{\frac{3c}{2} - 3} \pi^{\frac{c-3}{2}} \Delta^{\frac{c-2}{2}} (-1 + e^{2\sqrt{2}\pi\sqrt{k\Delta}})^{\frac{c+1}{2}}} \sum_{i=0}^{\frac{c-1}{2}} \sum_{j=0}^{\frac{c-1}{2}} \sum_{i=0}^{2} v_{1}(c, i, j) e^{2\sqrt{2}\pi\sqrt{k\Delta}j} (\sqrt{2k\Delta}\pi)^{i} \\ &+ \sum_{n=1}^{\infty} \frac{b(n) e^{2\sqrt{2}\pi n\sqrt{\frac{\lambda}{k}}} n^{\frac{c-1}{2}} \pi}{2^{\frac{c+13}{4}} k^{\frac{c+3}{4}} \Delta^{\frac{c-3}{4}} (-1 + e^{2\sqrt{2}\pi n\sqrt{\frac{\lambda}{k}}})^{\frac{c+3}{2}}} \sum_{i=0}^{\frac{c-1}{2}} \sum_{j=0}^{\frac{c-5}{2}} \frac{v_{2}(c, i, j) k^{\frac{j}{2}} e^{2i\sqrt{2}n\pi\sqrt{\frac{\lambda}{k}}}}{2^{\frac{j}{2}} n^{j} \pi^{j} \Delta^{\frac{j}{2}}} \Biggr] \\ &= \zeta(c-1) \Gamma\left(\frac{c-1}{2}\right) \pi^{\frac{1-c}{2}} \sum_{k} \alpha_{k} \left(\frac{c}{2} k^{\frac{c}{2}-1} + \left(\frac{c}{2}-1\right) k^{-\frac{c}{2}}\right), \qquad c \text{ odd, } c \ge 5 \end{split}$$
(D.11)

The sum over n in (D.11) can be done exactly, which gives:

$$\begin{split} \sum_{\Delta \in \mathcal{S}} \sum_{k} \alpha_{k} \Biggl[\frac{1}{2^{\frac{3c}{2} - 3} \pi^{\frac{c-3}{2}} \Delta^{\frac{c-2}{2}} (-1 + e^{2\sqrt{2}\pi\sqrt{k\Delta}})^{\frac{c+1}{2}}} \sum_{n=0}^{2} \sum_{m=0}^{2} \sum_{m=0}^{c-1} v_{1}(c, n, m) e^{2\sqrt{2}\pi\sqrt{k\Delta}m} (\sqrt{2k\Delta}\pi)^{n} \\ &+ \frac{e^{2\sqrt{2}\pi\sqrt{\frac{\Delta}{k}}}}{2^{\frac{3c}{2} - 7} \pi^{\frac{c-7}{2}} k^{2} \Delta^{\frac{c-4}{2}} (-1 + e^{2\sqrt{2}\pi\sqrt{\frac{\Delta}{k}}})^{\frac{c+1}{2}}} \sum_{n=0}^{2} \sum_{m=0}^{2} \sum_{m=0}^{2} v_{3}(c, n, m) e^{2\sqrt{2}\pi\sqrt{\frac{\Delta}{k}}m} (\sqrt{\frac{2\Delta}{k}}\pi)^{n} \Biggr] \\ &= \zeta(c-1)\Gamma\left(\frac{c-1}{2}\right) \pi^{\frac{1-c}{2}} \sum_{k} \alpha_{k} \left(\frac{c}{2}k^{\frac{c}{2}-1} + \left(\frac{c}{2}-1\right)k^{-\frac{c}{2}}\right), \qquad c \text{ odd, } c \ge 5. \end{split}$$

$$(D.12)$$

In the notation of (5.53),

$$\operatorname{vac}(k) = -\zeta(c-1)\Gamma\left(\frac{c-1}{2}\right)\pi^{\frac{1-c}{2}}\left(\frac{c}{2}k^{\frac{c}{2}-1} + \left(\frac{c}{2}-1\right)k^{-\frac{c}{2}}\right), \quad (D.13)$$

and $f(k, \Delta)$ is the term in the brackets of (D.12). Using these definitions, an explicit calculation verifies the claim in (5.54).

By examining the crossing equation (D.12), we notice something interesting. Acting on the crossing equation with $k^{3/2}\partial_k$ gives us an expression that is antisymmetric under $k \leftrightarrow k^{-1}$. This gives us another way to rewrite the crossing equation that will turn out to work for all *c* (not just odd *c*). Let us define the following function, using (D.6)

$$\begin{split} h(c,k,\Delta) &\coloneqq \sum_{m=1}^{\infty} k^{3/2} \partial_k \left(4\pi \Delta f_m^k(-c,2\pi\Delta) - c f_m^k(2-c,2\pi\Delta) \right) \\ &= \sum_{m=1}^{\infty} \left[2^{-\frac{c-4}{4}} k^{\frac{c+2}{4}} m^{\frac{c}{2}} \Delta^{-\frac{c-4}{4}} \pi c K_{\frac{c}{2}}(2\sqrt{2}m\pi\sqrt{k\Delta}) \right. \\ &\left. - 2^{-\frac{c+2}{4}} k^{\frac{c}{4}} m^{\frac{c-2}{2}} \Delta^{-\frac{c-2}{4}} (c(c-2) + 8\pi^2 \Delta k m^2) K_{\frac{c-2}{2}}(2\sqrt{2}m\pi\sqrt{k\Delta}) \right]. \end{split}$$
(D.14)

The sum can be evaluated exactly in closed form for odd c, but exists and converges for any c. An equivalent formulation of our scalar crossing equation is:

$$k^{3/2} \operatorname{vac}'(k) + \sum_{\Delta \in \mathcal{S}} h(c, k, \Delta) - h(c, k^{-1}, \Delta) = 0.$$
 (D.15)

The sum rules used in (5.55) are just the odd derivatives of k (evaluated at k = 1) of (D.15). Finally, note that the term $k^{3/2}$ vac'(k) is simply the contribution of the vacuum state:

$$k^{3/2} \operatorname{vac}'(k) = \frac{\Lambda\left(\frac{c-1}{2}\right)c(c-2)}{4} \left(k^{\frac{1-c}{2}} - k^{\frac{c-1}{2}}\right)$$
$$= \lim_{\Delta \to 0} \left(h(c, k, \Delta) - h(c, k^{-1}, \Delta)\right).$$
(D.16)

D.3 c = 1 and c = 2 revisited

In this appendix, we reconsider $U(1)^c$ theories at c = 1 and c = 2. Due to the pole structure of the function $\Lambda(s) := \pi^{-s}\Gamma(s)\zeta(2s)$, the spectral decomposition and scalar crossing equation for these theories are slightly different than for c > 2. This is related to the fact that the average genus 1 partition function for c = 1 and c = 2Narain CFTs diverges [166, 167]. For both c = 1 and c = 2 we will first consider Narain CFTs, and then the potentially more general $U(1)^c$ theories. We will use the notation

$$\tilde{E}_{s}(\tau) \coloneqq \Lambda(s)E_{s}(\tau)$$

= $\Lambda(s)y^{s} + \Lambda(1-s)y^{1-s} + \sum_{j=1}^{\infty} \frac{4\sigma_{2s-1}(j)\sqrt{y}K_{s-\frac{1}{2}}(2\pi jy)}{j^{s-\frac{1}{2}}}\cos(2\pi jx)$
(D.17)

which we can see obeys $\tilde{E}_s(\tau) = \tilde{E}_{1-s}(\tau)$. This will make $s \leftrightarrow 1 - s$ crossing manifestly invariant.

D.3.1 c = 1 reconsidered

The c = 1 free boson is labeled by a radius r. In our convention, we will take the self-dual point (i.e. the $SU(2)_1$ WZW model) to be r = 1 so that T-duality acts as $r \leftrightarrow r^{-1}$. The spectral decomposition of the reduced c = 1 partition function is:

$$\widehat{Z}^{c=1}(\tau,r) = r + r^{-1} + \frac{1}{4\pi i} \int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} ds 2\widetilde{E}_s(\tau)(r^{2s-1} + r^{1-2s}).$$
(D.18)

(See e.g. Sec. 3.1.1 of [165] for derivation.) Notice that there are no Maass cusp forms in (D.18).

At c = 1, our scalar crossing equation (5.38) reduces to

$$1 + \sum_{\Delta \in \mathcal{S}} e^{-2\pi\Delta y} = -1 + \varepsilon_{c=1}(\mu) y^{-\frac{1}{2}} + \sum_{k=1}^{\infty} \operatorname{Re}\left(\delta_{k,c=1}(\mu) y^{\frac{z_k}{2}}\right) + \sum_{\Delta \in \mathcal{S}} \sum_{n=1}^{\infty} b(n) \sqrt{\frac{2\Delta}{y}} e^{-\frac{2\pi n^2 \Delta}{y}}$$
(D.19)

where as usual μ is some abstract coordinate that we include to emphasize which terms are theory-dependent.

Let us verify (D.19) for a free boson at radius r. From the explicit spectral decomposition (D.18), we know that the free boson at radius r has $\varepsilon_{c=1}(\mu) = r + r^{-1}$ and $\delta_{k,c=1}(\mu) = 0$. Moreover, the set of scalar operators S are simply operators with either zero momentum or zero winding number (recall at c = 1, the spin of an operator is just the product of its momentum and winding number). Thus the set Sis simply operators of dimension $\frac{m^2}{2r^2}$ and $\frac{m^2r^2}{2}$ for $m \in \mathbb{Z}_{>0}$, each with degeneracy 2. Thus (D.19) reduces to

$$2y^{\frac{1}{2}} + 2\sum_{m=1}^{\infty} (e^{-\pi m^2 r^2 y} + e^{-\pi m^2 r^2 y})y^{\frac{1}{2}} = r + r^{-1} + 2\sum_{n=1}^{\infty} \sum_{m=1}^{\infty} b(n)m(re^{-\frac{\pi n^2 m^2 r^2}{y}} + r^{-1}e^{-\frac{\pi n^2 m^2}{yr^2}}).$$
(D.20)

We can rewrite the RHS with new variables m' = nm, n' = n (and dropping primes)

$$2y^{\frac{1}{2}} + 2\sum_{m=1}^{\infty} (e^{-\pi m^2 r^2 y} + e^{-\pi m^2 r^{-2} y})y^{\frac{1}{2}} = r + r^{-1} + 2\sum_{m=1}^{\infty} \sum_{n|m} b(n)\frac{m}{n}(re^{-\frac{\pi m^2 r^2}{y}} + r^{-1}e^{-\frac{\pi m^2}{yr^2}}).$$
(D.21)

It can be shown from properties of the M'obius μ function that

$$\sum_{n|m} b(n)\frac{m}{n} = 1$$
(D.22)

for all m. Our crossing equation is then equivalent to

$$2y^{\frac{1}{2}} + 2\sum_{m=1}^{\infty} (e^{-\pi m^2 r^2 y} + e^{-\pi m^2 r^{-2} y})y^{\frac{1}{2}} = r + r^{-1} + 2\sum_{m=1}^{\infty} (re^{-\frac{\pi m^2 r^2}{y}} + r^{-1}e^{-\frac{\pi m^2}{yr^2}}).$$
(D.23)

This simply follows from the modular transformation properties of the Jacobi theta functions.

We would now like to derive a more general bound for $U(1)^c$ CFTs at c = 1, without assuming the theory is a free boson compactified on a circle. This means we cannot assume that the $\delta_{k,c=1}$ terms in (D.19) necessarily vanish, so we need to apply the same functionals that we considered in Sec. 5.3.3. We first take a derivative with respect to y to remove the $\varepsilon_{c=1}(\mu)$ term. This gives the analog of (5.41):

$$\sum_{\Delta \in \mathcal{S}} \left[\left(\frac{4\pi\Delta}{t} - t \right) e^{-\frac{2\pi\Delta}{t^2}} + \sum_{n=1}^{\infty} b(n) 4\pi \sqrt{2}n^2 t^4 \Delta^{\frac{3}{2}} e^{-2\pi\Delta n^2 t^2} \right]$$
$$= 2t + \sum_{k=1}^{\infty} \operatorname{Re} \left(\delta_{k,c=1}(\mu) (z_k - 2) t^{z_k} \right).$$
(D.24)

We next would like to apply the functional (5.45) to (D.24), but there a slight subtlety. Recall that (5.45) was designed so that

$$\mathcal{F}[t^s] \propto \zeta(s). \tag{D.25}$$

The last line of (D.24) has a term 2*t*, which will naively give something proportional to $\zeta(1)$ which diverges. However, it can be shown the integral (5.45) converges. The reason is that $M_{\varphi}(s)$ in (5.48) vanishes at s = 1 which cancels the divergence of the zeta function. A careful analysis shows that if we choose $\varphi(t) = \sum_{i=1}^{N} \alpha_i e^{-\pi k_i t^2}$ (subject to the constraints (5.50)), then

$$\mathcal{F}^{\varphi}[2t] = \sum_{i=1}^{N} \alpha_i \left(-\frac{\log k_i}{2\sqrt{k_i}} \right). \tag{D.26}$$

We then apply the same functional \mathcal{F}^{φ} to the LHS of (D.24). This gives

$$\sum_{k} \alpha_{k} \left(\frac{\log k}{2\sqrt{k}} + \sum_{\Delta \in \mathcal{S}} \left[\frac{\pi\sqrt{\Delta}(\coth(\sqrt{2}\pi\sqrt{k\Delta}) - 1)}{\sqrt{2}} + \frac{\log(1 - e^{-2\sqrt{2}\pi\sqrt{k\Delta}})}{2\sqrt{k}} + \sum_{n=1}^{\infty} b(n) \frac{\sqrt{k}\coth\left(\sqrt{2}n\pi\sqrt{\frac{\Delta}{k}}\right) + \sqrt{2}n\pi\sqrt{\Delta}\operatorname{csch}^{2}\left(\sqrt{2}n\pi\sqrt{\frac{\Delta}{k}}\right)}{4kn} \right] \right) = 0.$$
(D.27)

The sum over *n* in (D.27) formally diverges but we can replace $\operatorname{coth}\left(\sqrt{2}n\pi\sqrt{\frac{\Delta}{k}}\right)$ with $\operatorname{coth}\left(\sqrt{2}n\pi\sqrt{\frac{\Delta}{k}}\right) - 1$ since the term we add is multiplied by 0 from (5.50). This gives the following convergent sum rule:

$$\begin{split} \sum_{k} &\alpha_{k} \left(\frac{\log k}{2\sqrt{k}} + \sum_{\Delta \in \mathcal{S}} \left[\frac{\pi \sqrt{\Delta} (\coth(\sqrt{2}\pi \sqrt{k\Delta}) - 1)}{\sqrt{2}} + \frac{\log(1 - e^{-2\sqrt{2}\pi \sqrt{k\Delta}})}{2\sqrt{k}} \right. \\ &+ \sum_{n=1}^{\infty} b(n) \frac{\sqrt{k} \left(\coth\left(\sqrt{2}n\pi \sqrt{\frac{\Delta}{k}}\right) - 1 \right) + \sqrt{2}n\pi \sqrt{\Delta} \operatorname{csch}^{2} \left(\sqrt{2}n\pi \sqrt{\frac{\Delta}{k}} \right)}{4kn} \right] \right) = 0. \end{split}$$

$$(D.28)$$

The sum over *n* can be done exactly to give:

$$\sum_{k} \alpha_{k} \left(\frac{\log k}{2\sqrt{k}} + \sum_{\Delta \in \mathcal{S}} \left[\frac{\pi \sqrt{\Delta} (\coth(\sqrt{2}\pi \sqrt{k\Delta}) - 1)}{\sqrt{2}} + \frac{\log(1 - e^{-2\sqrt{2}\pi \sqrt{k\Delta}})}{2\sqrt{k}} + \frac{\pi \sqrt{\Delta} (\coth(\sqrt{2}\pi \sqrt{\frac{\Delta}{k}}) - 1)}{\sqrt{2}k} - \frac{\log(1 - e^{-2\sqrt{2}\pi \sqrt{\frac{\Delta}{k}}})}{2\sqrt{k}} \right] \right) = 0.$$
 (D.29)

Again from the same arguments as used to derive (5.53) we know that the term in parenthesis in (D.29) must be $c_0 + c_1 k^{-1/2}$ for some (theory-dependent) constants c_0, c_1 . Moreover we see after evaluating $\partial_k|_{k=1}$ on each term, that $c_1 = -1$. Therefore we can write our crossing equation as

$$\operatorname{vac}^{(n)}(1) + \sum_{\Delta \in \mathcal{S}} (\partial_k)^n f(k, \Delta)|_{k=1} = 0, \quad n \ge 2, \quad n \text{ even},$$
(D.30)

with

$$\operatorname{vac}(k) = \frac{2 + \log k}{2\sqrt{k}}$$

$$f(k, \Delta) = \frac{\pi\sqrt{\Delta}(\operatorname{coth}(\sqrt{2}\pi\sqrt{k\Delta}) - 1)}{\sqrt{2}} + \frac{\log(1 - e^{-2\sqrt{2}\pi\sqrt{k\Delta}})}{2\sqrt{k}}$$

$$+ \frac{\pi\sqrt{\Delta}(\operatorname{coth}(\sqrt{2}\pi\sqrt{\frac{\Delta}{k}}) - 1)}{\sqrt{2}k} - \frac{\log(1 - e^{-2\sqrt{2}\pi\sqrt{\frac{\Delta}{k}}})}{2\sqrt{k}}.$$
(D.31)

Note that the equations (D.30) are indeed equivalent to derivatives (with respect to k, evaluated at k = 1) of (D.15) at c = 1.

D.3.2 c = 2 reconsidered

The c = 2 free boson is labeled by a metric and *B* field, which gives four real moduli in total. These can be repackaged into two elements of the upper half plane as [188]:

$$\rho = B + i\sqrt{\det G}, \quad \sigma = \frac{G_{12}}{G_{11}} + i\frac{\sqrt{\det G}}{G_{11}}.$$
(D.32)

T-duality acts as two independent elements of $SL(2, \mathbb{Z})$ acting on ρ and σ in the usual way. In terms of these coordinates, the spectral decomposition of the reduced c = 2 partition function is:

$$\widehat{Z}^{c=2}(\tau,\rho,\sigma) = \widehat{E_{1}}(\tau) + \widehat{E_{1}}(\rho) + \widehat{E_{1}}(\sigma) - \frac{3}{\pi} (4 - \gamma_{E} - 3\log(4\pi) - 48\zeta'(-1)) + \frac{1}{4\pi i} \int_{\frac{1}{2} - i\infty}^{\frac{1}{2} + i\infty} ds \frac{\widetilde{E}_{s}(\tau)\widetilde{E}_{s}(\rho)\widetilde{E}_{s}(\sigma)}{\Lambda(s)\Lambda(1-s)} + 8\sum_{n=1}^{\infty} \frac{\nu_{n}^{+}(\tau)\nu_{n}^{+}(\rho)\nu_{n}^{+}(\sigma)}{(\nu_{n}^{+},\nu_{n}^{+})} - 8i\sum_{n=1}^{\infty} \frac{\nu_{n}^{-}(\tau)\nu_{n}^{-}(\rho)\nu_{n}^{-}(\sigma)}{(\nu_{n}^{-},\nu_{n}^{-})}.$$
 (D.33)

(See e.g. Sec. 3.1.2 of [165] for derivation.) In (D.33), the function $\widehat{E_1}$ is defined as

$$\widehat{E_1}(\tau) \coloneqq \lim_{s \to 1} E_s(\tau) - \frac{3/\pi}{s-1}$$

= $y - \frac{3}{\pi} \log y + \frac{6}{\pi} (1 - 12\zeta'(-1) - \log 4\pi) + \sum_{j=1}^{\infty} \frac{12\sigma_1(j)e^{-2\pi jy}\cos(2\pi jx)}{j}.$
(D.34)

Let us derive the scalar crossing equation at c = 2. We first assume the theory is a Narain CFT. As usual let us denote the set of scalar operators under the $U(1)^2$ chiral algebra excluding the vacuum, as S. (Of course, S depends on the moduli of the theory, which for c = 2 we denote by ρ, σ , but we will suppress that.) The partition function of these scalars is given by

$$y\left(1+\sum_{\Delta\in\mathcal{S}}e^{-2\pi\Delta y}\right) \coloneqq \int_{-1/2}^{1/2} dx \widehat{Z}^{c=2}(\tau,\rho,\sigma)$$

= $y - \frac{3}{\pi}\log y + \widehat{E_1}(\rho) + \widehat{E_1}(\sigma) + \frac{3}{\pi}\left(-2 + 24\zeta'(-1) + \gamma_E + \log 4\pi\right)$
+ $\frac{1}{4\pi i} \int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} ds \frac{(\Lambda(s)y^s + \Lambda(1-s)y^{1-s})\widetilde{E_s}(\rho)\widetilde{E_s}(\sigma)}{\Lambda(s)\Lambda(1-s)}$
= $y - \frac{3}{\pi}\log y + \widehat{E_1}(\rho) + \widehat{E_1}(\sigma) + \frac{3}{\pi}\left(-2 + 24\zeta'(-1) + \gamma_E + \log 4\pi\right)$
+ $\frac{1}{2\pi i} \int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} ds \frac{y^s \widetilde{E_s}(\rho)\widetilde{E_s}(\sigma)}{\Lambda(1-s)}.$ (D.35)

Let us move the contour in *s* to the right past all the poles. The function \tilde{E}_s has simple poles at s = 0, 1 (which can be seen from (D.17)). Moreover, $\Lambda(1 - s) = \pi^{\frac{1}{2}-s}\Gamma(s-\frac{1}{2})\zeta(2s-1)$ has zeros whenever 2s-1 is a nontrivial zero of the Riemann zeta function. Thus the integrand has simple poles that we cross at $s = 1, \frac{1+z_n}{2}, \frac{1+z_n^*}{2}$, where z_n is a nontrivial zero of the Riemann zeta function (with positive imaginary part).¹ A picture of the pole structure is given in Fig. 5.1 (where we move the pole at $s = \frac{c}{2}$ to s = 1).

We then get the equation:

$$y\left(1 + \sum_{\Delta \in S} e^{-2\pi\Delta y}\right) = -\frac{3}{\pi} \log y + \frac{3}{\pi} \left(-2 + 24\zeta'(-1) + \gamma_E + \log 4\pi\right) + \widehat{E_1}(\rho) + \widehat{E_1}(\sigma) + \sum_{k=1}^{\infty} \operatorname{Re}\left(\frac{4\pi^{\frac{z_k}{2}}\Lambda(\frac{1+z_k}{2})^2 E_{\frac{1+z_k}{2}}(\rho) E_{\frac{1+z_k}{2}}(\sigma)}{2\Gamma(\frac{z_k}{2})\zeta'(z_k)}y^{\frac{1+z_k}{2}}\right) + \frac{1}{2\pi i} \int_{\gamma-i\infty}^{\gamma+i\infty} ds \frac{y^s \tilde{E}_s(\rho) \tilde{E}_s(\sigma)}{\Lambda(1-s)},$$
(D.36)

where $\gamma > \frac{c}{2} = 1$. This integral is a special case of the one studied in (5.37), which can be done exactly to give us:

$$1 + \sum_{\Delta \in \mathcal{S}} e^{-2\pi\Delta y} = -\frac{3}{\pi} \frac{\log y}{y} + \frac{\frac{3}{\pi} (-2 + 24\zeta'(-1) + \gamma_E + \log 4\pi) + \widehat{E_1}(\rho) + \widehat{E_1}(\sigma)}{y} + \sum_{k=1}^{\infty} \operatorname{Re}\left(\frac{4\pi^{\frac{z_k}{2}} \Lambda(\frac{1+z_k}{2})^2 E_{\frac{1+z_k}{2}}(\rho) E_{\frac{1+z_k}{2}}(\sigma)}{2\Gamma(\frac{z_k}{2})\zeta'(z_k)} y^{\frac{-1+z_k}{2}}\right) + \frac{1}{y\sqrt{\pi}} \sum_{\Delta \in \mathcal{S}} \sum_{n=1}^{\infty} b(n) U\left(-\frac{1}{2}, 1, \frac{2\pi\Delta n^2}{y}\right) e^{-\frac{2\pi\Delta n^2}{y}}.$$
 (D.37)

The sum over k in (D.37) falls off exponentially in k so the sum is indeed convergent.

The generalization to any $U(1)^2$ CFT at c = 2 is straightforward. We again need to subtract $\widehat{E_1}(\tau)$ to render the reduced partition function square-integrable, and a gap to the first excited state constrains the poles we cross in *s* to only be at $s = 1, \frac{1+z_n}{2}, \frac{1+z_n^*}{2}$ (see Appendix D.1). Finally, the same arguments as in Sec 5.3.2 let us compute the

¹The double pole at s = 1 in the numerator of the integrand becomes a simple pole when canceled by the simple pole at s = 1 in the denominator. There is also a pole at s = 0, but since we move the contour to the right we can ignore it.

non-perturbative corrections at high temperature to get:

$$1 + \sum_{\Delta \in \mathcal{S}} e^{-2\pi\Delta y} = -\frac{3}{\pi} \frac{\log y}{y} + \frac{\varepsilon_{c=2}(\mu)}{y} + \sum_{k=1}^{\infty} \operatorname{Re}\left(\delta_{k,c=2} \ y^{\frac{-1+z_k}{2}}\right) + \frac{1}{y\sqrt{\pi}} \sum_{\Delta \in \mathcal{S}} \sum_{n=1}^{\infty} b(n)U\left(-\frac{1}{2}, 1, \frac{2\pi\Delta n^2}{y}\right) e^{-\frac{2\pi\Delta n^2}{y}}.$$
 (D.38)